# EHT – 10/EHT – 10/2 System Development Guide

**OS Version 2.0** 

=

**EPSON** 

Y24399100801 M013B

### REMARKS

- All rights reserved.Reproduction of any part of this manual in any form whatsoever without SEIKO EPSON's expressed written permission is forbidden.
- (2) The contents of this manual are subject to change without notice.
- (3) All efforts have been made to ensure the accuracy of the contents of this manual. However, should any errors be detected, SEIKO EPSON would greatly appreciate being informed of them.
- (4) The above notwithstanding, SEIKO EPSON can assume no responsibility for any errors in this manual or their consequences.

 $CP/M^{TM}$  is a registered trademark of Digital Research, Inc.  $MS-DOS^{TM}$  is a registered trademark of Microsoft Corp.

(C) Copyright 1986 by SEIKO EPSON CORPORATION Nagano, Japan

## CONTENTS

INTRODUC	TION	0 - 1
PART	1 SOFTWARE	
1. SYSTE	M OVERVIEW	1 - 1
1.1	Characteristics of System 1.1.1 EHT-10/EHT-10/2 concepts 1.1.2 Characteristics of the System	1 - 1 1 - 1 1 - 1
1.2	System configuration	1 - 4
1.3	Software structure 1.3.1 Overview 1.3.2 Structure	1 - 5 1 - 5 1 - 5
1.4	Memory map         Memory space           1.4.1 Memory space	1 - 8 1 - 8 1 - 1
2. BASIC	SYSTEM OPERATIONS	2 - 1
2.1	Overview	2 - 1
2.2	System status change 2.2.1 Control flow 2.2.2 Status change	2 - 1 2 - 1 2 - 2
2.3	System initiation and termination 2.3.1 System initialization 2.3.2 Reset	2 - 4 2 - 4 2 - 7 2 - 7 2 - 8 2 - 8 2 - 1
2.4	Alarm/Wake 2.4.1 Overview 2.4.2 Alarm 2.4.3 Wake	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$
2.5	Self-Test2.5.1 Overview2.5.2 System area sum check2.5.3 User area sum check2.5.4 RAM disk sum check	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$
2.6	CP/M Operations 2.6.1 Overview 2.6.2 CP/M structure 2.6.3 CP/M Initiation and termination 2.6.4 CCP 2.6.5 I/O byte 2.6.6 Setting system parameters	2 - 2 2 - 2

_	511	2013
3.	Application program creation	3 - 1
	J.I UVErview	3 - 1
	3.2 Notes on designing application systems	2
	3.2.1 Programming language coloction	3 - 2
	3.2.2 Program distribution and maintenance	3 - 2
	Siziz riogram discribution and maintenance	3 - 2
	3.3 Setting development environment	2 – 3
	3.3.1 Computer used to create application programs	2 - 2
	3.3.2 Development software and development procedure	3 - 3
	3.3.3 Storing program in ROM	3 - 4
		5 .
	3.4 Creating ROM-based Programs	3 - 5
	3.4.1 Overview	3 - 5
	3.4.2 Setting	3 - 5
	3.4.3 Processing flow	3 - 5
	3.4.4 Program start address	3 - 6
	3.4.5 Lalling BDUS/BIUS	3 - 7
	3.4.0 KUM storing RUM-based programs	3 - 8
	3 A 9 32 White on more DOM based presso	3 - 8
	5.4.6 32 kuyte or more kum-based program	3 - 8
4.	BIOS Overview	4 - 1
	4.1 Overview	4 - 1
	4.1.1 Characteristics of EHT-10/EHT-10/2 BIOS	4 - 1
	4.1.2 BIOS processing	4 - 1
	4.2 Uverview of BIUS Commands	4 – 4
	4.3 Entering data with touch name) or keyboard keys	4 - 128
	4.3.1 Overview	4 - 128
	4.3.2 Functions supported by BIOS	4 - 129
	4.3.3 Position and function codes	4 - 129
	4.3.4 Key buffer	4 - 133
	4.3.5 Key input beep	4 - 135
	4.3.6 Touch panel (EHT-10)	4 - 135
	4.3.7 Keyboard (EHT-10/2)	4 - 141
	4.4 Uisplaying characters on LCDs	4 - 146
	4.4.1 UVERVIEW	4 - 140
	4.4.2 Functions supported by Blus	4 - 140
	A A A Display specifications in EHT-10/2	4 - 140
	4.4.5 Character opperator	4 - 172
	4.4.6 Displaying cursor	4 - 176
	4.4.7 Details on work areas for displaying	4 - 178
	4.5 Printer	4 - 183
	4.5.1 Overview	4 - 183
	4.5.2 Functions supported by BlOS	4 - 183
	4.5.3 Character set adjustment to a specific country	4 - 183
	4.5.4 Printer unit	4 - 184
	4.5.5 Uutputting data to RS-232C interface	4 - 193

	4.6 User BIOS 4.6.1 Overview 4.6.2 User BIOS 4.6.3 User BIOS area	4 - 195 4 - 195 4 - 195 4 - 195 4 - 197
5.	BDOS processing	5 - 1
	5.1 Overview	5 - 1
	5.2 BDOS processing flow	5 - 2
	5.3 BDOS error	5 - 3
6.	Disk system	6 - 1
	6.1 Overview	6 - 1
	6.2 Logical and physical drives	6 - 2
	6.3 RAM disk	6 - 4
	6.3.1 Overview	6 - 4
	6.3.2 Drive name and capacity	6 - 4
	6.3.3 Format	6 - 5
	6.3.4 Miscellaneous	6 - 7
	6 4 DOM cocket	6 - 8
	6 A 1 Overview	6 - A
	6 4 2 Drive name and canacity	6 - 8
	6 4.3 Format	6 - 9
	6.4.4 Executing ROM socket programs	6 - 13
	6.4.5 Miscellaneous	6 - 13
	6.5 IC card	6 - 14
	6.5.1 Overview	6 - 14
	6.5.2 Drive name and capacity	6 - 14
	6.5.3 Format	6 - 15
	6.5.4 Protocol	6 - 16
	6.5.5 Disk mode interface	0 - 2/
	b.5.6 Lommand through mode interface	0 - 20
	6.5.9 Miccollopour	0 - 20 6 - 30
	0.3.8 MISCELLANGONS	0 00
	6.6 Floppy Disks	6 - 33
	6.6.1 Overview	6 - 33
	6.6.2 Drive name and capacity	6 - 33
	6.6.3 Format	6 - 33
	6.6.4 Protocol	6 - 34
	6.6.5 Miscellaneous	b - 42
	6.7 Details on Disk-related work areas	6 - 43
7.	I/O interface overview	7 - 1
	7.1 Overview	7 - 1

	7.2	Serial interface 7.2.1 Overview 7.2.2 Setting a serial device 7.2.3 Serial ports supported by OS 7.2.4 Serial communication by the user	7 7 7 7 7		3 3 3 4 6
	7.3	Cartridge interface 7.3.1 Overview 7.3.2 Modes and how to set a mode 7.3.3 Determining the mode 7.3.4 I/O registers used for cartridge I/F	7 7 7 7 7		9 9 9 10 12
	7.4	IC card interface 7.4.1 Overview 7.4.2 I/O registers used for IC card I/F 7.4.3 Controlling IC card reader power supply	7 7 7 7		15 15 15 16
	7.5	Barcode reader interface 7.5.1 Overview 7.5.2 I/O registers for barcode I/F 7.5.3 Barcode reader power supply control	7 7 7 7		18 18 18 19
	7.6	7508 commands 7.6.1 7508 functions 7.6.2 7508 interface 7.6.3 7508 commands	7 7 7 7	-	21 21 21 25
	7.7	Other I/O 7.7.1 Overview 7.7.2 Buzzer 7.7.3 Timer 7.7.4 EL-backlight	7 7 7 7 7		52 52 52 52 53
8.	Interi	rupt processing	8	-	1
	8.1	Overview	8	-	1
	8.2	Interrupt vector	8	-	3
	8.3	Controlling Interrupts 8.3.1 Controlling interrupts by the system 8.3.2 Disabling and enabling interrupts 8.3.3 Interrupt processing time 8.3.4 Processing executed when an interrupt occurs	8 8 8 8	-	4 4 7 7
	8.4	7508 interrupts 8.4.1 Overview 8.4.2 Multiple interrupt processing 8.4.3 7508 interrupts 8.4.4 Suppressing alarm or power off (failure) interrupt	8 8 8 8	-	9 9 10 15
	8.5	ART interrupts	8	-	20
	8.6	ICF interrupts	8	_	21
	8.7	OVF interrupts	8	-	22

8.8 EXT interrupts 8.8.1 Overview 8.8.2 1-msec/8-msec timer interrupt 8.8.3 Interrupt sent from cartridge 8.8.4 Interrupt sent from IC card	8 8 8 8		23 23 23 24 25
8.9 Extending interrupt processing 8.9.1 Overview 8.9.2 Extension by using interrupt hook 8.9.3 Extension by rewriting interrupt vector 8.9.4 Checking the interrupt occurrence status	8 8 8 8		26 26 26 26 26
9. Utilizing system functions	9	-	1
9.1 Overview	9	-	1
9.2 System menu 9.2.1 Controlling system menu 9.2.2 Setting CONFIG parameters 9.2.3 Specifying DL/UL drives	9 9 9	-	2 2 2 6
9.3 Extending MENU processing 9.3 1 Changing/adding display drives 9.3.2 Modifying/adding file types	9 9 9		7 7 7
9.4 Extending Power-on processing	9	-	9
10. System hooks and jump tables	10	-	1
10.1 Overview	10	-	1
10.2 System hooks 10.2.1 Overview 10.2.2 Structure and usage procedure of each hook table 10.2.3 System hooks registered in system hook table (I) 10.2.4 System hooks registered in system hook table (II)	10 10 10 10		2 2 2 7 21
<pre>10.3 System jump tables 10.3.1 Overview 10.3.2 Jump table configuration 10.3.3 Resident area jump table 10.3.4 OS ROM jump table (I) 10.3.5 OS ROM jump table (II)</pre>	10 10 10 10 10 10	-	22 22 25 25 49 69
11. System expansion	11	-	1
11.1 Overview	11	-	1
11.2 Communication protocol expansion 11.2.1 Overview 11.2.2 Expansion procedure 11.2.3 Description of hook 11.2.4 Communication relation work area detail	11 11 11 11 11		2 2 2 3 19

.

11.3	IC card protocol expansion 11.3.1 Overview 11.3.2 Support at OS 11.3.3 Protocol expansion 11.3.4 Mount process hook (ICDHK1,ICDHK7) 11.3.5 Hook at 1 record read (ICDHK5) 11.3.6 Hook at 1 record write (ICDHK6) 11.3.7 Hook at format (ICDHK8)	11 11 11 11 11 11 11 11		22 22 32 35 36 36 37 37
	11.3.10 Work area detail related to IC card	ii	_	30
11.4	Cartridge Device expansion 11.4.1 Overview 11.4.2 Cartridge Mount process 11.4.3 Interrupt process from cartridge device	11 11 11 11		47 47 47 50
11.5	Addition of bar code decoder 11.5.1 Overview 11.5.2 Hand scanner 11.5.3 Work area details for bar code decoder	11 11 11 11	-	53 53 55 56

# PART 2. HARDWARE

1	Hardware overview	12	-	1
	1.1 Hardware structure1.1.1 Overview1.1.2 Hardware structure	12 12 12		1 1 3
	1.2 Address map1.2.1 Bank switching1.2.2 Notes on bank switching	12 12 12	-	7 7 13
2.	I/O register description	13	-	1
	2.1 I/O address table	13	-	1
	2.2 I/O register details	13	-	4
	2.3 Notes	13	-	42
3.	Interrupt description	14	-	1
	3.1 Overview	14	-	1
	3.2 Interrupt vector	14	-	2
	3.3 7508 interrupt	14	-	3
	3.4 ART interrupt	14	-	4
	3.5 OVF interrupt, ICF interrupt	14	-	5
	3.6 EXT interrupt	14	-	6

4	. Interface description	15	-	1	
	4.1 Cartridge interface	15 15 15 15 15 15 15 15 15 15		1 1 2 3 4 6 6 6 6 1 9 20 22 25	
	4.2 Bar code reader interface4.2.1 Connector in use4.2.2 Terminal and function4.2.3 I/O register4.2.4 Power characteristic	15 15 15 15		26 26 27 27 28	
	4.3 RS-232C interface	15 15 15 15 15 15		29 29 30 31 31 32 33	
	4.4 IC card interface 4.4.1 Connector in use 4.4.2 Terminal and function 4.4.3 IC card interface block 4.4.4 Power characteristics	15 15 15 15 15	-	34 34 34 35 36	
	4.5 Touch panel interface (EHT-10)         4.6 LCD control (EHT-10)         4.6.1 Hardware configuration         4.6.2 VRAM address map         4.6.3 I/O port         4.6.4 T6963 command         4.6.5 Sample program	15 15 15 15 15 15		37 39 39 40 41 43 46	
	4.7 Buzzer	15	-	48	
	4.8 Counter 4.8.1 1 second counter 4.8.2 1/10 second counter	15 15 15		49 49 49	
	4.9 Dip switch	15	-	50	
	4.10 LED (EHT-10/2)	15	-	51	

5.	Power	supply	16	-	1
	5.1	Overview	16	-	1
	5.2	Charging current to battery by adaptor	16	-	2
	5.3	battery capacity	16	_	2
	5.4	Charger and charging time	16	_	2
6.	Notes	on circuit design	17	-	1

# PART 3. BASIC

1.	BASIC	memory management	18	-	1
	1.1	Overview	18	-	1
	1.2	Memory map in RAM	18	-	2
	1.3	Options at BASIC start	18	-	3
	1.4	Disk buffer	18	-	4
	1.5	Text (program)	18	-	6
	1.6	Simple variables	18	-	7
	1.7	Array variables	18	-	9
	1.8	Variables in BASIC work area	18	-	10
2.	Comman	nd expansion	19	-	1
	2.1	Machine language hold	19	-	1
	2.2	Hook at BASIC start	19	-	2
	2.3	Reserved word EXTD	19	-	3
	2.4	Syntax analysis routine	19	-	4
	2.5	Reserved words 2.5.1 Internal code (statement) 2.5.2 Internal code (function)	19 19 19		9 10 11
3.	Sequer	ntial access device expansion	20	-	1
	3.1	Overview	20	-	1
	3.2	Device table	20	-	2
	3.3	DCB table	20	-	3
	3.4	DCB (Device control block)	20	-	4

## PART 4. APPENDIX

1.0	Character code table (1) Character generator code table (2) Character code table (2) Character code table	21 21 21		1 1 2
2.	Font table (1) EHT-10/EHT-10/2 mainframe font table (2) Printer unit font table	21 21 21	-	4 5 9
3.	Version (1) CP/M serial No (2) ROMID	21 21 21		13 13 14
4.	Table of system work area	21 21 21 21 21 21 21 21		15 16 22 29 34 47 49
5.	BJOS function list	21	-	55
6.	Display control functions (1) List of display control functions (2) Details on display control functions	21 21 21	- - -	60 60 64
7.	BDOS functions list	21	_	77
8.	Filink protocol	21	-	80
9.	Sample program lists         (1) BIOS CALL         (2) CONIN CONOUT         (3) PUNCH READER LIST         (4) DISK Access         (5) BEEP         (6) TIMDAT         (7) RSIOX         (8) GETPFK         (10) POWEROFF BACKLIGHT         (11) CONTINUE         (12) BARCODE         (13) TCAM         (14) GRAPHICS (LINE)         (15) GRAPHICS (CIRCLE)         (16) GRAPHICS (CIRCLE)         (17) KANJI         (18) MASKI         (19) STORX LDIRX         (20) Serial switch         (21) CALLX         (22) RDVRAM         (23) ICCARD         (24) User BIOS area	21 21 21 21 21 21 21 21 21 21 21 21 21 2		82 82 83 84 86 89 91 101 103 104 106 109 113 115 116 118 119 120 122 124 125 127 134 140

	(26)	BDOS Error	21	_	143
	(27)	Check the ALARM/POWER OFF	21	-	147
	(28)	Mount check of the cartridge device	21	-	149
	(29)	ART interrupt	21	-	151
	(30)	EXTHOOK	21	-	154
-	(31)	RIOSHOOK	21	-	156
	(32)	Extend communication protocol	21	-	158
	(33)	Extend IC-card protocol	21	-	169
	(34)	FDD Format/Copy utility for the EHT-10/EHT-10/2	21	-	185

## INTRODUCTION

< PURPOSE OF THIS MANUAL >

This manual describes the functions of the operating system for the SEIKO EPSON EHT-10/EHT-10/2 system. It is intended for system house users who are to develop applications programs which make the best of the EHT-10/EHT-10/2's capabilities.

The reader is assumed to be familiar with the following:

- Basic knowledge about the CP/M operating system
- General knowledge about machine-language programming
- Z80 instructions

#### < BEFORE READING THIS MANUAL >

This manual uses the following notational conventions:

(1) Types of the EHT-10 series

In EHT-10 series, there are three types as follows.

EHT-10 ----- Touch key type EHT-10/2 ----- Keyboard type with backlight EHT-10/2B ----- Keyboard type without backlight

In this manual, EHT-10/EHT-10/2 are the general names of the above three types unless there is the special note for the name.

(2) Data representation

This manual uses binary, decimal, and hexadecimal numbers. They are represented in the formats:

Binary:00100011B (Numbers are followed by 'B')Decimal:35 (Only numerals)Hexadecimal:23H (numbers are followed by 'H')

Character constants are enclosed in apostrophes (') or double quotation (").

Example: 'ABC' or "ABC"

#### (3) Register Representation

The EHT-10/EHT-10/2 registers are illustrated below.



Registers are expressed at A, B, DE, HL, and so on. They may sometimes be followed by the word "register" to clearly identified as the Z flag (or Z), the C flag (or C), and so on.

(4) Bit Representation

Bits are numbered 0, 1, and so on, from the lowest order bit (0) to the highest order bit. The lowest order bit is referred to as the least significant bit (LSB) and the highest order bit as the most significant bit (MSB).



#### (5) Address Representation

Addresses are generally represented in hexadecimal notation. I/O addresses are prefixed by "P".

Examples: 0010H Memory address 10H P10H 1/0 port address 10H

Note that the contents of I/O addressed may differ during read and write operations.

#### < THE CONSTITUTION OF THIS MANUAL >

This manual consists of four parts, which are software, hardware, BASIC, APPENDIX.

Part 1 - SOFTWARE - describes the features of the EHT-10/EHT-10/2 and the information which are necessary to develop software like as BIOS, BDOS, I/O, system function, system hook, and etc. Part 1 is divided into 11 chapters. The contents are as follows.

chapter 1, 2 Overview and basic operation chapter 3 to 8 BIOS, BDOS, I/O, interrupt, and etc chapter 9 System function chapter 10, 11 How to use expansive function

Part 2 - HARDWARE - explains the hardware and interfaces. It also explains the necessary information to use the universal unit. Part 2 is divided into 6 chapters. The contents are as follows.

chapter 1	Overview of the hardware
chapter 2 to 4 chapter 5	I/U registers, interfaces and etc Power supply
chapter 6	Advice for designing the circuit

Part 3 - BASIC - is divided into 3 chapters.

chapter	1	Memory con	trol	of BA	SIC			
chapter	2	Expansion	of	the	commands	including	the	commands
		analyzing	rout	ine.				
chapter	3	Expansion	of th	ne seq	uential ac	cess devices	i.	

Part 4 - APPENDIX - describes the various kinds of code tables and BIOS call list. And at the end of this manual is the index.

Because each section of this manual describes each subject independently, you don't have to read this manual from the beginning to the last. You can use the contents to search the part you want to read. However, we recommend to read the chapter 1 and the chapter 2 of software part and the chapter 1 of the hardware part first because the overview of the EHT-10/EHT-10/2 is described there.

## PART 1 SOFTWARE

#### CHAPTER 1 SYSTEM OVERVIEW

1.1 Characteristics of System

1.1.1 EHT-10/EHT-10/2 concepts

EHT-10/EHT-10/2 has the new functions in addition to the basic functions of PX-4/HX-40. EHT-10/EHT-10/2 is a hand-held computer much smaller than PX-4/HX-40.

Although the EHT-10/EHT-10/2 can be carried by one hand, EHT-10/EHT-10/2 has mass memory of 256-Kbyte (maximum) RAM, 128-Kbyte system ROM, and 128-Kbyte (maximum) application ROM, that is much more than general personal computers. EHT-10/EHT-10/2 is equipped with CP/M and BASIC in standard so that application programs can easily be created.

Further, EHT-10/EHT-10/2 has user-friendly functions such as the touch panel with LCD and IC card read/write functions so that the operators can easily handle the terminals without special education and training.

Smali-sized and high-performance mobile computer EHT-10/EHT-10/2 is most appropriate for order issuing and accepting jobs for distribution industry, liaison jobs for financial institutions, electricity, gas, and water gauge examination, stock-taking and supply order issuing for supermarkets, cargo booking for carrying trade and agriculture industry, and production management jobs.

1.1.2 Characteristics of the system

(1) Extended CP/M version 2.2

- 1 The extended CP/M Ver. 2.2 is used as OS for EHT-10/EHT-10/2. This OS enables the users to easily create application programs. The application programs of other devices that operate under CP/M can be converted by taking into account of display screen and input method.
- 2 Main memory can be extended up to 256 Kbytes. A part of main memory (up to 232 Kbytes) can be used as RAM disk. A RAM disk can be handled in the same way as a floppy disk and can be accessed in high speed. Further an IC card can be used instead of a floppy disk.
- 3 Application programs input from the ROM drive can be loaded to the main RAM for execution as in the previous CP/M way. However, these application programs can be directly executed in the ROM drive to use memory efficiently and to have less power consumption. Up to 1-Mbit (128-Kbyte) ROM can be mounted in the ROM drive so that a large-size application programs can be handled. Further, an application program stored in an application ROM inserted in the ROM drive can be automatically executed at power on.

- (2) Software development environment
- 1 Application programs in BIOS and BASIC level has the compatibility with PX-4/HX-40 and conversion can easily be done (however, the display size and input methods must be noted).
- 2 EPSON BASIC is supported so that programs can easily be created.
- 3 The development cartridge is supported in standard as an option. Using the development cartridge and development software(WAD), the users can easily create and debug application programs.
- (3) Touch panel and keyboard
- 1 The input screen and the display screen are the same for EHT-10 for higher operationability and extendability. The touch panel has 70 (horizontal 5 x vertical 14) input points and OS supports normal, alphanumeric and kana input modes for the touch panel. The keys can be freely redefined in normal mode by application programs so that user-friendly software can be created with united input and display screens.
- 2 EHT-10/2 has 34 keys and OS supports normal, alphabet, and kana modes. Each mode can be identified from the lit LED. The Keys can be redefined in normal or alphabet mode.
- (4) LCD and EL backlight
- 1 The size of EHT-10 real screen (LCD) is 12 columns x 14 lines and the size of EHT-10/2 real screen (LCD) is 20 columns x 4 lines. In addition, 12 columns x 42 lines of EHT-10 virtual screen and 20 columns x 25 lines of EHT-10/2 virtual screen are supported so that application programs requiring larger screen can be created and executed.
- 2 A model (EHT-10/2B) that has EL backlight is provided so that it can be used in a dark room.
- (5) IC card equivalent to ISO

By using the IC card reader/writer equipped in standard, IC cards can be handled in the same way as floppy disks (storing collected data, distributing programs, etc.).

(6) Barcode reader

Barcode reader I/F is equipped in standard. OS supports reading barcodes that are JAN/EAN/UPC, 3 of 9, codabar, and interleaved 2 of 5 so that barcodes can be easily used.

(7) Kanji support

OS supports 174 characters of Japanese Characters (kana) and symbols, and 996 kanji characters. Further JIS level-1 kanji characters can be used when JIS level-1 ROM is mounted in the ROM drive. Up to 6144 external characters (gaiji) can be registered (the maximum number differs depending on the media size). (8) Support devices

#### 1 For EHT-10/EHT-10/2, all memory I/O devices are handled as disk drives.

- a. RAM disk (A:) A part of RAM can be used as a disk to realize a disk that can be accessed in high speed.
- b. ROM drive (B:) The ROM drive is supported so that a program can be executed in ROM.
- c. IC card (C:)
   IC card is supported so that it can be used as an R/O disk (ROM card)
   or R/W disk (RAM card).
- 2 The following device I/F are supported in addition: Floppy disk (D: or E:) Barcode I/F RS-232C I/F Cartridge I/F (printer unit) Clock Buzzer
- (9) Timer management
- 1 Even when an application program is being executed, the alarm function displays the alarm screen to notify the user that specified time came.
- 2 The wake function turns on the main frame power to execute the specified program at the specified time.
- (10) Other functions
- 1 The MENU function enables the user to select programs easily.
- 2 The DLL function can receive and execute programs sent from the host computer. The received program is directly initiated. The protocol used for transmission can be freely changed. Further, the protocol can be extended to the one exclusively used by an application.
- 3 The system menu function can set the system environment, send or receive files, manage disks, and supports the test function.
- 4 The calculator function can perform arithmetic operations and the result of arithmetic operation can be fetched as data to be input to an application. Arithmetic operations are performed in BCD so that results with less errors can be obtained.
- 5 If the power is turned off in continue mode, processing can be continued when the power is turned on again.
- 6 Self-test is performed at power on for higher system reliability.
- 7 The sleep mode and automatic power off functions prevent unnecessary power consumption.

----

#### 1.2 System Configuration



Figure blow shows the devices supported by EHT-10/EHT-10/2.

Fig. 1.1 EHT-10/EHT-10/2 System Configuration

EHT-10/EHT-10/2 consists of two CPUs, using Z-80 as the main CPU and 7508 as the slave CPU.

Memory space are used as four main memory banks and subbanks. 64-Kbyte RAM and 128-Kbyte system ROM are equipped in standard. RAM can be extended up to 256 Kbytes. See "Section 1.4 Memory Map" for details on banks.

Slave CPU 7508 controls the keyboard, clock and power supply. Main CPU Z-80 supports RS-232C, cartridge I/F, IC card I/F, and barcode I/F.

#### 1.3 Software Structure

1.3.1 Overview

The EHT-10/EHT-10/2 system software is stored in the 128-Kbyte system ROM. Since the system software corresponds to EHT-10, EHT-10/2, and EHT-10/2B, the common system ROM is mounted in EHT-10, EHT-10/2, and EHT-10/2B. The system ROM consists of the OS, utility, BASIC, and kanji sections.



- The size of each section is 32 Kbytes.

Fig. 1.2 Structure of system ROM

#### 1.3.2 Structure

Figure 1.3 shows the structure of EHT-10/EHT-10/2 operating system functions.



Fig. 1.3 OS Structure

(1) Structure of system section

The system section consists of the kernel and CP/M sections.

- Starter
  The starter is the boot loader that initiates the system and performs initialization.
- 2 Relocater The relocater is a group of processing routines that relocate programs resided in RAM. The relocater is initiated by the starter or BIOS WBOOT.
- 3 Interrupt processing Interrupt processing performs various interrupts such as 7508 (key input, alarm, 1-second interrupt, and power). ART (serial receiving), ICF, OVF, and EXT (printer, interval, and IC card back panel).
- 4 Resident processing Resident processing is structured by a part of interrupt processing, bank switch processing, and the BIOS and BDOS interface sections.
- 5 MENU MENU displays executable program files in the menu screen so that the user can select and execute a program.
- 6 DLL DLL down-loads an execution program from the host computer through a communication line and executes the program.
- 7 CCP section The CCP section receives data related to program initiation from MENU and DLL and initiates a program.
- 8 BDOS section The BDOS section is the CP/M disk file management section. The BDOS section can be used by an application program calling BDOS. EHT-10/EHT-10/2 supports the ROM drive, IC card, and a part of RAM as disks.
- 9 BIOS section The BIOS section is the CP/M I/O handler. The standard CP/M BIOS is extended for EHT-10/EHT-10/2.
- (2) Utility section

The utility section consists of the calculator and system menu functions. Refer to EHT-10/EHT-10/2 Operating Manual for details.

(3) BASIC section

The BASIC section contains the EPSON BASIC interpreter. The BASIC interpreter is directly executed in ROM instead of expanded in RAM, so that memory space in main RAM can be used efficiently. Refer to "BASIC PART" and EHT-10/EHT-10/2 BASIC Reference Manual for details.

(4) Kanji section

The kanji section stores 996-kanji and 174-nonkanji fonts.

1.4 Memory Map

1.4.1 Memory space

System ROM (128 Kbytes), RAM (up to 256 Kbytes), application ROM (up to 128 Kbytes), in total of up to 512 Kbytes, are supported as EHT-10/EHT-10/2 memory space.

However, main CPU Z-80 can directly access only 64-Kbyte memory space starting from 0000H to FFFFH.

Because of this, the bank method is used for EHT-10/EHT-10/2 so that main CPU Z-80 can directly access 512-Kbyte memory space.

In the EHT-10/EHT-10/2 bank method, main banks and subbanks are used in memory map shown in FIgure 1.4.

Generally, OS switches banks so that application programs do not require to take into account of bank switching. However, high-level application programs can be created by taking into account of the concept of bank switching and using this concept for application program execution.

[Notes] Banks are specified as follows in this manual: [main bank]#[subbank]

Example

[bank 2#1]: Subbank 1 in bank 2. 2#1 starts from 8000H to FFFFH in application ROM and is mapped from 6000H to DFFFH in the memory map.

However, the system bank is called [system bank] because the system bank does not have any subbank.

000         011         012         013         014         015         015         111         112         113         210         211         212         21           111111         Resident RAM         Residen RAM         Resident	000         011         012         013         014         015         016         111         112         113         210         211         212         13           111111         Resident         Resident         RAM         Resident         Res	Adda	Syst BANK			BANK	0 (RAM	(			BANK1 (S	ystem	ROM)	BANK2 (1	Applic	ation	ROM)
0001 Resident RAM Resident RAM Resident RAM Standard	000II Resident RAM 2000II System for and active state of the state of	ress	em	010	011	012	013	014	015	910	111	112	113	210	211	212	213
C0001         Expansion         Ex	C0001         Expansion         Expasin         Expasion         Expas						Resid	dent R	AM								
Column         Same #1         RAM #2         RAM #2 <thram #2<="" th=""> <thram #2<="" th=""> <thram #2<="" t<="" td=""><td>Early area         Mark #1         Mark #2         Mark #2</td><td>[0001]</td><td>Standa</td><td>and a</td><td>Expansio</td><td>Expansio</td><td>Expansio</td><td>Expansion</td><td>xpansion</td><td>Expansio</td><td>System ROM</td><td>System ROM</td><td>System ROM</td><td>App1. Ron</td><td>Appl. Roh</td><td>App1. ROM</td><td>App1. ROM</td></thram></thram></thram>	Early area         Mark #1         Mark #2	[0001]	Standa	and a	Expansio	Expansio	Expansio	Expansion	xpansion	Expansio	System ROM	System ROM	System ROM	App1. Ron	Appl. Roh	App1. ROM	App1. ROM
60001 System 59001 System 60001 System 600001 System 60001 System 600001 System 600001 System 600001 System 600001 System 60000 System 6000 Sy	600II System 600II System 600II 000II 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		Bank	area	11 101	RAN 12	CI HVE	RAN 14	RANK #5	RAM 16	8000II I FFFFII	10000II I 17FFFII	18000II - 1 1FFFFI	0II 1 7FFFFII	8000II I FFFFII	1000011 I 17FF f 11	180001 I I
00001 00001 00001 1 1 1 1 1 1 1 1 1 1 1	00001 00001 1 1 11f[1] Standard RAM (common area) + +	10008	System														
00001 Standard RAM (common area)	00001 Standard RAM (common area)	10000	10000		1											}	
							Stand	lard R	AM (C	иошшо	area)						
		110000															

Fig. 1.4 Memory space

(1) System bank

The system bank does not have any subbank. The system bank is structured as follows: 0000H to 7FFFH: 00000H to 07FFFH (OS section) in system ROM 8000H to DFFFH: 24 Kbytes in standard RAM (8000H to DFFFH in bank 0#0) E000H to FFFFH: Resident RAM section (a part of standard RAM that can be directly accessed by CPU Z-80 regardless of bank switching.)

(2) Bank O (bank in all RAMs)

All RAMs have bank 0. Bank 0 has the following subbanks depending on the size of extended RAM:

Size of extended RAM		Subbanks								
	0#0	0#1	0#2	0#3	0#4	0#5	0#6			
OKB(only for Standard RAM)	0									
64KB	0	0	0							
128кв	0	0	0	0	0					
192KB	0	0	0	0	0	0	0			

An empty field indicates that there is no subbank.

Bank O is structured as follows:

0000H to SFFFH: This is a part of standard RAM. This can be accessed regardless of main bank and subbank switching. 8000H to DFFFH: This is switched with a part of another extended RAM depending on the subbank. When bank O#O is specified, this becomes a part (24 Kbytes) of standard RAM. E000H to FFFFH: Resident RAM (same as the system bank)

The BIOS and BDOS entries are relocated in bank 0#0. Bank 0#0 is used as the base of CP/M. Also, programs in load and execute mode are loaded in bank 0#0.

Generally an application program does not switch the bank in extended RAM to the one from 0#0 to 0#6 because the extended RAM is used as a disk by the system.

(3) Bank 1 (bank in system ROM)

Bank 1 is the extension of system bank and is bank 0 6000H to DFFFH replaced as a part of system ROM.

Bank 1 has the following subbanks: Bank 1#1: O7FFFH to OFFFFH (utility section) in system ROM are mapped. Bank 1#2: 10000H to 17FFFH (BASIC section) in system ROM are mapped. Bank 1#3: 18000H to 1FFFFH (kanji section) in system ROM are mapped.

(4) Bank 2 (bank in application ROM)

Bank 2 is bank 0 6000H to DFFFH replaced as a part of application ROM (ROM drive). Bank 2 has the following subbanks depending on the size of application ROM:



Address Address

Fig. 1.5 Structure of Application ROM

(Note 1) For application ROM, ROM physical addresses differ from the logical addresses mapped in memory.

1.4.2 Memory map

The resident RAM area (E000H to FFFFH) and bank O#O contains the system parameters, entries of resident processing routines BlOS and BDOS. Bank O#O is used as the base of CP/M.

The memory map of bank 0#0 for EHT-10 differs from that of EHT-10/2. Figures 1.6 and 1.7 show the memory maps of EHT-10 and EHT-10/2.

(1) RBDOS and RBJOS

RBDOS1 is functionally same as RBDOS2 and RBIOS1 is functionally same as RBIOS2. An application program in load and execution mode (loaded and initiated starting from 100H in RAM) uses RBIOS1 and RBDOS1 (by calling BIOS and BDOS).

An application program in ROM-based mode (operates at 6000H to DFFFH in bank 2) uses RBIOS2 and RBDOS2 (both have fixed addresses) because RBIOS1 and RBDOS1 may be placed at the different bank in ROM.

(2) RSYSPR system area

RSYSPR is a system program in resident section that performs bank switching interrupt processing. The system area consists of work area, jump table, and hook used by the system.

(3) Gaiji definition area and key redefinition area

Gaiji font is defined in the gaiji definition area and key codes are redefined in the key definition area.

(4) Virtual screen and system screen

The virtual and system screens are the display data storage areas. The size of virtual screen is fixed and the map is not changed even if the virtual screen size is modified.

(5) VRAM

For EHT-10, VRAM is used as the VRAM data save area. For EHT-10/2, VRAM consists of VRAM1, VRAM2, and VRAM3. VRAM1 is used as the virtual screen 1 VRAM, VRAM 2 is used as the virtual screen 2 VRAM, and VRAM 3 is used as the system screen VRAM.

(6) User BIOS area

The default size of user BIOS area is 0. However, the user BIOS area is reserved between the RAM disk and system common area 2 when there is user BIOS.



(Note) The position of user BIOS area differs depending on EHT-10 or EHT-10/2.

The user BIOS area is used when a part of the user program is made to remain resident after termination of user program.



352 bytes are required if the screen will be used from top to bottom. or 192 bytes for side-to-side use.

Fig. 1.6 EHT-10 Bank O#O Memory Map

- Note 1: DLL is executed when forced-load is specified.
- Note 2: A specific program is executed when forced-selection is specified. The above two items are specified by CONFIG utility Exec type.
- Note 3: Application is initiated by the wake string when Wake is used for starting.
- Note 4: At system initialization or reset, menu processing is executed even if there is only one execute-mode program in the disk.

2.2.2 Status change

Figure 2.2 shows the changes of EHT-10/EHT-10/2 status.

EHT-10/EHT-10/2 has the following status:

- (1) Power off (restart) status
- (2) Power off (continue) status
- (3) Application operation status(4) MENU screen
- (5) Alarm/Wake screen
- (6) System menu screen
- (7) Calculator screen
- (8) DLL screen

Power off status is in either restart or continue mode. The mode is determined according to the condition when the power is turned off.

EHT-10/EHT-10/2 is in MENU or DLL screen status when the power is turned on in restart mode. EHT-10/EHT-10/2 is in the status held at power off when the power is turned on in continue mode.



Fig. 2.2 Changes of System Section Status

('END' means the Return key for EHT-10/2.)
A Power switch off, power failure, automatic power off
B Power switch off, power failure, 50 seconds elapse, END
C Power switch off, power failure, 50 seconds elapse, END,
 power switch on
D Power switch on
E Wake

2.3 System Initiation and Termination

2.3.1 System initialization

(1) Overview

If program execution enters wild run status, 7508 (slave CPU) operation enters wild run status, RAM contents are destroyed due to lowered battery voltage, or the system is initialized for the first time, system initialization is performed to reset the EHT- 10/EHT-10/2 environment (the contents of ROM are protected in general when the battery voltage is lowered).

(2) Initiation

The system initialization is initiated under one of the following conditions:

- 1 7508 operation hangs up or the battery voltage is lowered. Make the AC adapter enter ready status and open the back panel to press the reset switch.
- 2 The power is turned off during system initialization and then the power is turned on again (during CONFIG in system initialization).
- 3 The result of system area sum check performed at power on is not equal to the result of sum calculated at power off.
- An error occurs as a result of the following check operations during reset processing and initialization is specified (see "Section 2.5 Self-Test):

(a) The total of the internal RAM disk size and the BIOS size exceeds the maximum. The maximum values are as follows:
EHT-10: 39.5 Kbytes
EHT-10/2: 40.0 Kbytes
(b) The BDOS or BIOS loading address in RAM is not equal to the value calculated from the RAM disk size during reset processing.

(3) Processing

The system is initialized in the following way:

- 1 A keyboard reset command is sent to CPU 7508 to initialize the keyboard.
- 2 The interrupt mask status and the bank status are initialized.
- 3 The system work area (RSYSAR1) is initialized.
- 4 System device cold boot (note 1) is performed.
- 5 The display screen is turned on to display system initialization messages. When the cause of initiating system initialization is 3 or 4 explained in "Section (2) Initiation", a system error message is displayed for about one second before system initialization message.
- 6 The default value is set as the size of RAM disk and the disk is formatted.
- 7 Reset processing is performed (explained later).
- 8 BIOS BOOT processing is performed.
- 9 System menu CONFIG is initiated so that the user can perform system initialization. Table 2.1 shows the system default values.

(Note 1) Device cold boot and device warm boot

- (1) Device warm boot
- 1 The I/O registers are initialized.
- 2 LCD is initialized.
- 3 Whether ROM is mounted in the ROM drive is checked.
- 4 Whether an extended RAM is mounted is checked.
- 5 Whether the cartridge device is mounted is checked and the cartridge device is initialized.
- (2) Device cold boot
- 1 The system parameters (RSYSAR2 and RSYSAR3) are initialized and the resident processing routines are loaded.
- 2 1-second interrupts are allowed and the alarm is suppressed.
- 3 The DIP switches are read to determine the corresponding country.
- 4 Device warm boot is performed.
- 5 The display and keyboard are initialized.

(4) End of system initialization

System initialization ends when CONFIG ends. Then, MENU or DLL processing is initiated.

Table 2.1 Default Values of System Para	meters
---	--------

Parameter	Contents	Default Value
Self Test	Self-test function ON/OFF	ON
BASIC	Parameters of BASIC	Open file 3 Record 128 bytes Buffer size 256 bytes
Calculate	Display mode of Calcu- lator operations result	Floating
Date/Time	Date Time	00/00/00 (note 3) 00:00:00
Disk size	Size of internal RAM disk Size of User BIOS	31КВ О
DLL	Communication para- meters for down load- ing.	Bit rate4800 bpsBit length8 bitsParityNONStop bit2 bitsProtocolFilinkConnectorRS-232C
RS-232C	Communication para- meters for RS-232C	Bit rate4800 bpsBit length8 bitsParityNONStop bit2 bitsControlNONSI/SODisable
Exec type	Initiation	Auto
Power OFF	Power control time	Main Power 5 min. Printer Unit 180 sec. Back light 3 min.
Printer	Printer output	Printer unit
Country	Character set for country	ASCII (note 1) or Japan

(Note 1) Japan or overseas is specified according to DIP switch setting.

(Note 2) The output destination for Calculate, Country, DLL, and Printer is initialized by reset processing. (Note 3) The default values of date and time are set only during

CPU 7508 reset processing.

2.3.2 Reset

(1) Overview

If program execution enters wild run status, press the reset switch at the side of main frame to reset data residing in RAM.

(2) Initiation

Reset is initiated under one of the following conditions:

- 1 The reset switch is pressed once in power on status.
- 2 After the power failed in power off status, the power switch is turned on or an alarm/wake occurs.
- 3 The result of user area (TPA or user BIOS area) sum check is not equal to the value calculated at power off.

(3) Processing

- 1 A keyboard reset command is set to CPU 7508 to initialize the keyboard.
- 2 Device cold boot is performed (see the section of system initialization).
- 3 The sizes of the standard RAM section and user BIOS area are added. System initialization is performed if the total exceeds the maximum value (39.5 Kbytes for EHT-10 and 40 Kbytes for EHT-10/2).
- 4 System initialization is performed if the result of adding the RBDOS1 or RBIOS2 load address and its size is not equal to the beginning of the standard RAM in the RAM disk.
- 5 The disk parameter blocks are set for RAM and ROM disks.
- 6 RAM disk sum check is performed. If the result unmatches, the user selects whether to perform RAM disk formatting.
- 7 User area sum check is performed if reset processing is initiated due to power failure in power off status.
- (4) End of reset processing

Reset processing is automatically ended after the above operations and BICS BOOT processing is started (however, control is returned to system initialization if reset processing is initiated from system initialization).

2.3.3 Restart mode

(1) Overview

The power is turned off in restart mode when the power switch is turned off during MENU or DLL processing.

- (2) Initiation from power off status in restart mode
- Power switch on When the power switch is turned on, MENU or DLL processing is restarted or an application is directly initiated (see "Section 2.2.1 Control flow").

- 2 Alarm time When the alarm time comes, the alarm is sounded and the alarm screen is displayed. Operation 1 is performed if the power switch is turned on at this point. 3 Wake time Operation 1 is performed when the wake time comes. In this case, an application can be initiated by assuming the wake string as key input data. (3) Restart mode conditions The power switch is turned off in MENU or DLL status or the power is 1 turned off automatically. Restart mode is set by BIOS CONTINUE in an application and the power is 2 turned off by the power switch or by BIOS POWER OFF in restart mode. (4) Processing Device warm boot is performed (see the section of system 1 initialization). The self-test is performed (see "Section 2.5 Self-Test"). 2 High-pitched tone is sounded to indicate power off in restart mode. 3 4 **BIOS WBOOT is performed.** 5 Control is returned to MENU, DLL, or an application according to the initiation condition (power switch on, wake) and the system status (existence of execute-mode file, execution type). (5) End Processing ends after initiating MENU, DLL, or an application. 2.3.4 Continue mode (1) Overview Continue mode is used as the default during application execution. When the power is turned on after turned off in continue mode, the program can be continued from the status held at power off. (2) Continue mode initiation from power off 1 Power switch on The program is continued from the status held at power off. 2 Alarm time When the alarm time comes, the alarm is sounded and the alarm screen is displayed. Operation 1 is performed if the power switch is turned on at this point.
- 3 Wake time When the wake time comes, alarm is sounded, the wake screen is displayed, and then the program is continued from the status held at power off. In this case, the wake string is ignored.
- (3) Continue mode conditions
- 1 The power switch is turned off during application execution.
- 2 The power is turned off because the automatic power off time comes during application execution.

- 3 The power is turned off by the BIOS POWER OFF in continue mode.
- 4 The power is turned off during calculator or system menu operation because the power switch is turned off or the automatic power off time comes.
- 5 The power switch is turned off when the power failed during operation other than system initialization, MENU, or DLL operation.

(4) Processing

- 1 Device warm boot is performed (see the section of system initialization).
- 2 Self-test is performed (see "Section 2.5 Self-Test").
- 3 A high-pitched tone is sounded to indicate power off in continue mode.
- 4 The register status held at power off is recovered and control is returned to the execution address used at power off.

(5) End

Control is returned to the address used at power off and then processing ends.

(6) Remarks

Generally, processing is continued from the status held at power off when the power is turned on in continue mode. However, the cartridge options are initialized at power off and cannot be continued. For the printer unit, complete continuation operation is not possible because specifications are reset during 1/0 operation at power on and the country specification and gaiji definition data are initialized.

See "Section 7.3 Cartridge Interface" to continue cartridge options.

2.3.5 Sleep function and automatic power off function

(1) Overview

To have less power consumption, EHT-10/EHT-10/2 supports the sleep and automatic power off functions. Further, EHT-10/2B also supports the automatic backlight off function.

(2) Sleep function

Sleep mode is enabled by using a Halt command to stop CPU when processing is waiting for input data during BIOS CONIN operation. When an interrupt occurs during Halt command execution, sleep mode is cleared and interrupt processing is executed. Processing enters in sleep mode again after interrupt processing when the interrupt is not a key input interrupt. When the interrupt is a key input interrupt, input operation is executed and CONIN is terminated.

The automatic power off, alarm, power switch off, and power failure functions can operate in sleep mode.
(3) Automatic power off function

The power is automatically turned off if any data is not input from the keyboard within the fixed time (default is 5 minutes) while CONIN operation is waiting for input data. Processing is continued from the status held at power off when the power is subsequently turned on (the power switch is turned on or the wake time comes). However, the power is turned on in restart mode if the power is automatically turned off during MENU or DLL processing.

The time for automatic power off is set by CONFIG.

(4) Automatic backlight off function

For EHT-10/28, the backlight is automatically turned off if any data is not input from the keyboard within the fixed time (default is 3 minutes) while BIOS CONIN operation is waiting for input data. The backlight is turned on again when data is input from the keyboard or the main power supply is turned off and on. The automatic backlight off function is effective only when the backlight switch is on.

The time for automatic backlight off is set by CONFIG.

(5) Using the automatic power off, automatic backlight off, and sleep functions in an application

Some application programs input data by using CONST to poll the keyboard status. The automatic power off, automatic backlight off, and sleep functions do not operate for such application programs.

In order to use these functions in such cases, processing shown in FIg. 2.3 is required in the application programs. (See APPENDIX 9 SAMPLE 25)



Fig. 2.3 Automatic Off Operation

(Note 1) Omit steps (2), (7), (8), and (11) for EHT-10/EHT-10/2 since EHT-10/EHT-10/2 does not have the automatic backlight off function. If the automatic backlight off function is not required although EHT-10/2B is used, the above steps should be omitted.

Step Explanation

SOFTWARE Page 2 - 11

- (2) The automatic backlight off time is set. The automatic backlight off time is set. ELOFFEND is set in two bytes as follows: (ELOFTIME) + (TIMERO) -> (ELOFEND) (F023H) (F02DH) (F387H)
- (3)&(4) Whether data is input from the keyboard is checked during BIOS CONIN.
- (5)&(6) The automatic power off time is checked. Whether the automatic power off time has come is checked. If the time has come, the power is turned off. Checking is done according to the expression below. The time has come if the result of calculation is negative. (TIMEEND) - (TIMERO) (F5F9H) (F02DH)
- (7)&(8) The automatic backlight off time is checked. Whether the automatic backlight off time has come is checked. If the time has come, the backlight is turned off. The following expression is used: (ELOFEND) - (TIMERO)
- (9) HALT Sleep mode is enabled by a Halt command. Processing jumps to operation (3) by a Jump command after the Halt command.
- (10) & Data input from the keyboard is fetched and the backlight
   (11) is turned on. Data input from the keyboard is fetched by BIOS CONIN and the backlight is turned on.
- 2.3.6 Power failure
- (1) Overview

Power failure occurs if main-battery voltage is lowered (to about 4.7 V or less). If power failure occurs, the power failure screen is displayed to notify the user of power failure.

(2) Cause

Processing for power failure starts when a power failure interrupt is sent from the slave CPU (7508) to the main CPU.

1 Slave-CPU (7508) operations

If the voltage of the main battery is lowered to about 4 V or less, a power failure interrunt is sent to the main CPU and the power is also supplied from the subbattery. A power failure interrupt is sent every 1 second. The power is forcibly turned off at the main CPU if power off processing is not executed at the main CPU within 50 seconds. In this case, reset processing is performed when the power is turned on again.

2 Main-CPU operations After accepting a power failure interrupt, the main CPU continues I/O operation until it can be left off if I/O operation is in progress, and then performs power failure processing.

SOFTWARE Page 2 - 12

## (3) Processing

If power failure occurs, power off processing is executed (excepting the actual power off operation) in continue mode and then the power failure screen shown in Figure 2.4 is displayed.

Charge Battery	Charge Battery
	EHT-10/2
ЕНТ-10	Fig. 2.4 Power Failure Screen

The power failure screen is displayed on and off every second and then the power is turned off under one of the following conditions:

- 1 30 seconds elapse after displaying the power failure screen is started.
- 2 The user turns off the power switch after displaying the power failure screen is started.
- 3 50 seconds elapse after power failure occurs.

The power failure screen is displayed by using the system screen so that displaying the power failure screen does not affect the user screen.

(4) Others

- 1 Recovering after the power is turned off - Connect the charger to charge sufficient electricity. - Replace the main battery. Perform one of the above operations and turn the power switch on to recover from power failure.
- 2 Continuation processing after power failure Processing can be continued after one of the operations explained in 1 if the power is turned off after the power failure screen is displayed. However, if the slave CPU forcibly turns off the power before the power failure screen is displayed (I/O operation is not ended within 50 seconds after power failure occurs), reset processing (same as processing executed when the reset switch is pressed) is executed but not continuation processing even if one of the operations explained in 1 is performed.
- 3 Power failure occurrence in power off status Power is consumed to backup 7508 and RAM even in power off status. The subbattery is supplying the required power if the voltage of main battery is lowered to the fixed value or less. In such a case, the power is not turned on even if the power witch is turned on. Therefore, perform one of the operations explained in 1 to execute reset processing and to recover the system. In this case, the system area sum check is executed and the result is compared with the value calculated at power off. If the result is not equal to the calculated value, it is assumed that the memory contents are destroyed in power

off status and system initialization is executed.

4

Alarm/wake at power failure When power failure occurs, the alarm/wake function is disabled and the status held at power failure is recovered at the subsequent power switch on. Because of this, the alarm/wake operation to be performed between power failure occurrence and the subsequent power on may be omitted.

## 2.4 Alarm/Wake

## 2.4.1 Overview

EHT-10/EHT-10/2 has the alarm function that sounds alarm and displays the alarm screen when the set time comes, and the wake function that automatically turns the power on if the power is off and performs the specified operation when the set time comes (however, either alarm or wake can be specified at the same time).

The alarm/wake operation differs depending on the power on or off status. Table 2.2 shows the outline of alarm/wake functions.

Туре	Power of	ff status	Power on status
	Restart mode	Continue mode	
Alarm	<ol> <li>The power is turned off in restart mode after the alarm screen is displayed.</li> <li>Power on pro- cessing in restart mode is executed if the power swi- tch is turned on while the alarm screen is display- ed.</li> </ol>	<ol> <li>The power is turned off in con- tinue mode after the alarm screen is displayed.</li> <li>Power on pro- cessing in continue mode is executed if the power switch is turned on while the alarm screen is displayed.</li> </ol>	<pre>(1) The alarm screen is dis- played.(at dis- play timing.)</pre>
Wake	(1) Alarm is sounded and power on processing in re- start mode is executed. Pro- cessing specified by the wake string is executed if it is specified.	<pre>(1) Alarm is sound- ed and power on processing in re- start mode is executed. A wake string is ignored even if it is specified.</pre>	<pre>(1) The wake sc- reen is displayed (at display timing.)</pre>

Table 2.2 Alarm/Wake Functions

2.4.2 Alarm

(1) Function

When the set time comes, alarm is sounded and the alarm screen shown in Figure 2.5 is displayed.



When alarm occurs in power off status and the power switch is turned on while the alarm screen is displayed, power switch on processing is executed.

(2) Setting and resetting

The alarm is set or reset by calling BIOS TIMDAT. See "Section 4.2 Overview of BIOS Commands" for TIMDAT.

- (3) Initiation
- 1 Alarm time comes in power off status.
- 2 Alarm time comes in power on status. In this case, the alarm function is initiated at a timing so that I/O execution is not affected.

(4) End

- 1 If the alarm function is initiated in power off status, processing returns back to the status held before initiation (power off status in restart or continue mode) under one of the following conditions:
  - (a) 50 seconds elapse after initiation.
  - (b) The 'Press' (EHT-10) or 'Return' (EHT-10/2) key is pressed.
  - (c) The power switch is turned off.
  - (d) Power failure is detected.
  - (e) The power switch is turned on. In this case, power switch on processing is executed.
- 2 If the alarm function is initiated in power on status, processing returns back to the status held before initiation under one of the following conditions:
  - (a) 50 seconds elapse after initiation (see Note 1).
  - (b) The 'Press' (EHT-10) or'Return' (EHT-10/2) key is pressed.
  - (c) The power switch is turned off.
  - (d) Power failure is detected.

(Note 1) The alarm display time can be modified by modifying the contents of system area ALRMTIME (F1CAH). The value is indicated in the units of seconds.

(5) Remarks

SOFTWARE Page 2 - 16

- The alarm screen is displayed on the system screen and the system 1 screen returns back to the status held before alarm occurrence after alarm processing ends.
- In the alarm screen, control codes (OOH to 1FH) are moved to "^" + "@" 2 (40H) and displayed.
- Alarm (or wake) occurrence is suppressed temporarily during one of the 3 following operations:
  - (a) Data is being received during DLL or DL.
  - (b) Data is being input in alphabet or kana mode for EHT-10.
  - (c) Power failure occurs.

### 2.4.3 Wake

(1) Function

The following operations are executed at the set wake time:

- 1 When processing is initiated in power off status in restart mode The power is turned on, alarm is sounded, and power on processing in restart mode is executed. If a wake string is set, the wake string is passed to application initiation processing and used as the program initiation parameters (for CP/M CCP).
- When processing is initiated in power off status in continue mode 2 The power is turned on, alarm is sounded, power on processing in continue mode is executed. A wake string is ignored even if it is specified.
- When processing is initiated in power on status 3 Alarm is sounded and the wake screen shown in Figure 2.6 is displayed.



Fig. 2.6 Wake Screen

**EHT-10** 

s)>

(2) Setting and resetting

Wake is set or reset by calling BIOS TIMDAT. See "Section 4.2 Overview of BIOS Commands" for TIMDAT.

- (3) Initiation
- Wake time comes in power off status. 1
- Wake time comes in power on status. 2 In this case, the wake function is initiated at a timing so that I/Oexecution is not affected.

# (4) End

- 1 If the wake function is initiated in power off status, the same operations executed when the power switch is turned on and the wake function is initiated is executed.
- 2 If the wake function is initiated in power on status, processing is returned to the status held before initiation under one of the following conditions:

  - (a) 50 seconds elapse after initiation.
    (b) The 'Press' (EHT-10) or 'Return' (EHT-10/2) key is pressed.
  - (c) The power switch is turned off.
  - (d) Power failure is detected.

2.5 Sell - Test

2.5.1 Overview

EHT-10/EHT-10/2 performs RAM sum check at power on to improve system reliability and compares the result with the check sum value calculated at power off.

The entire EHT-10/EHT-10/2 RAM areas are classified into the following three groups for RAM sum check:

- 1 System area 2 User area
- 3 RAM disk

Checking the user area and RAN disk can be skipped at power on when no checking is specified by CONFIG.

2.5.2 System area sum check

(1) Areas to be checked

The following system areas are sum-checked:

- 1 RBDOS1 area (80H bytes)
- 2 RBIOSI area (A5H bytes)
- 3 The address after the end of user BIOS area (RAM disk when size 0 is specified) to the address before the stack area used by the system For EHT-10: DCOOH to FCDFH For EHT-10/2: EDOOH to FCDFH This area contains resident processing routines and system parameters.
- 4 Area (FF60H to FFFFH) containing system hook, system jump vector, and interrupt vector
- (2) Checking sum

Simple sum is obtained for each byte in the units of 1 Kbytes. The sum up to the end of the area or remaining area is used as data for sum check if the size of an area or remaining area is less than 1 Kbytes.

(3) Check sum storage area

Check sum data is stored starting from the address determined from the following expression: Start address of RBDOS1 + COH

(4) Handling a sum check error

If a sum check error occurs in a system area, a message indicating "SYSTEM ERROR" is displayed unconditionally and system initialization is executed.

2.5.3 User area sum check

(1) Areas to be checked

The following areas are sum-checked: 1 From address 0 to the address before RBDOS1 2 User BIOS area

(2) Checking sum

Checking sum is executed in the same way as system area check sum.

(3) Check sum storage area

Check sum data is stored starting from the address determined from the following expression: Start address of RBDOS1 + 80H

(4) Handling a sum check error

If a sum check error occurs in an user area, message indicating "RAM SUM CHECK ERROR" is displayed unconditionally and reset processing is executed.

2.5.4 RAM disk sum check

(1) Overview

The RAM disk has a check sum area. Generally sum is determined for each sector when data is written in the RAM disk, and the sum is checked when data is read from the RAM disk.

However, since a part of main RAM is used as the RAM disk, data may be destroyed by an application program or RAM may be destroyed in power off status. Because of this, EHT-10/EHT-10/2 performs sum check of entire RAM disk at system initiation (power on or reset).

(2) Handling a sum check error

A message requesting to format the RAM disk is output if a sum check error occurs in the RAM disk.

In this case, either Yes or No can be selected. Application processing is continued if the status is power on in continue mode. "ERR #nnnn" in the message indicates the sector number (sector number is counted starting from track 0 and sector 0 in ascending order) of the error sector.

(3) Handling a sum check error by an application program

A serious error may occur if application program execution is continued after a sum check error occurs in the RAM disk. Therefore, application programs must detect and handle any RAM disk sum check errors. The sector number of an error sector is stored in system area ERRSEC (F6B7H to F6B8H). An application program can find out error occurrence by referencing this system area. However, the contents of ERRSEC is unknown until the first error occurs. Because of this, the application program must initialize this area in advance with a sector number that does not exist (for example, FFFFH).

SOFTWARE Page 2 - 20

### 2.6 CP/M Operations

2.6.1 Overview

EHT-10/EHT-10/2 OS is the extended CP/M version 2.2. The EHT-10/EHT-10/2 CP/M is structured as shown in Figure 2.7 and has extended device and BIOS functions.

(1) RS-232C, IC card reader/writer, barcode reader, cartridge device (printer unit is supported for the system), and touch panel are supported as the extended devices.

Various BIOS functions are provided to support these devices and unique BIOS functions that takes into account of hand-held terminals are also provided.

(2) RAM disk, ROM socket, IC card, terminal floppy disk drives are supported as the disk drives.

(3) The most part of EHT-10/EHT-10/2 OS is executed in ROM so that a larger user RAM area can be used. Because of this, up to 59.5-Kbyte (default is 28.5-Kbyte) CP/M for EHT-10 and up to 60- Kbyte (default is 29-Kbyte) CP/M for EHT-10/2 can be structured.



Fig. 2.7 CP/M Structure

2.6.2 CP/M structure

(1) Overview

The RAM (bank 0#0) memory map of EHT-10 slightly differs from that of EHT-10/2. These RAM memory maps are shown in Figures 1.6 and 1.7 in Chapter 1.

For EHT-10/EHT-10/2, the main sections of BDOS, BIOS, and CCP are in the system ROM and are not relocated in RAM. Only the BDOS and BIOS entries are relocated in RAM. Entry point RBDOS1 (RBDOS2) or RBIOS1 (RBIOS2) is called

SOFTWARE Page 2 - 21

when the user calls BDOS or BIOS.

When BDOS or BIOS is called, OS switches the bank to pass control to the main part of BDOS or BIOS in the system ROM. See Chapters 4 and 5 for details on calling BDOS and BIOS.

(2) RBDOS and RBIOS

RBDOS1 is functionally same as RBDOS2 and RBIOS1 is functionally same as RBIOS2. RBDOS1 is the jump table to RBDOS2 and RBIOS1 is the jump table to RBIOS2.

RBDOS2 or RBIOS2 calls a processing routine in the main section of BDOS or BIOS stored in system ROM, by using the bank switching routine stored in RSYSPR.

Application programs in load and execute mode (loaded and executed in addresses starting from 100H in RAM) use RBDOS1 and RBIOS1 (general BIOS and BDOS calling).

Application programs in ROM-based mode (operates at addresses from 6000H to DFFFH in bank 2) use RBIOS2 and RBDOS2 (they are at the fixed addresses) because RBDOS1 and RBIOS1 may be in a different bank.

FFFFH	
	RSYSPR
	RBIOS2
FROOH	RBDOS2(256B)
EAOOH	
	RAM Disk
	(EHT-10/2 0 to 40KB)
	RB10S1(256B)
	RBDOS1(256B)

Fig. 2.8 RBDOS and RBIOS

## 2.6.3 CP/M Initiation and Termination

## (1) CP/M initiation

Figure 2.9 shows CP/M initiation processing. (Note) RSYSAR1: System area initialized at system initialization RSYSAR2: System area initialized at reset RSYSAP3: System area initialized at WB00T



SOFTWARE Page 2 - 23

(2) CP/M interrupt and termination

```
1
     When the key defined with key code F6H is pressed:
   <Aim>
     I/O processing is immediately interrupted.
   <Processing>
     The keyboard buffer is cleared and I/O interrupt flag (CSTOPFLG:F22OH)
     is turned on.
     1/0 processing (RS-232C send or receive operation or processing waiting
     for the printer to become ready) is immediately interrupted according
     to the I/O interrupt flag.
2
     The power switch is turned off:
   <Aim>
     The power is turned orf in restart or continue mode and processing is
     terminated or interrupted.
   <Processing>
     I/O processing is continued until processing can be left off, I/O
     processing is interrupted, and the power is turned off. The subsequent
     power on is executed in restart or continue mode.
3
     When the automatic power off time comes:
   <Aim>
     The power is automatically turned off to save electricity if the user
     forgets to turn the power off.
   <Condition>
     Processing must be waiting for key input operation at the specified
     time.
   <Processing>
     The power is turned off in continue mode. The previous processing is
     continued when the power is turned on again. However, processing is
     not continued if the power is automatically turned off during MENU or
     DLL processing.
Δ
     When power failure is detected:
   <Cause>
     The battery voltage is lowered to the specified voltage or less.
   <Processing>
     I/O processing is continued until processing can be left off, I/O
     processing is interrupted, a message indicating "power off" is
     displayed, and the power is turned off in continue mode.
5
    When the reset switch is pressed:
   <Aim>
     The system is cold-started if wild run occurs. However, CP/M may not
     correctly be initiated if the CP/M size, RAM disk, etc are destroyed.
   <Processing>
     Reset processing is executed. The RAM disk and user BIOS area are
     saved but the BIOS entries are initialized.
    When 7508 (slave CPU) is reset:
6
   <Aim>
     The system is initiated if 7508 enters wild run.
   <Processing>
     The entire system including 7508 is initialized.
```

2.6.4 CCP

(1) Overview

For EHT-10/EHT-10/2, the CCP execution program initiation section is extended to have the following characteristics:

- 1 ROM-based programs can be initiated.
- 2 Execution programs (including BASIC files) received by using DLL can be initiated.
- 3 CCP command input operations are deleted so that operations are simplified. Differing from the previous CP/M CCP, EHT-10/EHT-10/2 CCP is not relocated to RAM but the CCP section is operated in the OS ROM.

(2) Error handling

The two errors listed below may occur when CCP initiates a file. If an error occurs, error handling displays an error message and executes warm boot.

- 1 "Bad load error" Cause: The size of the specified file to be initiated was too large and the file could not be stored in the TPA area. User response: Press the 'Press' or 'Return' key to perform warm boot.
- 2 "Command error" Cause: The specified file was not in the specified disk or there was an error in the command line. User response: Press the 'Press' or 'Return' key to perform warm boot.

2.5.5 I/O byte

The I/O byte is used to assign physical devices to logical devices. Table 2.3 shows the I/O byte assignment. A device is assigned by replacing the contents of RAM address 3.

Logical Device	LST:	PUN:	RDR:	CON	:
Bit Location	7,6	5,4	3,2	1,0	
Bit Value	Output	Output	Input	Output	Input
0 0	_	_	Keyboard (Touch key)	RS-232C	Keyboard (Touch key)
0 1	• Cartri- dge	LCD	-	* LCD	Keyboard (Touch key)
1 0	RS-232C	*RS-232C	*RS-232C	LCD	RS-232C
1 1	-	-	-	RS-232C	RS-232C

Table 2.3 I/O Byte Assignment

(Note 1) An asterisk (\*) indicates the default.

If a physical device is assigned to a logical device indicated by (Note 2) a negative sign (-) in the PUN: filed and if characters are output to the device in ready status, nothing is output. If a physical device is assigned to a logical device indicated by a negative sign (-) in the RDR: filed, if the device is ready, and if a character input request is issued, 1AH (CTRL/Z) is sent back.

## 2.6.6 Setting system parameters

Table 2.4 shows the conditions under which system parameters are set.

Continue:	Power on in continue mode
WBOOT:	Warm boot (JMP 0000H is executed by an application program.)
Restart:	Power on in restart mode
Reset:	The reset switch is pressed.
Initialization:	The 7508 reset switch is pressed.
System menu:	Modification using system menu CONFIG

The characters in the table have the following meanings:

S: The default value is set by the system.

(S): The default value is set by the system conditionally.u: This parameter is set by the user.

v: The value specified by the user is set by the system.

-: No modification

Parameter	Default value	Continue  WBOOT   Restart   Reset   Initial   Syste	ization m menu
Date Time Self-test BASIC parameters Calculator parameters	00/00/00 00:00:00 ON (Seanote 1.) Floating-point representation	u u u u u u s u u	
RAM disk size User BIOS size Character set	31KB OKB The default value differs depending on the DIP switch setting (see note 2)	u u u u - v v v u u	
DLL parameters RS-232C parameters Program execute mode	(See note 3.) (See note 3.) AUTO	s u u u u u u	
Power control Printer output Alarm/Wake I/O byte	(See note 5.) Cartridge No setting xx101001 (xx differs depending on the printer output parameters.)	u u s u u s - s -	
Key constant table Display parameters Keyboard status Gaiji (EOH to FFH) Subroutine key table	(See note б.) Normal mode (See note 4.) Only for return operation	- s s s s - - s s s s - - s s s s - - s s s s	
Current drive User BIOS entry	Drive A Only for return operation		

Fig. 2.4 System Parameter Set Timing

(Note 1) The BASIC parameters specify the following items: Number of files that can be opened at the same time (3 files) Maximum record length to be used for random file (128 bytes) RS-232C receive buffer size (256 bytes)

- (Note 2) The default value is changed according to the DIP switch. When Japan is specified: JAPAN When overseas is specified: USA ASCII
- (Note 3) The following initial values are used for communication parameters: 4800 BPS, 8 bits, nonparity, and 2 stop bits
- (Note 4) The gaiji default value differs depending on Japan or overseas specification as follows: Japan: Yen, year, month, and day (The remaining area is filled with blanks.) Overseas: (The entire area is filled with blanks.)
- (Note 5) The following parameters can be specified for power control: Automatic power off time (5 minutes) Printer power save time (3 minutes) Automatic backlight off time (3 minutes) (only for EHT-10/2B) A value enclosed in a pair of parentheses ( ) is the default value.
- (Note 6) The following parameters and default values are used as the display parameters: Screen size: 12 columns x 28 lines (20 columns x 25 lines for EHT-10/2) Scroll mode: Follow mcde Cursor type: Block and blink Attribute: None (normal status)

# CHAPTER 3 APPLICATION PROGRAM CREATION

## 3.1 Overview

This chapter explains how to create EHT-10/EHT-10/2 application programs. For EHT-10/EHT-10/2, the BASIC interpreter is generally stored in ROM so that application programs can be written in BASIC. Further, CP/M Ver. 2.2 is used as the OS so that programs can easily be created in machine language or in a high-level language.

EHT-10/EHT-10/2 application programs are created in the development machine and loaded to EHT-10/EHT-10/2 for execution and debug operations.

Application programs are debugged by using the development tool that is an option. To use this development tool, a computer that supports CP/M or MS-DOS (PC-DOS) as the OS is required as the development machine (refer to EHT-10/EHT-10/2 Development Tool User's Guide for details).

Figure 3.1 shows the overview of application system creation procedure.



Fig. 3.1 System Development Procedure

SOFTWARE Page 3 - 1

#### 3.2 Notes on Designing Application Systems

3.2.1 Programming language selection

For EHT-10/EHT-10/2, application programs can be written in machine language, BASIC, and high-level compiler-type language that can be executed under CP/M. A programming language is selected according to the processing contents of application programs.

Application programs in machine language can be stored in ROM in the format that enables the application programs to be executed in ROM. In this case, RAM space is saved, less power is consumed, and execution is done in high speed (see "Section 3.4 Creating ROM-Based Application Programs" for details).

3.2.2 Program distribution and maintenance

The method of program distribution and maintenance must be determined when application programs are designed. There are the following three ways to distribute and maintain programs:

(1) Programs are distributed and maintained by down-loaded to RAM (TPA or

RAM disk) by DLL or DL using communication lines. In this case, a communication program that down-loads application programs must be created at the host computer side. EHT-10/EHT-10/2 supports the Filink protocol and No protocol as the communication protocols. A protocol can be extended so that the user can create a user protocol. See "Section 11.2 Extending a Communication Protocol" for details. See Appendix for the Filink protocol.

(2) Application programs are stored in ROM for distribution.

By storing fixed data (such as kanji file) and processing in ROM, RAM space is saved and maintenance becomes easier.

(3) Application programs are distributed from IC card.

The master file and programs are stored in an IC card. Programs in IC card can easily be executed because EHT-10/EHT-10/2 can use an IC card as a disk.

Further, the files and programs in an IC card can be directly maintained from the host computer through the communication line.

#### 3.3 Setting Development Environment

3.3.1 Computer used to create application programs

Application programs are created by a development machine (computer) and are down-loaded to EHT-10/EHT-10/2 for debug operations.

The EHT-10/EHT-10/2 debugger (development tool) requires a CP/M or MS-DOS (PC-DOS) machine as a terminal. Refer to EHT-10/EHT-10/2 Development Tool User's Guide for the names of the computers that can be used.

3.3.2 Development software and development procedure

Table 3.1 shows the types of development software required to create application programs.

Language	Type of software		Example		
BASIC	Development tool (WAD)		Programs can also be edited by the general text editor.		
High-level language	Editor and compiler		BDS C Compiler (BD Software Co.) and other compilers that operate under CP/M Ver. 2.2.		
Machine language	Yhen the host OS is CP/M	Editor	Word Master (Micropro Co.) or ED		
		Assembler	MACRO80 (Microsoft Co.)		
		Linker	L80 (Microsoft Co.)		
		Development tool (WAD)			
	When the host OS is MS-DOS	Editor	Word Star (Micropro Co.) or EDLIN		
		Assembler	XMACR080 (Nikkei Co.)		
		Linker	XLINK80 (Nikkei Co.)		
		Development tool (WAD)			

## Table 3.1 Application Development Software

## (1) BASIC

Since the BASIC interpreter is internally built in the OS, a program in BASIC can be directly created and debugged in EHT-10/EHT-10/2 using the development tool. In this case, screen edit operations can be performed by using a development machine as the terminal.

## (2) Machine language

A program in machine language is created, assembled, and linked in the development machine to be generated as an object module (COM file) in execute form. As other CP/M programs, a program in machine language is generally assembled starting from address 100H. However, note that the

SOFTWARE Page 3 - 3

execution start address of a ROM- based program is not address 100H (see Section 3.4). The development tool is used to down-load and debug a program in machine language.

(3) High-level language

The high-level languages that satisfy the following conditions can be executed in EHT-10/EHT-10/2:

1 Compiler-type languages that can be executed under standard CP/M Ver. 2.2.

2 The compilers can generate an object module (COM file) in execute form.

As a program in machine language, the object module of program in high-level language is down-loaded and executed in EHT-10/EHT-10/2.

3.3.3 Storing program data in ROM

To store completed programs and data in ROM, the programs and data are converted in the ROM format by development tool WPROMFRM. If the development machine and the ROM writer are connected through RS-232C, WPROMFRM can directly transfer data to the ROM writer.

## 3.4 Creating ROM - Based Programs

3.4.1 Overview

An application program in ROM socket can be loaded to address 100H in ROM for execution in the same way as the previous CP/M. However, EHT-10/EHT-10/2 can also execute the application program directly in ROM to save memory space and to have less power consumption.

The rest of Section 3.4 explains how a ROM-based program is executed, how to set a ROM-based program, and notes on using a ROM-based program.

3.4.2 Setting

To identify a program that is directly executed in ROM (ROM-based program) from a program that is loaded and executed in ROM (program in load and execute mode), a 5-byte ID must be placed at the beginning of a ROM-based program.

The system determines a ROM-based program from this ID and performs processing for ROM-based program.



3.4.3 Processing flow

Figure 3.2 shows the processing flow until control is passed to a ROM-based program.



Fig. 3.2 Initiating a ROM-based program

3.4.4 Program start address

Since a program in load and execution mode is loaded starting from address 100H in RAM, addressing is started from 100H as a general CP/M file.

On the other hand, a ROM-based program is directly executed in ROM so that addressing must be started from the program start address determined by taking into account of ROM format and ROM in the memory map.

See "Section 6.4 ROM Drive" for ROM format.

The start address of a ROM-based program can be determined from the following expressions: Number of programs stored in ROM: n - Size of each program: Fk (k=1, 2,...n) Rounded up to the unit of 1 Kbvtes. - Number of directories for each program: Fk/16384 (16384 = 16 Kbytes) - Total number of directories: Dnum = F1 + F2 +...Fn - Size of directory area: Dsize = (1 + Dnum) x 32 (Rounded up to the unit of 128 bytes.) Header section (32 bytes). The execution start address of i-th program can be determined as follows:  ${Dsize + (F1 + F2 + ...Fi-1)}/{32768}$ (32768 = 32 Kbytes) Address = (remainder) + 6000H Bank data = (quotient x 10H) + 02H (Quotient is the subbank data.) Figure 3.3 shows the address map used when four files are stored in a 32-Kbyte ROM.

SOFTWARE Page 3 - 6



Fig. 3.3 Example of ROM structure (in EHT-10/EHT-10/2)

# 3.4.5 Calling BDOS/BIOS

A program in load and execute mode generally uses 0005H to 0007H to call BDOS and 0000H to 0002H to call BJOS. However, BDOS and BIOS may be placed at the back of the ROM capsule for a ROM -based program. Because of this, a ROM-based program must call BDOS and BIOS (RBDOS2 and RBIOS2) in the resident section.

RBDOS2	entry address	.FF90H (JP RBD0S2)
RB10S2	BOOT address WBOOT address CONST address I I INFORM address	.EB00H .EB03H .EB06H

SOFTWARE Page 3 - 7

3.4.6 ROM storing ROM-based programs

Formats P and M are provided as the ROM formats of ROM drive for EHT-10/EHT-10/2. ROM to be used to store ROM-based programs must be created in format P.

ROM to be used to store programs in load and execute mode can be either in format P or M.

3.4.7 Work area

A ROM-based program uses a work area different from the one used by a program in load and execute mode.

A program in load and execute mode can freely use areas inside the program and after the program as the work areas. A ROM-based program uses a work area starting from address 100H.

The upper limit of work area of ROM-based program is obtained as follows: MIN {(ROM start address), (RBDOS1 start address)}

The ROM start address is 6000H and the RBDOS1 start address is determined from the contents of addresses 6 and 7.

3.4.8 32-Kbyte or more ROM-based program

In EHT-10/EHT-10/2, ROM is allocated on the memory map for each subbank in the units of 32 Kbyres (see "Section 1.4 Memory Map"). One ROM-based program must not be extended over two or more subbanks. If the program size is more than 32 Kbytes, the following processing is required:

(1) Each module in the program must not be extended over bank end address (DFFFH in memory map).

(2) Dummy data must be inserted so that the end address of the last module in a bank is the subbank end address.

(3) The module next to the one explained in (2) must be addressed starting from address 6000H.

(4) JSCALLX (FF99H) must be used to perform subroutine call operations between banks (parameters same as that of BIOS CALLX are used).



Fig. 3.4 Structure of ROM-based file stored in 2banks

## CHAPTER 4 BIOS OVERVIEW

## 4.1 Overview

4.1.1 Characteristics of EHT-10/EHT-10/2 BIOS

As BDOS, the EHT-10/EHT-10/2 BIOS processing is performed mainly in system bank OSROM. Two BIOS RAM entries are provided so that an ROM-based program can use BIOS without taking into account of banks.

For EHT-10/EHT-10/2, the standard BIOS is extended so that communication with serial devices, touch panels, IC cards, and public lines can easily be done. Further, BIOS can be extended by the user since the user BIOS and BIOS hook are provided. See Section 4.6 for the user BIOS and Chapter 10 for BIOS hook.

- 4.1.2 BIOS processing
- (1) Processing flow

Processing flows as follows (Figure 4.1) when an application program calls EHT-10/EHT-10/2 BIOS:

- 1 The bank is switched to the system bank at the BIOS section in RAM.
- 2 The actual GIOS in OS ROM is called.
- 3 After BIOS processing, various return information and data are saved and control is returned to the bank used before BIOS is called.

An application program calls BIOS as follows:

- 1 A load-and-execute program obtains the BIOS address from the JMP WBOOT placed at address 0000H to call BIOS.
- 2 A ROM-based program directly calls the BIOS jump table placed in RBIOS2 (EBOOH to EBFFH).

The functions of BIOS are the same for programs in load and execute mode and for programs in ROM-based mode.



Fig. 4.1 BIOS Processing Flow SOFTWARE Page 4 - 2

(2) PREBIOS and PSTBIOS

EHT-10/EHT-10/2 BIOS processing is based on the PRE/PST BIOS concept to have higher system reliability.

If interrupt processing is executed immediately after an interrupt occurs during BIOS processing, the program execution may not be continued after control returns from interrupt processing. Therefore, interrupt processing is suppressed before actual BIOS processing is started and interrupt processing for interrupts that occurred during BIOS processing is executed after BIOS processing is completed. PRE/PST BIOS controls these operations.

PRE/PST BIOS is automatically executed when BIOS is called from a BIOS entry placed in RAM.

- PREBIOS PREBIOS sets the flag indicating BIOS processing in progress, the flag indicating alarm suppressed, and the flag indicating power off suppressed.
- 2 PSTBIOS PSTBIOS resets the flags set by PREBIOS and performs alarm or power off processing if alarm or power off has been requested during BIOS processing.

4.1.3 Notes on using BIOS

(1) The contents of registers other than the one used to store return parameters are not guaranteed unless other wise stated. The contents of the required registers must be saved in advance.

(2) The addresses of two BIOS entries are used as follows:

- 1 To indicate a BIOS entry address with an offset value based from WBOOT, the contents (WBOOT entry address) of addresses 001H and 0002H are fetched to determine the BIOS entry address.
- 2 There is no specific problems when a BIOS entry address is indicated with a fixed address. However, study the maps in RAM carefully to convert previous CP/M application programs.

### 4.2 Overview of BIOS Commands

This section explains each BIOS command. Each command is explained according to the conventions explained below. See Appendix 5 for the list of BIOS commands.

- Command name
   The command name is written with upper-case letters at the top left of
   a page. The commands are explained in the ascending order of
   addresses. If a command has subcommands, the subcommand names are
   written enclosed in a pair of parentheses ( ) after the command name.
- (2) Entry addresses The offset address from WBOOT and the absolute address are written at the top right of a page.
- (3) [Function] [Function] explains the overview of a BIOS command.
- (4) [Entry parameters] [Entry parameters] explains the input parameters required to call a BIOS routine.
- (5) [Return parameters] [Return parameters] explains the parameters set when BIOS routine execution is ended.
- (6) [Saved registers] [Saved registers] lists the registers of which the contents are saved even if BIOS is used. [Saved registers] is not written unless there is one or more saved registers.
- (7) <Explanation>
   <Explanation> explains the functions and usages of each BIOS command.
- (8) <Relations>
   <Relations> explains other related BIOS command.
- (9) <Reference> <Reference> explains sections to be referenced.

```
[Function]
BOOT performs CP/M cold boot.
[Entry parameters]
None
[Return parameters]
C=00H
```

## <Explanation>

The following operations are performed when BOOT is executed:

 The current drive is changed to A:.
 The I/O bytes are initialized.
 The keyboard standard (for Japan or overseas) and the display character set are initialized according to the DIP switches.
 RBIOS1 and 2 are loaded.

The rest of operations are the same as warm boot (see WBOOT).

(2) BOOT is used by the system for system initialize, reset, or 7508 reset operation but not by the user.

<Relations>

WBOOT (WBOOT+O)

<Reference>

Section 2.6 CP/M Operations

[FUNCTION] WBOOT performs CP/M warm boot. [Entry parameters] None [Return parameters] C=drive number

# <Explanation>

- The following operations are performed when WBOOT is executed:

   The display parameters are initialized.
   The key tables are initialized.

   The rest of operations are the same as BOOT.
   I/O close processing is executed.
   The jump address of BDOS WBOOT is set.
   RBDOS1, and 2 are loaded.
   RSYSAR3 is loaded.
   MENU or DLL is initiated.
- (2) WBOOT is initiated to branch out of an application when address JP=0000H is executed or the power is turned on in restart mode.

<Relations>

BOOT (WBOOT-03H)

<Reference>

Section 2.6 CP/M Operations APPENDIX 9 SAMPLE-LIST

SOFTWARE Page 4 - 6

[Funct] C c	ion] CONST ch current	ecks CON:	whether (default	data is t is the	input fro keyboard	m the do or toucl	evice al n panel)	located	as 1	the
[Entry N	parame lone	ters]								
(Retur A A	n param =00H: =FFH:	eters Conso No co	i] ble input bnsole in	t iput						

<Explanation>

- (1) The current CON: is determined according to the I/O bytes.
- (2) "No console input" is determined for undefined keys or keys without codes if the keyboard or touch panel is allocated as the current CON:. For EHT-10, BIOS TOUCH or PUTPFK must be executed at first to define keys in order to use CONST and CONIN because all keys are defined ineffective (function code=FFH) after an application program is initiated.
- (3) If the keyboard or the touch panel has been allocated as the current CON: and a function call key is pressed, the key operation corresponding to the key is executed, key input is checked, and then the status is set in register A. The function call keys have the key function codes from EOH to FFH used to call subroutines. See "Section 4.3 Key Input" for details.
- (4) The status of receive buffer is returned if the current CON: input is to RS-232C.
- (5) If the current CON: input is RS-232C, and if the user has opened the cartridge serial by using BIOS RSIOX or has opened an IC card by using BIOS ICCARD, A=OOH is set as the return parameter of CONST.
- (6) If CON: input is RS-232C and the user is using the RS-232C, the transmission-mode parameter specified by the user at open operation is used.
- (7) See Section explaining PUNCH to modify the RS-232C transmission mode.

<Relations>

CONIN (WBOOT+06H) PUNCH (WBOOT+0FH) PUTPFK(WBOOT+6CH) TOUCH (WBOOT+93H) KEYIN (WBOOT+99H) <Reference>

Section 2.6.5 I/O bytes Section 4.3 Key Input (Touch Panel/Keyboard)
```
[FUNCTION]
    CONIN inputs one character from the device allocated as the current
    CON: (default is the keyboard or touch panel) and sets the character in
    register A. Processing waits until data is input if there is no input
    data.
[Entry parameters]
    None
[Return parameters]
    A=input data (function code)
    C=position code
<Explanation>
(1) The current CON: is determined according to the I/O bytes.
(2) For EHT-10, BIOS TOUCH or PUTPFK must be executed at first to define
    keys in order to use CONST and CONIN because all keys are defined
    ineffective (function code=FFH) after an application program is
    initiated.
(3) If CON: input is the keyboard or the touch panel, the sleep mode and
    automatic power off functions are operated. See "Section 2.3.5 Sleep
    function and Automatic Power Off
                                        function" for details.
(4) If a subroutine call key that has a function code from EOH to FFH is
    pressed, the corresponding subroutine is executed and then processing
    waits for key input operation.
```

- (5) The return parameter in register C is meaningless if CON: input is RS-232C.
- (6) If CON: input is RS-232C, and if the user has opened the cartridge serial by using BIOS RSIOX or the IC card by using BIOS ICCARD, control returns without executing any operation. In this case, the return parameters are meaningless.
- (7) If CON: input is RS-232C and the user is using the RS-232C, the transmission-mode parameter specified by the user at open operation is used.
- (8) See Section explaining PUNCH to modify the RS-232C transmission mode.

<Relations>

CONST (WBOOT+03H) PUNCH (WBOOT+0FH) PUTPFK(WBOOT+6CH) TOUCH (WBOOT+93H) KEYIN (WBOOT+99H) <Reference>

Section 2.3.5 Sleep function and Automatic Power Off function Section 2.6.5 I/O bytes Section 4.3 Key Input (Keyboard/Touch Panel) APPENDIX 9 SAMPLE 2

[Function] CONOUT outputs one character to the device allocated as the current CON: (default is LCD). [Entry parameters] C=output data [Return parameters] None <Explanation> (1) The current CON: is determined according to the I/O bytes. (2) Display operations can be controlled with control codes (OOH to 1FH) and the ESC sequences if CON: output is LCD. See "APPENDIX 6 DISPLAY CONTROL FUNCTIONS" for details on control code and ESC sequence functions. 1 The control codes, ESC sequences, and character codes are classified as follows: Control codes: OOH to 1FH 1BH (ESC) + n1 + n2 +...nk ESC sequences: Character codes: 20H to FFH 2 In an ESC sequence, CONOUT is called as may times as the number of parameters.

3 Control codes and ESC sequences can be used to control not only LCD but also keyboard (touch panel) LED and buzzer.

4 If an ineffective control code or an ESC sequence parameter error is detected, no operation is executed and the original status is guaranteed. ESC sequence parameters are checked after all parameters are received. See "Section 4.4 LCD Display" for the LCD display functions.

(5) Processing same as PUNCH is executed if the CON: device is RS-232C.

<Relations>

PUNCH (WBOOT+OFH) GRAPHICS(WBOOT+90H) TOUCH (WBOOT+93H) KANJI (WBOOT+99H)

<Reference>

Section 2.6.5 I/O bytes Section 4.4 LCD Display APPENDIX 6 DISPLAY CONTROL FUNCTIONS

[Function] LIST outputs one character to the device allocated as the current LST (default is the cartridge I/F).
[Entry parameters] C=output data
<pre>[Return parameters] Only when a printer unit is mounted, (LSTERR)=00H: Normal termination ≠00H: Abnormal termination (indicates a printer unit not mounted or forced system termination.)</pre>
<explanation></explanation>

- (1) The current LST: is determined according to the I/O bytes.
- (2) Processing waits until the LST: device becomes ready if it is not ready. However, if LST: is a cartridge I/F and a printer unit is not mounted, error information is set in LSTERR and control returns.
- (3) See Section explaining BIOS INFORM for the LESTER address.
- (4) Print and other operations can be executed concurrently by using the spooling function if a printer unit has been allocated as the LST: device. See "Section 4.5 Printer" for details.
- (5) If LST: device is RS-232C, and if the user has opened the cartridge serial by using BIOS RSIOX or the IC card by using BIOS ICCARD, control returns without executing any operation.
- (6) If the LST: device is RS-232C and the user is using the RS-232C, the transmission-mode parameter specified by the user at open operation is used.
- (7) See Section explaining PUNCH to modify the RS-232C transmission mode.

<Relations>

PUNCH (WBOOT+OFH) LISTST(WBOOT+2AH) SCRNDUMP(WBOOT+33H) RSIOX (WBOOT+51H) ICCARD(WBOOT+96H) KANJI (WBOOT+9CH) INFORM(WBOOT+A2H)

<Reference>

Section 2.6.5 I/O bytes Section 4.5 Printer APPENDIX 9 SAMPLE 3

LIST

[Function] PUNCH outputs one character to the device allocated as the current PUN: (default is RS-232C).

```
[Entry parameters]
C=output data
```

```
[Return parameters]
None
```

<Explanation>

- (1) The current PUN: is determined according to the I/O bytes.
- (2) If PUN: device is RS-232C, and if the user has opened the cartridge serial by using BIOS RSIOX or the IC card by using BIOS ICCARD, control returns without executing any operation.
- (3) If the PUN: device is RS-232C and the user is using the RS-232C, the transmission-mode parameter specified by the user at open operation is used.
- (4) Transmission mode default value and how to modify the mode when the physical I/O device is RS-232C, the same transmission mode is used for BIOS CONIN (CONST), CONOUT, and LIST (LISTST, SCRNDUMP, KANJI printout) because a transmission mode is defined in the same system area. The transmission mode for these BIOS commands is modified by CONFIG. The transmission mode can also be modified by a user program rewriting the following system area:

```
(Example)
Address: Variable name(Byte length)
```

FO10H : SRSADR (2+2) First 2 bytes specifies the Start address of receive buffer (initial value: COMBUF F95AH) Second 2 bytes specifies the Size of receive buffer (initial value: 0100H)

F014H : SRSPAK (5) Each byte specifies the following parameter. Transmission speed (initial value ODH .....4800 BPS) Bit length (initial value: 03H .....8 bits) Parity (initial value: 00H .....nonparity) Stop bit (initial value: 03H .....2 stop bits) Special parameter (initial value: FFH)

1 The parameter is structured in the same way as RSIOX open operation.

2 RS-232C must be closed by BIOS RSIOX after modification when this parameter is modified by an application program.

<Relations>

CONST (WBOOT+O3H) CONIN (WBOOT+O6H) CONOUT(WBOOT+O9H) LIST (WBOOT+OCH) LISTST(WBOOT+2AH) SCRNDUMP(WBOOT+33H) RSIOX (WBOOT+51H) ICCARD(WBOOT+96H) KANJI (WBOOT+9CH)

<Reference>

Section 2.6.5 I/O bytes Section 7.2 Serial Interface APPENDIX 9 SAMPLE3 .

<Explanation>

- (1) The current RDR: is determined according to the I/O bytes.
- (2) Processing waits until data is input if there is no input data.
- (3) Processing is interrupted to perform power off or alarm operation if power off or alarm status occurs while processing is waiting for input data.
- (4) The return parameter output when the ROR: device is RS-232C is some as the one output by RSIOX Get.
- (5) The transmission mode used when RS-232C is specified is same as the transmission mode used for PUNCH. The default values of transmission mode are 4800 BPS, 8 bits, and nonparity. The communication parameters are modified by system menu CONFIG.
- (6) 1AH (EOF) is set in register A if an I/O device (DTR or UR2) not supported for RDR: is allocated as the RDR:.

<Relations>

PUNCH (WBOOT+OFH)

<Reference>

Section 2.6.5 I/O bytes Section 7.2 Serial Interface APPENDIX 9 SAMPLE 3

.

```
[Function]
	HOME sets the disk seek track to 0.
[Entry parameters]
	None
[Return parameters]
	None
```

<Explanation>

(1) For a floppy disk, the contents of blocking buffer are written and then the track is set to 0.

(2) HOME does not actually move the head to track 0.

<Reference>

CHAPTER 6 OVERVIEW OF DISK SYSTEM

- (1) A parameter error occurs if a disk is not connected.
- (2) The correspondence between logical and physical drives can be freely changed for EHT-10/EHT-1C/2. See "CHAPTER 6 OVERVIEW OF DISK SYSTEM" for details.

<Reference>

[Function] SETTRK	specifies	a	track	for	read/write	operations.
[Entry param BC=trac	neters] :k number					
[Return para None	ameters]					

The parameter is not checked. However, an error occurs during read/write operation if a value out of the allowed range is specified. The following tack numbers are allowed for drives:

Physical drive	Logical drive	Range	Remarks
RAM disk	A:	0 <bc<28< td=""><td>Maximum may vary.</td></bc<28<>	Maximum may vary.
ROM Socket	8:	0 <bc<15< td=""><td>Maximum may vary.</td></bc<15<>	Maximum may vary.
IC card	C:	0 <bc< 7<="" td=""><td>Maximum may vary.</td></bc<>	Maximum may vary.
External disk drive	D: E:	0 <bc<39< td=""><td>Maximum may vary.</td></bc<39<>	Maximum may vary.

Table 4.1 Track numbers for drives

<Reference>

CHAPTER 6 OVERVIEW OF DISK SYSTEM APPENDIX 9 SAMPLE4

- (1) The allowed range of sector numbers is 0 < C < 63.
- (2) An error occurs during read/write operation if a sector number out of the allowed range is specified.

<Reference>

[Function]
 SETDMA specifies the start address of 128-byte data area to be read or
 written.
[Entry parameters]
 BC=DMA start address
[Return parameters]
 None

<Explanation>

- (1) A DMA buffer address is specified. The DMA buffer must be reserved in main RAM.
- (2) The DMA buffer receives or sends data during read/write operation.

<Reference>

```
<Explanation>
```

- (1) READ reads 128-byte data from a disk according to the parameters set by SELDSK, SETTRK, SETSEC, and SETDMA.
- (2) The return parameter output at FDD access abnormal termination has the following meaning:

```
A=FAH : read error
FBH : write error
FCH : select error
FDH : read only disk or write protect
```

(3) The error information of disk read/write operation executed by the called BIOS can also be referenced from the following area:

Address : Variable name(Byte length)

F471H : BIOSERROR (1) BIOS return code =00H:Normal termination =01H:Read Error =02H:Write Error =03H:Write Protect Error =04H:Time out or Communication Error =FEH:Others

See Section explaining BIOS INFORM for the BIOSERR address.

<Relations>

INFORM (WBOOT+A2H)

<Reference>

[Function] WRJTE writes 128-byte data. [Entry parameters] C=writing specification (only for floppy disk) =00H: Standard writing (blocking) =01H: Forced writing (no blocking) =02H: Sequential file writing [Return parameters] A=return information =00H: Normal termination \$00H: Abnormal termination

<Explanation>

- (1) As READ, WRITE writes 128-byte data in a disk according to the set parameters.
- (2) The return parameters indicating abnormal termination and error information returned in BIOSERROR are same as the ones of BIOS READ.

<Relations>

READ(WBOOT+24H)

<Reference>

```
[Function]
    LISTST checks the use status of the device allocated as the current
    LST: (default is printer unit).
[Entry parameters]
    None
[Return parameters]
    A=printer status
     =FFH: Ready (Data can be output to the LST: device.)
     =OOH: Busy (Data cannot be output to the LST: device.)
    B=buffer status (only for printer unit. Details are explained later.)
     =FFH: No data (empty)
            There is data.
     =00H:
     (LSTERR) = error status
             =00H: Normal
            ≠00H: Printer unit not mounted
```

- (1) The current LST: is determined according to the I/O bytes.
- (2) When the current LST: device is a printer unit, 128-byte printer buffer is provided for LST: device output operation so that multiprocessing with printer output operation can be executed. The buffer status indicates whether dats to be output is in the printer buffer.
- (3) When the current LST: is a printer unit and the output buffer is not full, printer ready is set in register A and control returns.
- (4) See the explanation of BIOS INFORM for the LSTERR address.

<Relations>

INFORM(WBOOT+A2H)

<Reference>

Section 2.6.5 I/O bytes Section 4.5 Printer APPENDIX 9 SAMPLE 3 .

[Function]

SECTRN translates a logical sector to a physical sector.

[Entry parameters] BC=logical sector

[Return parameters] HL=physical sector

### <Explanation>

.

SECTRN only copies the register contents because logicalsectors are equal to the physical sectors for EHT-10/EHT-10/2.

(1) The current LST: is determined according to the I/O bytes.

- (2) The contents of the current LCD section in VRAM are output to LST: in bit image. For EHT-10, no operation is is executed and control returns.
- (3) See the explanation of BIOS INFORM for the LSTERR address.
- (4) Processing waits until the LST: device becomes ready if it is not ready. However, if LST: is a cartridge I/F and a printer unit is not mounted, error information is set in LSTERR and control returns.

<Relations>

INFORM(WBOOT+A2H)

<Reference>

Section 2.6.5 I/O bytes Section 4.5 Printer

```
[Function]
    BEEP sounds the buzzer.
[Entry parameters]
     There are the following three ways to specify the BEEP entry
     parameters:
     (1) Software beep
         1 Tone specification
           B=tone (13 < B < 60. No sound for B=0)
           C=period (1 < C < 255. The unit is 100 ms.)
         2 Frequency specification (Turn the MSB of register B to 1.)
           C=period (1 < C < 255. The unit is 100 ms.)
           D=small loop counter
           E=large loop counter (No sound for DE=0)
     (2) Hardware beep
           B=tone
             =100 (64H): 512 Hz
            =101 (65H): 1024 Hz
            =102 (66H): 2048 Hz
           C=period (1 < C < 255. The unit is 100 ms.)
[Return parameters]
    A=00H: Normal termination
     =FFH: Forced termination due to alarm or power off.
```

- (1) EHT-10/EHT-10/2 supports the software and hardware beeps. For software beep, the sound in the specified frequency is generated by the software timer and processing cannot branch out of the BEEP routine until the sound stops. Details on specification are explained later. For hardware beep, the sound in the fixed frequency is generated by hardware and processing can branch out of the BEEP routine after the buzzer is turned on and the timer is set. Beep sound generation is monitored by using 1-ms/8-ms timer.
- (2) Software beep is specified as follows: 1 The small and large loop values are set in registers D and E. The following expressions are used to determine values D and E: T1 = {13 \* (D-1) + 8} / (3.68 \* 10<sup>6</sup>) Sec T2 = {3307 \* (E-1) + 240} / (3.68 \* 10<sup>6</sup>) Sec Period T = 2(T1 + T2) Sec Frequency f = 1 / T Hz

```
2 Buzzer has the compass range of 200 to 4000 Hz.
```

3 The table below shows the relationship between tone specification value and actual tone. The values enclosed in a pair of parentheses ( ) are the large and small loop values.

	0	1	2	3	4
С	1	13	25	37	49
	(OSH, 30H)	(03H, 0FH)	(01H, FCH)	(01H, 75H)	(01H, 31H)
C ł	2	14	26	38	50
	(04H, F1H)	(02H, EFH)	(01H, EDH)	(01H, 6DH)	(01H, 2DH)
D	3	15	27	39	51
	(04H. B8H)	(O2H, D2H)	(01H, DFH)	(01H, 66H)	(01H, 2AH)
DI	4	16	28	40	52
	(O4H, 82H)	(O2H, B7H)	(01H, D1H)	(01H, 5FH)	(01H, 26H)
F	5	17	29	4 1	53
-	(04H, 4FH)	(O2H, 9DH)	(01H, C4H)	(01H, 59H)	(01H. 23H)
F	6	18	30	4 2	54
	(O4H, 1EH)	(02H, 85H)	(01H, BAH)	(01H, 53H)	(01H, 20H)
FI	7	19	31	43	55
•	(O3H, EFH)	(O2H, 6EH)	(01H, ADH)	(01H, 4DH)	(01H, 10H)
G	8	20	32	44	56
-	(O3H, C4H)	(02H, 59H)	(01H, A2H)	(01H, 4AH)	(01H, 18H)
GI	9	21	33	4 5	57
_	(O3H, 9BH)	(028, 448)	(01H, 98H)	(018, 438)	(01H, 18H)
A	10	• 2 2	34	46	58
	(O3H, 75H)	(02H, 31H)	(01H, 8FH)	(01H, 3EH)	(01H, 16H)
AI	11	23	35	47	59
///	(03H, 51H)	(02H, 1FH)	(01H, 85H)	(01H, 39H)	(01H, 13H)
A	12	24	36	48	60
U	(03H, 2FH)	(02H, 0EH)	(01H, 7DH)	(01H, 35H)	(018, 118)

Table 4.2 Software E3ep Tone (large and small loops)

Note) An asterisk (\*) indicates 440-Hz tone.

(3) By changing the system area, interrupt suppressed/allowed status for buzzer sounding period can be modified. The address can be found out by the BIOS INFORM.

Address : Name(Byte length)

F23BH : BPINTEBL (1) Interrupt control flag for buzzer sounding period Bit 7: 1-sec interrupt 1 = suppressed, 0 = no change Bit 6,5: Fixed to 0 Bit 4: EXT interrupt 1 = suppressed, 0 = no change Bit 3: Fixed to 0 Bit 2: ICF interrupt 1 = suppressed, 0 = no change

Bit 1: ART interrupt 1 = suppressed, 0 = no change Bit 0: Key interrupt 1 = suppressed, 0 = no change OVF interrupt is always suppressed.

- (4) No sound is generated when O is specified in register B for tone.
- (5) 'No sound is generated when O is specified in registers D and E for frequency.
- (6) Processing is terminated when function key F6H is pressed and alarm or power off occur.
- (7) While buzzer is sounded, the cursor is not blinked to prevent the tone from warping caused by OVF interrupt. Cursor blinking can be controlled for other interrupts for the same reason.

<Reference>

APPENDIX 9 SAMPLE 5

[Function]
TIMDAT supports the clock functions. The following nine functions are
provided depending on the value of register C:
C=OOH: Reads time. (Read Time)
=FFH: Sets time. (Set Time)
=80H: Allows alarm/wake. (A/W enable)
=81H: Suppresses alarm/wake. (A/W disable)
=82H: Sets alarm/wake. (Set A/W)
=83H: (TMDT83 hook)
=84H: Reads alarm/wake. (Read A/W)
=85H: (TMDT85 hook)
=86H: (TMDT86 hook)
No operation is executed when register C contains a value other than
above values.
Details on each function are explained later.
Registers D and E are saved for each function.
See "Section 10.2 System Hook" for TIMDAT hook (TMDT83, TMDT85, TMDT86).

Г

(1) The time descripter is used as the parameters when time, or alarm/wake is set or read.

Figure 4.2 Time descripter

+0	Year (lower two digits) 2-digit BCD	1	byte Year: 00 to 99
	Month 2-digit BLU	1	
2[	Day 2-digit BCD	1	byte Day: OI to JI
3	Hour 2-digit BCD	1	byte Hour: OO to 23
4	Minutes 2-digit BCD	1	byte Minute: 00 to 5916
5[	Second 2-digit BCD	1	byte Second: OO to 59
6	Day of the week	1	byte Day of the week
	-		(00: Sun, 01:Mon,06: Sat)
7 [	Type (Note 1)	3	byte
8	Address (Note 2)	2	bytes
9			
10	Status (Note 3)	1	byte

(Note 1) Type....The alarm/wake type is set. =00H: No setting. =01H: Alarm setting =02H: Wake setting

(Note 2) Address.....The start address of alarm message or wake string is specified.

<--1 byte--> <--Up to 32 bytes-->

	Length	Message (	or string	
^			_	
1				
	Start addre	224		

The length of data that can be displayed is 20 bytes for EHT-10 and 18 bytes for EHT-10/2. A control code (OOH to 1FH) is counted as two bytes. (Note 3) Status.....The alarm generation status is indicated. =00H: No generation =01H: Generated (2) The lower two digits of dominical year is set as the year. 24-hour clock is used for time indication. (3) Once set correctly, the parameters from year to the day of week are automatically adjusted for 1901 to 2099 including leap years. (4) The system uses the following area to manage alarm/wake data: Address : Name(Byte length) FO28H : ALRMTP (1) Alarm/wake set type =00H: No setting (default) =01H: Alarm setting =02H: Wake setting - This area is referenced by alarm/wake execution processing. FO29H : ALRMAD (2) Start address of alarm/wake message - This indicates ALRMMSG (F32FH). FO2BH : ALRMST (1) Alarm/wake generation status =00H: No generation (default) =01H: Generated - OOH is set by TIMDAT "Set Alarm/Wake" or "Read Alarm/Wake" and OlH is set at alarm/wake interrupt occurrence. F32FH : ALRMMSG (34) Alarm/wake message storage area The first byte contains the message length. - The message specified by TIMDAT "Set Alarm/Wake" is stored. <Reference> Section 10.2 System Hook APPENDIX 9 SAMPLE 6

- (1) Time is set according to the 7-byte data indicating year, month, day, hour, minute, second, and day of week specified by the time descripter.
- (2) When 1111B is specified in the time descripter digit that is not required to be updated, the previously set value of this digit remains effective.
- (3) Since parameter validity is not checked, the clock contents are not guaranteed if logically incorrect data is specified.

<Reference>

APPENDIX 9 SAMPLE 6

The current time (year, month, day, hour, minute, second, and day of week) is returned in the time descripter beginning from the start address.

### <Reference>

APPENDIX 9 SAMPLE 6

- (1) This command executes the alarm/wake operation at the set time.
- (2) The system automatically enables the alarm/wake function when Set Alarm/Wake operation is executed.

## <Relations>

TIMDAT (Set Alarm/Wake)

```
[Function]
	TIMDAT (Alarm/Wake Disable) disables the alarm/wake function.
[Entry parameters]
	C=81H: Function number
[Return parameters]
	None
[Saved registers]
	DE
```

- (1) Alarm/wake operations can no longer be executed after this command is executed.
- (2) The alarm/wake operation data is saved even if Alarm/Wake Disable is executed. This alarm/wake data becomes effective when Alarm/Wake Enable is executed again.

<Relations>

TIMDAT (Alarm/Wake Enable)

[Function] TIMDAT (Set Alarm/Wake) sets the time for alarm/wake operations. [Entry parameters] C=82H: Function number DE=start address of time descripter [Return parameters] None [Saved registers] DE

## <Explanation>

- 10-byte data in time descripter starting from year to address is set. However, 2-digit year and the lower digit of 2-digit second are ignored (10 is the minimum value that can be set for seconds).
- (2) If all 1 is set as a BCD digit, the digit is assumed equal to any number from 0 to 9. For example, all 1 is set to the month, day, and day of week digits and specific values are set to the hour, minute, and second digits, alarm/wake operations are executed at the specified hour, minute, and second every day.
- (3) The alarm/wake function is automatically enabled after this command is executed.
- (4) Since parameter validity is not checked, alarm/wake processing is not guaranteed if logically incorrect data is input. Especially, data other than OOH to O2H must not be specified as the type.

```
[Function]
	TIMDAT (Read Alarm/Wake) reads the alarm/wake time.
[Entry parameters]
	C=84H: Function number
	DE=start address of time descripter (11-byte area is required).
[Return parameters]
	The alarm/wake status is set in the time descripter.
[Saved registers]
	DE
```

When this command is executed, the current alarm/wake time is set in the time descripter.

However, all 1 is set as the year and as the lower digit of the second because these are not included in the set alarm/wake time.

[Function] MEMORY checks the current bank information. [Entry parameters]

[Return parameters] C=bank information

<Explanation>

None

An application program calls this routine to find out the bank in which the application program is being executed. The bank information is indicated as follows:

Bank information

Register C

	/	6	5	4	3	2	1	0
L		<u> </u>				<u> </u>		
1	ir	for	natio	on	- r	for	natio	on

Bank	information	Bank number	Explanation
	FFH	System bank	O to 7FFFH in system ROM
	00H	0#0	Bank in Standard RAM
	10H 20H 30H 40H 50H 60H	0#1 0#2 0#3 0#4 0#5 0#6	Banks in extended RAM
	11H 21H 31H	1#1 1#2 1#3	Banks in system ROM
	02H 12H 22H 32H	2#0 2#1 2#2 2#3	Banks in application ROM

# Fig. 4.3 Correspondence between bank information and bank

<Reference>

Section 1.4 Memory Map

[Fund	ction]	
-	RSIOX p	erforms serial communication. RSIOX has the following 10
	function	ns depending on the value in register B:
	8=1XH:	Opens device. (Open)
		Lower 4 bits: Device specification
		=U: KG-ZGZL
	-204.	=3: Cartridge SIU
	-20N.	Checks whather data has been received in the receive huffer
	(Insts)	CHECKS WHELHET DALA HAS DEEN FECETAED IN THE FECETAE DUITET.
	=40H:	Checks whether sending is possible. (Outst)
	=50H:	Receives 1-byte data from the receive buffer. (Get)
	=60H:	Sends 1-byte data. (Put)
	=70H:	Reads the control line status. (Ctlin)
	=80H:	Sets a control line. (Setctl)
	=90H:	Reads error status and clears error flags. (Ersts)
	=FOH:	Checks the current serial device use status. (Sens)
	Details	on each function are explained later.
	_	
<exp]< td=""><td>anation</td><td>&gt;</td></exp]<>	anation	>
(1)	Two type	es of interfaces (RS-232C and cartridge SIO) are provided for

(1)	EHT-10/EHT-10/2 serial communication. However, these two interfaces cannot be used at the same time and only one device can be used. Because of this, OS uses one device efficiently. See "Section 7.2 Serial Interface" for details.
(2)	The list below shows the system areas used by RSIOX.
Addr€	ess : Variable name(Byte length)
FOOD	I : RSXON (1) XON code used when XON/XOFF is specified. The initial value is 13H.
FOOEH	I : RSXOFF (1) XOFF code used when XON/XOFF is specified. The initial value is 11H.
F604H	A : RSPSTS (1) RSIOX status flags Bit 7: DSR status (0: Active) Bit 6: Framing error status (1: Error) Bit 5: Receive overrun status (1: Overrun) Bit 4: Parity bit error status (1: Error) Bit 3: CD status (1: Active) Bit 2: Receive buffer overflow status (1: Overflow) Bit 1: Receive buffer status (1: Full) Bit 0: Open status (1: Open)
F605H	i : RSPRBGP (2) Receive buffer get pointer

```
F607H : RSPRBPP (2)
     Receive buffer put pointer
F609H : RSPRBAD (2)
     Start address of receive buffer
F60BH : RSPRBSZ (2)
     Size of receive buffer
F60DH : RSPBITR (1)
     Bit rate (parameter set at open operation)
     =02H:110bps =0AH:1200bps =10H:38400bps
     =04H:150
                     =0CH:2400
                                    =80H:75/1200
     =05H:200
                   =0DH:4800
                                     =81H:1200/75
     =06H:300
=07H:600
                    =0EH:9600
                    =0FH:19200
F60EH : RSPBITL (1)
     Bit length (Parameter set at open operation)
     =02H:7bits
                    =03H:8bits
F60FH : RSPPAR (1)
     Parity (parameter set at open operation)
     =OOH:NON =OIH:ODD
                                    =02H:EVEN
F610H : RSPSTOPB (1)
     Stop bit (parameter set at open operation)
     =01H:1bit
                   =02H:2bits
F611H : RSPSPP (1)
     Special parameter (parameter set at open operation)
     Bit 7: Unused
Bit 6: RTS/CTS control (0: Control)
BIt 5: DTR/DSR control (0: Control)
     Bit 4: XON/XOFF control (0: Control)
     Bit 3: Unused
     Bit 2: SI/SO control (1: Control)
     Bit 1: RTS control (1: Active)
Bit 0: DTR control (1: Active)
     - XON/XOFF, RTS/CTS, and DTR/DSR are used for buffer control. Only one
     of these can be specified.
F612H : RSRBFAD (2)
     Receive buffer end address + 1
F614H : RSXONSZ (2)
     Receive data length at the beginning of XON code sending (Receive
     buffer size/4)
F616H : RSXOFSZ (2)
     Receive data length at XOFF code sending (Receive buffer size x 3/4)
F618H : CHR5MSK (1)
     Receive data mask pattern
     7FH...6-bit length, FFH...8-bit length
```

```
F619H : RSSSPP (1)
     Special parameter for system reference
     Bit 7: XOFF receive flag (1: Received)
     Bit 6: RTS/CTS control (1: Active)
     Bit 5: DTR/DSR control (1: Active)
     Bit 4: XON/XOFF control (1: Active)
     Bit 3: XOFF send flag (1: Sent)
Bit 2: SI/SO control (1: Active)
     Bit 1: RTS control (1: Active)
     Bit 0: DTR control (1: Active)
F61AH : RSPAKAD (2)
     Address of RSIOX parameter packet (Value of RSIOX parameter HL)
F61CH : RSRDL (2)
     Length of data stored in receive buffer
F61EH : SISOCNT (2)
     Number of SI and SO codes in PSRDL. Actual data length can be
     determined by the following expression: RSRDL - SISOCNT
F620H : SSXMODE (1)
     SI/SO send status
      =03H: SI send status
      =01H: SO send status
      =02H: SI and SO unsend status (initial status)
F621H : RSXMODE (1)
     SI/SO receive status
      =00H: SI status (initial status)
      =80H: SO status
F622H : RSFDEV (1)
     RSIOX parameter 4-bit LSB in register B
F623H : RSODEV (1)
     Device mode used by RSIOX
      =00H: RS-232C
=03H: Cartridge SIO
      =FFH: Nothing has been opened.
     Among the device specifications, only the serial communication mode
(3)
     switching is controlled by the system. Because of this, the cartridge
     mode (DB, IO, HS, or UT) is not changed even if the cartridge SIO is specified as the device. See "Section 7.3 Cartridge Interface" for
     details.
(4) The following terms are used for the RSIOX functions:
     Serial: Generic for RS-232C and cartridge SIO
     RS-232C: Communication through RS-232C connector
     Cartridge SIO: Communication through cartridge connector
<Reference>
```

Section 7.2 Serial Interface Section 7.3 Cartridge Interface APPENDIX 9 SAMPLE 7

- (1) This command is executed to switch the device to the specified one, set the communication status according to the specified parameters, allow serial interrupts, and enable sending and receiving.
- (2) A parameter block is structured as follows:

(HL)-> +1	Start address of receive buffer
+2 +3	Size of receive buffer (bytes)
+4	Bit rate
+5	Character_length
+б	Parity check
+7	Stop bit
+8	Special parameter

Fig. 4.3 RSIOX parameter block

- 1 Start address of receive buffer This parameter specifies the start address of receive buffer.
- 2 Size of receive buffer (bytes) This parameter specifies the size of receive buffer.
- 3 Bit rate This parameter specifies the transmission speed.

Value	Send bit	rate	Receive bit rate
02H	110	bps	110 bps
04H	150	·	150
05H	200		200
06H	300		300
08H	600		600
OAH	1200		1200
OCH	2400		2400
ODH	4800		4800
OEH	9600		9600
OFH	19200		19200
10H	38400		38400
80H	75		1200
<b>8</b> 1H	1200		75

4 Character length This parameter specifies 1-character length for communication. 02H: 7 bits/character 03H: 8 bits/character

- 5 Parity check This parameter specifies parity checking. 00H: No parity checking 01H: Odd parity 03H: Even parity
- 6 Stop bit This parameter specifies the stop bit length for communication. 01H: 1 bit 03H: 2 bits
- 7 Special parameter Each bit of this parameter specifies condition or status for serial communication

Bit position	Contents
0	Data-terminal-ready (DTR) line control 0:Inactive1: Active
1	Request-to-send (RTS) line control 0: Inactive 1: Active
2	Shift in/sift out (SI/SO) control O: Control 1: No control
3	Unused
4	XON/XOFF control 0: Control 1: No control
5	DTR/DSR control 0: Control 1: No control
б	RTS/CTS control 0: Control 1: No control
7	Unused

DTR and RTS are effective only when RS-232C is specified. XON/XOFF, DTR/DSR, and RTS/CTS are used for buffer control. Only one of them can be specified. (3) A return information block is structured as follows:

Fig. 4.4 RSIOX return information block

(HL)	->	Status flag
+1		Receive get pointer
+3 +4 +5	+3	Receive put pointer
	Start address of receive buffer	
	+0 +7	Size of receive buffer
	+8	

1 Status flag

Each bit indicates serial status.

Bit position	Contents		
0	Open status O: Not open 1: Open status		
I	Receive buffer status 0: Buffer sufficient 1: Buffer full		
2	Overflow status of receive buffer O: No overflow 1: Overflow		
3	Carrier detector (CD) status O: Inactive 1: Active		
4	Parity error status 0: No error 1: Parity error		
5	Receive overrun error status O: No error 1: Receive overrun error		
б	Framing error status O: No error 1: Framing error		
7	Data set ready (DSR) status O: Active 1: Inactive		

CD and DSR are effective only when RS-232C is specified. Error information indicated by bits 2 and 4 to 6 is held once an error occurs until RSIOX (Ersts) is called.

- 2 Receive buffer get pointer This pointer is used to get data from the receive buffer.
- 3 Receive buffer put pointer This pointer is used to put receive data in the receive buffer.
- 4 Start address of receive buffer Start address of receive buffer
- 5 Size of receive buffer This is the byte length of receive buffer.
- (4) Buffer control is ineffective when the size of receive buffer is less than 16 bytes.

- (5) SI/SO is specified to send 8-bit data when 7 bits/character is used. Codes OEH and OFH cannot be sent when SI/SO processing is specified. Because of this, binary data cannot be sent. See "Section 7.2 Serial Interface" for details.
- (6) XON/XOFF, DTR/DSR, or RTS/CTS is specified to synchronize the receive and send sides when the communication speed is faster than the processing speed at the receive side. Codes 11H and 13H cannot be sent when XON/XOFF processing is specified. Because of this, binary data cannot be sent. See "Section 7.2 Serial Interface" for details.

<Relations>

RSIOX (Ersts)

<Reference>

Section 7.2 Serial Interface APPENDIX 9 SAMPLE 7
[Function] RSIOX (Close) closes the used device. [Entry parameters] B=20H: Function number

```
[Return parameters]
None
```

<Explanation>

- (1) RSIOX (Close) closes the serial device and prohibits interrupts at the receive side.
- (2) The serial device is automatically closed when WBOOT processing is executed.

<Referance>

APPENDIX 9 SAMPLE 7

.

<Explanation>

- (1) RSIOX (Insts) checks whether there is receive data in the receive buffer.
- (2) XON or XOFF codes (when XON/XOFF is specified) or SI or SO codes (when SI/SO is specified) are not included in the byte longth of receive data.
- (3) The current serd or receive status is set in the return information block.

SOFTWARE Page 4 - 46

.

[Function] RSIOX (Outst) checks whether sending can be done. [Entry parameters] B=40H: Function number HL=start address of return information block (9-byte area is required.) [Return parameters] Z-flag=1: Normal termination A=FFH: Sending possible =00H: Sending not possible HL=saved (return information in the same structure as Open) Z-flag=0: Abnormal termination A=O3H: Open operation had not been executed.

<Explanation>

- (1) Whether sending is possible is determined by checking TXReady. Sending is not possible under one of the following three conditions:

   XON/XOFF control is specified and XON is received.
   DTR/DSR control is specified and DSR is inactive.
   RTS/CTS control is specified and CTS is inactive.
- (2) The current send or receive status is set in the return information block.

<Reference>

[Function] RSIOX (Get) fetches 1-byte data from the receive buffer.
<pre>[Entry parameters]     B=50H: Function number     HL=start address of return information block (9-byte area is required.)</pre>
[Return parameters] Z-flag=1: Normal termination A=receive data
HL=saved (return information in the same structure as Open.) Z-flag=0: Abnormal termination A=03H: Open operation had not been executed. =04H: Processing was forcibly terminated. =05H: Receive-buffer-overflow occurred.

- (1) RSIOX (Get) fetches 1-byte data from the receive buffer and sets it in register A.
- (2) When data is not yet received, processing waits until data is received.
- (3) Power off or alarm/wake operation is executed if power off or alarm/wake occurs while waiting for data to be received. Processing waits for data to be received after the power off or alarm/wake operation.

[Function] RSIOX (Put) sends the specified 1-byte data.
[Entry parameters] B=60H: Function number C=send data HL=start address of return information block (9-byte area is required.)
[Return parameters] Z-flag=1: Normal termination HL=saved (return information in the same structure as Open) Z-flag=0: Abnormal termination A=03H: Open operation had not been executed. =04H: Processing was forcibly terminated.

- (1) RSIOX (Put) checks the send possible (ready) status and sends the specified data if sending is possible.
- (2) Whether sending is possible is determined in the same way as RSIOX (Outst). If sending is not possible, processing waits until sending becomes possible.
- (3) If forced termination occurs while processing is waiting for sending to become possible, A=04H is set.
- (4) Power off or alarm/wake operation is performed if power off or alarm/wake occurs while processing is waiting for sending to become possible. Processing waits again for sending to become possible after the power off or alarm/wake operation.

<Relations>

RSIOX (Outst) APPENDIX 9 SAMPLE7 

### <Explanation>

The control line status is structured as follows in the return parameter:

Bit position	Contents
7	Data set ready (DSR) status 0: Active 1: Inactive
6	Unused
5	Clear to send (CTS) status 0: Inactive 1: Active
4	Unused
3	Carrier detect (CD) status D: Inactive 1: Active
2,1,0	Unused

CTS, CD, and DSR are effective only when RS-232C is specified.

### <Explanation>

(1) The control line status is structured as follows in the return parameter:

Bit position	Contents
From 7 to 2	Unused (0)
1	Request to send (RTS) control 0: Inactive 1: Active
Ö	Date transmit ready (DTR) control 0: Inactive 1: Active

RTS and DTR are effective only when RS-232C is specified.

(2) This command is used to reset the control line status specified at open operation.

(1) The error information is structured as follows in the return parameter:

Bit position	Contents
7	Data set ready (DSR) status O: Active 1: Inactive
б	Framing error status O: No error 1: Error
5	Receive overrun error status O: No error 1: Error
4	Parity error status O: No error 1: Error
3	Carrier detect (CD) status O: Inactive 1: Active
2	Receive buffer overflow error status 0: No error 1: Error
1,0	Unused

The same information is set for CD and DSR regardless of the device specification.

- (2) The status of errors is reset by the software after the current error status is read.
- (3) If a receive buffer overflow occurs, receive data is discarded until data is fetched from the receive buffer by RSIOX(Get).

<Reference>

[Function]

RSIOX (Sens) checks the use status of the current serial device.

[Entry parameters] B=FOH: Function number

[Return parameters] A=serial device use status



<Explanation>

RSIOX (Sens) checks the use status of the current serial device and sets the information in register A.

```
[Function]
MASKI sets interrupt mask and checks the current mask status.
[Entry parameters]
B=interrupt mask data
C=7508-related interrupt mask data (explained later)
[Return parameters]
B=old interrupt mask status
C=old 7508-related interrupt mask status
[Saved registers]
AF, DE, HL
```

- (1) MASKI controls five types of interrupts for EHT-10/EHT-10/2 and three types of interrupts related to 7508.
- (2) The entry parameters are structured as follows:
  - 1 Register B



2 Register C



SOFTWARE Page 4 - 55

[Function] LOADX reads 1-byte data at the specified address from the specified bank. [Entry parameters] C=bank information HL=data address [Return parameters] A=read data [Saved registers] BC, DE, HL, IX, IY

<Explanation>

- (1) Bank information is in the same format as the return parameter of MEMORY.
- (2) Since parameter validity is not checked, operation is not guaranteed if incorrect parameters are specified.
- (3) The data address is the address of data mapped in memory. The data address is not equal to ROM address when data is in a bank in application ROM.

<Relations>

MEMORY (WBOOT+4EH)

<Reference>

Section 1.4 Memory Map

SOFTWARE Page 4 - 56

```
[Function]
    STORX writes 1-byte data at the specified address in the specified
    bank.
[Entry parameters]
    C=bank information
    A=data
   HL=data address
[Return parameters]
     None
[Saved registers]
     AF, BC, DE, HL, IX, IY
<Explanation>
(1) All registers are saved.
(2) Since parameter validity is not checked, operation is not guaranteed if
     incorrect parameters are specified.
(3) Data can be actually written only in RAM.
<Relations>
     MEMORY (WBOOT+4Eri)
     LOADX (WBOOT+5AH)
<Reference>
     Section 1.4 Memory Map
     APPENDIX 9 SAMPLE 19
```

```
[Function]
    LDIRX moves data as much as the specified number of bytes from the
    specified bank to another bank.
[Entry parameters]
    BC=byte length of data to be transferred
    DE=start address of data in the destination bank
    HL=start address of data in the original bank
    (SRCBNK)=information of the original bank (same as MEMORY)
    (DISBNK) = information of the destination bank (same as MEMORY)
[Return parameters]
    BC=0000H
    DE=entry parameter DE + entry parameter BC
    HC=entry parameter HL + entry parameter BC
    (Same as the ones at Z-80 instruction LDIR execution)
[Saved registers]
    AF, ĪX, IY
```

- (1) This command is used to take into account of the bank used by instruction LDIR.
- (2) See the explanation of BIOS INFORM for the SRCBNK and DISBNK addresses.
- (3) Since parameter validity is not checked, operation is not guaranteed if incorrect parameters are specified.

<Relations>

MEMORY (WBOOT+4EH) INFORM (WBOOT+A2H)

<Reference>

```
[Function]
JUMPX jumps to the specified address in the specified bank.
[Entry parameters]
IX=jump address
(DISBNK)=bank information (same as MEMORY)
[Return parameters]
None
[Saved registers]
AF, BC, DE, HL, IX, IY (when control jumps to the destination)
```

- (1) At the jump destination, all registers remain in the same status as the one held when control is passed to JUMPX.
- (2) See the explanation of BIOS INFORM for the DISBNK address.
- (3) Since parameter validity is not checked, operation is not guaranteed if incorrect parameters are specified.
- (4) The BIOS stack provided by the system is used after this command is executed. Therefore, the processing routine to be executed after jump operation must set the stack again. If this is not done, wild run may occur when BDOS/BIOS is used.
- (5) The system is in "BIOS in progress" status after this command is executed. Call JPSTBIOS (FF96H) to make the system out of "BIOS in progress" status.

#### <Relations>

MEMORY (WBOOT+4EH) INFORM (WBOOT+A2H)

# <Reference>

Section 1.4 Memory Map

[Funct	tion]
(	CALLX calls the specified address in the specified bank.
[Entry	y parameters]
1	IX=call address
(	(DISBNK)=bank information (same as MEMORY)
[Retur	rn parameters]
A	Among the registers returned from call destination, the ones other than
I	IX and IY are saved.

- (1) The registers held when this routine is called are passed to the call destination.
- (2) See the explanation of BIOS INFORM for the DISBNK address.
- (3) Since parameter validity is not checked, operation is not guaranteed if incorrect parameters are specified.
- (4) After this command is executed, the BIOS stack provided by the system is used and the system is in "BIOS in progress" status. Therefore, the notes (4) and (5) in the explanation of JUMPX must also be taken into account.
- (5) When control returns, the original stack must be set again if the stack is modified at the call destination.
- (6) This command is used to call a utility stored in system ROM.

<Relations>

MEMORY (WBOOT+4EH) JUMPX (WBOOT+63H) INFORM (WBOOT+A2H)

<Reference>

Section 1.4 Memory Map APPENDIX 9 SAMPLE 21

```
[Function]
    GETPFK reads the code of the current set key.
[Entry parameters]
    C=function code
      =01H: Reads the key table in normal mode.
             Reads the key table in alphabet mode (effective only for
     =02H:
             EHT-10/2 but not for EHT-10).
    B=position code
     #FFH: Reads the key code of the key corresponding to the position
            code.
     =FFH: Reads all codes.
    When B=FFH.
        HL=start address of buffer used to store key code data
[Return parameters]
   (1) When GETPFK is called with B≠FFH:
       A=key code
    (2) When GITPFK is called with B=FFH:
       (HL)=key code data
[Saved registers]
    BC, HL
```

- (1) All the key codes beginning from the address specified in HL are read from the key table when GETPFK is called with B=FFH.
- (2) A 70-byte buffer (for EHT-10) or a 32-byte buffer (for EHT-10/2) must be reserved after address indicated by HL when B=FFH is specified. The read data is stored in this buffer in the ascending order of the key position codes.

<Relations>

CONST (WBOOT+O3H) CONIN (WBOOT+O6H) TOUCH (WBOOT+93H)

<Reference>

Section 4.3 Key Input (Touch Panel/Keyboard) APPENDIX 9 SAMPLE 8

SOFTWARE Page 4 - 61

```
[Function]
     PUTPFK stores a key code in the user key table.
[Entry parameters]
    C=function code
     =OOH: Initializes the user key table.
      =01H: Stores a key code in the key table in normal mode.
     =02H*: Stores a key code in the key table in alphabet mode
            (* is effective only for EHT-10/2 but not for EHT-10)
     B=position code
     #FFH: Stores the key code of the key corresponding to the position
             code.
    When B#FFH.
        A=key code
         =FFH: Sets the entire user key table.
    When B=FFH,
       HL=start address of key code buffer
[Return parameters]
    Hune
[Saved registers]
    BC, HL
```

- (1) The contents of register B are meaningless when the user key table is initialized.
- (2) Data stored at addresses starting from the one indicated by HL is set in the key table when the entire key table is set. The required length of data is 70 bytes for EHT-10 or 32 bytes for EHT-10/2. Data must be stored in the ascending order of the key position codes.
- (3) The default values are stored in normal or alphabet mode key table when warm boot is executed.

<Relations>

CONST (WBOOT+03H) CONIN (WBOOT+06H) TOUCH (WBOOT+93H)

<Reference>

Section 4.3 Key Input (Touch Panel/Keyboard) APPENDIX 9. SAMPLE 9 <Explanation>

When the DIP switches are read, the return parameter isstructured as follows:



<Reference>

APPENDIX 9 SAMPLE LIST

(1)	<pre>RDVRAM reads one character from the character buffer of the current active screen. The values that can be specified in the entry parameters are limited as follows: 1 EHT-10 (vertical display) (i) Virtual screen section 1 &lt; B &lt; 12 1 &lt; C &lt; virtual screen size (ii) Fixed display section 1 &lt; B &lt; 12 1 &lt; C &lt; 14</pre>
	<pre>2 EHT-10 (horizontal display) 1 &lt; B &lt; 25 1 &lt; C &lt; virtual screen size</pre>
	3 EHT-10/2 1 < B < 20 1 < C < virtual screen size

(2) The character attributes are indicated in the same way as BIOS CONOUT Set Attribute (ESC+D9H) as follows:



- (Note 1) Only the reverse and secret attributes are effective for EHT-10 horizontal display.
- (Note 2) The contents of register H are meaningless for EHT-10/2 because EHT-10/2 does not have any character attributes.

<Relations>

CONOUT (WBOOT+09H)

<Reference>

Section 4.4 LCD Display APPENDIX 6 FUNCTION DISPLAY CONTROL APPENDIX 9 SAMPLE 22

```
[Function]
    POWEROFF turns the system power supply off.
[Entry parameters]
    C=power off mode
    =00H: Turns the power off in continue mode.
    =01H: Turns the power off in restart mode.
[Return parameters]
    None
```

- POWEROFF saves the current system status and turns the system power supply off.
- (2) When the power is turned off in restart mode, the power may be turned off in continue mode instead of restart mode unless continue mode is reset by BIGS CONTINUE in advance.

<Relations>

CONTINUE (WBOOT+87H)

<Reterence>

Section 2.3 Sleep function and Automatic Power-off function APPENDIX 10 SAMPLE 10

Function] USERBIOS registers a user BIOS.
Entry parameters] Parameters differ depending on the user.
Return parameters] Parameters differ depending on the user.

- (1) USERBIOS is used to extend a user BIOS. A user BIOS area is reserved and the user BIOS is stored in this area.
- (2) The execution start address of the user BIOS is stored as the entry address.
- (3) The default value is the address indicating only return operation.
- (4) The default value is set at reset or system initialize operation.

### <Reference>

Section 4.6 User BIOS

[Function] CONTINUE sets/resets the continue flag. [Entry parameters] C=00H: Reset =01H: Set

=oin: set [Return parameters] None

<Explanation>

If the power switch is turned off after the continue flag is reset, the power is turned on in restart mode during subsequent power on operation. The default is continue mode.

<Reference>

[Function] BARCODE supports the barcode reader functions. The following four functions are provided depending on the contents of register C: C=OOH: Allows to use barcode reader. (Open) =01H: Prohibits to use barcode reader. (Close) =02H: Reads one character. (Read) =03H: Indicates the buffer status. (Status) Details on each function are explained later.

## <Reference>

Section 7.5 Barcode Reader Interface Section 11.5 Addition of Bar Code Decoder APPENDIX 9 SAMPLE 12

```
[Function]
BARCODE (Open) turns the barcode reader power on and enables the
barcode reader operations.
[Entry parameters]
C=00H: Function number
A=code type
=01H: EAN/UPC-A/UPC-E/JAN
=02H: 3 of 9
=03H: Codabar
=04H: Interleaved 2 of 5
D=option (explained later)
E=delimiter (explained later)
[Return parameters]
None
```

(1) The following options can be set:

Bit in register D	Corresponding code	Explanation
b7	All codes	Buzzer for normal reading O: Sounds the buzzer. 1: Suppresses the buzzer. The buzzer is sounded for about 100 ms at normal reading when 0 is specified
b6	All codes	Delimiter O: Uses the delimiter. I: Suppresses delimiter. A delimiter is a special code (one character) inserted between two codes to indicate the end of a code. When the delimiter is specified, the code specified in register E is used as the delimiter.
b5	All codes	LED control for barcode reader O: Performs LED control. 1: Suppresses LED control. LED remains lit when LED control is suppressed. LED blinks to have less power consumption when LED control is on.
b4,b3	Unused	

b2	Only for UPC-E	Zero addition O: Zero is not added. 1: Zeros are added.
b1	Only for 3-of-9	Full ASCII conversion specification O: Suppresses full ASCII conversion. 1: Performs full ASCII conversion.
bO	Only for 3-of-9	Check digit O: Does not assume the last data as the check digit. 1: Assumes the last data as the check digit. When the check digit is specified, the last data is assumed as the check digit and is not indicated as a character. When the check digit does not match, the entire code is assumed invalid and ignored.

(2) The delimitor is specified as follows:

The value specified in register E is used as the delimiter when option b6 is 0. A value from OOH to FFH can be specified but this value must not be equal to any codes used for the barcodes in the character set. Other wise, the code is not effective as the delimiter.

<kererence>

[Function]
 BARCODE (Close) turns the barcode reader power off and disables the
 barcode reader operations.
[Entry parameters]

[Entry parameters] C=01H: Function number

[Return parameters] None

<Reference>

[Function] BARCODE (Read) reads one character from the barcode reader. [Entry parameters] C=02H: Function number [Return parameters] Z-flag=0: There is data. When Z-flag=0, A=data Z-flag=1: There is no data.

<Explanation>

- A barcode is read during interrupt processing and stored in a buffer. This command only fetches one character from data stored by interrupt processing.
- (2) If 0 is specified as option b6 and delimiter is specified at Cpan operation, the delimiter is inserted between barcodes.

<Reference>

CHAPTER 8 INTERRUPT PROCESSING APPENDIX 9 SAMPLE 12 [Function] BARCODE (Status) indicates error information and the data stored in buffer by interrupt processing. [Entry parameters] C=03H: Function number [Return parameters] BC=byte length of data stored in buffer DE=size of empty area in buffer (byte length) (BC + DE = buffer size. The buffer size is fixed to 80 bytes.) A=error information (explained later) =00H: Normal termination #00H: Abnormal termination

<Explanation>



(3) Error information is reset once this routine is called.

<Reference>

[Fur	TCAM sends or receives data through public line. TCAM is the abbreviation of telecommunication access method. TCAM has he following four functions depending on the contents of register A: A=01H: Connects to the host computer. (Connect) =02H: Sends data to the host computer. (Send) =03H: Receives data from the host computer. (Receive) =04H: Disconnects from the host computer. (Disconnect) Details on each function are explained later.
<exp< th=""><th>planation&gt;</th></exp<>	planation>
(1)	No protocol and Filink are the usable protocols. Further, a protocol such as BSC protocol can be extended (extended protocol).
(2)	See "APPENDIX 8 FILINK PROTOCOL" for the Filink protocol.
(3)	The serial line and transmission protocol are specified by system menu CONFIG.
(4)	TCAM uses the following work areas:
Addr	ress:Variable name(Byte length)
FICO	DH : TCAMPRM(7) This area stores the initiation conditions used at TCAM open operation. The initiation conditions are set and modified by CONFIG but they can also be modified by an application program.
+0 +1 +2 +3 +4 +5 +6	TypeType:Protocol typeLine=0: protocol direct-CBit rate=1: protocol direct-BCharacter length=2: Filink (default)Parity check=3: Extended protocolStop bitSpecial parameter
	Line: Serial line to be used for sending and receiving =0: RS-232C (default) =3: Cartridge I/F The bit rate, character length, parity check, stop bit, and special parameter are same as that of BIOS RSIOX. The default values are 4800 Bps, 8 bits, Nonparity, and 2 stop bits.
F1D4	IH : TDFLTCNT (1) This area is used to specify the number of retry operations executed when the Filink protocol does not match with the host protocol. (The default is 3.)

F1D5H : T1STTIME (2)
F1D7H : T2NDTIME (2)
This area is used to specify the timer used for timeout during data
receiving (unit: Seconds).
T1STTIME: Timer for the first data receive operation (30 seconds)
T2NDTIME: Timer for the second data receive operation (3 seconds)
In the first data receive operation, data up to file name is received
when Filink protocol is used, or 1-byte data is received when protocol
is not used (Direct-B or -C).

<Reference>

Section 11.2 Extending Communication Protocol APPENDIX 8 FILINK PROTOCOL APPENDIX 9 SAMPLE 13

[Function] TCAM (Connect) allows serial interrupts and connects the line to the host computer. [Entry parameters] A=01H: Function number **B**=connection information [Return parameters] CY=0: Normal termination =1: Abnormal termination When CY=1. A=error code A=01H: Parameter error =02H: Open operation had been executed. =03H: Open operation was not executed. =04H: Forced termination =05H: Receive buffer overflow =06H: Timeout =07K: Protocol arror =08H: Communication error

### <Explanation>

- (1) This command allows serial communication interrupts and opens a serial line. Processing differs depending on the protocol as follows:
  - 1 For No protocol This command ignores the contents of register B and opens a serial line.
  - 2 For Filink

The send or receive file operation is specified in entry parameter register B. B=00H: Send file =01H: Receive file The file even must be transmitted by using TCAMAPEA. See the

The file name must be transmitted by using TCAMAREA. See the explanation of BIOS INFORM for the TCAMAREA address. When send file is specified, set a file name in TCAMAREA and then perform the connect operation.

When receive file is specified, the name of a file to be received is stored when connect operation is normally terminated. TCAMAREA is structured by a file name (8 bytes) and a file type (3 bytes).

3 For an extended protocol Processing differs depending on the extended protocol. In this case, registers BC, DE, and HL can be used as parameters.

<Relations>

INFORM (WBOOT+A2H)

<Reference>

<Explanation>

- The sent data specified by register HL is sent as much as the number of bytes specified in register BC.
   BC=128(80H) is used when the Filink protocol is specified.
- (2) No operation is performed when the Filink protocol is specified and "Receive File" is specified by Connect.

<Explanation>

- TCAM (Receive) stores receive data in the receive buffer specified in register HL. The byte length of receive data is set in return parameter register 3C.
- (2) When No protocol or Filink protocol is specified, up to 128-byte data can be received by one Receive operation. Receive end is determined from BC=0.
- (3) No operation is performed when the Filink protocol is specified and "Send File" is specified in Connect.
- (4) When the Filink protocol is specified, register BC indicates 11 (OBH) at the start of receiving the second or later file to indicate that two or more files are received. In this case, a file name is stored in TCAMAREA in the same way as Open. Note that register BC contains 128 (80H) when used as a return parameter for general data receive operation.

<Reference>
<Explanation>

TCAM (Disconnect) closes the serial line.

<Reference>

.

[Function]	
GRAPHICS	S supports 12 graphics functions depending on thecontents of
registei	r C as follows:
C=00H:	Initializes a graphic package. (Init)
=01H:	Sets a view port. (View)
=02H:	Clears the contents of view port. (Cls)
=03H:	Sets a dot. (Pset)
=04H:	Indicates the attribute of a dot. (Point)
=05H:	Draws a line. (Line)
=06H:	Draws a circle. (Circle)
=07H:	Paints an enclosed area. (Paint)
=08H:	Fills an enclosed area with tile pattern. (Tile)
=09H:	Saves the drawing information of the specified area in the
	specified storage area. (Get)
=OAH:	Loads the drawing information from the specified storage area
	to the specified area. (Put)
=08H:	Draws the specified kanji in the specified area. (Kanji)
Details	on each function are explained later.

<E::planacion>

- (1) GRAPHICS is a group of functions used to draw graphic images with dots and lines in LCD.
  - Coordinates Coordinates for EHT-10 differ from that of EHT-10/2. For both EHT-10 and EHT-10/2, the origin is placed at the left top (0,0). This is same for EHT-10 horizontal display.

Fig. 4.5 Coordinates for EHT-10 and EHT-10/2

i) Coordinates for EHT-10 vertical display



SOFTWARE Page 4 - 82

ii) Coordinates for EHT-10 horizontal display





2 View port

A view port is a defined drawing area in actual screen. Drawing by using the functions is allowed only in a view point. Coordinates are specified with X and Y. Values indicating out of a view port or exceeding the physical LCD sizes can be specified for the Pset, Line, Circle functions but drawing can only be done in a view port. Coordinates can be specified with integers from -32768 to 32767. The specified coordinates must be within a view port for the Point, Paint, Tile, Get, Put, and Kanji functions.

3 Data package

A data package is a group of data used to draw lines and circles to be displayed in LCD. The data package structure differs depending on the function. The start address of data package must be set in register HL when GRAPHICS is called.

4 Attribute

This is the dot drawing attribute and is specified with 0 or 1 as follows: 0: Dot reset

1: Dot set

```
5 Data format
2-byte data (such as coordinate values) used in GRAPHICS is
specified in the order of low and high unless other wise stated.
```

(2) The cursor must be turned off before this BIOS command is used.

[Function] GRAPHICS (Init) sets a view port containing the entire LCD. The screen is not changed. [Entry parameters] C=00H: Function number

[Return parameters] A=end condition =00H: Normal termination

<Explanation>

(1) This routine is automatically called during warm boot operation. This is one of the routines that initialize EHT-10/EHT-10/2. The user does not require this routine excepting to reset a view port containing the entire LCD.

<Relations>

WBOOT (WBOOT+O)

```
[Function]
GRAPHICS (View) specifies a view port. This can also paint the view
port or draw the outer frame of the view port.
[Entry parameters]
C=01H: Function number
HL=start address of data package (explained later)
[Return parameters]
A=end condition
=00H: Normal termination
=01H: Parameter error
=02H: (X,Y) was not in LCD.
```

(1) After this command is executed, drawing is allowed only within the view port specified by this command. In the example below, drawing is allowed only within the area filled with oblique lines.



Cb: Attribute used to draw the outer frame of the view port OOH: Dot reset OIH: Dot set FFH: No drawing

```
[Function]
     GRAPHICS (Cls) clears the contents of view port.
[Entry parameters]
     C=02H: Function number
     HL=start address of data package (explained later)
[Return parameters]
     A=end condition
     =00H: Normal termination
```

The data package is structured as follows:

C: Attribute OOH: Dot reset O1H: Dot set

```
[Function]
    GRAPHICS (Pset) sets a dot at the specified coordinates.
[Entry parameters]
    C=03H: Function number
    HL=start address of data package (explained later)
[Entry parameters]
    A=end condition
    =00H: Normal termination
```

The data package is structured as follows:

	+0		+2		+4	+5
HL	->	х		Y		С
		X: Y:	X C Y C	coor coor	din din	ate ate
		ι.	001	1: 1:	Dot Dot	reset set

[Function] GRAPHICS (Point) indicates the attribute of the specified coordinates. [Entry parameters] C=04H: Function number HL=start address of data package (explained later) [Return parameters] A=end condition =00H: Normal termination =02H: (X, Y) was not in the view port. C=attribute =00H: Dot reset =01H: Dot set

#### <Explanation>

The data package is structured as follows:

X: X coordinate Y: Y coordinate

[Function] GRAPHICS (Line) draws a line between two specified points or draws a rectangle that has a diagonal line placed between two specified points.
[Entry parameters] C=05H: Function number HL=start address of data package (explained later)
[Return parameters] A=end condition =OOH: Normal termination =O1H: Parameter error
<explanation></explanation>

The data package is structured as follows:

+0	+;	2 +4	4 +6	i +8	8 +9	+1	0 -	+12			
HL ->	XI	Yl	X2	¥2	C	F	Ρ				
	X1: Y1: X2: Y2:	X coord Y coord X coord Y coord Attribu	dinate c dinate c dinate c dinate c	of draw of draw of draw of draw	ing s ing s ing e ing e	tart tart nd p nd p	point point oint oint	t			
	F:	00H: ( 01H: ( Drawing 01H: ( 02H: F 03H: F	Dot rese Dot set g code Line dra Rectangl Rectangl	e draw e draw e paint	ing ting ern)	(P i	s igno	ored	when	this is	specified.)
	15	14 13 1	12 11 10	98	7 (	5 5	4 :	3 2	1	0	
	   <	Left	line da	  ta>	<	Rig	ht li	ne da	ata -	·>¦	
	The corr is (	attribu respond: 0, noth:	ing to t ing is c	ified L he bit lonc.	by C wh <b>en</b>	iss ab	et to it is	the 1.	dot Wher	the bit	
< <b>Refere</b> APP	nce> ENDIX	9. SAM	PLE 14								

```
[Function]
     GRAPHICS (Circle) draws an ellipse or a circle according to the
     specified center and the specified radii on X and Y directions.
     GRAPHICS (Circle) can also draw an arc or fan shape when start and end
     points are specified.
[Entry parameters]
     C=O6H: Function number
    HL=start address of data package (explained later)
[Return parameters]
     A=end condition
      =00H: Normal termination
<Explanation>
(1) The data package is structured as follows:
    +0
           +2
                  +4
                         +6
                                +8 +9 +10 +11
                                                   +13
                                                          +15
        СХ
                                          F2
HL ->
               CY
                      RX
                             RY
                                   С
                                      F1
                                                TS
                                                       TE
        CX: X coordinate of center
        CY: Y coordinate of center
        RX: Radius in X direction
        RY:
            Radius in Y direction
        C:
            Attribute
             OOH: Dot reset
             01H: Dot set
        F1: Fan shape flag
```



A fan shape can be drawn when 1 is set to b0 and b7.

SOFTWARE Page 4 - 91

- F2: This specifies what to draw when the start point is the end point.
  OOH: Draws a full circle.
  OIH: Draws the point only.
  This specification is meaningless when the start point is not the end point.
  TS: Start point (OOOOH to FFFFH)
  TE: End point (OOJOH to FFFFH)
- (2) The relationship between the start and end points and the specified values can be determined by the following expression where the angle of the start (end) point is t radian:
   65536 x t/(2 x PI) where PI is 3.141592.
   (0 degree is indicated as 0000H and 180 degrees is indicated as 8000H.)

The left figure shows the angle relation between X and Y axes.



<Reference>

[Function] GRAPHICS (Paint) paints the area enclosed in boundary color and including the specified point, with the specified color.	
[Entry parameters] C=07H: Function number HL=start address of data package (explained later)	
[Return parameters] A=end condition =00H: Normal termination =01H: Parameter error =02H: (X, Y) was not in the view point. =03H: Insufficient work area	

The data package is structured as follows:

+0		+2	+4 +	5 +6	5 +8	8 +10	נ
HL ->	X	Y	Ср	Сь	WS	WL	

X:	X coordinate of paint start point
Υ:	Y coordinate of paint start point
Cp:	Area color (attribute)
	00H: Dot reset
	01H: Dot set
Cb:	Boundary color (attribute)
	OOH: Dot reset
	01H: Dot set
WS:	Work area start address
WL:	Work area size
	At least 12 bytes are required for the work area.
	A larger work area is required to paint a complicated figure.
	If the work area becomes insufficient during maint operation.
	procession is terminated and control error-returns
	processing is cerminated and control error returns.

```
[Function]
    GRAPHICS (Tile) fills the area enclosed in the boundary color and
    including the specified point, with the specified tile pattern.
[Eatry parameters]
    C=08H: Function number
   HL=start address of data package (explained later)
[Return parameters]
    A=end condition
     =00H: Normal termination
     =01H: Parameter error
     =02H: (X, Y) was not in the view port.
     =03H: Insufficient work area
```

Tb

(1) The data package is structured as follows:

	+0	+2	! +4	+	5 +3	7 +	·g +	10 +	12 +	14
HL -:	> _ ;	ĸ	Y	L	Tf	ТЬ	СЪ	WS	WL	

```
X: X coordinate of paint start point
 Y: Y coordinate of paint start point
L: Length of tile pattern ( 1 to 64 bytes)
Tf: Tile pattern storage address
Tb:
     Background tile pattern storage address
Cb:
     Boundary color (attribute)
     OOH: Dot reset
     01H:
           Dot set
WS:
     Work area start address
WL:
     Work area size
 Τf
                                . . . . . . . .
                             Up to 64 bytes
               1 byte
```

At least 12 bytes are required for the work area. A larger work area is required to paint a complicated figure. If the work area becomes insufficient during paint operation, processing is terminated and control error-returns.

(2) The tile pattern differs depending on the EHT-10/EHT-10/2 mode. In vertical display mode, 8 dots in Y direction are defined as 1 unit and units are defined as much as required in X direction. In EHT-10 horizontal display mode or EHT-10/2 mode, 8 dots in X direction are defined as I unit and units are defined as much as required in Y direction.

Mode	X direction	Y direction
EHT-10 Vertical	n	8 dots
EHT-10 Horizontal	8 dots	n
EHT-10/2	8 dots	n

A tile pattern is specified with up to 64 bytes. Each byte contains a binary value (0: white, 1: black) in 8 dots.

Example) In EHT-10 vertical display mode



Example) In EHT-10 horizontal mode or in EHT-10/2 mode

*	*			*	*			> CCH
	*	*			*	*		> 66H
		*	*			*	*	> 33H
*			*	*			*	> 99H

Y The pattern shown at the above figure is specified as follows: CCH, 66H, 33H, 99H

<Reference>

[Function] GRAPHICS (Get) saves the drawing information of the specified area in the specified storage area. [Entry parameters] C=90H: Function number HL=start address of data package (explained later) [Return parameters] A=end condition =00H: Normal termination =01H: Parameter error =02H: (X, Y) was not ir the view port. =03H: Insufficient work area

<Explanation>

```
(1) The data package is structured as follows:
```

	+0	+2	2 +4	4 + j	i +8	3 +	10 +	12
٩L	->	X1	¥1	X2	¥2	WS	WL	]

X1: X coordinate of the left top of specified area
Y1: Y coordinate of the left top of specified area
X2: X coordinate of the right bottom of specified area
Y2: Y coordinate of the right bottom of specified area
WS: Storage area start address
WL: Storage area size

The required size of the storage area is determined as follows:

Number of required bytes =
4 + ((DY + 7) /8) \* DX : EHT-10 vertical display mode
4 + ((DX + 7) /8) \* DY : EHT-10 horizontal display mode or EHT-10/2
where DX = (X2 - X1) + 1
 DY = (Y2 - Y1) + 1
 "/" indicates division of which the decimal positions in results are
truncated.

- (2) Data read into the storage area by Get operation is in the following format:
  - 1 Image in screen



EHT-10 Vertical display mode

2 Store order in storage area



A numeric value indicates the number of bytes.

where m = DY, n = DX, k = (DY + 7)/8 (EHT-10 vertical display mode) m = DX, n = DY, k = (DX + 7)/8 (EHT-10 horizontal display mode) (EHT-10/2 mode)

/ indicates division of which the decimal positions in results are truncated.

[Function] GRAPHICS (Put) loads drawing information from the specified storage area to the specified area. [Entry parameters] C=OAH: Function number HL=start address of data package (explained later) [Return parameters] A=end condition =OOH: Normal termination =OIH: Parameter error =O2H: (X, Y) was not in the view port.

<Explanation>

```
(1) The data package is structured as follows:
                         +6 +7
    10
           +2
                  +4
                            F
HL ->
        X1
               Y1
                      hS
        X1: X coordinate of the left top of drawing area
        Y1: Y coordinate of the left top of drawing area
        MS: Storag∟ area start address
        F:
             Drawing mode
             =0:PSET
              1:PRESET
              2:0R
              3:AND
              4:XOR
```

(2) The data format in the storage area must be equal to the format of data read into storage by Get operation.

```
[Function]
GRAPHICS (Kanji) draws the specified kanji in the specified area.
[Entry parameters]
C=OBH: Function number
HL=start address of data package (explained later)
[Return parameters]
A=end condition
=OOH: Normal termination
=OIH: Parameter error
=O2H: (X, Y) was not in the view port.
```

The data package is structured as follows:



<Relations>

KANJI (WBOOT+9CH)

<Reference>

[Func	<pre>[Functions] TOUCH sets and displays key blocks of touch-panel keys. This command is effective only for EHT-10. For EHT-10/2, no operation is performed and control returns. [Entry parameters] C=number of key blocks to be set DE=start address of key block descripter (explained later)</pre>								
[Entr D									
[Retu	rn parameters] None								
<expl< th=""><th>anation&gt;</th></expl<>	anation>								
(1)	If the same code is set to two or more adjoining touch-panel keys,								

- (1) If the same code is set to two or more adjoining touch-panel keys, these touch keys are assumed as one key and are called a key block. The TOUCH command sets key blocks and displays the specified key blocks in LCD.
- (2) The key block descripter is structured as follows:

```
Key block
     position | Number of keys
|<---->|<--->|
    +0 +1 +2 +3 +4 +5 +6 +7
                                                 +N
                             C/G Display Data
DE->
                                                       Y
     X
          Y
             NX.
                 NY
                      Κ
                          Α
                                                   X
                                                          • • • • • •
    |<-----> Key block 1 -----> <--- Key block 2</pre>
                      K = Key code
                      A = Attribute
     1 Key block position (X, Y)
          This specifies the position of the key at the left top of the key
          block to be set. The values of X and Y must be as follows:
          1 < X < 5, 1 < Y < 14
```

	1	2 3	4 5	6 7	8 9	10 11 1	2
1	1	(1, 1)	(2, 1)	(3, 1)	(4, 1)	(5,1)	
2		(1, 2),	(2.2)	(3, 2)	(4,2)	(5.2)	
3		(1, 3)	(2.3)	(3, 3)	(4, 3)	(5, 3)	
4		(1, 4)	(2,4).	(3.4)	(4, 4)	(5.4)	
5		(1.5)	(2, 5)	(3.5).	(4, 5)	(5.5)	
6		(1,6)	(2.6)	(3, 6)	(4,6)	(3, 6);	
7		(1,7)	(2,7)	(3,7)	(4, 7);	(5.7)	Key block position.
8		(1,8)	(2,8)	(J. 8) <sup>1</sup>	(4, 8);	(5.8)	in a pair of
9		(1, 9)	(2.9)	(3.9)	(4,9)	(5.9)	parentheses ( ) are the coordinates of
10		(1, 10)	(2, 10)	(3, 10)	(4, 10)	(5, 10)	Position where a
11		(1, 11)	(2.11)	(3, 11)	(4, 11)	(5, 11)	
12		(1. 12)	(2.12)	(3. 12)	(4, 12)	(5, 12)	- character is displayed.
:3		(1, 13)	(2.13)	(3, 13)	(4, 13)	(5, 13)	
14	1	(1, 14)	(2, 14)	(3, 14)	(4, 14)	(5, 14)	

The following table shows the coordinates of touch keys:

Fig. 4.6 Touch key display positions

2 Number of keys (NX, NY)

This specifies the number of keys in the X direction and the number of keys in the Y direction for the key block to be set. The total number of keys is NX multiplied by NY. The values of NX and NY must be as follows: 1 < X + NX - 1 < 5, 1 < Y + NY - 1 < 14

3 Key code

This specifies a key code defined to the key block. In other words, the specified key code is used in the entire key block. a value from 00H to FFH can be specified. However, fonts supported by the system in standard can be used if a value from F6H to FEH is specified.

The following fonts are supported by the system in standard:



# 4 Attributes

This specifies the display attributes of the key block.



The attribute specification is effective for the entire block to be set. The following figure shows the display status specified by attributes in an example of key block (vertical 2 x horizontal 2):



Fig 4.7 Key block attributes

<Reverse> inverts data in the area enclosed in the vertical and horizontal ruler lines (including the frame).

5 C/G

This specifies whether to display the key block in character or graphics mode. The structure of data in the display data section differs depending on the C/G specification.

C/G=00H:	Character mode
=01H:	Graphics mode
=FFH:	Uses the system fonts (this is effective only when the
	key code is a value from F6H to FEH).

б Display data

The structure of data differs depending on the C/G specification (character mode (including using system fonts) and graphics mode).

i) Character mode

Data as much as the number of bytes determined by the following expression is required when the key block size is NX multiplied by NY: NY x (2NX - 1)



ii) Graphics mode Graphic data as much as the number of bytes determined by the following expression is required when the key block size is NX multiplied by NY: NY x NX x 2 x 11

SOFTWARE Page 4 - 103



The contents of 11 bytes in key block G (Nxi, Nyi) is structured as follows:



SOFTWARE Page 4 - 104

### (Example)

Data is displayed in LCD as shown in lower right when this BIOS routine is called with the following data descripter:



- (3) Data is directly written in VRAM instead of the character buffer for key block display. All data written in a key block is replaced with key block display data.
- (4) Data written by this BIOS routine is handled in the same way as graphic data in the view of CONOUT.

<Reference>

Section 4.3.6 Touch Panel APPENDIX 9 SAMPLE LIST

SOFTWARE Page 4 - 105

[Fun	ction] ICCARD inputs data from or outputs data to an IC card. ICCARD has the following five functions depending on the value in register A: A=OOH: Opens an IC card. (Open) =OIH: Closes an IC card. (Close) =O2H: Reads 1-character data. (Read) =O3H: Writes 1-character data. (Write) =O4H: Checks the status of data received from an IC card. (Status) Details on each function are explained later.
<exp< th=""><th>lanation&gt;</th></exp<>	lanation>
(1)	Data is sent to or received from an IC card in serial communication by SIO I/F with 9600 bps, 8 bits, and even parity. The receive buffer (256 bytes) is provided by the system.
(2)	Another serial interface (RS-232C) cannot be used at the same time because data is sent to or received from an IC card in serial communication by SIO I/F (terminal floppy disks can be used).
(3)	The following system areas are used by ICCARD:
Addr	ess : Variable name(Byte length)
FICI	ዛ : ICCDTIME (2) Timeout time used when data is received from an IC card
F1R3	H : ICCDPRM (5) Line open parameters used when ICCARD open is specified.
F1E9	H : ICTSDT (1) Data used for matching when IC card reset response is received.
F1AE	H : ICRSTPNT (2) Points the stroage area of the Reset response. The default is to point ICRSTDT.
F663	H : ICRSTDT (10) Area to store data at reset response.
< <b>Re</b> 1	ation>
	RSIOX (WBOOT + 51H)
<ref< td=""><td>erence&gt;</td></ref<>	erence>
	11.3 IC card protocol expansion APPENDIX 9 SAMPLE 23

- (1) The ICCARD (Open) enables sending data to and receiving data from on IC card using SIO I/F in serial communication. Open processing is executed as follows:
  - 1 For the IC card, VCC CLK is supplied, RST is frecd, and then "Answer to Reset" is received from the IC card.
  - 2 Allows receive interrupts from the IC card.
- (2) Close an IC card with the function CLOSE when any error is occurred.

### <Reference>

# ICCARD (Close)

[Function] ICCARD (Close) closes an IC card.

[Entry parameters] A=01H: Function number

[Return parameters] None

<Explanation>

Receive interrupts sent from IC card is prohibited and the power supply to the IC card is turned off.

<Reference>

[Function] ICCARD (Read) reads 1-byte data from an IC card. [Entry parameters] A=02H: Function number [Return parameters] CY=0: Normal termination When CY=0, A=read data CY=1: Abnormal termination When CY=1. A=error code =01H: Parameter error =03H: Open operation was not executed. =04H: Forced termination =05H: Receive buffer overflow =06H: Timeout =08H: Communication error =09H: Power off end =OAH: Over current =OBH: A serial device has already been used.

<Explanation>

- (1) 1-hyte data re\_eived from an IC card and stored in receive buffer is read.
- (2) If data is not received for a fixed time (default is 3 seconds), a timeout error occurs and control returns.
- (3) The communication error code (O8H) is set for subsequent data if a communication error occurs in receive data.

<Reference>

# ICCARD (Write)

[Function] ICCARD (Write) writes 1-byte data to an IC card. [Entry parameters] A=03H: Function number C=write data [Return parameters] CY=0: Normal termination =1: Abnormal termination When CY=1, A=error code =01H: Parameter error =03H: Open operation was not executed. =09H: Power off end =0AH: Over current

<Explanation>

ICCARD (Write) sends data specified in register C to the IC card.

# <Reference>

[Function] ICCARD (Status) checks the status of data received from an IC card. [Entry parameters] A=04H: Function number [Return parameters] CY=0: Normal termination When CY=0. A=status =FFH: There is receive data. =00H: There is not receive data. BC=byte length of receive data CY=1: Abnormal termination When CY=1. A=error code =01H: Parameter error =03H: Open operation was not executed. =09H: Power off end =OAH: Over current

<Explanation>

- (1) ICCARD (Status) checks whether there is data received from an IC card, and indicates the byte length of data stored in the receive buffer when there is data received from the IC card.
- (2) To continue processing after control returns due to a communication error, insert the following operations to reset the flag:

RSPSTS	EQU	F604H
	LD	A, (RSPSTS)
	AND	100010118
	LD	(RSPSTS),A

<Reference>

[Function]	
KEYIÑ i	nitiates the system input function and receives entered data.
KEYIN ha	as the following four functions depending on the contents of
register	r B.
B=ŌOH:	Initiates the calculator. (Calc)
=01H:	Initiates the alphabet input function. (Alph)
=02H:	Initiates the kana input function. (KANA)
=03H:	Initiates the normal input function. (Norm)
Details	on each function are explained later.

This command initiates the calculator, alphabet, kana, or normal input function supported by the system in standard.

The calculator can perform simple arithmetic operations and input the results of arithmetic operations. The alphabet function can input alphanumeric (EHT-10) or alphabet (EHT-10/2) characters.

The kana function can input kana characters. The normal input function does not perform anything but enables normal input operations for EHT-10/2.

<Reference>

```
[Function]
   KEYIN (Calc) initiates the calculator and enables arithmetic operations
   and inputting the results of arithmetic operations.
[Entry parameters]
   B=00H: Function number
   C=data format of arithmetic operation results
   =00H: BCD format
   =FFH: ASC11 format
   HL=start address of area used to store arithmetic operation result
[Return parameters]
   A=byte length
   (HL)=result of arithmetic operation
[Saved registers]
   HL
```

- (1) KEYIN (Calc) displays the calculator screen to perform arithmetic operations. The result of arithmetic operation is set when the SAVE or QUIT key is pressed. When control is returned by the SAVE key, the arithmetic operation result held before return operation is set in the result storage area. When control is returned by the QUIT key, register A contains OOH. The BCD or ASCII format can be specified for the arithmetic operation results.
- (2) The size of result storage area differs depending on the result data format as follows: 11 bytes for BCD format 13 bytes for EHT-1J ASCII format 21 bytes for EHT-10/2 ASCII format
- (3) The byte length is set in a return parameter (register A) as follows: When control is returned by the SAVE key, A=11 for BCD format A=1 to 13 for EHT-10 ASCII format A=1 to 21 for EHT-10/2 ASCII format When control is returned by the QUIT key, A=0

(4) The following formats are used for arithmetic operation results:

1 BCD format



```
[Function]
	KEYIN (Alph) initiates the alphanumeric (EHT-10) or alphabet (EHT-10/2)
	input function. Processing differs depending on EHT-10 or EHT-10/2
	(details are explained later).
[Entry parameters]
	B=01H: Function number
	HL=start address of input alphanumeric character storage area
	(This is not required for EHT-10/2.)
[Return parameters]
	A=byte length (0 for EHT-10/2)
	(HL)=input alphanumeric characters (only for EHT-10)
[Saved registers]
	HL
```

- (1) Differences between EHT-10 and EHT-10/2 processing
  - 1 EHT-10

The current screen status is saved and alphanumeric input screen (shown in the next page) is displayed. At the top three lines of the screen, display data is copied so that the cursor position held at alphabet specification is placed at line 3. (However, data is copied so that the cursor position is in line 1 or 2 when the cursor position is in line 1 or 2 of the wind.)

### Fig. 4.8 EHT-10 alphanumeric input screen



Input keys are displayed sequentially from the current cursor position. Up to 31 characters can be input.

- 'Return'key: This key ends input operation and stores input data in the buffer specified by register HL. CR (ODH) is added at the end of data. The number (1 to 32) of input keys is set in register A.
- '<-' key: This key is used to correct input data. This key deletes the latest column and shifts the cursor one column back.

When this BIOS routine is called in EHT-10 horizontal display mode, the top three lines are cleared with blanks and the cursor is moved to the home position. Display and operations are equivalent to vertical display mode.

2 EHT-10/2

This command makes the keyboard enter alphabet mode and turns on the alphabet LED. The display screen does not change. The input alphabet character storage area is not required because this command only modifies the mode. O is set in register A (return parameter)
```
[Function]
	KEYIN (KANA) initiates the kana input function. Processing differs
	depending on EHT-10 and EHT-10/2. This routine can be called only when
	Japan is specified by the DIP switch.
[Entry parameters]
	B=02H: Function number
	HL=start address of input kana character storage area (This is not
	required for EHT-10/2.)
[Return parameters]
	A=byte length (0 for EHT-10/2)
	(HL)=input kana character (only for EHT-10)
[Saved registers]
	HL
```

## <Explanation>

- (1) Differences between EHT-10 and EHT-10/2 processing
  - 1 EHT-10

The kana input screen shown in the next page is displayed. The screen is displayed in the same way as Alph.

Fig. 4.9 EHT-10 kana character input screen

7	1	ゥ	I	*
力	+	2	ケ	
#	2	ス	セ	y
3	チ	3	テ	۲
+	Ξ	3	7	1
$^{\wedge}$	E	7	~	赤
$\overline{\mathbf{v}}$	111	4	×	Ŧ
7	L	Ξ		•
ラ	IJ	N	V	
7	17	ン	-	
エイ	小		+	1

SOFTWARE Page 4 - 117

# 2 EHT-10/2

This command makes the keyboard enter kana mode and turns on the kana LED. The display screen does not change. The input kana character storage area is not required because this command only modifies the mode. O is set in register A (return parameter). Data is input in Roman letters and converted from Roman letters.

## <Reference>

4.3.7. Keyboard

.

[Function]	
KEYIN (Norm) makes the keyboard enter normal mode and turns on the normal LED. The screen does not change. This command is effective only for EHT-10/2. No operation is executed for EHT-10.	2
[Entry parameters] B=03H: Function number	
[Return parameters] A=00H	
[Saved registers] HL	

[Function]	
KANJI c	reates a gaiji (external character) or displays and prints a
kanji.	This command has the three functions depending on the contents
of regi	ster B as follows:
B=00H:	Specifies a gaiji file. (Gaiji)
=01H:	Displays a kanji (gaiji). (Disp)
=02H:	Prints a kanji (gaiji). (Print)
Details	on each function are explained later.

<Explanation>

(1) Kanji codes indicate the place where the font is stored as follows:

Kanji codes 0000H to 00ADH: Font (non kanji) in system ROM 03E8H to 07CBH: Font (kanji) in system ROM 0800H to 1FFFH: Font in gaiji (external character) file 2000H to 4FFFH: Font in JIS level-1 ROM

 Use a gaiji file that was specified by the Gaiji function.
 The JIS level-1 ROM can be inserted to the application ROM drive of EHT-10/EHT-10/2. The code is same as the JIS code.
 All kanji codes must be specified in the order of low and high.

- (2) All the Kanji and Gaiji characters are made by 16 \* 16 dots.
- (3) The cursor must be turned off before this BIOS routine is called.

<Relations>

GRAPHICS (Kanji) (WBOOT+90H)

<Reference>

APPENDIX 9 SAMPLE 17

```
[Function]
KANJI (Gaiji) specifies a gaiji file.
[Entry parameters]
B=00H: Function number
HL=start address of the file name (explained later)
[Return parameters]
None
```

<Explanation>

- (1) A gaiji is a font designed by the user and is accessed in a file. If a gaiji file is specified, the font fetched from the file can be displayed in LCD or output to the printer. In order to display or print a gaiji, this routine must be called to specify a gaiji file in advance.
- (2) The file name is structured as follows:

+(	<b>)</b> +)	l		+9	+	12
HL ->	Dr	File	Name		Extension	

Dr: Drive name

(3) Format of gaiji file The font in a gaiji file is structured with 32 bytes/character. The first character corresponds to kanji character code 800H, the second character corresponds to 801H, the third character corresponds to 802H, and so on.

Code 800H Code 801H Code 802H Code 803H	• • • • • •	32 32 32 32 32	bytes bytes bytes bytes

The font structure of one character is horizontally scanned. In the example shown below, one character must be stored in the order of 00, 00, 73, E0,...

Fig. 4.10 Gaiji file structure

(Example)

					•				•			•		•	•	0000H
	*	+	+			+	+	*								7380H
												-	-			2100H
l .	•	+	•	•	•	•	*	•	•	•	•	•	•	•	•	21008
•	•		•	•	•	•		•	•	•	•	•	•	•	•	21001
•		*	•		•				•	•	•	•			•	2100H
Ι.		*					+									2100H
		*					•				*	*	*			213EH
l •	•		4	-	4	-		•	•							35024
٠	٠	-	-	-	-	-	-	•	•	•	•	٠	٠	-	٠	Jruzn
		+		•						•	•					2102H
١.		*					-			•			-			2104H
		٠										*				2108H
Ι.	•		•	•	•	•	+	•	•	•	- <b>-</b>		•	•	•	21100
•	•	-	•	•	•	•	-	•	•	•	-	•	•	•	•	
		•				•	*	•	•	+			•	•	•	2120H
																2120H
1.	1		-		•	1	4	- 1	•		1	4	4	÷.	•	72050
٠		-		٠	٠		-	-	•	-		-				13050
	•	•	•	•	•	•	-					•	•	•	•	оооон
								_								J

```
[Function]
    KANJI (Disp) displays kanji (gaiji) in LCD.
[Entry parameters]
    B=01H: Function number
    HL=start address of data package (explained later)
[Return parameters]
    None
```

<Explanation>

The data package is structured as follows:



Kanji (gaiji) characters are displayed as much as N starting from the specified coordinates (X, Y) (dot position) toward right in the order of kanji codes 1, 2, 3,...

<Peference>

APPENDIX 9. SAMPLE 17

```
[Function]
KANJI (Print) prints kanji (gaiji).
[Entry parameters]
B=02H: Function number
HL=start address of data package (explained later)
[Return parameters]
None
```

```
<Explanation>
```

(1) The data package is structured as follows:

```
+0 +1 +2 +3
HL -> X Y N Code 1 Code 2 ..... Code N
```

Kanji (gaiji) characters are printed starting from the specified position X toward right in the order of kanji codes 1, 2,... Position Y is ignored even if specified. Position X must be a multiple of 8. If it is not a multiple of 8, the largest multiple of 8 but less than the specified value is used.

(2) The maximum value of N is 9 for a printer. Value N differs depending on the number of printer columns for a printer connected to RS-232C. However, N must satisfy the following expression: Number of dots that can be printed starting from the current head position  $\geq 8 \times X + 16 \times N$ 

<Reference>

Section 4.5 Printer APFENDIX 9 SAMPLE 17

```
[Function]
BACKLIGHT performs the backlight software control. This command is
effective only for EHT-10/2B. No operation is executed for EHT-10 and
EHT-10/2.
[Entry parameters]
C=backlight control
=00H: Backlight off
=01H: Backlight on
[Return parameters]
None
```

<Explanation>

(1) The following table shows the relationship between EHT-10/2B power switch and backlight:

Switch status	1	2	3
Power status	OFF	ON	ON
Backlight	X	X	0

O indicates that software control is allowed. X indicates that software control is not allowed.

- (2) System default is backlight on.
- (3) The system supports automatic backlight off. Backlight is automatically turned off when data is not input by key operation for a fixed time (default is 3 minutes) in key input wait status. See "Section 7.8.3 Backlight" for details.

<Reference>

APPENDIX 9 SAMPLE 10

[Function] INFORM checks the address of work area used by the system.						
[Entry parameters] C=system information code (explained later)						
[Return parameters] HL=address of system information						

<Explanation>

When this routine is called with register C containing one of the following system information codes, this routine sets the address of the system information in HL register:

System information code	Name	Contents
0	BJUSERROR	Error information for disk access operations. See BIOS READ.
1	BPINTEBL	Interrupt status for BEEP processing See BIOS BEEP.
2	CONFMOD	Specification of whether CONFIG is allowed to set each item. See "Section 9.4 System Menu Functions"
3	DISBNK	Bank information for bank switching
4	FUNC_TBL	Subroutine call table for key input operations. See BIOS CONIN.
5	LSTERR	Error information for printer output See BlOS LIST.
6	SKEYFLG	Specification of whether to suppress or allow calling a subroutine by key input operations
7	SRCBNK	Bank information for bank switching
8	SYSMMOD	Specification of whether to suppress or allow setting each item by system menu operations
9	TCAMAREA	Area used to pass the parameters for communication. See BIOS TCAM.

<Relations>

CONIN	(WBOOT+06H)
LIST	(WBOOT+OCH)
READ	(WBOOT+24H)
BEEP	(WBOOT+36H)
TCAM	(WBOOT+BDH)

<Reference>

Section 9.4 System Menu Functions APPENDIX 9 SAMPLE LIST 4.3 Entering Data with Touch Fanel or Keyboard Keys

4.3.1 Overview

Data is input for EHT-10 and EHT-10/2 by using the touch panel and keyboard, respectively. Both the touch panel and keyboard are controlled by the slave 7508 CPU when commands and data are exchanged with the main CPU through serial communications.

- (1) Entering data with touch-panel keys
  - 1 The 7508 CPU checks at every 30 msec whether any key has been pressed and, if one or more keys have been pressed, sets 80H in the key buffer of the 7508 CPU and sends an interrupt signal to the main CPU.
  - 2 During interrupt processing, the main CPU scans the touch panel and generates a position code only for the first key that it scanned. The main CPU then generates a function code from the position code and the key state, and stacks the position and function codes in the key buffer.
  - 3 The stacked key buffer data items are fetched and processed one by one by BlOS functions CONIN, CONST, and KEYIN.
- (2) Entering data with EHT-10/2 keyboard keys
  - 1 The 7508 CPU checks all keys at every 30 msec and , if one or more keys have been pressed, stacks the corresponding scan codes in the key buffer of the 7508 CPU and sends an interrupt signal to the main CPU.
  - 2 During interrupt processing, the main CPU receives the scan codes from main CPU, generates position and function codes, and stacks them in the key buffer.
  - 3 The stacked key buffer data items are fetched and processed one by one by BIOS functions CONIN and CONST.
- (3) 7508 CPU functions for controlling data input through keys

The 7508 CPU has the functions listed below to control data input through keys.

See Section 7.6.3 for 7508 CPU commands.

- 1 When a key is pressed In EHT-10, 80H is output as a scan code. (The actual scan code is generated during OS interrupt processing.) In EHT-10/2, the matrix position is output as the scan code (see Figure 4.11).
- 2 When two or more keys are pressed sequentially in EHT-10/2, corresponding scan codes are output sequentially.
- 3 When two or more keys are pressed simultaneously, only the code of the key scanned first is output.
- 4 All keys are scanned once in approximately every 30 msec.

- 5 The automatic repeat function is supported. The prohibition or permission of the automatic repeat function can be specified in a command or the BIOS CONOUT ESC sequence (the default is prohibition). The automatic repeat start time and repeat interval can be modified by a command.
- 6 The prohibition or permission of key interrupt can be specified in a command or BIOS MASKI.
- 7 A seven-byte area is allocated as a key input buffer. Seven bytes or one byte can be specified as the size of this key input buffer (the default is seven bytes).
- 8 The keys pressed after the key input buffer becomes full of data are ignored.
- 4.3.2 Functions supported by BIOS

The following BIOS functions are supported.

- (1) CONST : Checks the CON: input status.
- (2) CONIN : Inputs one character from the CON: device.
- (3) READER: Inputs one character from the RDR: device.
- (4) GETPFK: Reads the key code currently set.
- (5) PUTPFK: Registers a key code in the user toble.
- (6) TOUCH : Sets the indication for setting the key block of the touch-panel keys (only in SHT-10).
- (7) KEYIN : Initiates the input function that the system has, and receives data.

See Section 4.2 for details on the BIOS functions.

## 4.3.3 Position and Function Codes

(1) Position and function codes

EHT-10 and EHT-10/2 use two types of key codes: position codes and function codes.

A position code is a physical code which indicates the position where the key was pressed. This position code is set in the C register when BIOS function CONIN is executed. Figure 4.11 shows the relationship between key positions and position codes.

A function code is a logical value assigned to a position code. This function code is set in the C register when BIOS function CONIN is executed. Function codes can also be set by BIOS function PUTPFK or TOUCH (only in EHT-10). Table 4.4 lists the functions of function codes.

01	02	03	04	05
----	----	----	----	----

06	07	80	09	0A
OB	0C	٥٥	0E	OF
10	11	12	13	14
15	16	17	18	19
1A	18	1C	1D	10
1F	20	21	22	15

EHT-10/2 keyboard

Note : Values are represented in hexadecimal notation.

01	02	03	04	05
06	07	08	09	OA
ОВ	0C	00	OE	OF
10	11	12	13	14
15	16	17	18	19
1A	18	10	10	1E
1 <b>F</b>	20	21	22	23
24	25	26	27	28
29	2A	28	20	2D
2E	2F	30	31	32
33	34	35	36	37
38	39	38	30	3C
30	3E	3F	40	41
42	43	44	45	46

EHT-10 touch-panel keys

Fig. 4.11 Position code array

Table 4.4 Functions of function codes

Function code (hex)	Function
00 - DF	Returns the specified function code to the user without performing any processing.
E0 - EF	Calls a user function. The default function is only to return control. This function also enables an application program to modify the subroutine call address and add operations.
F0 - F5	Calls a system function. The default function is only to return control.
F6	<ul> <li>I/O forcible termination:</li> <li>If this function key is pressed, the I/O operation being performed can be suspended. The following I/O operations can be suspended:</li> <li>1. Waiting for sending or receiving through an RS-232C interface</li> <li>2. Waiting for printer ready status</li> <li>3. Beep processing</li> </ul>

F7	Paper feed (FEED): Feeds paper of the printer.
F8	Print (COPY): Outputs the data being displayed on the screen to printer as bit image data. In EHT-10, this function code performs no processing.
F9	000: If this function key is pressed, code 0 (30H) is returned three times consecutively.
FA	System MENU: If this function key is pressed, the system menu screen is initiated.
FB	Normal table (NORMAL): If this function key is pressed, the key status is set to normal mode.
FC	Kana table: If this function key is pressed, the keys status is set to kana mode. If overseas mode has been specified, chis function key performs no processing. (In EHT-10, string data is input.)
FD	Alphabetic character table (ALPH): If this function key is pressed, the key status is set to letter (alphabetic character) mode. (In EHT-10, string data is input.)
FE	Calculator (CALC): If this function key is pressed, the calculator is initiated.
FF	Invalid: If this function code is defined, the key is made ineffective and, even if the key is pressed, no operation is performed.

## (2) Subroutine call system

The subroutine call system has been incorporated in function codes EOH to FEH so that a specific service program can be executed by pressing the corresponding key. Table 4.4 lists the system default functions. The called subroutine is executed when BIOS function CONIN or CONST is called.

To execute a specific user service program, set the bank information and address of the service program in the address of the corresponding function code shown in the function table FUNC\_TBL) listed in Table 4.5.

Note the following when registering user service programs:

- 1 To use a stack frequently, set a new stack. In this case, return the stack to its original state when control is returned.
- 2 BIOS and BDOS cannot be called during service program execution. If required, directly call the module stored in ROM.

## Notes:

A function table consists of three bytes: one byte for bank information and two bytes for address information. The top address of the table is determined by FUNC TBL of BIOS INFORM.

Function code (HEX)         Relative position from the table         Remarks           Bank         Address         Remarks           FUNC TBL>         E0         +0         +1         +2           (F45ĀH)         E0         +0         +1         +2           E1         3         4         5         5           E2         6         7         8         9         10,11           E4         12         13,14         14         15         16,17           E6         18         19,20         21,22         22         28         23         24,25           E8         23         24,25         24         25         55         56           F0         48         49,50         51         52,53         Default is return           F0         F1         51         52,53         Default is return           F2         54         55,56         58,59         F4         60         61,62           F3         57         58,59         F4         60         61,62         F5           F4         60         61,62         F5         73,74         Print           F9         75					
Bank         Address           FUNC_TBL>         E0         +0         +1         +2           E1         3         4         5           E2         6         7         8           FUNC_TBL         ->         E2         6         7         8           E3         9         10,11         Default is return           E4         12         13,14         15         16,17           E6         18         19,20         24,25         16           E7         21         21,22         24,25         16           E8         23         24,25         16,47         17           F0         48         49,50         51         52,53           F2         54         55,56         56         57           F3         57         58,59         56         56           F4         60         61,62         67,68         1/0 forcible           F7         69         70,71         Paper feed         78           F8         72         73,74         Print         79,80         System menu           F8         81         82,83         Normal table		Function code (HEX)	Relative from the	e position e table	Remarks
FUNC_TBL      >       E0       +0       +1       +2       4       5         E1       3       4       5       5       2       6       7       8       0efault is return         E4       12       13,14       14       15       16,17       0efault is return         E4       12       13,14       15       16,17       0efault is return         E6       18       19,20       21,22       28       23       24,25         E8       23       24,25       24       25       24       25         E8       23       24,25       0efault is return       76         F0       48       49,50       51       52,53       0efault is return         F2       54       55.56       0efault is return       78         F4       60       61,62       65       66       67,68       I/O forcible termination         F7       69       70,71       Paper feed       78       79,80       System menu         F8       72       73,74       Print       79       79       70,71       000         FA       78       79,80       System menu       F8       81			Bank	Address	
F0       48       49.50         F1       51       52,53         F2       54       55.56         F3       57       58,59         F4       60       61,62         F6       66       67.68       1/0 forcible termination         F7       69       70,71       Paper feed         F8       72       73,74       Print         F9       75       76,77       000         FA       78       79,80       System menu         F8       81       82,83       Normal table         F0       87       88,89       Letter table         F0       87       88,89       Letter table	FUNC_TBL> (F45AH)	E0 E1 E2 E3 E4 E5 E6 E7 E8 L EE EF	+0 3 6 9 12 15 18 21 23 42 45	+1 +2 4 5 7 8 10,11 13,14 16,17 19,20 21,22 24,25 43,44 46,47	Default is return
F66667,68I/O forcible terminationF76970,71Paper feedF87273,74PrintF97576,77000FA7879,80System menuFB8182,83Normal tableFC8485,86Kana tableFD8788,89Letter tableFE9C91,92Calculator		F0 F1 F2 F3 F4 F5	48 51 54 57 60 63	49,50 52,53 55,56 58,59 61,62 64,65	Default is return
F7       69       70,71       Paper feed         F8       72       73,74       Print         F9       75       76,77       000         FA       78       79,80       System menu         FB       81       82,83       Normal table         FC       84       85,86       Kana table         FD       87       88,89       Letter table         FE       9C       91,92       Calculator		F6	66	67,68	I/O forcible termination
F8       72       73,74       Print         F9       75       76,77       000         FA       78       79,80       System menu         FB       81       82,83       Normal table         FC       84       85,86       Kana table         FD       87       88,89       Letter table         FE       9C       91,92       Calculator		F7	69	70,71	Paper feed
F9       75       76,77       000         FA       78       79,80       System menu         FB       81       82,83       Normal table         FC       84       85,86       Kana table         FD       87       88,89       Letter table         FE       9C       91,92       Calculator		F8	72	73,74	Print
FA7879,80System menuFB8182,83Normal tableFC8485,86Kana tableFD8788,89Letter tableFE9C91,92Calculator		F9	75	76,77	000
FB8182,83Normal tableFC8485,86Kana tableFD8788,89Letter tableFE9C91,92Calculator		FA	78	79,80	System menu
FC8485,86Kana tableFD8788,89Letter tableFE9C91,92Calculator		FB	81	82,83	Normal table
FD8788,89Letter tableFE9C91,92Calculator		FC	84	85,86	Kana table
FE 9C 91,92 Calculator		FD	87	88,89	Letter table
		FE	90	91,92	Calculator

Table 4.5 Function table

(3) Function code check mode

By setting the function check mode flag (YPFCMFLG: F21DH) to FFH, subroutine calling is not performed for function codes EOH to FEH and function codes can be returned by CONIN.

(4) Suppressing system key functions

The functions of function codes FO to FEH can be suppressed by setting the corresponding system key flag (SKEYFLG) bits to 0. If suppression is set it is assumed, unlike the function key check mode, that the key has not been pressed.



#### 1.3.4 Key buffer

In EHT-10 and EHT-10/2, the slave 7508 CPU has a key buffer and the main CPU has another key buffer. The main CPU also has a special buffer for inputting string data.

(1) 7508 CPU key buffer

Seven characters (default) or one character can be specified. Output a command directly for the 7508 CPU when making specification. (See Section 7.6.3 for details.)

Key scan codes are stacked in the 7508 CPU key buffer, and these codes are fetched one by one by interrupt processing performed by the main CPU. In EHT-10, however, the scan code is fixed to 80H.

(2) Main CPU key buffer

A 66-byte area capable of buffering the codes for 32 keys is allocated as the key interrupt buffer.

Two bytes are used for one key: one byte for a position code and one byte for a function code. The key buffer constitutes the ring structure.



F229H : YPFKCNT (2) Counter which indicates the data length . This counter is decremented whenever a character is fetched. 4.3.5 Key input beep When pressing a key is accepted, key input beep of 1024 Hz sounds for approximately 100 ms. Output, nonoutput, frequency, and the sounding time of the key input beep can be specified or modified by directly modifying the system area contents. If an external interrupt is suppressed while a key input beep is sounding. the sounding time is made longer because an external interrupt continues for 8 ms. Address : Variable name ( Number of bytes ) FO94H : BUZ FLG (1) Specifies output or nonoutput of key input beep. 0: Nonoutput Not O: Output (default) The default value is set when reset is performed. FO95H : KDFLTBZ (2) Specifies the key input beep sounding time in units of milliseconds. The default value is set when reset is performed. FO97H : BUZZHZ (1) Specifies the beep frequency. COH: 2048 Hz 80H: 1024 Hz (default) 40H: 512 Hz **OOH: Nonoutput** The default value is set when reset is performed. 4.3.6 Touch panel (EHT-10) (1) Overview In EHT-10, data is input from the transparent lattice-shaped keyboard consisting of five horizontal and 14 vertical grids (input points) mapped on the LCD screen. This keyboard is called a touch-panel keyboard. An application program assigns keys (see the explanation of BIOS TOUCH for function key codes and key displaying) to these touch grids and use them. By using this keyboard, data can be input at the very position where the entered data is displayed, and which enables the users to

EHT-10 also supports calculator mode so that simple four fundamental arithmetic operations can be performed by the application program. See Section 9.5 for calculator mode.

create applications flexible to their needs. EHT-10 supports letter and

kana modes so that alphanumeric and kana characters can be input.

(2) Positional relationship between LCD display areas and touch panel

A total of 70 switches (five horizontal and 14 vertical switches) have been set on the 12-column x 14-line LCD display possible area.

The size of one touch grid (name of one rectangular part) is the same as that of an area for displaying two characters, and a character is displayed at the center of the touch grid.

Figure 4.13 shows the positional relationship between the display areas and the touch panel.



- Fig. 4.13 Positional relationship between display areas and touch panel
- (3) Normal mode
- 1 Overview

Normal mode is a standard mode in which the user can use the keyboard by defining arbitrary function codes for position codes. (See the explanation of BIOS functions TOUCH and PUTPFK.)

2 Setting conditions

The key status is set to normal mode under the following conditions:

- (a) When the application program is initiated (including initiation by Menu. DLL) immediately after warm boot or restart power on In this case, the user must define function codes because all function codes are defined as invalid keys (FFH). Also, no data is displayed on the LCD screen.
- (b) When letter, kana, or calculator mode is released When the normal mode is restored from one of the above modes, the LCD display status is returned to the status set before the mode was switched to the above mode.

#### 3 Function codes

The user can define desired function codes in normal mode.

Figure 4.14 lists the system default values.

Notes: 1. The values are presented in hexadecimal notation. 2. Code FFH specifies suppression of the key function.

1	FF	FF	FF	FF	FF
2	FF	FF	FF	FF	FF
3	FF	FF	FF	FF	FF
4	FF	FF	FF	FF	FF
5	FF	FF	FF	FF	FF
6	FF	FF	FF	FF	FF
7	FF	FF	FF	FF	FF
8	FF	FF	FF	FF	FF
9	FF	FF	FF	FF	FF
10	FF	FF	FF	FF	FF
11	FF	FF	FF	FF	FF
12	FF	FF	FF	FF	FF
13	FF	FF	FF	FF	FF
14	FF	FF	FF	FF	FF

Fig. 4.14 Function codes (default values) in normal mode

# (4) Letter mode

- 1 Overview Letter mode is a standard input mode supported by the system for inputting alphanumeric characters. Up to 31 characters can be input by using keys in this mode.
- 2 Setting conditions The key status is set to letter mode under the following conditions:

  - (a) Function key FD is pressed.(b) Letter mode is specified as a parameter of BIOS KEYIN.
  - (c) Key 'EI' is pressed in kana mode.

## 3 Display format

- (a) Figure 4.15 shows the letter input screen.
- (b) Three lines of screen data are copied so that the current cursor line becomes the third line on the LCD screen. If the cursor is positioned in the first or second line of the window, however, the cursor line data is copied into the first or second line on the LCD screen. (In this case, the cursor is positioned in the first or second line.)



Fig. 4.15 Screen display in letter mode

# 4 Input procedure

(a) Up to 31 characters can be input in line-input mode. In this case, the input character string is processed as follows:

- When function key FD is pressed: The input character string is stored in the key input buffer (YPFKBUF: F434H) and is input one character at a time by CONIN.

- When BIOS KEYIN is executed: The input character string is stored in the buffer specified by the user.

(b) The pressed key is displayed at the current cursor position. If normal mode is restored after letter mode termination, the pressed key is not displayed.

- (c) The functions of special keys pressed in letter mode are as follows.
  - 'Return'key: Terminates inputting.

Pressing this key stores the input character string in the buffer and returns the current mode to normal mode. Value ODH is added to the end of the character string.

- 'kana' key: Sets the input mode to kana mode and display the kana input screen. This key is only effective when Japanese has been specified by the DIP switch.
- 5 Termination procedure

Pressing the 'Return' key terminates letter mode and returns to the original screen.

6 Function codes

Figure 4.16 shows the function codes used in letter mode.

Note : The values are represented in hexadecimal notation.

_				
FF	FF	FF	FF	FF
FF	FF	FF	FF	FF
FF	FF	FF	FF	FF
FF	FF	FF	FF	FF
FF	FF	FF	FF	FC
41	42	43	44	45
46	47	48	49	4A
48	<b>4</b> C	4D	<b>4</b> E	4F
50	51	52	53	54
55	56	57	58	59
37	38	39	2F	5A
34	35	36	2A	08
31	32	33	20	20
30	2C	<b>2</b> E	28	00
	FF         FF         FF         FF         41         46         48         50         55         37         34         31         30	FF       FF         FF       FF         FF       FF         FF       FF         41       42         46       47         48       4C         50       51         55       56         37       38         34       35         31       32         30       2C	FF       FF       FF         41       42       43         46       47       48         4B       4C       4D         50       51       52         55       56       57         37       38       39         34       35       36         31       32       33         30       2C       2E	FF       FF       FF       FF         41       42       43       44         46       47       48       49         4B       4C       4D       4E         50       51       52       53         55       56       57       58         37       38       39       2F         34       35       36       2A         31       32       33       2D         30       2C       2E       2B

Fig. 4.16 Function codes in letter mode



- 7 Remarks
  - (a) In letter (alphabetic character) mode, data is entered from the system screen. In this case, the data of the original user screen is saved.
  - (b) The input screen for letter mode is called the window which uses LCD lines 1 to 3.
- (6) Kana mode
- 1 Overview

Kana mode is a standard mode supported by the system for inputting kana characters. Up to 31 characters can be entered by using keys in this mode.

If overseas mode has been specified, this mode is ineffective.

- 2 Setting conditions
  - The key status is set to kana mode under the following conditions:
  - (a) Function key FC is pressed.
  - (b) Kana mode is specified as a parameter of BIOS KEYIN.
  - (c) The 'kana' key is pressed in letter mode.
- 3 Display format
  - (a) Figure 4.16 (is omitted in this manual) shows the kana input screen.
  - (b) Three lines of screen data are copied so that the current cursor line becomes the third line on the LCD screen. If the cursor is positioned in the first or second line of the window, however, the cursor line data is copied into the first or second line on the LCD screen. (In this case, the cursor is positioned in the first or second line.)
- 4 Input procedure
  - (a) Up to 31 characters can be input in line-input mode. (See letter mode.)
  - (b) The pressed key is displayed at the current cursor position.
  - (c) The functions of special keys used in kana mode are as follows.

5 Termination procedure

Pressing the 'Return' key terminates kana input mode and returns to the original screen.

#### 6 Function codes

Figure 4.18 shows the function codes used in kana mode.

Note : The values are represented in hexadecimal notation.

				_	
1	FF	FF	FF	FF	FF
2	FF	FF	FF	FF	FF
3	FF	FF	FF	FF	FF
4	B1	82	83	B4	85
5	86	B7	88	89	BA
6	88	BC	BD	8E	BF
7	СО	C1	C2	C3	C4
8	C5	C6	C7	C8	٢9
9	CA	CB	22	CD	CE
10	CF	DO	D1	D2	D3
11	D4	D5	D6	DF	DE
12	D7	D8	D9	DA	DB
13	DC	A6	DD	80	A5
14	FD	00	20	08	OD

Fig. 4.18 Function codes in kana mode

## 4.3.7 Keyboard (EHT-10/2)

## (1) Overview

In EHT-10/2, data is input from the keyboard.

The EHT-10/2 keyboard configuration is as follows: 1 Number of keys: 34 2 Number of LEDs: 3

All keys can be redefined arbitrarily by the application program (see BIOS PUTPFK).

Although the LEDs are used by the system for displaying the key mode, they can also be used by the user for BIOS CONOUT execution.

EHT-10/2 supports normal, letter, and kana modes as input modes. In kana mode, the Roman-character-to-kana-character conversion system has been incorporated.

EHT-10/2 also supports calculator mode so that simple four fundamental arithmetic operations can be performed by the application program. See Section 9.5 for calculator mode.

(2) Key arrangement



Fig. 4.19 EHT-10/2 keyboard key arrangement

- (3) Normal mode
- 1 Overview

Normal mode is a standard mode in which the user can use the keyboard by defining arbitrary function codes for position codes. (See BIOS PUTPFK.)

2 Setting conditions

The key status is set to normal mode under the following conditions:

- (a) Function key FB is pressed.
- (b) Normal mode is specified in BIOS KEYIN.
- (c) The application program is initiated immediately after warm boot or restart power on.
- 3 Functions

Sets the function key tables in normal mode tables, and turns all LEDs off.

### 4 Function codes

The user can define desired function codes in normal mode. Figure 4.20 lists the system default values.

FE	FD	FC	FB	FA
01	02	F7	F8	FA
04	05	06	07	08
37	38	39	2F	09
(7)	(8)	(9)	( /)	
34	35	36	2A	25
( 4)	(5)	(6)	(*)	(%)
31	32	33	2D	00
(1)	(2)	(3)	( -)	
30	(*1)	2E	28	UU
( 0)	(000	( .)	( +)	(Ret

F8: COPY F7: FEED FA: SYSTEM FB: NORM FD: ALPH FE: CALC

Fig. 4.20 Function codes in normal mode

\*1 Value 0 (30H) is returned three times sequentially. (Setting code is F9H.)

Notes:

1. The values in the figure are represented in hexadecimal notation.

2. The value enclosed in parentheses is an ASCII code corresponding to the function code.

3. Codes FAH to FEH, F7H, and F8H are assigned with system functions. Even if these keys are pressed, the key codes are usually not returned to the user.

- (4) Letter mode
- 1 Overview Letter mode is a standard input mode supported by the system for inputting letters (alphabetic characters).
- 2 Setting conditions The key status is set to letter mode under the following conditions:
  - (a) Function key FD is pressed.(b) Letter mode is specified in BIOS KEYIN.
- 3 Functions Sets the function key tables in normal mode tables, and turns only the LED for "letter (ALPH)" on.

4 Function codes The function codes to be used in letter mode can be defined arbitrarily by the user (see BIOS PUTPFK). Figure 4.21 shows system default values.

	Y			Y.
FE	FD	FC	FB	FA
41	42	43	44	45
( A)	( B)	( C)	( D)	( E)
46	47	48	49	4A
(F)	(G)	(H)	( 1)	( ))
4B	4C	4D	4E	4F
(К)	( L)	( M)	( N)	(0)
50	51	52	53	54
( P)	(Q)	( R)	( S)	( 1)
55	56	57	58	
(U)	( V)	(₩)	( X)	
59	5A	20	08	(Fet
(Y)	( Z)	(¿Þ)	(BS)	

Fig. 4.21 Function codes in letter mode

Notes:

 The values are represented in hexadecimal notation.
 The value enclosed in parentheses is an ASCII code corresponding to the function code.
 Keys FAH to FEH are assigned with system functions. Even if these keys are pressed, the key codes are usually not returned to the user.

(5) Kana mode

## 1 Overview

Kana mode is a standard mode supported by the system for inputting kana characters. Roman letters can be entered instead of kana characters. The entered Roman characters are converted to kana characters by the system. If overseas mode has been specified, this mode is ineffective.

#### 2 Setting conditions

The key status is set to kana mode under the following conditions:

- (a) Function key FC is pressed.
- (b) Kana mode is specified in BIOS KEYIN.

# 3 Functions

Sets function key tables in letter mode tables and turns only the LED for FC (kana) on. After this, kana characters can be entered through Roman character to kana character conversion.

4 Conversion table

Table 4.6 (is omitted in this manual) shows the Roman-character-to-kana-character conversion system.

4.4 Displaying Characters on LCDs

4.4.1 Overview

The LCD-related hardware configuration for EHT-10 is different from that for EHT-10/2. Accordingly, the OS specifications for displaying characters in EHT-10 are also different from those in EHT-10/2. Sections 4.4.3 and 4.4.4 explain the display specifications in EHT-10 and those in EHT-10/2, respectively.

Sections 4.4.2, 4.4.5, 4.4.6, and 4.4.7 provide integrated explanation of the functions supported by BIOS, character generator, cursor displaying, and details on the display work area for both EHT-10 and EHT-10/2.

4.4.2 Functions supported by BIOS

The following BIOS functions are supported:

(1) CONOUT: Outputs one character to the CON: device.
 (2) PUNCH: Outputs one character to the PUN: device.
 (3) GRAPHICS: Supports graphic functions.
 (4) TOUCH: Sets the indication for setting the key block of the touch-panel keys (only in EHT-10).
 (5) KANJI: Displays and prints kanji characters.

See Section 4.2 for details on BIOS functions.

4.4.3 Display specifications in EHT-10

(1) Overview

1 Hardware specifications

EHT-10 uses LCD controller T6963, and exchanges data and commands with the main CPU through parallel communications. The hardware specifications are as follows:

LCD display panel: 84 x 154 dots (12 characters x 14 lines)
Controller: T6963
Driver: X driver T7778 x 4

Y driver T6961 x 1

Duty ratio: 1/48
Frame frequency: 58.6 Hz
VRAM: 2 Kbytes (housed in the controller)

All display operations including displaying of characters and cursor and screen scrolling are performed in the graphic mode of the LCD controller under control of the operating system.

2 Sottware specifications

Character fonts
 Number of fonts: 228 + 13 (external characters for system)
 Font size: 5 x 7 dots (ordinary characters)
 6 x 8 dots (graphic characters)
 7 x 11 dots (external characters)

- Display area Character: 12 columns x 14 lines Graphic: 84 (horizontal) x 154 (vertical) dots - Screen User screen: Scroll and fixed screens are used as virtual screens, and these screens have a character buffer capable of storing 42 lines of data. System screen: The system screen has a character buffer capable of storing 14 lines of data (contents of one screen). - Character attributes Vertical, upper and lower ruler lines, comma, reverse, and secret - Cursor attributes Block/underline Blink/nonblink - Scroll The screen can be scrolled only in the vertical direction. - Character set adjustment to a specific country Character set registration can be performed for ten countries including Japan. (2) Screen mode Screen configuration EHT-10 has two virtual screens: system screen and user screen. These screens are automatically switched from one to the other by the system, and either of them is displayed on the LCD. Figure 4.11 shows the relationship between the system screen and the user screen. <System screen> The system screen is exclusively used by the system, and it has a character buffer capable of storing 12 columns x 14 lines. Use of the system screen does not affect the status of the user screen. The system screen is used for the following functions: System menu (CONFIG, DL/UL, DISK, or TEST) Calculator mode Letter/kana mode BDOS/CCP error screen Alarm screen Charge battery screen Disk Sum Check error screen

1



<System screen>

<User screen>

Fig. 4.22 System and user screens

The user screen can be used freely by the user. This screen is already active when the application program is initiated.

The user screen supports two modes: vertical display mode and horizontal display mode. Either of these modes can be selected in the BIOS CONOUT ESC sequence. The user screen is in vertical display mode immediately after application program initiation. The user screen has a character buffer consisting of up to 1008 bytes. A logical screen larger than the actual LCD screen size can be configured by assuming the user screen to be a virtual screen. Therefore, only part of the virtual screen contents is actually expanded in VRAM and displayed on the LCD screen. This part of the LCD screen on which the expanded VRAM data is displayed is called the "window".

The maximum virtual screen size is 12 columns x 42 lines in vertical display mode and 25 columns x 20 lines in horizontal display mode.

The virtual screen in vertical display mode consists of the scroll screen and the fixed screen. By making the window size on the scroll screen less than the LCD screen size, both the window and fixed screen can be displayed simultaneously on the LCD screen. The part of the fixed screen contents overlapped with the window contents cannot be viewed. Scroll and fixed screens can be controlled each individually by BIOS CONOUT.

2 Memory map (display buffer)

Figure 4.23 shows the memory map for the display buffer.



\*1 Two bytes (one byte for a character code and one bytes for the character attribute) are used for one display character in the character buffer.

Fig. 4.23 Display buffer memory map

3 Display area configuration

<Vertical display mode>

A total of 168 characters (12 characters x 14 lines) can be displayed in the LCD display area (84 (horizontal) x 154 (vertical) dots).

One display area consists of 7 x 11 dots. A character font and character attribute are written in this area. Each character is displayed on the LCD screen when the corresponding bit image data including the character attribute is written in the VRAM located in the LCD controller.



Fig. 4.24 Display area in vertical display mode

<Horizontal display mode>

In horizontal display mode, a total of 250 characters (25 columns x 10 lines) can be displayed by using 80 x 150 dots of the LCD display area (84 (vertical)  $\times$  154 (horizontal) dots) on the user screen. One display area consists of 6 x 8 dots. A character font and character attribute are written in this area.



Fig. 4.25 Display area in horizontal display mode

4 VRAM addresses and display positions

In EHT-10, VRAM is located in the LCD controller. The VRAM contents can be accessed directly from the main CPU by using a command.

Figure 4.26 shows the relationship between the VRAM addresses and display positions in the LCD controller.

Notes: 1. Each address is represented in hexadecimal notation, and indicates the relative address from the head of VRAM.

2. The VRAM save area configuration is the same as the VRAM's. Therefore, assume each address in this figure as the relative address from the head of the VRAM save area.



Fig. 4.26 Relationship between VRAM addresses and display positions
### (3) System screen

The system screen is used by the system, and it has a character buffer capable of storing the contents of one screen (12 columns x 14 lines) and a link flag table. Figure 4.27 shows the character buffer structure.

<--12 columns x 2 -----> (including attribute columns) Т T 1 0 2 1 3 22 23 2 24 25 26 27 46 47 3 48 49 50 51 1 70 71 4 b y t e S 14



System screen

Link flag table

Fig. 4.27 Character buffer structure

Notes:

1. Each value indicates the relative address from the head of the system screen.

2. The even-number-byte data is character (ASCII) data, and the odd-number-byte data is its attribute data.

Saving VRAM data 1

> When initiating the system screen, the system saves the contents of VRAM that has been used for user screen processing in the main memory. When system screen processing is terminated, the system restores the saved VRAM data. This enables the system screen to be used without affecting the user screen display data. The structure of the VRAM data save area is the same as that of VRAM. (See Figure 4.26.)

2 Normal and direct modes

> The system screen can be internally used in normal or direct mode. The normal mode is used to write the display data into the character buffer and VRAM during execution of the following functions:

- System menu (CONFIG, DL/UL, DISK, or TEST) - Calculator mode - Letter/kana mode - BDOS error - CCP error (Bad Load or Command Error) The direct mode is used to display data during processing such as interrupt processing. In this mode, no data is written into the character buffer but data is directly written into VRAM. The direct mode is used for the following functions: - Displaying Alarm - Displaying Charge Battery - Displaying Disk Sum Check error The direct mode can also be initiated in the system screen status. If the direct mode is initiated from the system screen, data other than the graphic data is redisplayed when the direct mode is terminated. (\*1) The operations performed at direct mode initiation and termination are as follows: (a) When initiated while the system screen is in normal mode At initiation: The display parameters for part of the system screen contents are saved. At termination: The saved display parameters are restored and the screen buffer contents are redisplayed according to these parameters. When initiated as the user screen (b) At initiation: The mode is switched to system screen mode, and VRAM data is saved. At termination: The saved VRAM data is restored, and the system screen mode is switched to user screen mode. Figure 4.28 shows the transition of user and system screens (normal and direct modes). \*1 Because graphic data is usually not used for the system screen, the screen contents are completely restored after direct mode termination. If user graphic data (including kanji

characters) is written in the input data display area in letter/kana mode, the graphic data is erased after message "Charge Battery" is displayed.



### Note:

If the Alarm screen is output when a Disk Sum Check error occurred, the sequence escapes from the Alarm screen, following which the data is redisplayed by the Disk Sum Check error processing routine.

Fig. 4.28 Screen mode transition

(4) Scroll and fixed screens

The user screen of EHT-10 has incorporated the concept of virtual screen. In vertical display mode, the user screen consists of the scroll screen and the fixed screen, and has a character buffer capable of storing up to 12 columns x 42 lines. The scroll screen is a character screen consisting of up to 12 columns x 42 lines, and part of the screen contents is displayed through the window on the LCD screen. The fixed screen is a character screen having a fixed size of 12 columns x 14 lines. This screen is output logically behind the window and, when the window is made smaller than the fixed screen, only the part not hidden by the window can be viewed. The fixed screen is made ineffective when the scroll screen size exceeds 28 lines.

SOFTWARE Page 4 - 155

Fixed and scroll screens can be controlled each independently by BIOS CONOUT (\*1). Handling one screen does not affect the other screen.

Figures 4.29 and 4.30 shows the fixed screen structure and the relationship between the scroll and fixed screens, respectively.

\*1 The screen that can currently be controlled is called an active screen. An active screen can be modified by the BIOS CONOUT ESC sequence.

Command: ESC+D1H+n n=0: Scroll screen n=1: Fixed screen



Fig. 4.29 Scroll screen structure (14 < R < 42)

Notes:

1. Each value indicates the relative address from the head address of the virtual screen.

2. The even-number-byte data is character (ASCII) data, and the odd-number-byte data is its attribute data.



Fig. 4.30 Relationship between fixed and scroli screens

1 Scroll screen and window

The number of the scroll screen columns is fixed to 12 which is the same as the number of LCD display columns. The number of the scroll screen lines can be modified to a maximum of 42 by CONOUT (\*1). Part of the scroll screen contents is displayed on the LCD screen by the window (\*2).

The window size and the display position on the LCD screen can be specified by CONOUT (\*3). The window size is set to 14 lines when the application program is activated.

\*1 ESC+DOH+n [Setting the screen size]

\*2 The window is usually displayed starting from the home position on the scroll screen, and then scrolled sequentially.

The window can also be moved to an arbitrary position on the scroll screen by using a CONOUT control code.

10H or ESC+96H [Scroli up by one line] 11H or ESC+97H [Scroll down by one line] \*3 ESC+D8H+M+N [Setting the window]

M: Window start line N: Window end line

2 Window scroll mode

There are two window scroll modes: follow mode and unfollow mode. The scroll mode can be changed by CONOUT (\*1).

<Follow mode>

In this mode, the window follows the cursor. If the unfollow mode in which the cursor is positioned outside the window is changed to follow mode, the window is automatically moved to the cursor position.

<Unfollow mode>

In this mode, the cursor does not follow the window. Therefore, the window does not move when data is written for the virtual screen (scroll screen). If the cursor is positioned outside the window when data is written for the virtual screen, the screen is automatically scrolled (\*2). However, the display position remains unchanged.

\*1 ESC+95H+M [Setting a scroll mode]

M=0: Follow mode (default) M=1: Unfollow mode

\*2 By modifying the system area contents, the mode in which the cursor is always positioned inside the window can be set (LSCROLMD).

3 Fixed screen

The fixed screen is automatically allocated by the system when the scroll screen size does not exceed 28 lines. If the window size is not less than the LCD size, none of the fixed screen contents can be viewed because the fixed screen is output logically at the behind of the window screen of the scroll screen. Because the fixed screen is not scrolled, it is very advantageous when used for displaying fixed data such as touch-panel keys. The structure of the fixed screen is the same as that of the scroll screen whose size is set to 14 lines.

(5) Horizontal display mode

If horizontal display mode is specified in BIOS CONOUT (\*1), 25 columns x 10 lines can be displayed on the LCD screen (\*2). Because one display area consists of 6 x 8 dots in horizontal display mode, the external font of 7 x 11 dots for vertical display mode cannot be used (\*3).

Only reverse and secret can be specified as the character attributes.

In horizontal display mode, the virtual screen size can be set to a maximum of 25 columns x 20 lines. The fixed screen is not allocated and the window is fixed to 25 lines x 10 lines (LCD size).

SOFTWARE Page 4 - 158

\*1 ESC+DAH+n [Changing display mode] n=0: Vertical display mode n=1: Horizontal display mode
\*2 See Item (2) 3 "Display area configuration" for the configuration of the display area on the LCD screen.
\*3 In horizontal display mode, external characters of 6 x 8 dots can be registered.
ESC+EOH+n+P(1)+ ... +P(8) n: External character code (EOH to FFH) P(i): Font pattern

Notes:

Because BIOS TOUCH does not support horizontal display mode, characters are displayed vertically by the BIOS TOUCH function.

(6) Attributes

Attributes can be specified for each display character. Each display character is displayed according to the specified attributes and font data.

The attribute bit configuration is as follows.



- \* : Can be specified only in vertical display mode
- 1. Horizontal ruler line
  - (a) When the horizontal ruler line is specified and the vertical ruler line is not specified, the horizontal ruler lines are displayed in full stroke width. (See Figure 4.31 (a).)
  - (b) When both the horizontal and vertical ruler lines are specified

    When the horizontal ruler line is specified, only the right half of the horizontal ruler line is displayed. (See Figure 4.31 (b).)
    When the horizontal ruler line is specified at the previous character position, only the left half part of the horizontal ruler line is displayed. (See Figure 4.31(c).)



- (a) Specifying only the horizontal ruler line
- (b) Specifying both horizontal and vertical ruler lines
- (c) Specifying the vertical ruler line and specifying the horizontal ruler line for the previous character

Fig. 4.31 Font data and ruler line specification

2 Vertical ruler line

When the vertical ruler line is specified, the vertical ruler line is displayed at the character position.

3 Comma

When a comma is specified, a comma is displayed at position (7,10) of the font data.

4 Reverse

When reverse is specified, the font data of  $7 \times 11$  dots is inverted and displayed. If a comma or ruler line is also specified, the data corresponding to the attribute is also inverted and displayed.

5 Secret

When secret is specified, the corresponding data is displayed as blank. Even if another attribute is specified in this case, the attribute is ignored but written in the attribute data.

SOFTWARE Page 4 - 160

(7) Link flag

The link flag indicates whether the line is logically continuous from the previous line.

The state of this flag is referenced by BASIC when keys are pressed, and set or released under the following conditions:

1 Setting

A character is displayed when the cursor is positioned in the last column of the line, and the cursor has moved in the first column of the next line. (Value OlH is set.)

- 2 Release
  - (a) When "Clear Screen & Home" is output The link flag table of the current screen is reset.
  - (b) When "Erase End of Screen" is output The link flags of the current cursor line and subsequent lines are reset.



Figure 4.32 shows the link flug table structure.

Link flag table of User screen

Notes:

1. The link flag value is OOH or O1H.

OOH: Continuous OIH: Incontinuous

2. The link flag table of the user screen is the one when the scroll screen consists of 28 lines. If the scroll screen size exceeds 28 lines, the link flags for the lines that exceeded the 28 scroll screen lines are set in the link flag table area of the fixed screen. In horizontal display mode, the first 20 bytes of the scroll screen are used of the link flag table.

(8) Graphics

Graphic data is directly written into VRAM. Graphic data can be written and output with character data. If the screen is scrolled and the displayed graphic data overflows from the screen, the overflowed graphic data is lost and, even if the screen is scrolled back, the lost data cannot be redisplayed (\*1).

The graphic screen consists of horizontal 84 dots x vertical 154 dots.

GRAPHICS and KANJI are supported as BIOS functions for writing graphic data.

Note:

Graphic data is also lost when the display position or size of the window is modified.

- 4.4.4 Display specifications in EHT-10/2
- (1) Overview
- 1 Hardware specifications

The LCD display screen in EHT-10/2 uses a uniform matrix of 120 x 32 dots that can be dynamically activated. The hardware specifications are as follows:

- Scroll function: Vertical direction dot scroll
- Others
   Display ON/OFF function is supported.
- 2 Software specifications

Character fonts
Number of fonts: 228 + 13 (external characters for system)
Font size: 5 x 7 dots (ordinary characters)

6 x 8 dots (graphic and external characters)
Display area

Character: 20 columns x 4 lines
Graphic: 120 (horizontal) x 32 (vertical) dots
Screen
User screen: Two screens (each a virtual screen consisting

of 20 columns x 25 lines)

System screen: One screen

Character attributes: None (the system screen has)

- Cursor attributes
   Block/underline
   Blink/nonblink
   Scroll
   The screen can be scrolled only in the vertical direction.
   Character set adjustment to a specific country
   Character set registration can be performed for ten countries including
   Japan.
- (2) Screen mode

1

Screen configuration EHT-10/2 has a total of three screens: one system screen and two user screens. Each screen has VRAM and character buffer, and can be controlled completely independently. However, only the two user screens can be used by the user. The user cannot directly control displaying data on the system screen. Figure 4.33 shows the relationship between the system screen and user screens.

<System screen>

The system screen has a character buffer with the same size as the LCD screen size (20 columns x 4 lines), and used for the following functions:

- System menu (CONFIG, DL/UL, DISK, or TEST)
- Calculator mode
- BDOS/CCP error
- Alarm screen
- Charge Battery screen
- Disk Sum Check error screen

<User screens>

User screens can be used freely by the user. These screens are already active when the application program is initiated. The configurations of two user screens are the same, and can be switched freely from one to the other by the application program. A user screen has a character buffer consisting of 20 columns x 25 lines. A logical screen larger than the actual LCD screen size can be configured by assuming the user screen to be a virtual screen. Therefore, only part of the virtual screen contents is actually expanded in VRAM and displayed on the LCD screen. This part of the LCD screen on which the expanded VRAM data is displayed is called the "window".



User screen 1 User screen 2 System screen Fig. 4.33 System screen and user screens

2 Memory map (display buffer)

Figure 4.34 shows the memory map for the display buffer.



Fig. 4.34 Display buffer memory map

# 3 VRAM structure and display area

This Item explains the VRAM structure and how to use it. A VRAM relative address corresponds to an LCD panel dot. Figure 4.36 shows the relationship between VRAM data and LCD panel dots.



\*1 : Relative address from the VRAM head

Fig. 4.36 Relationship between VRAM relative addresses and LCD display panel dots

The display start address for displaying data on the LCD panel is determined from the following:

VRAM head address (LSCRVRAM) Vertical offset value (LVRAMYOF)

The vertical offset value indicates the vertical relationship between VRAM locations and LCD panel dots. The VRAM data is displayed starting from the location indicated by the offset from the bottom end of VRAM and, when processing has reached the VRAM bottom end, VRAM data is displayed starting from the top of VRAM.

The system uses this function when scrolling the screen vertically. Figure 4.35 shows the relationship between VRAM relative byte addresses and LCD panel dots when the offset value is set to 2.

32	33	34		46	47	
			VRAM (Vertical offset =2)			
496	497	498		510	511	
0	1	2		14	15	< *1
16	17	18		29	30	
				~		•

Relative address from the VRAM head

\*1 :This line is the first VRAM line.

Fig. 4.35 Relationship between VRAM and LCD panel (vertical offset value = 2)

The configuration of the system area used for displaying VRAM data is as follows.

Address : Variable name (Number of bytes)

```
F253H : LSCRVRAM (2)
```

The head address of VRAM1 or VRAM2 whose contents are being displayed on the LCD screen is stored. The value must be E000H or larger, and a 512-byte boundary must be taken into account.

```
F266H : LVRAMYOF (1)
The vertical VRAM offset value is stored.
O < LVRAMYOF < 31
A multiple of 8 is usually stored.
```







Fig. 4.37 Display area configuration

(3) System screen

The system screen is used by the system, and it has a character buffer capable of storing the contents of one screen (20 columns x 4 lines) and a link flag table.

One display character shares two bytes (one byte for the character code and one byte for the attribute) in the system screen character buffer. The system screen character buffer size is 160 bytes.

Figure 4.38 shows the relationship between the character buffer and link flag table for the system screen.



Character buffer

link flag (4 bytes)

Fig. 4.38 Relationship between character buffer and link flag table

\*1 Each value is the relative address from the head of the character buffer.

SOFTWARE Page 4 - 168

\*2 The even-number-byte data is character data, and the oddnumber-byte data is its attribute data.

<Normal and direct modes>

The system screen can be internally used in normal or direct mode.

The normal mode is used to write the display data into the character buffer and VRAM during execution of the following functions:

System menu (CONFIG, DL/UL, DISK, or TEST)
Calculator mode
Letter/kana mode
BDOS error
CCP error (Bad Load or Command Error)

The direct mode is used to display data during processing such as interrupt processing. In this mode, no data is written into the character buffer but data is directly written into VRAM.

The direct mode is used for the following functions:

Displaying an alarm
Displaying Charge Battery
Displaying a Disk Sum Check error

The direct mode can also be initiated in the system screen status. If the direct mode is initiated from the system screen, the following operations are performed:

At initiation: After the display parameters for part of the system screen contents are saved, data is written into VRAM (displayed on the screen). At termination: After the screen buffer contents are rewritten into VRAM, the saved display parameters are restored.

The system screen contents other than graphic data are redisplayed by the above operations. (Because the system usually does not use graphic data, the screen contents are completely restored after use in direct mode.)

Figure 4.39 shows the transition of user and system screens (normal and direct modes).



### Note:

If the Alarm screen is output when a Disk Sum Check error occurred, the sequence escapes from the Alarm screen, following which the data is redisplayed by the Disk Sum Check error processing routine.

- (4) User screen
- 1 Virtual screen

Like EHT-10, EHT-10/2 has incorporated the concept of virtual screen which enables the application program to use a screen logically larger than the actual LCD screen (20 columns x 4 lines). The maximum size of the EHT-10/2 virtual screen is 20 columns x 25 lines.

The virtual screen is a character screen, and part of the screen contents is displayed through the window on the LCD screen.

Figure 4.40 shows the virtual screen structure. The attribute data is not added to the virtual screen contents.





Note:

Each value used in the figure is a relative address from the head of the virtual screen.

2 Switching user screen

There are two types of user screens: user screen 1 and user screen 2. These two screens are identical and can be switched arbituarily from the one to the other by the user for use. The user screen can be switched by BIOS CONOUT.

Sequence: ESC+D1H+n n=0: Screen 1 n=1: Screen 2

## 3 Window scroll mode

There are two window scroll modes: follow mode and unfollow mode. The scroll mode can be changed by CONOUT. The window scroll modes for EHT-10/2 are the same as those for EHT-10. See Section 4.4.3 for details on scroll modes.

(5) Link flag

The link flag indicates whether the line is logically continuous from the previous line. The state of this flag is referenced by BASIC when keys are pressed, and set or released under the following conditions:

1 Setting

A character is displayed when the cursor is positioned in the last column of the line, and the cursor has moved in the first column of the next line. (Value OIH is set.)

- 2 Release
  - (a) When "Clear Screen & Home" is output The link flag table of the current screen is reset.

SOFTWARE Page 4 - 171

(b) When "Erase End of Screen" is output The link flags of the current cursor line and subsequent lines are reset.

Figure 4.41 shows the link flag structure.



Note:

The link flag value is OOH or O1H. DOH: Continuous D1H: Incontinuous

Fig. 4.41 Link flag table

(8) Graphics

Graphic data is directly written into VRAM. Graphic data can be written and output with character data. If the screen is scrolled and the displayed graphic data overflows from the screen, the overflowed graphic data is lost and, even if the screen is scrolled back, the lost data cannot be redisplayed.

The graphic screen consists of horizontal 84 dots x vertical 154 dots.

GRAPHICS and KANJI are supported as BIOS functions for writing graphic data.

- 4.4.5 Character generator
- (1) Overview

EHT-10 and EHT-10/2 have a character generator in the system ROM. The character generator codes and font sizes are as follows.

OOH to 7FH: 5 x 8 dots 80H to 9FH: 6 x 8 dots AOH to OFH: 5 x 8 dots EOH to FFH EHT-10 (vertical display mode): 7 x 11 dots EHT-10 (horizontal display mode): 6 x 8 dots EHT-10/2: 6 x 8 dots EOH to FFH indicate external characters that can be defined freely by the user (\*1). \*1 ESC sequence for registering external characters ESC+EOH+n+p(1)+ ... +p(i) n: External character code (EOH<sup>S</sup> < <sup>Sn<sup>S</sup></sup> < <sup>SFFH</sup>) P: Font pattern (i = 8 or 11)

(2) Font format

Each font consists of five, six, or 11 bytes, and is recorded in the memory. Each character is recorded in a laid format as shown below. (The \*\* grid indicates bit=1 and the blank grid indicates bit=0.)





Fig. 4.42 Front format

(3) Character set adjustment to a specific country

EHT-10/2 supports character sets for ten countries including Japan. See Section 2 "CHARACTER CODE TABLE FOR OVERSEAS SPECIFICATIONS" in APPENDIX for the character set used in each country.

The following table lists the character sets to be adjusted to specific countries.

Table 4.7 Characters sets to be adjusted to specific countries

Country name	YLCOUNTRY
U.S.A. (ASCII)	OFH
France	OEH
Germany	0DH
U.K.	ОСН
Denmark	Овн
Sweden	OAH
Italy	09H
Spain	08H
Japan	07H
Norway	06H

SOFTWARE Page 4 - 174

The character set can be changed by CONOUT (\*1) or CONFIG. If the character set is changed, the printer character set is also changed automatically. For Norway, however, the printer character set is set to the ASCII character set. Address : Variable name (Number of bytes) F5F3H : YLDFLTC (1) Default value displayed for each country This default value is initialized by the system reset and set by CONFIG. F5F4H : YLCOUNTRY (1) Current value displayed for each country These specifications are set by the ESC sequence. YLDFLTC contents are copied during WBOOT processing. (4) Character generator table configuration 1 Pointer table EHT-10 and EHT-10/2 have five character generator tables according to the character generator codes. These character generator tables are pointed according to the following pointer table. Address : Variable name (Number of bytes) F1B8H : RLCGENX (3) Pointer data for character generator tables OOH to 1FH F1BBH : RLCGENN (3) Pointer data for character generator tables 20H to 7FH F18EH : RLCGENG (3) Pointer data for character generator table containing 80H to 9FH F1C1H : RLCGENK (3) Pointer data for character generator table containing AOH to DFH F246H : RLCGENU (3) Pointer data for character generator table containing EOH to FFH The pointer data structure is as follows. +0—> Font size of one character EHT-10: 5, 6, or 11 is possible. +1 EHT-10/2: 5 or 6 is possible +2 Character generator table address (8000H or later of the system bank or standard

RAM)

2 Character generator table

A character generator table contains font data items in the ascending order of character generator codes.

3 Modifying character generator table contents

By modifying the above pointer table, user-dependent character generator tables can be created. In this case, the new character generator tables should be set in address 8000H or later of the standard RAM (subbank 0). The pointer table value is not initialized until system reset is performed. The RLCGENU value is set to the default value when an error such as system menu, Alarm, Menu, DLL, or BDOS error occurred.

- 4.4.6 Displaying cursor
- (1) Cursor types

There are four types of cursors, and one of which can be specified by CONOUT (\*1).

- 1 Block cursor that blinks
- 2 Elock cursor that does not blink
- 3 Underline cursor that blinks
- 4 Underline cursor that does not blink

\*1 ESC+D6H+n [Setting the cursor type]

(2) Block cursor

All the contents of one display area are inverted and displayed. EHT-10 vertical display: 7 (horizontal) x 11 (vertical) dots EHT-10 horizontal display: 6 (horizontal) x 8 (vertical) dots EHT-10/2: 6 (horizontal) x 8 (vertical) dots

(3) Underline cursor

The bottom dot line of one display area is inverted and displayed.





EHT-10 Horizontal Display EHT-10/2

EHT-10 vertical display

Fig. 4.43 Underline cursor

(4) Blinking

Cursor blinking is performed by OVF interruption. The cursor blinks at intervals of approximately 500 milliseconds. The blink interval can be changed by modifying the system area (BLNKTIME) contents.

(5) Cursor-related system work areas

Address : Variable name (Number of bytes)

F075H : BLNKSTAT (1) Flag which indicates whether the cursor blinks OOH: Blink O1H: Not blink This flag is referenced by the blink processing routine.

FO76H : BLNKCNT (1) Blink counter This counter is incremented by 1 wnenever an overflow interrupt is made. When the value of this counter exceeds the BLNKTIME value, the cursor is inverted.

FO77H : BLNKRVRS (1) This indicates the cursor status during blinking. OOH: Cursor ON FFH: Cursor OFF F078H : BLNBKTIME (1) This specifies the blink interval. The initial value is O4H, and repeats cursor inversion at intervals of approximately 500 milliseconds. F25FH : LCURSOR (1) Flag which indicates the cursor status. Bit 7: Cursor mode 0: Display 1: Not display Bits 6 to 2: Don't care Bit 1: Cursor type 0: Block 1: Under line Cit 0: Specifies blink. 0: Blink 1: Not blink 4.4.7 Details on work areas for displaying (1) Work areas related to screen modes Address : Variable name (Number of bytes) F242H : LDSPMOD (1) Flag which indicates the current display mode (only for EHT-10) 0: Normal mode 1: Alphanumeric mode 2: Kana mode 3. Calculator mode F244H : LLCDMOD (1) Flag holding the current screen mode 0: User screen 1: System screen (normal mode) 2: System screen (direct mode) F245H : LUWDMOD (1) Flag which indicates existence or nonexistence of a fixed screen area 0: A fixed screen exists. 1: A fixed screen does not exist. F24FH : LWORKBFO (47) Area holding the current screen display status (See Item  $(\overline{2})$  for details.) F27EH : LWORKBF1 (47) Area used to replace the current screen data with the old screen data when the user screen mode is changed. The configuration of this area is the same as that of LWORKBFO. The contents of this area are replaced by the LWORKBFO contents when the screen is switched between the scroll screen and the fixed screen in EHT-10 or between user screen 1 and user screen 2 in EHT-10/2.

F2ADH : LWORKBF2 (47) All contents of this area are replaced by the LWORKBFO contents when the screen is switched between the user screen and the system screen. The configuration of this area is the same as that of LWORKBFO. F2DCH : LSCMODE (1) This indicates the current status of the display parameter. 0: System screen (Data is being replaced between LWORKBFO and LWORKBF2.) 1: EHT-10: Scroll screen EHT-10/2: User screen 1 2: EHT-10: Fixed screen EHT-10/2: User screen 2 (Data is being replaced between LWORKBFO and LWORKBF1.) F2DDH : LSCMDSV (1)Area for saving the LSCMODE value when the screen is switched between the user screen and the system screen. F2DFH : DSPTYPE (1) Flag which indicates the vertical or horizontal display mode in EHT-10 0: Vertical display mode 1: Horizontal display mode F2EOH : DSPTPSV (1) Area for saving the DSPTYPE value when the screen is switched to the system screen in EHT-10 F6A7H : SVCRSR (6) Area for saving the screen status when the system screen mode is switched from normal to direct F6B5H : BTRYDSP (2) Area for saving the old screen status when Battery Fail occurred (2) Work areas whose contents are saved at screen switching Address : Variable name (Number of bytes) F24FH : LSCADDR (2)Screen buffer head address (address 8000H or later) F251H : LSCSIZE (2) Screen buffer size F253H : LSCRVRAM (2) EHT-10: VRAM data save area head address EHT-10/2: VRAM head address The above address must be saved in address 8000H or later. F255H : LLNKFLG (2) Link flag table head address (Must be saved in address 8000H or later) F257H : LSCSIZEX (1) Number of screen buffer columns (fixed value) EHT-10: 12 EHT-10/2: 20 SOFTWARE Page 4 - 179



SOFTWARE Page 4 - 180

```
F260H : LSCCPOSX (1)
     This indicates the vertical cursor position in the screen buffer.
F261H : LSCCPOSY (1)
     This indicates the horizontal cursor position in the screen buffer.
F262H : LWDXMIN (1)
F263H : LWDYMIN (1)
     This indicates the upperleft position of the window in the screen
     buffer.
     LWDXMIN: Horizontal direction (fixed to 1)
     LWDYMIN: Vertical direction
F264H : LWDCPOSX (1)
F265H : LWDCPOSY (1)
     This indicates the cursor position in the window.
     LWDCPOSX: Horizontal direction
     LWDCPOSY: Vertical direction
F266H : LVRMYOF : (1)
     This indicates the vertical VRAM offset value in EHT-10/2.
F267H : LCURATR (1)
     Cursor position attribute data
F2_08H : LOLGATR (1)
     Attribute data for the column immediately before the cursor position
F_{269H} : LCRWAIT (1)
     Flag for adjusting the cursor position when CR is specifieu immediately
     after one character is displayed in the rightmost column on the screen
     O: Being in CR-waiting check status
     1: Not being in CR-waiting check status
F26AH : (Reserved)
F26BH : LFKSTAT (1)
F26CH : LFKADDR (2)
     The status (LFKSTAT) of the function according to the BIOS CONOUT
     control code and execution address (LFKADDR)
     The status and execution address obtained after retrieval of tables
     LSCRTB1, LESCTG1, and LESCTB2 are stored.
F26EH : LESCFLG (1)
F26FH : LESCCNT (1)
F270H : LESCPRM (14)
     ESC sequence processing work area
     LESCFLG: ESC sequence receiving status
          0: The ESC code has not been accepted.
          1: Only the ESC code has been accepted.
          2: The ESC and first parameter (command code) have been accepted.
     LESCCNT: This indicates the number of parameters (excluding the command
     code) accepted by the ESC sequence.
     LESCPRM: This is an area for storing the parameters accepted by the ESC
     sequence.
```

```
(3) Work areas used for displaying control
Address : Variable name (Number of bytes)
F074H : DSPFLAG (1)
    Display flag for EHT-10/2
    OOH: LCD display OFF
    80H: LCD display ON
F243H : WTONLY (1)
    Flag which indicates whether to move the cursor after displaying one
    character
    OOH: Move the cursor
    01H: Does not move the cursor
F249H : LSCRTB1 (2)
F24BH : LESCTB1 (2)
F24DH : LESCTB2 (2)
     Function of the control code used in BIOS CONOUT
    Head addresses of the command tables for ESC sequence functions
    LSCRTB1: For control codes OOH to 1FH
    LESCTB1: For ESC sequences common to EHT-10 and EHT-10/2
    LESCTB2: For ESC sequences different between EHT-10 and EHT-10/2
F683H : LVRAMADR (2)
F685H : LVRAMDT (24)
    Work area used to read VRAM data
    LVRAMADR: Read start address
    LVRAMDT: Area to store the read data
F69DH : SCRLFG (1)
    The contents of this work area are set to 1 when the screen buffer is
    scrolled by displaying one character or executing LF.
F69EH : LW_SADDR (2)
F6A0H : LW DADDR (2)
F6A2H : LW BPOS (2)
F6A4H : LW LADDR (2)
    Work area used to scroll the screen in EHT-10
```

4.5 Printer

4.5.1 Overview

EHT-10 and EHT-10/2 support a printer unit and a printer having an RS-232C interface as LST: devices. The printer unit is connected to the cartridge interface. The LST: device can be selected by using CONFIG as well as the I/O byte (see Section 2.6.5).

4.5.2 Functions supported by BIOS

The following BIOS functions are supported:

- (1) LIST : Outputs one character to the LST: device.
- (2) LISTST : Checks the LST: device status.
- (3) SCRNDUMP: Dumps the screen contents (not supported by EHT-10).
- (4) KANJI : Prints out kanji characters.

See Section 4.2 for details on the BIOS functions.

4.5.3 Character set adjustment to a specific country

(i) Adjusting a character set to a specific country

In EHT-10 and EHT-10/2, the printer character set is automatically changed according to the display character set by outputting the international character specification sequence. (ESC+'R'+n) to the printer.

The international character specification sequence is output when the first LIST is output under one of the following conditions:

- 1 After the LST: device contained in the I/O byte is modified
- 2 After warm boot
- 3 After power on (\*1)
- 4 After the display character set is changed

Table 4.8 shows the relationship between display character sets and international character specification codes (value n of ESC+'R'+n).

\*1 Note that, if the power is turned off while the application program is outputting the ESC sequence, the application program outputs the international character specification sequence when the power is turned on subsequently. Therefore, subsequent data may not be output correctly by the printer. (This symptom occurs only when outputting to the RS-232C interface.)

Display character set	Printer unit	RS-232C
ASCII France Germany U.K. Denmark Sweden Italy Spain Japan	0 1 2 3 4 5 6 7 8	0 1 2 3 4 5 6 7 8
Norway	0(*2)	9

Table 4.8 Display character sets and international character specification codes

\*2 For a printer unit in Norway, the display character set is set to ASCII. (This is because the printer unit does not support Norway.)

(2) Prohibiting character set adjustment to a specific country

Registering a character set for a specific country can be prohibited by modifying the contents of system parameter PRTINIT (F1CCH).



- 1.5.4 Printer unit
- (1) Overview

The printer unit is a cartridge option, and connected to the cartridge interface when used. The cartridge is used in HS mode (\*1), and the device code (\*2) is 07H. The printer unit is micro dot printer Model-180 capable of printing 24 columns and copying two sheets. Because LA-180 is used as the printer controller, the application program can control the printer unit in a variety of ways. Because the printer unit consumes not a small amount of current, the controller is placed in power down mode to save power consumption when the printer is not used for a fixed time. Data is output to the printer by making ready interrupt, and this enables the system to perform printing concurrently with other processing.

\*1 and \*2 See Section 7.3 for cartridge mode and device codes.

- (2) Functions supported by the system
- 1 Character set adjustment to a specific country

See Section 4.5.3. If the display character set is set to Norway, the system automatically sets the display character set to ASCII because the printer unit does not support Norway.

2 Screen dump function

This function is supported only by EHT-10/2.

3 Paper feed function

If the paper feed (FEED) key is pressed, the paper is fed one line (8 dots). The paper feed function is enabled only by pressing a desired key to which function key code F8H has been assigned.

4 Concurrent processing during printing

The system has a printer unit output buffer (PRNBUF) consisting of 128 bytes, and from which data items are fetched one by one by the ready interrupts made by the printer. This enables the system to perform printer output processing concurrently with other processing. When the output buffer is full of data, output processing is suspended until the buffer full state is released. The output buffer state can be checked by using 'ISTST.

Address : Variable name (Number of bytes)

FB2AH : PRNBUF (128)

Printer unit output buffer
 Put Pointer: PPUTPTR
 Get Pointer: PGETPTR

F4BDH : PPUTPTR (2)

 Pointer to store data in the printer unit output buffer (PRNBUF)
 This pointer is post-incremented so that it always points the head of the empty area.

- The buffer is assumed to be full when PPUTPTR+2=PGETPTR
- F4BFH : PGETPTR (2)

Pointer to fetch data from the printer unit output buffer (PRNBUF)
 This pointer is pre-incremented so that it always points the subsequent data item to be fetched.
 The printer unit output buffer is assumed to be empty when PPUTPTR=PGETPTR.

5 Power-down function

The power-down function saves the power consumption of the printer unit. If the printer is not used for a fixed time (default value is three minutes), the system sets the printer to power-down mode. The power-down time (the time from when the printer unit is used most recently to when the power-down mode is set) can be specified by CONFIG or by directly modifying the PRNPWTM contents (F025H) by using the application program. However, the printer unit is set to power-down

SUFTWARE Page 4 - 185

mode approximately five seconds after the printer unit power is first turned on, regardless of the specified power-down time. The user need not be concerned with power-down mode because the power-down mode is automatically released by the system when a data output request is issued for the printer. Note: Output one data line in a batch as far as possible so that the interval between data outputs does not exceed the power-down time. If the power-down time has expired midway through outputting one data line, the printer unit prints the data that it received before the power-down mode was set and then feeds a line. Consequently, subsequent data is printed starting at the head the new line. [Remarks] The power-down time is monitored by one-second interruption. When the power-down time has expired, value -1 is set in PRNPWFLG (F3BEH), indicating that the power-down time has expired. The printer unit power is actually set to power-down mode by PSTBIOS and key input routine (being in key input wait status). Address : Variable name (Number of bytes) FU25H : PRNPWTM (1) - Power-down time (initial value) - Ur.it: Second - Default: 80 (seco.ds) - A value from 0 to 255 can be specified. - If value 0 is specified, the power-down mode is not sit. F3BDH : PRNPWCNT (1) - Counter which indicates the time left up to power-down time expiration. First, the PRNPWTM value is copied into this counter, following which the counter value is decremented by each one-second interruption. When the value of this counter becomes 0, the power-down mode is set. F3BEH : PRNPWFLG (1) - Flag which indicates the printer unit power status 1: Normal mode 0: Power-down mode -1: The power-down time has expired (but the printer has not been set to power-down mode). Power off in continue mode 6 The printer unit is reset when the mainframe power is furned on, and all states and data that have been set for the printer such as character set adjustment to a specific country and external character definition are initialized. The system output buffer contents are also cleared. Therefore, note that the specifications made for the printer unit are not held effective. To initialize the printer at power on, use cartridge device HOOK. See Section 10.2 for HOOK.

(3) Control functions

Table 4.9 lists the I/O ports used by the printer unit. The printer unit can execute the following functions through these I/O ports.

1 Switching between offline and online

When SLIN is set to 1, the printer is set to offline mode (\*1).

When SLIN is set to 0, the printer is set to online mode.

\*1 (a) The printer unit cannot be accessed when the development unit is used.

(b) The printer unit is set to online mode by the following operations:
Turn the power on.
Execute DLL.

- Press the PF (paper feed) key.
- 2 Paper feed function

When the PF key is set to 1 for 50 ms or more in offline mode, paper feeding is started.

3 Buffer-full function

When the same amount of print data (including spaces) as the total number of printer columns (24 columns for character data or 18 bytes for graphic data) is input, the printer prints printer buffer data and feeds paper.

4 Emergency stop function

If one of the abnormalities listed below is detected while the printer is operating, the power that has been supplied to the print head and drive motor is cut, and the printer is set to power-down mode. In this case, reset the controller by turning the mainframe power off and then on to release the emergency stop status.

- (a) Motor lock
- (b) Malfunctioning of the timing detector (signal ungeneration)
- (c) Malfunctioning of the reset detector (signal ungeneration)

Port Address (R/W) : Port name (RAM data address)

P16H (R) : IOSTR bit O: (PBUSY): Printer busy signal 1: Busy O: Ready

P17H (W) : ICCTLR (FODBH) bit 6:(PRIE): Printer interrupt control 1: Disable 0: Enable

P19H (W) : IOCTLR (FODDH) bit 0:(PF):Paper feed signal bit 1:(SLIN):Printer select-in signal bit 3:(PINI)Cartridge reset and power-down mode release signal **P23H (R) : ITSR** bit 1:(IPBUSY): Printer busy signal Same as IOSTR pbusy P10H (W) : CHSOR bit 0 to 7 : Printer output data P11H (R) : CHSSR bit 0:(OBF): Out put buffer status 1:Busy 0:Ready (4) Control commands Printer unit control commands are listed below. 1 Print (LF: OAH) This command prints printer buffer data and feeds paper. 2 Printing enlarged characters (SO: OEH) This command prints subsequent print data in double-width enlarged characters. In double-width enlarged character mode, up to 12 characters can be printed in one line. The double-width enlarged character mode is released by control code DC4 (14H), LF (OAH), DC2 (12H), or DC3 (13H). 3 Cancelling input data (CAN: 18H) This command cancels all data input in the same line. Releasing double-width enlarged character mode (DC4: 14H) 4 This command releases the double-width enlarged character mode. Power down function 1 (DC2: 12H) 5

This command sets the controller to power-down mode. The power-down

mode is released by keeping **PINI** (bit 3 in address 19H of the I/O register) to 1 for five microseconds or more. After the power-down mode is released, the controller is restored to the status set before it was set to power-down mode. Even if the controller is set to power-down mode by DC2, controller oscillation does not stop.
#### 6 Power-down function 2 (DC3: 13H)

This command sets the controller to power-down mode. The power-down

mode is released by keeping PINI (bit 3 in address 19H of the I/O register) to 1 for one millisecond or more. After the power-down mode is released, the controller is restored to the status set before it was set to power-down mode. If the controller is set to power-down mode by DC3, controller oscillation stops.

[Remarks]

The amount of power consumption in normal mode and that in power-down mode are as follows. Normal mode: approx. 5 mA Power-down mode When set by DC2: approx. 1.5 mA When set by DC3: approx. 1 microA

Therefore, considerably larger amount of power can be saved when the controller is set to power-down mode by DC3 than by DC2. However, more time is required to return the power-down mode set by DC3 to normal mode.

Notes: Because the printer unit power is automatically controlled by the system, the user usually need not use control codes DC2 and DC3. Note the following when controlling the printer unit power by using control codes DC2 and DC3 in the user program:

(a) Be sure to set the system power-down time to infinity (PRNPWTM=0). (b) Release the power-down mode by using the application program. When accessing an I/O port to release the power-down mode, access it in the I/O access procedure explained in Chapter 7.

7 Escape alphabet control commands

Printing can also be controlled by escape alphabet control commands each consisting of ESC (18H) followed by an alphabetic code and binary data. Value n in the escape alphabet command indicates a one-byte binary data. The plus sign (+) is only given in this document as a separator for clarity, and need not be entered nor output actually. If an escape alphabet command is input midway in a line, the printer unit prints the data input up to the command entry and then terminates the previous sequence. Therefore, the specification made by the entered escape alphabet command is made effective for the subsequent lines.

(a) Setting line space (ESC+'A'+n)
This command sets the line space for each dot line. Value n must satisfy the following condition:
1 < n < 255</li>
If continuous printing can be performed, value 0 can be specified as n.

(b) Transferring bit image data (ESC+'K'+n1+n2+n3) This command processes subsequent data as bit image data. Values n1,n2 and n3 indicate the amount of bit image data to be transferred as follows:

```
n1: Number of horizontal bytes (1 < n < 18)
n2: The low-order byte for the number of vertical dot
    lines (0 < n < 255)
n3: The high-order byte for the number of vertical
    dot lines (0 < n < 1)</pre>
```

After all bit image data is transferred, the printer is automatically returned to text mode.

# Note:

In case of bit image data transmission, the printer may operate abnormally if all transmission data is not sent consecutively. The data of one dot line must be sent at transfer rate 18 bytes/15 ms or more.



DO to D7 indicate dot positions. To print a dot at a dot position, binary O must be set for the position. To print a space at a dot position, binary O must be set for the position.

Fig. 4.44 Relationship between data and print out

```
(c) Setting international character set (ESC+'R'+n)
This command sets the character set of the specified country.
```

SOFTWARE Page 4 - 190

Table 4.10 Print character sets and international character specification codes.

n	Country
0	U.S.A. France
2	Germany
3	U.K.
4	Denmark
5	Sweden
6	Italy
7	Spain
8	Japan

Value n greater than 8 is ignored, and the previous n value premains effective.

(e) Paper feed command (ESC+'B'+n) This command feeds the paper n dot-lines. Value n must satisfy the following condition: 1 < n < 255

(f) Transferring external character registration data (ESC+'&'+n1+n2) By entering this command followed by pattern data, an arbitrary pattern consisting of u 6 x 7 dot matrix can be registered in LSI. Up to eight characters can be registered in desired addresses of the audress area (20H to FFH). If a new character pattern is registered in an address in which another character pattern has already been registered, the old pattern is cleared and the new one is made effective. If more than eight character patterns are registered, all external character data that has been registered is cleared.

#### [Setting addresses]

The address that has oeen set matches the character code and, after registration, can be accessed like other fixed characters. If a fixed character has been defined in the address that has been set, the fixed character is made ineffective. Values n1 and n2 indicate the registration start and end addresses, respectively. Therefore, the number of characters to be registered is determined by values n1 and n2, and up to eight characters can be registered in addresses from n1 to n2.

[Pattern data configuration] Each pattern data to be registered consists of 6 x 7 dots that share six bytes. This pattern data is vertically divided into six portions each consisting of one byte, and transferred as 6-byte data as a total.



By repeating one character registration, up to eight characters can be registered. To register multiple characters in consecutive addresses (character codes) starting from registration start address n1 and ending with registration end address n2, register (n2-n1+1) characters of pattern data sequentially. In this case, the following condition must be satisfied: n1 < n2, n2 - n1 < 7 <Example> Assume that three 6 x 7 dot matrix full-dot patterns are to be registered in addresses E4H to E6H (the digits are represented in hexadecimal notation).

18,	26,	Ε4,	Ε6,	FF FF	FF FF	. FF FF
ESC	'&'	nl	n2			
				Pattern registered in E4 (6 bytes)	Pattern registered in E5 (6 bytes)	Pattern registered in E6 (6 bytes)

Total  $6 \times 3 = 18$  bytes

- 4.5.5 Outputting data to RS-232C interface
- (1) Overview

By specifying RS-232C as printer output in CONFIG or I/O byte, data can be output to an ordinary terminal or portable printer having an RS-232C interface. In EHT-10 and EHT-10/2, the Busy signal of the printer is checked by RS-232C DSR and, when DSR is active, data is output. The system supports BIOS functions LIST, LISTST, SCRNDUMP, and KANJI (kanji print). SCRNDUMP and KANJI cannot be used in some types of printers because they use printer control codes (see Item (2)). As for kanji character printing by KANJI, a non-kanji printer can also be used because the kanji fonts stored in EHT-10 or EHT-10/2 are output as bit image data.

(2) Connectible printers

Although the terminal printers (\*1) and portable printers produced by Seiko Epson can be basically connected to the system, note the following because some control odes and graphic characters are different:

1 Graphic characters

Fonts 80H to 9FH may differ according to the printer. For the printers for which fonts can be down-loaded, use the printers after down-loading the fonts as required. Fronts EOH to FFH cannot be printed in any printers. Use these fonts after down-loading as required.

2 SCRNDUMP (screen dump)

The screen dump function uses the control codes listed below. The printers that do not have these control codes cannot output screen dump data. Although any specific problems are not caused when the printer does not have control code ESC+'2', subsequent paper feed rate becomes 8/72 inches.

- (a) ESC+'K'+n1+n2 (single-density bit image mode)
- (b) ESC+'A'+n (paper feed rate is set to n/72)
- (c) ESC+'2' (paper feed rate is set to 1/6)

### 3 KANJI (kanji print)

The kanji print function uses control codes (a) and (b) above. The printers that do not have these control codes cannot output kanji characters.

4 Character set adjustment to a specific country

The following control code is used to adjust the printer character set to the mainframe: ESC+'R'+n (international character specification) Character set adjustment to a specific country can be suppressed when it is unrequired or the printer does not have the above control code. (See Section 4.5.3.)

\*1 All terminal printers having a parallel interface produced by Seiko Epson can be controlled by the RS-232C after connected with the serial interface board supplied as an optional device.

(3) Setting serial parameters

Like other I/O devices, the system uses the system default values for the serial parameters used for outputting data to the RS-232C interface of the printer. The system default values can be modified by using CONFIC. The system default values are listed below.

Baud rate: 4800 bps Data length: 8 bits Parity bit: None Stop bit: 2 bits 4.6 User BIOS

4.6.1 Overview

In EHT-10 and EHT-10/2, a user BIOS entry is allocated so that the user can extend BIOS and add new entries to it.

The user BIOS area can also be allocated as the area for storing the user BIOS processing section. The user BIOS area can also be used by machine-language routines (e.g., barcode input routine) used by multiple programs and hook extend processing (e.g., extensions of communication protocols and IC card protocols).

this Section explains the structure of the user BIOS and how to use the BIOS area.

# 4.6.2 User BIOS

(1) User BIOS expansion procedure

The user BIOS can be extended in the following procedure:

- 1 Cetermine the user BIOS extend processing area.
- 2 Load the user BIOS extend processing routine.
- 3 Update the user BIOS entry address and sets the new address as the extend processing start address.
- (2) User BIOS extend processing area

The user BIOS extend processing area can be reserved in one of the following three areas.

1 User BIOS area

If the user BIOS extend processing area is reserved in the user BIOS area, the user BIOS processing that has been set by one application program can also be used by other application programs. See Section 4.6.3 for the user BIOS area allocation procedure. Be sure to pay attention to the notes provided in Section 4.6.3 when using the BIOS area.

2 TPA (user area)

If the user BIOS expand processing area is reserved in the user BIOS area, the extended user BIOS processing is made local and cannot be used by other application programs. In this case, be sure to return the user BIOS entry address to its original address in the procedure explained in Item (6) when application program execution is terminated.

3 Part of the application ROM

Part of the application ROM can also be used as the BIOS extend processing area. In this case, the application program must take note of the bank where the extend processing routine resides when using the user BIOS. (3) Modifying entry address

EHT-10 and EHT-10/2 each has two BIOS entries RBIOS1 and RBIOS2. To extend the user BIOS, update the RBIOS1 jump table and sets the jump address in the extend processing start address area reserved in Item (2).

 WBOOT + 7EH
 C3H
 Jump instruction code

 WBOOT + 7FH
 - Jump address ---> Extend processing start address

 WBOOT + 80H
 - Jump address ---> Extend processing start address

The WBOOT address is contained in addresses 0001H and 0002H.

- (4) User BIOS call procedure
- 1 Calling by a load and execute program

To execute the user BIOS after loading it in TPA, obtain the WBOOT entry address from the contents of addresses 1 and 2 and call the (WBOOT + 7EH) address in the same way as that for other BIOSs. In this case, the user can freely use other BIOSs within the user BIOS. When control is returned to the user BIOS, the user stack is used.

2 Calling by a ROM-based program

Like the call procedure 1, obtain the entry address and call the user BIOS entry address by using BIOS CALLX (\*1). Set the bank information to be set by CALLX in bank O#O (OOH). (\*2) After the user BIOS is called by BIOS CALLX, other BIOSs cannot be used within the user BIOS (\*3). When control is passed to the user BIOS, the BIOS fack is used.

\*1 See Section 4.2 for CALLX.
\*2 Bank 0#0 is subbank 0 of bank 0, which is a standard RAM bank.
\*3 Use system jump table JSCALLX when calling the user BIOS which uses a BIOS from a ROM-based program. See Section 10.3 for JSCALLX.

(5) Starting user BIOS extend processing

Note the following when performing user BIOS extend processing:

- 1 The user BIOS is terminated by the RET instruction and returns control to the user program.
- 2 If the stack was modified during user BIOS execution, return to the stack to its original status when user BIOS execution is terminated.
- 3 If the user BIOS is called by BIOS CALLX, the user BIOS is in DI status when it received control. Set the user BIOS to EI status as required.
- (6) Terminating user BIOS extend processing

If user BIOS extend processing is no more required when the application program is terminated, be sure to disconnect the user BIOS extend processing routine from the user BIOS in the following procedure.

- 1 Return the user BIOS entry address to its original one as shown below.
  - 1. Jump instruction code
  - 2. RBIOS2 user BIOS entry address
- 2 When the user BIOS area is used If the user BIOS area is used as the user BIOS extend processing area, be sure to release the are in the procedure explained in Section 4.6.3.



(7) Initializing the user BIOS

Because the operating system initializes the BIOS entry at system reset or system initialization, the user BIOS entry is also initialized. After initialization, the user BIOS does returns only.

- 4.6.3 User BIOS area
- (1) User BIOS area position

The user BlOS area is allocated starting in the address immediately after the RAM disk area and ending in address DBFFH for EHT-10 or DDFFH for EHT-10/2.



Fig. 4.45 User BIOS area

SOFTWARE Page 4 - 197

(2) User BIOS area size

The user BIOS area size can be set in units of pages (256 bytes) by the user. (The default value is 0 page.) The maximum user BIOS area size is determined by the size of the standard RAM area in the RAM disk.

1 EHT-10

User BIOS area size + RAM disk standard RAM area size < 39.5 Kbytes

2 EHT-10/2

User BIOS area size + RAM disk standard RAM area size < 40 Kbytes

(3) Allocating user BlOS area

The user BIOS area can be allocated in one of the following three methods.

- 1 Allocate by using CONFIG at system initialization.
- 2 Select CONFIG on the system menu screen, and allocate it there.
- 3 Allocate by using the user program.
- (4) Allocating user BIOS area by using a user program

Figure 4.46 shows the procedure to allocate the user BIOS area by using a user program.





Step Explanation

- (1) Checking current user BIOS area size

   Check whether the current user BIOS area size satisfies the size of the area to be allocated.
   The current user BIOS area size can be determined from the USERBIOS (FOOCH) contents. (The unit is 256 bytes.)
- (2) Checking new user BIOS area size - Check whether the sum of the new user BIOS are size and the RAM disk standard RAM area size satisfies the following condition: EHT-10: < 39.5 Kbytes EHT-10/2: < 40 Kbytes</p>

SOFTWARE Page 4 - 199

If the sum exceeds the maximum value, the new BIOS area cannot be allocated. - The current RAM disk standard RAM area size can be determined from the SIZRAM (FOOBH) contents. (The unit is 1 Kbytes.) (3) Setting new user BIOS area size - Set the new user BIOS area size in USERBIOS (FOOCH) in units of 256 bytes. (4) SETRAMAD - Set the CP/M size and RAM system parameters. - SETRAMAD resides in the jump table allocated in the system bank, and its address is OO16H. - Call SETRAMAD by using BIOS CALLX. (5) CHGRAMD - Change the position of the RAM disk standard RAM area according to the modification of the user BIOS area size. - CHGRAMD resides in the jump table allocated in the system bank, and its address is 0019H. - Call CHGRAMD by using BIOS CALLX. - CHGRAMD entry parameters A register = 2 : Position modification B register = (CRAMD SIZE: F068H) : Extended RAM size eqister = (CSYZRAM: FOOBH) : RAM disk standard RAM area size (6) Replace RAM disk - Move information in a standard RAM as changing the size of user BIOS (7) BIOSJILD - Load the BIOS jump table in RAM. - BIOSJTLD resides in the jump table allocated in the system bank, and its address in OO13H. - Call BIOSJTLD by using BIOS CALLX. (8) Setting RBDOS1 and RBIOS1 - Set the RBDOS1 and RBIOS1 entry addresses. - BDSLAD (F005H) + 6 and BDSLAD (F005H) + 7 ---> (0006H and 0007H) - CI1LAD (F007H) + 3 and BI1LAD (F007H) + 4 ---> (0001H and 0002H) Notes: 1 RBIOS2 can be used to call the BIOS command for allocating the user BIOS area, and RBIOS1 cannot be used for this purpose. This is because

2 After the user BIOS area is allocated, do not call 0005H to execute BDOS until WBOOT is executed. This is because, after the user BIOS area size is modified, RBDOS1 is not loaded until WBOOT is executed. Use RBDOS2 BDOS when using BDOS.

the RBIOS1 area is relocated due to user BIOS area size modification

and the RBJOS1 entry address becomes undefined.

Remarks:

The system work areas related to user BIOS area size modification are listed below. Address : Variable name (Number of bytes) FOOBH : SIZRAM (1) RAM disk standard RAM area size EHT-10: 0 < SIZMAM < 39 Kbytes EHT-10/2:  $\overline{O}$  < SIZRAM < 40 Kbytes The unit is I Kbytes. FO68H : RAMD SIZE (1) RAM disk extended RAM area size 0 < RAMD SIZE < 6The unit is 32 Kbytes. FOOCH : USERBIOS (1) User BIOS area size EHT-10: 0 < USERBIOS < 158EHT-10/2:  $\overline{0}$  < USERBIOS < 160 The unit is 256 bytes. FCO5H : BDSLAD (2) **RBDOS1** loading address FOO7H : BI1LAD (2) **RBIOS1** loading address FO5CH : TOPRAM (2) User BIOS area head oddress (5) User BIOS area usage procedure Overview 1 The user BIOS area cannot be used simultaneously by multiple programs. To manage sharing the user BIOS area by multiple programs, the user BIOS area has a header in its end. (This header must be ado⊾d by the user during user BIOS creation.) The application program that is to use a program or data stored in the user BIOS area can determine whether the corresponding program or data really resides in the area. This header is also used to determine whether any other program has already used the user BIOS alea when loading a program or data in the area. The user BIOS area contents remain unchanged until the system is initialized or the user BIOS area size is modified. 2 Header structure (2) (3) (4) (5) (6) (7) (1)Si Release 'U' 'B' \*2 •1 address OOH routine name ze

\*1 = Over write flag

\*2 = Check sum

The header consists of 16 bytes. Because the bottom address of the user BIOS area is fixed, the header always resides in the following address area: EHT-10: DBFOH to DBFFH EHT-10/2: DDFOH to DDFFH (a) Header contents No. : Address of EHT-10, Address of EHT-10/2 : Item (Number of bytes) (1) : DBFOH, DDFOH : Header ID (2) - ID which identifies a header - This ID is fixed to 'UP' (ASCII). (2) : DBF2H, DDF2H : Routine name (8) - Name which indicates the name of the routine loaded in the user BIOS area - A desired name can be set in ASCII codes. (3) : DBFAH, DDFAH : Size (1) - Size of the routine loaded in the user BIOS area - This size can be set in units of 256 bytes in binary notation. (4) : DBFBH,DDFBH : Over write flag (1) - Flag which indicates whether a new routine can be overwritten on the routine that has already been loaded. OOH: Prohibits loading of a new routice. Not OOH: A new routine can be loaded after release processing is performed. (5) : DBFCH, DDFCH : Release address (2) - Address of the routine to be executed before loading a routine when the user BIOS area has already been used by another routine - The release processing can be only performed when the Overwrite flag is not OOH. - Release addr. must be an address within the user BIOS area. Release processing must terminate with a return instruction. (6) : DBFEH, DDFEH : Unused (1) - Fixed to OOH (7) : DBFFH, DDFFH : Check sum - The contents of each byte of the 15 bytes from the header head through the field immediately before the Check sum field (CBFOH to CBFEH) are subtracted from OOH, and the results are stored in this field. (b) Detailed on Overwrite flag The Overwrite flag must be set to OOH (prohibiting loading a new routine) when the routine must be resident after loaded in the user BIOS area. The routine for which the above flag has been set can be deleted from the user BIOS area only by the program that !baded the routine or after the system is initialized.

The Overwrite flag must be reset to a value other than OOH (a new routine can be loaded after release processing is performed) when the

SOFTWARE Page 4 - 202

system area can be restored to its original status by performing release processing and, after that, a new routine can be loaded without causing any problems.

(c) Details on release processing

When a routine stored in the user BIOS area modifies the system area contents, the system area contents must be saved in the user BIOS area before the start of modification. Release processing (processing executed by the routine indicated by Release addr.) is performed to restore the saved system area contents in the system area, thus restoring the system to the status set before the user BIOS routine was loaded. After that, the release processing routine initializes the contents of all header fields to 00H. The header contents must be cleared even when the system area contents need not be restored to the original. Release processing must be performed using the last 256 bytes of the user BIOS area. Release processing must terminate with a RETURN instruction.

3. Use procedure

The application program that loads a routine in the user BIOS area performs processing (see Figure 4.47) to check whether the user BIOS area can be used.



Fig. 4.47 User BIOS header check

# Step : Explanation

- (1) Checking header

   Sums the contents of all header bytes and checks whether the result becomes 00H.
   (Sum check)
   If the sum check result is valid, check whether the first two header bytes contain 'UB'.
- (2) Checking whether the routine has been loaded

   Check the routine-name field of the header area to determine whether the routine to be executed has already been loaded.
   This check step can be omitted.
- (3) Checking overwrite flag
   Check the Overwrite flag of the header area. If overwriting cannot be performed, the routine is not loaded.
- (4) Checking user BIOS area size
   Allocate the user BIOS area in the procedure explained in Section 4.1.3 (2).
- (5) Release processing
   Call the routine indicated by Release addr of the header area.
- (6) Loading user BIOS routine
   Load the new routine in the user BIOS area and create a new header area.

# CHAPTER 5 BDOS PROCESSING

5.1 Overview

The operating systems used in EHT-10 and EHT-10/2 are enhanced versions of CP/M Version 2.2. In EHT-10 and EHT-10/2, two 8DOS reside in different locations so that:

- (1) The upperlimit of the usable RAM memory space (TPA) where an ordinary CP/M application program as is can operate can be indicated. (RBDOS1)
- (2) The ROM-based program can call BDOS without taking note of bank switching. (RBDOS2)

In EHT-10 and EHT-10/2, extended devices have some restrictions on use. This chapter mainly explains the specifications unique to EHT-10. See APPENDIX 7 for BDOS functions.

### 5.2 BDOS Processing Flow

When BDOS is called by an application program in EHT-10 or EHT-10/2, the control is stored in BDOS allocated in RAM, following which the bank is switched and the actual BDOS stored in the OS ROM allocated in the system bank is called. When processing is terminated, control and return information are returned

to the bank first called and then returned to the application program. The BIOS being used by the BDOS stored in ROM directly calls the BIOS stored in the OS ROM.

The application program calls BDOS in the following procedure.

- (1) The load-and-execute program calls JMP RBDOS1 stored in address 0005H.
- (2) The ROM-based program calls JMP RBDOS2 stored in address FF90H.

The procedure to use BDOS called by a load-and-execute program is the same as the procedure to use BDOS called by a ROM-based program. Figure 5.1 shows the BDOS processing flow from when BDOS is called by an application program till when control is returned to the application program.



### 5.3 BDOS Error

There are following types of BDOS errors.

- (1) BAD SECTOR ERROR
- 1 Cause

Data cannot be input to or output from the disk normally.

2 Response

Pressing the 'Abort' or 'Return' key suspends processing. After processing is suspended, the system is warm-booted and application program execution is terminated. Pressing the 'Ignore' key or a key other than 'Return' ignores the defective sector and continues processing.

- (2) SELECT ERROR
- 1 Cause

An unexisting drive name was specified.

2 Response

Pressing any key including 'Press' sets drive A to the logged drive.

- (3) R/O ERROR
- 1 Cause

An attempt was made to write data on a read-only disk.

2 Response

Pressing any key including 'Press' warm-boots the system.

The states including the following are referred to as "the drive is not ready":

The ROM socket has not been mounted.
 The IC card has not been mounted.
 The IC card protocol is defective.
 The floppy disk drive power is off.
 The floppy disk drive cable has not been connected.
 The floppy disk drive diskette has not been set.

Table 5.1 shows the relationship between disk devices and BDOS errors.

Cause	RAM disk	ROM socket	IC card	FDD	BDOS processing
Check sum error	0	-	-	_	Bad Sector
Directory full	0	-	0	0	Return with A=FFH
Disk full	0	-	0	0	Return with A=FFH
Write processing	-	0	-	-	R/0
Write processing in write protect mode	0	-	0	0	R/0
The file is not in the directory	0	0	0	0	Return with A=FFH
The drive is not ready.	-	0	0	0	Select

Table 5.1 Disk device and BDOS error

0:Occurs -:Does not occur

As shown above, BDOS displays four types of errors. Because these errors are handled by BDOS independently, a message or input request unrelated to the application program may be output or the system may be warm-booted by a key pressed by the user after the error is displayed.

To avoid them, the application program must notify and recover the errors by itself. Only the following two procedures are explained here.

- (1) The procedure to receive a BDOS error as a return code
- (2) The procedure to update the BDOS error processing jump vector and perform discrete error processing
- (1) The procedure to receive a BDOS error as a return code
- 1 Modification procedure

Modify so that the application program can call OS ROM (system bank) address OOOAH. This enables the application program to store the BDOS error information in a register . (SETERR) Also, modify so that the application program can call OS ROM address OOODH. This enables BDOS to notify errors in an ordinary way. BIOS CALLX (WBOOT + 66H) is used by the application program to call a routine that resides in OS ROM. (See APPENDIX 9 SAMPLE26)

### 2 Return codes

The following return codes are returned after SETERR execution.

Error	Register A	Register H	Explanation
BAD SECTOR BAD SELECT R/O Disk R/O File	FFH FFH FFH FFH	01H 02H 03H 04H	CP/M Standard BDOS Error
		00H	Not an error

When the H register contains OOH, a return code corresponding to the CP/M return information has been set in the A register.

When BAD SECTOR ERROR occurred, more detailed error information are set in system area BIOSERROR (F417H).

Address : Variable Name (Number of bytes)

F417H : BIOSERROR (1)

Return code for BIOS disk read/write operation

- =00H: Normal termination
- =01H: Read Error
- =02H: Write Error
- =03H: Write Protect Error
- =04H: Time Over Error
- =05H: Seek Error
- =06H: Break Error
- =07H: Power Off Error
- =08H: Mount Error
- =FEH: Other Error
- 3 Notes

(a) Once SETERR is executed, BDOS only returns the error status to a register and does not perform error processing until RSTERR or WBOOT is executed.

(b) If the application program does not determine the error and recovers it by itself after SETERR execution, normal subsequent processing is not guaranteed.

- (2) Procedure to update the BDOS error processing jump vector
- 1 Modification procedure

The BDOS error processing jump vector is allocated at the head of BDOS in RAM. By updating the contents of this jump vector, the application program can perform its discrete error processing.

The jump vector configuration is shown below. (The RBDOS1 address can be obtained from the contents of addresses OOO6H and OOO7H.) See Appendix 9. Sample 26.

Address Data		Contents
RBDOS1+03H	DW PERERR	Permanent error processing address (BAD SECTOR)
RBDOS1+05H	DW SELERR	Select error processing address (BAD SELECT)
RBDOS1+07H	DW RODERR	R/O disk error processing address (R/O DISK)
RBDOS1+09H	DW ROFERR	R/O file error processing address (R/O FILE)

# 2 Notes

(a) Because the BDOS stack that belongs to the system is used, the BDOS stack must be switched to the application stack when returning control directly to the application program.

(b) Because the bank is in all-RAM (bank 0#0) state, be careful when updating this jump vector with a ROM-based program.

(c) Take the bank into consideration when updating the jump vector contents with a ROM-based program because the jump vector may be positioned behind the application ROM where the ROM-based program resides.

(d) Keep the following two items when returning control to the system (BDOS) after performing error processing by the application program.

- 1 Do not call BDOS during error processing.
- 2 Return control to the system after switching the current bank to the system bank.

# CHAPTER 6 DISK SYSTEM

# 6.1 Overview

In EHT-10 and EHT-10/2, the I/O devices related to the memory are assigned to disk drives so that they can be handled easier. The relationship between I/O devices and disk drives is as follows.

Drive A: RAM disk (internal and extended RAMs) B: ROM socket C: IC card D: External disk (floppy disk) E: External disk (floppy disk)

#### 6.2 Logical and Physical Drives

In EHT-10 and EHT-10/2, the correspondence between logical and physical drives can be modified arbitrarily. The default values are as given in Section 6.1.
 The correspondence table for logical and physical drives is allocated in system area DSKTBL. By updating the contents of this table, the correspondence can be modified. The contents of this table are held until the system is reset (BOOT). The DISKTBL contents are shown below. The correspondence is modified by changing the physical drive codes assigned to the logical drives in DISKTBL.

Example: To change drive B to a floppy disk drive, replace OlH, which is the contents of address DISKTBL + 1 (FOE5H), with O3H.

Address : Variable name (Number of bytes)

FOE4H : DISKTBL (5) Logical/physical drive correspondence table

Address	Initial value	Corresponding	logical	drive	name
F0E4H F0E5H F0E6H F0E7H F0E8H	00H 01H 02H 03H 04H	Logical drive Logical drive Logical drive Logical drive Logical drive	A: B: C: n: E:		

Physical drive codes

OOH: RAM disk O1H: ROM socket O2H: IC card O3H: Floppy disk O4H: Floppy disk

(2) Notes on changing correspondence

Note the following when changing the correspondence between logical and physical drives.

- 1 If a physical drive code between O5H and FFH is specified, the corresponding drive cannot be accessed.
- 2 Two or more identical physical drive codes must not 5e specified in a five-byte area of DISKTBL.
- 3 If the DISKTBL contents are modified, the corresponding DISKROV (FDE9H) vector contents that indicate whether the drive is used for R/O or R/W must also be modified. However, DISKROV modification is only made effective after warm BOOT is executed.
- 4 If the DISLTBL contents are modified, the contents of the area which indicates the ROM socket position in the table must also be modified.



Above bits correspond to logical drives A: to E:. The initial value for drive B: is 1 (R/O).

### 6.3 RAM Disk

6.3.1 Overview

The extended RAMs and part of the standard RAM are allocated as the RAM disk. Standard RAM and extended RAM are handled as a logically consecutive drive disk. Check sum is performed for the RAM disk during read or write processing to determine whether data was not destroyed.

Data can be written into and read from the RAM disk at higher speeds than for a floppy disk, and which largely increases the performance of EHT-10 and EHT-10/2. A maximum of 231 Kbytes for EHT-10 and 232 Kbytes for EHT-10/2 can be allocated as the RAM disk when all extended RAMs are mounted.

- 6.3.2 Drive name and capacity
- (1) Drive name

The drive name is A:.

- (2) Capacity
- 1 When no extended RAM is used: 0 to 40 Kbytes (\*1)
- 2 When one or more extended RAMs are used: Total extended RAM size + size of the disk area allocated in the main RAM
- 3 The RAM disk capacity can be modified by CONFIG on the system menu screen. However, only the RAM disk area size allocated in the main RAM can be modified by the user because all extended RAMs are automatically assumed as part of the RAM disk area by the system.
- 4 Fig 6.3 lists the specifications of the RAM disk for each extended RAM size.

\*1 Although a maximum of 40 Kbytes can be allocated in EHT-10/2, a maximum of 39 Kbytes can be allocated in EHT-10. This is because the system area size in EHT-10 is different from that in EHT-10/2.

Extended RAM	0 КВ	64 KB	128 KB	192 KB
Disk size (*1)	0 to 40KB	64 to 104KB	128to 168KB	192to 232KB
Sector/track	128 bytes			
Track/sector	64 sectors			
Track	0 to 4	0 to 12	0 to 20	0 to 28
Sector	0 to 63	0 to 63	0 to 63	0 to 63
Number of directories	16(512B) 32(1 KB) 64 (2 KB)			КВ)
Check sum area	512B 1 KB		2 1	KB

Table 6.1 RAM disk specifications for each extended RAM size

\*1 The maximum disk sizes listed in this table are those in EHT-10/2, and corresponding maximum disk sizes in EHT-10 are one Kbyte less than these values (that is, 39 Kbytes, 103 Kbytes, 167 Kbytes, and 231 Kbytes).

S.3.3 Furmiat

Figure 6.4 shows the RAM lisk format.



Fig. 6.1 RAM disk format

# (1) Directory area

1 Each directory consists of 32 bytes.

2 The number of directories depends on the extended RAM capacity. They are 16, 32, and 64.

# (2) Sum check area

1 The such check area size also depends on the extended RAM capacity. They area 512 bytes, 1 Kbytes, and 2 Kbytes. Check sum is performed for each 128-byte data, and the result is stored in a one-byte area. The check sum results are updated when data is written, and check sum is performed when data is read or the power is turned on.

2 Figure 6.1 shows the logical format of the RAM disk, and actual directory positions differ according to the extended RAM size. Figure 6.2 shows the positional relationship between extended RAM sizes and directory positions.



Fig. 6.2 Positional relationship between extended RAM sizes and directory positions

#### Notes:

1. Values #O to #6 indicate subbank numbers.

2. The arrow indicates from data top toward data bottom.

3. A value between 0 and 40 (or 39) Kbytes can be specified as the capacity of the RAM disk to be used as the subbank #0 disk.

### 6.3.4 Miscellaneous

- (1) Because the RAM disk area is usually allocated on the back of the application ROM (subbanks #0 to #6 of the bank), the bank must be switched in order to access the RAM disk data. However, the user need not be aware of it because switching is performed by the operating system.
- (2) The operating system checks subbanks, starting from subbank #0, for whether the extended RAM has been mounted. When a subbank for which the extended RAM has not been mounted is detected, the operating system does not check subsequent subbanks.
- (3) Setting RAM disk capacity with CONFIG.

Note the following when setting the standard RAM disk size.

- 1 If the new size is less than the current size, format all the RAM disk.
- 2 If the new size is greater than the current size, format only the enlarged area part. The contents of the old area part are remain unchanged.
- 3 If the new size is equal to the current size, the contents of the current area remain unchanged.
- 4 Although the RAM disk contents remain unchanged when the user BIOS size is enlarged, the RAM disk contents are destroyed when the user BIGE size is made smaller.
- 5 When ROM has been mounted in the ROM socket, the user BIOS size cannot be changed.

### 6.4 ROM Socket

6.4.1 Overview

The EHT-10/EHT-10/2 mainframe is equipped with one ROM socket in which ROM can be mounted. ROM is mapped in the EHT-10/EHT-10/2 memory space, and can be accessed by switching the bank. A CMOS masked ROM or CMOS EPROM with a capacity of 256 Kbits (e.g., 27C256), 512 Kbits, or 1 Mbits (e.g., HN62301) can be used as the ROM. (\*1) The programs and data stored in the ROM socket can be loaded in the RAM and executed and processed by the EHT-10/EHT-10/2 application programs.

The system supports the following ROM program formats as standard formats:

M format: Format used for load-and-execute programs in HX-40, PX-4 and PX-8.

P format: Format which enables PX-4, HX-40 ROM contents execution.

Note:

When using a 256-Kbit EPROM or masked ROM, the ROM contents switching jumper must be opened. When using a 512-Kbit or 1-Mbit EPROM or masked ROM, the switching jumper must be closed. Refer to EHT-10 Operation Manual.

- 6.4.2 Drive name and capacity
- (1) Drive name

The drive name is 8:.

(2) Capacity

Table 6.2 shows the ROM socket disk capacity.

Capacity	256 Kbits 32 KB	512 Kbits 64 KB	1 Mbits 128 KB
Track	0 to 3	0 to 7	0 to 15
Sector	0 to 63	0 to 63	0 to 63
Maximum number of directories	31	31	31

Table 6.2 ROM capacity

\*1 The values used in this table are those when the number of directories is 28 to 31. If the number of directories is less than 28, the number of logical tracks increases.

Example: When the ROM capacity is 256 Kbits and the number of directories is 1 to 3, logical tracks 0 to 4 and sectors 0 to 6 can be used. This is because the directory area is allocated in units of 128 bytes (one sector), and the area less than one block (1 Kbytes) is used as a data area. Consequently, the data area size increases by a maximum of seven sectors.

# 6.4.2 Format

(1) Overview

There are two types of application programs executed under control of the EHT-10/EHT-10/2 CP/M.

1 Load-and-execute application programs

These application programs are loaded in TPA and executed like an ordinary CP/M.

(a) Files can be loaded from the disk.(b) Files can be loaded through a communication line.

2 ROM-based application programs

In addition to the load-and-execute programs, the programs stored in the ROM socket ROM can be directly executed in ROM. For this purpose, the programs contained in the ROM mounted in the ROM socket have the following two formats.

(a) M format This ROM program format is the same as that of the PX-8 ROM socket and PX-4/HX-40 ROM cartridge, and used when the programs are loaded in TPA and executed.

(b) P format This ROM program format is the same as that of the HX-40,PX-4 ROM socket (ROM-based), and used when the programs are directly executed in the ROM.

Since format type of the ROM mounted in the ROM socket is automatically determined and format addresses are also automatically calculated by the operating system, the user need not be aware of the difference in the ROM program format.

# (2) M format



Fig. 6.7 Relationship between ROM addresses and data addresses (M format)

Nate:

For a 256-Kbit ROM, only subbank #O is assumed to be a consecutive ROM area. For a 512-Kbit ROM, subbanks #O and #1 are assumed to be a consecutive ROM area. For a 1-Mbit ROM, subbanks #O, #1, #2, and #3 are assumed to be a consecutive ROM area.

1 Header area (32 bytes)

The header area consists of 32 bytes, and used to store data such as the format, size, and the number of directories of the ROM. Figure 6.4 shows the header structure.



Fig 6-8 ROM header structure

2 Directory area

One directory consists of 32 bytes. Up to 31 directories can be registered. Refer to the corresponding CP/M-related manual for directory configuration. The header directory can be automatically created by using EHT-10/EHT-10/2 development tool WPROMFRM.

3 Data area

The top address of the data area changes according to the number of directories. The data area top address is calculated as follows:

Header top address + 20H x m

SOFTWARE Page 6 - 11

(Value m is (n + 1) rounded to a multiple of 4 where n is the number of directories.)

The data top address is fixed to track 0 and sector 8 regardless of the number of directories. If an attempt is made to read the data on an unexisting sector (m/4 to m/7), return code E5H is always returned.

(3) P format



Fig 6.5 Relationship between ROM addresses and data addresses (P format)

Note:

for a 256-Kbit ROM, only subbank #O is assumed to be a consecutive ROM
area. For a 512-Kbit ROM, subbanks #O and #1 are assumed to be a
consecutive ROM area. For a 1-Mbit ROM, subbanks #C, #1, #2, and #3
are assumed to be a consecutive ROM area.

1 Header area (32 bytes)

If value 50H instead of 37H is set in byte 2 of the M-format header, the header becomes a P-format header. Other values are the same as those of the P-format header. (See Figure 6.4.)

2 Directory area

Same as the M-format directory area

SOFTWARE Page 6 - 12

### 3 Data area

The top address of the data area is the same as that of the M-format data area. The following five-byte data must be added to the head of a ROM- based application program (see Item 3.4):

DDH,DEH,OOH,OOH ("DDH,DEH" is an ID which indicates that the program is a ROM-based program.)

The application program is initiated starting from the byte immediately after the above five bytes.

- (3) Miscellaneous
- 1 The above five-byte data must not be added to the head of a program other than a ROM-based program.
- 2 There are two types of P-format programs: load-and-execute programs without an ID area and ROM-based programs with an ID area. These two types of programs can simultaneously reside in the same ROM.
- 6.4.4 Executing ROM socket programs
- (1) Load-and-execute programs

The M- or P-format program having no ID at the head is loaded in TPA and executed starting from address 100H.

(2) ROM-based programs

If a P-format program having an ID area at the head is specified as an execution file, the BIOS ROM handler reads one sector and checks the ID area to determine whether the format of the specified file is P format (ROM-based program). If the program is determined as a ROM-based program, control is passed to the ROM address next to the (file top address + 5) address.

# 6.4.5 Miscellaneous

- (1) The relationship between logical addresses and ROM addresses in an M-format program is different from that in a P-format program. This should be taken note of when creating a ROM-based program.
- (2) When executing a ROM-based program by using a 512-Kbit or 1-Mbit ROM, sufficient address management must be performed and bank switching must be reflected in the program.
- (3) When creating a ROM-based program, determine the program execution start address by taking the directory area capacity beforehand because the data top address changes according to the number of directories.

6.5 IC Card

6.5.1 Overview

EHT-10/EHT-10/2 supports IC card I/F as a standard feature that conforms to ISO standards DP7816/1 and DP7816/2.(\*1) (However, part of the power specifications is different.) Commands and data are exchanged through serial communications between EHT-10/EHT-10/2 and the IC card. In this case, data is transferred according to an specified protocol. The IC cards that conform to the Toppan printing protocol used in EHT-10/EHT-10/2 OS version 2.0 are supported as disk drives. For other IC cards, hooks are prepared. The user can use these IC cards as disks by creating hook processing programs. This chapter explains IC cards supported only as disks.

(\*1) ISO (International Standardization Organization) standards stipulate items on the IC card, including the size, physical characteristics, and contact part.

6.5.2. Drive name and capacity

(1) Drive name

The drive name is C:.

(2) Capacity

The capacity of the disk is as follow.

Capacity		8KB(*1)
Track		0
Sector		0-63
Number of	directories	8(*2)

(\*1) The capacity and the number of directories can be changed by using IDSK function. Please see 6.5.4 (\*2) If the number of directories is 8, directory area is actually from 0 track 0 sector to 0 track 1 sector. However, 0S specifies from 0 track 0 sector to 0 track 7 sector as directory area. Therefore, the data area always starts from 0 track 8 sector. In such a case user cannot access from 0 track 2 sector to 0 track 7 sector.
# 6.5.3 Format

(1) Format

Fig. 6.6 shows the IC card format.

Directory area	256B	$ 32B \times 8 = 256B (0 track 0 sector to 1)$
Virtual directo area	ory 7688	User cannot access. (O track 2 sector to 7 sector.)
Data area	7KB	<pre> 128B x 56 record = 7KB (0 track 8 sector - 63 sector) * Number of directories and records can be changed by using IDSK function. ( Numbers should be even.)</pre>
-		

Fig. 6.5 IC card format (1)

By using ISET, IDSK function, user can make plural disks on one IC card. Fig. 6.7 shows the such a example.

Directory area 1 Data area 1	<ul> <li>Disk n torms one CP/id disk. In this case, File name (*1),</li> <li>System PSW, and User PIN should be set for each disk.</li> <li>And user cannot access 2 disks at the same time.</li> </ul>
Directory area 2 Data area 2	<ul> <li>(*1) For the IC card, one disk area is regarded as a file.</li> <li>k 2 Therefore, this "File name" means the name of a disk area.</li> </ul>
	(*2) Fig. 6.6 and 6.7 are both logical architecture. Actually the area for file control or check sum is included.

Fig. 6.7 IC card format (2)

# (2) Directory area

Fig. 6.8 shows the architecture of directory area.

Directory 1 (32B)	*Each directory has 32 bytes. Physically,
Directory 2 (32B)	sector(64B:*1) automatically.
Directory 3 (32B)	(*1) This 1 sector means one unit for
Directory 4 (32B)	from 1 sector of CP/M)
	(*2) The number of directories can be changed by IDSK function.
Directory 8 (32B)	

Fig. 6.8 Directory area of IC card

(3) Data area

Fig. 6.9 shows data area of the IC card.

Record 1 (128B) Record 2 (128B)	*Each record has 128 bytes. Physically the data for check sum is added each 1 sector(648:*1) automatically.
Record 3 (128B)	(*1) This 1 sector means one unit for data control of IC card. (Different from 1 sector of CP/M).
	(*2) The number of records can be changed by IDSK function.
Record 56 (123B)	

Fig. 6.9 Data area of IC card

- 6.5.4 Protocol
- (1) Overview

The specification of communications between EHT-10/EHT-10/2 and an IC card are as follow.

Transfer rate:	9600 bps
Data length:	8 bits
Stop bit:	1 bit
Parity bit:	Even parity
Protocol:	Toppan printing
Control line:	Unused

### (2) IC Card interface

There are following 2 mechods for accessing the IC card for the EHT-10/EHT-10/2.

A. Command Through mode B. Disk mode

A. Command through mode

IC card has a  $CP\bar{U}$  and EEPROM so that we can regard the IC card as a kind of computers. Command through mode is that the EHT-10/EHT-10/2 send the data to (and receive the data form ) the IC card directly .It is just like as to communicate with another computer via RS-232C port. In case of using command through mode, application program should send the command block to the IC card according to the IC card protocol. Also, application should control the answer from the IC card (response block). If you want to make an application by using the command through mode, therefore, you should know the IC card protocol which you want to use. For the EHT-10/EHT-10/2, the following 2 ways are supported for using command through mode. (a) BIOS ICCARD BIOS ICCARD can send/receive 1 byte in each time. (b) ICMD function If you use the Toppan printing IC card, you can send/receive 1 command block in each time by using ICMD function.

B. Disk mode Disk mode is the mode that to use the IC card just like as FDD or RAM disk. Drive "C:" is assigned for the IC card in the EHT-10/EHT-10/2. Therefore, OS should be able to control the IC card as other disk devices. For example, to control the data as each record or be able to format the disk. OS should be extended according to the IC card protocol. For the OS version 2.0, Toppan printing IC card (64Kbit type) can be used as drive C:. If you want to use other manufacturer's IC card, OS can be extended by using HOOKs. Toppan printing IC card has such functions as password, ID and so on. OS Vers. 2.0 supports 8 function interface commands for using these functions.

(3) Function Interface commands

EHT-10/EHT-10/2 IC card function interface commands have the following characteristics.

(a) For the purpose of optimum efficiency in memory use, the length of one block (the minimum unit used in file management) of a CP/M is 256 bytes; 512- or 1024-byte block units can also be selected as required.

(b) "n" number of files (with CP/M, independent disks can exist on the IC Card) can be created on a single IC Card. Each file can convert System PSW or User PIN and it is possible to prevent unintentional access from the external world.

(c) The size of one file (with CP/M, the size of a message of one disk) of the IC Card can be specified. In addition, the quantity of

directories in each disk can be specified according to the amount of data therein.

(d) The following 8 functions are provided: ISTS, ISET, IDSK, IFMT, EJCT, IOPN, ICLS, and ICMD. These support the exclusive functions of the IC Card.

- 1. ISTS: Checks the loading status of the Expansion Program, the installation status of the IC Card R/W Unit, and the insertion status of the IC Card.
- 2. ISET: Sets the file name and the contents of System PSW and User PIN.
- 3. IDSK: Sets the disk size and the quantity of directories.
- 4. EJCT: Eject the IC card from IC card reader/writer unit.
- 5. IFMT: Formats the IC Card based on the values specified by ISET and IDSK, enabling use of the IC Card as a disk.
- 6. IOPN: Starts the COMMAND THROUGH Mode and sets the IC Card to OPEN status.
- 7. ICLS: Terminates the COMMAND THROUGH Mode and sets the IC Card to CLOSED status.
- 8. ICMD: Transmits text to the IC Card, then stores and returns the response to that transmission.

The remainder of this section describes each of the above functions separately.

Function: Checks the status of the IC Card.

Format: ISTS = &HFFB7 N% = -1 CALL ISTS (N%)

Processing: Checks and reports OS version is 2.0 and the insertion status of the IC Card.

- N = -1: OS version is 1.0 ( IC card cannot be used.)
  - = 0: IC Card power is ON. Or during the auto power OFF.
  - = 1: IC Card power is OFF.
  - = 2: Excess current goes through IC card

Caution: This function can be used at any point during execution. It merely checks the status of the IC Card without accessing it.

Remarks: This function should be executed at the beginning of the BASIC program. After this function is executed, other 7 function can be used.

Function: Sets the file name and password onto RAM.

Format:	ISET	=	&HEAA4	
	FILE\$	=	"File name"(8 characters) +	CHR\$(N)
	PSWS	Ξ	"System PSW" (16 characters)	+ CHR\$(R)
	PINS	=	"User PIN" (16 characters) +	CHR\$(R)
	CALL		ISET (FILE\$, PSW\$, PIN\$)	
	LALL		ISEI (FILE3, PSW3, PIN3)	

Processing: Memorizes the file name, system password, user password(PIN), and the number of retries onto RAM. After execution of this function, the file (\*1)specified therein can be accessed.

Caution: In case a disk will be used, this function must be used to set the values for that disk in advance. The default value of FILES, PSWS and PINS are all OOH.

Remarks: Within the escape code CHR\$, "N" represents the file number. and "R" represents the quantity of retry errors permitted. The "N" value must be sequentially incremented from "OO". The "R" value (OOH to OFH) is referred during the execution of IFMT.

Errors: An I/O error will occur unless the length of FILE\$ is nine bytes, and that of PSU\$ and PIN\$ is 17 bytes each.

(\*1) For the IC card, one disk area is regarded as a file. Therefore, this "File name" means the name of a disk area. Function: Specifies the size of the IC Card as a disk.
Format: IDSK = &HEAA8
 N% = (Size of the data area)
 M% = (Quantity of directories)

L% = (!ength of one data block)

CALL IDSK (N%, M%, L%)

Processing: Sets the parameters required for using the IC Card as a disk. Default velue is N%=56, M%=8, L%=256.

Caution: In case a disk will be used, this function must be used to set the values for that disk in advance. After executing this function, the RESET command must always be executed.

Remarks: The "N%" value specifies the size of the data area in 128-byte units up to a maximum of 56 records using multiples of two. The "M%" value specifies the quantity of directories up to a maximum of 16 directories using multiplies of two. The "L%" value specifies the length of the data block to either 256, 512 or 1024 bytes. The default values are as follows: N% = 56, M% = 8, and L% = 256. If you want to make plural disks on an IC card, you should consider the size of the IC card and specify the each parameter.If you make only one disk on an IC card, you can specify the each parameter according to the following expression.

 $N_{\%} \times 2 + M_{\%} / 2 < 764$  (118)

Errors: No error checking is performed.

Function: Formats the IC Card as a disk. Format: IFMT = &HEAAC CALL IFMT

Processing: This functions creates a file on the IC Card based on the file name(\*1) and the PSW/PIN contents that were specified by the ISET function, allocates the area corresponding to the size specified by the IDSK function, then formats the file to enable its use as a disk.

Caution: This function can only be used in DISK Mode. It takes about 1 or 2 minutes to finish the formatting. If you push the reset switch or remove the IC card during the formatting, IC card may be broken. Don't push the reset switch or remove the IC card.

Error: Any error that occurs will be regarded as an I/O error and its occurrence will terminate any processing that is currently being executed.

(\*1) For the IC cand, one disk area is regarded as a file. Therefore, this "File name" means the name of a disk area. Function: Eject the IC card from IC card R/W unit.
Format: EJCT = &HEABO

CALL = EJCT

Processing: After closing the IC card, eject the IC card from IC card R/W unit. This function is for the Ic card R/W unit. Except the ejection, the process is the same as ICLS function.

Function: Supplies power to the IC Card and sets it to COMMAND THROUGH Mode.

Format: IOPN = &HEAB4 CALL IOPN

Processing: This function switches OFF the power supply to the JC Card, then switches it back ON again.

Caution: After this function is executed, the IC Card will be in COMMAND THROUGH Mode till ICLS or EJCT function is executed.

Remarks: When the power to the IC Card is switched ON, the Reset response from the IC Card is automatically sent to RAM.

Error: In case an error occurs because the IC Card has not been inserted, such error will be regarded as an I/O error.

Function: Discontinues the power supply to the IC Card and sets it to DISK Mode.

Format: ICLS = &HEAB8 CALL ICLS

Processing: This function switches OFF the power supply to the IC Card.

Caution: After this function is executed, the IC Card will be in DISK Mode.

Function: Transmits text to the IC Card and receives the response to that transmission.

Format: ICMD = &HEABC TEXT\$ = "Command data" ANS\$ = SPACE\$(N) CALL ICMD (TEXT\$, ANS\$)

Processing: This function transmits the command specified by TEXT\$ to the IC Card, then stores the response from the IC Card in ANS\$.

Caution: This function can only be executed in COMMAND THROUGH Mode.

Remarks: Since the Start code, Length, and Check will be automatically appended, only the Command, Reference, and Data are required for a CALL operation. (The same format is also applicable for the response data.) The "N" value specifies the size of the response data.

Error: In case an error occurs due to, for example, a NC response from the IC Card, it will regarded as an I/O error.

#### 6.5.5 Disk Mode Interface

### (1) Beginning Use

When the IC Card is accessed in DISK Mode, power is automatically supplied to the IC Card, collation of ID, PSW, PIN, etc. is performed according to your command specification, and the IC Card assumes a status wherein Read/Write operations can be performed. Before accessing the IC card as a disk, therefore, the disk status must be specified in advance at the application program, using the ISET and IDSK functions. And Before 1st disk access, you should check the IC card status by using ISTS function.

(2) READ/WRITE Operations

Read/Write operations can be executed just as the other disks. Since the block length of a data area is determined by the "L%" parameter of the IDSK function, the data boundaries can be in units of 256, 512 or 1024 bytes, depending on the "L" value. (For example, when L%=256, 500 bytes of data will occupy a data area of two blocks (512 bytes)).

(3) Terminating Use

Use of the IC Card in DISK Mode is terminated by the CLOSE command of BASIC. At this time, however, the power supply to the IC Card will remain ON. To terminate the power supply, execute the ICLS or EJCT function.

- (4) Miscellaneous
  - a. Power Switch OFF Status or Low Battery Status

In case the POWER SW is set OFF or the battery runs low during access of the IC Card, processing is performed to the end of the current command then the power supply is stopped. When the power is next switched back ON, it is possible to continue the previous processing. (There is no need for special consideration of the Power OFF status within application programs.)

b. Auto Power OFF Function

To reduce power dissipation when using the IC Card in DISK Mode, the power supply of the IC Card can automatically be switched OFF. In case the IC Card has not been accessed during a fixed time period, the Auto Power OFF function automatically stops the power supply to the IC Card. This function can be inhibited by the application program.

c. Use of Multiple Disks

A single IC Card can be used as multiple disks, but such use requires that the following points be strictly observed: (i) Before changing to another disk (that is, accessing a file of the IC Card which has a different file name, PSW, and PIN), the ICLS function must first be used to close the currently open file. (If you wish to perform disk access using BASIC, be sure to first execute the CLOSE command.)

(ii) The RESET command will initialize the disk status of BDOS. (iii) Use of the ISET and IDSK functions sets the status of the IC Card as a new disk.

(iv) Performing Steps (i) to (iii) completes the preparation for accessing a new disk, so the OPEN command can be used to start disk access.

- 6.5.6 Command Through Mode Interface
- (1) Beginning Use

The IOPN functions sets the COMMAND THROUGH Mode. At this time, power is supplied to the IC Card and a Reset response is received.

(2) Access

Access is performed using the ICMD function. In case a communication error, data error, No response error or other such errors occur, they will be regarded as I/O errors. Even if the status of the response data from the IC Card consists of an error code, however, it will not be regarded as an error and the corresponding response data will be returned.

(3) Terminating Use

The ICLS functions terminates the COMMAND THROUGH Mode and activates the DISK Mode. At this time, the power supply to the IC Card is also stopped.

- (4) Miscellaneous
  - a. Power Switch OFF Status or Low Battery Status In case the POWER SW is set OFF or the battery runs low during access of the IC Card, processing is performed to the end of the current command then the power supply is stopped. Special attention should be paid to the fact that, in this case, when the power is next switched back ON, the previous processing cannot be continued.
  - b. Disk Access When a disk access operation is executed in COMMAND THROUGH Mode, a Bad Select or Bad Sector error will occur.

6.5.7 Machine Language Interface

If you want to use the IC card interface functions ( such as ISET, IDSK) in the machine language application, do as the following.

- (1) Check whether the OS version is 2.0.
- (2) Call the ICBASCMD(bank:1#1,Address:6031H) using the BIOS CALLX. At that time, specified parameters should be set.
- (3) Return parameter is the following.
   CY=0: The function is terminated normally.
   =1: The function is terminated abnormally.

The followings are the parameters of each function. For detailed information about the each function, please see "6.5.4. Protocol"

(Note) The architecture of the parameter depends on the parameter type.



<Example>

ISTS DISBNK CALLX ICBASCMD	EQU EQU EQU	00H F41BH EQU EB69H 6031H	
	LD LD	A,11H (DISBNK),A	; SET THE BANK
	L D L D	HL,-1 (0200H),HL	; PARAMETER FOR ISTS
	LD LD LD CALL	HL.0200H A,ISTS IX,ICBASCMD CALLX	; PARAMETER FOR ISTS ; BIOS CALLX

#### 6.5.8 Miscellaneous

(1) Relationship with BIOS

If BIOS ICCARD and an IC card as a disk are used simultaneously, the protocol may not be satisfied because BIOS ICCARD also access the IC card. To solve this problem, the operating system prohibits simultaneous use of an IC card and BIOS ICCARD by returning an error code when an attempt is made to use either of them while the other is being used. (For example, an error is assumed if an attempt is made to open BIOS ICCARD while an IC card is open as a disk.)

(2) Relationship between an IC card and serial communication

An IC card is connected to EHT-10/EHT-10/2 through an IC card interface. Although three types of I/Fs (RS-232C, IC card and Cartridge SIO) are available for serial communications with EHT-10/EHT-10/2, none of these I/Fs can be used simultaneously with an IC card because only one port is used internally by way of switching. Supporting automatic serial port switching by one operating system enables an IC card to be used while, for example, a floppy disk (RS-232C) is being used.

(3) Registration of the card ID

When you start to access the IC card, ID check is required. Please do the following to register the card ID.

a. If the card ID is not set, register the ID by command through mode.

b. Set the ID (which is registered to the IC card) onto RAM by using the ICDIDPNT (F643H:\*1).

c. After a. and b., you can use the IC card. First, you should format the IC card by using ISET, IDSK and IFMT functions.

(\*1) The architecture of the ICDIDPNT is the following.

F643H: ICDIDPNT (4)

Represents the storage area for ID data during registration of the card ID. (Valid only in DISK Mode)



"Bank" specifies the bank indicated by the pointer. Default value is EFEAH and points RAM bank. With a RAM bank, the default data area is 22 bytes.

(4) Erasure of files

Basically, it is not possible to erase a file (one disk on CP/M) of the IC Card. (That is, once a file has been created by the IFMT command, that file cannot be erased.) Consequently, erasure of a file must be performed by directly erasing the file in COMMAND THROUGH Mode.

(5) BASIC DSKF command

Execution of the DSKF command will report the size of the remaining memory area of the specified disk. In case of the IC Card (Disk C), however, the size will be reported in the units specified by the "L%" parameter of the IDSK function.

(6) Reset during the accessing

Do not the reset the EHT-10/EHT-10/2 during accessing to the IC card (Especially during the writing). IC card may be broken.

(7) Default value related to the IC card

If you want to use the IC card without executing the ISET or IDSK function, you should set the status of IC card registration to the default value of the IC card. The default values are the following.

ID registration	Number of the registered ID data = $0$
Password	System password = All Os
	User password (PIN) = All Os
Disk	Number of directories = 8
	Block size = 256 bytes
	Data area = 56 records

If you register ID only, passwords are automatically registered by executing IFMT function or format of the CONFIG utility. All the values are initialized by reset.

(8) Formatting IC card

The IC card can be formatted by the DISK utility on the system menu screen.

(9) IC card power-off time

To save the power consumed by the IC card, the operating system stops supplying the power (power off) to the IC card when the IC card is not accessed for a fixed time (one minute as the default value). Therefore, the next access is started from reset response. When BIOS ICCARD is used, the power-off time for the IC card becomes infinite, and power is supplied until the IC card is clcsed.

```
6.6 Floppy Disks
6.6.1 Overview
An external 5.25-inch or 3.5-inch floppy disk drive can be connected to EHT-10/EHT-10/2 through an RS-232C interface. Read and write operations are performed for the connected external floppy disk drive in units of KB. The connectible floppy disks are as follows:

TF-15 (single and dual)
PF-10

6.6.2 Drive name and capacity

(1) Drive name
The drive names are D: and E:.
The relationship between drive names and floppy disks is as follows:
```

TF-15 (single drive)D:TF-15 (dual drives)D: and E:PF-10D:

(2) Capacity

Capacity per drive: 320 Kbytes Number of tracks per drive: 80 Number of sectors per track: 16 Storage capacity per sector: 256 bytes Number of directories: 64 User capacity: 278 Kbytes

6.6.3 Format

Figure 6.10 shows the disk format on the media. Although tracks 0 to 3 are used by the system, the user can perform read and write operations for these tracks by using BIOS. Sectors 1 to 16 of track 4 are used as the directory area.

Track



Fig 6.10 Disk format

## 6.6.4 Protocol

(1) Overview

The specifications of communications made between EHT-10/EHT-10/2 and a disk drive are as follows:

Transfer rate: 38400 bps Data length: 8 bits Stop bit: 1 bit Parity bit: none Protocol: Used Control line: Unused

(2) EPSP (EPSON Serial Communication Protocol)

Communications is made between a disk drive and EHT-10/EHT-10/2 according to the EPSP (EPSON Serial Communication Protocol). The general EPSP format is shown below.

FMT	FMT: Head block format
DID	EHT-10/EHT-10/2
SID	OIH: Indicates data sent from the FDD
FNC	DID: ID of the destination device SID: ID of the source device
SIZ	Device IDs are as follows:
Text data	EHT-10/EHT-10/2: 23H FDD (D: or E:): 31H

Therefore, the following is assumed: When data is sent from EHT-10/EHT-10/2 to the FDD DID=31H and SID=23H

When data is sent from the FDD to EHT-10/EHT-10/2 DID=23H and SID=31H

FNC: Command issued for the FDD SIZ: Text data length, which is the actual text data length minus 1

The data sent from the floppy disk drive unit (FDD) has a return code at the end of the text data.

Tables 6.3 and 6.4 lists EPSP functions and return codes, respectively.

Item No.	Command	FNC	Function
1	RESET	0DH	Resets the disk drive.
2	READ	77H	Directly reads data from the disk.
3	WRITE	78H	Directly writes data to the disk.
4	WRITEHST	79H	Forcibly writes data to the disk.
5	COPY	7AH	Copies the disk contents to a volume.
6	FORMAT	7CH	Formats the disk.

Table 6.3 List of EPSP functions codes

Return code		Contents
00H		Normal termination
FAH FBH FCH FDH FFH	BDOS error	Read error Write error Drive select error Write protection Other error

Table 6.4 List of EPSP return codes

### <Function>

The RESET command resets the disk drive.

<send< th=""><th>data&gt;</th><th></th><th colspan="4"><receive data=""></receive></th></send<>	data>		<receive data=""></receive>			
+00H	ООН	(FMT)	+00H	01H	(FMT)	
01H	31H	(010)	01H	23H	(010)	
02H	23H	(510)	02H	31H	(SID)	
03H	ИООН	(FNC)	03н	ODH	(FNC)	
04H	ООН	(SIZ)	04H	ООН	(SIZ)	
05H	ООН		05H	Return code	1	

<Explanation>

- Upon receiving the RESET command, the FDD initializes itself and enters receive wait state.

- Return code ODH is sent as response to the mainframe.



<Explanation>

- Upon receiving the READ command, the FDD transfers data (128 bytes) corresponding to the specified logical track and sector numbers and a return code.

#### <Function>

The WRITE command writes data onto the specified sector of the disk.



## <Explanation>

 Upon receiving the WRITE command, the FDD writes the specified data (128 bytes) onto the disk area indicated by the specified logical track and sector numbers.

Write type = OOH: Standard writing (The FDD performs blocking.)
 O1H: Forcible writing (The FDD does not perform blocking and immediately writes data to the disk.)
 O2H: Sequential file writing (The FDD performs blocking and writes data at a high speed.)



# <Explanation>

- Upon receiving the WRITEHST command, the FDD forcibly writes the contents of the 1-Kbyte host buffer containing the data sent by the WRITE command to the disk.

#### <Function>

The COPY command copies the disk contents into a volume.



Number of the track being copied = 0 to 39 (\*1) Number of the track being copied (low-order) (\*2) Number of the track being copied (high-order)

<E.:planation>

- Upon receiving the COPY command, the FDD copies the specified disk contents to another disk in the same FDD.

- This command cannot be used by an FDD having only a single drive.
- Refer to Appendix 9. sample 34

<Function>

The FORMAT command formats the disk.



<Explanation>

- Upon receiving the FORMAT command, the FDD formats two tracks and returns the corresponding logical track number and return code to the system. The FDD repeats this operation.

- When formatting is completed, the logical track number of the receive data becomes FFFFH.

(3) Using system utilities

Although the operating systems for EHT-10/EHT-10/2 support read/write processing for the FDD by using BIOS, the application program can directly exchange data with the FDD. The application program uses the following system utilities when directly exchanging data with the FDD:

EPSPSND (EPSP data send utility) EPSPRCV (EPSP data receive utility)

See Section 10.3 for how to use these system utilities.

6.6.5 Miscellaneous

(1) Forcible writing

When floppy disk TF-15 or PF-10 is used, blocking and deblocking of data are performed to upgrade the data read/write efficiency. Therefore, when data is written to this floppy disk by using EHT-10/EHT-10/2, data may not yet be actually written to the disk even when the processing by EHT-10/EHT-101I is completed. To solve this problem, the operating system of EHT-10/EHT-10/2 sends a forcible write command to the disk drive unit at warm BCOT or power off execution to prevent write data from being lost.

(2) Relationship between a disk and serial communications

A floppy disk is connected to EHT-10/EHT-10/2 through an RS-232C Interface. Although three types of I/Fs (RS-232C, IC card, and cartridge SIO) are available for serial communications with EHT-10/EHT-10/2, none of these I/Fs can be used simultaneously with a floppy disk because only one port is used internally by way of switching. Supporting automatic serial port switching by the operating system enables a floppy disk to be used while, for example, an IC card (BIOS ICCARD) is being used.

6.7 Details on Disk - Related Work Areas Address : Variable name (Number of bytes) FOOBH : SIZRAM (1) RAM disk standard RAM area size (unit: Kbyte) FO5FH : QT ROM CP1 (1)ROM disk capacity 0: None 20H: 32 Kbytes 40H: 64 Kbytes 80H: 128 Kbytes FOGOH : OT RAM IN (1) RAM disk capacity (unit: Kbyte) F061H : DR ROM CP1 (1) Number of RAM disk directories FO62H : DR RAM IN (1) RAM disk directory area size (unit: 128 bytes) FOG3H : AD ROM CP1 (2) ROM disk start address P format: 6000H M format: COOOH FO65H : AD RAM IN (2) Bottom address of the RAM disk standard RAM area F067H : CS RAM IN (1) RAM disk check sum area size (unit: 128 bytes) FO68H : RAMD SIZE Size of the extended 'RAM used as the RAM disk (unit: 32Kbytes) FOE4H : DISKTBL (5) Logical/physical drive set table (See Section 6.2.) FOE9H : DISKROV (2) Disk R/O vector (See Section 6.2.) FOEBH : FTSTAB (10) Initial disk select jump vector Jump vector when the first disk select is specified by BIOS SELDSK FOF5H : READTAB (10) Jump vector for disk reading This vector is used for jumping to each read process during BIOS READ execution. FOFFH : WRTTAB (10) Jump vector for disk reading This vector is used for jumping to each write process during BIOS WRITE execution.

F109H : DPBASE Refer to the CP/M-related manual for the disk parameter header configuration. DPEO (16): RAM disk DPH DPE1 (16): ROM disk DPH DPE2 (16): IC card DPH DPE3 (16): Floppy disk drive 1 DPH DPE4 (16): Floppy disk drive 2 DPH F159H : DPBO (15) RAM disk DPB (Disk Parameter Block) D168H : DPB1 (15)ROM disk DPB D177H : DPB2 (15) IC card DPB D186H : DPB3 (15)Disk drive DPB F4D7H : DIRBUF (128) Directory access buffer F557H : ALVO (29) RAM disk allocation area F574H : CSVO (0)RAM disk check sum area (Defined label name only) F574H : ALV1 (16) ROM disk allocation area F584H : CSV1 (0)ROM disk check sum area (Defined label name only) F584H : ALV2 (8) IC card allocation area F58CH : CSV2 (8)IC card check sum area F59CH : ALV3 (18) Floppy Disk drive 1 allocation area F5AEH : CSV3 (16)Floppy Disk drive 1 check sum area F5BEH : ALV4 (18)Floppy Disk drive 2 allocation area F5DOH : CSV4 (16)Floppy Disk drive 2 check sum area F82DH : SYSFCB (36)System FCB area

F851H : SYSDMA (128) System DMA buffer

# CHAPTER 7 I/O INTERFACE OVERVIEW

## 7.1 Overview

This chapter explains the EHT-10/EHT-10/2 I/O interface.

(1) I/O registers

For EHT-10/EHT-10/2, all I/O operations are controlled through I/O registers in the gate array. The I/O registers in the gate array can be directly controlled by I/O commands from the main CPU.

EHT-10/EHT-10/2 uses the 39 I/O ports listed in the table below among 256 I/O ports.

I/O port	Explanation
POOH to POFH	Used by the system.
P10H to P13H	Used by the system. However, this address space is used to extend a cartridge.
P14H to P2CH	Used by the system.
P27H tc PrFH	Unused

# Table 7.1 I/O Ports

Refer to "PART 3 HARDWARE" for details on I/O structure.

(2) Accessing I/O registers

The EHT-10/EHT-10/2 OS stores the current output contents of an I/O register as the I/O register exit in the system area. An application program can modify I/O register contents as follows:



Fig 7.1 Accessing an I/O Register

Table 7.2 shows the I/O registers that require the operations shown in Figure 7-2. I/O address : Register name (RAM data address :RAM variable name) POOH : CTLR1 (FOD6H : RZCTLR1) PO2H : CTLR2 (FOD8H : RZCTLR2) PO4H : IER (F42EH : RZIER) Interrupts are suppressed while this register is being rewritten. **PO5H : BANKR (F42CH : RZBANKR)** Interrupts are suppressed while this register is being rewritten. PO8H : VADR (F254H : LSCRVRAM +1) PO9H : YOFF (F266H : LVRAMOF) Only bits 6 to 0 are stored. (FO74H : DSPFLAG) Only bit 7 is stored. P15H : ARTMR (FOD9H : RZARTMR) **P16H : ARTCR (FODAH : RZARTCR)** P17H : ICCTLR (FODBH : RZICCTLR) Interrupts are suppressed while this register is being rewritten. P18H : SWR (FODCH : RZSWP) **P19H : IOCTLR (FODDH : RZIOCTLR)** P22H : SBKR (F42DH : RZSBENKR) Interrupts are suppressed while this register is being rewritten. P23H : CTLR3 (FODEH : RZCTLR3) Interrupts are suppressed while this register is being rewritten. Table 7.2 I/O Registers of Which System Area Must be Rewritten

### 7.2 Serial Interface

### 7.2.1 Overview

EHT-10/EHT-10/2 has RS-232C, cartridge SIO, and IC card as the serial communication devices. However, EHT-10/EHT-10/2 has only one serial I/O port and the output destination is switched by the internal serial switch. Because of this, these serial devices cannot be used at the same time as a rule. The user is not required to switch the serial switch since the OS automatically switches the serial mode.

# 7.2.2 Setting a serial device

Table 7.3 shows the I/O registers used by the serial I/F. Refer to "PART 3 HARDWARE" for details on each I/O register.

(1) Switching the serial port

The three serial ports for RS-232C, IC card, and cartridge S10 are switched by SWR (P18H).

R/W	I/O Addr <b>e</b> ss	Register name	bit7	bit6	bit5	bit4	bit3	bit2	bit1	bitð	Notes
W	P18H	SWR					SSW1	SSWO			(*1)

(\*1) Bits other than bits 3 and 2 are used for others.

Serial mode	SSW1	SSWO	RXD/TXD	Example of corresponding device
0	0	0	Cartridge SIO	
1	0	1	IC card	IC card
2	1	0	RS-232C	FDD, printer, or coupler

Table 7.3 Serial Port Switching

(Note) Both SSW1 and SSW0 cannot be turned to 1 at the same time.

(2) Serial parameters and data control line

Table 7.4 shows the I/O registers related to setting the serial parameters, reading/writing data, and setting/reading the control line.

R/W	I/O Address	Register name	bit7	bit6	bit5	bit4	bit3	bit2	bit1	bit0	Notes
	P14H	ARTDIR		7 or 8 bits data							
R	P15H	ARTSR	RDSR		FE	OE	PE	TX Emp	RX RDY	TX RDY	
	P16H	IOSTR			+ RCTS	+ RCD	RXD			(*1)	
	POOH	CTLR1	BRG3	BRG2	BRG1	BRGD					(*2)
	P14H ARTDOR 7 or 8 bits data										
W	P15H	ARTMR	STOP		EVEN	PEN			DATA Lng.		
	P16H	ARTCR			+ RRTS	ER	SBRK	RXE	RDTR	TXE	

(\*1) Bits other than bits 5 to 3 are used for others.
(\*2) Bits other than bits 3 to 0 are used for others.
(Note) A bit indicated by an asterisk (\*) is effective only when the serial switch is set to RS-232C.

Table 7.4 I/O Registers Related to Control Line

# 7.2.3 Serial ports supported by OS

(I) Overview

The EHT-10/EHT-10/2 OS supports the following methods to effectively use the three serial ports:

1 By the user using BIOS RSIOX

- 2 By the user using BIOS ICCARD
- 3 As a system J/O device (LST:, PUN:, RDR:, or CON:)
- 4 As a terminal floppy disk (TF)

5 As an IC card disk

OS classifies 1, 2, and 3 as one mode among the above five and automatically switches the serial switch so that each of the following three modes can operate independently from each other:

```
    Used by the user (BIOS RSIOX, TCAM, ICCARD, and I/O device)
    As TF
    As IC card disk
```

Table 7.5 and Figure 7.2 show the relationship between the modes and devices.

Module to be newly used		User		System			
Currer	nt use mode	RSIOX TCAM	ICCARD	I/O device	FDD	IC card (disk)	
User	RSIOX, TCAM	-	x	x	o(*1)	o(*1)	
	ICCARD	x	-	x	o(*1)	x	
	I/O device	x	x	-	o(*2)	o(*1)	
System	ystem FDD		٥	o(*2)	-	0	
	IC card (disk)		x	O	0	-	

Table 7.5 Relationship between Devices and Modes

(\*1) Receive data may be ignored if FDD or IC card is used while the user is using the serial port. To prevent this, use a protocol that does not accept receive data while FDD or IC card is being used.

(\*2) This is meaningless because the same device (RS-232C I/F' is actually used.



Fig. 7.2 Concept of Using Serial Port
Serial port switch processing is performed as follows: The current serial use mode is stored in SRMODE (F241H). Processing is executed without changing the current serial use mode when the use mode is equal to the current one. Processing is executed after the new serial use mode is set when the use mode is not equal to the current one. The parameters used to change the serial use mode are stored in the 15-byte area starting from SRTABL (F196H).

- 7.2.4 Serial communication by the user
- (1) Overview

The serial interface is generally controlled by BIOS RSIOX and TCAM and not by the user directly. However, the user must directly control the serial interface to extend serial communication or the IC card protocol explained in Chapter 11. This section explains the procedure used by the user to directly control the serial I/F.

(2) Open processing

The serial I/F is opened in the following procedure:

Open chcck Whether the serial I/F is being used is checked, RSODEV (F623H) =00H: RS-232C =01H: IC card (BIOS ICCARD) =03H: Cartridge SIO =FFH: The serial I/F is not in open status.

An error occurs if the serial I/F is in open status (see note 1).

The code of the device to be used is set if the serial I/F is not in open status.

(Note 1) The serial I/F has been opened for RS-232C if RS-232C is being used as the I/O device (LST:, PUN:, RDR:, or CON:). In this case, the serial I/F can be closed by using BIOS RSIOX.

Setting the serial parameters A 15-byte area starting from SRSTABL (F196H) contains three 5-byte packets for IC card (used as a disk), user, and FDD in this order. The serial parameters are set in one of these packets. The serial parameters are set in the IC card packet for an IC card (used as a disk) or in the user packet in any other cases (see the following).

SOFTWARE 7 - 6

Address : Variable name (Byte length) F241H : SRMODE (1) - This indicates the current serial use mode as follows: =FFH: Unused =OOH: IC card (as a disk) is being used. =01H: The user is using the serial I/F. =02H: FDD is being used. - The initial value is FFH and this parameter is initialized when warm boot is executed. F196H : SRTABL (15) - This table is used to switch the serial use mode. Byte length Name Contents SYSCILR1 IC card (disk) CTLR1 1 IC card (disk) ARTMR SYSARTMR 1 IC card (disk) SWR SYSSWR 1 IC card (disk) SWR IC card (disk) ARTCR unused User CTLR1 User ARTMR User SWR User AKTCR unused FDD CTLR1 FDD ARTMR FDD SWR FDD ARTCP SYSARTCR SYSSOUT 1 SYSSOUT 1 

 STSSOUT
 1

 RS2CTLR1
 1

 RS2ARTMR
 1

 RS2SWR
 1

 RS2ARTCK
 1

 KS2SOUT
 1

 HSCTLR1
 1

 HCARTMR
 1

 HSSWR
 1

 HSSWR 1 HSARTCR HSSOUT FDD ARTCR 1 unused Ł - The table is separated in the units of 5 bytes and data corresponds to I/O registers CILR1, ARTMR, SWR, ARTCR, or IOCTLR. - The bits not related to serial I/F must be left as O - See "Part 2 Hardware chapter 2 I/O register" for details. 3 Switching the serial switch According to the parameters explained in 2, the serial switch is changed and the serial parameters are set for the serial controller. The serial switch is changed by using FSELSER shown in APPENDIX 9 SAMPLE20. In this case, the entry parameter is set in register C as follows: Register C=OOH: IC card as a disk =01H: Other than above 4 Allowing receive interrupts Receive interrupts must be enabled during open processing if receive interrupts are required for serial communication (including IC card) extension processing (see "CHAPTER 8 ART INTERRUPTS). (3) Sending and receiving data and controlling the control line

The serial mode must be switched by SELSER before send/receive operation every time data is sent or received (see note 1). The entry parameters explained in Section (2) for FSELSER is used for SELSER.

The serial mode must also be switched to control the control line or to recover from an error.

(Note 1) When data is received or sent during extension processing or when FDD and IC card are used at the same time, the current mode is not necessarily be the one set at open processing because OS automatically changes the serial switch.

- (4) Close processing
- 1 FFH is set in RSODEV (F623H) to indicate that there is not a device in open status.
- 2 FFH is set in SRMODE (F241H) to indicate that a serial device is not being used.
- 3 Receive interrupts must be disabled if they were enabled.

7.3 Cartridge Interface

7.3.1 Overview

The EHT-10/EHT-10/2 functions can be extended by connecting an option cartridge to the cartridge interface. The printer unit is supported in standard as a cartridge option. However, the user can create a cartridge device by using the universal cartridge. The cartridge I/F has HS, IO, DB, and OT modes. One of these mode can be selected according to the characteristics of the cartridge device. See the following sections for the cartridge interface:

- 1 PART 3 HARDWARE Section 4.1 Cartridge Interface
- 2 Section 11.4 Extending a Cartridge Device
- 7.3.2 Modes and how to set a mode
- (1) Modes

EHT-10/EHT-10/2 cartridge I/F has four modes (HS, IO, DB, and OT). One of these modes can be selected according to the characteristics of the cartridge device.

- Hand shake (HS) mode This is CPU-to-CPU interface mode used for the device that has CPU at the option side. Data is sent or received through the input or output buffer. Data transmission is controlled according to the flags (IBF and OBF).
- 2 Input output port (IO) mode The interface is in the form of 4-bit input and output ports.
- 3 Data bus (DB) mode In this mode, an option looks as a general I/O device in the view of the main frame. The cartridge interface simply connects the data bus of the main frame to the cartridge data bus.
- 4 Output port (OT) mode The interface is in the form of 8-bit output port.

See "PART 3 HARDWARE Section 4.11 Cartridge Interface" for details on each mode.

(2) Setting mode

The cartridge interface mode can be selected by using the cartridge switches CSW1 and CSW0 (bits 1 and 0) in SWR (P18H). CSW1 and CSW0 are set by the initialization routine in the gate array as follows: CSW1=0 and CSW0=0 (HS mode)

CSW1	CSWO	Mode
0	0	HS mode
0	1	10 mode
1	0	DB mode
1	1	OT mode

(Note 1) OS automatically sets a mode so that a user program is not required to set it. See the next section for the mode setting procedure done by OS.

(Note 2) HS mode is set after the power on of the EHT-10/EHT-10/2 main frame. However, OS automatically resets the specified mode at power on processing so that the user is not required to consider the mode.

- 7.3.3 Determining the mode
- (1) Overview

Since the EHT-10/EHT-10/2 OS automatically sets the cartridge mode, user programs need not set the mode. Mode setting is performed in the following timing:

- 1 EHT-10/EHT-10/2 main frame power on
- 2 Reset processing
- **J** System initialization

The device number of cartridge device and the CSEL signal must be set for a cartridge option created by the user because the mode is determined from the device number and CSEL signal (see "PART 2 HARDWARE Section 4.1 Cartridge Interface" for details).

- (2) CSEL signal and device number
- CSEL signal The CSEL signal is the signal line used to check whether the cartridge option is in HS mode. The CSEL signal is in bit 6 of I/O register P16H. CSEL=1: HS mode =0: Other mode (IO, DB, or OT mode)
- 2 Device number The device number indicates the type of cartridge option. The device number is in the higher 4 bits of I/O register P13H read in DB mode. A device number is indicated by a code and is set according to the operation mode of the cartridge option (see Table 7.6).
- 3 Device management by OS OS manages the cartridge device according to the device code consisting of the cartridge mode and device number and stored in CRGDEV (F42FH).



Device number	CSEL=1		CSEL=0			
ОН		No option				
1 H 2 H 3 H	Printer unit	HS mode		For DB mode extension		
4H 5H 6H 1 7H	(M160 Printer)	For HS mode extension		10 moje		
8H 9H AH BH			,	DB mode		
CH DH EH FH				OT mode		

## Table 7.6 CSEL and Device Number

(3) Mode determination procedure

Figure 7.3 shows the mode determination procedure. See "Section 11.4 Extending Cartridge Device" for CRGHOOK in the figure.



Fig 7.3 Cartridge Mode Determination

7.3.4 1/0 registers used for cartridge I/F

Table 7.7 shows the I/O registers used to control the cartridge devices. The contents of I/O registers P10H to P13H differ depending on the cartridge mode. See "PART 2 HARDWARE Section 4.1 Cartridge Interface" for details.

See the beginning of this chapter for accessing I/O registers. [Note] The following I/O registers among the ones related to cartridge I/F must not be accessed (modified) by the user: 1 DCTG (P17H bit 4) This bit is controlled at debugger operation and is usually set to 0. 2 CSWO and CSW1 (P18H bits 0 and 1) They are the cartridge mode switch and are usually set in device management processing executed by OS. 3 PINTDS (P23H bit 3) This masks IC card and cartridge interrupts. Generally, interrupt allowed status is set. IC card and cartridge interrupts are masked independently in ICCTLR (P17H). Table 7.7 I/O Registers Related to Cartridge Devices Port Address (R/W) : Port name (RAM data address) P10H to P13H (R) : General input register The contents differ depending on the cartridoe mode (see "PART 3 HARDWARE" for details). P10H to P13H (W) : General output register The contents differ depending on the cartridoe mode (see "PART 3 HARDWARE" for details). P16H(R) : IOSTRbit 0 : (PBUSY) Interrupt status 0: Interrupt requested 1: No interrupt requested bit 6 : (CSEL) Cartridge option determination signal 0: Mode other than HS mode 1: HS mode P17H (W) : ICCTLR (FODBH) bit 4 : (DCTG) Development cartridge switching signal 0: Normal mode 1: Development cartridge mode (The user must not modify this bit.) bit 6 : (PRIE) Interrupt mask 0: Interrupt allowed 1: Interrupt suppressed P18H(W) : SWR(FODCH)bit 0 to 1 : (CSWO and CSW1) Cartridge mode switch (The user must not modify these bits.)

P19H (W) : IOCTLR (FODDH) bit 0 : (PF) Outputting to general output line (CCTLO) bit 1 : (SLIN) Outputting to general output line (CCTL1) bit 3 : (PINI) Cartridge reset 0: Reset on 1: Reset off P23H (R) : ITSR bit 1 : (IPBUSY) Status of interrupt from IC card or cartridge 0: Interrupt requested 1: No interrupt requested P23H (W) : CTLR3 (FODEH) bit 3 : (PINTDIS) Masking an interrupt sent from IC card or cartridge 0: Interrupts allowed 1: Interrupts suppressed (0 is generally set.) 7.4 IC Card Interface

7.4.1 Overview

EHT-10/EHT-10/2 has the IC card interface and OS supports BIOS ICCARD and an IC card as a disk. See "CHAPTER 4 BIOS OVERVIEW", "Section 6.5 IC card" and "Section 11.3 Extending IC Card Protocol" for details.

This section explains the I/O registers used to control IC card I/F and power control related to the IC card I/F.

## 7.4.2 I/O registers used for 1C card I/F

The I/O address space related to the IC card I/F is structured as shown below.

R/W	I/O Address	Register name	bit7	bit6	bit5	bit4	bit3	bit2	bit1	bit0	Notes
R	P16H	IOSTR						1C CLS			(*1)
	P17H	ICCTLR			IC ITE			IC DIR	100	1C0 SC	
	P19H	IOCTLR						ICR			(*1)

(\*1) Bits other than bit 2 are used for others.

Table 7.8 I/O Registers Related to IC Card I/F

The serial communication I/O port is used in addition to the I/O ports shown above because an IC card is accessed through serial communication. Other bits must be saved when data is written in a write register (ICCTLR or IOCTLR). Write data saved in memory is fetched, only the bit to be modified is manipulated, and then the value is written in memory and the register. The memory addresses of data corresponding to registers are listed below.

1/0		PAM			
I/O address	I/O name	Address	Name		
P17H	ICCTLR	FODBH	RZICCTLR		
P19H	IOCTLR	FODDH	RZIOCTLR		

Register name

```
'OSTR (10 Status Register)
    ICCLS: Indicates the lid status of IC card reader.
    =1: Closed
    =0: Opened
```

```
ICCTLR (IC card Control Register)
ICITE: Interrupt control according to the lid status of IC card reader
=0: Disabled
=1: Enabled
ICDIR: IC card reader read/write switch
=0: Write
=1: Read
ICC: IC card reader power control
=0: Vcc not supplied
=1: Vcc supplied
ICOSC: IC card reader clock oscillation control
=0: Not oscillated
=1: Oscillated
IOCTLR (IO Control Register)
ICR: IC card reader reset signal control
=0: Reset off
=1: Reset on
```

7.4.3 Controlling IC card reader power supply

Power (+5 V), clock, and reset signals are supplied to the IC card reader. The power and clock require a fixed time after turned on to become stable.

This section explains the power, clock, and reset control timings at power on and oif.

(1) At power on





### 7.5 Barcode Reader Interface

#### 7.5.1 Overview

EHT-10/EHT-10/2 has the barcode interface and OS supports the barcode interface with BIOS BARCODE. BIOS BARCODE supported by OS is explained in detail in "CHAPTER 4 BIOS OVERVIEW" and "Section 11.5 Adding Barcode Decoder".

This section explains the I/O registers used for barcode I/F and the controlling barcode reader power supply.

### 7.5.2 I/O registers for barcode I/F

The EHT-10/EHT-10/2 barcode interface consists of +5 V logic power supply, +5 V LED power supply, and barcode input signals. The following figure shows the I/O address space related to the barcode I/F:

R/W	I/O Address	Register name	bit7	bit6	bit5	bit4	bit3	bit2	bit1	bit0	Notes		
	POOH	ICRL.C	Lo	Lower 8 bits of the current FRC.									
	P01H	ICRH.C	U	Upper 8 bits of the Current FRC.									
0	PO2H ICRL.B Lower 8 bits of FRC held when the barcode input signal is changed.												
ĸ	P03H	ICRH.B	Upper barco	per 8 bits of FRC held when the rcode input signal is changed.									
	P04H	ISR				ICF			(*1)				
	P05H	STR							BCRD		(*2)		
	POOH	CTLR1					SW BCR	BCR1	BCRO		(*3)		
W	P04H	IER									(*1)		
	P17H	ICCTLR								(*4)			

(\*1) Used by others except for bit 2
(\*2) Used by others except for bit 1
(\*3) Used by others except for bit 3 to 0
(\*4) Used by others except for bit 3

Other bits must be saved when data is written in a write register (CTLR1, IER, or ICCTLR). Write data saved in memory is fetched, only the bit to be modified is manipulated, and then the value is written in memory and the register. The memory addresses of data corresponding to registers are listed below.

1/0		RAM			
I/O address	1/O name	Address	Name		
POOH	CTLR1	FOC6H	RZCTLR1		
РО4Н	IER	F42EH	RZIER		
P17H	ICCTLR	FODBH	RZICCTLR		

Register name

- ICRL.C (Input Capture Register Low Command Trigger) Lower 8 bits of the current free running counter (FRC)
- ICRH.C (Input Capture Register High Command Trigger) Higher 8 bits of the current FRC. ICRL.C must be read first if it is required to be read.
- ICRL.B (Input Capture Register Low Barcode Trigger) Lower 8 bits of FRC held when the barcode input signal is changed.

ICRH.B (Input Capture Register High Barcode Trigger) Higher 8 bits of FRC held when the barcode input signal is changed. The interrupt occurred due to barcode input signal change is reset when this register is read.

- ISR (Interrupt Status Register) ICF: Interrupt that occurs when FRC is latched to ICR due to barcode input signal change
- STR (Status Register) BCRD: Barcode input signal

CTLR1 (Control Register 1) SWBCR: +5 V logic power switch for barcode reader 0 = Off 1 = On BCR1 or 0: Latch trigger polarity selection 00 = Trigger suppressed 01 = Fall trigger

- 10 = Rise trigger
- 11 = Rise and fall triggers

IER (Interrupt Enable Register) ICT: ICF interrupt control

ICCTLR (IC card Control Register) BCRP: Barcode reader LED power switch

7.5.3 Barcode reader power supply control

Logic power (+5V) and LED power (+5V) are supplied to the barcode reader. Logic power and LED power require a fixed time after turned on to become stable.

SOFTWARE 7 - 19

Fig 7.6 shows the logic and LED power timings.



Fig 7.6 Barcode Reader Power Supply Control Timing

Barcode output is not stable for 1 second after the logic power is turned on. Also, barcode output is not stable for 800 micro seconds after the LED power is turned on. This section explains the function overview of slave CPU 7508, data transmission to and from 7508, and various commands. See "Section 4.3 Key Input" and "Section 8.4 7508 Interrupts" for reference.

7.6.1 7508 functions

7508 has the following functions:

- (1) Keyboard scan and control
- (2) Main-CPU power on/off
- (3) Reset switch control
- (4) Battery voltage management and switching
- (5) Alarm function
- (6) 1-second timer
- (7) Power switch control
- (8) Calendar clock
- (9) D-RAM refresh signal control
- (10) Serial data transmission to and from main CPU

Commands and data are transmitted to or from Z-80 through serial data line in handshake mode.

For functions (1) to (7), an interrupt is sent to Z-80 and the Z-80 side can find out the cause by reading the 7508 status.

7.6.2 750a interface

(1) Input and output ports

EHT-10/EHT-10/2 provides the following I/O ports to transmit commands and data to or from 7508:

1 For serial data transmission

PO6H (R) [SIOR]

7 6 5 4 3 2 1 0

8 bits data

Data sent from 7508.

PO6H (W) [SIOR]

Command and data sent to 7508

PO5H (R) [STR]



SOFTWARE 7 - 21

PO1H (W) [CMDR]



> RESET RDYSIO: Controls RDYSIO signal. =0: No operation is performed. =1: The RDYSIO signal is reset.

2 Interrupts

PO4H (R) [ISR]

This indicates that an interrupt is sent from 7508. INTO=0: No interrupt 1: An interrupt is sent.

PO4H (W) []ER]

7 6 5 4 3 2 1 0

This indicates that interrupts sent from 7508 are suppressed or allowed. IERO=0: Interrupts suppressed

1: Interrupts allowed

x indicates that this bit is not related to 7508.

(2) 7508 and transmission procedure

This section explains the procedure to transmit data to or from 7508 and notes on transmission operations.

Sending data to 7508 Figure 7.7 shows the procedure to send a command or data to 7508.



Fig 7.7 Command (Data) Send Procedure

To send data after each command, the above operation is repeated as much as the number of commands and data items.

Step : Explanation

- (1) RDYSIO? Whether 7508 is ready to receive a command (data) is checked. PO5H is read, processing goes to step (2) if bit 2 in PO5H is 1, and step (1) is repeated if bit 3 is 0.
- Write SIOR
   A command (data) is sent to 7508.

   Data or command to be sent is written in PO6H.
- (3) Reset RDYSIO 7508 RDYSIO signal is reset. 02H is written in POlH.
- 2 Receiving data from 7508 Figure 7.8 shows the procedure to receive data from 7508.



Fig 7.8 Data Receive Procedure

SOFTWARE 7 - 23

Step : Explanation

- (1) Command sending A command is sent according to the send procedure shown in the above figure.
- (2) RDYSIO? Whether data sent from 7508 can be received is checked. PO5H is read, processing goes to step (3) if bit 3 in PO5H is 1, and step (2) is repeated if bit 3 in PO5H is 1.
- Read SIOR
   Data received from 7508 is read.
   PO6H is read to fetch data sent from 7508.
- (4) Data end? Whether data items as much as the number of sent commands are received is checked. Processing goes to step (5) if there are more data items to be received.
- (5) Reset RDYSIO 7508 RDYSIO signal is reset. 02H is written in PO1H.
- (3) Notes

This section explains the notes on transmitting commands and data to or from 7508.

- I Interrupts sent from 7508 mist be suppressed while a command or data is transmitted to or from 7508. Interrupts can be suppressed in the following three ways:
  - (a) DI instruction(b) Rewriting IER (PO4H)(c) BIOS MASKI

Interrupts must be suppressed while processing shown in Figures 7.7 or 7.8 is being executed.

2 Send or receive data sequence for 7508 commands must be completed. In other words, subsequent operations are not guaranteed unless required number of data items are sent or received.

7.6.3 7508 commands

This section explains the 7508 commands. Table 7.10 shows the list of 7508 commands.

Number : Function (Code)

```
1 : Power OFF (01H)
2 : Read Status (02H)
3 : KB Reset (03H)
4 : KB Repeat Timer 1 Set (04H)
5 : KB Repeat Timer 2 Set (14H)
6 : KB Repeat Timer 1 Read (24H)
7 : KB Repeat Timer 2 Read (34H)
8 : KB Repeat OFF (05H)
9 : KB Repeat ON (15H)
10 : KB Interrupt OFF (06H)
11 : KB Interrupt ON (16H)
12 : Clock Read (07H)
12 : Clock Read (07A)

13 : Clock Write (17H)

14 : Power Switch Read (08H)

15 : Alarm Read (09H)

16 : Alarm Set (19H)

17 : Alarm OFF (29H)

18 : Alarm OFF (29H)
18 : Alarm ON (39H)
19 : DIP Switch Read (OAH)
20 : 7 Characters Buffer (OCH)
21 : 1 Character Buffer (1CH)
22 : 1 Sec. Interrupt OFF (ODH)
23 : 1 Sec. Interrupt ON (1DH)
24 : KB Clear (OEH)
25 : System Reset (OFH)
```

Table 7.10 7508 Commands Command or data can be identified from MSB. MSB is 0 for a command and 1 for data.

This command turns off the main-CPU power.

[Sequence]

7	6	5	4	3	2	1	0	
0	0	0	۵	0	0	0	1	(Write)

<Explanation>

(1) The main-CPU (Z-80) power is turned off.

(2) This command cannot be used in an application program. BIOS POWER OFF is used in an application program in order to turn the power off.

This command reads the 7508 status or key code.

[Sequence]



#### <Explanation>

(1) This command is used to read the 7508 status when an interrupt is sent from 7508 or after reset is cleared.

(2) The status is mainly classified as follows:

- I Key code 2 Status
- 2 Status

The rest of this section explains the key code and status.

```
1 Key code (scan code)
```

When the read /508 status is BEH or less, data is a key scan code. Figure 7.9 shows the scan codes of EHT-10/2 keys. 30H is indicated as the status when a touch-panel key on the EHT-10/2 is pressed. Actually, touch-panel key is scanned during OS interrupt processing.



Fig 7.9 Scan Codes (EHT-10/2)

(Note) The values in the left figure are represented in hexadecimal notation .

SOFTWARE 7 - 27

2 Status

When the read 7508 status is COH or more, the status has been changed due to interrupt occurrence or reset clearance. BFH is indicated when the status has not been changed.

The cause of status change is indicated by a bit as shown below. If there are two or more causes of status change, two or more bits are turned to 1.



To determine the subsequent operations, the EHT-10/EHT-10/2 OS reads the 7508 status during interrupt processing when an interrupt occurs, or during O-address start processing (system initialization, reset, power on, or alarm/wake processing) when reset is cleared.

This command initializes the keyboard status.

[Sequence]



<Explanation>

- (1) The keyboard status is initialized.
  - 1 656 ms is set as the keyboard repeat start time.
  - 2 70 ms is set as the keyboard repeat interval.
    3 The key buffer is cleared.

  - 4 Interrupts caused by key input operation are allowed.

(2) The keyboard is scanned and the information of the key that has been pressed is stored in the buffer.

(3) OS uses this command during system initialization and reset processing.

This command sets the keyboard repeat start time.

[Sequence]



<Explanation>

(1) When a general key other than switches and special keys is kept pressed, the input is repeated. This command sets the time after a key is pressed until repeat operation is started.

(2) A value up to 2 seconds can be specified in the unit of 1/64 seconds (about 15 ms) as send data. 1 must be set as MSB of data.

(3) The initial value is 656 ms.

This command sets the keyboard repeat interval.

## [Sequence]



### <Explanation>

(1) This command sets the repeat interval to be used when a key is kept pressed.

(2) A value up to 0.5 seconds can be specified in the unit of 1/256 seconds (about 3.9 mp) as send data. 1 must be set as MSB of data.

(3) The initial value is about 70 ms.

This command reads the keyboard repeat start time.

[Sequence]



<Explanation>

(1) The current keyboard repeat start time is indicated.

(2) As the command explained in 4, a value is indicated in the units of 1/64 seconds (about 15 ms) as receive data. MSB of receive data is 1 but MSB of data indicating the repeat start time is 0.

This command reads the keyboard repeat interval.

## [Sequence]



#### <Explanation>

(1) The current keyboard repeat interval is indicated.

(2) As the command explained in 5, a value is indicated in the units of 1/256 seconds (3.9 ms) as receive data. MSB of receive data is 1 but MSB of data indicating the receat interval is 0.

This command disables the keyboard repeat function.

[Sequence]



<Explanation>

(1) This command disables the automatic keyboard repeat function. Only one key code is sent even if a key is kept pressed.

(2) OS specifies to disable the automatic keyboard repeat function as the default for 7508.

(3) OS supports BJOS CONOUT (ESC+FOH) to switch the automatic keyboard repeat function on/off.

This command enables the keyboard repeat function.

[Sequence]



<Explanation>

(1) This command enables the automatic keyboard repeat function. When a key is kept pressed, a key code is repeated every specified repeat interval after the specified repeat start time elapses.

(2) The automatic keyboard repeat function is disabled as the default.

(3) OS supports BIOS CONOUT (ESC+FOH) to switch the automatic keyboard repeat function on/off.

This command disables all interrupts caused by key input operations.

[Sequence]



<Explanation>

(1) Interrupts caused by key input operations and sent to Z-80 are disabled. When a key is pressed, only a code is input to the 7508 key buffer but an interrupt is not sent to Z-80. When the command "KB Interrupt ON" explained in Section II is executed after a key is pressed, interrupts caused by the pressed keys are sent to Z-80 unless the key buffer is empty.

(2) Key codes accurred after the key huffer becomes full are discarded.

(3) OS supports BIGS MASKI for this operation.

This command enables all interrupts caused by key input operations.

[Sequence]



<Explanation>

(1) After this command is executed, interrupts are sent to the main CPU.

- (2) Key interrupts are enabled as the default.
- (3) OS supports BIOS MASKI for this operation.

This command reads the current 7508 time.

### [Sequence]

7	6	5	4	3	2	1	0	
0	0	0	0	0	1	1	1	(Write)
0	0	0	0					(Read) Tens digit of year
0	0	0	0					(Read) Unit digit of year
								(Read) Month — Tens digit in
								(Read) Day (bits 7 to 4) and unit digit in
								(Read) Hour (bits 3 to 0)
								(Read) Minutes
								(Read) Second
0	0	0	0					(Read) Day of week

<Explanation>

(1) This command reads the 7508 calendar clock.

(2) The receive data indicates year, month, day, hour, minute, second, and day of week in this order. Each item is represented in BCD code.

(3) The hour is indicated by 24-hour clock. As the day of week, 1 indicates Sun, 2 indicates Mon,...6 indicates Sat. If a logically incorrect calendar clock is set, the contents of read data is not guaranteed because 7508 does not check the set data.

(4) OS supports BIOS TIMDAT for this operation.

This command sets the 7508 calendar clock.

### [Sequence]

7	6	5	4	3	2	1	0			
0	0	0	1	0	1	1	1	(Write)		
1	0	0	0					(Read)	Tens digi	it of year
1	0	٥	0					(Read)	Unit digi	it of year
1								(Read)	Month	— Tens digit in
1								(Read)	Day	higher 4 bits (bits 7 to 4) and unit digit in
1								(Read)	Hour	lower 4 bits (bits 3 to 0)
1								(Read)	Minutes	
1								(Read)	Second	
1	0	0	0					(Read)	Day of we	eek

<Explanation>

(1) Send data specifies year, month, day, hour, minute, second, and day of week in this order. Each item is represented in BCD code. 1 must be set as MSB of the data.

(2) The hour is indicated by a 24-hour clock. The day of week is automatically updated between 0 to 6. If a logically incorrect data is set, the contents of the calendar clock are not guaranteed because 7508 does not check the set data.

(3) To set only one or more item, set 1 to all the bits of other items that are not to be set and send the entire data.

(4) OS supports BIOS TIMEAT for this operation.

This command reads the status of current power switch.

# [Sequence]



<Explanation>

(1) This command reads the status of the power switch placed at the side of the EHT-10/EHT-10/2.

(2) OS supports BIOS READSW for this operation.

This command reads the current alarm time.

# [Sequence]



<Explanation>

(1) This command reads the current alarm time in the order of month, day, hour, minute, second, and day of week. Each item is represented in BCD code.

(2) OS supports BIOS TIMDAT for this function.
This command sets the alarm time.

## [Sequence]



<Explanation>

(1) This command sets the alarm time in the order of month, day, hour, minute, second, and day or week. Each item is represented in BCD code. 1 must be set as MSB of the data.

(2) The item is ignored when 1 is set to all bits of an item (for example, every minute is assumed when 1 is set to the entire bits indicating minute).

(3) The year and the unit digit of second cannot be set.

(4) If the command "Alarm ON" explained in Section 18 is executed after this command is executed, an alarm is generated at the specified time.

(5) OS supports BIOS TIMDAT for this operation.

SOFTWARE 7 - 42

This command disables the alarm function.

[Sequence]



<Explanation>

(1) This command disables interrupts caused by the alarm function and sent to the main CPU. The alarms occurred during alarm off status are ignored (the alarm interrupts occurred in this status will not be generated even after the status is changed to alarm on).

(2) OS supports BIOS TIMDAT and MASKI for this operation.

This command enables the alarm function.

[Sequence]



<Explanation>

(1) This command enables alarm interrupts sent to the main CPU. An alarm interrupt occurs when the set time comes after this command is executed.

(2) OS supports BIOS TIMDAT and MASKI for this operation.

This command reads the status of DIP switches placed at the back of the main frame.

# [Sequence]



<Explanation>

(1) The DIP switches correspond to the read data bits as follows:



(2) Each receive data bit indicates on when it is 1 and off when it is 0. However, MSB (bit 7) for EHT-10 differs from the one for EHT-10/2 as follows: MSB=1: EHT-10/2 =0: EHT-10

(3) OS uses the dip switch 1 (bit 0 of the received data) to change the setting of the EHT-10/EHT-10/2.

DIP switch 1 = 0: Overseas = 1: Japan

The reading of the dip switch setting is supported by READSW of the BIOS.

This command specifies 7 characters as the length of key code buffer.

[Sequence]



<Explanation>

(1) This command reserves a 7-character buffer at the 7508 side and stores the key code when a key is pressed and the key code is not fetched.

(2) The key and switch codes of keys and switches pressed after the buffer tecomes full are ignored.

(3) A 7-character buffer is used as the default.

This command specifies 1 character as the length of key code buffer.

[Sequence]



<Explanation>

(1) This command sets 1 character as the length of key buffer at the 7508 side.

(2) The key buffer function is same as the one explained in 20. " 7 Character Buffer".

(3) OS also has a key buffer and this buffer can store up to 32 characters.

This command disables interrupts.

[Sequence]



<Explanation>

(1) This command disables interrupts sent every second from 7508. 1-second interrupts are no longer sent to the main CPU after this command is executed.

- (2) OS supports BIOS MASKI for this operation.
- (3) OS uses I-second interrupts for the following operations:
  - 1 Automatic power off time monitoring
  - 2 Alarm screen display time monitoring

The above operations can no longer be executed when 1-second interrupts are disabled.

This command enables 1-second interrupts sent from 7508.

[Sequence]



<Explanation>

(1) This command enables interrupts sent every second from 7508. 1-second interrupts are sent to the main CPU every second after this command is executed.

(2) OS supports BIOS MASKI for this operation.

(3) OS sends the "1 Sec. Interrupt ON" command during reset or system initialization.

This command resets the keyboard section.

[Sequence]



<Explanation>

(1) This command clears the 7508 key buffer, scans the keyboard, and stores the information of the current key.

(2) Differing from the "KB Reset", the "KB Clear" command does not initialize the repeat start time, etc.

This command resets 7508.

[Sequence]



<Explanation>

(1) This command initializes entire 7508.

(2) An application program cannot use this command. System initialization is executed for EHT-10/EHT-10/2 when the "System Reset" command is sent.

7.7 Other 1/0

7.7.1 Overview

This section explains controlling the buzzer, timer and backlight.

7.7.2 Buzzer

EHT-10/EHT-10/2 has piezo-buzzer that is controlled by software. OS supports BIOS BEEP for controlling this buzzer (see "CHAPTER 4 BIOS OVERVIEW"). Figure 7.10 shows the overview of the circuit. Software can specify 512 Hz, 1024 Hz, or 2048 Hz for the frequency. Further, an optional frequency can be output by outputting the frequency generated by the software timer to the SP terminal.



#### Fig 7.10 Buzzer Interface

(Note 1) O is generally set to the SP terminal (P19H, bit 7) to enable outputs from CTLR3.

7.7.3 Timer

Figure 7.11 shows the EHT-10/EHT-10/2 timer circuit. FRC is a 16-bit free running counter and the input clock is 614.4 KHz (1.6276 sec). Therefore, FRC overflows every 0.106667 sec determined as follows: 1.6276 x 2<sup>16</sup> = 0.106667 sec

The value in FRC can be found out by reading ICRL.C (PODH) and ICRH.C (PO1H). (Note 1) ICRL.C and ICRH.C must be read in this order because the FRC value is latched in the input capture register(ICR) when ICRL.C is read. When FRC overflows, an interrupt (OVF interrupt) is sent to the main CPU and the status is set in ISR (PO4H) bit 3 (INT3). The OVF status is reset when 1 is set to CMDR (PO1H) bit 2 (Res OVF) and write operation is executed. To use the timer in a user program, disable OVF interrupts and structure the timer by using ICRL.C, ICRH.C, and ISR bit 3.

SOFTWARE 7 - 52

(Note 1) The FRC value is also latched in ICR when the signal sent from the barcode reader changes. In this case, the FRC value is read from ICRL.B (PO2H) and ICRH.B(PO3H).



614.4 KHz clock

Reset OVF (POCH bit3)

Fig 7.11 Timer Clock

7.7.4 EL backlight (EHT-10/2B)

EHT-10/2B model that has the EL backlight is provided so that it can be used in a dark room. The EL backlight can be turned on or off by the backlight switch or controlled by software. OS supports BIOS BACK LIGHT and automatic backlight off function at key input for this operation (see "CHAPTER 4 BIOS OVERVIEW" and "Section 2.3.5 Sleep function and automatic power off function"). Controlling the EL backlight by software is effective only when the backlight switch is on.

# CHAPTER 8 INTERRUPT PROCESSING

8.1 Overview

EHT-10/EHT-10/2 supports five types of interrupts. When an interrupt
occurs, the vector corresponding to the interrupt vector is set and control
is passed to the interrupt processing routine.
EHT-10/EHT-10/2 supports the following five types of interrupts:
 1 7508 (4-bit CPU)
 2 ART (RXRDY)
 3 ICF (Input capture)
 4 OVF (FRC overflow)
 5 EXT (timer, cartridge, and IC card)
7508 (4-bit CPU) interrupts are caused by the following events:
 1 Keyboard
 2 Power switch on/off
 3 Alarm time
 4 Power failure
 5 1-second interrupt
EXT interrupts are caused by the following events:
 1 1-msec/8-msec timer interrupt
 2 Interrupt sent from cartridge I/F

3 Interrupt sent from IC card (back panel or overcurrent)

Interrupt	Cause	Reset conditions
7508	<ul> <li>Keyboard input</li> <li>1-second interval</li> <li>Alarm time</li> <li>Power switch on or off</li> <li>Power voltage failure</li> </ul>	- A response is sent to 7508
ART (RXRDY)	<ul> <li>Serial data receive flag RXRDY in the ART section (gate array) is set.</li> </ul>	- Receive data register ARTDIR (P14H) is read.
ICF	- The input signal sent from the barcode reader is changed.	- The input capture register (PO3H) is read.
OVF	<ul> <li>Free running counter (FRC) overflows. The cycle is about 106.7 msec because FRC is 16 bits and the input clock is 614.4 kHz.</li> </ul>	<ul> <li>"Reset OVF" command (that writes 1 to bit 2 of command register CMDR (PO2H))</li> </ul>
EXT	- 1-msec or 8-msec interval.	- 1 is written in P23H bit 1 to reset the interrupt.
	- An interrupt signal is received from the cartridge I/F.	<ul> <li>A response is sent to the cartridge device (in the case of a printer unit, the printer unit is made busy by data output because the interrupt is a ready interrupt).</li> </ul>
	<ul> <li>The back panel of IC card is opened.</li> <li>Overcurrent is flowed in IC card.</li> </ul>	- This cannot be reset by software.

Table 8.1 shows the causes and reset conditions of interrupts.

Table 8.1 Causes and Reset Conditions of Interrupts

### 8.2 Interrupt Vector

EHT-10/EHT-10/2 supports five types of interrupts. The interrupt vector table is stored in FFFOH to FFFFH. Interrupts requested are accepted in the priority order.

Priority order	Interrupt cause	Vector	Address in RAM
1 (Highest priority) 2 3 4 5	7508 (4bit CPU) ARī (RXRDY) ICF (Input Capture) OVF (Overflow of FRC) EXT (timer, cartridge, and IC card)	F0H F2H F4H F6H F8H	FFFOH, FFF1H FFF2H, FFF3H FFF4H, FFF5H FFF6H, FFF7H FFF8H, FFF9H

Table 8.2 Interrupt Vector Table

EHT-10/EHT-10/2 controls interrupts as follows:

(1) Setting interrupt mode and interrupt vector

The following operations are executed when processing is started from address GUOGH (system initialization, reset, or power or star\*): 1 moue-2 interrupt is specified as the interrupt mode. 2 FFH (indicating to use FFFOH to FFFFH as the interrupt vector table, is set in register I.

(2) Loading the interrupt vector table

The interrupt vector data stored in OS ROM is loaded to FFFOH to FFFFH at reset or system initialization.

#### 8.3 Controlling Interrupts

- 8.3.1 Controlling interrupts by the system
- (1) Setting by the system

The EHT-10/EHT-10/2 OS sets the interrupt initial status at the timing listed in the table below.

	7508	ART	ICF	OVF	EXT	7508		EXT			
						Кеу	1Sec	Alarm	1/8mSec	Cartridge	IC card
System initiali- zation	0	x	x	0	0	0	0	x	X	x	x
Reset	0	X	X	0	0	0	0	-	X	X	x
Warm boot	-	X	X	-	-	-	-	-	-	-	-
Restart power on	0	X	X	0	0	-	-	-	-	-	-
Continue power on	-	-	-	-	-	-	-	-	-	-	-

O: Enabled X: Disabled -: No change

### Table 8.3 Setting Interrupt Mask Status

(2) Modules in which interrupts are disabled

While data is read from or written to FDD or an IC card as a disk, DI status is set so that any interrupts are disabled. OVF interrupts are disabled during BEEP operation.

(3) Interrupts occurred during interrupt processing

Interrupt processing is generally executed in DI status so that multiple interrupt processing is not executed. However, interrupts other than 7508 interrupts are enabled during 7508-interrupt processing. Also, any interrupts are enabled during 7508-alarm interrupt processing.

# 8.3.2 Disabling and enabling interrupts

(1) Overview

Interrupts are disabled or enabled in the following two ways: 1 By BIOS MASKI

2 Interrupt enable register IER (PO4H) is directly rewritten.

See "CHAPTER 4 BIOS OVERVIEW" for using BIOS MASKI.

Further, the following interrupts can be enabled or disabled individually by using a command or rewriting the I/O register: - 7508 interrupts - Key interrupt - Alarm interrupt - 1-second interrupt - EXT interrupts - 1-msec/8-msec interrupt - Cartridge interrupt - IC card interrupt (2) Directly rewriting IER Figure 8.1 shows how to directly rewrite IER. DI RZIER (F42EH) is the interrupt (1)enable or disable status storage area. RZIER is rewritten (2)(Example) External interrupts are enabled. Data is output to IER (3)10 LD A. (RZIER) 10H ΕI (4)OR LD (RZIER),A OUT (IER), A EI Fig. 8.1 Rewriting IER Address : Variable name (Byte length) F42EH : RZIER (1)- Interrupt enable/disable status storage area 7 5 4 3 2 1 0 6 -> 7508 interrupt -> ART interrupt -> ICF interrupt -> OVF interrupt -> EXT interrupt (Each bit indicates enable status when it is 1 and disable status when it is 0.- The RZIER bit structure is same as IER (PO4H) bit structure. (3) Controlling 7508 interrupts See "Section 7.6 7508 Commands" for controlling 7508 interrupts.

(4) Controlling EXT interrupts

SOFTWARE Page 8 - 5

Generally, 1 (interrupt enabled) is set to the IER EXT interrupt bit (bit 4) for EXT interrupts so that each interrupt can be disabled or enabled. Table 8.4 shows the I/O registers used to control each EXT interrupt. The contents of each I/O register are output by rewriting the data stored in memory and outputting the rewritten data to the I/O port in the same way as rewriting IER explained in (2). Port Address (R/W) : Port name (RAM data address) P17H (W) : ICCTLR (FODBH) bit 6 : (PRIE) cartridge interrupt =0: Enabled =1: Disabled bit 5 :(ICITE) IC card interrupt =0: Disabled =1: Enabled P23H (W) : CTLR3 (FODEH) bit 3 : (PINTDIS) cartridge and IC card interrupt =0: Enabled =1: Disabled bit 4 : (OVFITV) 1-msec/8-msec interrupt switch =0: 8 msec =1: 1 msec bit 5 :(OVFEN) 1-msec/8-msec interrupt =0: Disabled =1: Enabled Table 8.4 EXT Interrupt Mask I/O Registers

.

# 8.3.3 Interrupt processing time

Interrupt		Number of states	Time (micro Sec)	Notes	
	Key input	4720	1282	For EHT-10 (touch panel)	
7 5 0 8		2384	647	For EHT-10/2 (keyboard)	
	1 Sec	1913	519		
	Alarm	2089	568	Alarm display disabled	
	Power switch on	1642	446	Power off processing disabled	
	Power switch off	1701	462	Power off processing disable	
	Power failure	1713	465	Power failure processing disabled	
	ART	1716	466	No buffer control specified	
ICF		-	-	Barcode reading is performed in ICF interrupt processing.	
OVF		1263	343	No cursor blinking	
E X T	Timer	1572	427		
	Cartridge	1603	436	For printer unit	
	IC card	1757	477		

Table 8.5 shows the interrupt processing time required for EHT-10/EHT-10/2.

# Table 8.5 Interrupt Processing Time

(Note 1) The above table lists the standard processing time for each interrupt and the time required for actual operation may slightly differ.

(Note 2) The time required for an 7508 interrupt is determined by using the communication time to 7508 CPU as 0 micro Sec. For actual operations, about 3000 to 6000 states (800 to 1600 micro Sec) must be added for communication.

8.3.4 Processing executed when an interrupt occurs

(1) Interrupt processing

EHT-10/EHT-10/2 has four types of banks and processing may be in progress in any bank when an interrupt occurs. Because of this, the entry section of interrupt processing is in the resident section (EOOOH to FFFFH), the bank is switched to the appropriate one in this entry section, and actual interrupt processing is executed in the OS ROM that stores the main section of interrupt processing.



Fig 8.2 Processing Flow at Interrupt Occurrence

(2) Relationship between interrupt processing and BIOS If an interrupt occurs during BIOS and interrupt processing is executed immediately, the current program may not be continued or power off processing in continue mode cannot be done after control returns. Therefore, BIOS turns on the BIOS execution flag (PREBIOS) at the beginning of processing and the interrupt flag is set to indicate interrupt occurrence if such an interrupt occurs. At the end of BIOS processing, the interrupt flag is checked and actual interrupt processing is executed if the flag indicates interrupt occurrence (PSTBIOS).

In BIOS processing that forms a loop (CONIN, RSIOX, TCAM, or ICCARD), the interrupt flag is checked as explained above and interrupt processing is executed if the flag indicates interrupt occurrence.

See "CHAPTER 4 BIOS OVERVIEW" for PREBIOS and PSTBIOS.

#### 8.4 7508 Interrupts

8.4.1 Overview

When an interrupt is sent from 7508, interrupt processing performs serial communication with 7508 to read the 7508 status. One of the following processing is executed according to this status:

- Key interrupt
- 1-second interrupt
- Power failure interrupt
- Alarm interrupt
- Power switch interrupt

7508 interrupt processing can perform multiple 7508 interrupt processing so that key input operations are enabled during alarm screen display.

## 8.4.2 Multiple interrupt processing

7508 interrupt processing can perform multiple interrupt processing by taking into account of another interrupt occurrence during interrupt processing. Multiple interrupt processing is performed by using interrupt level INTLEVEL (F0S2H) and stack. Figure 8.3 shows the structure of multiple interrupt processing.



Fig. 8.3 Multiple 7508 Interrupt Processing

SOFTWARE Page 8 - 9

### 8.4.3 7508 interrupts

(1) Overview

There are five types of 7508 interrupts and the type can be identified by reading 7508 status. When an interrupt is sent from 7508, a Status Read command (O2H) is sent to 7508 to read the 7508 status. The status indicates a hardware code for key interrupt when the read status is less than BFH. A power switch, 1-second, alarm, or power failure interrupt is indicated when the status is COH or more.

This status is stored in STS7508 (F3C3H). Power switch interrupts are further classified into power switch on and power switch off interrupcs.

Interrupts processing is executed after the status indicating power switch on/off, i-second, alarm, or power failure interrupt is set in INTFG (F3C2H). The operations to be executed in interrupt processing is determined from the status according to the TBL7508 (F0A3H).

The execution status of 1-second interrupt processing is set in FG7508 (F3C4H).

Address : Variable name (Byte length)

```
FOA3H : TBL7508 (16)
```

 This table is used to determine 7508 interrupt processing.
 Each byte is structured with the following bits: (Processing is executed when a bit is 1.)



FOB1H	C7H OR	E7H	OFF	<b>H80</b>
FOB2H			ON	<b>H80</b>

- Whether 1-second interrupt processing should be executed is determined by software after conversion.

F3C2H : INTFG (1)

This indicates the type of 7508 interrupt processing executed by OS.
 OOH is set in INTFG when key interrupt processing is executed.



When an interrupt is sent from 7508, the 7508 status is read through serial communication from 7508. If this status is less than BFH, the

interrupt is handled as a key interrupt.

SOFTWARE Page 8 - 11

For EHT-10, the 7508 status indicates 80H when an interrupt is sent from the touch panel. When 7508 status indicates 80H during interrupt processing, the touch panel is scanned to find out the pressed touch-panel key. See "HARDWARE Section 4.5 Touch Panel Interface" for scanning the touch panel.

For EHT-10/2, the read 7508 status indicates a position code (hardware code) in the keyboard matrix. See "Section 7.6.2 7508 interface" for the relationship between keys and position codes. During key interrupt processing, the key hardware code is stored in the key buffer (KBUF). The key code is discarded if the key buffer is full.

(3) 1-second interrupt processing

1-second interrupt processing performs the following operations: (a) 16-bit timer TIMERO (F02DH) is increased by 1. (b) 16-bit timer TIMER1 (F02FH) is decreased by 1.

- (c) Timer function TMFUNC (F205H) processing The following operations are performed when TMFUNC (F025H) is not equal to OOH: TMSEC (F3C6H) is decreased by 1. If TMSEC becomes OOH as a result of decrease operation, OuH is set in TMFUNC and FFH is set in TMFLAG (F206H). (d) The power down mode time for printer unit is counted down.
  (e) The power off time of IC card is counted down.
  (f) The alarm repeat count is counted down.

(a), (b), and (c) are the 1-second counters for application programs. (d), (e), and (f) are the counters used for system control. TIMERO is also used by the system to monitor automatic-power-off time.

- Using TIMERO and TIMER1 1 TIMERO and TIMER1 are the 16-bit 1-second counters and are increased or decreased by 1 every time a 1-second interrupt occurs. These counters can be used as reference counters without changing the contents to measure processing time.
- 2 Using TMFUNC TMFUNC is a user function used to monitor a fixed time. Figure 8.4 shows how to use TMFUNC.



SOFTWARE Page 8 - 13

the alarm time determination. When the specified alarm time comes, an alarm interrupt is generated up to 10 times. OS assumes only one alarm interrupt among the generated alarm interrupts as the actual alarm interrupt. A hook is set in alarm interrupt processing so that processing can be extended by application programs. See "CHAPTER 10 SYSTEM HOOK AND JUMP TABLE" for details on hook. Using YALRMDS (FOB6H), alarm screen display can be suppressed by software during alarm interrupt processing. (5) Power switch on interrupt processing This interrupt is generated when the power switch is turned on. Power switch on interrupt processing sets power switch flag PWSWONFG (F3C8H). Address : Variable name (Byte length) F3C8H : PWSWONFG (1) - Power switch on flag =00H: Power switch on interrupt not occurred =FFH: Power switch on interrupt occurred - OOH is set at power on start when the main power is off. (6) Power failure interrupt processing A power failure interrupt occurs when the voltage of the main battery is lowered. Power failure occurs when the voltage of NiCd battery is lowered to about 4.7 V or less. When a power failure interrupt occurs, power failure interrupt processing sets power failure interrupt flag BTRYFG (FOB3H) and displays the power failure screen to warn the user. A power failure interrupt is sent every second after occurrence. 7508 forcibly turns off the main power supply if the main power supply is not turned off within 50 seconds after interrupt occurrence. Address : Variable name (Byte length) FOB3H : BTRYFG (1) - Power failure interrupt flag =00H: Power failure interrupt not occurred =FFH: Power failure interrupt occurred - OOH is set at power on start. (7) Power switch off interrupt processing A power switch off interrupt occurs when the power switch is turned off. Power switch off interrupt processing sets the power switch status to power switch off flag PWSWOFFG (F3C9H) and turns the power off. Address : Variable name (Byte length) F3C9H : PWSWOFFG Power switch off flag =00H: Power switch off interrupt not occurred =01H: Power switch off interrupt occurred

8.4.4 Suppressing alarm or power off (failure) interrupt processing

(1) Overview

For an alarm, power off, or power failure interrupt, interrupt processing can suppress the interrupt and only indicates the interrupt occurrence so that the screen is not displayed and actual power off processing is not executed. Suppressing interrupts is used when processing should not be interrupted or terminated during serial communication or I/O processing for cartridge I/F until group of consecutive operations are completed. OS performs this suppress processing by using PREBIOS and PSTBIOS. During BIOS processing, the alarm screen is not displayed or the power is not turned off (interrupt processing is executed if an interrupt occurs when processing is waiting for key input operation or RSIOX receive or send operation).

(2) Method of suppressing interrupts

The system provides YPOFDS (FOB4H) and YALMDS (FOB6H) as the power off processing and alarm/wake processing suppress specification flags. Each flag is structured as follows:



When a bit is 1, power off is not executed or the alarm screen is not displayed.

(Note 1) Generally, alarm processing is executed at key input operation regardless of YALMDS bits. However, if YALMDS bit 0 is 1, alarm processing is not executed even in key input operation.

SOFTWARE Page 8 - 15

Interrupt processing copies the YPOFDS and YALMDS values to YPOFST (FOB5H) and YALMST (FOB7H) and executes power off or alarm screen display processing if these values are OOH. If these values are not OOH, no operation is executed and processing is terminated.

(4) Procedure to suppress interrupts

Figures 8.5 and 8.6 show how to suppress interrupts. When a general application program suppresses an interrupt, use the bit (bit 6) for application programs and other bits should not be manipulated.



Fig 8.5 Suppressing Power off Interrupts

Step : Explanation

- (1) The YPOFDS specification bit is turned on.
   The YPOFDS (FOB4H) specification bit is turned to 1.
   Bit O is used by an application program.
- Processing in power off suppress status

   Processing is executed in power off suppress status.
   Power off occurrence must be checked often if processing is executed for a long time in power off suppress status.
- (3) The YPOFDS specification bit is turned off.
   The bit turned to 1 in step (1) is turned to 0.
- (4) Checking YPOFST
   The YPOFST (FOB5H) value is checked.
   If this value is not equal to OOH, it means that a power off interrupt occurred while power off interrupts are suppressed.
- (5) The YPOFST specification bit is turned off. - The YPOFST (FOB5H) bit turned on in step 1 is turned to 0.
- (6) Checking YPOFST

   The YPOFST value held after step (5) execution is checked.
   If this value is not equal to OOH, it means that another module is suppressing the power off intercupts.
- (7) Checking power off conditions

  The power off status is checked.
  If BTRYFG (FOB3H) is not equal to 00H, the power is turned off in continue mode. If BTRYFG is equal to 00H, PWSWOFFG (F3C8H) is checked. The power is turned off in restart mode if PWSWOFFG is not equal to 02H. In any other cases, the power is turned off in continue mode.
- (8) BIOS POWER OFF is called.
   BIOS POWER OFF is called according to the power off status obtained in step (7).

(note In step (7) and (8), you may call JPSTBIOS by turning bit 7 on.)



Fig 8.6 Suppressing to Display Alarm Screen

### Step : Explanation

- (1) The YALMDS specification bit is turned on.
   The YALMDS (FOB6H) specification bit is turned to 1.
   An application program uses bit 0.
- (2) Processing in alarm/wake suppress status

   Processing is executed in alarm/wake suppress status.
   Alarm/wake interrupt occurrence must be checked often if processing is executed for a long time in alarm/wake suppress status.

- (3) The YALMDS specification bit is turned off.
   The bit turned to 1 in step (1) is turned to 0.
- (4) Checking YALMST
   The YALMST (FOB7H) value is checked.
   If this value is not equal to OOH, it means that an alarm interrupt occurred while alarm interrupts are suppressed.
- (5) The YALMST specification bit is turned off.- The YALMST (FOB7H) bit turned on in step 1 is turned to 0.
- (6) Checking YALMST

   The YALMST value held after step 5 execution is checked.
   If this value is not equal to OOH, it means that another module is suppressing the alarm interrupts.
- (7) YALMST bit 7 is turned on.
   YALMST (FOB7H) bit 7 is turned on.
   After bit 7 is turned on, the alarm/wake screen is displayed by PSTBIOS executed in step (8).
- (8) RSPSTBIOS is called. - JPSTBIOS (FF96H) is called to display the alarm/wake screen.
- (4) Notes

When processing is made to wait for key input operation by BIOS CONIN, power switch off. power failure, or alarm/wake interrupt processing is executed unconditionally even if interrupts are suppressed. Interrupt processing is also executed unconditionally when an interrupt occurs during BIOS RSIOX waiting for send or receive operation or BIOS waiting for printer to become ready (other than screen dump).

Whether a power off or alarm/wake interrupt occurred is checked often if processing is executed for a long time in power off or alarm/wake suppress status.

## 8.5 ART Interrupts

An ART interrupt occurs when serial data receive flag RXRDY in the ART section is set.

The operations listed below are executed when an ART interrupt occurs. See the sections explaining BIOS RSIOX and ICCARD "Section 11.2 Extending Communication Protocol" and "APPENDIX9 SAMPLE 29" for ART interrupt processing.

- (1) No operation is executed if RSIOX Open has not been executed.
- (2) The error status is checked (framing error, receive overrun error, parity error, or receive buffer overflow).
- (3) Receive data is read and stored in the receive buffer.
- (4) The byte length of receive data stored in the receive buffer is checked if buffer control (XON/XOFF, DTR/DSR, or RTS/CTS) is specified. An XOFF code is sent if data is stored in 3/4 or more of the receive buffer.

### 8.6 ICF Interrupts

An ICF interrupt occurs when the input signal sent from the barcode reader is changed. Generally, ICF interrupts are masked by IER [PO4H] and this masking is cleared when BIOS BARCODE performs open operation.

See "Section 11.5 Adding Barcode Decoder", "Section 7.5 Barcode Reader Interface", and "Chapter 4 BIOS OVERVIEW" for details on ICF interrupts and barcode reader.

#### 8.7 OVF Interrupts

An OVF interrupt occurs when the free running counter (FRC) overflows. An OVF interrupt is generated every 106.7 msec because FRC is a 16-bit counter and input clock 614.4 KHz is used.

The system uses an OVF interrupt for cursor blinking. The cursor blinks every 500 msec. This blink interval can be changed by rewriting BLNKTIME (F078H).

Address : Variable name (Byte length)

F078H : BLNKTIME (1) - Cursor blink interval - The unit is 100 msec and the initial value is 04H. 8.8.1 OVerview An EXT interrupt occurs when one of the three events listed below occurs. The interrupt cause can be determined by reading the I/O register status (see Table 8.6). 1 1-msec/8-msec timer interrupt Interrupt sent from the cartridge I/F 2 The back panel of IC card is opened or overcurrent flows. 3 I/O address (R/W) : Register name P16H (R) : IOSTR bit0 :(PBUSY) status of interrupt sent from cartridge I/F =0: No interrupt =1: Interrupt occurred bit2 :(ICCLS) Interrupt caused by opened IC card back panel or by overcurrent =0: Interrupt occurred =1: No interrupt P23H (P) : ITSP bitO :(OVFINT) 1-maec/8-msec timer interrupt status =0: No interrupt =1: Interrupt occurred bit1 :(IPBUSY) Status of interrupt sent from IC card or cartridge I/F =0: Interrupt occurred =1: No interrupt Table 8.6 EXT Interrupt Status E.3.2 1-msec/8-msec timer interrupt OS uses a 8-msec timer interrupt for key input sound, BIOS BEEP, or barcode reader LED blinking. The user can use 1-msec/8-msec timer interrupts as explained below. (1) Enabling 1-msec/8-msec interrupts BIOS CALLX is used to call system jump table ENINTVL (002EH). The following entry parameters are used: Register B: 1 msec/8 msec specification 8=00H: 8 msec is specified. B=10H: ] msec is specified (see note 1). Register C: Specification of module that uses the timer interrupt.

The user must specify 40H.

8.8 EXT Interrupts

(Note 1) If I msec is specified and the barcode reader is used, LED blinking is quickened.

(2) Disabling 1-msec/8-msec interrupts

BIOS CALLX is used to call system jump table DISINTVL (0031H) (see note 2). The following entry parameters are used: Register C: Code of the used module The user must specify 40H.

If 1-msec interrupt is used, RZCTLR3 (FODEH) bit 4 must be turned to 0 after DISINTVL is called.

(Note 2) Even if DISINTVL is called while the key input sound is generated or the barcode reader is used, the interrupts are not actually disabled and the system only checks that the user operation is completed.

(3) 1-msec/8-msec timer

2-byte timer TIMER1M (F032H to F033H) is provided for 1-msec/8-msec interrupt processing and this timer can be referenced freely by the user. However, TIMER1M cannot be rewritten by the user because it is also used by the system. TIMER1M is 1-msec timer and can count up to 65,536 msec.

Address : Variable name (Byte length)

F032H : TIMER1': (2) 1-msec/8-msec counter. Unit is 1 msec.

F098H : INTVLFG (1) Use status of 1-msec/8-msec interrupts



(Note "APPENDIX 9 SAMPLE 30" shows the way to extend the interrupts of EXT.)

8.8.3 Interrupt sent from the cartridge

See the following Sections for details on interrupts sent from the cartridge:

- 1 4.3 Printer
- 2 7.3 Cartridge Interface
- 3 11.4 Extending Cartridge Device

It is needed to prohibit the interrupts from the cartridge I/O in the standard execution of the system. See "11.4.3" for details.
### 8.8.4 Interrupt sent from IC card

An interrupt is sent from the IC card when one of the following events occurs:

- 1 The back panel of IC card is opened (see note 1).
- 2 Overcurrent flowed to IC card.

The cause of an interrupt sent from IC card cannot be determined by software. Also, an interrupt sent from IC card cannot be reset by software.

OS forcibly closes the IC card when an interrupt is sent from IC card. Check the following when an interrupt is sent from IC card:

- 1 Close the back panel.
- 2 Check that the IC card is not broken.

8.9 Extending Interrupt Processing

8.9.1 Overview

The following three ways are provided to extend interrupt processing:

- The interrupt hook is used.
- 2 The interrupt vector is rewritten.
- 3 Interrupt occurrence status is checked.

8.9.2 Extension by using interrupt hook

The following four interrupt processing hooks are provided:

- 1 HK8251 (ART interrupt hook)
- 2 ICFHOOK (ICF interrupt hook) 3 OVFHOOK (OVF interrupt hook)
- 4 EXTHOOK (EXT interrupt hook)

See "Section 10.2 System Huck" for details on each hook.

8.9.3 Extension by rewriting interrupt vector

Interrupt processing created by the user can be executed by rewriting the contents of interrupt jump vectors (FFFOH to FFFFH). Note the following for rewrite operation:

1 Rewrite operation must be done in interrupt suppress status. (DI status).

2 The new interrupt processing must be set in the resident section (E000H to FFFFH) (in other words, the contents of rewritten jump vector must indicate E000H to FFFFH). If required at the jump destination, the bank is switched and processing routine is called again.

3 Interrupt processing must reserve its stack area.

4 Interrupt processing cannot use BDOS or BIOS.

5 The register contents held at interrupt processing start must be recovered after interrupt processing is ended (in other words, all registers must be saved).

8.9.4 Checking the interrupt occurrence status

(1) Overview

In this method, interrupt processing is not directly executed but whether interrupt has occurred is checked often and extended processing is executed if an interrupt has occurred. Whether an interrupt has occurred is determined in the following two wavs:

Interrupt occurrence flag INTTYPE (F093H) is checked. 1 2 Interrupt status register [ISR: PO4H] is checked.

7508 and EXT interrupts have two or more interrupt causes. Whether a 7508 or EXT interrupt has occurred can also be determined.

(2) Determination by INTTYPE

The user program must set OOH in INTTYPE in advance to check the interrupt status by using INTTYPE (F093H).

SOFTWARE Page 8 - 26

Each INTTYPE bit has the following meaning:



The corresponding bit is turned to 1 when an interrupt occurs.

(3) Determination by ISR

See "PART 3 HARDWARE" for the meaning of each ISR [PO4H] bit. To check the interrupt status from ISR, interrupts must be suppressed by using IER [PO4H] because each ISR bits are reset during interrupt processing.

(4) Checking the cause of 7508 interrupt

The cause of 7508 interrupt can be found out from INTFG (F3C2H). The user must set 00H in INTFG in advance. The INTFG contents set for the first interrupt is erased if two different interrupts (for example, 1-second interrupt and key interrupt) occurs consecutively.



(Interrupt occurrence is indicated when a bit is 1.)

The causes of alarm, power off, and power failure interrupts can be also found out by another way. See "Section 8.4.4 Suppressing alarm and power off (power failure) interrupt processing".

(5) Checking the cause of EXT interrupt

The cause of EXT interrupt is checked as follows: 1 Interrupts to be checked are suppressed. 2 The interrupt status is checked. 3 Processing is executed if the checked interrupt status indicates interrupt occurrence.

See Table 8.4 and Table 8.6 for suppressing interrupts and checking interrupt status.

# CHAPTER 9 UTILIZING SYSTEM FUNCTIONS

# 9.1 Overview

This chapter explains how to use such systems functions as MENU, system menu, and power-on in application programs.

This chapter does not explain how to operate the above system functions. Refer to EHT-10/EHT-10/2 Operation Manual for how to operate these system functions.

#### 9.2 System Menu

- 9.2.1 Controlling system menu
- (1) Enabling/disabling system menu functions

Although DL/UL, DISK, and TEST functions are disabled during application program execution, these functions can be enabled by modifying the SYSMMOD (FGBBH) contents. However, special attention must be paid because normal operation may not be guaranteed when a specific system function is used under a specific condition. (\*1)



The SYSMMOD contents are reset when WBOOT or restart power-on is executed.

\*1 If DISK is used while the RAM disk is open, for example, subsequent processing becomes abnormal.

(2) Enabling/disabling CONFIG processing

Each CONFIG parameter function can be enabled or disabled by modifying the corresponding CONFMOD (F6BCH and F6BDH) contents.



The CONFMOD contents are reset when WBOOT or restart power-on is executed.

### 9.2.2 Setting CONFIG parameters

(1) Overview

The system parameters set by the CONFIG functions can also be set by the user program. This chapter explains how to set each system parameter.

(2) Self test ON/OFF

Self test ON/OFF can be set by updating the contents of the following system areas:

1 RAM sum check: RAMTFLG (F07BH) 2 RAM disk sum check: DSKTFLG (F07AH)

When OlH is set in both of the above areas, self test ON is set. When OOH is set in both of the above areas, self test OFF is set.

- (3) BASIC parameters
  - 1 Number of files that can be opened: BASICF (F07CH)
  - 2 Number of random file records: BASICR (FO7DH and FC7SH)
  - 3 Receive buffer size: BASICB (F07FH and F080H)

The values greater than the values that can be set by CONFIG can also be set in these parameters as long as the BASIC program can be initiated actually.

Note: Although each BASIC parameter can also be set by the BASIC program, each parameter is made effective after the BASIC program is reinitiated.

(4) Calculator display mode

RNDFIG (F1EOH) is used to set the decimal place up to which the decimal values are displayed. If FFH is set in RNDFIG, values are displayed in a floating-point representation.

(5) Setting date and time

Set the date and time in BIOS TIMDAT. (See Chapter 4.)

(6) Setting RAM disk size and user BIOS area

See Item "CHGRAMD" in Section 10.3 and Section 4.6.

(7) Setting DLL parameter

The DLL parameters can be set by modifying the contents of the seven bytes starting from TCAMPRM (FICDH).



1 Type 00H: Specifies No protocol Direct-C. 01H: Specifies No protocol Direct-B. 02H: Specifies Filink protocol. 03H: Specifies extended protocol.

2 Line OOH: RS-232C O3H: Cartridge SIO

All the above parameters except the type and line parameters are the same as those when open processing was performed by BIOS RSIOX. (See Chapter 4.)

(8) Setting RS-232C parameters (for the system)

The RS-232C parameters for the system can be set by modifying the contents of five bytes starting from SRSPAK (F014H). Each parameter is the same as that of BIOS RSIOX. (See Chapter 4.)

SRSPAK (F014H)	-4	Receive buffer address
	-2	Receive buffer size
	+0	Bit rate
	+1	Character length
	+2	Parity check
	+3	Stop bit
	+4	Special parameter

The low-order four bytes starting from SRSPAK contain the default values of the address and size of the receive buffer. The address and size of the receive buffer can be modified by updating these default values. If the default receive buffer address is not modified, the maximum buffer size that can be specified is 256 bytes.

(9) Specifying execution type

The execution type can be set by modifying the contents of the following system area:

Address : Variable name (Byte length)

```
F019H : EXECTYPE (1)
OOH: Automatic determination
O1H: Forcible selection
O2H: Forcible loading
```

When forcible selection is specified, the execution file name and parameters must be set in the area staring from EXESTRNG (F30DH).

Character string (Max 33 bytes)

Character string length (one-byte area)

- (10) Setting power-off time
  - 1 Automatic power-off time

ATSHUTOFF (FO2OH): This value must be set in units of minutes. ATSOTIME (FO21H and FO22H): This value must be set in units of seconds.

Be sure to set both ATSHUTOFF and ATSOTIME. If OOH is specified for ATSHUTOFF, automatic power-off is not performed.

2 Setting printer power save mode

Set a value in PRNPWTM (FO25H) in units of seconds. If OOH is set, the power save mode is not enabled.

3 Setting automatic backlight-off time (only for EHT-10/2B)

Set a value in ELOFTIME (FO23H and FO24H) in units of seconds. If OOH is set, the automatic backlight-off function is not enabled.

(11) Setting printer I/F

Set an I/O device code ir RIOBYTE (F415H) and bits 6 and 7 of address 3. See Section 2.6.5 for details.

(12) Specifying a character set for a specific country

Set the code corresponding to the specific country in YLDFLTC(F5F3H) and YLCOUNTRY (F5F4H).

Table 9.1 shows the correspondence between country names and codes.

Country name	Code
U. S. A. (ASCII)	OFH
France	OEH
Germany	ODH
U.K.	ОСН
Denmark	ОВН
Sweden	OAH
]taly	09H
Spain	08H
Japan	07H
Norway	Joh

Table 9.1 Country names and codes for character set adjustment

# 9.2.3 Specifying DL/UL drives

Although drive A (RAM disk) has been specified in DL/UL as the drive for which down-load or up-load operation is to be performed, another drive (disk) can be specified by modifying the DSKNUM (F1DFH) contents. The relationship between the values and drive names is as follows.

00H:	Drive	Α	(RAM disk)
018:	drive	B	(ROM socket)
02H:	Drive	С	(IC card)
03H:	drive	D	(Floppy disk)
04H:	drive	Ε	(Floppy disk)

The value that has been set remains valid until the system is reset. Because the DSKNUM contents are also referenced by DLL and system menu DISK, the corresponding drive for DLL and DISK is also changed. (The DISK format is not modified.)

### 9.3 Extending MENU processing

9.3.1 Changing/adding display drives

The current menu display drives can be changed or new menu display drives can be added by modifying the contents of five bytes starting from MDRV (FOB9H).

MDRV +0H 01H ---> Drive A (RAM disk) (FOB9H) +1H 02H ---> Drive B (ROM disk) +2H 03H ---> Drive C (IC card) ---> Terminator (This and subsequent ones are not +3H FFH disks.) +4H FFH

The following values can be specified: 01H: Drive A (RAM disk) 02H: drive B (ROM socket) 03H: Drive C (IC card) 04H: Drive D (floppy disk) 05H: Drive E (floppy disk)

Up to four d.ives can be specified, and FFH must be written at the end. The specified values remain valid until they are reset. If the MDRV contents are updated, the menu drives for which forcible selection has been specified in the CONFIG EXECTYPE parameter are also changed.

### 9.3.2 Modifying/adding file types

Although the types of files that can be displayed by the MENU function are COM and BAS, these file types can be changed or new file types can be added by updating the contents of 10 bytes starting from MFTYP (FOBEH). However, the files of types other than COM and BAS cannot be processed. (If the file type is not specified (space), COM file type is assumed to be specified.)



Up to three file types can be specified, and FFH must be written at the end.

The specified values remain valid until the system is reset.

If the MFTYP contents are updated, the menu drives for which forcible selection has been specified in the CONFIG EXECTYPE parameter are also changed.

#### 9.4 Extending Power-on Processing

In EHT-10/EHT-10/2, continue mode is always set during application program execution, and the application program need not take power-off into account.

If a device such as the printer unit that has been used since it was set at power off/on need be initialized, power-on processing must be performed by the application program.

There are the following two power-on processing procedures.

(1) Power-on processing by using the cartridge mount processing hook (CRGHOOK)

(2) Power-on processing through checking the power-on flag (SRSTMODE: F389H)

For procedure (1), see Section 11.4. CRGHOOK can perform power- on processing required for the application program regardless of the cartridge device type. SRSTMODE (F389H) used in procedure (2) is a flag which is set to OIH at power on. Procedure (2) is as follows.

- 1 OOH is set in SRSTMODE at the beginning of the program.
- 2 The SRSTMODE contents are always checked and, if they are OIH, Step 3 is performed.
- 3 OOH is set in SRSTMODE.
- 4 The power-on processing routine created by the user is called.

# CHAPTER 10 SYSTEM HOOKS AND JUMP TABLES

10.1 Overview

The EHT-10 and EHT-10/2 systems have system hooks and jump tables so that the application program can extend system functions or use specific functions that the system has. There are 27 system hooks. These hooks are used in interrupt processing, extension and modification of BIOS functions, and other operations.

There are the following three jump tables, each used to jump to a specific system processing routine.

- 1 Resident area jump table 2 OS ROM jump table (I)
- 3 OS ROM jump table (II)

There are 31 system processing routines.

This chapter explains the structure and usage procedure of each jump table and the system hook control flow.

10.2 System Hooks

10.2.1 Overview

The EHT-10 and EHT-10/2 systems have 27 system hooks so that the application program can easily perform interrupt processing, extension of BlOS and serial communication functions, and other operations.

The user can enable the above processing simply by replacing the jump address of the hook with the address of the extend processing routine. This section explains the structure and usage procedure of each hook table and the system hook control flow. The functions of some hooks are also explained in detail in relevant chapters. Refer to these chapters also.

10.2.2 Structure and usage procedure of each hook table

(1) Hook table structure and processing flow

A total of 27 system hooks are configured into two tables shown in Figures 10.1 and 10.2. Figures 10.3 and 10.4 show control flow through each hook table.



SOFTWARE Page 10 - 2



Note:

The current bank is automatically switched according to the bank information, and the corresponding processing routine is called through system hook table (II).

Fig 10.2 Structure of system hook table (II)







Fig 10.4 Control flow through hook table (II)

(2) Hook types

System hooks can be classified to the following three types according to the usage procedure.

1 System hooks that jump to the hook processing routines under the system bank condition. The 10 hooks listed below belong to this system hook group. The hook processing routines for these hooks must reside in RAM address 8000H or later.

ALMHKO, ALMHK1, ALMHK2, ALMHK3, ALMHK4, HK8251, TMDT83, TMDT85, TMDT86, and BIOSHK

2 System hooks that jump to the hook processing routines after switching the current bank to all-RAM bank (bank 0#0) The four hooks listed below belong to this system hook group. The hook processing routines for these hooks must reside in RAM.

ICFHOOK, OVFHOOK, EXTHOOK, and BASHOOK

3 System hooks that jump to the hook processing routines after switching the current bank to the user-specified bank All hooks in hook table (II) belong to this system hook group. The user can specify the bank and address of the hook processing routine. The operating system switches the current bank to the bank specified by the user and then jumps to the specified address.

(3) Updating hooks

A hook can be updated after being set to DI status. Also, the hook processing routine must be loaded before the hook is updated.

1 To update a hook registered in hook table (I), set the execution start address of the corresponding hook processing routine in the two-byte area starting from the address obtained by adding 1 to the entry address of the nook.

СЗН	JP instruction
Hook processing — address —	This address must be set by the user.

2 To update a hook registered in hook table (II), set bank information and hook processing address (low-order byte and high-order byte) sequentially in this order starting from the entry address of the hook.

Bank	information	
Hook — addı	processing ress –	

These information must be set by the user.

(4) Notes on creating hook processing routine

Note the following when creating a hook processing routine.

1 Saving registers

Because only the minimum number of required registers are saved for each hook processing, all registers to be used must be saved. Especially, if all registers to be used are not saved before an interrupt hook is executed, the program may go into an abnormal loop when control is returned from the interrupt routine.

2 Saving interrupt status

Some interrupt hook routines are performing processing in DJ status, and multi-interrupt processing is not taken into account for some of these interrupt hook routines. Therefore, the interrupt disabled status for each hook must be saved. The interrupt disabled status for each hook is as follows.

ALMHKO: DI status ALMHK1: DI status ALMHK2: EI status, interrupt by 7508 CPU disabled ALMHK3: EI status, interrupt by 7508 CPU disabled ALMHK4: EI status HK8251: DI status ICFHOOK: DI status OVFHOOK: DI status EXTHOOK: DI status

3 Hook processing is usually terminated by an RET instruction.

(5) Miscellaneous

1 Hook processing routines can be placed in the user BIOS area. In this case, the routine must conform to the user BIOS area usage procedure. See Section 4.6 for the user BIOS area usage procedure.

2 Hook processing can be invalidated by the following:

For a hook in hook table (I): Set jump address F000H. (\*1)
 For a hook in hook table (II): Set bank information FFH (system bank) and jump address F000H.

The hook table is initialized when the system is reset or initialized.

\*1 Because an RET instruction has been stored in address FOOOH, the hook returns control without performing any processing when it is called.

10.2.3 System hooks registered in system hook table (1)

This chapter explains how to use 14 system hooks registered in system hook table (1).

(1) Alarm hooks

EHT-10 and EHT-10/2 each has five hooks for alarm processing. These hooks are used for such operations as automatic updating of alarm/wake data. These hooks can be called at the following point in time:

mese nooks can be carried at the forlowing point in time.

- ALMHKO: Immediately before the alarm screen is displayed during the processing after an alarm occurred in the power-off status
- ALMHK1: Immediately after the alarm screen is erased during the processing after an alarm occurred in the power-off status
- ALMHK2: Immediately before the alarm screen is displayed during the processing after an alarm occurred in the power-on status
- ALMHK3: Immediately after the alarm screen is erased during the processing after an alarm occurred in the power-on status
- ALMHK4: Immediately before the sequence escapes from the alarm screen during the processing after an alarm occurred

Figures 10.5 to 10.7 show control flow among these hooks.

For example, if an alarm or wake occurred when ALMHKO and ALMHK2 are used during updating alarm/wake data, the next alarm or wake time is set by using these hooks.



Fig. 10.5 Alarm processing in power-off status



Interrupt postprocessing

\*1 If the displaying of the alarm screen has been disabled, alarm screen display processing is skipped. The processing after (\*2) are OS alarm interrupt processing

Fig. 10.6 Alarm processing in power-on status



Fig 10.7 Alarm screen display processing

(2) Interrupt hooks

EHT-10 and EHT-10/2 each has four hooks for interrupt processing. These hooks can be called at the following point in time.

HK8251: Before data is received during ART interrupt processing ICFHOOK: During ICF interrupt processing OVFHOOK: Immediately before blink processing during OVF interrupt processing EXTHOOK: During EXT interrupt processing

SOFTWARE Page 10 - 10

#### 1 HK8251

HK8251 is stored in OS ROM so that it can extend ART interrupt processing.

Figure 10.8 shows where HK8251 is used in ART interrupt processing. As indicated in the figure, no processing is performed and control is returned even if an interrupt occurs before BIOS serial OPEN is executed. This should be noted. See "APPENDIX 9 SAMPLE 29".



(Processing in resident area)

(Processing in OS ROM)

Fig 10.8 ART interrupt processing

(\*1) This step is performed to check whether serial open processing has been done by BIOS RSIOX, ICCARD, or TCAM CONNECT.

(\*2) This step is performed to determine whether a send suspend request is to be issued for the send destination when a buffer control such as XON/XOFF, RTS/CTS, or DTR/DSR has been specified. 2 ICF, OVF, and EXT interrupt hooks

Before an ICF, OVF, or EXT interrupt hook is executed, the current bank is switched to bank O (RAM) and, when hook execution is terminated, the bank is switched again to the original bank. Figure 10.9 shows when each hook is called in interrupt processing. Because the size of the stack area allocated for each interrupt processing is restricted to the minimum, a new stack area size must be set when extending the interrupt processing. There are three causes for EXT interrupts: 1 ms or 8 ms timer, cartridge I/F, and IC card I/F. EXTHOOK must check which of the above three is the cause of the current interrupt, and perform extend

processing only for the determined interrupt cause. See Section 8.8 for details.



(3) BIOS TIMDAT hooks

EHT-10 and EHT-10/2 each has the following three hooks for BIOS TIMDAT extend processing.

TMDT83: For BIOS TIMDAT when C=83H TMDT85: For BIOS TIMDAT when C=85H TMDT86: For BIOS TIMDAT when C=86H Only the DE register contents are saved when BIOS TIMDAT is called.



(4) In addition to user BIOS functions, EHT-10 and EHT-10/2 each has a BIOS hook for extending each BIOS function. This BIOS hook enables conventional BIOS processing to be modified arbitrarily. How to use this BIOS hook is explained below. See "APPENDIX 9 SAMPLE31".

1 BIOS hook calling point in time

The BIOS hook is called through the hook table allocated in OS ROM in the system bank immediately before the sequence jumps to a specific BIOS function. Figure 10.11 is a system processing flowchart in which the BIOS hook is executed.

If the BIOS function is called by BDOS, only Steps (3) to (8) are executed as BIOS processing.



Fig 10.11 BIOS processing flowchart

Step : Explanation

(1) Switching the current stack pointer to the BIOS stack pointer The current stack pointer value is saved in USRSBI. The current stack pointer is switched to the BIOS stack pointer.

SOFTWARE Page 10 - 16

- (2) Switching the current bank to the system bank The BIOS function number is checked. The DMA address is translated to the system DMA address. The current IO byte is checked and its contents are set in RIOBYTE. The current bank is switched to the system bank and the original bank information is saved.
- (3) Calling jump table in OS ROM The jump table stored in OS ROM is called according to the BIOS function number.
- (4) Calculating the BIOS execution start address The BIOS execution start address is calculated according to the address of the called jump table.
- (5) Executing PREBIOS PREBIOS (see Section 4.1.2) is executed.
- (6) Executing BIOS hook BIOS hook (FFE7H) is called. The register contents remain unchanged since BIOS was called.
- (7) Executing a specific BIOS function A EIOS processing routine is called according to the BIOS execution scart address calculated in Step (4).
- (8) Executing PSTBIOS PSTBIOS (see Section 4.1.2) is executed.
- (y) Switching the current bank to the original one The current bank is switched to the original bank according to the bank information saved in Step (2). For a read function, the DMA data is copied.
- (10) Switching the current stack pointer to the original one The stack pointer value saved in Step 1 is restored.

The following table lists the system areas used in BIOS common processing. Address : Variable name (Number of bytes) F415H : RIOBYTE (1) 10 byte storage area. See Section 2.6.5 for the configuration of this area. F418H : OLDBNK (1) Bank information save area FFH: System bank 00H: Bank 0 01H: Bank 1 02H: Bank 2 F424H : USRSBJ (2) BIOS user stack pointer save area

```
F426H : BIOSFN (1)
BIOS function number storage area
OOH: BOOT
O3H: WBOOT
A5H: INFORM
F430H : SAVEIX (2)
IX register save area
F432H : SAVEIY (2)
IY register save area
```

2 Logic of the hook processing routine

The BIOS hook is always called when any BIOS function is called. When the BIOS hook is called, the user must determine whether the called BIOS function is the one to be extended.



Fig 10.12 Example of BIOS hook processing

SOFTWARE Page 10 - 18

Step : Explanation

- (1) Switching the stack pointer Because the system BIOS stack pointer is currently used, a new stack area must be allocated when the processing is to use much stack area is to be used in the processing.
- (2) Saving register contents Because the parameters corresponding to each BIOS function have been set in the current registers, the current register contents must be saved when the registers are to be used for a new BIOS function.
- (3) Calculating BIOS function number Calculate the BIOS function number from the BIOS execution start address and the BIOS jump table top address.



The difference of (A) - (B) is the offset value from BOOT(OOH, O3H, ...).

(4) Determining whether the BIOS function is to be extended The function number of the called BIOS function is compared with the function number obtained in Step (3) to determine whether the called BIOS function is the one to be extended. If the following condition is satisfied when the function number of the called BIOS function is n, it is determined that the object BIOS function has been called:

(A) - (B) = 3 x n n = 00H ... BOOT = 01H ... WBOOT = 02H ... CONST = 37H ... INFORM

- (5) BIOS extend processing BIOS extend processing is performed. The user inserts BIOS extend processing here.
- (6) Restoring saved register contents The register contents saved in Step (2) are restored.
- (7) Switching the stack pointer to the original one and returning control The stack pointer value saved in Step (1) is restored. Control is returned, following which subsequent BIOS processing is performed by OS. To suppress BIOS processing by OS, control is returned after poped two times (4 bytes). In this case, the sequence skips Step (7) and directly goes to Step (8) in the BIOS processing flowchart shown in Figure 10.11.

3 Updating BIOS hook jump address

The jump address of the BIOS hook can be updated.

FFE8H	BIOS	hook		(L)
FFE9H	—jump	address	-	(H)

The initial value is FOOOH. RET has been stored in address FOOOH. By replacing the contents of addresses FFE8H and FFE9H with the top address of the hook processing routine, the hook processing routine is executed after the BIOS function is called.

4 Notes on using BIOS hook

(a) Because the system bank is used when control is passed to the BIOS hook, the hook processing routine must reside in address 8000H or later. Reserve a user BIOS area and process the hook processing routine in the area.

(b) The sequence of each BIOS function used in BDOS also goes through this BIOS hook.

(c) The hook processing routine cannot call BIOS and BDOS (RBIOS), RIBIS2, RBDOS1, and RBDOS2) that reside in RAM. To use BIOS, BIOS that reside in OS ROM must be directly called. (BDOS cannot be used.)

(d) To pass return information to IX and IY, the IX and IY values must be set in SAVEIX (F430H) and SAVEIY (F432H), respectively.

(5) BASIC activation hook

BASHOOK has been provided as a hook for BASIC activation. See "PART 3 BASIC" for details.

10.2.4 System hooks registered in system hook table (II)

The 13 system hooks registered in system hook table (II) are explained in detail in relevant sections as shown below.

```
      TEXHK1
      Section 11.2 Communication Protocol Expansion

      DLHK1
      Section 11.2 Communication Protocol Expansion

      CRGHK1
      Section 11.4 Cartridge Device Expansion

      ICDHK1
      Section 11.4 Cartridge Device Expansion

      ICDHK2
      Section 11.3 IC card Protocol Expansion

      ICDHK5
      Section 11.3 IC card Protocol Expansion

      ICDHK6
      Section 11.3 IC card Protocol Expansion
```

10.3 System Jump Tables

10.3.1 Overview

EHT-10 and EHT-10/2 each has the following two types of system jump tables.

1 Resident area jump table

This jump table is used to jump to system routines related to the banks relocated in the resident area.

2 OS ROM jump table

This jump table is used to jump to system routines that reside in OS ROM. The OS ROM jump table is further classified to the following two types:

1 System bank jump table
2 Bank 2 (subbank 2#1) jump table

10.3.2 Jump table configuration

(1) Configuration of resident area jump table

The resident area jump table resides in addresses FF90H to FFBFH in the resident RAM, and can be accessed from any bank through a CALL instruction. Figure 10.13 shows the configuration of the resident area jump table. The detailed function of each system routine called through this jump table is explained in Section 10.3.3.

Address		
FF90H	JRBDOS	Entry point of RBDOS2
FF93H	JPREBIOS	PREBIOS processing
FF96H	JPSTBIOS	— PSTBIOS processing
FF99H	JSCALLX	Almost same as BIOS CALLX
FF9CH	JSELBNK	Bank switch routine
FF9FH	JLDAXX	Almost same as BIOS LOADX
FFA2H	XXATZL	—— Almost same as BIOS STORX
FFA5H	JLDIRXX	Almost same as BIOS LDIRX
FFA8H	JJUMPXX	Almost same as BIOS JUMPX
FFABH	JCALLXX	Almost same as BlOS CALLX
FFAEH	JINTCPR	—— Almost came as CIOS MASKI
FERIN	J SOBGIOX	Debugger control routine (for the system)
FFB4H	JHDBG10X	— Debugger control routine (for the user)
	Reserved	
FFCOH	L	

Fig 10.13 Resident area jump table configuration

(2) Configuration of OS ROM jump table (1)

OS ROM jump table (I) is allocated staring from the head of OS ROM (system bank address 0000H). Figure 10.14 shows its structure. The current bank must be switched to the system bank when calling a routine through OS ROM jump table (I).

1 BIOS CALLX is used when the application program calls a system routine.

2 JCALLX or JSCALLX of the resident area jump table is used when calling a system routine during interrupt extend or BIOS extend processing.

The detailed function of each system routine called through this jump table is explained in Section 10.3.4.

Address

0000H	(NOP)	
0001H	(STARTER)	Address O start
0004H	BDOSTABL	— BDOS function table address
0007H	BIOSTABL	— BIOS function table address
000AH	SETERR	Setting BDOS special error code
000DH	RSTERR	Setting BDOS ordinary error code
0010H	GOBACK	— BDOS error recovery processing
0013H	BIOSJTLD	— Relocate processing of RBIOS1, RBIOS2
0016H	SETRAMAD	— Setting load address of RBDOS1, RBIOS1
0019H	CRGRAMD	—— Modifying RAM disk size
001CH	FMTRAMD	Formatting RAM disk
001FH	EPSPSND	Sending under EPSP protocol
0022H	EPSPRCV	— Receiving under EPSP protocol
0025H	MELODY	Melody
0028H	WRT7508	—— Outputting a command to 7508 CPU (only for a non-necessary command)
0 <b>02</b> BH	CMD7508	- Outputting a command to 7508 CPU (only for a
002EH	EINTVL	— Enabling 1 mS / 8 mS timer interrupt
0031H	DISINTVL	— Disabling 1 mS /8 mS timer interrupt

Fig 10-14 Configuration of OS ROM jump table (I)
(3) Configuration of OS ROM jump table (II)

OS ROM jump table (II) is allocated in an OS ROM area starting from address 8000H (address 6000H of bank 2#1). Figure 10.15 shows its structure. The current bank must be switched to bank 2#1 when calling a routine through OS ROM jump table (II).

1 BIOS CALLX is used when the application program calls a system routine.

2 JCALLX or JSCALLX of the resident area jump table is used when calling a system routine during interrupt extend or BIOS extend processing.

The detailed function of each system routine called through this jump table is explained in Section 10.3.5.

6000H	NOP		
6001H	Checking IC card mounting		
6004H	Formatting IC card		

Fig 10.15 Configuration of OS ROM jump table (11)

10.3.3 Resident area jump table

13 system routines have been registered in the resident area jump table. This section explains the function and parameters of each system routine called through the resident area jump table.

JSELBNK, JKDAXX, JSTAXX, JJUMPXX, JCALLXX, JSCALLX, and JINTOPR are used to control banks and interruption without using BIOS such as in interrupt extend or BIOS extend processing.

[Function]
 JRBDOS indicates the entry address of RBDOS2.
[Entry parameters]
 Same as those of each BDOS function

[Return parameters] Same as those of each BDOS function

[Explanation]

(1) In EHT-10 and EHT-10/2, there are two BDOS entries at different locations in RAM. This is because:

1 The upperlimit of the usable RAM memory range can be indicated so that an ordinary CP/M application program can run without being modified.

2 ROM-based program can use BDOS without taking bank switching into account.

- (2) JRBDOS indicates the entry address of RBDOS2 that resides in the system common area.
- (3) A ROM-based program calls JRBDOS (FF90H) instead of calling address 0005H that must be called when using conventional BDOS.
- (4) See APPENDIX 7 and Chapter 5 for details on each BDOS function.

[Function] JPREBIOS sets the BIOS execution flag on, disabling subsequent operations such as alarm and power-off operations. JPSTBIOS performs operations that occurred after JPREBIOS execution such as alarm and power-off operations, and sets the BIOS execution flag off. [Entry parameters] None [Return parameters] None

- (1) The contents of all registers are saved.
- (2) PRSBIOS is used to disable interruption to be caused by such an event as power-off or alarm during BIOS processing by OS. PSTBIOS is used to release the interrupt disabled status.
- (3) JPREBIOS and JPSTBIUS are routines enhanced from PREBIOS and PSTBIOS, respectively, so that they can be used by application programs.
- (4) The application program uses JPREBIOS and JPSTBIOS under the following conditions:
  - 1 The power must not be turned off during program execution.
  - 2 The alarm screen must not be displayed during program execution.
  - 3 Power failure must be suppressed during program execution.
  - 4 BIOS functions that reside in OS ROM must be directly used.
- (5) Because JPREBIOS and JPSTBIOS performs processing after internally switching the current stack pointer to the BIOS stack pointer, these routines cannot be used when the BIOS stack pointer has already been used.
- (6) If JPREBIOS is executed, JPSTBIOS must be executed later. If JPSTBIOS is not executed after JPREBIOS, power-off and alarm operations are not performed.
- (7) If a BIOS function is used after JPREBIOS execution, PSTBIOS is automatically executed at the end of the BIOS processing. After that, alarm and power-off are immediately processed when they occurred. To use a BIOS function after JPREBIOS execution, the BIOS function that resides in OS ROM must be directly called.
- (8) If the processing time between JPREBIOS and JPSTBIOS becomes too long, the processing is separated. This is because, if power-off operation is not performed within 50 seconds after a power failure occurred, the mainframe power is forcibly turned off.

Figure 10.16 shows processing flow of PREBIOS and PSTBIOS.



Figure 10.17 shows the processing flow of JPREBIOS and JPSTBIOS.



Fig 10.17 Processing flow of JPREBIOS and JPSTBIOS

```
[Function]
    JSCALLX calls a subroutine that resides in the specified bank.
[Entry parameters]
    DISBNK (F41BH) = Call destination bank information
    IX = Call destination address
[Return parameters]
    None. (This depends on the return parameters of the called
    subroutine.)
```

(1) This routine has the same function as that of BIOS CALLX except the following points:

1 SCALLX can also be used by the routine called by BIOS CALLX or JCALLXX.

2 SCALLX can be executed without switching the stack pointer.

- (2) The contents of all parameters set by the user are saved when this routine is called and restored when control is returned to the calling program.
- (3) Because the stack pointer area that has been used when this routine is called is used as it is, the stack pointer area must be allocated in a location that can be referenced even if the current bank is switched to the bank where the calling routine resides.
- (4) This routine enters EI status after returning control.

```
[Function]
	JSELBNK switches the current bank to the specified bank.
[Entry parameters]
	C = Information about the bank to replace
[Return parameters]
	C = Information about the bank to be replaced
```

- (1) The contents of all registers except for C are saved.
- (2) The bank information is the same as that of BIOS MEMORY.
- (3) Parameter errors are not checked.
- (4) Because the stack pointer area that has been used when this routine is called is used as it is, the stack pointer area must be allocated in a location that can be referenced even if the current bank is switched to another bank.
- (5) This routine enters EI status after returning control.

```
[Function]
	JLDAXX reads one byte from the specified address in the specified bank.
[Entry parameters]
	C = Information about the bank containing the data to be read
	HL = Address of the data to be read
[Return parameters]
	A = Read data
```

- (1) The contents of all registers except for A are saved.
- (2) The function and parameters of this routine are the same as those of BIOS LOADX except that this routine can be executed without switching the stack pointer.
- (3) Parameter errors are not checked.
- (1) Because the stack pointer area that has been used when this routine is called is used as it is, the stack pointer area must be allocated in a location that can be referenced even if the current bank is switched to the bank from which data is to be read.
- (5) This routine enters El status after returning control.

[Function] JSTAXX writes one-byte data into the specified bank. [Entry parameters] A = Data to be written C = Information about the bank into which data is to be written HL = Write address [Return parameters] None

- (1) The contents of all registers are saved.
- (2) This routine has the same function as that of BIOS STORX except that this routine can be executed without switching the stack pointer.
- (3) Because the stack pointer area that has been used when this coutine is called is used as it is, the stack pointer area must be allocated in a location that can be referenced even if the current bank is switched to the bank into which data is to be written.
- (4) This routine enters EI status after returning control.

```
[Function]
JLDIRXX transfers the specified bank data to another bank.
[Entry parameters]
SRCBNK (F41DH) = Transfer source bank information
DISBNK (F41BH) = Transfer destination bank information
HL = Top address of the transfer data
DE = Transfer data storage start address
BC = Number of transfer data bytes
[Return parameters]
HL = HL + BC
DE = DE + BC
BC = 0000H
(Same as LDIR instruction)
```

- (1) This routine has the same function as that of BIOS LDIRX except that this routine can be executed without switching the stack pointer.
- (2) Parameter errors are not checked.
- (3) Because the stack pointer area that has been used when this routine is called is used as it is, the stack pointer area must be allocated in a location that can be referenced even if the current bank is switched to the bank from which data is to be copied.
- (4) This routine enters EI status after returning control.

```
[Function]
    JJUMPXX jumps to the specified bank.
[Entry parameters]
    DISBNK (F41BH) = Jump destination bank information
    IX = Jump destination address
[Return parameter]
    None
```

- (1) This routine has the same function as that of BIOS JUMPX except that this routine can be executed without switching the stack pointer.
- (2) Control is passed to the jump destination routine with all register contents saved.
- (2) Error check is not performed.
- (4) Because the stack pointer area that has been used when this routine is called is used as it is, the stack pointer area must be allocated in a location that can be referenced even if the current bank is switched to the bank to which this routine is to jump.
- (5) This routine enters El status after returning control.

```
[Function]
    JCALLXX calls a subroutine that resides in the specified bank.
[Entry parameters]
    DISBNK (F41BH) = Call destination bank information
    IX = Call destination address
[Return parameters]
    None. (This depends on the return parameters of the called
    subroutine.)
```

- (1) This routine has the same function as that of BIOS CALLX except that this routine can be executed without switching the stack pointer.
- (2) The contents of all parameters set by the user are saved when this routine is called and restored when control is returned to the calling program.
- (3) Because the stack pointer area that has been used when this routine is called is used as it is, the stack pointer area must be allocated in a location that can be referenced even if the current bank is switched to the bank from which a subroutine is to be called.
- (4) This routine enters EI status after returning control.

```
[Function]
	JINTOPR sets or resets the interrupt mask.
[Entry parameters]
	B = Interrupt mask data (IER)
	C = 7508 CPU or EXT interrupt mask data
[Return parameters]
	BC = Old interrupt mask status
```

- (1) This routine has the same function as that of BIOS MASKI except that this routine can be executed without switching the stack pointer.
- (2) The stack pointer area must be allocated in address 8000H or later.

[Function]

JSDBGIOX and JRDBGIOX each supports communications with the development cartridge. JSDBGIOX and JRDBGIOX each has the following 10 functions according to the B register contents.

B = OOH: Opens the development cartridge (OPEN).

- = OlH: Checks the receive buffer status (INSTS).
- = 02H: Checks whether sending is possible (OUTST).
- = 03H: Receives one-byte data from the receive buffer (GET).
- = O4H: Sends one-byte data (PUT).
- = 05H: Switch the cartridge I/F status (SWITCH).
- = O6H: Resets the development cartridge (DRSINT).
- = OBH: Checks the development cartridge mount status (RDSTS).
- = 09H: Changes the user cartridge mode (MODECHG).
- = OAH: Initializes the receive buffer (INITBUF).

The details on each function are explained later.

(Explanation)

- JSUBGIOX and JRDBGIOX each controls the RS-232C port of the development cartridge.
- (2) The difference between the functions of JSDBGIOX and those of JRDBGIOX lies in that JSDBGIOX switches the current stack pointer to the BIOS stack pointer but JRDBGIOX uses the current stack pointer without switching it. Therefore, the stack pointer area must be allocated in RAM address 8000H or later when calling JRDBGIOX. (The stack pointer area must be allocated in address E000H or later when calling JSDBGIOX from the application ROM.)
- (3) Because neither JSDBGIOX nor JRDBGIOX has a close function, perform close processing in the following procedure.
  - 1 Set the current mode to ordinary mode by using the SWITCH function.
  - 2 Initialize the receive buffer by using the INITBUF function.

JSDBGIOX	(0051)	FFB1H
JRDBGIOX	(OPEN)	FF84H

- (1) This function performs the following operations:
  - 1 Sets the receive buffer.
  - 2 Changes the cartridge mode to debugger mode.
  - 3 Resets the debugger.
  - 4 Enables interruption from the cartridge I/F.
- (2) If the cartridge has already been opened, this function does not reset the debugger. However, the receive buffer is set again.
- (3) If the development cartridge has not been mounted, this function performs no operation.
- (4) The receive buffer must be allocated in RAM address 8000H or later. The receive buffer size is automatically set to 256 bytes.
- (5) This function sets the following flag according to the development cartridge mount status:

DBGFLG (5203H): OOH: Not mounted 01H: Mounted

JSDBGIOX	(INCTC)	FFB1H
JRDBGIOX	(11212)	FFB4H

```
[Function]
This function checks the receive buffer status.
[Entry parameters]
B = 01H: Function number
[Return parameters]
A = 00H: The receive buffer contains no data.
= FFH: The receive buffer contains data.
```

- (1) This function checks whether any data has been received in the receive buffer and sets the information in the A register.
- (2) Note that this function does not change the cartridge mode.

JSDBGIOX		FFB1H
JRDBGIOX	(ourst)	FFB4H

- (1) This function checks the send buffer status and set the status in the A register.
- (2) If the cartridge mode is not debugger mode, this function forcibly changes the mode to debugger mode and checks the send buffer status.

JSDBG10X		FFB1H
JRDBGIOX	(GET)	FFB4H

```
[Function]
This function fetches one-byte data from the receive buffer.
[Entry parameters]
B = 03H: Function number
[Return parameters]
A = Received data
```

- (1) If the receive buffer contains no data, the routine waits until data is received.
- (2) Because this function does not change the cartridge mode, be sure to check the current mode and, if the current mode is not debugger mode, change it to debugger mode. (See Item "SWITCH".)

JSDBGIOX	
	(PUT)
JRDBGIOX	

[Function] This function sends one-byte data.

```
[Entry parameters]
    B = 04H: Function number
```

```
[Return parameters]
None
```

- (1) If sending is not possible, this routine waits until sending is made possible.
- (2) If the cartridge mode is not debugger mode, this function changes the current mode to debugger mode and sends data.

IDDRCIOX	(SWITCH)	
J SDBG I OX	X	FFB1H

```
[Function]
This function switches the cartridge I/F status.
[Entry parameters]
B = 05H: Function number
A = 00H: Changes the current mode to debugger mode.
= 01H: Close debugger
= FFH: Changes the current mode to ordinary mode.
[Return parameters]
None
```

This function changes the current mode to debugger or ordinary mode.

To access the cartridge operating in ordinary mode after OPEN, OUTST, or FUT execution, set FFH in the A register and execute this function.

JSDBGIOX (DRSINT) JRDBGIOX	
[Function] This function resets the development cartridge.	
[Entry parameters] B = O6H: Function number	
[Return parameters] None	

FFB1H

FFB4H

- (1) This function resets the development cartridge.
- (2) This function must be executed after a communication error occurred.
- (3) Check that the current mode is debugger mode before executing this function.

JSDBGIOX		FFB1H
JRDBGIOX	(2120)	FFB4H

```
[Function]
This function checks the development cartridge mount status.
[Entry parameters]
B = 08H: Function number
[Return parameters]
CY = 1: Development cartridge mounted
= 0: Development cartridge not mounted
```

- (1) This function checks the development cartridge mount status.
- (2) This function is also effective even if the OPEN function has not been executed.

JSDBGIOX		FFB1H
JRDBGIOX	(MODECHG)	FFB4H

- (1) This function changes the mode of the cartridge device connected to the development cartridge.
- (2) This function is also effective even if the Jevelopment cartridge has not been mounted.
- (3) The mode values that can be specified in the A register are as follows.

A register = OOH: HS mode = O1H: IO mode = O2H: DB mode = O3H: OT mode

- (4) After this function is executed, the mode is changed to ordinary mode.
- (5) This function can be executed even if the cartridge has not been opened.

JSDBG IOX	(INITRUE)		FFB1H
JRDBGIOX		_	FFB4H

```
[Function]
   This function initializes the receive buffer.
[Entry parameters]
   B = OAH: Function number
```

[Return parameters] None

[Explanation]

This function actually resets the buffer pointer to the status set when the development cartridge was opened.

10.3.4 OS ROM jump table (I)

16 system routines have been registered in OS ROM jump table (I).

This section explains the specifications of each system routine registered in OS ROM jump table (I).

. .

0004H

[Function]

BDOSTABL indicates the top address of BDOS function on vector that resides in OS ROM.

- (1) This jump vector is used to directly reference a BDOS function that resides in OS ROM.
- (2) Because each BDOS function that resides in OS ROM neither changes the stack pointer nor performs error processing, special attention must be paid when using it.

BDOSTABL -> DW FUNCO +02H DW FUNC1 +04H DW FUNC2 +06H DW FUNC3 ī I 1 1 1 +50H DW FUNC40

[Function]

BIOSTABL indicates the top address of the BIOS jump table that resides in OS ROM.

- (1) This jump address is used to directly use a BIOS function that resides in OS ROM.
- (2) Because each BIOS function that resides in OS ROM neither changes the stack pointer nor executes PREBIOS and PSTBIOS, special attention must be paid when using it.

BIOSTABL -> JP BOOT +03H JP WBOOT +06H JP CONST +09H JP CONIN JP INFORM +A5H

```
None
```

- If this routine is called, this routine updates the BDOS error processing vector so that, even if a disk access error occurred, no error message will be displayed and return parameters will be passed.
- (2) This routine is used when a file access error is processed by an application program.
- (3) There are following types of BDOS errors.
  - 1 Bad Sector Data input/output for the disk is abnormal. 2 Bad Select
  - An unexisting drive or a drive being in unready state has been selected.
  - 3 R/O disk An attempt was made to write data to a read-only disk. 4 R/O file
    - An attempt was made to write data to a read-only file.
- (4) After SETERR execution, an error code corresponding to the BDOS error is set in registers A and H.

Register name	A	Н
Error type		
BDOS processing normal termination	BDOS return code	00H
Bad Sector	FFH	01H
Bad select	FFH	02H
R/O Disk	FFH	03H
R/O File	FFH	04H

(5) When OOH is set in the H register, a value corresponding to the CP/M return information is set in the A register. See Section 5.3 for the SETERR usage procedure.

- If this routine is called, this routine initializes the BDOS error processing vector and, if a disk error occurred after that, outputs a corresponding error message.
- (2) If WBOOT is executed, the BDOS error processing vector is also initialized.

## **GOBACK**

```
[Function]
    GOBACK performs BDOS termination processing.
[Entry parameters]
    None
[Return parameters]
```

```
None
```

- (1) GOBACK performs BDOS termination processing.
- (2) After a BDOS function is executed, this routine performs the following operations as termination processing:
  - 1 Restores the current disk contents.
  - 2 Sets the return information.

[Function]
BIOSJTLD generates the RBIOSI jump table.
[Entry parameters]
None
[Return parameters]
None

- (1) BIOSJTLD is used to regenerate the RBIOS1 jump table after the RAM disk or user BIOS area size is changed.
- (2) According to BIILAD (FOO7H), this routine generates the jump table to be linked to RBIOS2 (resident area).

[Function] SETRAMAD sets the top addresses of the user BIOS area, RBDOS1, and RBIOS1 according to the sizes of the RAM disk and user BIOS area. [Entry parameters] SIZRAM (FOOBH) = RAM disk standard RAM area size USERBIOS (FOOCH) = User BIOS area size [Return parameters] TOPRAM (FO5CH) = user BIOS area top address BIILAD (FOO7H) = RBIOS1 top address BDSLAD (FOO5H) = RBDOS1 top address

[Explanation]

SETRAMAD is used to set the top addresses of RBDOS1, RBIOS1, and user BIOS area after the RAM disk or user BIOS area size is changed.

[Related routines]

BIOSJTLD (0013H) CHGRAMD (0019H)

```
[Function]
	CHGRAMD modifies the RAM disk size.
[Entry parameters]
	A = function
	= 0: Modifies only the size.
	= 1: Modifies the size and formats the disk.
	= 2: Increase the size and relocate the disk contents.
	B = extended RAM size (in units of 32 Kbytes)
	C = new RAM disk size (in units of Kbytes)
[Return parameters]
	CY = error information
```

```
= 0: Normal termination
```

= 1: Error occurred

- This function is used to modify the size of the RAM disk or user BIOS area.
- (2) Although an arbitrary extended RAM size can be specified in units of 32 Kbytes in the B register, the size to be specified must not exceed the total size of the actually-mounted extended RAM. If the value specified in the B register is less than the actually-mounted extended RAM size, no access is made by OS for the extended RAM area that exceeds the B register value. Therefore, this area can be arbitrarily used by the user. The user can know the size of the actually-mounted extended RAM by referencing the contents of RAM SET (F069H) in the system area.
- (3) Relocating RAM disk contents When the RAM disk size is increased, the RAM disk contents can be relocated (without being destroyed) by calling CHGRAMD with A register = 2, thus the sizes of the areas allocated in the RAM disk can be modified. However, the extended RAM area size cannot be modified.
- (4) Modifying user BIOS area size CHGRAMD can also modify the user BIOS area size. See Section 4.6 for details.



Fig 10.18 Modifying RAM disk size

```
Step : Explanation
```

 Determining RAM disk size The user can know the current RAM disk size by referencing the following area contents.

SOFTWARE Page 10 - 58

- (2) Determining user BIOS area size The user can know the current user BIOS area size from the contents of system area USERBIOS (FOOCH). The unit is 256 bytes (one page).
- (3) Setting system RAM area Set the SIZRAM and USERBIOS values determined in Steps 1 and 2 and call SETRAMAD (0016H). In this case, the following condition must be satisfied:

(SIZRAM) - (USERBIOS) < 39.5 Kbytes (EHT-10) or 40.0 Kbytes (EHT-10/2)

 (4) Modifying RAM disk size Modify the RAM disk size according to the parameter values determined in Steps (1) and (2).

> C: New Q<sup>T</sup> RAM IN value B: New RAMD\_SIZE value A: OOH, O1H, or O2H

Call CHGRAMD by using these parameters.

- (5) Loading BIOS jump table Call BIOSJTLD (0013H).
- (6) Set RBDOS1 and RBIOS1 entry =ddresses. BDSLAU (F005H) = 6, = 7 ==> (0006H, 0007H) BIILAD (F007H) = 3, = 4 ==> (0001H, 0002H)

## FMTRAMD

[Function] FMTRAMD formats the RAM disk.

[Entry parameters] None

[Return parameters] None

[Explanation]

If this function is executed, the entire RAM disk is formatted, erasing all files contained.
```
[Function]
     EPSPSND sends EPSP data from SIO.
[Entry parameters]
    HL = EPSP send packet top address
    A = Send mode
         LSB = 0: Performs only send operation.
             = 1: Performs both send and receive operations.
[Return parameters]
    CY = return information
       = 0: Processing completed
       = 1: Processing suspended
    A = return code
      = 00H: Normal termination
      = 61H: Device unconnected
      = 62H: Communication error
      = 63H: Time over
      = 64H: Function key F6H
```

- (1) EPSPSND sends the packet data indicated by the HL parameter to a disk drive.
- (2) If both send and receive operations are specified in the A register, the data received from the disk drive is stored in the area starting from the address indicated by the HL parameter.
- (3) The packet data is transferred according to the EPSP (EPSON Serial Communication Protocol) protocol.
- (4) The packet format is shown below. See Section 6.6.4 for details on the function.

FMT	FMT: Head block format OOH: Indicates data sent from
DID	EHT-10/EHT-10/2
SID	OlH: Indicates data sent from the FDD
FNC	SID: ID of the destination device
SIZ	Device IDs are as follows:
Text data	EHT-10/EHT-10/2: 23H FDD (D: or E:): 31H

Therefore, the following is assumed: When data is sent from EHT-10/EHT-10/2 to the FDD DID=31H and SID=23H

- When data is sent from the FDD to EHT-10/EHT-10/2 DID=23H and SID=31H
- (5) Because the bank is not taken into account in send operation, the packet top address (HL register of the entry parameter) must be 8000H or later.

```
[Function]
EPSPRCV receives EPSP data from S10.
[Entry parameters]
HL = EPSP receive packet top address
[Return parameters]
A = return code
= 00H: Normal termination
= 61H: Device unconnected
= 62H: Communication error
= 63H: Time over
= 64H: Function key F6H
B = received packet information (effective when A=00H)
= 00H: Data with a header was received.
= 01H: Data without a header was received.
(HL) = received data
```

- (1) EPSPRCV stores the data received from the disk drive in the packet indicated by the HL parameter.
- (F) The receive packet format is the same as that of EPSPSND. See Section 6.6.4 for details.
- (3) Because the bank is not taken into account in receive operation, the packet top address (HL register of the entry parameter) must be 8000H or later.

## MELODY

```
[Function]
    MELODY sounds the specified melody.
[Entry parameters]
    HL = Melody table top address
    C = repeat count
```

```
[Return parameters]
None
```

- (1) MELODY sounds the specified melody the specified number of times.
- (2) The melody table configuration is as follows.



- (3) The melody table must be allocated in RAM address 8000H or later.
- (4) This routine runs using the BIOS BEEP function.

```
[Function]
WRI7508 sends a command to the slave 7508 CPU.
[Entry parameters]
C = command data to be sent to the slave 7508 CPU
[Return parameters]
None
```

- (1) The contents of all registers other than A are saved.
- (2) WRT7508 sends a command to the slave 7508 CPU.
- (3) See Section 7.6.3 for details on the commands for 7508 CPU.
- (4) Before sending a command to the 7508 CPU, the interrupt by the 7508 CPU must be disabled.
   BIOS MASKI is used to cortrol interruption

[Function] CMD7508 sends a command to the slave 7508 CPU and receives its response data. [Entry parameters] C = command data to be sent to the slave 7508 CPU [Return parameters] A = response data sent from the slave 7508 CPU

- (1) The contents of all registers other than A are saved.
- (2) CMD7508 sends the specified command to the slave 7508 CPU, receives one-byte response data, and sets the response data in the A register.
- (3) See Section 7.6.3 for details on the commands for 7508 CPU.
- (4) Before using CMD7508, the interrupt by the 7508 CPU must be disabled. BIOS MASKI is used to control interruption.
- (5) If multiple data items are returned from the 7508 CPU, the remaining data items must be read directly from the port.

```
[Function]
ENJNTVL enables 1 ms or 8 ms timer interrupt.
[Entry parameters]
B = Specifies 1 ms or 8 ms interrupt.
= 00H: 8 ms interrupt
= 01H: 1 ms interrupt
C = 04H: Indicates that the interrupt is to be used by the user.
[Return parameters]
None
```

See Section 8.8.2 for details.

```
[Function]
DISINTVL disables 1 ms and 8 ms timer interrupts.
[Entry parameters]
C = 04H: Indicates that using timer interrupt by the user is to be
terminated.
[Return parameters]
None
```

See Section 8.8.2 for details.

10.3.5 OS ROM jump table (II)

OS ROM jump table (II) is allocated starting from address 6000H in bank 2#1. The user can use only the following two routines through this table.

- (1) ICDKMNT (IC card mount processing)(2) ICDFMT (IC card formatting)

This section explains the specifications of each above routine.

## SOFTWARE Page 10 - 69

[Function] ICDKMNT checks whether an IC card has been mounted.							
[Entry parameters] None							
<pre>[Return parameters] CY = return information = 0: Normal termination = 1: Abnormal termination A = return code (effective when CY=1) = 01H: No IC card has been mounted. = 02H: An error was detected during communications with the IC card.</pre>							

- ICDKMNT checks whether an IC card has been mounted and, if an IC card has been mounted, checks its capacity and sets the information in DPB (Disk Parameter Block).
- (2) If ICDKMNT is called while power is being supplied to the IC card (open state), it returns control without performing any operation. If ICDKMNT is called while power is not being supplied to the IC card (close state), it supplies the power and the clock signal to the IC card and releases the reset status.

```
[Function]
    iCDFMT formats the IC card so that the IC card can be used as a CP/M
    disk.
[Entry parameters]
    None
[Return parameters]
    CY = return information
        = 0: Normal termination
        = 1: Abnormal termination
        A = return code (effective when CY=1)
        = 01H: No IC card has been mounted.
        = 02H: IC card formatting error
```

- (1) ICDFMT formats the IC card by writing E5H in all data area of the IC card.
- (2) ICDFMT is used by an application program for formatting an IC card. The operating system calls this routine through the IC card format processing displayed on the system menu screen.

CHAPTER 11 System Expansion

11.1 Overview

This chapter mainly describes I/O devices for system expansion.

The following I/O devices are covered in this chapter.

- Serial communication protocol
   IC card
   Cartridges

- 4. Bar code

Refer to the following chapters while reading this chapter for I/O device control and system expansion.

....

Chapter 7 I/O Interface overview
 Chapter 8 Interrupt processing
 Chapter 10 System Hook and Jump Table

11.2 Communication Protocol Expansion

11.2.1 Overview

EHT-10/EHT-10/2 supports communication procedures with protocol, but protocol may differ according to the host computer type. The OS is structured to use protocol for easy expansion. The communication procedure can be selected from the following three types.

(1) No sequence (Data is sent/received without specific protocol)

(2) Filink (Common communication procedure for EPSON computers)

(3) Expansion sequence (protocol for original application is added and data is sent/received)

These protocols are used in:

(1) System utility DLL, DL/UL

(2) TCAM of the BIOS

to expand the communication procedure, expansion must be performed at the two processing locations mentioned above. To operate the expansion procedure, select one of the following.

- (1) Modify the value of TCAMPRM (F1CDH) + 0 to 03H.
- (2) Select "Extend" of the CONFIG DLL item.

Then,

- (1) DLLHK1 call in DLL processing
- (2) ULHK1 call in UL processing
- (3) DLHK1 call in DL processing
- (4) TEXHK1 call in BIOS TCAM processing

are executed in the system by parameter modification as mentioned above.

The bank can be specified via these hooks, the application expands the communication protocol by adding an expanded section utilizing the hook necessary for the expansion procedure. (See "APPENDIX 9 SAMPLE 32")

11.2.2 Expansion Procedure

(1) Loading of expansion program

The program for protocol expansion can be placed; 1 In user BIOS area. 2 On application ROM. 3 In RAM (TPA).

See "10.2 System Hook" for each procedure and notes.

(2) Rewrite of Hook

Rewrite the contents of the hook corresponding to the process to be expanded. The communication protocol hook is configured of 3 bytes as follows.

Bank information (1 byte) + address information (2 bytes).

See "10.2 System Hook" for further information.

(3) Communication protocol expansion specifications

The communication protocol hook does not jump to the expansion process by only a rewrite of the hook. To start the expansion process, protocol expansion must be specified.

To specify expansion, select either of the following.

1 Modify the value of TCAMPRM+0 to 03H. 2 Set the protocol of the CONFIG DLL item to "Extend".

(Example)

TCAMPRM EQU OFICDH

- LD A,O3H } expansion specification code LD (TCAMPRM),A
- 11.2.3 Description of Hook
- (1) Overview

The following 4 types of communication protocol hook are available.

1 TEXHK1 (Expansion hook in BIOS TCAM) 2 DLLHK1 (Expansion hook in DLL process) 3 DLHK1 (Expansion hook in DL process) 4 ULHK1 (Expansion hook in UL process)

When using DLL or DL/UL process which is standard-support of the system, the hooks of (2) to (4) mentioned above must be used for expansion. When not used, expansion is not necessary, but correct operation of DLL and DL/UL is not guaranteed. (Interface may not match as BIOS TCAM is used in DLL and DL/UL.)

DLLHK1, DLHK1 and ULHK1 are types to call a specified hook address in the proper section of each process. The state of TEXHK1 stores the entry parameter of BIOS and uses it to jump to the specified hook address. (The processes after TEXHK1 are skipped.)



(2) TEXHK1

1 Overview

TEXHK1 is used to expand communication protocol. This hook is included in the BIOS TCAM process and the following advantages are created by using this hook to expand the communication protocol.

Communication with devices which have a different communication protocol is enabled.
 Protocol which expands the system utility (DLL,DL/UL) enables operation.
 Easy application development.

2 Hook location

Figure 11.1 shows the location of TEXHK1 during process.



Fig. 11.1 Flow of BIOS TCAM

3 Description

(a) When TEXHK1 is called, TCAM entry parameter is stored excluding the flag register.

(b) When TEXHK1 is returned and the error process at OS is required to be invalid, process as follows.

\* Request the location of the address returned from the hook by the value of the saved stack pointer.

\* Rewrite the contents of the return address to an address with only a return instruction.

(Ex.)

DEC HL and omit OS error check process	TSPSAVE RETADR	EQU EQU LD DEC LD DEC LD	OF6C3H OF000H HL,(TSPSAVE DE,RETADR HL (HL),D HL (HL),E	) 	Address with RET instruction. Rewrite the contents of the sta and omit OS error check process
--	-------------------	--	--	-------	---

(c) Input parameter at expansion process is as follows.

A register ... function specification A=01H : Connect =02I: : Send =03H : Receive =04H : Disconnect

Other registers are different depending on the function. Further information is available in the item of each function.

(Note) When the addition of a function is necessary, add by the value of A register.

(d) Align return data of the expansion routine with the return data of the system. The system return data is as follows.

CY-Flag ... Indicates normal/abnormal end. CY=0 : Normal end CY=1 : Abnormal end A register ... Indicates error data at abnormal end. A=01H : Parameter error =02H : Open =03H : Not open =04H : Forced end =05H : Received buffer overflow =06H : Timeout =07H : Protocol error =08H : Communication error

Other registers are different depending on the function, details are described later.

(Note) When the system returns by CY=1 at expansion, A register=08H is force set. Thus, when error data is required to remain in the application, the process of previous item (b) must be added.

(e) "Connect" function process

When entry becomes "A register=O1H" for expansion routine, the loop is connected.

The expansion process uses the following data depending on requirements. (Details of each area is described in the work area item.)

condition		TCAMPRM(F1CDH -	F1D3H)
		TDFLTCNT(F1D4H)	
		TISTTIME(F1D5H -	- F1D6H)
		T2NDTIME(F1D7H -	- F1D8H)
	condition	condition	condition TCAMPRM(F1CDH - TDFLTCNT(F1D4H) T1STTIME(F1D5H - T2NDTIME(F1D7H -

The following items are to be processed when expansion is added.

<Parameter Check>

Send and receive process of Filink are different, the process is specified by the B register.

B register=OOH : Send process =01H : Receive process When a distinction of send/receive data is required in the expansion process, refer to this data. <Open Check> To check whether it is open, if open, an error occurs. See RSODEV to check whether it is open. If open, set value to RSODEV =00H : RS-232C RSODEV (F623H) =03H : Cartridge SIO =FFH : Not open <Communication Condition Set> To set the bit rate, bit length, parity, stop bit and other status of the control line. See "7.2 Serial Interface" for setting the communication condition. (Note) Required processes for application is as follows. 1 Set value of 5 bytes of RS2CTLR1 (F19BH) to RS2SOUT (F19FH) 2. Set OOH to RSODEV (F62CH) 3. Set OOH to SPMODE (F241H) 4. Call serial line switching process (SELSER) (entry parameter A=01H) <Communication Interrupt Process> The OS contains a standard receive interrupt process, but is used for BIOS RSIOX and cannot be used for application. Therefore, when receive interrupt for application is used, the interrupt vector must be rewritten or an interrupt expansion must be performed using the interrupt hook. Interrupts used in applications must be enabled interrupts. See "8.5 ART Interrupt" for interrupt process. (Note) When a timer is necessary in communication process, 1 ms or 8 ms, 100 ms ,1 s are enabled (See "Chapter 8 Interrupt Process" for further information.) <Connection With Host> Loop connection with a host computer must be performed as a connect process. Due to this, the expansion process does not only set serial communication, but also performs linkage with the host computer. The utility (DLL,DL/UL) of the system mainly sends/receives data, a filename process is necessary for the first send/receive data. With the receive process, received filenames that follow must be set to TCAMAREA. With the send process, filename data is stored in TCAMAREA (FICHDH), and data following is received. However, when operation of a utility is not necessary or filenames are not of concern, proper operation of data in TCAMAREA is only necessary.

(Note) When extension (9th to 11th byte of TCAMAREA) is "COM", DLL starts load to TPA, when "BAS", DLL starts as a BASIC program, other than these, DLL is stored in RAM disk by specified filename. DL is stored in RAM disk by filename specified in TCAMAREA. See MODEFG (FOOAH) to distinguish whether DLL, DL/UL is being processed. (f) "Send" function process When entry is "A register=02H" in expansion routine, data send process is performed. Input parameter at "Send": BC register ... Number of bytes to be sent. HL register ... Leading address of send data. Expansion process sends data for the number of BC bytes from HL register. <Open Check> To check whether it has been opened, if not, an error  $(\Lambda=0.3H, CY=1)$ occurs. Serial Variance> When IC cards are used at the same time, the serial line switches. Due to this, the serial line must be switched to the mode in present use, before "Send" process in expansion process. (See "7.2 Serial Interface") However, if other serials are not used between "Connect" and "Disconnect", this is not necessary. <Send Data Process> To send data for BC bytes from address indicated by HL to the host computer. At this time, data to be sent is assumed to be always in RAM. The number of bytes to be sent by the system is always 128 bytes. (Note) Since the location of send data may enter the end bank of the process routine, send data must be fetched with consideration for the bank. <Send Data Completed> When sending data completes normally, CY=0 is returned. (g) "Receive" function process When entry is "A register=03H" in expansion routine, data receive process is performed. Input parameter at "Receive": HL register .. Receive data stored at leading address. Expansion process stores the unaffected data received at addresses which follow the address indicated by HL, and return the number of bytes to BC. 30FTWARE 11 - 8

<Open Check>

To check whether it has been opened, if not, an error (A=O3H, CY=1) occurs.

<Serial Variance>

When IC cards are used at the same time, the serial line switches. Due to this, the serial line must be switched to the current mode before the "Receive" process in expansion process. However, if other serials are not used between "Connect" and "Disconnect", this is not necessary.

<Receive Data Process>

To store the data received at addresses which follow the address indicated by HL. Data storage is always performed in RAM; data is only unaffected data. The receive buffer of COMBUF (256 bytes) is used to receive control code.

(Note) Since storage of receive data is performed in RAM and may enter the last bank of the process routine, storage of receive data must be processed with consideration for the bank.

<Receive Data Completed>

When 1 block of receive data is completed, the number of receiving bytes is set to BC register and CY=0 is returned.

(Note 1) In the system (DLL,DL) process, the process is based on the following regulations.

\* Completion of receive data is decided by BC=O. When receive data is completed, "Disconnect" function is called to complete the process.

\* Receive process is performed in 1 record (128 byte) units. If it cannot be performed in 128 byte units, blocking process must be performed by the expansion process.

(Note 2) When expansion protocol operates DLL and DL, condition of (Note 1) mentioned above must be satisfied or the system process must be expanded using DLLHK1 or DLHK1.

(h) "Disconnect" function process

When entry becomes "A register=04H" in the expansion process routine, the loop is released.

The expansion process performs serial line close process after the process necessary for loop release from the host computer. The close processes to be performed at "Disconnect" in the expansion process are as follows.

<Close Process>

\* Receive interrupt is disabled. When interrupts other than receive interrupt is used, it must be returned to the state prior to use.

\* Set OOH to RSPSTS, RSINTST.

\* Set FFH to SRMODE, RSODEV.

(Note) Close process must be always executed regardless of "Connect" function execution. ("Disconnect" is not error-returned.)

4 Remarks

(a) Development cartridge

The system forces communication through the development cartridge when the development cartridge is installed (in open state) at BIOS TCAM. Therefore, this process is not performed when TCAM expansion is processed, development with the development cartridge cannot be performed. However, as only DLL process out of TCAM relates to the development cartridge, the development cartridge operation is enabled by considering the system process to operate only at DLL process.

(b) Receive buffer

COMBUF (256 bytes) is used to receive protocol control code. This receive buffer is used to communicate at BIOS RSIOX and does not compete with other buffers during RSODEV check.

(3) DLLHK1

1 Overview

DLLHK1 is used to expand DLL (Down Line Loader) process. The DLL process loads the program from the host computer and starts it, when an application program does not reside in each drive of A:, B: or C:, it starts by turning on Restart. As DLL uses TCAM in BIOS, it is automatically expanded to TCAM expansion protocol along with TCAM expansion mentioned in the previous item. However, as DLL cannot operate well with TCAM expansion protocol, DLLHK1 is used for support.

2 Location of hook

The location of DLLHK1 during process is shown in Figure 11-4 to 11-6.



Fig 11.2 DLL Main Process

(Note 1) When the state of each type changes by system menu, it returns here.

(Note 2) Main process starts the received program, special consideration is not necessary in applications.







3 Remarks

(a) DLL process uses BIOS TCAM to receive files. DLL process receives data in 1 record (128 byte) units. If the receive data process is not a 1 record unit, expansion using DLLHK1 is necessary.

(b) The conditions in which the main hook is not used at protocol expansion are as follows.

\* File name must be stored in TCAMAREA at BIOS TCAM "Connect".

\* Data must be received in 1 record units at BIOS TCAM "Receive".

\* Receive data completion condition (BC=0) must be satisfied at BIOS TCAM "Receive".

\* Loop cut process must be performed at BIOS TCAM "Disconnect".

\* Normal/abnormal completion return parameter of BIOS TCAM must be satisfied.

(c) When an error occurs at the disk function of BIOS TCAM, BDOS, the process stops, loop cut and file close are processed.

(d) When BAS file is received, it goes through DLL process then gives control to BASIC. Basic receives data one record at a time storing them in RAM using BIOS TCAM "Receive".

(e) When a development cartridge is installed, receiving data is performed not from the RS-232C in the DLL process, but through the development cartridge. Thus, protocol expansion using a development cartridge with DLL process operation is impossible.

## (4) DLHK1, ULHK1

1 Overview

DLHK1 and UKHK1 are used to expand DL (Down Loading) and UL (Up Loading) processes. DL/UL process starts when the system menu DL/UL is specified. DL/UL process is to send/receive data to/from a file host computer for Å drive (RAM dick). As DL/UL process uses BIOS TCAM internally, it is expanded to TCAM expansion protocol automatically along with TCAM expansion mentioned in item (1). However, TCAM expansion protocol may not operate DL/UL well, DLHK1/ULHK1 is used for support.

2 Location of hook

The location of NLHK1 and ULHK1 during processing is shown in Figure 11.5 to 11.7.



SOFTWARE 11 - 15



Fig 11.6 Download process





3 Remarks

(a) DL/UL process uses BIOS TCAM to send/receive data. DL/UL process sends/receives data in 1 record (128 byte) units. If send/receive process by expansion is not in 1 record units, expansion using DLHK1 or ULHK1 is necessary.

(b) The conditions in which the main hook is not used at protocol expansion are as follows.

\* File name must be stored in TCAMAREA at BIOS TCAM "Connect" when data is received. (Filename is stored in TCAMAREA at send data.)

\* Data send in 1 record units at BIOS TCAM "Send" must be performed.

\* Data receive in 1 record units at B1OS TCAM "Receive" must be performed.

\* Receive data completion condition (BC=O) must be satisfied at BlOS TCAM "Receive".

\* Loop cut process must be performed at BIOS TCAM "Disconnect".

\* Normal/abnormal completion return parameter of BIOS TCAM must be satisfied.

(c) When an error occurs at the disk function of BIOS TCAM, BDOS, the process stops, loop cur and file close are processed.

(d) As the installed development cartridge does not effect DL/UL, special care is not necessary for applications.

11.2.4 Communication Relation Work Area Detail

Address : Variable name (Number of bytes)

```
F1CDH : TCAMPRM(7)
```

\_

This area stores the initiation conditions used at TCAM open operation. The initiation conditions are set and modified by CONFIG but they can also be modified by an application program.

+0	pe ر T	Туре:	Pro	otocol type	
+1	Line	=	:0	protocol direct-C	
+2	Bit rate	=	1:	protocol direct-8	
+3	Character length	=	2:	Filink (default)	
+4	Parity check	=	-3:	Extended protocol	
+5	Stop bit				
+6	Special parameter				
<ul> <li>=0: RS-232C (default)</li> <li>=3: Cartridge I/F</li> <li>The bit rate, character length, parity check, stop bit, and special parameter are same as that of BIOS RSIOX. The default values are 800 Bps, 8 bits, Nonparity, and 2 stop bits.</li> <li>(Note) When expansion protocol is specified by type without adding expansion protocol and this BIOS is executed, an error returns. See protocol expansion for details</li> </ul>					
F1D4H	: TDFLTCNT (1) This area is used t when the Filink pro default is 3.)	o specify th tocol does n	ie nu iot π	umber of retry operations executed match with the host protocol. (The	
F 105H F 107H	: T1STTIME (2) : T2NDTIME (2) This area is used for receiving (unit: S T1STTIME: Timer for T2NDTIME: Timer for In the first data for when Filink protoco is not used (Direct	to specify the beconds). For the first the second teceive operation to used, o to or -C).	data data dat ition or 1-	imer used for timeout during data a receive operation (30 seconds) ta receive operation (3 seconds) n, data up to file name is received -byte data is received when protocol	
S	(Note) When expansi expansion protocol ee protocol expansi	on protocol and this BlO on for detai	iss Sis Is.	specified by type without adding s executed, an error returns.	

F1D9H : T1STFLG (1) Flag to identify "initial", "subsequent" when timeout time is determined. =0 : "Initial" =1 : "Subsequent" = -1 : "For future receive" F1D8H : DRCVCNT (2) FIDDH : DRCVBUF (2) Specifies the area for storing data at DLL, DL/UL. DRCVCNT ... Size of data storage area. Default is 128 bytes. DRCVBUF ... Leading address of data storage area. Default indicates DLLBUF. FIDFH : DSKNUM (1) Specifies the drive for file operation at DL/UL. Default is drive A. If the value of this area is "C" drive (=3) for example, drive for DL/UL becomes IC card and send/receive of IC card file is enabled. F6C3H : TSPSAVE (2) Area to save the stack pointer when BIOS TCAM starts. F6C5H : TCNTDT (2) Work area to check the number of times of retry when Filink protocol is specified. F6C7H : TERRTIME (2) Work area to check timeout time at BIOS TCAM. F6C9H : TCAMAREA (12) Work area to pass parameters of applications at BIOS TCAM. It has the following meaning at Filink protocol. At file send ... Connect process is performed after filename (11 bytes) are stored in this area. At file receive ... When connect process is performed, filename (11 bytes) received in this area is stored and F6C9H returns. At no sequence: F6C9H is not used at TCAM but is used as the area for filename memory in DL/UL or DLL process. (To equalize the interface with that at Filink.) At expansion protocol: It is desirable to use an identical interface as Filink. F6D5H : TCAMIF (1) Area to store whether connection is performed in send/receive mode when Filink protocol is specified at BIOS TCAM. =0 : Send mode =1 : Receive mode F6D6H : DLLSVSP (2) Work area to save the stack pointer when DLL process starts. F6D8H : LOADAD (2)Work area which stores the loading address of data which follows when loading a file to DLL process TPA area. SOFTWARE 11 - 20

F6DAH : DLLRVS (1) Flag to indicate reverse status of the message "Down Loading", "Up Loading" at DLL and DL/UL processes. F6DBH : DLLFILE (1) Work area to store file types received at DLL process. =0 : COM file receive =1 : BAS file receive = -1 : For future receive = -2 : File other than COM/BAS receive (While open) F6DCH : ERRADR (2) Indicates the execute address of the error process when an error occurs in the disk process during communication. F6DEH : ULDLSVSP (2) Work area to save the stack pointer when DL/UL process starts. F6EOH : ULDLTYPE (1) Flag to indicate which process is executing in DL/UL process. =0 : Down Loading = -1 : Up Loading FGE1H : RENFLG (1) Flag to indicate whether Rename process is executed at write process to disk in DL/UL process.

11.3 IC Card Protocol expansion

11.3.1 Overview

EHT-10/EHT-10/2 contains an IC card I/F and can read/write IC cards based on ISO standard. OS supports the following two types of usage as IC card use format.

(1) Use of IC card as disk(2) Use of IC card as serial communication

(1) IC card as disk

IC card can be assumed to be a disk for R/W or R/O and can be used as a floppy disk or RAM disk. OS communicates with the IC card with Toppan printing protocol. (See Section 6.5) Therefore, "protocol expansion" must be performed for IC card with another protocol. Protocol expansion is performed using a hook arrangement in the OS process.

(2) IC card as serial communication

IC card is assumed to be media containing a CPU and only supports serial communication. OS supports this with BIOS ICCARD. Special care of protocol expansion is not necessary for applications, it can be assimilated in its process.

- 11.3.2 Support at OS
- Disk support

OS supports the use of the IC card as a disk. The IC card is assigned to drive C and is accessed using SELKSK, READ and WRITE of BIOS. (When BDOS is used, BDOS accesses using the BIOS of SELDSK, READ and WRITE in its process.)

SELDSK ... Drive selection (Fig. 11.8) READ .... Data read in 1 record units (128 bytes) — Fig 11.9 WRITE .... Data write in 1 record units (128 bytes)

ICDKMNT ... IC card installation check (See item 3) ICRECRD ... 1 record read (See item 4) ICRECWT ... 1 record write (See item 4)

The routines mentioned above are actually in charge of access to the IC card.



Fig 11.8 SELDSK Process



SOFTWARE 11 - 24

## **3 ICDKMNT**

(a) Overview

ICDKMNT starts at the first IC card access or at the IC card access after OFF operation is generated while IC card is in use. ICDKMNT checks IC card installation, IC card capacity and sets data to DPB (Disk Parameter Block). ICDKMNT supplies power and clock to the IC card, releases reset, accepts reset response and puts the IC card in the open state.

<Note> When OFF operation occurs while IC card is in use, one of the following conditions is satisfied.

\* When the main power supply of the main unit is turned off (power switch, battery charge or auto-power is off.)

\* When access to the IC card is not performed for a preset time (approx. 60 seconds as default).

\* When the rear cover of the IC card opens.


(Note 1) Serial line is performed at 9600 bps, 8 bit, even parity. (Set values are changeable.)

Fig 11.10 IC Card Mount Process



Fig. 11.11 IC Card Open Process



(Note 1) When IC card protocol is different, an error occurs in this process, thus, expansion using a hook is necessary. When this process is expanded, maximum number of records, capacity and DPB setting are necessary.

(Note 2) Set values corresponding to capacities are as follows.

QTICCARD	0 to 2 KB	2 to 8 KB	More than 8 KB	
DPB2 + 7 (Directory size)	7	15	31	
TPICCARD	2	4	8	
DPB2 + 11, 12 (Check vector size	2	4	8	

QTICCARD - 1 -> DPB+5, 6 (Disk max. size)

Fig 11.12 1C Card Capacity Check Process

## 4 ICRECRD, ICRECWT

# (a) ICRECWD

ICRECWD computes the physical address to access SEKTRK (track), SEKSEC (sector) IC card, reads data of 128 bytes from the computed address, stores in DMAADR (buffer) sequentially and returns. At his time, additional data of the header or checksum should not enter the 128 bytes and data stored in DMAADR is only unaffected data.

(b) ICRECWT

ICRECWT computes the physical address to access SEKTRK (track), SEKSEC (sector) IC card, then writes and returns data of 128 bytes stored in DMAADR (buffer), which follows the computed address. As data stored in DMAADR is unaffected data, header or checksum is added for processing, according to requirements.

<Note> When IC card is in close state and ICRECRD or ICRECWT is specified, ICDKMNT process is executed for open state before this process starts.

(c) ICRECRD Process



(Note 1) See "7.2.4 Serial Communication by User" for serial line switching.

(Note 2) For example, directory is always assumed to have 8 sectors (1 Kbyte) in the system, the data section starts with 0 track 8 sectors. However, the directory area has 256 bytes or 512 bytes actually, if 0 track 2 sectors or 0 track 4 sectors are specified, it becomes virtual sector. (See below)

(Note 3) If protocol is different, expansion is performed using a hook.



(d) ICRECWT Process



Fig 11.14 IC Card Write Process

# **5** ICDFMT

ICDFMT format-processes IC card to use IC card as CP/M disk. In format process, E5H is written to IC card data area using ICDKMNT, ICRECRD, ICRECWT as mentioned above. This process starts when the disk format of the system menu is specified.



(Note 1) When the protocol is different, expansion is performed using a hook.

Fig 11.15 IC Card Format Process

(2) Serial support

Due to IC card confidentiality of its original characteristics, OS supports BIOS ICCARD as standard, which supports only serial communication function. The following 5 functions are provided for this BIOS.

Open (puts IC card in open state.) Close (Puts IC card in close state.) Read (1 byte data receive) Write (1 byte data send) Status (Status of receive buffer)

BIOS ICCARD is different from a disk and receives data using RXRDY interrupt. (Disk performs send/receive data in coded units, the software timer monitors timeout time.) See item ICCARD in "4.2 BIOS Function Description" for further information.

- 11.3.3 Protocol Expansion
- (1) Overview

This section describes expansion when using the IC card which utilizes a protocol other than the protocol OS supports as standard. The following two types of usage of the IC card are as mentioned in the previous item.

1 Use of IC card as a disk 2 Use of IC card as serial communication

For the latter, special care of the protocol difference of the IC card is not necessary. (It can be assimilated in the application.) However, with BIOS ICCARD open function, reset response check is executed after Reset is released. Thus, IC card of which process at reset is different must be expanded using ICDHK1 (open process hook).

For the former, expansion is performed depending on the difference of protocol from standard, but basically the following 4 types of hooks are used.

ICDHK7 ... Hook in ICDKMNT ICDHK5 ... Hook in ICRECRD ICDHK6 ... Hook in ICRECWT ICDHK8 ... Hook in ICDFMT

(2) Expansion procedure

1 Expansion program loading

Program for protocol expansion can be placed in/on any of the following.

(a) In user BIOS area
(b) On application ROM
(c) In RAM (TPA)

With (a), see "4.6 User BIOS" for maintaining. With (b), care of location (address) of expansion program in memory map is necessary. (When the hook is rewritten, it directly jumps to the specified ROM, relating to the location on ROM and memory map.) With (c), no problems occur while the application is executing, but if another application starts, it may unconsciously speed-up. Expansion process must be closed in applications.

#### 2 Rewrite of Hook

Rewrite the contents of the hook corresponding to the process to be expanded.

Bank Address FF6FH -> Bank Sub-bank ICDHK1 Address Main bank (low,high) =0 : RAM =1 : System ROM +3H Bank =2 : Application ROM ICDHK? Address (Ex. 1) When expansion process is at 6C80H or application ROM (27C256)+6H Bank Bank = 20HICDHK3 Address = 6080HAddress (Ex. 2) When expansion process is at DOOOH in user BIOS area in main RAM. Bank = 00HAddress = DOOOH +21H Bank ICDHK8 Address

#### Fig 11.16 Rewrite of Hook

3 IC card expansion specification

IC card hook does not jump to the expansion process only by rewrite of a hook. IC card expansion specification is also necessary.

ICHOKDT1 ... Specifies whether jump to IC card hook must be (FIE6H) performed.



Bits correspond to hooks, specification is as shown below. Set bit corresponding to the proper process to 1 in order to jump from the next IC card access to the address specified by the bank which is set in item 2.

(Ex. 1) When specifying reset response process expansion. ICHOKDT1=80H

(Ex. 2) When expanding IC card installation check, 1 record read/write and format processes. ICHOKDT1=78H

(3) Process flow of each hook

Flow chart shows the positions of the hooks (ICKH1,ICDHK5 to a). Each hook decides whether to jump to the specified address using the following process.



As control moves to the expansion process by jump instruction, if return is executed in the expansion process, original standard process is ignored and only the expansion process becomes optimizing.

11.3.4 Mount process Hook (ICDHK1, ICDHK7) (1) Hook at reset response (ICDHK1) 1 Overview This hook starts when access is going to execute to IC card in close state and puts IC card in open state. This hook opens IC card. 2 Items where expansion is required in this hook (a) To check whether IC card rear cover is open. When open, it must return with an error (CY=1. A=OAH) (b) To enable IC card rear cover interrupt. (c) To set IC card serial line to read mode. (d) To supply power to IC card. (e) To supply clock to IC card. (f) To set IC card power ON flag (ICONFLG) after waiting for at least 50 msec and releasing the reset signal. (g) To receive reset response from IC card. Return is performed by CY=1, A=06H at timeout error and by CY=0 at normal end. 3 Note (a) Register does not need to be held. (b) As communication conditions (transfer rate, parity ... etc.) have been already set (9600 bps, 8 bit, even parity), if modification is required, SYSCTLR1 TO SYSOUT must be modified in advance. (2) Hook at IC card installation check (ICDHK7) 1 Overview This hook starts when IC card is in the close state and access is executed to IC card. This hook computes IC card capacity and regulates DPB (Disk Parameter Block). (See ICDKMNT in the previous item.) 2 Items where expansion is required in this hook (a) To put IC card in open state for software. To perform specified collation and enable read/write execution of data that follows. (b) To set IC card DPB (Disk Parameter Block) as drive "C" when IC card is in open state. 3 Note (a) The following require area setting in this hook. QTICCARD: IC card size (data area) Numeric value is specified in ICBLKS2. When the value is 0, no IC card is assumed. ICMAXREC: Number of access enable records in (F65CH) IC card data area, (the value that represents the net amount of data that can be held in 128 byte units). TPICCARD: Number of records which is held as IC card directory area. SOFTWARE 11 - 35

DPB3(F177H)... Disk parameter black (Drive C:) DPB3 + 5, 6 (Disk size max) DPB3 + 7 (Directory size) DPB3 + 11, 12 (Check vector size) (b) Return must be performed by CY=0 at normal end and by CY=1 at error end. 11.3.5 Hook at 1 Record Read (ICDHK5) (1) Overview This hook starts when IC card is specified by BIOS READ. (2) Item where expansion is required in this hook (a) To set serial line to SIO. (b) To compute access address from SEKIRK, SEKSEC to IC card. To fill NUL code in DMA buffer for return with virtual record. (c) To read data of 128 bytes from IC card using "read command" and store in DMA buffer. (3) Note (a) Each register of BE, DE, HL must be preserved. (b) See "6.5 IC Card" for track/sector correspondence with IC card address. (c) Return must be performed by CY=O at normal end and by CY=1 at error end. 11.3.6 Hook at 1 Record Write (ICDHK6) (1) Overview This hook starts when IC card is specified by BIOS WRITE. (2) Items where expansion is required in this hook. (a) To set serial line to SIO (IC card). (b) To compute access address from SEKTRK, SEKSEC to IC card. With virtual record, just return. (c) To write DMA buffer data of 128 bytes to IC card using "write command". (3) Note (a) Each register of BC, DE, HL must be stored. (b) See "6.5 IC Card" for track/sector correspondence with IC card address. (c) Return must be performed by CY=O at normal end and by CY=1 at error end.

11.3.7 Hook at Format (ICDHK8)

(1) Overview

IC card formatting is processed after installation check in a close state. This hook is in real formatting start point.

(2) Items where expansion is required in this hook.

(a) To write E5H to the entire area using "write command" to IC card. (b) Return must be performed by CY=0 at normal end and by CY=1, A register=2 after IC card close process at error end.

# (3) Note

IC card close process is executed as follows.

(a) Reset RXENB bit.
(b) Reset IC card.
(c) Disable IC card rear interrupt.
(d) Stop clock supply to IC card.
(e) Stop power supply to IC card.
(f) Clear each flag. (write 0) (ICCPWFLG, ICUSEDV, ICONFLG, ICCVFLG)
(g) Set serial line to RS 232C.
(h) keturn PXENB bit to original.

11.3.8 IC Card Protocol Expansion Example

"Peference 9. Sample 33" describes expansion example using IC card.

#### 11.3.9 Remarks

(1) Communication parameter modification

EHT-10/EHT-10/2 sends/receives data at 9600 bps, 8 bit, even parity as a communication parameter. To modify this communication parameter, follow the procedure as described below.

1 Set SYSCTLR1 and SYSARTMR to required communication parameter.



SYSCTLR1 specifies transfer rate, 9600 bps is constant for IC card.

2 When parity is modified, data for TS byte collation at reset response of ICTSDT(F1E9H) must be modified.

(Ex.) No parity ICTSDT = FBH Even parity ICTSDT = 3BH

(2) Interrupt

El state is always held at jump to hooks, if another interrupt is issued in each process, malfunction may occur. Thus, IER (Interrupt Enable Register) operates and all interrupts are disabled in order to jump to a hook. Therefore, the following process must be inserted at return from each hook.

ZIER EQU 04H LD A,(RZIER) — Return interrupt to the state prior OUT (ZIER),A — to the hook jump.

(3) Remarks for using IC card

1. Error check at command send and response receive. As IC card operates in DI state normally at command send and response receive, error check is performed as follows.

(a) Check open/close state of rear cover at command send and response receive, if open, an error occurs.

LD A.(ICCVFLG) Check rear cover interrupt of OR A IC card SCF RET NZ LD A.(ICONFLG) Check power off INC A RET Z OR A RET

(b) Check timeout in the process of which response receive loops.

(4) IC card power-on sequence

With EHT-10/EHT-10/2 OS, power-on sequence to IC card supplies power and clock, releases reset and resets response process by release of reset. Reset response process is as follows.

1 Receive TS byte.

2 Check whether received TS byte data is equal to the value of ICTSDT(F1E9H). If different, an error occurs so do not execute the following processes.

3 Receive TO byte

4 Determine whether TA1 to TD1 that follow are received, following the data of received TO byte.

5 If there is data of TA1 to TD1, receive it.

```
6 Open process completes with end of data receive of all reset
     response.
     TS byte, TO byte and data from TA1 are stored in the area that
     ICRSTPNT(F1AEH) points.
     In expansion process, when either of the following conditions occurs,
     reset response process must be expanded using ICDHK1.
     1 When timing of power supply, clock or reset is modified.
11.3.10 Work Area Detail Related to IC Card
     This section describes the work area related IC card. "*" means that
     this variable is different from OS Version 1.0. For the difference of
     system work area between Version 1.0 and 2.0, see Appendix 10.
(Example)
Address(*): Name of Variable (Byte Qty.)
EACOH*: ICBASIF (29)
     Area for accessing the IC card from BASIC.
EADDH*: BDOSIF (33)
     Area for accessing the IC card from BDOS.
EFCOH*: ICFILNAM (42)
     Area for the file names of the IC card.
EFEAH*: ICDIDDAT (22)
     Area for ID data of the IC card.
F196H: SYSCTLR1 (1)
F197H: SYSARTMR (1)
F198H: SYSSWR (1)
F199H: SYSARTCR (1)
     Communication parameter with IC card when using IC card as a disk.
     Default is 9600 bps, 8 bits, EVEN parity.
     Bit configuration is the same as that of 1/0 port data.
          SYSCTLR1 ----> CTLR1 (PO1H)
          SYSARTMR ----> ARTMR (P15H)
          SYSSWR ----> SWR (P18H)
          SYSARTCR ----> ARTCR (P16H)
F1ADH*: ICRSTCNT (1)
     Specifies the number of receive bytes when the answer to reset.
          = OOH : Ignore the answer to reset
                    number of receive bytes. When answer to reset, receives
          ≠ 00H :
                    this number of bytes, and stores the data to the area
                    which is pointed by ICRSTPNT. If the time out coror is
                    detected during the receiving, system stores the data
                    till the error is detected. default value is O9H.
```

F1AEH: ICRSTPNT (2) Represents the storage area of the Reset response upon cancellation of the IC Card's Reset status. ICRSTPNT ----> ....Size of area (in bytes) n Default value = 10 bytes. TS TO. The procedure for storage to this area is specified by the value of ICRSTCNT. F1BOH: IDFILCNT (1) F1B1H: ICCDTIME (2) F1B3H: ICCDPRM (5) Area to regulate each access condition of BIOS ICCARD (independent from disk access) IDFLTCNT: Specifies the number of times of retry when IC card error occurs. Default is 3 times. ICCDTIME: Regulates timeout time when data is received from IC card. Default is approx. 3 seconds. ICCDPRM : Har loop open parameter when ICCARD OPEN is specified by IC card. (Note) In BIOS ICCARD OPEN process, this section of communication parameter is copied after SPBADR+4 and BIOS RSIOX open process is executed using COMBUF area as receive buffer. Therefore, when other modules have already used RSIOX, open error occurs. It must be used after RSIOX close or ICCARD close process is executed. ICCOPRM +0 Bit rate Default value is 9600bps 8bit, Even parity, 1 stop +1 Bit length bit. +2 Parity +3 Stop bit +4 Special parameter F1E2H\*: ICSRACD (1)SRA code (the 1st byte during command transmission) Default value = 3AH F1E3H\*: ICFLCLS (1) File class code. This is the class code when using file-related commands. Default value = 00H. The File No. of the ISET function is entered here. F1E4H\*: ICCLASS (1) System class code. Default value = 20H SOFTWARE 11 - 40

```
F1E5H*: ICRECSZ (1)
     Specifies the number of bytes of 1 record. Default value = 40H
F1E6H*: ICHOKDT1 (1)
     Specifies whether execute the hook processing or not.
         Bit = 1 : Jump to the hook.
              = 0 : Do not jump (Default)
     7 6 5 4 3 2 1 0
                  0
                     0
                        0
                            -> ICDHK5 (1 record read)
                            -> ICDHK6 (1 record write)
                            -> ICDHK7 (IC card mount check)
                            -> ICDHK8 (IC card format)
                            -> ICDHK1 (Answer to reset)
F1E7H^*: ICWAIT1 (1)
     Specifies the interval (in ms units) for awaiting the start of Reset
     response reception after the Reset status is canceled. Default value =
    50 ms(32H)
FIE8H*: ICWAIT2 (1)
     Specifies the interval (in ms units) for awaiting the start of command
     transmission after the Reset response is received. Default value = 100
    ms(64H)
FIE9H: ICTSDT (1)
    Data used for TS byte collation during the reception processing of the
    Reset response.
    During Reset response reception processing, only the TS byte is
    collated and the other data is merely received.
    Default value = 3BH
F643H*: ICDIDPNT (4)
    Represents the storage area for ID data during registration of the
    card ID. (Valid only in DISK Mode)
    For the detailed information, see Section 6.5
Fo47H*: ICFILPNT (3)
    Represents the storage area of the File Name data during file creation.
     All of the above data are valid only in DISK Mode and represent the
     address of the data set by the ISET function.
```



```
F65CH: ICMAXREC (2)
F65DH*: QTICCARD (1)
F65FH: TPICCARD (1)
     Indicates data for the IC card capacity.
          ICMAXREC: Indicates maximum number of physical records of IC
                    card. Default is 40H (64records)
          QTICCARD: Indicates physical capacity of IC card. Unit is
                    specified by ICBLKSZ. Default is 20H (32 x 256bytes).
          TPICCARD: Indicates the number of records used for directory.
                    Default is O8H (8 records).
F660H: ICONFLG (1)
F661H: ICCVFLG (1)
F662H*: ICUSEDV (1)
     Flag to indicate each state while IC card is in use.
     ICONFLG --- Indicates IC card power supply state.
                    =0 : No power supply (close state)
                    =1 : Power is supplied (open state)
                    =-1: Power supply is temporarily canceled as there is
                         no access for a present time.
     ICCVFLG ----- Indicates state of the excess current while IC card is
                    in use.
                    =0 : Normal current
                    =: : Excess current goes through the IC card
     ICUSEDV ----- Indicates type of module which is using the IC card.
                    =-1: Command through mode uses IC card,
                    =0 : Not used.
                    =1 : BIOS ICCARD uses IC card.
                    =? : Disk mode uses IC card.
     (Note) Continue to supply power to the IC card while IC card is used in
     BIOS ICCARD.
F663H: ICRSTDT (10)
     Area to store data at reset response (Answer to reset)
     This area is pointed by ICRSTPNT.
F66DH*: ICRETRY (1)
     Reserved.
F66EH*: ICWAIT3 (2)
     Specifies the interval (in 10-ms units) for a time out error for the
     response from the IC Card after command transmission. Default value =
    OOFFH (approx. 2.5 Sec)
F670H*: ICOPNSTS (1)
     Specifies the processing during File OPEN status (after the IC Card
     power is ON). (Valid only in DISK Mode)
     7 6 5 4 3 2 1 0
                                    Default value = 07H
   0
      0
         0
            0
               0
                                 -> ID collation
                                                         (1: ON)
                                                         (1: ON)
                                -> User PIN collation
                                 -> System PSW collation (1: ON)
```

F671H\*: ICFMTSTS (1) Specifies the processing during formatting. (Valid only in DISK Mode) 7 6 5 4 3 2 1 0 Default value = 10H 0 10 0 0 0 0 > = 00 : Do nothing. = 01 : Delete the specified file only. = 1X : Create a file and format after deleted the specified file. F8D2H\*: ICERRSTS (3) Returns the error information generated during access processing of the IC Card. +0 Command code +1 Status (1) +2 Status (2) For details on the data, see the following. F8D5H\*: ICDWORK (69) Area for sending/receiving data to the IC card. Command codes of ICERRSTS (F8D2H) Command Code Command Name(Remarks) OOH. Error unrelated to command processing 01H Open Card ID Open System Password 02H 03H Open User Password 04H Create File 05H Directory Read 06H Erase File 07H Read Write 084 **0**9H Set Address Pointer

Status codes of ICERRSTS (EF8D2H) Status 1 Status 2 OOH OPEN error OlH Excess current (During Open) 02H (EHT-IO side) Reserved. 03H TS byte not received (IC Card not installed) 04H TS bytes do not match 05H Reserved. O6H Reserved. 07H Reserved. **OBH** Reserved. 09H Power off. OAH Excess current (During Write) 01H Abnormal 01H Parity error (see NOTE below) reception 02H Framing error (see NOTE below) (EHT-10 side) 03H Over-run error (see NOTE below) 04H BCC error 05H Reserved. 06H Time-out error 07H Format error(SRA code not match) Status 1 Status 2 ElH CLS is abnormal **DOH** E2H COD is abnormal 00H **F**FH Others 13H Command mode error (for details see the IC card status) (IC Card side) 14H Length error 15H Syntax error (parameter error) 16H Duplicate file names Specified file does not exist 17H 19H Formatting impossible (file already exists) 20H Format error 21H Sum error 22H Memory defect 23H Chain error 24H Memory full 25H Address overflow 30H Memory defect during formatting 32H Password collation error 33H Quantity of valid passwords exceeded 34H Card ID collation error 35H Comparison error 41H File has been forced OPEN (Chain error included) 42H File has been forced OPEN (Sum error included) FOH Abnormal 01H Parity error reception 02H Framing error (IC Card side) 03H BCC error 04H Length error 05H Time-out error

NOTE: When the above three errors occur simultaneously, they are checked in the sequence of: Over-run error -> Framing error -> Parity error.

11.4 Cartridge Device Expansion

11.4.1 Overview

This chapter describes:

- (1) Mount process expansion
- (2) Interrupt process expansion

as expansion method of cartridge device control.

Be sure to read this chapter when you create a new cartridge device, control it or expand the standard device (printer unit) control.

See "7.3 Cartridge Interface" and "4.1 cartridge Interface" in Hardware Version for further understanding.

(1) Mount process

Mount process checks the cartridge device installation and sets the mode. Mount process can add the expansion process of the cartridge device initialization using a system hook (CRGHOOK).

(2) Interrupc process

When creating a cartridge device for interrupt process, user's original interrupt process can be performed using a system hook (EXTHODK).

- 11.4.2 Cartridge Mount Process
- (1) Overview

EHT-10/EHT-10/2 performs cartridge device installation check (mount process) at power-on and reset of main unit and system initialization. OS reads cartridge device code from the cartridge device, stores in memory and determines the cartridge mode using a device code at the same time in order to set the mode. See "7.3 Cartridge Interface" for relation of device code and cartridge

mode.

Mount process routine jumps to system hook (CRGHOOK) after all processes complete so that the cartridge device initialization is enabled at installation check by adding a hook process.

(2) Contents of mount process and location of hook.

Contents of mount process is shown in Figure 11.17.

1 Device code (CRGDEV)

Device code is configured of the cartridge device number and cartridge mode, and is stored in CRGDEV (F42FH) in memory.



2 Decision of mode

Cartridge mode is determined by device number and CSEL (this P16H bit 6 data is set to DB mode and is read). See "7.3 Cartridge Interface" for details.

3 Mode setting

Cartridge mode is set according to the device code. Mode in the development cartridge is modified at development cartridge connection.

.....



Fig 11 17 Cartridge Mount Process

## (3) CRGHOOK expansion

CRGHOOK is used to expand the cartridge device mount process and initializes the cartridge device at power on.

1 Expansion program loading

Hook process program can be placed in the user BIOS area, on application ROM, or in RAM (TPA). See "10.2 System Hook" for procedure and remarks.

2 Rewrite of hook

Rewrite the contents of a hook corresponding to the process to be expanded. CRGHOOK configuration is as follows.



3 Hook process execution specification

To execute hook process, set hook process execution specification flag (USERCRG:FODFH) in addition to items 1 and 2. Set value other than 0 to USERCRG to execute hook process.

- 11.4.3 Interrupt Process from Cartridge Device
- (1) Overview

CINT terminal of cartridge I/F is connected to external interrupt signal line of EHT-10/EHT-10/2 main unit and can accept interrupt from

cartridge devices using CINT. (See Hardware Version or Chapter 7 1/0 Description, Chapter 8 Interrupt Process) OS uses Ready interrupt process of the printer unit for interrupt from a cartridge.

To create a new cartridge device for an interrupt process, expand the external interrupt hook (EXTHOOK).

This section describes the interrupt process from a cartridge device using EXTHOOK.

(2) I/O port related to interrupt

I/O port related to interrupt from the cartridge I/F is as described below.

Port Address (R/W) : Port name (RAM data address) P16H (R) : IOSTR bit 0 : (PBUSY) Interrupt status 0: Interrupt requested 1: No interrupt requested P17H (W) : ICCTLR (FODBH) bit 6 : (PRIE) Interrupt mask 0: Interrupt allowed 1: Interrupt suppressed P23H (R) : ITSR bit 1 : (IPBUSY) Status of interrupt from IC card or cartridge 0: Interrupt requested 1: No interrupt requested P23H (W) : CTLR3 (FODEH) bit 3 : (PINTDIS) Masking an interrupt sent from IC card or cartridge 0: Interrupts allowed 1: Interrupts suppressed (0 is generally set.) (Note) PINTDIS of CTLR3 (P23H) must be always set to O (enable) normally. (3) EXTHOOK expansion 1 Expansion program loading EXTHOOK automatically switches to the entire RAM (bank 0 # 0) to call the expansion process, thus, the expansion process routine must be loaded in RAM. When the expansion process is used in multiple application programs, use the user BIOS area. See "4.6 User BIOS" for further information for how to use the user BIOS area. (See APPENDIX9 SAMPLE 30) 2 Rewrite of hook EXTHOOK is a 3 byte configuration as shown below. The address section is rewritten. **EXTHOOK** - JP instruction FFDBH +0 C3H - Expansion process address

+1

+2

Address information

3 Remarks in expansion process

(a) The expansion process is called in "D1" (Disable Interrupt). "EI" (Enable Interrupt) must not be called in the expansion process.

(b) When the stack is often used, set a new stack. In this case, the stack must be returned after the expansion process.

(c) EXTHOOK is called by the cartridge I/F interrupt, 1 msec / 8 msec interval interrupt and IC card interrupt. Therefore, the contents of the process must be classified by the interrupt cause in the expansion process. PBUSY (P16H bit 0) can acknowledge the cartridge I/F interrupt.

(d) Expansion process completes with the following format. System standard process continues after the end. (See Note 1.)

<Entry Parameter>

DE register : Interrupt state after the end of the interrupt process.

DE=ODEBH ... Enable interrupt =ODE1H ... Disable interrupt

RETAD	EQU EF4F4	Cost-process address of EXT interrupt.
	LD HL,RETAD	Exchange of post-process address with return address.
	EX (SP),HL PUSH DE	Save specification as enable/disable
	JP (HL)	Jump to return address.

When loading the expansion process, modify a section of the system interrupt process as shown below.

Modification address	EFB9H
Before modification	18H (JR instruction)
After modification	C9H (RET instruction)

Following this, jump to enable/disable interrupt process which is set at the end of the expansion process. Return to the post-process routine following RET instruction of the jump target.

(Note 1) Return to the original state following the process shown above to obligingly disable the interrupt from the cartridge I/F in the system standard process. 11.5 Addition of Barcode Decoder

11.5.1 Overview

ICF interrupt is issued when the input polarity from barcode changes (0"\*1 or 1"\*0). ICF interrupt is issued when barcode is attached to white paper, the LED continuously lights then polarity changes from white to black. If a barcode is now scanned to read black bars, ICF interrupt is issued. ICF interrupt routine ICFINT is as shown below.



Fig. 11.18 ICF interrupt Process

Decoder for specified barcode is called in ICFINT routine. Called decoder correspond to types of barcode specified by BARCODE open function. ICFDSP is the table which registers the vector (entry address) of the decoder corresponding to the types of barcode. ICFINT calls the decoder referring to the types of barcode and ICFDSP at open. ICFDSP of the vector (entry address) to 6 decoders is stored in a 12 byte table up to F6ECH-F6F7H. IF the vector is set to 0, the decoder is not called.

The vector to the user decoder of ICFDSP must be rewritten to add the barcode decoder. 6B vector is stored in ICFDSP as shown below.

Name	Type of barcode	Address	No. of bytes	Contents
	0	F6ECH	2	Reserved for system expansion
	1	F6EEH	2	Vector to decoder for JAN/EAN/UPC-A/UPC-E
ICFDSP	2	F6F0H	2	Vector to decoder for 3-of-9
	3	F6F2H	2	Vector to decoder of codabar
	4	F6F4H	2	Vector to decoder for interleaved 2-of-5
	5	F6F6H	2	Vector to decoder for user

ICFDSP is referred in INTICF routine and vectors are called. The memory map is as shown below.



Decoders for the user are created as described below.

(1) Time of white or black is queried using the value of FRC(Free Running Counter). This can be queried by calling RDTIME routine. ICFINT routine is set so that the time of black can be read when RDTIMR is first called. The time of white can be read when RDTIMR is called for a second time. After this, every time RDTIMR is called, the time of black and white can be read alternately.

```
RDTIMR

Address : AFOOH

Input condition : HL = overilow time

Output condition : HL = FRC value (when CY-Flag=O)

CY = 1 ... timeout

Description: Overflow time is the maximum wait time till

polarity changes and is given by the value of FRC.

If polarity does not change after wait overflow

time, the scan by the barcode reader is assumed to

have completed or canceled.
```

(2) Decode the bar code to character by the time ratio of black and white. Store the decoded character in the temporary buffer. This can be performed by calling TPUTQ routine. The temporary buffer is initialized by ICFINT routine.

TPUTQ	
Address	: AD57H
Input condition	: A = character
Output condition	: CY = 0 Normal end
	1 Buffer overflow
Description: The	temporary buffer stores the decoded character
temp	orarily. The temporary buffer shares the area
with	the character buffer. BIOS BARCODE read
func	tion takes one character out of the character
buff	er. The temporary buffer pointer must be
reas	signed to the character buffer using SETQ
rout	ine.

(3) When decode completes without an error, reassign the temporary buffer pointer to the character buffer to store it as data. This is performed in SETQ routine. SETQ appends CR to the end of data after referring to the specification whether CR must be appended at BARCODE open.

> SETQ Address : ADA3H Input condition : None Output condition : CY = 0 ... Normal end I ... Buffer overflow

11.5.2 Hand scanner

ICF interrupt routine does not support a hand scanner, driver for a hand scanner must be created. The hand scanner serial-transfers in ASCII code after it reads a barcode and self-decodes. The parameter at serial transfer is as shown below.

Communication rate ... 1200 bps Start bit ..... 1 bit Data length ..... 8 bit/char Parity bit ..... No parity Stop bit ..... 1 bit

The program which converts to characters using the barcode routine can be created by input of serial-transferred data to the bar code terminal. Start bit 1, data length 8, stop bit 1, so 10 bits per character is transferred. As 1200 bps, pulse width per bit is;

1/1200 = 833.3 micro Sec.

change this to the number of counts of the free running counter.

833.3 / 1.6276 = 512 counts



```
1 = Open
```

F6F9H : BCTYPE (1) Type of bar code 0 = For system 1 = JAN/EAN/UPC-A/UPC-E 2 = 3-of-9 3 = Codabar 4 = Interleaved 2-of-5 5 = For user

```
F6FAH : BCOPTN (1)
     Option parameter at open
          bit 7 = Buzzer disable at normal read
                =0 : Buzz
                =1 : No buzz
          bit 6 = Delimiter disable
                =0 : Delimiter enable
                =1 : Delimiter disable
                 (Use delimiter of TRMCHR)
          bit 5 = LED pulse control disable
                =0 : Pulse control enable
                =1 : Pulse control disable
          bit 4 = For future use
          bit 3 = For future use
          bit 2 = Zero addition specification
                 (Valid only with UPC-E)
                =0 : Zero not added
                =1 : Zero added
          bit 1 = Full ASCII conversion specification
                 (Valid only with 3-of-9)
                =0 : Full ASCII conversion
                =1 : No full ASCII conversion
          bit 0 = Check digit specification
                 (Valid only with 3-of-9)
                =C : Last dath is not assumed as check digit
                =1 : Last data is assumed as check digit.
FL_{TBH} : TRMCHR (1)
     Used when delimiter, BCOPTN bit 6 = 0
F6FCH : SCNERR (1)
     Error code detected at end
           bit 7 = Buffer overflow
           bit 6 = For future use
           bit 5 = For future use
           bit 4 = For future use
           bit 3 = For future use
           bit 2 = For future use
           bit 1 = For future use
           bit 0 = Scan error
F6FDH : LEDTIM (1)
     Time for LED continuous lighting
           Default is 80H
F\in FEH : LEDCNT (1)
     Counter for LED continuous lighting
           0 = Pulse lighting
           Other then O = continuous lighting
F5FFII : PLSFRQ (1)
     Period of LED pulse flash
           Default is 8
F700H : PLSCNT (1)
     Counter for LED pulse control
           0 = LED on
           Other than O = LED off
                             SOFTWARE 11 - 57
```

F701H : PWRCNT (1) Counter till logic power off Logic power goes down 1 to 2 seconds after close When closed, this becomes 2 F702H : FRCBUF (2) Address (FAAAH) of FRC (Free Running Counter) buffer address TIMBUF enters. F704H : FRCSIZ (2) Number of counters which can enter FRCBUF, it must be the value of 2<sup>n</sup>. default is 64. F706H : OVTIME (2) Timeout counter value Default is 65536/8 F708H : TIMCNT (2) The last read FRC value F70AH : TIMLOC (2)Number of counters stored in FRC buffer F70CH : BCGETP (2) Get-pointer of character F70EH : BCPUTP (2)Put-pointer of character F710H : BCLOC (1) Number of characters in character buffer. F711H : BCLOF (1) Capacity empty in character buffer. F712H : BCTPTP (2) Pointer to store temporarily in character buffer. F714H : BCTLOC (1) Number of characters temporarily stored in character buffer. F715H : BCTLOF (1) Remaining capacity which can be temporarily stored in the character buffer. F716H : SPCTHR (2)Space threshold value (FRC value) F718H : NRWSPC (2) Narrow space value (FRC value) F71AH : WIDSPC (2) Wide space value (FRC value) F71CH : BARTHR (2)Bar threshold value (FRC value)

F71EH : NRWBAR (2) Narrow space value (FRC value) F720H : WIDBAR (2)Wide space value (FRC value) (F716H to F721H) : UPCBUF (12) Work area for UPC-E zero addition F722H : BITPIN (2) Bit pattern of space and bar SPCBIT (F722H) ... Space bit pattern BARBIT (F723H) ... Bar bit pattern F724H : DIRECF (2) Direction flag or parity flag LFLAG (F724H) ... Left parity for JAN RFLAG (F725H) ... Right parity for JAN F726H : MODULS (2) Sum of module (JAN) or sum of check digit (3-of-9) FA5AH : BCDBUF (80)Character buffer Decoded bar code is stored in character (ASCI) units. F822H : TIMBU⊮ (128) FRC (Free Running Counter) buffer. Value of FRC at change of black and white polarities is stored.

# Part 2 HARDWARE

CHAPTER 1 Hardware Overview

1.1 Hardware Structure

1.1.1 Overview

EHT-10/EHT-10/2 uses 2 CPU types, C-MOS Z-80 compatible CPU ( $\mu$ PD70008) is used as the main CPU, and 4 bit C-MOS CPU7508 as a slave CPU. The slave CPU7508 controls key input, clock, power supply and performs serial communication with the main CPU. Main memory contains 256 KB (maximum) of RAM and 256 KB of ROM (128 KB for the system with a maximum of 128 KB for applications) for a total of 512 KB, switched by the bank register.

The main circuitry of the EHT-10/EHT-10/2 consists of 3 semi-custom LSIs (GAPNIO, GAPNIT, GAWIS), the I/O register in each custom LSI can be accessed by software using I/O instructions. Relation of the display and input of EHT-10 and EHT-10/2 is hardware dependent.

(1) EHT-10

Display : 154x84 dot LCD and LCD controller T6963. 2 KB is dedicated for VRAM.

Input : Touch panel with 14x5 input points

(2) EHT-10/2

Display : 120x32 dot LCD with the controller in GAWIS. VRAM uses part of main RAM. EHT-10/2 is available in a model (EHT-10/28) with an EL back light.

Input : Keyboard with 34 keys and 3 LEDs

The EHT-10/EHT/10/2 includes RS-232C I/F, cartridge I/F, bar code I/F and IC card I/F for external interface connection. The cartridge I/F consists of 4 modes (HS mode, DB mode, IO mode, OT mode), and is structured for easy creation and control of each cartridge option.

The EHT-10 and EHT-10/2 power supply uses a nickel-cadmium battery pack as a main battery which is charged through an AC adapter . The nickel-cadmium sub-battery is built-in and performs RAM backup as a design safeguard, even when the main battery is discharged, RAM backup is supported.

Figure 1.1 shows EHT-10 and EHT-10/2 Hardware.



Fig 1.1 Hardware Structure HARDWARE Page 12 - 2
#### 1.1.2 Hardware Structure

(1) Main CPU (µPD70008)

Z-80 compatible C-MOS CPU CLK=3.6864 MHz

It sleeps via the HALT instruction optimizing power consumption. µPD70008 operates without WAIT, and neither DMA nor non-maskable interrupts are used.

- (2) Memory
  - 1. ROM

Mask ROM of 128 KB is available as the system ROM, mask ROM and EPROM of 128 KB, 64 KB and 32 KB are available as application ROM.

System ROM µPD23C1000G ... CMOS 128 KB, mask ROM

Application ROM (use CMOS structure with access time faster than 200 ns).

- EPROM : HN27C301G or ones based on this pin arrangement. 128 KB Mask ROM : HN62301P or ones based on this pin arrangement.
- EPROM : 27C256, 27C512 64 KB/32 KB Mask ROM : Based on the EPROM pin arrangement mentioned above.

HN62301P

#### HN27C301G

JZ VCC

30 NC

29 A 14

28 A 13

27] A 8

26 A 9

25 A 11

241 A 16

23 A 10

22 CE

2107

20 0 6

190s

10 4

10 CT

31 PGM

A 15 1 A 12 2 A 7 3 A 6 4 A 5 5 A 4 6 A 3 7 A 2 9 A 1 100 D 0 11	28 Vcc 20 A 14 28 A 13 A 13 A 14 28 A 13 A 14 A 13 A 14 C E 7 19 D 7 18 D 6	V DD 1 OE 2 A 15 3 A 12 4 A 7 5 A 6 6 A 5 7 A 4 9 A 3 9 A 2 10 A 10
A 4 6 A 3 7 A 2 8 A 1 9	222 A 16 211 A 10 200 CE	
A 0 10 D 0 11 D 1 12	19 D 7 18 D 6 177 D 5	A 210 A 1111 A 012
V 25 (11)	11 D 3	0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1
		GND

Fig 1.2 HN62301P/HN27C301G Pin arrangement (Top view)

Page 12 - 3 HARDWARE

2. RAM CMOS 32 KB pseudo-stacic RAM is used. The standard arrangement of 64 KB with a maximum of 192 KB (every 64 KB additions are possible) using an additional RAM card brings the total of available memory to 256 KB. Standard arrangement CMOS 32 KB pseudo-static RAM µPD42832G x 2 RAM card CMOS 32 KB pseudo-static RAM uPD42832G x 6 RAM controls read/write and refresh by G.A. (gate array) for memory control. Memory backup is performed for all RAM to avoid data loss at power-off. (3) 7508 (slave CPU) 4 bit C-MOS CPU supports sleep timer functions. CLOCK = approx. 270 kHz Battery backup supports operation when the power switch is off. Use: Clock function Keyboard scan and control Power on/off. reset signal control, battery voltage monitor calendar, clock, alarm and timer function. (4) GAWIS (memory control G.A.) Control of pseudo-static RAM read/write and refiesh. Read signal control of the system ROM and application ROM. Timer interrupt Control of LCD unit (EHT-10/2) (5) GAPNIT (interrupt, timer control G.A.) Interrupt controller Timer & baud rate generator (with input capture function) Interface with slave CPU (7508) Barcode interface (6) GAPNIO (I/O control G.A.) ART (Asynchronous Receiver Transmitter) Cartridge interface RS-232C interface LED interface IC card interface (7) GAP10 (touch panel control G.A. : Installed only in EHT-10 Gate array for touch panel interface. (8) T6963 (LCD control installed only in EHT-10) LCD controller

```
(9) RS-232C interface
    Level : RS-232C level +-5 V
     Bit transfer rate: 110, 150, 200, 300, 600, 1200, 2400, 4800, 9600,
    19200, 38400, 75 (bps)
    Start bit :
                       1 bit
    Stop bit
                       1 or 2 bits
                :
     Parity
                       (Even, odd) parity/no parity
                 :
    Error check : Parity error, framing error, overrun error
Communication Type : Full duplex
(10) Input
     The touch panel for the EHT-10 and keyboard for the EHT-10/2 are
     available as input device. The major difference is as described below.
     1 Touch panel (EHT-10)
       Number of touch keys
                                   5x14=70
      Area of touch keys
                                   10.5x9.35 mm
    2 Keyboard (EHT-10/2)
      Number of keys
                                   34 keys
       3 built-in LEDs indicate key mode display
       7 character key buffer
     Common functions of the EHT-10/EHT-10/2 touch panel and keyboard are as
     follows.
       Auto repeat function
      Repeat time reset function
(11) LCD display
    LCD (liquid crystal) is used for the display of both the EHT-10 and
    EHT-10/2. The difference in LCDs for both devices is as described
    below.
     1 EHT-10
       1/48 duty liquid crystal
       154x84
                    Dot matrix
       Driver
                    X:T7778x4
                    Y: T6961x1
      Controller
                    T6963
       VRAM area 2 KB (independent of the memory in the main unit.)
       Drive voltage Variable from 12 V to 18 V volume Controls view angle
     2 EHT - 10/2
       1/32 duty liquid crystal
       120x32
                    Dot matrix
                    X:SED1180x2
       Driver
                    Y:SED1190X1
       Drive voltage Variable from 14 V to 19 V volume Controls view angle
       VRAM area
                    512 bytes (of main memory)
                    GAWIS (function of VRAM area redefinition and vertical
       Controller
                    dot scroll.)
```

### (12) Buzzer

Piezo-electric buzzer is used with the selection of three frequencies; 512 Hz, 1024 Hz, 2048 Hz. In addition to this, various frequencies can be generated by writing "1" or "0". Buzzer drive voltage is +12 V.

.

(13) IC card interface

Read/write data is enabled by the IC card based on ISO DP7816. Write/read voltage is 5 V single.

(14) Interrupt

Mode 2 interrupt is used to support 5 types of interrupt in a prioritized order.

#### 1.2 Address Map

The following 4 types of memory are placed in the memory address space of the EHT-10/EHT-10/2.

- (1) RAM (64 KB) : Standard in the main unit
- (2) System ROM : (128 KB)
- (3) Application ROM (32/E4/128 KB)
- (4) Expansion RAM card (Maximum of 192 KB)

As the RAM capacity is 64 K + 192 K = 256 KB, the 2-80 with an address capacity of 64 KB cannot read/write the entire memory. The EHT-10/EHT-10/2 incorporates bank switching to alleviate this, supporting a large memory. The bank switching I/O port is shown in the table below above the actual description.

R/W	1/0 Address	Register name	bit7	bit6	bit5	bit4	bit3	bit2	bit1	bit0	Notes
W	PO5H	BANKR	BANK 3	BANK 2	BANK 1	BANK O	0	0	0	0	
W	P22H	SBKR	APBN K1	APBI: KO	SYSB NK1	SYSB NKO	RAMB NK3	RAMB NK2	RAMB NK1	RAMB NKO	

### Table 1.1 Bank Switch I/O Port

Cire is required as there are 2 I/O ports for bank switching. Address map of EHT-10/EHT-10/2 changes according to the contents of the bank register.

### 1.2.1 Bank Switching

The memory concept divides the EHT-10/EHT-10/2 memory into 4, which gives an easy-to-operate environment to the OS and various application programs. These 4 environments are specified by I/O address PO5H [bank register] and are named by each value as shown below.

BANK3	BANK2	BANK1	BANKO	Name
0	0	x	0	System bank
0	1	x	0	Bank O
1	0	1	0	Bank 1
1	1	1	0	Bank 2

Either 1 or 0 can be entered in the boxes marked with an x. (Normally 0)

Table 1.2 Bank Name



(Notes)

RAM is divided into 8 areas every 32 KB and individually named.
 Only 8 KB from 32 KB of common RAM can be used in the system.
 System ROM of 128 KB is divided into 4 areas every 32 KB and named as sub-bank 0 to 3 as shown above.

#### Fig 1.3 System Bank Memory Structure

The figure above is the memory structure of the system bank, showing 32 KB of system ROM from address OOOOH to 7FFFH, and address 8000H to FFFFH is RAM area.

The system ROM is divided into 4 areas every 32 KB, when RAM is maximum (256 KB), it is divided into 8 areas every 32 KB.

The system ROM sub-bank O, common RAM and RAM sub-bank O are selected for the system bank when EHT-10/EHT-10/2 is reset or the system bank is specified for the bank register (PO5H).



(Note)

One from RAM #O to #6 is specified by sub-bank register [P22H]

Fig 1.4 Bank O Memory Structure

The figure above shows the memory structure when bank O. Bank O is sometimes called "all RAM mode" as the entire address space is RAM in the Z-80. Address OOOOH to 5FFFH and EOOOH to FFFFH is assigned to common RAM in this mode. One RAM (#O to #6), divided into a maximum of 7 every 32 KB, can be assigned to address space of 6000H to DFFFH of the Z-80. RAM is selected by the value of bit O to 3 (RAMBNK O to 3) of sub-bank register (P22H). RAM (#O to #6) selected by the value of RAMBNK O to 3 is shown in the next page.

RAMBNK3	RAMBNK2	RAMBNK1	RAMBNKO	Selected RAM
0	0	0	0	#0
0	0	0	1	#1
0	0	1	0	#2
0	0	1	1	#3
0	1	0	0	#4
0	1	0	1	#5
0	1	1	0	#6
0	1	1	1	This value
1	x	x	x	be set.

Table 1.3 sub-bank Selected by RAMBNK





1. RAM #O to #6 are not used

#### Fig 1.5 Bank 1 Memory Structure

The figure above shows the memory structure when bank 1. Address 0000H to 5FFFH and E000H to FFFFH is assigned to common RAM. One system ROM which has been divided into 3 every 32 KB can be assigned to the address space of 6000H to DFFFH. One sub-bank (1 to 3) is selected by the value of the sub-bank register (P22H) bit 4 or 5 (SYSBNK 0 or 1). The sub-bank selected by the value of SYSBNK 0 or 1 is shown in the table below. Both SYSBNK1 and SYSBNK0 cannot be 0 at the same time.

SYSBNK1	SYSBNKO	Selected sub-bank
0	0	This value should not be written
0	1	Sub-bank 1 (#1)
1	0	Sub-bank 2 (#2)
1	1	Sub-bank 3 (#3)

Table 1.4 Sub-bank Selected by SYSBNK



(Note)

RAM #O to #6 are not used

#### Fig 1.6 Bank 2 Memory Structure

The figure above shows the memory structure when bank 2. This structure is similar to bank 1 but the selected ROM is application ROM instead of system ROM. There are 3 application ROM types according to capacity; 32 KB, 64 KB and 128 KB. One sub-bank (0 to 3) is selected by the value of sub-bank register (P22H) bit 6 or 7 (APBNKO, 1). The sub-bank selected by the value of APBNKO or 1 is shown in the table below.

APBNK1	APBNKO	Selected sub-bank
0	0	Sub-bank 0 (#0)
0	1	Sub-bank 1 (#1)
1	0	Sub-bank 2 (#2)
1	1	Sub-bank 3 (#3)

Table 1.5 Sub-bank Selected by APBNK

HARDWARE Page 12 - 12

### 1.2.2 Notes on Bank Switching

When a user's program is operating in RAM or ROM of EHT-10 and EHT-10/2 and the bank is switched, the bank register of the program must be in common RAM. However, common RAM is normally controlled by OS and a user's program cannot occupy common RAM. Therefore, when a user changes banks, OS BIOS call is used. See "Software 4.2 BIOS Details" for further information.

CHAPTER 2 I/O Register Description

2.1 I/O Address Table

A device placed in the I/O address space of the EHT-10/EHT-10/2 consists of GAPNIT, GAPNIO, GAPAWIS and GAPIO of 4 gate arrays, LCD controller T6963, and I/O device cartridge option.

Table 2.1 shows the contents of the I/O address space of EHT-10 and EHT-10/2. Access to addresses for future use from POOH to P26H in the table is disabled. Correct access of disabled addresses cannot be guaranteed. Port addresses and the corresponding device names are written in the table.

40	Read	Write	0
res	(bit)	(bit)	vice
I/O Add	7 6 5 4 3 2 1 0	7   6   5   4   3   2   1   0	De
POOH	ICRL-C ICR - Low Command trigger	CTLR1 Control register	
	8-bit data	SRGJ SRGZ BRGT BRGD SWBCR BCR1 BCR0 1	
POIN	ICRH - C ICR - High Command trigger	CMDR Command register	
	8-bit data	RES SET OVF RDYSIO RDYSIO	1
PO2H	ICAL - 3 ICA - Low Bar code trigger	CTLR2 Control register 2	
	8-bit data	0 ELCM	
POSH	ICRH-3 ICR - High Bar code trigger		
	8-bit data		NIT
POAN	ISR Interrupt status register	IER Interrupt enable register	<b>JAP</b>
	INT4 INT3 INT2 INT3 INT0	IERA IERA IERA IERA IERA IERA	2
POSH	STR Status register	MAKR Bank register	BANIS.
	BANKS BANKS BANKS BANKS ROYSID ROY BCSD	BANKS ZANKS ZANKS ZANKS 0 0 1 0	Dud
POGH	SIOR Serial IO register	SIOR Serial IO register	0 A 9
Í	8-bit data	8-bit data	BANF
2078			E: 1
			NOT
POSH		VAOR VRAMStart address register	
		A14 A13 A12 A11 A10 A9	Í
POSH		YOFF Y offset register	118
		05P 11 13 12 11 10	GAW
POAN		FR Frame register	
		783 FR2 FR1 FR0	İ ı
PICH			
P11M			
P 1 III	CTG IF	CTG IF	0 I N
2124	(CTG interface )	( CTG interface )	3API
145	address area	address area	-
P179			
1.41			

Table 2.1 EHT-10/EHT-10/2 I/0 Address Table

_	Read	Write		
ess	(bit)	(bit)	ce	
I/O Addr	7   6   5   4   3   2   1   0	7   5   5   4   3   2   1   0	Devi	
01/14	ARTOIR ART data input register	ARIDOR ART data input register	1	
( lat	7-,8-bit data	7-,8-bit data	İ	
ne ful	ARISA ART status register	ARIMR ART mode register		
Phan	RDSR FE DE PE TA RAY RDY	STOP EVEN PEN DATA		
PIEH	10572 IO status register	ARTCS ART command register		
1	CEEL ACTS ACD ACD ICCLS PRUSY	RRIS ER SERK REE RDIR THE	CIN	
P173		ICTLR IC Card control register	GAP	
		PR         IC         BC         BC         BC         ICD18         ICC         ICCSC           LCD         IE         ITE         BC         BC         ICD18         ICC         ICCSC		
P188		SWR Switch register		
ĺ		6423 1423 6422 1422 0		
PISH		10CTLR IOControl register		
		SP LEER LEDT LEDO PINI ICK SLIN PF		
	LCCCOIR	LCCCDOR		
P20H	LCDC data input register	LCDC data output register		
ŀ	8-bit data	8-bit data		
	LCCCS TR	LCCCC3	1.6	
P2:8	LCDC status register	LCDC command register		
	8-bit data	8-bit data		
		SBKR Sub Bank register		
1221		APEN 1924 SYS SYS BAN RAN RAN RAN RAN RAN RAN RAN RAN RAN R	3	
:	153 Interrupt status register	CILR3 Control register	GAW	
238	IPSU OVF SY INT	SP1 SP0 CVFEN OVF PINT W/P OVF RESET		
		TPSC Touch panel scan register		
		0155 0 1 185 184 183 182 181		
258	TRAITTouch panel return register		010	
Í	8-bit data		01	
258	19212 Touch panel return register			
	6-bit data			

HARDWARE Page 13 - 3

### 2.2 I/O Register Details

(1) POOH (Read)

ICRL.C (Input Capture Register Low Command Trigger)

bit	Name	Function
7	ICR7	
6	ICR6	
5	ICR5	
4	ICR4	Low order 8 bits of Input Capture Register
3	ICR3	
2	ICR2	
1	ICR1	
0	ICRO	

<Description>

1. Low order 8 bits of Input Capture Register. The moment this register is read, the contents of FRC (Free Running Counter) is latched (both high and low orders) to ICR (Input Capture Register). The value of high order is gained by reading address POIH (ICRH.C).

2. This register is used when reading the current FRC.

3. FRC and ICR are described in "3.5 OVF, ICF Interrupt".

# (2) POOH (Write)

CRLR1 (Control Register 1)

bit	Name	Function
7	BRG3	
6	BRG2	
5	BRG 1	Baud rate generator select
4	BRGO	
3	SWBCR	Power switch of +5V for barcode =1 : Power ON =0 : Power OFF
2	BCR1 (UP)	
1	BCRO (DOWN)	Barcooe mode select
0		Always set to 1

:Description>

1. BRG3 to O specifies serial transfer baud rate, BCR1 to O specifies ICR trigger polarity.

RBG	Send		Receive	
3210	TXC	Bit rate	RXC	Bit rate
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1.74545 K 2.4 4.8 9.6 19.2 38.4 76.8 153.6 19.2 1.2 307.2 614.4 3.2	110 150 300 600 1200 2400 4800 9600 1200 75 19200 38400 200	1.74545 K 2.4 4.8 9.6 19.2 38.4 76.8 153.6 1.2 19.2 307.2 614.4 3.2	110 150 300 600 1200 2400 4800 9600 75 1200 19200 38400 200

## 2. Baud Rate Generator Select

### 3. BarCode Mode Select

BCR1	BCRO	Trigger polarity
0	0	Trigger disable
0	1	Fall trigger
1	0	Rise trigger
1	1	Rise and fall trigger

<Notes>

In EHT-10 and EHT-10/2 OS, this register stores write data to CTLR1 in the system area RZCTLR1 (FDD6H) in RAM to be used at renew. Thus, when writing to CTLR, the value of RZCTLR1 is rewritten. Bit configuration of RZCTLR1 is the same as CTLR1.

(Ex.) This register turns on bar code.

l.D	A, (RZCTLR1)
0r	081
LD	(RZCTLR1),A
OUT	(C1_R1).A

In OS, this register modifies the value of the baud rate (BRG3 to 0) by BIOS RSIOX, floppy disk or IC card access.

## (3) PO1H (READ)

ICRH.C (Input Capture Register High Command Trigger)

bit	Name	Function
7	ICR15	
6	ICR14	
5	ICR13	
4	1 <b>CR1</b> 2	High order 8 bits of Input Capture Register
3	1 <b>CR1</b> 1	
2	ICR10	
1	ICR9	
0	1 CR8	

# <Description>

This register is used when reading the current FRC. However, the values of ICR15 to ICR8 are latched at the moment ICRL.C (POOH) is read. Thus, ICRL.C must be the first one to read.

# (4) PO1H (Write)

# CMDR (Command Register

bit	Name	Function
7		
6		
5		Don't care.
4		
3		
2	RESOVF	<pre>=1 : Resets OVF interrupt generated by FR overflow. =0 : Nothing is performed.</pre>
1	RESRDYSIO	<ul> <li>=1 : Resets RDYSIO signal [P05H bit 3] used for communication with 7508.</li> <li>=0 : Nothing is performed.</li> </ul>
0	SETRDYSIO	<ul> <li>=1 : Sets RDYSIO signal used for communication with 7508.</li> <li>=0 : Nothing is performed.</li> <li>Normally this bit is not used.</li> </ul>

### <Notes>

In EHT-10 and EHT-10/2 OS, this register uses RES OVF in OVF interrupt process and uses RES RDYSIO in the communication process with 7580.

# (5) PO2H (Read)

ICRL.B (Input Capture Register Low Bar Code Trig	jer)	
--	------	--

bit	Name	Function
7	ICR7	
6	I CR6	
5	ICR5	
4	ICR4	Low order 8 bits of Input Capture Register at signal change.
3	I CR3	
2	ICR2	
1	ICR1	
0	I CRO	

<Description>

1. The value of the low order 8 bits of ICR which is latched from FRC to ICR by a signal change (rise and fall trigger) from the bar code reader.

2. INT2 signal (ICF) becomes active when the signal changes from the bar code reader.

# (6) PO2H (Write)

```
CTLR2 (Control Register 2)
```

bit	Name	Function
7		
6		
5		
4		
3		
2		
1		Always set to O
0	ELON	EL panel ON/OFF =0 : OFF =1 : ON

<Description>

This register controls on/off of the back light (EL panel) for the EHT-10/2 in software. However, the power switch must be at B.L position to operate.

## (7) PO3H (Read)

ICRH.B (Input Capture Register High Bar Code Trigger)

bit	Name	Function
7	ICR15	
б	ICR14	
5	ICR13	
4	1CR12	High order 8 bits of Input Capture Register at
3	ICR11	signal change
2	ICRIO	
1	ICR9	
0	ICR8	

constitute

1. The value of the high order 8 bits of ICR which is latched from FRC to ICR by a signal change (rise and fall trigger) from the bar code reader.

2. Reading this register resets INT2 signal (ICF) generated by change of signal from the barcode reader.

# (8) PO4H (Read)

## ISR (Interrupt Status Register)

bit	Name	Function
7		
6		Always set to O
5		
4	INT4(EXT)	<pre>Interrupt signal from the cartridge, IC card, or timer of 1 msec or 8 msec. =1 : Interrupt request =0 : No interrupt request</pre>
3	INT3(OVF)	Interrupt signal generated by FRC overflow. Write 1 to RESOVF [PO1H bit 2] to reset. =1 : Interrupt request =0 : No interrupt request
2	INT2(ICF)	This interrupt signal is generated at the same time when the value of FRC is latched to ICR by a change (rise and fall) from the bar code signal. However, this interrupt is not generated when both BCR1 and BCRO [POOH] are O. Read ICRH.B [PO3H] to reset. =1 : Interrupt request =0 : No interrupt request
1	INT1(ART)	Interrupt signal generated when ART RXRDY=1 (receive data enters). Read receive data from ART to reset. =1 : Interrupt request =0 : No interrupt request
0	INTO(7508)	Interrupt signal generated from 7508. Respond to 7508 to reset. =1 : Interrupt request =0 : No interrupt request

<Description>

INT4 to INTO can be read when an interrupt is masked. INTO has the highest priority and INT4 the lowest priority of interrupt priority order.

See "Chapter 3 Interrupt Description" for details.

HARDWARE Page 13 - 12

## (9) PO4H (Write)

## IER (interrupt Enable Register)

bit	Name	Function
7		
6		Don't care.
5		
4	IER4(EXT)	INT4 (EXT) interrupt =1 : Enable =0 : Disable
3	IER3(OVF)	INT3(OVF) interrupt =1 : Enable =0 : Disable
2	IER2(ICF)	INT2 (ICF) interrupt =1 : Enable =0 : Disable
1	IER1(ART)	INT1 (ART) interrupt =1 : Enable =0 : Disable
0	IERO(7508)	INTO (7508) interrupt =1 : Enable =0 : Disable

<Description>

Each bit corresponds to an interrupt and sets enable/di.zble.

<Notes>

In EHT-10 and EHT-10/2 OS, this register saves write data to IER in the system area RZIER (F42EH) in RAM to be used at renew. Thus, when writing to IER, the value of RZIER is rewritten. Bit configuration of RZIER is the same as IER.

(Ex.) This register enables EXT interrupt.

DI		
LD	A, (RZ!ER)	RZIER: (F42EH)
OR	10H	. ,
LD	(RZIER), A	
OUT	(IER), 4	
EI		

In EHT-10 and EHT-10/2 OS, this register supports masked interrupt control using BIOS MASKI. At normal state, 7508 and OVF interrupts are enabled, ART interrupt is enabled when OPEN is executed at BIOS RSIOX, it is disabled when CLOSE is executed. ICF interrupt is disabled. EXT can be masked individually by an interrupt factor.

Interrupt from cartridge -> ICCTLR [P17H bit 6]
Interrupt from IC card -> ICCTLR [P17H bit 5]
1 msec/8 msec timer interrupt -> [P23H bit 5]

# (10) POSH (Read)

STR (Status Register)

bit	Name	Function
7	BANK3	
6	BANK2	Main memory bank register
5	BANK1	
4	BANKO	
3	RDYSIO	This serial bus control signal is an I/F with the 7508. =1 : Access enable to 7508 =0 : Access disable to 7508
2	RDY	RDY signal from 7508. Not used normally.
1	BCRD	Data input signal of bar code reader.
0		Don't care.

~Description>

See "1.2 Address Map" for the value of BANK 3 to 0.

:Notes>

1. Value of RDYSIO is 1 directly after power on. RDYSIO signal must be reset using RESRDYSIO [PO1H bit 1] when data or command is set using SIOR [PO6H] or after data is received.

2. Signal from the bar code reader can be read through BCRD directly.

## (11) PO5H (Write)

### BANKR (Bank Register)

bit	Name	Function
7	BANK3	
6	BANK2	Main memory bank register
5	BANK1	
4	BANKO	
3		
2		Always set to u
1		Always set to 1
0		Always set to O

-

<riescription>

See "1.2 Address Map" for values of BANK 3 to 0.

< res>

1. In EHT-10 and EHT-10/2 OS, this register saves write data to BANKR in system area RZBANKR (F42CH) in RAM to be used at renew. Thus, when writing to BANKR, the value of RZBANKR is rewritten. Bit configuration of RZBANKR is the same as BANKR.

2. LOADX, STORX, LDIRX, JUMPX and CALLX are provided in EHT-10 and EHT-10/2 OS as BIOS for bank control.

# (12) PO6H (Read/Write)

.

```
SIOR (Serial I/O Register)
```

bit	Name	Function
7	\$107	
6	S106	
5	\$105	
4	\$104	Register for 7508 data send/receive.
3	\$103	
2	\$102	
1	SI01	
0	S100	

<Description>

At read : A bit data which is received after parallel conversion of serial data transferred from 7508.

At write : 8 bit gata serial-transferred to 7508.

HARDWARE Page 13 - 17

.

## (13) PO8H (Write)

bit	Name	Function
7		Don't care.
6	A14	
5	A13	
4	A12	VOAN start address
3	A11	VKAM SLATL AUUTESS
2	A10	
1	A9	
0		Don't care.

VADR (VRAM Start Address Register (EHI-10/2 only)

## <Description>

VADR is the register to specify the VRAM area and specifies the high order 6 bits of the address. Due to this, 512 bytes from (A14, A13, A12, A11, A10, A9 0 0000 0000) to (A14, A13, A12, A11, A10, A9 1 1111 1111) are assigned as the VRAM area. VRAM address can be set in 512 byte units.

#### <Notes>

VRAM address must be fetched within OOOOH to 5FFFH and EOOOH to FFFFH.

## (14) PO9H (Write)

```
YOFF (Y Offset Register) (EHT-10/2 only)
```

bit	Name	Function		
7	DSP	This register controls the LCD panel display. =1 : Screen display =0 : No screen display		
6	<b></b>			
5		Don't care.		
4	Y4			
3	Y3			
2	Y2	Offset value in Y direction		
1	Y1			
0	YO			

### <Description>

This register specifies the corresponding relation of VRAM and LCD in the vertical direction. Display starts at the point moved by YOFF dot of VRAM. When the display reaches the bottom of VRAM, it returns to the top of VRAM and displays YOFF=1 dot line to end a screen.

## <Notes>

EHT-10/2 OS uses Y Offset Resister scrolling LCD screen vertically. The value of the current YOFF is saved in LVRAMYOFF (F266H).

# (15) POAH (Write)

```
FR (Frame Register) (EHT-10/2 only)
```

bit	Name	Function
7		
6		
5		juon t care.
4		
3	FR3	
2	FR2	Enome Desister
1	FR1	
0	FRO	

<Cescription>

This register regulates the LCD frame frequency. The relation of Frame Register and frame frequency is as listed below.

FR				LCD frame frequency (Hz)
FR3	FR2	FR1	FRO	3.68 M Hz
0 0 0 1 1 1 1 1	1 1 1 0 0 0 0 0	0 0 1 1 0 0 1 1 0 0	0 1 0 1 0 1 0 1 0	450 360 300 257 225 200 180 164 150 138
i	i	1	Ô	128

(0000) to (0011) and (1111) are disabled.

<Notes>

Fixed value OEH is set to FR in EHT-10 and EHT-10/2 OS at power on, reset or system initialization.

# (16) P10H to P13H

# <Description>

PlOH to Pl3H is address space for the cartridge interface, the configuration differs depending on cartridge mode (HS, DB, IO and OT mode).

See "4.1 Cartridge Interface" for further information about P10H to P13H.

## (17) P14H (Read)

.

ARIDIA (ARI DUCU INPUC REGISCEL	ARTDIR (	ART	Data	Input	Register
---------------------------------	----------	-----	------	-------	----------

bit	Name	Function
7	RD7	
6	RD6	
5	RD5	
4	RD4	
Ē	RD3	
2	RD2	
1	RD1	
0	RDO	

#### <Description>

Serial data input from RXD is parallel-converted to be fetched. When the data length is 7 bits, 0 is input to bit 7 (RD7).

#### <Notes>

1. In EHT-10/EHT-10/2 OS, Interrupt routine processes data receive and BIOS RSIOX interfaces with application of receive data.

2. EHT-10 and EHT-10/2 has 3 serial interfaces; RS-232C, IC card reader and cartridge interface. However, the register for send/receive data has only one send/receive data register. Thus, you must select which interface to use this register for before register access. SWR (Switch Register P18H) can be used for this occasion.

## (18) P14H (Write)

ARTDOR	(ART	Data	Output	Register)
--------	------	------	--------	-----------

bit	Name	Function
7	TD7	
6	TD6	
5	TD5	
4	TD4	
3	TD3	Send data
2	TD2	
1	TD1	
0	TDO	

<Description>

This data is converted to serial data and is output from TXD. When the data length is 7 bits, bit 7 (TD7) is "Don't care".

## <Notes>

1. BIOS RSIOX interfaces with the application for serial data send process in EHT-10 and EHT-10/2 OS.

2. See (17) <Notes>.

## (19) P15H (Read)

ARTSR (ART Status Register)

bit	Name	Function
7	RDSR	DSR (Data Set Ready) signal. When RS-232C DSR terminal is active, RDSR=1.
6		Always O
5	FE	Framing error (occurs when FE=1).
4	OE	Overrun error (occurs when OE=1)
3	PE	Parity error (occurs when PE=1).
2	TXEMP	This register indicates no data in the send section. It is set when send data buffer ARTDOR and the parallel/serial transducer are both empty.
1	RXRDY	When RXRDY=1, INT1 (ART) interrupt is requested to CPU (2-80) and it indicates receive data is accepted. RXRDY is reset when the receive buffer ARTDIR (P14H) is read, it is also reset by reset input or error reset command.
0	TXRDY	This register is set when the output buffer ARTDOR is empty. It is reset when data is written to the output buffer ARTDOR.

<Description>

This register is equal to Status Register of 8251 (USART).

FE (bit 5) Framing error does not effect data receive operation, if the next data is fetched sequentially, framing error is checked for new data. FE is reset if the stop bit is correct.

OE (bit 4) Overrun error does not effect data receive operation, OE is not reset even if next data is fetched correctly. Error reset command (ER=1) or reset input is necessary for OE reset.

PE (bit 3) PE reset condition is the same as FE. Parity is checked only when PEN=1. PE is 0 when PEN=0.

## (20) P15H (Write)

### ARTMR (ART Mode Register)

bit	Name	Function				
7	STOP	<pre>Specifies the number of stop bits. =1 : 2 bits =0 : 1 bit</pre>				
6		Don't care.				
5	EVEN	<pre>Specifies parity Even/Odd. (Valid only when PEN=1.) =1 : Even parity =0 : Odd parity</pre>				
4	PEN	Specifies Parity/No parity. =1 : Parity =0 : No parity				
3		Don't care.				
2	DATA LENGTH	Specifies data length. =1 : 8 bit =0 : 7 bit				
1						
0		Don't care.				

### <Description>

This register is equal to Mode Instruction Register of 8251 (USART).

#### <Notes>

In EHT-10 and EHT-10/2 OS, this register saves write data to ARTMR in the system area RZARTMR (FOD9H) in RAM to be used at renew. Thus, when writing to ARTMR, the value of RZARTMR is rewritten. Bit configuration of RZARTMR is the same as ARTMR. Output to ARTMR is supported by BIOS RSIOX. Set TXE[P16H bit 0 W] and RXE[P16H bit 2 W] to 0 in advance to renew this register.
# (21) P16H (Read)

IOSTR	(10	Status	Register)
-------	-----	--------	-----------

bit	Name	Function
7		Don't care.
6	CSEL	Signal to identify the cartridge option type. =1 : HS (Hand Shake) mode =0 : Other mode
5	RCTS	RS-232C CTS signal This signal becomes 1 when RS-232C CTS signal is active.
4	RCD	RS-232C CD signal This signal becomes 1 when RS-232C CD signal is active.
3	RXD	Serial data input signal.
2	ICCLS	This signal becomes 0 when excess current goes through the IC card or IC card reader covar is open.
1		Don't care.
0	PBUSY	Interrupt request from the cartridge option can be checked using the cartridge interface CINT signal. =0 : Interrupt request from cartridge option =1 : No interrupt request from cartridge option

#### <Notes>

1. See "4.3.7 Notes" for RCTS, RCD and RXD signals. Serial data from RS-232C, IC card and cartridge interface can be directly read using RXD.

2. See "3.6 EXT Interrupt" for ICCLS and PBUSY.

# (22) P16H (Write)

ARTCR	(ART	Command	Register)	
-------	------	---------	-----------	--

bit	Name	Function	
7			
6		Don't care.	
5	RRTS	RS-232C RTS signal RS-232C RTS signal becomes active when RRTS=1.	
4	ER	OE, FE and PE are reset. ER is set to 1 when RXE=1. ER is pulse output, pulse generates only while writing with ER=1. Therefore, ER=0 does not need to be returned after ER=1.	
3	SBRK	Break output TXD is forced set to 0 when SBRK=1. (Valid when TXE=1)	
2	RXE	This enables data receive. =1 : Enable =0 : Disable	
1	RDTR	RS-232C DTR signal RS-232C DTR signal becomes active when RDTR=1.	
0	TXE	This enables data send. =1 : Enable =0 : Disable TXD=1 (mark) while TXE=0	

### <Description>

Bit 0 to 5 of this register is equal to Command Instruction Register of 8251 (USART).

#### <Notes>

In EHT-10 and EHT-10/2 OS, this register saves write data to ARTCR in the system area RZARTCR (FODAH) in RAM to be used at renew. Thus, when writing to ARTCR, the value of RZARTCR is rewritten. Bit configuration of RZARTCR is the same as ARTCR. Data output to ARTCR is supported at BIOS RSIOX.

# (23) P17H (Write)

# ICCTLR (IC Card Control Register)

bit	Name	Function
7	PLCD	LCD power supply =0 : Supplied =1 : Not supplied
6	PRIE	Signal from cartridge I/F CINT =0 : Enable =1 : Disable
с'n	ICITE	IC card interrupt enable =0 : Disable =1 : Enable
4	DCTG	Development cartridge switch =0 : Normal operation =1 : Development cartridge access enable
3	BCRP	Par code reader LED control =0 : UFF =1 : ON
2	ICDIR	IC card Read/Write switch =0 : Write =1 : Read
1	ICC	IC card power control =0 : OFF =1 : ON
0	ICOSC	IC card clock control =0 : Oscillation stops =1 : Oscillation

# <Notes>

1. As this register is not effected by the reset signal, 54H must be set in the initialize routine.

2. Hold 15 msec from the IC card clock oscillation start till it becomes stable.

1

# (24) P18H (Write)

SWR (Switch Register)

bit	Name	Function	
7			
6		Don't care.	
5			
4		Always set to O	
3	SSW1	Conial and	
2	SSWO	Serial mode	
1	CSW1		
0	CSWO	lartridge interface mode	

<Description>

This register sets the serial interface mode and cartridge incerface mode.

Serial mode

SSW1	SSWO	RXD	TXD
0	٥	Cartridge	Cartridge
0	1	IC card	IC card
1	0	RS-232C	RS-232C
1	1	This value sho	ould not be set.

Cartridge interface mode

CSW1	CSWO	Mode	
0	0	HS (Hand Shake) mode	
0	1	10 (Input Output) mode	
1	0	DB (Data Bus) mode	
1	1	OT (Output port) mode	

In EHT-10 and EHT-10/2 OS, this register saves write data to SWR in the system area RZSWR (FODCH) in RAM to be used at renew. Thus, when writing to SWR, the value of RZSWR is rewritten. Bit configuration of RZSWR is the same as SWR.

(Ex.)

This register switches to IO mode.

LD A, (RZSWR) RZSWR : (FODCH) AND OFCH SWR : (18H) OR O1H LD (RZSWR), A OUT (SWR), A

Serial mode switching is supported at BIOS RSIOX.

# (25) P19H (Write)

bi+	Namo	
7	SP	Output signal to speaker. Set SP=0 when not used.
6	LED2	
5	LEDI	LED output port
4	LEDO	
3	PINI	Cartridge set signal This signal is used to reset the cartridge for software. Cartridge is reset when PINT=0.
2	ICR	IC card reset signal =1 : Reset ON =0 : Reset OFF
1	SLIN	This signal controls cartridge interface CCTL1. Printer select-in output. Select-in when SLIN=0.
0	PF	This signal controls cartridge interface CCTLO. This form-feeds the printer.

#### <Notes>

1. In EHT-10 and EHT-10/2 OS, this register saves write data to IOCTLR in the system area RZIOCTLR (FODDH) in RAM to be used at renew. Thus, when writing to IOCTLR, the value of RZIOCTLR is rewritten. Bit configuration of RZIOCTLR is the same as IOCTLR. In OS, SP is used in BIOS BEEP process, LED 2 to 0 are used in key input process. Control LED 2 to 0 is supported by BIOS CONOUT (ESC+AOH to A5H)

2. PINT, SLIN and PF are general purpose outputs and can be used when a user created cartridge is used. The functions mentioned above are enabled only when a dedicated printer for EHT-10 and EHT-10/2 is used.

3. When form-feed is performed in the printer, set SLIN to disable, PF to enable; SLIN=1, PF=1. See "4.1 Cartridge Interface" if you would like to use PINT, SLIN or PF for general purpose output.

# (26) P20H (Read) (EHT-10 only)

LCDCDIR	(LODC	Data	Input	Register	)
---------	-------	------	-------	----------	---

bit	Name	Function
7		
6		
5		
4		Data mod monister from LCD controller 16963
3		
2		
1		
٥		

<netcription>

This register is to read data from LCDC controller T6963.

1. This register is valid for EHT-10 only.

2. Check LCDCSTR[P21H] status bit whether it can be read before data read from this register.

# (27) P20H (Write) (EHT-10 only)

LCDCDOR (LCDC Data Output Register)

bit	Name	Function
7		
6		
5		
4		
ū		Data write register from LLD controller 16963
2		
1		
0		

<Description>

This register is to write data to the LCD controller T6963.

<Notes>

1. This register is valid for EHT-10 only.

2. Check LCDCSTR[P21H] status bit whether it can be written before data write to this register.

### (28) P21H (Read) (EHT-10 only)

# LCDCSTR (LCD Status Register)

bit	Name	Function
7	STA7	To check the blink state (Approximate cycle 1 second duty 50%) =1 : Normal display =0 : OFF
6	STA6	If a point other than on the real screen is set while copy, the flag is set. If the instruction is executed, the flag is reset.
5	STA5	To check whether controller is operatable. =1 : Operatable =0 : Not operatable
4	STA4	Don't care.
3	STA3	To check whether data can be written (Valid only when AU10). =1 : Write enable =0 : Write disable
2	STA2	To check whether data can be read (Valid only when AUTO). =1 : Read enable =0 : Read disable
1	STAL	To check whether data can be written or read. =1 : Data enable =0 : Data disable (during internal process)
0	STAO	To check whether the instruction is executable. =1 : Executable ≈0 : Not executable (during instruction execution).

# <Description>

1. LCDC status can be read using this register. Check for STAD=STA1=1 when writing a command to LCDC [P21H W] for timing. Theck for STAD==STA1=1 also when writing data to LCDC.

2. When T6963 (LCDC) is in AUTO mode, it can be performed only with identification of STA2 or STA3.

(29) P21H (Write) (EHT-10 only)

```
LCDCCR (LCDC Command Register)
```

bit	Name	Function
7	LCDCMD7	
6	LCDCMD6	
5	LCDCMD5	
4	LCDCMD4	
3	LCDCMD3	Command Register to LCD controller 16963.
2	LCDCMD2	
1	LCDCMD1	
0		

<Description>

This register gives command to the LCDC controller T6963.

<Notes>

1. This register is valid for EHT-10 only.

2. Check whether command write is enabled with LCDCSTR[P21H] status bit before command is given to this register.

See "4.6.4 T6963 Command" for T6963 command table.

# (30) P22H (Write)

### SBKR (SUBBANK Register)

bit	Name	Function
7	APBNK1	Application ROM sub-bank
6	APBNKO	= 00 : SUD-Dank 0 = 01 : sub-bank 1 = 10 : sub-bank 2 = 11 : sub-bank 3
5	SYSBNK1	System ROM sub-bank
4	SYSBNKO	= 00 : sub-bank 0 = 01 : sub-bank 1 = 10 : sub-bank 2 = 11 : sub-bank 3
3	RAMBNK3	RAM sub-bank
2	RAMBNK2	= 0000 : sub-bank 0 = 0001 : sub-bank 1 = 0010 : sub-bank 2
1	RAMBNK1	= 0010 . Sub-bank 2 = 0011 : sub-bank 3 = 0100 : sub-bank 4
0	RAMBNKO	= 0100 : sub-bank 5 = 0111 : sub-bank 6 = 0111 : sub-bank 7

#### <Description>

This register switches memory space for the CPU (Z-80). See "1.2 Address Map" for further information.

### <Notes>

1. This register is not affected by reset.

2. In EHT-10 and EHT-10/2 OS, this register saves write data to SBKR in the system area RZSBBNKR (F42DH) in RAM to be used at renew. Thus, when writing to SBKR, the value of RZSBBNKR is rewritten. Bit configuration of RZSBBNKR is the same as SBKR.

# (31) P23H (Read)

ITSR (Interrupt Status Register)

bit	Name	unction					
7							
6		Don't care.					
5							
4							
3							
2							
1	IPBUSY	Interrupt request signal from IC card and cartridge option. =1 : No interrupt request =0 : Interrupt request					
ð	OVFINT	<pre>Interrupt request signal from timer of 1 msec or 8 msec. =1 : Overflow (interrupt request) =0 : No overflow (no interrupt request)</pre>					

# <Des:ription>

1. See "Chapter 3 Interrupt Description" for OVFINT signal.

2. Interrupt from IC card and cartridge option, and timer interrupt of 1 msec and 8 msec cause INT4(EXT) interrupt. Even when INT4 (EXT) interrupt mask [PO4H bit 4] is performed, interrupt can be checked using this status bit IPBUSY or OVFINT.

# (32) P23H (Write)

CTLR3 Control Register 3

bit	Name	Function
7	SP1	Speaker ON/OFF and frequency switching.
6	SPO	=00 : 0FF =10 : 1024 H2 =01 : 512 Hz =11 : 2048 Hz
5	OVFEN	1 msec, 8 msec interrupt enable =0 : Disable =1 : Enable
4	OVFITV	1 msec, 8 msec interrupt frequency select =0 : 8 msec =1 : 1 msec
3	PINTDIS	External interrupt enable =0 : Enable =1 : Disable
2	W/P	Always set to 1
l	OVFRESFT	1 msec, <sup>A</sup> msec timer interrupt reset =1 : Reset
0		Don't care.

#### <Description>

1. SPO, SP1 Any of the 3 types of frequencies can be selected by changing this value. In this case, set SP[P19H bit 7] to 0. When the signal is output to the speaker using SP, set SPO and SP1 to 0.

2. OVFRESET Set this bit to 1 to reset timer interrupt (when OVFEN=1). When OVFRESET=1, write pulse generates, OVFRESET=0 does not need to be returned after OVFRESET=1.

### <Notes>

These registers become all O after reset. In EHT-10 and EHT-10/2 OS, this register saves write data to CTLR3 in the system area RZCTLR3 (FODEH) in RAM to be used at renew. Thus, when writing to CTLR3, the value of RZCTLR3 is rewritten. Bit configuration of RZCTLR3 is the same as CTLR3.

### (33) P24H (Write)

TPSC (Touch Pane	l Scan Register)
------------------	------------------

bit	Name	Function						
7	DISC	scharge of the touch panel charge						
6		Don't care.						
5		Always set to O						
4	TR5							
3	TR4							
2	TRS	Touch panel scan output data						
1	TR2							
0	TR1							

<Description>

1. When the touch panel scan output data is entered TR1 to TR5, the bit is set to P25H or P26H corresponding to the pressed key. This snows which touch panel has been pressed.

Data to set

- (a) When interrupt wait Set TR1 to TR5 to 1. DATA 10011111
- (b) When scan Set DISC to 1 and SET one bit of TR1 to TR5 to 1. DATA 100 ....
- (c) When discharge Set all to 0 DATA 00000000

When scan, set one bit of TR1 to TR5 to 1 and set the remaining bits to 0. However, when writing 0 to this register, the corresponding bit does not become 0 but becomes high impedance, thus, the charge remains in polarity. To discharge this, the operation in item (c) mentioned above is necessary.

See "4.5 Touch Panel I/F" for touch panel scan flow.

# (34) P25H (Read)

TPRT1	(Touch	Panel	Return	Register	1)	
-------	--------	-------	--------	----------	----	--

.

bit	Name	Function
7	TC8	
6	TC7	
5	TC6	
4	TC5	Touch ascal mature register 1
3	TC4	j louch panel recurn register i
2	ТСЗ	
1	TC2	
0	TC1	]

<Description>

See "4.5 Touch Panel I/F" for TC1 to TC8.

# (35) P26H (Read)

TPRT2 (Touch Panel Return Register 2)

bit	Name	unction					
7							
6		uon't care.					
5	TC14						
4	TC13						
3	TC12	Touch and a second star O					
2	TC11	Touch panel return register 2.					
1	TC10						
0	TC9						

<Description>

See "4.5 Touch Panel I/F" for TC9 to TC14.

(1) I/O register initial reset

Some of the control registers included in EHT-10 and EHT-10/2 gate array set output directly after reset to 0 using a method to reduce the number of gates.

As shown in figure 2.1, mask register output with flip-flop to set output to 0 even if the contents of the register is unstable, in order to control the register output directly after the reset.

Register output mask release after the CPU starts and the contents of the register initialized by a program is described (See Figure 2.1). By doing this, register output operates similar to register reset. Hereafter this method is called pseudo-reset.

The EHT-10 and EHT-10/2 register has 3 types of state including this pseudo-reset at reset.

- 1. Pseudo-reset state
- 2. Reset state
- 3. State which is not reset and unstable



Fig 2.1 Pseudo-Reset

1. Pseudo-reset register bits



The 3 registers mentioned above have output masked with the same output control flip-flop directly after reset and the mask is released by writing to CTLR1 [POOH].



Mask release procedure



The 4 registers mentioned above have output masked with the same output control flip-flop directly after reset and the mask is released by writing to ARTMR [P15H].



## Mask release procedure

\*: Value which makes the control signal DISABLE.

2. Reset register bits



FR [POAH]			F3	F2	F1	FO

ARTDR [P14H]

7/8 Transmit / Receive DATA

ICCTLR [P17H]	PWLCD	PRIE	ICITE	DCTG	BCRP	ICDIR	100	ıcosc

(2) Output to I/O port

When EHT-10 and EHT-10/2 outputs data to the I/O port, it saves the contents of the current output I/O register in RAM so that a part of the I/O register can be modified and the I/O port state can be recovered at power on. Thus, when data is output to the I/O port, the contents of RAM must be modified at the same time. Relation of I/O register and area in RAM is as listed below.

1/0	address	Register name	RAM address	Variable	Notes
	РООН	CTLR1	FOD6H	RZCTLR1	
	P04H	IER	F42EH	RZIER	DI state at modification
	P05H	BANKR	F42CH	RZBANKR	Further processing for actual bank switching.
	P15H	ARTMR	FOD9H	RZARTMR	
	P16H	ARTCR	FODAH	RZARTCR	
	P18H	SWR	FODCH	RZSWR	
	P19H	IOCTLR	FODDH	RZIOCTLR	
	P22H	SBKR	F42DH	RZSBBNKR	
	P23H	CTLR3	FODEH	RZCTLR3	

# CHAPTER 3 Interrupt Description

3.1 Overview

5 types of interrupt are supported in the EHT-10/EHT-10/2. When an interrupt is issued, the vector is set corresponding to the interrupt vector and control moves to the interrupt routine. CPU mode 2 is used for the interrupt, EHT-10/EHT-10/2 OS sets the interrupt vector at system initialization, reset, and power- on. Interrupts the EHT-10/EHT-10/2 supports are as listed below.

(1) 7508 (4 bit CPU)
(2) ART (RXRDY)
(3) ICF (Input Capture)
(4) OVF (FRC overflow)
(5) EXT (External)

Factors of 7508 (4 bit CPU) interrupt are as listed below.

Keyboard (EHT-10/2)
 Touch panel (EHT-10/2)
 Alarm time
 Power failure
 Power switch ON/OFF
 One second interrupt

Tantors of EXT interrupt are as listed below.

- (1) Cartridge option
- (2) IC card reader
- (3) 1 msec, 8 msec timer

#### 3.2 Interrupt Vector

The EHT-10/EHT-10/2 has 5 types of interrupt, the interrupt vector Table is located at address FFFOH to FFFFH. The interrupts are prioritized, requested interrupts are accepted following the priority order. I/O address PO4H [IER : Interrupt Enable Register] can disable/enable interrupts, when an interrupt is disabled, the interrupt request status can be checked by reading I/O address PO4H [ISR: Interrupt Status Register].

Priority order	Interrupt cause	Vector	Address in RAM	IER	ISR	NAME
1 (Highest priority)	7508 (4bit CPU)	FOH	FFFOH, FFF1H	IERO	ISRO	INTO
2	ART (KXRDY)	F2H	FFF2H,FFF3H	IER1	ISR1	INT1
3	ICF (Input Capture)	F4H	FFF4H,FFF5H	IER2	ISR2	INT2
4	OVF (Overflow of FRC)	F6H	FFF6H,FFF7H	IER3	I SR3	INT3
5	EXT (timer, cartridge, and IC card)	F8H	FFF8H, FFF9H	IER4	ISR4	INT4

# Tab'e 3.1 Interrupt Priority Order

Write 0 to the corresponding IER for interrupt disable and write 1 for interrupt enable. "1" is set to each ISR at interrupt.

#### 3.3 7508 Interrupt

Factors of 7508 (4 bit CPU) interrupts are as listed below.

- (1) At key input (EHT-10/2)
- (2) When the touch panel is pressed (EHT-10)
- (3) One second interval interrupt
- (4) At power failure (when main battery is discharged)
- (5) Specified alarm time
- (6) At power switch ON/OFF

Read status (command code 2) is issued to 7508 to check the factor of interrupt, the interrupt factor is analyzed and interrupt process is performed in the interrupt routine. See "Software Version 7.6 7508 Command" for interface and command of 7508 and CPU.



Fig. 3.1 ART Interrupt Generation Block

RS-232C interface is built-in the EHT-10/EHT-10/2, it issues ART (RXRDY) interrupt when receive data enters. When the interrupt is not used, write "O" to IER1 so that the interrupt can be masked and the status can be read by ISR and RXRDY. Interrupt signal reset is performed when I/O port P14H is read (receive data is read), 1 is written to error reset command ER or reset signal is input.

3.5 OVF Interrupt, ICF Interrupt

ICF is set when the input signal changes from the barcode reader and is reset when input capture register [ICRH.B: PO3H] is read.

OVF is set when FRC (Free Running Counter) overflows and is reset by RESOVF command [CMDR: Write 1 to PO1H bit 2]. As FRC is 16 bits and the input clock is 614.4 kHz, frequency is approx. 106.67 msec.

The interrupts are enabled when IER [PO4H] corresponding bit is set to 1 and are disabled when it is set to 0. Interrupt request can be checked by reading ISR [PO4H] regardless of the value of IER. While the value of IER is rewritten, the CPU must be in the interrupt disable state.

IER is initial-reset. Neither ICF nor IVF is initial-reset. Input capture register ICR contains 16 bits and FRC of 16 bits is fetched. There are 2 types of timing to fetch. One is according to a read instruction, when ICRL.C [POOH] read instruction is issued, FRC of 16 bits at that time is latched. This operation is used when the value of FRC is read. Read ICRH.C [PO1H] to read the remaining high order 8 bits. The other is according to the barcode. When the bar code signal changes, ICR latch pulse generates and FRC 16 bits are fetched to ICRL.B [PO2H], ICRH.B [PO3H]. ICF is set at this time. ICF is reset at ICRH.B read. The bar code signal is directly read by reading STR [PO5H] bit 1.

Figure 3.2 shows OVF, ICF interrupt generation block.



Fig 3.2 OVF, ICF Interrupt Generation Block

3 factors of EXT interrupt are as listed below.

- (1) Interrupt from cartridge option (CINT signal)
- (2) Interrupt from IC card reader
- (3) 1 msec, 8 msec timer interrupt

These interrupts are logical-added (ORed) and become one EXT interrupt. They can be individually enabled or disabled. Even when an interrupt is disabled, reading the I/O port status can check the interrupt request. Figure 3.3 shows EXT interrupt block and Table 3.2 describes the terminals.

(1) IC card reader

When excess current goes through the IC card or IC card cover is open, the IC card interrupt is issued. This state reflects ICCLS [P16H bit 2]. When this interrupt is not required, write "O" to ICITE [P17H bit 5]. Turn off IC card power supply or close the IC card cover to release the interrupt signal.

(2) Cartridge option

The EHT-10/EHT-10/2 accepts interrupts from the cartridge option. Set PRIE [P17H bit 6] to "O" so that this interrupt is accepted. This interrupt signal reflects PBUSY [P16H bit 0]. Ways of releasing the interrupt signals are different according to the cartridge option.

(3) Timer

When 1 msec or 8 msec is counted by the timer, the timer interrupt is issued. Interrupt of 1 msec or 8 msec is selected by OVFITV [P23H bit 4], writing "1" for 1 msec, "0" for 8 msec. Write "0" to OVFEN [P23H bit 5] when this interrupt is not required. Write "1" to OVFRESET [P23H bit 1] to release the interrupt.



Fig 3.3 EXT (External) Interrupt Block

Terminal	Description	1/0 port	R/W
IC card interrupt	This becomes 1 when IC card cover is open or excess current goes through the IC card.		-
ICCLS	IC card interrupt signal	P16H bit2	R
	ICCLS=IC card interrupt		
ICITE	IC card interrupt mask.	P17H bit5	W
CINT	Interrupt signal from cartridge option.		-
2Bij2A	CINT signal can be read.	P16H bit0	R
PRIE	CINT interrupt signal mask.	P17H bit6	W
OVFITV	Frequencies of interrupt 1 msec and 8 msec are set.	P23H bit4	W
OVFRESET	Interrupts of 1 mscc and 8 msec are resei.	P23H hitl	Ħ
IPBUSY	IC card interrupt and CINT interrupt status can be read.	P23H bitl	R
OVFINT	Interrupt status of 1 msec and 8 msec can be read.	P23H bitO	R
PINTDIS	Mask of IC card interrupt and CINT interrupt.	P23H bit3	W
OVFEN	Mask of interrupt 1 msec and 8 msec.	P23H bit5	W
ISR	Signals of IC card interrupt, CINT interrupt, 1 msec and 8 msec interrupt.	PO4H bit4	R
IER4	EXT interrupt signal mask.	PO4H bit4	W
INT4(EXT)	CPU mode 2 Interrupt signal.		-

Table 3.2 EXT Interrupt terminal

#### CHAPTER 4 Interface Description

4.1 Cartridge Interlace

4.1.1 Overview

EHT-10/EHT-10/2 contains a cartridge interface so that various types of option cartridges can be connected. The following 4 modes are available for cartridge interface and can correspond to various options.

(1) Hand Shake mode (HS mode)

This CPU to CPU mode interfaces with devices which use the CPU as an option, similar to 8255 hand shake mode. HS mode handles data through the input and output buffer. Data handling is controlled by flag.

(2) Input Output mode (IO mode)

This mode uses an interface format of 4 bit input port and 4 bit output port.

(3) Data bus mode (DB mode)

The option looks like a normal IO device in this mode viewing from the main unit. The cartridge I/F only connects the main unit date bus directly to the cartridge data bus.

(4) Output Port mode (OT mode)

This mode interfaces at the 8 bit output port.

Functions with (1) to (4) mentioned above are performed using gate array GAPNIO.

The cartridge I/F consists of a value signal line for modes mentioned above. serial communication signal line and battery outputs of -5 V and +5 V.

# 4.1.2 Cartridge I/F circuitry



Fig 4.1 Cartridge I/F Circuitry

4.1.3 Connector in use

The following connector is used for the cartridge I/F.

Connector	PICL-30P-LT
Ground metal fitting	Cartridge ground metal fitting A (Seiko Epson)

Use the following connector for the other side

Connector	PICL-30S-LT
Ground metal fitting	Micro cassette ground metal fitting B (Seiko Epson)

4.1.4 Cartridge Connector Terminal Name and Function

Pin arrangement (front of main unit connector)



Fig 4.2 Cartridge Connector Pin Arrangement

Table 4.1 Cartr	idge I/F	Pin	Arrangement
-----------------	----------	-----	-------------

Signal	Pin No.	1/0	Description	At main unit reset
FPOP	1	0	When main battery voltage is less than 3.6 V, more then 3 V of H level is output. This signal is used for cut-off when the printer is in low voltage.	Not effected
+5	2	-	+5 V is output at power on.	
CINT	3	I	General use input terminal with interrupt function. When "L" level is input, interrupt generates. Status read is enabled by the I/O port. This signal is used for ready detection of the printer, care is required for other uses.	

Signal	Pin No.	1/0	Description	At main unit reset
CCTL1	4	0	General use output.	"L" at reset.
CCTL3	5	0	General use output.	"L" at reset.
CWR	6	1/0	Write pulse input from the cartridge option in HS mode. Write pulse output to the cartridge option in DB mode.	Input state
CEN	7	I	When CPU 6301 is used in the cartridge, 6301E is connected. Other than that "H" level is fixed.	
CDO	8		The data bus I/O with cartridge	They are set to
CD1	26		are port outputs in OT mode, CD7 to	in the input
CD2	25		CD3 to CD0 are port inputs.	state.
CD3	11	1/0		
CD4	10			
CD5	28			
CD6	27			
CD7	13			
CSCT	g	1/0	Chip select input from the cartridge option in HS mode. Chip select output to the cartridge option in DB mode. It is for future use in OT or IO mode.	Input state
CRXD	12	1	This Serial receive line is connected to ART section through the serial switch and can be used for general input when set to SSW1=0 and SSW0=0.	
CTXD	14	0	This serial send/receive line is connected to ART section through the serial switch. (See 4.3.1)	"H" level
090	18	I	Cartridge option type identification signal. OPO=1 in HS mode, OPO=0 in other modes.	

Signal	Pin No.	1/0	Description	At main unit reset
CSTR7	19	0	Development cartridge switch signal. Normal operation at "O", development cartridge access enable at "1".	Inconsistent output at power on Reset does not effect this signal
RESET	20	0	Reset signal of the main unit system	
INTC	21	0	Interrupt signal used only in HS mode. When the main unit outputs data or command to the output buffer	"H" at reset.
			for HS mode, INTC=O and it requests interrupt to the cartridge option. When the cartridge option reads data	
			the request is canceled and INTC=1.	
CCTLO	22	0	General use output	"H" at rejet.
CABO	23	1/0	Address input from the cartridge option in HS mode. Address output to the cartridge option in DB mode. It is for future use in OT or IO mode.	Input state.
CAB1	29	1/0	General use input line from cartridge option in HS mode. Address output to cartridge option in DB mode. It is for future use in OT or 10 mode.	Input state.
CRD	24	1/0	Read pulse input from the cartridge option in HS mode. Read pulse output to the cartridge option in DB mode.	Input state.
GND	16	-	Signal ground	
- 5	17	-	This signal outputs -6 +1 V at power on.	
VB	15	-	This signal is connected to (+) of the main battery.	
VG	30	-	This signal is connected to (-) of the main battery.	

# 4.1.5 Cartridge I/F I/O Address Map (Table 4.2)



(Note)Terminal names are written in ( ) under bit names for I/O address bit names which are different from I/F terminal names.

# 4.1.6 Mode Setting

The cartridge I/F mode is set by cartridge switch CSW1, CSW0 (bit 1, 0) in the switch register (SWR:P18H). CSW1 and CSW0 are set to HS mode (CSW1, CSW0=0) at reset.

CSW1	CSWO	Mode		
0	0	HS mode		
0	1	10 mode		
1	0	DB mode		
1	1	OT mode		

Set to HS mode for mode setting after initialization (in HS mode) and for mode switching at cartridge option exchange.

### 4.1.7 Mode Description

 Symbols with meaning that are changeable according to modes. Table 4.3 shows signals with meaning that are changeable according too modes (HS, DB, IO, OT).

Signal	Mode	HS	DB	10	10
CSCT		(Input)Chip select input	(Output)Chip select output	unused	unused
CAB1		(Input) General use input line	(Output) Address output	unused	unused
CABO		(Input) Address input	(Output) Address output	unused	unused
CRD CWR		(Input) Read/write pulse	(Input) Read/Write pulse	unused	unused
CB7 -	0	(I/O) Data bus I/O	(I/O) Data bus I/O	(Output) Port output (CDE7 to 4) (Input) Port input (CDB3 to 0)	(Output) 8 bit port output
INTC		(Output) Interrupt signal	unused	unused	unused

Table 4.3 Mode Signal Name

(Input) cartridge --> EHT-10 and EHT-10/2 (Output) EHT-10 and EHT-10/2 --> cartridge (2) Signals enabled for general use I/O

Table 4.4 shows the signal names enabled as general use input or output.

Signal	1/0	Port	and bit position	Note	At	reset
	0	P19H	bitO (PF)	CCTLO terminal becomes 0 when 1 is written.		1
CCTL1	0	P19H	bitl (SLIN)			0
CCTL3	0	P19H	bit3 (PINI)			0
CRXD	I	P16H	bit3 (RXD)	Select SSWO=SSW1=0 at switch register P18H.		
CABI	I	P11H	bit2 (CAB1)	Only HS mode is valid.		
CINT	i	P16H	bitO (PBUSY)	With interrupt function When CINT=0, PBUSY=0		

(Input) cartridge --> EHT-10/EHT-10/2 (Output) EHT-10/EHT-10/2 --> cartridge

Table 4.4 General Use I/C Signal

(Note 1) See "3.6 EXT Interrupt" for CINT.

(Note 2)  $\overline{\text{CCTLO}}$ ,  $\overline{\text{CCTL1}}$ ,  $\overline{\text{CCTL3}}$  and  $\overline{\text{CINT}}$  are used for the printer in OS, when used for general use I/O, care is suggested. See "Software Version 8.8 EXT Interrupt" for further information.

HARDWARE Page 15 - 8

.


Fig 4.3 HS Mode I/F Diagram

### 1. Description

The main unit interfaces with the cartridge option through the output or input buffer in HS mode. When data is written to the output buffer by the main unit, it is read by the option. When data is written to the input buffer it is read by the main unit. Hand shake is performed by OBF (Output Buffer Full) and IBF (Input Buffer Full).

Option CD 7 to O, CABO  $\overline{CRO}$ ,  $\overline{CWR}$ ,  $\overline{CCS}$  are output from the option. When the main unit writes to the output buffer, ABO=1 sets the command and ABO=0 sets the data. The value of ABO is a status flag and is fetched by FO, this is shown from the option side by status read. A write to the output buffer regardless of the command data sets OBF=1

and INTC=0, and an interrupt to the option is requested.

The value of OBF can be shown from either the main unit or the option by status read. (However, OBF and IBF are reversed between the main unit and the option, OBF viewed from the main unit is the option for IBF.)

The option acknowledges, write is performed from the main unit to the buffer by interrupt or status read, and the data is read with command/data flag FO. When the option reads the output buffer, it returns to OBF=0, INTC=1. The option reads the value of the output buffer as CABO=0 and reads the status as CABO=1.

When the option writes to the input buffer (command/data are not distinguished when written from the option, thus, the value of CABO is originally "Don't care" but must be used as CABO=0), IBF=1 and the main unit checks this value using the status read and reads the input buffer.

IBF=O is returned by input buffer read. Data (input buffer) is read as ABO=O from the main unit and the status is read as ABO=1.

CAB1 is not an address in HS mode but a general use input from the option. The value of this CAB1 can be read from the main unit as a part of the status.

2. I/O address space

R/W	10 address	Register	7	6	5	4	3	2	1	0	Flag	3	
	P10H	CHSIR			8	bits	dat	ta			IBF	rese	t
	P11H	CHSSR						CABO	IBF	OBF			
ĸ	P12H								-				
	P13H		e use	! (ac	ces	5 019	Sad	e).					
	P10H	CHSOR			8	bits	dat	ta		_	OBF	set,	F0=0
	P11H	CHSOR			8	bits	cor	nmand			OBF	set,	F0=1
	P12H		For future use (access disable)						*				
	P13H	ror future											

Note When OBF=1, INTC=0.

View from the cartridge option

R/W	CABO	Register	7	6	5	4	3	2	1	0	Flag	
	0	OPIR			8 1	bits	data				OPIBF	reset
ĸ	1	OPSR					FO		OP I FB	OPO BF		
	0	OPOR			8	bits	data				OPOBF	reset
W	1		For	fut	ure i	Jse	(acce	222	disal	ble)		

OPIR=CHSOR, OPOR=CHSIR, OPIBF=OBF, OPOBF=IBF. Data when FO is O, Command when FO is 1.

3. AC characteristic (Main unit protection point of 1 K ohm resistance)



Fig 4.4 HS Mode AC Characteristic

(4) IO mode



Fig 4.5 IO Mode I/F Diagram

1. Description

The IO mode consists of a 4 bit output port (CD7 to CD4) and input port (DC3 to DCO). The output side consists of an 8 bit latch (CIOR), the value of terminal CD3 to CDO is input without the latch at input side. I/O address is PIOH. The contents of the 8 bit data bus is written to CIOR at output to PIOH. CIOR high order 4 bits are output directly to CD7 to CD4, low order 4 bits are not connected. After PIOH is read, the values of CD terminal 3 to 0 are input directly to the data bus low order 4 bits are input to the high order 4 bits.

CSCT, CAB1, CABO, CWR and CRD are high impedance in IO mode, CSCT, CWR

and CRD are pulled up, CAB1 and CABO are pulled down. INTC output becomes 1.

# 2. I/O address space

R/W	10 address	Register	7	6	5	4	3	2	1	0
	PIOH	CIOR	(Cont CB 3	en to	ts of 0	out	tput	port	:)	
	P11H									
	P12H	For fut	ire us	ie.	(acce	<b>s</b> s (	lisat	ole)		
	P13H									
	PIOH	CIOR	4 bit	: <b>s</b> (	data		Don	't ca	ire.	
Ц. Ц.	P11H									
	P12H	For fut:	ure us	e	(acce	ss c	lisat	ole)		
	P13H									

Table 4.5 10 Mode 1/0 Address Space

3. AC characteristic (Main unit protection point of 1 K chm resistance)



Fig 4.6 IO Mode AC Characteristic

# 4. IO Mode I/F Circuitry



Cartridge interface side

Fig 4.7 I/O Mode Interface

(5) DB mode

DB 7 - 0 <=

⇒ CD 7 - O

Fig 4.8 DB Mode I/F Diagram

1. Description

The DB mode uses the cartridge option as normal I/O device, the option has the appearance from the main unit, of an I/O device with 4 address spaces. CD7 to CDO is directly connected to the system data bus,  $\overline{\text{CSCT}}$ ,  $\overline{\text{CRP}}$ ,  $\overline{\text{CWR}}$ , CAB1, CAB0 are all supplied from the main unit, control lines are all output.  $\overline{\text{CSCT}}$  is 0 for address 10H to 13H, CAB1 and CAB0 are address low order 2 bits.  $\overline{\text{CRD}}$  and  $\overline{\text{CWR}}$  are I/O read/write pulse from the main unit.  $\overline{\text{INTC}}$  output is 1 in DB mode.

2. I/O address space

R/W	10 aduress	Register	7	6	5	4	3	2	1	0
	PIOH									
	P11H		Defin	Defined at cartridge o	opti	on si	de			
ĸ	P12H									
	P13H		Disab	led						
	P10H									
L	P11H		Defined at	te he		rtridge o	onti	on si	de	
W	P12H						= uption	ו כי ווט	uc	
	F13H									

## Table 4.7 DB Mode I/O Address Space

<Note> I/O address P13H READ is used by OS to get the device address, thus, circuitry must be structured as the cartridge option does not operate by P13H read.



\* -> Value determined by option Fig 4.9 DB Mode AC Characteristic HARDWARE Page 15 - 15



Cartridge inte.face\_side

Fig 4.10 DB Mode Interface

Page 15 - 16 HARDWARE



Fig 4.11 OT Mode/IF Diagram

1. Description

OT mode consists of an 8 bit output port (latch). The output buffer (COTR) address is PlOH, Contents of the data bus DB7 to DB0 are fetched to COTR at output to POOH and are output to CD7 to CD0 at the same time. Contents of COTR is read by reading PlOH.

CSCT, CAB1, CABO, CWR and CRD are high impedance, CSCT, CWR and CRD are pulled up, CAB1 and CABO are pulled down. INTC output is 1.

2. 1/0 address space

R/W	10 address	Register	7	6	5	4	3	2	1	0
	PIOH	COTR	DTR (Contents of output port)							
	P11H									
R	P12H	For futur	euse (	acce	ss di	sable	)			
	P13H									
	PIOH	COTR		8 1	oits	data				
	P11H				-					
W	P12H	For future	euse (	acce	ss di	sable	)			
	P13H									

Table 4.8 OT Mode I/O Address Space

<Note> Register CHSOR, CHIOR and COTR are physically the same.

3. AC characteristic resistance)



Fig 4.12 OT Mode AC Characteristic

4. OT mode J/F circuitry



Fig 4.13 OT Mode Interface HARDWARE Page 15 - 18

### 4.1.8 Device Address

Cartridge optional devices are assigned specific addresses for cartridge type recognition when connected. The OS identifies the type by putting this device address main unit in DB mode and reading I/O port P13H. Some of the device addresses have already be defined as in Table 4.4, most are available for future expansion. A user should utilize an available address if a device address is required. In this case, OS only sets the cartridge I/F mode to switch register (SWRP18H) following modes in the table, it does not access the cartridge, but must access the cartridge in the user's application program. Timing is used by OS to identify the cartridge option device address and set to switch registers after power switch on , reset and at system initialization.

(1) How to set device address

When a user sets the device address, pull up the data buss CD4 to CD7 at cartridge option by 10 Kohm resistance. Fix terminal OPO (CSEL) to high or low. CDO to CD7 have been already pulled down by 100 k ohm resistance by the main unit, busses that are not pulled up are assumed as "L".

(2) Device address set example

Ex. 1 : Figure 4.14 (a) snows when cartridge ID=8 in DB mode.

Ex. 2 : Figure 4.14 (b) shows when cartridge ID=5 in HS mode.

Device number	CSEL-i		CSEL=0	
ОН		No option	١	
1H 2H 3H	Printer unit	HS mode		For DB mode extension
4H 5H 6H 7H	(M160 Printer)	For HS mode extension		10 mode
8H 9H AH 8H				DB mode
CH DH EH FH				OT mode

### Table 4.9 Device Address Table

(Note) As cartridge IDO, 3 and 7 are defined in EHT-10/EHT-10/2, they should not be used.





# 4.1.9 Cartridge I/F DC Characteristic

(1) Signal line other than **RESET** FPOF

ltem		Symbol	Condition	Spec	ifica	ation	Unit
				Min	Тур	Max	
Output voltage	H level	VOH	10H=-0.4mA	3.7		5.25	V
	L level	VOL	IOL=2mA	0		0.4	V
Input voltage	H level	VIH		2.6			٧
	L level	VIL				0.7	٧
Input leakage c (See Note 2)	urrent		V1=0 to 5.25	-10		10	μΑ

(Note 1) Value at main unit protection point of 1 K ohm resistance.

(Note 2) Value at IC9 input terminal (including I/O common terminal at input time) excluding resistance leakage of pull-up and pull-down.

(2) RESET

Item		Symbol	Condition	Spec	ifica	ation	Unit
			(See note)	Min	Тур	Мах	
Output voltage	H level	VOH	IOH=-0.35mA	4.2			٧
	Llevel	VOL	10L=2.7mA			0.4	٧

(Note) Power consumption at main unit is excluded.

(3)	FPOF
-----	------

ltem		Symbol	Condition	Specification			Unit
	_		(See note)	Min	Тур	Max	
Output voltage	H level	VOH	IOH=-5mA		3.4		٧
(See Note I)	L level	VOL	IOL=5mA		0		v

(Note) Power ON/OFF depends on battery voltage.

(4) +5 V

Voltage range	Output current	(See	Note	1)
5 V + 10%	200 mA MAX			

(Nole 1) main unit power consumption is included.

(Note 2) Power consumption at power ON/OFF switching (during reset) must be as small as possible. V BK (backup voltage) may drop at switching, contents of RAM cannot be guaranteed.

(Note 3) Rush current at power on must be as small as possible.

(4) -5 V

Voltage range	Output current (See Note)
-6 V <u>+</u> 1 V	10 mA MAX

(Note) Power consumption at main unit is included.

<Reference> Power consumption at RS-232C in use approx. 6 mA.

(5) VB

Voltage	range	(See	Note)
4.6	to 6	.0 V	

(Note) It may be less than 4.6 V at printer operation.

### 4.1.10 Application Circuit Introduction

This chapter briefly introduces application circuits in IO mode, DB mode and OT mode from the cartridge interface mode. Please check operation first to use these circuits.

See the comment on each application circuit as reference. Read thoroughly "Chapter 6 Notes on Circuit Design".

(1) IO mode application circuit



Fig 4.15 IO Mode Application Circuit

Simple circuit using the buffer, LED and dip switch is introduced as an 10 mode application. CD6 is pulled up to :5 V by resistance of 10 K ohm and OPO="L" to obtain the device address 40H in this example. When this circuit is connected to the cartridge J/F of EHT-10/EHT-10/2 and the main unit power is turned on, OS automatically sets IO mode to switch register [P18H] so that the application program can read the contents of the dip switch and turn on/off the LED by P1CH read/write. CD4 to CD7 pull-down resistance of 100 k ohm is for input protection (74HC244 in this case). Resistance value of 100 K ohm to 1 M ohm is suggested.



## Fig 4.16 DB Mode Application Circuit

We will connect 82C55 to DB mode interface as an application example. EHT-10/EHT-10/2 OS reads the contents of I/O address P13H and checks the device address. This operation is an 82C55 access disable condition, CRD disable gate is necessary to avoid malfunction. When the device address is necessary in the application program, pull-up CD4 to CD7 to +5 V using resistance (about 10 K ohm). Device address is

set to 80H (Cartridge ID=8, CSEL=0) CSCT, CWR and CRD are output when I/O address P10H to P13H is specified.



# Fig 4.17 OT Mode Application Circuit

A simple circuit using buffer and LED is introduced as the OT mode application example. CD7 and CD6 are pulled up to +5 V by resistance of 10 K ohm and OPO="L" to obtain the device address COH in this example. Writing P10H can turn on/off LED individually.

(Notes)

(1) When EHT-10/EHT-10/2 turns off the power supply in the continue mode, the reset signal is input to 82C55, thus, the application program cannot operate correctly.

See "Software Version I1.4 Cartridge Device Expansion" to operate the application program in continue mode.

(2) Use CMOS-IC as the circuit to connect to the cartridge option as much as possible in order to conserve power consumption.

4.1.11 Universal Unit Circuit Board Size

To design various option unit using cartridge IO, the universal unit is prepared.

The circuit board size for the universal unit is described below.

Circuit board 1.6 mm glass Pitch 1/10 inch (2.54 mm)



Unit mm

Fig 4.18 Universal Unit Circuit Board Size



Fig 4.19 BarCode Reader Interface Circuit

Barcode can be decoded using the timer/counter in gate array GAPNIT. In addition to barcode data input, the light emitting diode power supply and logic power supply are barcode reader interfaces which can be controlled individually. NP-410 type barcode reader is used for EHT-10/EHT-10/2.

4.2.1 Connector in Use

The following barcode reader I/F connector is used.

TCS7560-01-101

## 4.2.2 Terminal and Function

Terminal	Pin number	1/0	Contents		
BRD	2	I	Barcode reader read signal "L" when white		
LGND	3	-	Logic ground		
GND	4	-	LED ground		
VLOG	5	-	Barcode reader logic section power supply.		
VLED	6	-	Barcode reader LED power supply.		
CG	E	-	Barcode reader case ground		

Table 4.10 Barcode reader I/F Terminal Function



Fig 4.20 Bar Code Reader I/F Pin Arrangement

### 4.2.3 I/O Register

(1) Barcode Reader LED Power Control

Write "1" to ICCTLR [P17H] bit 3 in order to supply power to the barcode reader LED. As this register is not effected by the reset signal, care is required.

(2) Barcode Reader Logic Power Supply Control

Write "1" to CTLR1 [POOH] bit 3 in order to supply logic power to the barcode reader. This register becomes all "0" by reset.

(3) Input signal BRD

The signal input by the barcode reader is fetched at gate array GAPNIT. This signal line is directly read by reading the I/O register STR [PO5H] bit 1. As Input Capture Register (ICR) latches the value of Free Running Counter (FRC) every time the signal changes, the time difference between the latch pulse and the next latch pulse can be measured. The latch pulse generation can be acknowledged by Input Capture Flag ICF [PO4H bit 2], this ICF is reset by reading ICRH.B [PO3H]. Latch pulse generation method can be selected by setting control register CTLR1 [PO0H] as follows.



Fig 4.21 latch Pulse Generation Method

4.2.4 Power Characteristic

Power characteristic of the barcode reader interface section is as listed below.

LED power voltage : 5V + 10% Service current to LED power supply : 50 mA MAX (See note) Logic power voltage : 5V + 10% Service current to logic power supply.: 50 mA MAX (See note)

(Note) This value is determined by the transistor characteristic for control, not provided power capacity for barcode. When this I/F is used with another load (IC card for example), it is limited by the suppliable capacity at the power supply.



Fig. 4.22 RS-232C I/F Circuit

EHT-10/EHT-10/2 serial interface has built-in ART (Aperiodic Receiver/transmitter) which is the same level as 8251. This ART consists of the functions from 8251 functions, which are only necessary for EHT-10/EHT-10/2. The send data section is independent from the receive data section and clocks sent from the baud rate generator are individually input. Send/receive data sections have a double buffer structure.

If buffer control function is used, even after host requests to stop send data, EHT-10/EHT-10/2 may output data of 2 to 3 bytes before send data stops. Receive buffer holds 256 bytes in main memory in addition to this. ART outputs interrupt signal RXRDY to the main CPU after receiving data.

# 4.3.1 Serial Mode Switching

Gate array GAPNIO has 3 systems of serial interface; RS-232C, IC card, cartridge SIO (Serial IO). Use the Switch Register SWR [P18H] to select. Selected serial interface is connected to ARI (same level as 8251) in the gate array. This is called serial mode switching. Serial mode switching by switch register SWR is as listed below. Pay attention to the change in the contents of RCTS [P16H bit 5].

Serial mode	SSW3(Bit 3)	SSW2(Bit2)	Contents 1/0	Port P16H Value of RCTS
0	0	0	Cartridge SIO	
1	0	1	IC card	1
2	1	0	RS-232C	Reversed RS-232C CTS terminal

Serial mode switching block is as shown below.





4.3.2 Connector in Use

The following connector is used for RS-232C I/F.

TCS7580-01-101

# 4.3.3 Terminal and Function

Terminal	Pin number	1/0	Contents
GND	1	-	Ground
RTX	2	0	Serial data output
RRX	3	1	Serial data input
RTS	4	0	Request to send
СТЅ	5	I	Clear to send
OSR	6	I	Data set ready
DTR	7	0	Data transmit ready
CD	8	I	Carrier detect
CG	E	-	Case ground

Table 4.11 Terminal functions of RS-232C interface



Fig 4.24 RS-232C I/F Pin Arrangement

# 4.3.4 Function Overview

Table 4.12 Specification of RS-232C Interface

Signal level	RS-232C level (+5 V)
Bit transfer rate	110,150,200,300,600,1200,2400,4800,9600,19200 38400,75 (bps)
Start bit	1 bit
Stop bit	1 or 2 bit
Parity	Even or odd parity/no parity
Error check	Parity error, framing error, overrun error
Communication format	Full duplex
Receive level	Within ±15 V

Set Switch Register SWR [P18H] SSW1 and SSW0 to SSW1=1 and SSW0=0 respectively to use the RS-232C interface.

1

\* Send level of -5 V fluctuates slightly depending on the other load condition ( $-6 \pm 1$  V).

### 4.3.5 Difference From 8251 (USART)

The major differences of the 8251 (USART) from the EHT-10/EHT-10/2 RS-232C interface are as listed below.

Contents	8251	EHT-10/EHT-10/2 RS-232C Interface		
Communication format	2 types: Synchronous and asynchronous	Only asynchronous		
Stop bit	3 types: 1, 1 1/2, and 2bits	2 types: 1 and 2 bits		
Data length	4 types: 5, 6, 7, 8 bits	2 types: 7 and 8 bits		
Clock rate 3 types of clock: 1, 16, and 64 times		Fixed to clock of 16 times		
CTS terminal	Sent when CTS=0. It is hardware controlled.	Send data can be controlled by reading the CTS signal through the I/O port. It is not hardware controlled.		
Enter mode (EH)	Enabled	Disabled as only asynchronous is supported.		
Internal reset(1R)	Enabled	Disabled (See Note 1)		
Break Detect (BD)	Enabled	Disabled (See Note 2)		

Table 4.13 Difference From 8251 (USART)

(Note 1) IR is used in 8251 for return from command instruction to mode instruction. AS EHT-10/EHT-10/2 RS-232C I/F has a different address for Command Register [P16H] and Mode Register [P15H], Internal Reset is not necessary. Set TXE=0, RXE=0 before rewrite of the mode register.

(Note 2) 8251 status register bit 6 is BD (Break Detect) but EHT-10/EHT-10/2 RS-232C 1/F does not support this function by the hardware.

### 4.3.6 Error Check During Receive Data

The EHT-10/EHT-10/2 RS-232C interface can acknowledge through the I/O port that an error has occurred while receiving data the same as 8251. FE (Framing Error) corresponds to I/O port P15H bit 5, OE (Overrun Error) corresponds to bit 4 and PE (Parity Error) corresponds to bit 3. They operate as follows.

FE (Framing Error) ... Even when a framing error occurs, data receive operation is not effected and continues. When successive data is fetched, the newly received data is checked for framing error and FE is reset if the stop bit is correct. Reset is also performed by reset input or error reset command (ER=1).

OE (Overrun Error) ... Even when overrun error occurs, the data receive operation is not effected and continues. OE is not reset even if successive data is correctly fetched. To reset OE, error reset command (ER=1) or reset input is necessary.

PE (Parity Error) ... PE reset condition is the same as FE. Parity is checked only when PEN=1. PE is 0 when PEN=0.

### 4.4 IC Card Interface

The EHT-10/EHT-10/2 can read/write data from/to the IC card based on the ISO DP7816. However, to write, only the IC card can be used as program voltage is +5 V constant. Close the rear cover of the main unit when using IC card. The IC card connector is stored in the IC card, power is supplied and data is sent/received by connecting to the IC card connector, thus, avoiding an increase of resistance by touch. Set switch register serial mode to 1 (See 4.3.1) when using the IC card.

4.4.1 Connector in Use

The following IC card I/F connector is used.

1CC-8P

4.4.2 Terminal and Function

Terminal	Pin number	1/6	Description
RST	2	0	Reset signal to IC card which resets the CPU in the IC card.
i vcc	1	-	IC card CPU power supply
GND	5	-	Ground
VPP	6	-	IC card EPROM read/write power supply
1/0	7	1/0	Serial data I/O line which handles data with the IC card.
CLK	3	0	Clock supplied in the IC card 4.9152 MHz

## Table 4.14 Pin contact function of IC card I/F

(Note) Neither pin 4 nor pin 8 is connected.



Fig 4.25 IC Card Pin Arrangement

## 4.4.3 IC Card Interface Block



(Note)

Interrupt : Two factors of interrupt from the IC card to the EHT-10/EHT-10/2 are as follows.

(1) EXT (external) interrupt
(2) ART interrupt

(1) EXT interrupt issues when excess current goes through the IC card and IC card cover is open.

(2) ART interrupt issues when data is received from IC card. These interrupts can be read as status through the I/O port, but when interrupt enable, it can be processed in the interrupt program. See "Chapter 3 Interrupt Description" for further information.

## 4.4.4 Power Characteristics

Power characteristics of the IC card interface section is as listed below.

CPU drive voltage (Vcc)	5 V + 5 %
IC memory, read/write voltage (Vcc)	5 V + 5 %
Clock frequency (CLK)	4.9152 MHz
Data transfer rate	9600 bps
Supply current	140 mA MAX (See note)
Excess current detect current	200 mA

(Note) This is a suppliable current value when only the IC card is in use, when this I/F is used with another load (bar code reader for example), it is limited by suppliable current at the power supply. So care is required.

Characteristics other than mentioned above are based on ISODP7816.

EHT-10 touch panel interface is configured as shown below.



Fig 4.27 Touch Panel I/F Block

1 is written to all bits of port P24H. When any switch of the touch panel is pressed, it transmits to 7508 after the keyboard interface scan line is fetched and ANDed, key code "80H" is assumed to be pressed.

Key interrupt routine clears port P24H, raises 1 every 1 bit, reads touch key data of port P25H and P26H, scans thoroughly and detects the pressed key. After scan, write "1" to all bits of P24H again.

(Note) 7508 key scan starts after 16 msec after key interrupt issues, if it overlaps the touch panel scan, the touch panel switch may be assumed to separate.



Reading the key twice avoids the LCD frame signal from entering the touch panel as noise. This noise width is about 20 micro-seconds, therefore, read 30 micro-seconds of gap twice. As the interval of noise is about 35 ms, the number of times to read twice per scan is once at most.

4.6 LCD Control (EHT - 10)

EHT-10 LCD panel is a full graphic panel of 84 dots wide by 154 dots long which uses T6963 as an LCD controller. Display duty is 1/48.



Fig 4.28 LCD Panel Block:

For controller convenience, the 84 dots are configured as 48 dots x 2, a metric of 6 dots x 2 is outside the display area which is assigned to VRAM (write 0 to reduce power consumption).

the display is entirely performed as graphics. T6963 has a built-in character generator, but VRAM address is different from the actual display position as shown in Table 4.15, character output using the character generator is not enabled.

3.6.1 Hardware configuration

EHT-10 LCD hardware configuration is as shown below.

1	2	*** *** *** *** *** *** *		42	43.	 
0	28		668	14	30	 670
1	29		669	15	30	670
2	2A		66A	16	3E	675
3	28		668	17	3F	67F
4	20		66C	18	40	680
5	20		66D	19	41	681
6	<b>2</b> E		<b>66</b> E	1A	42	682
7	2F		66F	18	43	683
8	30		670	10	44	684
9	31		671	10	45	685
A	32		672	1E	46	686
B	33		673	1F	47	687
C	34		67:	20	48	689
D	35		67a	21	49	689
8	36		676	22	4A	68A
F	37		677	23	<b>4</b> B	688
10	38		678	24	4C	68C
11	39		679	25	4D	68D
12	3A		67A	26	<b>4</b> E	68E
13	38		67B	27	4F	 68F

Relation of the position between VRAM address and display is listed below (address is HEX).

Table 4.15 Relation of Position between VRAM Address and Display

VRAM valid addresses are B6 and B7 for data in the address (13H, 3BH ...) which is the same level as the bottom line in OFH to 68FH.

4.6.2 VRAM Address Hap



With T6963, this VRAM is virtual memory area and the area actually displayed on screen is the real area (1680 bytes fixed). The real area can be variable according to command.

The address map on the left shows the set values of EHT-10 and EHT-10/2 OS.

4.6.3 I/O Port

]/O port	READ	WRITE
P20H	LCDC data	LCDC data
P21H	LCDC status	LCDC command

LCDC T6962 register is assigned to I/O port P2OH and P21H. When these ports are accessed in an application program, LCDC status must be read and timing of read/write is necessary (however, LCDC status read or timing is not necessary with a relatively slow language such as BASIC).

Table 4.16 shows T6963 status and Figure 4.29 shows the communication flow with CPU.

Status	Contents
STAO (BUSYI)	To check whether instruction is executable. =1 : Executable =0 : Non-executable (during instruction execution)
STA1 (Busy2)	To check whether data can be written/read. =1 : Data enable =0 : Data disable (during internal process)
STA2 (DAV)	To check whether data can be read (valid only when AUTO). =1 : Read enable =0 : Read disable
STA3 (DAV)	To check whether data can be written (valid only when AUTO). ≈1 : Write enable ≈0 : Write disable
STA4	Don't care.
STA5 (CLR)	To check whether the controller is operatable. =1 : operatable =0 : Not operatable
STA6 (ERROR)	If a point other than on the real screen is set while copy, the flag is set. Then if the instruction is executed, the flag is reset.
STA7 (Blink)	To check blink state (approximate cycle 1 second duty 50%) =1 : Normal display =0 : OFF

Table 4.16 T6963 Status Table

MSB

LSB

i

STA7	STA6 ST	STA4	STA3	STA2	STA1	STAO	
------	---------	------	------	------	------	------	--

# (Description)

STA5 : The clock is unstable for 1 to 2 ms after HALT release. T6963 does not operate during this time.

(a) Command write



(b) Data write



Fig 4.29 Communication Flow of T6963 with CPU

## 4.6.4 T6963 Command

The T6963 contains a great number of commands, but the commands related to text display cannot be used due to the hardware of the EHT-10/EHT-10/2. Only the commands used in the EHT-10/EHT-10/2 are described. Initial data must be written before writing commands (WTRG, WTRM, DR/W), necessary for the T6963.

(1) Internal Register Write (WTRG)



Command		0	0	1	0	0	1	0	0
Data	D1	Address (low)							
	02	Ac	ldı	res	ss	()	nig	gh)	

A user uses this to specify a RAM point when writing data to RAM or reading data from RAM. (OH < address < 68FH)

(2) Internal RAM Write (WTRM)



N	Meaning	D1	D2
10	Display graphic home address	Address(low)	Address(high)
11	Number of graphic columns	Column	0

Display graphic home address: Specifies the point in the virtual memory graphic area to place the home position on the real screen. EHT-10 and EHT-10/2 initial value is D1=0, D2=0.

Number of graphic columns : Specifies the number of columns per line to use the real graphic area in the virtual memory graphic area. The EHT-10/EHT-10/2 initial value is D1=(160 dots x 2)/8=40. A user should not set values without proper preparation for the number of graphic columns and graphic home address. (3) Display Mode Set (DSPM)



No data. This command enables LCD screen on/off.

(4) Data Read/Write (DR/W)



N	Meaning		
000	Data write (Address pointer up after execution)		
001	Data read (Address pointer up after execution)		
010	Data write (Address pointer down after execution)		
011	Data read (Address pointer down after execution)		
1+0	Data write (Address pointer change after execution)		
1+1	Data read (Address pointer change after execution)		

Data write executes instructions after data set.

Ν

(5) Auto Mode (AS/R)


N	Meaning
00	Data auto write set
01	Data auto read set
1*	Auto reset

Data auto write sends data after executing this instruction. Address pointer goes up every time data is written (read) till auto reset is accepted.

(6) Screen Peek (PEEK)



If the point specified by the address pointer equals the point of the graphic pointer on the real screen, 1 byte of displayed data on the aligned point is transferred to the stack so it can be read by a user. If the pointer specified by the address pointer is not in the graphic pointer area on the real screen, this instruction is ignored and the status flag is set.

(7) Screen Copy (COPY)



If the point specified by the address pointer equals the point of the graphic pointer on the real screen, 1 byte of data is written at a time sequentially to the aligned graphic pointer RAM area, this data is displayed for one line of real screen that follows the aligned point. If the point specified by the address point is not in the graphic pointer area, this instruction is ignored and the status flag is set.

(8) Bit Set/Reset (BS/R)



HARDWARE Page 15 - 45

N	Contents					
000	Bit O (LSB)					
001	1					
010	2					
011	3					
100	4					
101	5					
110	6					
111	7 (MSB)					

This command sets (1) and resets(0) bit specified by memory N which is specified by the address pointer.

## (S) Notes

When auto commands of data auto write set and data auto read set are executed, all commands that follow are ignored. Therefore, if a new command is issued, auto reset must be executed.

#### 4.6.5 Sample Program

Three programs which write "1" (black) over the entire LCD screen, put it in the auto mode, and write "A" in the upper left corner using the write command are shown below.

Sample program 1 Screen filled with 1. (When Data Write command is used.)

100	CLS	'Auto reset
110	OUT &H21,&H2B	
120	OUT &H20,0	
130	OUT &H20,0	
140	OUT &H21,&H24	'Set address pointer
150	FOR I=1 TO 20*84	
160	OUT &H20,&HFF	'Data written to screen
170	OUT &H21,&HCO	'Data Write command
180	NEXT	
190	END	

Sample program 2 Screen filled with 1. (When Auto Mode command is used.) 100 CLS 'Auto reset 110 OUT \$H21, \$H28 120 OUT 8H20.0 130 OUT &H20,0 140 OUT 8H21,8H24 'Set address pointer 'Data auto write set 150 OUT &H21,&HB0 160 FOR J=1 TO 20\*84 170 OUT &H20,&HFF 'Data written to screen 180 NEXT 190 END Sample program 3 Display "A" in the upper left corner of the screen 100 CLS 'Auto reset 110 OUT &H21,&HB2 120 ADRS=0 'Address pointer initial setting 130 GOSUB 260 140 FOR I=0 TO 260 150 READ AS D=VAL("&H" + AS)160 170 OUT &H20,D 190 OUT &h21,&HC4 Data Write command 190 ADRS=AGKS+40 'Pointer is moved to the next line 200 ©OSUB 260 210 **NEXT** 220 END 230 240 DATA 3E,48,88,48,3E 250 260 HI=INT(ADRS/256) 'Address of pointer write 270 LO=ADRS-H]\*256 280 OUT &H20,L0 290 OUT &H20,HI 300 OUT &H21,&H24 310 RETURN

T6963 status is not checked when data and command are written to T6963 in the sample program 1 to 3 as it is not necessary due to the process rate of BASIC. If a user writes in assembler, the status must be checked before data and command read/write.

4.7 Buzzer

EHT-10 and EHT-10/2 used a piezo-electric buzzer. Various frequencies can be selected through software using BIOS BEEP, and various frequencies can be hardware generated by writing "1" or "0".

The circuit for buzzer is as shown below.



# Fig. 4.30 Buzzer Block

Set everything to "O" (SP1=SP0=SP=O) when the buzzer does not sound, SP and CRLR3 should not mask each other.

4.8 Counter

4.8.1 1 Second Counter

This counter is a memory area which counts I second signals of the calendar clock. It counts through the OS and can be set to any as desired. Normally, these counters are used at auto power off.

F02EH,F02DH : Up count F030H,F02FH : Down count

10 A=PEEK(&HF02F) 20 B=PEEK(&HF030) 30 PRINT B\*256+A 40 GOTO 10

4.8.2 1/10 Second Counter

The 16 bit timer counter issues an interrupt when FFFFH -> 0000H. This interrupt is issued every 106.667 ms and is used as a count lock. The interrupt flag (F093H) monitors whether the interrupt is issued. This flag must be reset after reading.

F093H bit 7: 7508 6: ART 4: ICF 3: OVF Timer counter OVF 2: EXT 10 A=PEEK(&HF093) 20 B=A AND &H08 30 IF B=0 GOTO 10 40 POKE &HF093,&H0C 50 C=C+1 60 PRINT C\*(65.563/614.4) 70 GOTO 10 4.9 Dip Switch

The dip switch state is acknowledged by memory address F5F2H (value should not be written in this address).

The dip switch corresponds to F5F2H bits as shown below. However, DP1 is different from the other switches. If DP1=ON, bit 3=0, if DP1=OFF, bit 3=1. DP2 to DP4 are set to 1 at ON, they are set to 0 at OFF. When a user changes the dip switches, press the Reset button or turn the Power switch on again. If not, the contents of the dip switch is not reflected to F5F2H.





Fig. 4.31 Dip Switch

## 4.10 LED (EHT - 10/2)

The EHT-10/2 has 3 LEDs on the keyboard. The LEDs can be controlled by IOCTLR [P19H], but may destroy other bits. Due to this, OS writes the same contents written to IOCTLR to memory address FODDH. Therefore, when a user controls LED on/off, set the bit in the content of FODDH, write to IOCTLR and return the value to FODDH. When 1 is written to the corresponding bit, the LED turns on.



When turning on LED 2.

10 IO=PEEK(&HFODD) 20 LED=IO OR &H40 30 POKE &HFODD,LED 40 OUT &H19,LED 50 END

When turning off LED 2.

10 IO=PEEK(&HFODD) 20 LED=IO AND &HBF 30 POKE &HFODD,LED 40 OUT &H19,LED 50 END 5.1 Overview

In the EHT-10/EHT-10/2 power supply section, the step-up circuit and series regulator generate backup power and +5 V. DC-DC converter generates +12 V for the sub-battery charge, -5 V for serial communication and -15 V for LCD display. (See Figure 5.1.)

At power on, step-up circuit operates and steps up battery voltage to approx. 5.3 V and controls this to 5 V using a series regulator. DC-DC converter operates and outputs the voltages.

At power off, DC-DC converter and step-up circuit stop, the battery voltage which by-passes the step-up circuit becomes backup power through the series regulator.

Battery voltage is always monitored, when the voltage goes less than approx. 4.8 V, power failure signal generates and supply from the sub-battery starts.

The main battery and sub-battery are recharged using an AC adapter. Values of charging current are shown in the table on the next page, the AC adapter charges only the battery and does not supply the main unit. Thus, if the main unit consumes more than charging current, the battery supplements with extra current.



Power supply for printer

Fig. 5.1 EHT-10/EHT-10/2 Power Source Block

HARDWARE Page 16 - 1

Table 5.1 shows power supply specifications.

Output voltage range DC 4.5 V to	,	U	•	
----------------------------------	---	---	---	--

Volt	tage	Standard	Current capacity
	+5	+5+10%	Total of 200 mA MAX
VBK	Power ON	+5+5%	
	Power OFF	+5+5%-8%	5 mA MAX
	- 5	-6V+1V	10 mA MAX
-15		-15V+0.5V	5 mA MAX
	+12	+12V+0.5V	5 mA MAX

Table 5.1 EHT-10/EHT-10/2 Power Source Section Specification

(Note) VBK, -15, +12 are not user available.

Power failure detection voltage: Approx. 4.7 V

Power failure release voltage after power failure occurs : Approx. 5.05 V

#### 5.2 Charging Current to Battery by adaptor

Charging current by the AC adaptor (standard charger) is as listed below.

Main battery		Approx. 120 mA
Sub-battery	At power ON	Approx. 3 mA
	At power OFF	Approx. 1.5 mA

5.3 Battery Capacity

Main battery 700 mAH Nominal voltage 4.8 V

Sub-battery 45 mA Nominal voltage 4.8 V

Memory backup time (256 KB)

Approx. 700 hrs. with main and sub-batteries full.

Approx. 45 hrs. with sub-battery full.

#### 5.4 Charger and Charging Time

EHT-10/EHT-10/2 uses HOOCA\* (AC adaptor/standard charger). Charging time is normally about 10 hours when using the AC adaptor, it takes 45 hours to charge the sub-battery for initial use or after not using for an extended period of time.

Power failure is detected when the battery voltage is less than approx. 4.7 V, CHARGE BATTERY is displayed. If not charged, memory contents are eventually destroyed. If the memory contents are destroyed, when the battery recovers, system initialization starts.

## CHAPTER 6 Notes on Circuit Design

- 1. Cartridge interface signal guarantees operation for circuits which can be fixed to the main unit.
- 2. RS-232C interface is configured of CMOS circuitry, and does not use a line driver or receiver for the RS-232C. Therefore, a 10 to 20 m cable can be connected and operated, but capability cannot be guaranteed.
- 3. Power supply can be obtained from the main unit to an external circuit, but the supply capacity is limited, care is required.
- 4. Use CMOS or NMOS for external circuit ICs.
- 5. Cartridge interface output terminal CSTR7 is dedicated to the EHT-10/EHT-10/2 development cartridge and should not be used for other purposes by a user. This terminal must be left open (NC).
- 6.





HARDWARE Page 17 - 1

When the power supply is supplied from the main unit to an external circuit (when A connection), an external circuit pull-up and pull-down must be performed in the same way as at the main unit side.

When an outside power supply is supplied to an external circuit (when B connection), the input signal must be pulled down. The external circuit output signal is tri-state output. This can avoid unnecessary current to flow from the external circuit when the main unit power is off. R1 to R3 are resistance for input terminal protection.

# Part 3 BASIC

Chapter 1 BASIC Memory Management

1.1 Overview

The following memory map is used at the start of BASIC. The BASIC interpreter is in ROM (bank 1 #2), and BASIC program and variable area are in RAM.



<sup>(</sup>Note) Words and numbers in parentheses are for the EHT-10/2.

Fig 1.1 Memory Map at BASIC Start

## 1.2 Memory Map in RAM

The BASIC interpreter controls the area from address 100H in the user bank to directly before BDOS. However, as the high order address is variable depending on the CLEAR statement, it can be used as machine language area. When CLEAR, xxxx, yyyy is used, the user bank memory map is as shown below. With the CLEAR statement, xxxx indicates the high order address and yyyy indicates the stack size.

0000H		- Pointers held in work area(Address)
0100H	Work area RS-232C receive buffer Disk buffer Stack (yyyy byte)	<pre>&lt;- CCMBUF (099AH,099BH) &lt;- FILPTR (04A6H,04A7H) &lt;- LOWMEM (0962H,0963H) &lt;- TYTTAR (03D0H 03D0H)</pre>
	Program area Simple variable	<- VARTAB (03094,030AH) <- VARTAB (07C6H,07C7H) <- ARYTAB (07C8H,07C9H)
	Array variable	<- STREND (07CAH,07CBH) <- FRETOP (079FH,07A0H)
XXXXH	User area (machine language etc.) BDOS/BIOS	<- HIGMEM (077AH,077BH) <- MEMSIZ (0964H,0965H)
	Fig 1.2 User Ba	nk Memory Map

8ASIC Page 18 - 2

# 1.3 Options at BASIC Start

The following can be optionally set by CONFIG at the start of BASIC, this is equal to rewriting the following addresses. When BASIC starts, il references these areas, so if rewritten after the start of BASIC, it has no effect.

Option (Variable)	Address	NO. of bytes	Contents	Default
BASIC_F (BASICF)	F07CH	1	Number of files to open at the same time (Maximum number is 15.)	3
BASIC_S (BASICR)	F07DH	2	Maximum value of random file buffer at open in "R" mode	128
BASIC_C (BASICB)	F07FH	2	RS-232C receive buffer size	256

# Table 1.1 Options at BASIC Start

As options BASIC\_F, BASIC\_S and BASIC\_C effect the user area size (program, variable) according to the values set, care is required.

1.4 Disk Buffer

The disk buffer size is determined by option BASIC\_F or BASIC\_S and cannot be modified by a BASIC statement. The disk buffer is used as shown below. File 0 is used for commands such as LOAD, SAVE. Files starting from file 1 are used according to the file number specified by the OPEN statement.



# Fig 1.3 Disk Buffer Size

The file buffer for each file is used as shown below. A offset from the leading address of each file buffer is used. File 0 consists of 177 bytes from offset 0 to BOH. Other files consist of 186+n bytes. The VARPTR function points to the random buffer address at open in the "R" mode, otherwise at sequential buffer address.

The character variable specified in the Field statement points to an address in the random buffer.

Offset	No. of	bytes	Contents
00H	01H		File mode = 0: Not open = 1: "1" mode = 2: "0" mode = 3: "R" mode = 4: "A" mode
01H	24H		FCB (CP/M format)
25H	02H		Offset to sequential buffer
27H	01H		Pointer for INPUT#
28H	01H		Number of characters or output digits remaining in the buffer
29H	03H		Unused
2CH	01H		Device number = FFH : KYBD: = FEH : SCRN: = FDH : LPTO: = FCH : COMO: = FBH : COM1: = FAH : COM2: = F9H : COM3: = F8H : BRCD: = F7H : DLLO:
2 D H	01H		Maximum number of digits at output
2EH	01H		Unused
2FH	01H		Flag for INPUT#
30H	01H		Flag for PRINT#
31H	80H		Sequential file buffer
B1H	02H		Record size
взн	02H		Current physical record number
85H	02H		Current logical record number
B7H	01H		Flag
88H	02H		Number of output digits
BAH	n		Random file buffer n is the number of bytes specified by option BASIC_S.

Table 1.2 File Buffer

# 1.5 Text (Program)

The text (program) is chain structured as shown below. Pointers indicate the pointer address of the next line. Line numbers follow the pointers and text is entered after the line number. The end of the line is recognized by the pointer becoming 0,0 (2 bytes of 0).



## 1.6 Simple variables

When simple variables in a program are used, they are indexed in the order of use and in the format corresponding to type. Variables are indexed in the following format. VARPTR function indicate the first byte of data.



# Variable name



# \* String descriptor



\* Data



Single-precision type







BASIC Page 18 - 8

## 1.7 Array Variables

When the DIM statement is executed in a program or the definition of an array is less than 10 elements, the array is indexed in the order of use. Array variables are indexed in the following format.



- Type ... Same as single variable (Value of 2, 3, 4 or 8)
- \* Variable name ... Same as single variable
- Size ... Memory capacity used after dimension (2 bytes)
- \* Dimension ... Árray Number (1 byte)
- \* Number of arrays ... Single-dimension (dimension x 2 bytes)
- \* Element ... One element is 2, 3, 4 or 8 bytes according to variable type.

#### 1.8 Variables in BASIC Work Area

Address : Variable name (Byte length)

036FH : LPWAIT (1)

This variable specifies the wait time to which LPRINT and LLIST refer when printer is not ready. The default value is 30. ("DT error" will be displayed on the LCD after waiting 30 seconds.) Unit is second. But, system will wait eternally if you set this value to 0.

03F7H : RSWAIT (1)

This variable specifies the wait time when COM n: device has opened and control line check is ON. The default value is 30. ("DT error" will be displayed on the LCD 30 seconds after, if the control line of RS-232C have not changed to READY.) Unit is second. But, system will wait eternally if you set this value to 0.

- O3FAH : DEVNUM (2) Pointer for extend sequential device. See Chapter 3.
- O44BH : DCBTAB (2) Pointer for DCB of extend sequential device. See Chapter 3.
- 0995H : BLDADR (2) Loading address for BLOAD.
- U997H : BLDLNG (2) Loaded data length when BLOAD is executed.

# CHAPTER 2 Command Expansion

2.1 Machine Language Hold

The following three ways to hold machine language are available.

- (1) Reduce the BASIC user area by the CLEAR statement to hold an area which the BASIC interpreter does not access.
- (2) Hold the area between the BASIC work area and RS-232C receive buffer, where the BASIC interpreter does not access, using a hook at the start of BASIC. (See 2.2 for further information.)
- (3) Use the user BIOS area.

Work area (2) RS-232C receive buffer Disk buffer Stack BASIC user area (1) RAM disk USER BIOS System work area FFFFH	0000н		-
COOH DCOOH DCOOH DCOOH FFFFH		Work area	
RS-232C receive buffer Disk buffer Stack BASIC user area (1) RAM disk USER BIOS System work area FFFFH			- (2)
Disk buffer Stack BASIC user area - (1) RAM disk USER BIOS DCOOH (DEOOH) System work area FFFFH		RS-232C receive buffer	
Stack BASIC user area - (1) RAM disk USER BIOS JCOOH (DEOOH) System work area FFFFH		Disk buffer	]
BASIC user area - (1) RAM disk USER BIOS USER BIOS System work area FFFFH		Stack	1
DCOOH (DEOOH) System work area FFFFH		BASIC user area	]
RAM disk USER BIOS (DEOOH) System work area FFFFH			- (1)
DCOOH (DEOOH) System work area (3) FFFFH		RAM disk	
(DEOOH) System work area	рсоон	USER BIOS	- (3)
FFFFH	(DEOOH)	System work area	
	FFFFH		

(Note) The word in parentheses is for EHT-10/2.

Fig 2.1 Machine Language Hold

2.2 Hook at BASIC Start

Call the hook after initialization of the work area at the start of BASIC. The default jump target of this hook is an address with the RET instruction. The user routine in the user BIOS area can be called by rewriting this to the address in the user BIOS area.



(0000H to FFFFH) in the user bank can be called by BASHOOK, but only the user BIOS area can actually be used. Hooks controlled by BASIC can be rewritten in the routine called by BASHOOK. One hook, INIT, enables to move the location that follows the RS-232C receive buffer.



0000H to 5FFFH and E000H to FFFFH in the user bank can be called by INIT, (0C50H to 0CCFH) temporary area for hooks in BASIC work area is actually used. Therefore, load the necessary program into this area by BASHOOK. The area where the BASIC interpreter does not access can be held by changing the leading address of the HL register RS-232C receive buffer in the routine called by INIT.

#### 2.3 Reserved Word EXTD

Reserved word EXTD is provided for command (statement, function) expansion. After EXTD is interpreted, the BASIC interpreter jumps to the EXTDS routine when used as a statement and jumps to the EXTDF routine when used as a function. Both EXTDS and EXTDF routines consist of 3 bytes and can write the JUMP instruction. Default is JMP FCERR (FC error).



to use EXTD as a function, (p1, p2 ...) are analyzed in the EXTD function main unit.

Place the EXTD statement and function main unit between the BASIC WORK area and RS-232C receive buffer. Syntax analysis routine in the BASIC interpreter can be used while doing this. See 2.4 for Syntax Analysis Routine. 2.4 Syntax Analysis Routine

(1) SYNCHK Address: 09ACH (or 000BH) Input: HL = Text pointer Output: Same as CHRGET Description: This routine checks syntax. SN error occurs if the character which the current text pointer indicates is different from the character to be checked. Let's assume the character to be checked is xx and call as follows.

> CALL SYNCHK DB xx

When the character the text pointer indicates is xx, advance the text pointer to the next character and return. As SYNCHK has entry at address 8, the following call is enabled.

RST OBH DB xx

SYNCHK is the same as the following routine.

LD	A, (HL)			
Ελ	(SP), HL			
СР	(HL)			
JP	N2, SHERR	;qive	"SN	Error"
INC	HL			
EX	(SP),HL			
JP	CHREET			

(2) CHRGET Address: 7007H (or 0010H) Input: HL = Text pointer Output: A = Character HL = Text pointer

Z flag = 1 ... When the end of the statement is reached.

Description : This routine returns the character before the one indicated by the text pointer to the A register. Space is skipped. When the text pointer reaches the end of the statement, Z flag = 1 is returned.

## (3) CHROUT Address : BE78H (or 0018H) Input : A = character Output : No output Description : This routine outputs one character to the device currently selected. Output target is checked as follows.

¥ 0 -----> File < PTRFIL > = 0 *¥* 0 < PRTFLG > -> Printer = 0 Screen \* FCB leading address is stored in PTRFIL when executing the PRINT# statement. As  $PTRFIL \neq 0$  in CHROUT, the character is output to the file. PRTFIL=0 at the end of the PRINT# statement. \* PRTFLG≠O when executing LPRINT statement. As PTRFIL=O and PRTFLG≠O in CHROUT, the character is output to the printer. PRTFLG=0 at the end of the LPRINT statement. \* Both PTRFIL and PRTFLG are 0 when executing the PRINT statement. Thus, the character is output to the screen in CHROUT. Address : Name (Number of bytes) U3D3H : PTRF1L (2) FCB leading address O3CCH : PRTFLG (1)Flag at printer output (4) COMPAR Address : C9A6H (or OO2OH) Input : HL, DE Output : Z flag, CY flag HL > DE --- Z=0, CY=0HL = DE --- Z = 1, CY = 0HL < DE --- Z=1, CY=1Description : This routine compares the HL register with DE register. The A register is cleared. (5) GETYPE Address : 7A52H (or 0028H) Input : No input Output : Flag (Z tlag, CY flag, P flag, S flag) CY Ρ S Ζ Integer type 0 1 Ε M Character type Ρ 1 1 Ε 0 Ρ Single-precision type 0 1 Ε Ρ Double-precision type 0 0

	Desc type VAL1 (076	cript e, th TYP 68H)	ion : 1 e flag .2 = In 3 = Ch 4 = Si 8 = Do	his ro can ac teger aracte ngle-p uble-p	butine knowle type r type recisi recisi	returned dge t on ty on ty	rns FAC his. pe pe	type.	Variable	VALTYP	has F	AC
(6)	GOBI Addi Inpi Outj	DOS ress ut: put:	: A861H Same as Same a	(or 0 BDOS s BDOS	030H) input outpu	param It par	eter ameter					
	Desc cal C ro BAS erro	cript led a egist CAL DB IC ca or.	ion: Is follo er. L GOB n auses a	This a ws. T DOS n erro	routin The fun	e cal nctior the	ls BDO: number followi	S. Fur does n ng sequ	ot need 1 ence wher	mbern tobese 1 BDOS 1	BDOS t to t reads	is he an
	< A	Reg.	A <b>≢FFH</b> >	>	Return	1						
	< H	Reg. H≢0	H=0 >	>	Return	ו						
				>	The food	0110wi 0SERR =1 : F =2 : F =3 : F	ng erro (add Read err Irite er Irite pr	rs occu ress F4 or (RD ror (WR otect e	r dependi 17H). Error) Error) rror (WP	ng on th Error)	e valu	эс
		H=2		>	Drive	selea	t error	· (DU Er	ror)			
		H=3		>	Read	only c	Irive	(WP Er	ror)			
		H=4		>	Read	only 1	file	(WP Er	ror)			
(7)	FRM Add Inp Out Des FAC	EVL ress ut : put : cript and	: 76F11 HL = T( : HL = <sup>-1</sup> tion : sets t	ext ooi Text po This r This r	inter Dinter Doutine E to V	e eval ALTYP	uates a	n expre	ssion, se	ets the	value	to

(8) FRCINT Address : 870AH Input : No input Output : HL = Integer value Description : This routine converts FAC to an integer and returns it to the HL register. (9) GETBYT Address : 7D79H Input : HL = Text pointer Output : E = A = Value (0 to 255) HL = Text pointer Z flag = 1 When the end of the statement is reached. Description : This routine evaluates an expression and returns the value to E(A) register if numeric. An error occurs if not numeric or greater than 256. GETBYT is the same as the following routine. CALL FRMEVL ;evaluate formula ;save text pointer PUSH HL CALL FRCINT ;convert to integer EX DE,HL ; integer to DE reg POP :restore text pointer HL A.D ;get high order LD OR A,A ; is it O? JP NZ,FCERR ;no, give "FC Error" DEC HL :back text pointer CHRGET ;set condition on terminator A,E ;return the result in A and E CALL LD RET (10) MAKINT Address : B767H Input : HL = Integer Output : No output Description : This routine sets an integer value to FAC. (11) FRESTR Address : CF1AH Input : No input Output : HL = String descriptor Description : This routine puts the string descriptor to the HL register. "TM Error" occurs when not a character. Ex.) This routine evaluates an expression, puts the character string length in A register and the character string address in DE register. ;Formula evaluate FRMEVL CALL PUSH HL ;Save text pointer ;get string descriptor CALL FRESTR ;A = string length LD A,(HL) INC HL  $E_{\rm HL}$ LD INC HL D(HL) ;DE = address LD POP ;restore text pointer HL

BASIC Page 19 - 7

RET

(12) ERROR Address : 69E9H Input : E = Error code Description : Error handler for BASIC

AUTO, AND, ABS, ATN, ASC, ATTRS, ALARH Α BEEP, BLOAD, BSAVE, BACKLIGHT, BRCD R CLOSE, CONT, CLEAR, CINT, CSNG, CDBL, CVI, CVS, CVD, COS, CHR3, CALL. С COMMON, CHAIN, CLS, COLOR, CIRCLE, COPY, CSRLIN, COM DELETE, DATA, DIM, DEFSTR, DFFINT, DEFSNG, DEFDBL, DEF, DAY, DATE. D DSKF ELSE, END, ERASE, EDIT, ERROR , ERL, ERR, EXP, EOF, EQV, EXTD Ε F FOR, FIELD, FILES, FN, FRE, FIX, FONT G GOTO, GOSUB, GET H. HEXS INPUT. IF. INSTR. INT. INP. IMP. INKEYS I J **KILL, KEY, KANJI, KINPUTS** K LPRINT, LIST, LLIST, LPOS, LET, LINE, LOAD, LSET, LIST, LOCATE, LOG, 1 LOC, LEN, LEFTS, LOF, LKANJI, LFONT, LOGIN HERGE, HOD, HKIS, HKSS, HKDS, HIDS, HENU, HOUNT, HOTOR M NEXT, NAME, NEW, NOT N OPEN, OUT. ON, OR. OCTS, OPTION, OFF 0 PRINT, PUT, POKE. POS. PEEK. PSET. PRESET, PAINT, POINT, POWER. PCOPY Ρ Q RETURN, REND, RUN, RESTORE, REM, RESUME, RSET, RIGHTS, RND, RENUM, R RESET, RANDOHIZE, REHOVE STOP, SWAP, SAVE, SPC(, STEP, SGN, SQR, SIN, STRS, STRINGS, SPACES, S SYSTEM, SOUND, SCREEN, SET, STAT THEN, TRON. TROFF, TAB(, TO, TAN, TIHE, TITLE, TAPCNT Т U USING, USR V VAL. VARPTR, VIEW WIDTH, WAIT, WHILE, WEND, WRITE, WINDOW, WIND W X XOR Y Ζ

BASIC Page 19 - 9

.

(HEX)	(H <b>EX</b> )	(HEX)	(HEX)
- 08	A0 -	CO - FIELD	EO - USR
81 - END	A1 - WIDTH	C1 - GET	E1 - FN
82 - FOR	A2 - ELSE	C2 - PUT	E2 - SPC(
83 - NEXT	A3 - TRON	C3 - CLOSE	E3 - NOT
84 - DATA	A4 - TROFF	C4 - LOAD	E4 - ERL
85 - INPUT	A5 - SWAP	C5 - HERGE	E5 - ERR
80 - DIM	AG - ERASÉ	CG - FILES	EG - STRINGS
87 - READ	A7 - EDIT	C7 - NAME	E7 - USING
88 - LET	A8 - ERROR	C8 - KILL	E8 - INSTR
89 - GOTO	A9 - RESUME	C9 - LSET	E9 - '
8A - RUN	AA - DELETE	CA - RSET	EA - VARPTR
88 - 1F	AC - AUTO	CB - SAVE	EB - INKEYS
8C - RESTORE	AC - RENUH	CC - RESET	EC - OFF
8D - GOSUB	AD - DEFSTR	CD - CLS	ED -
JE - RETURN	AE - DEFINT	CE - LOCATE	EE -
8F - REN	AF - DEFSNG	CF - BEEP	EF - >
90 - STOP	BO - DEFDBL	DO - SOUND	FO - =
91 - PRINT	B1 - LINE	D1 -	F1 - <
92 - CLEAR	<b>B</b> 2 -	D2 - COLOR	F2 - •
93 - LIST	B3 -	D3 - PSET	F3
94 - NEW	B4 - WHILE	D4 - PRESET	F4 - •
95 - ON	B5 - WEND	D5 - CIRCLE	F5 - /
96 -	BG - CALL	DG - PAINT	F6
97 - WA]T	B7 - WRITE	07 -	F7 - AND
98 - DEF	B8 - COMMON	08 -	F8 - OR
99 - POKE	B9 - CHAIN	D9 - COPY	F9 - X0R
YA - CONT	BA - OPTION	DA - KEY	FA - EQV
98 - BSAVE	BB - RANDOHIZE	DB - COM	FB - IMP
9C - BLOAD	BC -	DC - 10	FC - HOD
9D - OUT	BD - SYSTEH	DD - THEN	FD - ¥
9E - LPRINT	86 -	DE - TAB(	FE -
9F - LLIST	BF - OPEN	DF - STEP	FF - (function)

# 2.5.2 Internal Code (Function)

Function use a 2 byte code. The first byte is always FFH and the second byte and corresponding reserved words are shown below.

(HEX)	(HEX)	(HEX)	(HEX)
- 08	A0 -	CO -	EO - ALARH
81 - LEFTS	A1 -	C1 -	E1 - WEND
82 - RIGHTS	A2 -	C2 -	E2 - EXTD
83 - MID <b>S</b>	A3 -	C3 -	E3 - MOTOR
84 - SGN	A4 -	C4 -	E4 - KANJI
85 - INT	A5 -	C5 -	E5 - LKANJI
86 - ABS	A6 -	C6 -	EG - FONT
87 - SOR	A7 -	C7 -	E7 - LFONT
88 - RND	- 8A	C8 -	E8 - VIEW
89 - SIN	- PA	C9 -	E9 -
8A - LOG	AA -	CA -	EA - WINCOW
8B - EXP	AB - CVI	CB -	EB - SET
8C - COS	AC - CVS	CC -	EC - ATTRS
8D - TAN	AD - CVD	CD -	ED - KINPUTS
8E - ATN	AE -	CE -	EE - BACKLIGHT
8F - FRE	AF - EOF	CF -	FF - BRCD
90 - INP	80 - LOC	DO - CSRLIN	F0 -
91 - POS	81 - LOF	D1 - POINT	F1 -
92 - LEN	82 - HKI <b>s</b>	D2 - DAY	F2 -
93 - STR <b>S</b>	B3 - MKS <b>S</b>	D3 - DATE	F3 -
94 - VAL	B4 - HKD <b>\$</b>	Dá - TIHE	F4 -
95 - ASC	85 -	D5 - SCREEN	F5 -
96 - CHR <b>S</b>	86 -	DG - DSKF	F6 -
97 - PEEK	87 -	D7 - MENU	F7 -
98 - SPACES	B8 -	D8 - LOGIN	F8 -
99 - 0013	89 -	D9 - TITLE	F9 -
9A - HEXS	BA -	DA - STAT	FA -
9B - LPOS	BB -	D8 - PÜŨPY	FB -
9C - CINT	BC -	DC - HOUNT	FC -
9D - CSNG	80 -	DD - POWER	FD -
9E - CD8L	8E -	DE - REMOVE	FE -
9F - FIX	8F -	DF - TAPCNT	FF -

BASIC Page 19 - 11

# CHAPTER 3 Sequential Access Device Expansion

#### 3.1 Overview

Data is input/output to a file based on the data block called DCB (Device Control Block) in devices which access sequentially. DCB is required in each device such as COMO. DCB must be set and the device name and number must be indexed at the same time to expand sequential access of the device. These can be performed by creating or adding a table. Required tables are as follows.

- \* Device table ... Device names and numbers are indexed.
- \* DCB table ..... DCB addresses for devices are stored in
  - this table.
- \* DCB ..... This table lists routine address which perform data I/O.

3.2 Device Table

This table indexes device names and numbers. The device name and number must be indexed to expand the device. BASIC supports 9 sequential devices and the device names and numbers are as shown below.



Fig 3.1 Device Name and Device Number

DEVNuM indicates the number of devices indexed in the device table. The device table consists of 4 character device names and numbers par device, with a maximum of 16 devices to index.

Increase DEVNUM to expand the device and index the devices of device number F6H - in the device table.

## 3.3 DCB Table

This table stores the address of DCB. DCB address for each device must be stored to expand the device.



## Fig 3.2 DCB Table

Indexing to the DCB table must be performed in the position corresponding to the device numbers indexed in the device table.
#### 3.4 DCB (Device Control Block)

This table stores the entry address of routines such as each device open, close, data I/O. 10 entry addresses per device are required. DCB configuration and contents are as shown below.

Item	Offset	Size	Name	Contents
1	0,1	2	OPEN	Open
2	2,3	2	CLOSE	Close
3	4,5	2	OUTPUT	One character output
4	6,7	2	INPUT	One character input
5	8,9	2	LOC	For LOC function
6	10,11	2	LOF	For LOF function
7	12,13	2	EOF	For EOF function
8	14,15	2	PUT	Saves preread data
9	16,17	2	WIDTH	Maximum number of digits at output
10	18,19	2	RND	For GET# and PUT#

#### Table 3.1 DCB Contents

(1) OPEN

Procedure :

1. Check whether open mode is correct. (If "O" mode is specified in the device only with input, an error occurs.) If "R" mode is specified, the mode ("I" or "O" or both) which can open the device is assumed specified.

2. Actual open process must be performed for the device.

3. If open is performed without an error, PTRFIL and FCB must be initialized. a. Set FCB leading address to PTRFIL (03D3H, 03D4H). b. Initialize the area for FCB sequential device. FCB + 0Open mode + 28H Initial value of digit position at output + 2DH Maximum number of digits at output (See WIDTH) However, the digit position and maximum number of digits at output are meaningless in the "I" mode. 4. Remove the file number text pointer from the stack. Put the text oointer in HL. \* File name can be specified in the Open statement. For example: OPEN "I", #1, "DEVO:(ABC)" is executed, character string of "(ABC)" enters FILNAM+1 (11 bytes 0504H to 050EH) and device number enters FILNAM (0503H). (?) CLOSE Function : Device close process Input condition : (SP) = FCB leading address (SP+2) = Text pointer Output condition : HL = Text pointer Procedure : 1. Perform close process for the device. 2. Remove the FCB address from the stack clearing the 49th byte to 0, from the address. 3. Set PTRFJL (address 03D3H, 03D4H) to 0. 4. Remove the text pointer from the stack, put it in HL and return. OUTPUT (3) Function : One character output Input condition : (SP) = Output character HL = FCB leading address Output condition : None Procedure : 1. Remove the character to be output from the stack and output to the device. 2. Remove AF, BC, DE and HL respectively from the stack and return.

BASIC Page 20 - 5

(4) INPUT Function : One character input Input condition : HL = FCB leading address Output condition : A = Input area CY = 0 : When normally input. 1 : When there is no data for input (EOF) or forced to cancel input. Procedure : 1. Check whether there is data saved by PUT before input from the device, if there is, return the value. 2. Input one character from the device. (5) LOC Function : LOC function Input condition : None Output condition : Set value to FAC. (6) LOF Function : LOF function Input condition : None Output condition : Set value to FAC. (7) EOF Function : EOF function Input condition : None Output condition : Set value to FAC. 0 = Not EOF1 = EOFMAKING routine is suggested for use to set the return parameter to FAC (Floating Accumulator) in LOC, LOF or EOF for convenience. (8) PUT Function : This DCB saves preread data. Input condition : C = Data to be saved Output condition : None **Description** : Data preread in the INPUT# statement is saved in PUT. 1 byte per device is required to save the data. This area holds 16 bytes (device number FFH to FOH respectively) from PUTBUF (OA9AH) Access to PUTBUF is as follows. \* OPEN ..... Clears PUTBUF. \* PUT ..... Saves data to PUTBUF. \* INPUT .... Returns value of PUTBUF as data when PUTBUF is not 0. O must be set after PUTBUF read.

Offset	Address	Device number	Device name	Note
0	0 <b>A9A</b> H	FFH	KYBD	
1	0A9BH	FEH	SCRN	Not used
2	0A9CH	FDH	LPTO	Not used
3	0 A9DH	FCH	СОМО	
4	0 <b>A9EH</b>	FBH	COM1	
5	0A9FH	FAH	COM2	
				•

(Note) PUT is not required for a device with output only.

(9) WIDTH

Function : This DCB saves the maximum number of digits at output. Input condition : C = Maximum number of digits Output condition : None

#### Description :

This DCB is called at execution of WIDTH "device" statement. WIDTH saves the maximum number of digits (WIDTH "device", W of W) at output to enable use of this value in OPEN or OUTPUT. The user must hold the area to save the maximum number of digits. The maximum number of digits which is saved by WIDTH is accessed as follows.

\* OPEN .... Sets maximum number of digits to FCB. \* WIDTH ... Obtains maximum number of digits from WIDTH statement. \* OUTPUT ... Outputs CR and LF after the output of the number of characters in the statement with the maximum number of digits set in FCB. However, if the maximum number of digits is FFH, neither CR nor LF is output.

Default value must be set in advance as the maximum number of digits is used at OPEN.

(Note) WIDTH is not necessary for a device with input only.

#### (10) RND

Function : Block I/O Input condition : (SP) = Text pointer (SP+2) = O : GET# Other than O : PUT# HL = FCB leading address Output condition : None

Description : The device which BASIC supports as standard does not perform block I/O (GET#, PUT#). (FC error occurs.) GET# and PUT# for sequential device are handled as follows.

BASIC Page 20 - 7

GET# n,m...Input m characters in the file buffer of file number n.

PUT# n,m...Outputs m characters in the file buffer of file number n.

In GET# and PUT# statement, syntax analysis is only performed till file number (directly before ","), analysis must be performed in RND after ",".

# **PART 4 APPENDIX**

# 1. CHARACTER CODE TABLE

(1) Character generator code table

1								Upp	per	byte	2						
	$\backslash$	0	1	2	3	4	5	6	7	8	9	A	B	C	D	E	F
	٥	à	Æ	SP	0	Ø	P	•	D	_	T	SP	-	9	III	۳J	
	1		Ø	1	1	A	Q	а	q	_	Т	a	7	チ	4	年	
	2	¢	Å	"	2	В	R	Ь	r		-	ſ	1	ッ	X	月	
	3	5	æ	#	3	С	S	с	s		-	J	ゥ	$\bar{\tau}$	ŧ	日	
	4	é	ø	£9	4	D	Т	d	t		-	•	I	+	t		
	5	ů	a	%	5	E	U	е	u		—	•	*	+	٦		
	6	è	É	&	6	F	V	ſ	v			7	מ	_	Е		
e	7	-	a	•	7	G	W	0	w			7	+	R	ラ		
byt	8	Á	ם	(	8	н	X	ħ	x		Г	4	2	7	IJ		
ower	9	Ö	ô	)	9	1	Υ	i	У		٦	ゥ	ケ	1	N		
	A	Ū	1	ж	u v	J	Ζ	j	Z		L	I		$\overline{\Lambda}$	V		
	B	ā	Pt	+	:	к	[	k	(			R	サ	۲	۵		
	C	Ō	1	•	<	L	¥				$\boldsymbol{\ell}$	tr	シ	7	7		
	D	ū	Ň	-	-	Μ	]	m	}		2	د	ス	$\wedge$	ン		
	E	ß	2	·	>	N	-	n	~		4	Э	t	ホ	•		ì
	F	£	Ē	1	?	0	_	0	1	+	۷	ッ	Y	7	đ		

The shapes of some character fonts displayed on the LCD screen are partially different from those printed by the printer. See the font table in (3) for details.

#### (2) Character Code Table

The basic character codes are as listed below. The character codes indicated by shaded portions depend on the countries as listed in the table on the next page.

1							ι	Jppe	r by	te							
	/	0	1	2	3	4	5	6	7	8	9	A	В	C	D	E	F
	0			SP	0		Ρ		p	_	T				_		
	1			!	1	А	Q	а	q	_	Т						
	2			~	2	В	R	b	r		+						
	3				3	С	S	С	s		ŀ						
	4				4	D	Т	d	t								
	5			%	5	Е	U	е	u		-						
	6			&	6	F	٧	f	v	-	$\square$						
	7			,	7	G	W	g	w								
byte	8			(	8	н	х	h	x	1	Г						
wer	9			)	9	I	Y	i	У		٦						
Lo	A			*	:	J	Ζ	j	z		L						
	В			+	:	К		k			Г						
	С				<	L		١			1						
	D			-	=	М		m		e.	٦						
	E				>	Ν		n			1						
	F			/	?	0	-	0	SP	+	2						

	ASCII	France	Germany	England	Denmark	Sweden	Italy	Spain	Norway
23н	#	#	#	£	#	#	#	Pt	#
24н	\$	\$	\$	\$	Ś	Ħ	`\$	\$	2
40н	@	à	6	<u>a</u>	@	Ē	@	@	É
5BH	۱ (	0	Â	1	Æ	Ä	0	1	Æ
5CH		Ç	Ô	\	Ø	Ō	×	Ň	Ø
5DH	1	5	Ū	]	A	A	é	2	A
5Ен	-	•	-	•	•	Ū	•	-	Ű
60н	4					é	ù		é
78н	L .	é	ā	[	æ	ā	à	-	æ
7Сн	1	ů	ō		0	ū	ò	n	ø
7Dн	1	è	ū	)	â	å	è	1	A
7Ен	~		ß	1	~	ú	1	~	ū

#### Notes:

- The printer unit (cartridge printer) for EHT-10/EHT-10/2 does not support the Norwegian font. If Norway is selected, ASCII is assumed.

- The terminal printer connected to the system through an RS-232C interface can only normally print front characters of a specific country normally that it supports.

# 2. FONT TABLE

In EHT-10/EHT-10/2, the character and graphic patterns supported by the mainframe are the same as those supported by the printer as far as the CG codes are the same. However, the shapes of some character and graphic patterns displayed on the LCD screen are partially different from those printed by the printer. These character and graphic patterns are listed in the following table.

ООН	а	12H	A	68H	h	85H		8 F H	+	B2H	1
02H	ç	15H	â	6DH	m	86н		91H	Т	C4H	F
04H	е	1FH	ñ	6EH	n	87H		92H	-	CP.	7
05H	ù	26H	&	73H	S	88H		93н	i	CDH	~
06н	é	2AH	*	75H	u	89н	I	96H		DAH	V
овн	ä	37н	7	80H	_	SAH		97H		ЕОН	PS
ОДН	ů	40H	0	810	-	88H		98H	Г		
OEH	в	59H	Y	82H	-	вСн		99H	٦		
10H	. ~	61H	а	взн		8DH		9СН	1		
11H	Ø	65H	e	84H		BEH		9DH	<u>ר</u>		

# (1) EHT-10/EHT-10/2 font table



#### Page 21 - 5 APPENDIX

			4CH			
		52H				
					\$FH	
		72 H				
	79H					

BOH	<u>81H</u>	82H	83H	84H	85H	86 H	87H
		92H	93H	94H	95H	96H	97H
98H				9CH			9FH

.

#### Page 21 - 7 APPENDIX

			•	

# (2) Printer unit font table

Note: The patterns listed in this table correspond to the EHT-10/EHT-10/2 mainframe front pattern codes.

				07H
				27H
				37H
		3CH		

		5FH	GFH	77H	
		SFH Q Q Q			
45H 000000 0 111 0 4 4 2 0 4 5 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4				75H	
	4CH				
				72 H	
			69H		
40 H					

BIH         B2H         B3H         B4H         B5H         B6H         B7H           100000         100 <th></th> <th></th> <th></th> <th></th> <th></th> <th></th>						
--	--	--	--	--	--	--

Page 21 - 11 APPENDIX

			СССС	

# 3. Version

# (1) CP/M Serial Number

The device and version number is located in the serial number (6 bytes) of BDOS CP/M. The serial number is located in the leading 6 bytes of RBDOS1 and RBDOS2. To refer to the serial number of RBDOS1, check the leading 6 bytes before the BDOS entry address located at address OOO6H and OOO7H. To refer to the serial number of RBDOS2, check the 6 byte area that follows address EAOOH. The same value is stored at these two address areas.



### (2) ROMID

ROMID indicates the OS ROM state which is located in the 8 byte area that follows 7FF8H of OS ROM. To check ROMID in an application, use BIOS LDIRX and read the data from OS ROM. ROMID configuration is as shown below and is set by ASCII code.



# 4. Table of System Work Area

<How to read the table>

Address : Indicates the address of each variable. Addresses are arranged from low to high order for easy location of the address contents when the address is known.

Variable name: Label name to refer to each variable in the system. When the variable is known and its address and contents is desired, refer to the list of labels at the end of this table. Only the first 8 letter are written for variables with more than 8 letters.

Number of bytes : Indicates variable size in bytes (decimal notation).

Type : Indicates kind of variables.

- R/W : A user can rewrite, changing the state of the system as a result.
- R/O : A user cannot rewrite, but reference enables to check the state of the system.
- -: Used as temporary work of the system. A user should not rewrite.
- initial value : Indicates initial value of each variable. However, the system may rewrite some variables but they are meaningless. Variables without an initial value written are clarified in the description. (Some are not described.)
- Reference : Indicates chapters in the Software Version which give further information of each variable. When parentheses are used, a description of the variables is not written but a related discussion is given by the indicated chapter.

(1) System Work Area I (RSYSAR1) Variables in this area are initialized only when the system initialization process is executed. Address : Variable (Number of bytes) : Initial value (Type) : Reference FOOOH : RETADD (1) : C9H (R/O): 10.2 System Hook RET instruction (C9H) is set and system hook initial value indicates this address. FOO1H - FOO4H : For future use. FOOSH : BDSLAD (2) : 5E00H (EHT-10),6000H (EHT-10/2) (R/0) : (5.2) BDOS Process Leading address of RBDOS1. This address changes according to RAM disk and size of user BIOS area. F007H : BI1LAD (2) : 5F00H (EHT-10),6100H (EHT-10/2) (R/O) : (4.1.2) BIOS Process Leading address of RBIOS1. This address changes according to RAM disk and size of user BIOS area. FOO9H : For future use. FOOAH : MODEFG (1) : OOH (R/O) : (2.2) System State Transition This flag indicates the system module currently executing in the system. 6 5 4 3 2 7 1 0 Alarm screen is displaying (power off). - Alarm screen is displaying (power on). During system menu. Electronic calculator screen is displaying. Menu is processing. DLL is processing. System is initializing. WBOOT and BOOT are processing. FOOBH : SIZRAM (1) : 1FH (R/O) : (6.3) RAM Disk Indicates size of standard RAM of the RAM disk, unit is i KB. FOOCH : USERBIOS (1) : OOH (R/O) : 4.6 User BIOS Indicates size of user BIOS area, unit is 1 page (256 bytes). FOODH : RXON (1) : 11H : 4.2 BIOS (RSIOX) Code XON used in the XON/XOFF process when XON/XOFF is specified by BIOS RSIOX. FOOEH : RXOFF (1) : 11H (R/O) : 4.2 BIOS (RSIOX) Code XOFF used in the XON/XOFF process when XON/XOFF is specified by BIOS RSIOX.

FOOFH : For future use.

FOIOH : SRSADR (9) : See below (R/W) : 4.2 BIOS (PUNCH) Open parameter used when RS-232C I/F is used with an I/O device such as PUN:, LST:, RDR:, CON:. The configuration is the same as BIOS RSIOX open parameter.

Address	: Contents	:Initial	value	:Meaning
F010H	Receive buffer address	F95AH		-
F012H	Receive buffer size	0100H		
F014H	Transmission rate	ODH		4800 BPS
F015H	Bit length	03H		8 BITS
F016H	Parity	00H		NON Parity
F017H	Stop bit length	03H		2 BITS
F018H	Special parameter	FFH		DTR/RTS Active

- F019H : EXECTYPE (1) : 00H (R/W) :9.9.2 CONFIG Parameter Setting Indicates starting method of an application program. =00H : Automatic decision specification =01H : Forced DLL specification =02H : Forced execution specification
- FOIAH : PWONSOND (2) : 3687H (R/W) : (10.3) System Jump Table Indicates the data address of sound generation when the restart mode is on. Sound generation data must be after RAM address 8000 for a user to rewrite and the data configuration is the same as MELODY in the system jump table. The initial value indicates the data table of OS DOM, the contents are 01H, 37H, 00H.

	- ( PWONSON	D)		
->	Length	(Unit is 100 m	Sec)	
	Tone	(Refer to BIOS	call	BEEP)
	Length			
	1	]		
		=		
	00H	(End mark)		

Sound Generation Data Table

- FOICH : CNTNSOND (2) : 36BAH (R/W) : (10.3) System Jump Table Indicates the sound generation data address when the continue mode is on. Table configuration is the same as PWONSOND. The initial value indicates the data table of OS ROM, the contents are O3H, 37H, OOH.
- FOIEH : ALRMSOND (2) : 36BDH (R/W) : (10.3) System Jump Table Indicates the sound generation data address at alarm/wake. Table configuration is the same as PWONSOND. However, with alarm/wake, the sound generation data pattern repeats 3 times. The initial value indicates data table of OS ROM, the contents are OIH, 25H, OIH, 29H, OOH.

FO2OH : ATSHUTOFF (1) : O5H (R/W) : 2.3.5 Sleep Function and Auto Power Off Function. Indicates the time of auto power off in minutes, auto power is not off when OOH. As this variable is not used to check the time of real auto power off, ATSOTIME (FO21H, FO22H) must be modified when the time is changed. FO21H : ATOSTIME (2) : O12CH (R/W) : 2.3.5 Sleep Function and Auto Power Off Function Indicates the time of auto power off in seconds. The contents of ATSHUT OFF (F020H) must be multiplied by 60 to set. F023H : ELOFTIME (2) : OOB4H (R/W) : 2.3.5 Sleep Function and Auto Power Off Function Indicates the time of auto back light off in seconds. Auto back light does not turn off when 0000H. F025H : PRNPWTN (1) : B4H (R/W) : 4.5.4 Printer Unit Indicates the time of the power-down mode of the printer unit in seconds. When OOH is specified, power-down mode is disabled. FO26H : JCCPWIN (1) 3CH (R/W) : (11.3) IC Card Protocol Expansion Indicates the time in seconds after IC card access until the current supplied to IC lead is cut. When OOH is set, current is supplied until close. F027H : For future use. FO28H : ALRMTP (1) : OOH (R/W) : 4.2 BIOS (TIMDAT) Indicates the status of alarm/wake. =OOH : Not set =01H : Alarm set =02H : Wake set F029H : ALRMAD (2) : F32FH (R/W) : 4.2 BIOS (TIMDAT) This pointer indicates the address of the alarm message or wake string. This must be set after RAM address 8000H for user modification. FO2BH : ALRMST (1) : OOH (R/O) : 4.2 BIOS (TIMDAT) Indicates generation state of alarm/wake. =00H : For future generation =01H : Generated This is set to OOH by alarm/wake of BJOS TIMDAT or read alarm/wake is set to OlH by alarm interrupt. FO2CH : ALRMFG (1) : DOH (-) : 4.2 BIOS (TIMDAT) This flag checks whether the alarm interrupt is initial in the system. 7508 CPU returns the status of alarm generation at every second interrupt for 10 seconds after alarm interrupt is issued, the system ignores this by using this flag. =00H : Alarm interrupt is not ignored. =01H : Alarm interrupt is ignored. F02DH : TIMERO (2) : 0000H (R/O) : 8.4 7508 Interrupt Process This 16 bit up-counter is counted through a second interrupt and should not be rewritten by a user.

- FO2FH : TIMER1 (2) : 0000H (R/O) : 8.4 7508 Interrupt Process This 16 bit down-counter is counted through a second interrupt and should not be rewritten by a user.
- F031H : ISTS7508 (1) : OAH (R/O) : 4.2 BIOS (MASKI) This flag indicates the current state of interrupt disable/enable related to 7508. See MASKI in "4.2 BIOS Function" for further information.
- FO32H : TIMERIM (2) : 0000H (R/O) : 8.7 EXT Interrupt Process This 16 bit up-counter is counted by 1 msec/8 msec. The unit is always 1 msec, a user should not rewrite this counter.
- F034H : BNKDTTBL (40) : See below (R/W) : (1.4) Memory Map This table stores the bank data which is set to Bank Register (BANKR), Sub-bank Register (SUBBNKR). Refer to this table for bank switching in the system.

Address	:	Main bank	data	:	Sub-bank	data	:	Bank number
F034H,35H		02H			00H			System bank
36H,37H		02H			00H			System bank
38H,39H		02H			00H			System bank
3AH,3BH		02H			00H			System bank
3CH,3DH		<b>42</b> H			00H			Bank 0 # 0
3EH,3FH		42H			01H			Bank O 🕖 1
40H,41H		42H			02H			Bank 0 # 2
42H,43H		<b>4</b> 2H			<b>03</b> H			Bank O # 3
44H,45H		A2H			00H			(Dummy)
46H,47H		A2H			10H			Bank 1 # 1
<b>48</b> H,49H		A2H			<b>20</b> H			Bank 1 # 2
4AH,4BH		<b>A2</b> H			<b>30</b> H			Bank 1 # 3
4CH,4DH		E2H			00H			Bank 2 # 0
4EH,4FH		E2H			<b>4</b> 0H			Bank 2 # 1
50H,51H		E2H			80H			Bank 2 # 2
52H,53H		E2H			СОН			Bank 2 🛚 3
54H,55H		42H			04H			Bank O 🖸 4
56H, 57H		42H			05H			Bank 0 🖸 5
58H, 59H		42H			06H			Bank 0 🖸 6
5AH, 5BH		<b>4</b> 2H			07H			(Du <b>mmy)</b>

#### BNKDTBL Initial Value

- FO5CH : TOPRAM (2) : DCOOH (EHT-10), DEOOH (EHT-10/2) (R/O) :4.6 User BIOS Indicates leading address of user BIOS area. Initial value is the value when user BIOS size is page 0.
- FO5EH : For future use.
- F05FH : QT\_ROM\_CP1 (1) : OOH (R/O) : (6.4) ROM Socket Indicates ROM capacity which is set to ROM socket (ROM disk) in 1KB units.
- FOGOH : QT\_RAM\_IN (1) : OOH (R/O) : (4.6) User BIOS Indicates RAM disk capacity in 1 KB units.

- FO61H : DR\_ROM\_CP1 (1) : OOH (R/O) : (6.4) ROM Socket Indicates the number of ROM disk directories.
- FO62H : DR\_RAM\_IN (1) : OOH (R/O) : (6.3) RAM Disk Indicates RAM disk directory area size in 128 B units.
- FO63H : AD\_ROM\_CP1 (2) : OOOOH (R/O) : (6.4) ROM Socket Indicates the leading address (header position) of ROM disk.
- F065H : AD\_RAM\_IN (2) : 0000H (R/O) : (6.3) RAM Disk Indicates the leading address of the RAM disk. When the expansion RAM card is used, this indicates the leading address of standard RAM. If the expansion RAM disk is not added, the address is 0000H.
- FO67H : CS\_RAM\_IN (1) : OOH (R/O) : (6.3) RAM Disk Indicates RAM disk checksum area size in 128 B units.
- F068H : RAMD\_SIZE (1) : OOH (R/O) : (6.3) RAM Disk Indicates the expansion RAM size in 32 KB units, which is used in the RAM disk.
- F069H : RAM\_SET (1) : OOH (R/O) : (1.4) Memory Map Indicates the expansion RAM size in 32 KB units.
- FOGAH FO7OH : This is used as a work disk for the read/write process.
- F071H : QT\_RAM\_OLD (1) : OOH (R/O) : (4.6) User BIOS
   Stores the previous size at RAM disk size modification, the unit is 1
   KB.
- F072H : AD\_RAM\_OLD (2) : 0000H (R/O) : (4.6) User BIOS Indicates the leading address of the previous RAM disk at RAM disk size modification and user BIOS size modification.
- F075H : BLNKSTAT (1) : OOH (R/O) : 4.4.6 Cursor Display This flag indicates whether blink is specified. =OOH Blink =O1H No blink
- F076H : BLNKCNT (1) : 00H (-) : 4.4.6 Cursor Display This work area is used in the system in order to measure the blink interval.
- F077H : BLNKRVS (1) : OOH (R/G) : 4.4.6 Cursor Display Indicates the current cursor display state =OOH : Displayed =FFH : Not displayed
- F078H : BLNKTIME (1) : O4H (R/W) : 4.4.6 Cursor Display Specifies the cursor blink interval. The unit is 100 ms.

F079H : BTRYALM (1) : OOH (R/W) : (2.3.6) Power Failure This flag indicates whether the alarm should be disabled while power is off after power failure. =00H : Disabled =01H : Enabled FO7AH : DSKTFLG (1) : O1H (R/W) : 9.2.2 CONFIG Parameter Setting This flag indicates whether RAM disk checksum should be performed at power on. =01H : Checksum =00H : No checksum FO7BH : RAMTFLG (1) : O1H (R/W) : 9.2.2 ConFIG Parameter Setting This flag indicates whether RAM checksum should be performed at power on. =01H : Checksum =00H : No checksum FO7CH : BASICF (1) : O3H (R/W) : 9.2.2 CONFIG Parameter Setting Indicates the number of files which can be opened in BASIC at a time. F07DH : BASICR (2) : 0080H (R/W) : 9.2.2 CONFIG Parameter Setting Random record size of files handled in BASIC. F07FH : BASICB (2) : 0100H (R/W) : 9.2.2 CONFIG Parameter Setting Size of data send buffer handled in BASIC. FO81H : SYSINFLG (1) : FFH (R/W) : (2.3.1) System Initialize This flag indicates system initialization has been processed. =FFH : System initialization has been processed. =OOH : System initialization has not been processed. FO82H : SSYSMMOD (1) : OFH (R/W) : (9.2.1) System Menu Control This flag disables the system menu functions during the MENU and DLL processes and is copied to SYSMMOD (F6BBH) at WBOOT. See SYSMMOD for bit configuration. FO83H : USYSMMOD (1) : 01H (R/W) : 9.2.1 System Menu Control This flag disables the system menu functions during application execution and is copied to SYSMMOD (F6BBH) at application start. See SYSMMOD for bit configuration. FO84H : SCONFMOD (2) : 07FFH (R/W) : 9.2.1 System Menu Control This flag disables CONFIG functions during the Menu and DLL processes and is copied to CONFMOD (16BCH) at WBOOT. See CONFMOD for bit configuration. FO86H : UCONFMOD (2) : O7EFH (R/W) : 9.2.1 System Menu Control This flag disables CONFIG functions during application execution and is copied to CONFMOD (F6BCH) at application start. See CONFMOD for bit configuration.

F088H - F08FH : For future use.

(2) System Work Area II (RSYSAR2)

Variables in this area are initialized when the system initialization or reset process is executed. Address : Variable (Number of bytes) : Initial value (Type) : Reference F090H : For future use. FO91H : INBIOS (1) : OOH (R/W) : (4.1.2) BIOS Process This flag indicates whether the BIOS process is being executed. =00H : BIOS not processing =FFH : BIOS processing F092H : INTLEVEL (1) : FFH (R/O) : 8.4 7508 Interrupt Process This flag indicates 7508 interrupt level. =FFH : Not 7508 interrupt processing =00H : 7508 interrupt processing FO93H : INTTYPE (1) : OOH (R/W) : 8.9 Interrupt Process Expansion This flag indicates interrupt generation state. F094H : BUZ FLG (1) : FFH (R/W) : 4.3 Key Input This flag disables key input sound. =OOH : Disabled =FFH : Enabled F095H : KDFLTBZ (2) : 0064H (R/W) : 4.3 Key Input Key input sound length (1 ms unit). F097H : BUZZHZ (1) : 80H (R/W) : 4.3 Key Input Key input sound frequency =40H : 512 Hz =80H : 1024 Hz =COH : 2048 Hz F098H : INTVLFG (1) : OOH (R/W) : 8.8 EXT Interrupt Process This flag indicates the module which uses the timer interrupt of 1 msec/8 msec. F099H : INTS7508 (10) : See below (R/O) : (8) Interrupt Process Interrupt stack area leading address table Address Variable Value Contents FF99H.9AH INTS7508 FE3FH For 7508 interrupts 9BH.9CH INTS8251 FDC7H For ART interrupts 9DH,9EH INTSICF FD5FH For ICF interrupts 9FH, AOH INTSOVE FD97H For OVF interrupts AlH, A2H INTSEXT FD2FH For EXT interrupts FOA3H : TBL7508 (16) : -(R/0) : (8.4) 7508 Interrupt Process Interrupt type process distribution table at 7508 interrupt generation. FOB3H : BTRYFG (1) : OOH (R/O) : (2.3.6) Power Failure Power failure interrupt generation flag. =00H : For future generation =FFH : Generated

- FOB4H : YPOFDS (1) : 00H (R/W) : 8.4 7508 Interrupt Process This flag temporarily disables the power-off process. This flag is copied to YPOFST (FOB5H) at the power-off interrupt process.
- FOB5H : YPOFST (1) OOH (R/W) : 8.4 7508 Interrupt Process This flag indicates power-off interrupt generation state when the power-off process is disabled by YPOFDS. The value of YPOFDS (FOB4H) is copied at power-off interrupt with this flag.
- FOB6H : YALMDS (1) : OOH (R/W) : 8.4 7508 Interrupt Process This flag temporarily disables the alarm/wake process. The alarm interrupt process copies this flag to YALMST (FOB7H).
- FOB7H : YALMST (1) : OOH (R/W) : 8.4 7508 Interrupt Process This flag indicates the alarm interrupt generation state when the alarm/wake process is disabled by YALMDS (FOB6H). The alarm interrupt process copies the value of YALMDS of this flag.
- FOB8H : SMENUFLG (1) : OOH (-) : (9.3) MENU Process Expansion This flag indicates whether there is modification of the disk state or execution of the system menu functions during the MENU process.
- FUB9H : MDRV (5) : See below (R/W) : 9.3 MENU Process Expansion Specifies a maximum of 4 drives for the MENU process. Initial values are as follows.

Address	Initial value	Meaning
FOB9H	01H	Drive A
BAH	02H	Drive B
BBH	03H	Drive C
BCH	FFH	End mark
BDH	FFH	End mark

FUBEH : MFTYP (10) : See below (R/W) : 9.3 MENU Process Expansion Specifies a maximum of 3 file extension names for the MENU process. Initial values are as follows.

Address Initial value FOBEH - COH "COM" FOC1H - C3H "BAS" FOC4H - C6H FFH,FFH,FFH FOC7H FFH

When not specified, end mark (FFH) is filled.

FOC8H - FOD5H : For future use.

FOD6H : RZCTLR1 (1) : 61H (R/W) : Hardware Part Chapter 2 Area to store the output state of control register 1 (CTLR1 : POOH), bit configuration is the same as CTLR1.

FOD7H : For future use.

FOD8H : RZCTLR2 (1) : O3H (R/W) : Hardware Part Chapter 2 Area to store the output state of control register 2 (CTLR2 :PO2H), bit configuration is the same as CTLR2.

- FOD9H : RZARTMR (1) : 84H (R/W) : Hardware Part Chapter 2 Area to store the output state of the ART mode register(ARTMR : P15H), bit configuration is the same as ARTMR.
- FODAH : RZARTCR (1) : OOH (R/W) : Hardware Part Chapter 2 Area to store the output state of ART command register (ARTCR: P16H), bit configuration is the same as ARTCR.
- FODBH : RZICCTLR (1) : C4H (R/W) : Hardware Part Chapter 2 Area to store the output state of IC card control register(ICCTLR : P17H), bit configuration is the same as ICCTLR.
- FODCH : RZSWR (1) : O8H (R/W) : Hardware Part Chapter 2 Area to store the output state of switch register (SWR :P18H), bit configuration is the same as SWR.
- FODDH : RZIOCTLR (1) : O6H (R/W) : Hardware Part Chapter 2 Area to store the output state of 10 control register (IOCTLR : P19H), bit configuration is the same as IOCTLR.
- FODEH : RZCTLR3 (1) : O4H (R/W) : Hardware Part Chapter 2 Area to store the output state of control register 3 (CTLR3: P23H), bit configuration is the same as CTLR3.
- FODFH : USERCRG (1) : OOH (R/W) : 11.4 Cartridge Device Expansion
  This flag indicates whether the mount check process hook(CRGHOOK) must
  be processed when a user expands the cartridge device.
  =OOH : No hook process
  #OOH : Hook process
- FOEOH : AHSTWRT (1) : OOH (R/O) : (6.6.4) Protocol Indicates whether write pending data to FDD (drive D:, E:) resides. =OOH : No pending data =31H : Pending data This flag is set when write/read to/from FDD is performed.
- FOE1H : ADSKOPN (1) : OOH (R/O) : (6.6.4) Protocol This flag indicates the open state of FDD (drive D:, E:). =OOH : Open =FFH : Not open This flag is set when BIOS SELDSK is executed to FDD.
- FOE2H : DSKDID (1) : 31H (R/O) : (6.6.4) Protocol Indicates FDD device code in DID code of EPSP for FDD send/receive.
- FOE3H : DSKSID (1) : 23H (R/O) : (6.6.4) Protocol Indicates EHT-10/EHT-10/2 code in SID code of EPSP for FDD send/receive.
- FOE4H : DSKTBL (5) : (R/W) : 6.2 Logical Drive and Physical Drive Logical drive and physical drive corresponding table.
- FOE9H : DISKROV (2) : OUO2H (R/W) : 6.2 Logical Drive and Physical úrive Indicates disk read only status.
- FOEBH : FTSTAB (10) : (R/W) : (6) Disk System Description Jump table for the initial select process of each drive at BJOS SELDSK.

FOF5H : READTAB (10) : - (R/W) : (6) Disk System Description Jump table to the read process routine of each drive at BIOS READ.

- FOFFH : WRTTAB (10) : (R/W) : (6) Disk System Description Jump table to the write process routine of each drive at BIOS WRITE.
- F109H : DPBASE (80) : See below (R/O) : (6) Disk System Description Disk parameter header corresponding to each drive has a 16 byte configuration and is sequentially set from drive A:. Disk parameter header configuration is as shown below.

Relative address



- F159H : DPBO (60) : (R/O) : (6) Disk System Description Disk parameter block corresponding to each drive is sequentially set from A: in a 15 byte configuration for each drive after DPBO. However, drive D and E have the same configuration and use the same parameter block.
- F195H : For future use.
- F196H : RSTABL (15) : (R/W) : 7.2 Serial Interface This serial data send/receive parameter packet modifies the serial switch in the system. Configuration is shown in 7.2.4.
- F1A5H : EPTRCN (1) : 03H (R/W) : (6.6.4) Protocol
   Specifies the number of retries at no response at EPSP data send/recei ve.
- F1A6H : EPTIMO (1) : OAH (R/W) : (6.6.4) Protocol
  Specifies the number of retries at time-over at 1 byte of EPSP data
  receive.
- F1A7H : EPEIMO (1) : 64H (R/W) : (6.6.4) Protocol Specifies the number of retries at time-over at 1 block of EPSP data receive.
- FIA8H : EPATMO (1) : OAH (R/W) : (6.6.4) Protocol Specifies the number of retries at time-over when ACK is received by EPSP protocol.

F1A9H : EPMODE (1) : OOH (R/W) : (6.6.4) Protocol
This flag specifies the mode at EPSP data send/receive.
=00H : Master mode (when header is sent)
#00H : Slave mode (when header send is omitted)

F1AAH : For future use.

- F1ABH : ET1BRTO (2) : 1C2FH (R/W) : (6.6.4) Protocol
   Specifies the time until time-over at 1 byte of EPSP data receive.
   Unit is approx. 13.86 micro-seconds, initial value is approx 100 msec
   and is time-over.
- FIADH : ICRSTCNT (1) : 09H (R/W) : 11.3 IC card Protocol Expansion
   Specifies the number of receive bytes when the answer to reset
   = 00H : Ignore the answer to reset
   # 00H : number of receive bytes.
- F1AEH : ICRSTPNT (2) : F663H (R/W) : 11.3 IC card Protocol Expansion This pointer indicates the data address for IC card reset response check.
- F1BOH : IDFLTCNT (1) : O3H (R/W) : 11.3 IC card Protocol Expansion Specifies the number of retries at IC card send/receive.
- FIGIH : ICCDTIME (2) : 0003H (K/W) : 11.3 IC card Protocol Expansion The timeout time when no response from the IC card (unit 1 sec).
- F1B3H : ICCDPRM (5) : See below (R/W): 11.3 IC Load Protocol Expansion Serial parameter packet for send/receive to/from IC card. Initial value is Oth,O3H,O3H,O1H,FFH
- F1B8H : RLCGENX (3) : See below (R/W) : 4.4.5 Character Generator F1BBH : RLCGENN (3) F1BEH : RLCGENG (3)
- FICIH : RLCGENK (3)
  - 4 types of character generator table pointer data. See 4.4.5 Character Generator for further information.
- FIC4H : LTOUCHLD (3) : (-) : (4.2) BIOS (TOUCH)
  Indicates routine address for TOUCH redisplay (used only in the
  system).
  - Address Meaning F1C4H,C5H Routine address of TOUCH redisplay. C6H For future use.

F1C7H - F1C8H : For future use

F1C9H : THSYSFLG (1) : OOH (R/W) : (4.2) BIOS (TOUCH)
This flag speeds up the BIOS TOUCH process.
=00H : Normal mode
=01H : Retrieves display data from the system bank.
(For use by a user, data must be retrieved after RAM address
BOOOH.)
=FFH : Omits key definition besides the process mentioned above.(01H)

FICAH : ALMTIME : 32H (R/W) : (2.4) Alarm/Wake Specifies alarm screen display time (unit : second). F1CBH : For future use FICCH : PRTINIT (1) : COH (R/W) : (4.5.3) Destination Process Specifies whether the initial printer process corresponding to each country is performed. bit 7 RS-232C bit 6 Printer unit = 1: Initialization = 0: No initialization F1CDH : TCAMPRM (7) : - (R/W) : 4.2 BIOS (TCAM) Start condition is stored at BIOS TCAM open. This area includes protocol, communication parameter ... etc. See BIOS TCAM for details. F1D4H : TDFLTCNT (1) : 03H (R/W) : 4.2 BIOS (TCAM) Specifies the number of retries when Filink protocol is specified at BIOS TCAM and the host protocol does not coincide. F1D5H : T1STTIME (2) : 001EH (R/W) : 4.2 BIOS (TCAM) Specifies timeout time until initial data is received at BIOS TCAM (unit : second). F1D7H : T2NDTIME (2) : 0003H (R/W) : 4.2 BIOS (TCAM) Specifies timeout time until subsequent data data is received at BIOS TCAM (unit : second). F1D9H : T1STFLG (1) : OOH (R/O) : 4.2 BIOS (TCAM) This flag identifies BIOS TCAM "initial time", "subsequent time". =OOH : Initial time =01H : Subsequent time =FFH : For future receive F1DAH : For future use. F1DBH : DRCVCNT (2) : 0080H (R/W) : 11.2 Communication Protocol Expansion Indicates area size for data storage at DLL, DL/UL. F1DDH : DRCVBUF (2) : F85AH (R/W) : 11.2 Communication Protocol Expansion Indicates the area of the leading address for data storage at DLL, DL/UL. F1DFH : DSKNUM (1) : 01H (R/W) : 9.2 System Menu Disk numbers for DL/UL, (DLL). F1EOH : RNDFIG (1) : FFH (R/W) : 9.2.2 CONFIG Parameter Setting Specifies the display mode after the decimal point of the electronic calculator. =FFH : Float mode #FFH : Number of display columns after the decimalpoint. FlElH : For future use



- F1E7H : ICWAIT1 (1) : 32H (R/W) : 11.3 IC card Protocol Expansion Specifies the interval (in ms units) for awaiting the start of Reset response reception after the Reset status is canceled. Default value = 50 ms(32H)
- F1E8H : ICWAIT2 (1) : 64H (R/W) : 11.3 IC card Protocol Expansion Specifies the interval (in ms units) for awaiting the start of command transmission after the Reset response is received. Default value = 100 ms(64H)
- F1E9H : ICTSDT (1) : 3BH (R/W) : 11.3 IC card Protocol Expansion TS byte collation data at IC card reset response. The initial value (3BH) is LSB First, even parity, Z=1. Collate this data and receive data at reset response, if same, move to the receive process after TO byte. If different, an error occurs.

F1EAH - F1FFH : For future use.

(3) System work Area III (RSYSAR3)

Variables in this area are initialized when system initialization, reset, restart or the WBOOT process is executed. Address : Variable (Number of bytes) : Initial value (Type) : Reference F200H : CNTFG (1) : FFH (R/0) : (2.3) System Start and End This flag indicates the restart/continue power-off mode. It is set at power off and is checked at power on. =00H : Continue mode =FFH : Restart mode F201H : FRCECNTN (1) : FFH (R/W) : (2.3) System Start and End Indicates the power-off mode during application execution. =OOH : Restart mode =FFH : Continue mode F202H : ICCPWCNT (1) : 3CH (R/O) : (11.3) IC Card Protocol Expansion This timer counts time after IC card access till power supply stops automatically. F2O3H : ICCPWFLG (1) : OOH (R/O) : (1.3) IC Card Protocol Expansion This flag indicates the power supply state after IC card access. =01H : ON =OUH : OFF =FFH : Time for OFF (still ON) F204H : ELCTFLG (1) : OOH (R/O) : 2.3.5 Sleep Function and Auto Power Off Function This is used in the system as EL back light. F205H : TMFUNC (1) : OOH (R/W) : (8.4) 7508 Interrupt Process Indicates valid/invalid for the user 1 sec timer function. =00H : Invalid **≠**00H : Valid F206H : TMFLAG (1) : OOH (R/W) : (8.4) 7508 Interrupt Process This flag indicates whether specified time has elapsed at user 1 sec timer function specification. =OOH : Specified time not elapsed =FFH : Specified time elapsed This flag decrements TMSEC (F3C6H) by one using a 1 second interrupt, when it becomes OOOOH, it is set to FFH. F207H : ROMEXQ (1) : OOH (R/O) : (3.4) ROM Execution Type Program Creation This flag is used in the system to start ROM execution type program. =00H : Not ROM execution type ≠00H : ROM execution type F208H : DBGFLG (1) : OOH (R/O) : (10.3) System Jump Table This flag indicates the normal mode/development cartridge mode for the current cartridoe mode. =00H : Normal mode =FFH : Development cartridge mode

F209H : COMPCOL (1) : OOH (-) : (5) BDOS F20AH : STRCOL (1) : OOH (-)F20BH : COLMN (1) : OOH (-) This work area is used for character display by BDOS. F20CH : LISTCP (1) : OOH (R/W) : (5) BDOS This flag indicates whether LST: output is performed at CON: output by BDOS P(IOH). =00H : No LST: output (normal mode) =01H : LST: output (<sup>^</sup>P mode) F20DH : KPCHAR (1) : OOH (R/O) : (5) BDOS CON: input buffer in BDOS =OOH : No input ≠00H : Input (key code) F20EH : USRCODE (1) : OOH (R/O) : (5) BDOS Current user number F20FH : CURDSK (1) : OOH (R/O) : (5) BDOS Current disk number F210H : EFCB (1) : E5H (-) : (5) BDOS This work area is used to search the empty directory section area at BDOS make file function execution. F211H : RODSK (2) : 0000H (R/W) : (5) BDOS This read only vector is controlled by BDOS. 7654321076543210 000000 0 00 OE 0 0 D C В =0 : R/W->-=1 : R/0F213H : DLOG (2) : 0000H (R/O) : (5) BDOS Indicates the disk in login. 7654321076543210 01010 0 0 0 0 0 Ε C В 0 0 D =0 : Not log in -F214H--->┴<---F213H-=1 : Log in F215H : DMAAD (2) : 0000H (R/0) : (5) BDOSDMA address used in BDOS. (Normally set to SYSDMA F851H). F217H : USRDMA (2) : 0000H (R/W) : (5) BDOS Indicates user DMA address which is set by BDOS SETDMA function. F219H : For future use F21AH : KEYF (1) : OOH (R/O) : 4.3 Key Input Indicates the input state of the console buffer (KEYD:F21BH). =00H : No input data =FFH : Input data APPENDIX Page 21 - 30

F21BH : KEYD (1) : OOH (R/O) : 4.3 Key Input Console input buffer in which key function code is set. F21CH : KEYS (1) : OOH (R/O) : 4.3 Key Input Console input buffer in which key location code is set. F21DH : YPFCMFLG (1) : OOH (R/W) : 4.3 Key Input Function key check mode flag =OOH : Function EO to FEH is executed. =FFH : Function EO to FEH is not executed and function code is returned. F21EH : SKEYFLG (2) : 0000H (R/W) : 4.3 Key Input This flag makes all functions invalid for function code FOH to FFH. F220H : CSTOPFLG (1) : 00H (R/W) : 4.3 Key Input I/O forced break flag by function code F6H. =00H : Function code F6H key not pressed. =01H : Function code F6H key pressed. F221H : KB FLG (1) : OOH (-) : (4.3) Key Input This flag indicates the key input mode modification (normal, alphanumerics, kana, electronic calculator) in the key interrupt process. =OOH : No input mode modification =FFH : Input mode modification F222H : KALMMOD (1) : OOH (-) : (4.3) Key Input This flag indicates conversion of Roman/Kana characters in the EHT-10/2 Kana mode. =00H : conversion =FFH : No conversion F223H : SMKFLG (1) : OOH (R/O) : (4.3) Key Input This flag indicates lowercase letter input in the EHT-10 Kana mode. =OOH : Normally kana input mode =01H : Lowercase kana input mode F224H : RM SIINFLG (1) : OOH (-) : (4.3) Key Input This flag indicates the number of the consonant inputs in the EHT-10/2 kana mode. F225H : RM HATSUFLG (1) : OOH (-) : (4.3) Key Input Indicates syllabic nasal generation in the EHT-10/2 kana mode. =00H : No syllabic nasal generated =FFH : Syllabic nasal generated F226H : RM SOKUFLG (1) : OOH (-) : (4.3) Key Input This flag indicates double consonant generation in the EHT-10/2 kana mode. =OOH : No double consonant generated =FFH : Double consonant generated.
F227H : KMODE (2) : FBOOH (R/O) : (4.3) Key Input Indicates the current key input mode. Address Initial value Meaning F227H User/system 00H =OOH : User mode =01H : System mode F228H FBH Key mode =FEH : Electronic calculator mode =FDH : Alphanumeric (alphabet) mode =FCH : Kana mode =FBH : Normal mode F229H : YPFKCNT (1) : OOH (R/O) : 4.3 Key Input Indicates the number of characters which are not returned to a user at character string input (electronic calculator or '000' key input). F22AH : RM FSIIN (1) : OOH (-) : (4.3) Key Input Area to save consonant codes which is first input in the EHT-10/2 kana input mode. F22BH : RM\_BJONCD (1) : OOH (-) : (4.3) Key Input Area to save vowel code in the EHT-10/2 kana input mode. F22CH : RM KEYDTR (1) : OOH (-) : (4.3) Key Input Area to save consonant code in the EHT-10/2 kana input mode. F22DH : PUT KEYPTR (2) : F7E9H (R/O) : 4.3 Key Input Key input buffer (KEY BUFFER) put pointer. F22FH : GET KEY<sup>D</sup>TR (2) : F7E9H (R/O) : 4.3 Key Input Key input buffer (KEY BUFFER) get pointer. F231H : PFKPTR (2) : F434H (R/W) : 4.3 Key input Indicates the leading address of the input character string at character string input. F233H : KTABPTR (2) : - (R/W) : (4.3) Key Input Indicates the head of the pointer table of the key table. Address Meaning Normal table (KTABPTR) +0 +2 Kana table Alphanumeric table +4 Electronic calculator table +6 F235H : PTR KEYTBL (2) : - (R/W) : (4.3) Key Input Indicates the head of the key table in the current key input mode. F237H : RM BUFPTR (2) : - (R/O) : (4.3) Key Input Indicates the head of the area of converted roman/kana character data storage in the EHT-10/2 kana input mode. F239H : USERKTBL (2) : - (R/O) : (4.3)Key Input Indicates the head of user normal key table. F23BH : BPINTEBL (1) : O1H (R/W) : 4.2 BIOS (BEEP) This flag disables interrupts during BEEP. APPENDIX Page 21 - 32

F23CH : INHCOPY (1) : OOH (R/O) : (4.2) BIOS (SCRNDUMP) This flag indicates that screen dump is being executed in EHT-10/2. =00H : Screen dump not executing ≠00H : Screen dump executing F23DH : For future use. F23EH : PRTFLG (1) : OOH (R/W) : 4.5 Printer This flag indicates whether the destination process for the printer has been executed. =00H : Future output ≠00H : Output The destination process must be executed to the printer after I/O byte modification and display character set modification, even when PRTFLG # **OOH**. F23FH : DISBRK (1) : OOH (R/O) : (4.3) Key input This flag makes the I/O process forced break invalid temporarily using function key (F6H). =00H : I/O process forced break valid #OOH : I/O process forced break invalid. F240H : For future use. F241H : SRSMODE (1) : FFH (R/W) : 7.2 Serial Interface This flag indicates current serial use status. =FFH : For future use =00H : IC card (DISK) in use =01H : Being used by a user =02H : Being used by FDD F242H : SCRLT3 (158) : - (-) : 4.4 LCD display Work area related to display. See 4.4.7 Work Area Related to Display for further information. F2E1H : MEMK (1) : OOH (R/O) : -Indicates the index number of the electronic calculator memory data (radix is 10). F2E2H : MEMO (1) : OOH (-) : -Work area used for operation of electronic calculator memory. F2E3H : MEMD (10) : - (R/O) : -Electronic calculator memory data mantissa. F2EDH - F2FFH : For future use.

(4) System Work Area IV (RSYSAR4) Variables in this area are generally not initialized. However, some variables set an initial value in the system. Address : Variable (Number of bytes) : Type : Reference F300H : XUSERBIOE (3) : R/O : (4.6) User BIOS Instruction to jump to the leading address in the user BIOS area is stored. F303H : CNTNSP (2) : R/O : (2.3.4) Continue Mode Area to save the current stack pointer at power off in the continue mode. F305H : CNTNILVL (2) : R/O : (2.3.4) Continue Mode Area to save the current 7508 interrupt level at power off in the continue mode. F307H : YPWSWST (1) : R/O : (2.3) System Start and End This area stores the current power switch state. =00H : Power switch off =01H : Power switch on 5305h : YMAINST (1) : 7/0 : (2.3) System Start and End This flag checks the main CPU power supply state at adoress 0 start (power-on, reset, system initialization). =OOH : Address O started from power-off state. =01H : Address 0 started from power-on state. This flag is normally set to OlH, when power-off command is output to 7508 CPU, it becomes OOH. IJO9H : ZSTARTFG (1) : R/O : (2.3) System Start and End This flag distinguishes system address O start. =00H : WBOOT =01H : Power ON =02H : Alarm (power off) =03H : Wake (power off) =05H : Reset =O6H : System initialize F30AH - F30CH : For future use. F30DH : EXESTRNG (34) : R/W : 9.2.2 CONFIG Parameter Setting Area to set execute filename and start parameter at forced selection specification. First byte is the length of the character string. F32FH : ALRMMSG (34) : R/W : (2.4) Alarm/Wake Area for alarm message/wake string setting. First byte is the length of the character string. F351H : STIMEBUF (24) : - : -Work area used for time setting and time display in CONFIG. F369H : TIMEWK (32) : - : -Work area used for time setting and time display in CONFIG.

F389H : SRSTMODE (1) : R/W : 9.4 Process Expansion at Power On Area to store the main power supply state (YMAINST : F308H) at address O start. The system uses this as the flag which indicates whether RAM check must be performed in the reset process. It is also used to check whether power is turned on/off in a user program. =OOH : Power is not turned on/off ≠00H : Power is turned on/off F38AH : MENUMOD (45) : - : -45 byte work area used in the MENU process. F3B7H : ELOFFEND (2) : R/W : 2.3.5 Sleep Function and Auto Power Off Function Time for auto back light off is set to this area by the auto back light off function. F3B9H : BUZZLNG (2) : - : (4.3) Key Input This counter counts key input sound length, the value of KDFLTBZ (F09511) is copied first. F3BBH : RSINTST (1) : R/W : (8.5) ART Input This flag indicates data has been received from serial I/F. =00H : No data received =OlH : Data received F3BCH : BCINTST (1) : R/W : (8.6) ICF Interrupt This flag indicates bar code is read. =00H : No bar code read =01H : Bar code read F3BDH : PRNPWCNT (1) : R/W : (4.5) Printer This counter counts the time of the power-down mode in printer units. The value of PRNPWTM (F025H) is copied first and decrementation is performed by 1 second interrupt. F3BEH : PRNPWFLG (1) : R/W : (4.5) Printer This flag indicates printer unit power supply state. =OlH : Normal mode =00H : Power-down mode =FFH : Time for power-down (power-down has not been performed) F3BFH : TBUZ FLG (1) : - : (4.3) Key Input This work area is used for key input sound or buzzing in the BIOS BEEP process. F3COH : ENTSINT (2) : R/O : (8) Interrupt process Area to save the current stack in the interrupt process. F3C2H : INTFG (1) : R/W : (8.4) 7508 Interrupt This flag indicates the type processed in the 7508 interrupt process. F3C3H : STS7508 (1) : R/O : (8.4) 7508 Interrupt Area to store status read from 7508 by OS at 7508 interrupt generation.

F3C4H : FG7508 (1) : R/W : (8.4) 7508 Interrupt Indicates the process execution status of 1 second interrupt and alarm interrupt. 7 6 5 4 3 2 1 0 . -> Initial alarm interrupt -> 1 second interrupt (All bytes indicate the process is performed at 1.) F3C5H : ALRMCT (1) : R/O : (8.4) 7508 Interrupt This flag ignores alarm interrupt within 10 seconds after initial alarm interrupt generation. F3C6H : TMSEC (2) : R/W : (8.4) 7508 Interrupt This area specifies user timer execution time in seconds. Timer starts at TMFUNC (F205H). F3C8H : PWSWONFLG (1) : R/W : (8.4) 7508 Interrupt This flag indicates power switch on interrupt generation. =00H : Not generated =FFH : Generated 73C9H : PWSWOFFG (1) : R/W : (8.4) 7508 Interrupt This flag indicates power switch off interrupt generation. =00H : Not generated =01H : Generated F3CAH : AFTER3 (1) : - : (8.4) 7508 InterruptArea used to decide time for alarm display end. F3CBH - F3CCH : For future use. F3CDH : BDOLT4 (66) : R/O : 5. BDOS Used as a BOOS work area. F40FH : USRSBD (2) : R/O : (5) BDOSArea to save the user stack pointer in the resident BDOS (RBDOS) process. F411H : USRFCB (2) : R/O : (5) BDOS Area to store user specified FCB address in the resident BDOS (RBDOS) process. F413H : BDOSFN (1) : R/O : (5) BDOS Area to store user specified BDOS function number in the resident BDOS (RBDOS) process. F414H : BDSBNK (1) : R/O : (5) BDOS Area to store user BDOS call bank data in the resident BDOS (RBDOS) process.

F415H : RIOBYTE (1) : R/W : (2.6.5) I/O Byte Area to store the contents of the current 1/0 byte (address 0003H) when the resident BDOS (RBDOS) and BIOS (RBIOS) are called. RIOBYTE configuration is the same as I/O byte. F416H : ERRFLG (1) : R/O : (5.3) BDOS Error This flag indicates the current BDOS error process mode. =00H : RSTERR mode (normal mode) =FFH : SETERR mode F417H : BIOSERROR (1) : R/O : 4.2 BIOS (READ) (WRITE) See 4.2 BIOS Function READ for BIOS error code configuration at disk read/write. F418H : OLDBNK (1) : R/O : (4.1.2) BIOS Process Arca to store the user BIOS call bank data in the resident BIOS (RBIOS) process. F419H : SVOLDBNK (1) : - : -Area to save OLDBNK contents in the system. F41AH : For future use. F416H : DISBNK (1) : P/W : 4.2 BIOS (CALLX) (JUMPX) (LDIRX) This bank data indicates the destination of BIOS CALLX, destination of JUMPX, jump target and LDIRX data transfer target. (This is user set.) F41CH : CALBNK (1) : - : (4.2) BIOS (CALLX)Area to store bank data returned at BIOS CALLX. F41DH : SRCBNK (1) : R/W : 4.2 BIOS (LDIRX) Area to store bank data which has transferring source data at BIOS LDIRX. (This is user set.) F41EH : RETBNK (1) : - : -Area to store bank data returned after process at BIOS LDIRX. F41FH : SBNKDT (2) : 2 : -This data changed the transferring source bank data to 1/0 register output at BIOS LDIRX. F421H : DBNKDT (2) : - : -This data changed the transferring source bank data to 1/0 register output at BIOS LDIRX. F423H : CURBNK (1) : R/O : (1.4) Memory Map Area to store current bank data. F424H : USRSBI (2) : R/O : (4.1.2) BIOS Process Area to save the user stack pointer when BIOS call is performed in the resident BIOS (RBIOS) process. F426H : BIOSFN (2) : R/O : (4.1.2) BIOS Process Area to store called BIOS function numbers (EBxxH) in resident BIOS (RBIOS) process.

F428H : ROMEXQON (1) : - : (3.4) ROM Execute Type Program Creation
This flag is for ROM execute decision when ROM execute type program
starts in the system.
=00H : ROM executed
=01H : ROM not executed

- F429H : ROMEXBNK (1) : : (3.4) ROM Execute Type Program Creation Bank data when moving control of ROM execute type program.
- F42AH : ROMEXADD (2) : : (3.4) ROM Execute Type Program Creation Address when moving control to ROM execute type program.
- F42CH : RZBANKR (1) : R/W : Hardware 2. IO Register Description Area to store the output state of bank register (PO5H). The configuration is the same as BANKR (PO5H).
- F42DH : RZSBBNKR (1) : R/W : Hardware 2. IO Register Description Area to store the output state of Sub-bank Register (PO4H). The configuration is the same as SBBANKR (P22H).
- F42EH : RZIER (1) : R/W : Hardware 2. IO Description Area to store the output state of interrupt enable register (PO4H). The configuration is the same as IER (PO4H).
- F42FH : CRGDEV (1) : R/W : 11.4 Cartridge Device Expansion Area to store the device number of the optional unit connected to cartridge I/F. It is set to OOH when not installed.
- F430H : SAVEIX (2) : R/O : (4.1.2) BIOS Process Area to save IX register when BIOS is called.
- F432H : SAVEIY (2) : R/O : (4.1.2) BIOS Process Area to save IY register when BIOS is called.
- F434H : YPFKBUF (32) : R/W : 4.3 Key Input This buffer stores the character string at character string input in the alphanumeric mode or kana mode of the electronic calculator and EHT-10.
- F454H : RM\_BUF (6) : : (4.3) Key Input This buffer stores converted kana code in EHT-10/2 kana mode.
- F45AH : FUNC\_TBL (96) : R/W : (4.3) Key Input Function bank and address for function code EOH to FEH is set in a 3 byte configuration.
- F4BAH : SVKMODE (2) : : (4.3) Key Input Area to save the current key input mode KMODE (F227H) in the system.
- F4BCH : SVPFMOD (1) : : (4.3) Key Input Area to save YPFCMFLG (F21DH) in the system.
- F4BDH : PPUTPTR (2) : R/O : 4.5 Printer PUT pointer to buffer PRNBUF (FB2AH) for printer unit output.
- F4BFH : PGETPTR (2) : R/O : 4.5 Printer GET pointer from PRNBUF (FB2AH) for printer unit output.

F4C1H : SEKDSK (1) : R/O : (4.2) BIOS (SELDSK) Physical number of the disk selected by BIOS SELDSK is set. F4C2H : SEKTRK (2) : R/O : (4.2) BIOS (SETTRK) Track number which is set by BIOS SETTRK is stored. F4C4H : SEKSEC (1) : R/O : (4.2) BIOS (SETSEC) Sector number which is set by BIOS SETSEC is stored. F4C5H - F4D4H : For future use. F4D5H : DMAADR (2) : R/O : (6) Disk System Description DMA buffer address used by BIOS at disk read/write is set. It points to SYSDMA (F851H) normally. F4D7H : DIRBUF (128) : R/W : (5) BDOS DMA buffer used by BDOS at directory read/write. F557H : ALVO (29) : R/O : (5) BDOS Allocation area for RAM disk (A:) F574H : CSVO (0) : - : (5) BDOS Checksum area for RAM disk (A:) (Only label is defined.) 557 h : ALV1 (16) : R/C : (5) BDOS Allocation area for ROM disk (B:) 1584H : CSV1 (0) : - : (5) BDOS Checksum area for ROM disk (B:) (Only label is defined.) F584H : ALV2 (8) : R/O : (5) BDOS Allocation area for IC card (C:)  $F_{2} = CSV2 (16) : R/O : (5) BDOS$ Checksum area for 1C card. F59CH : ALV3 (18) : R/O : (5) BDOS Allocation area for FDD (D:) **F5AEH : CSV3 (16) : R/O : (5) BDOS** Checksum area for FDD (D:) F5BEH : ALV4 (18) : R/O : (5) BDOS Allocation area for FDD (E:) r5DOH : CSV4 (16) : R/O : (5) BDOS Checksum area for FDD (E:) F5E0H - F5E1H : For future use. F5E2H : DISK\_BNK (1) : - : (6) Disk System Description Indicate the bank where the terget sector is located at read/write to RAM disk or ROM disk.



F624H -F625H : For future use

- F626H : EPWKTP (5) : R/W : 6.6.4 Protocol Area to store data of FMT, DID, SID, FNC, SIZ at EPSP data is send.
- F62BH : EPACKC (1) : R/W : 6.6.4 Protocol AREA to store ACK or NAK sent to the terminal side (FDD) at EPSP data send/receive.
- F62CH F62EH : For future use.
- F62FH : R0 (18) : : 6.6.4 Protocol
  Work area used in the system at EPSP data send/receive.
- F641H : For future use.
- F642H : BEEPBASE (1) : : (4.2) BIOS (BEEP) Work area used to measure BIOS BEEP time.
- F643H : ICDIDPNT (4) : R/W : 11.3 IC card protocol Expansion Represents the storage area for ID data during registration of the card ID. (Valid only in DISK Mode)
- F647H : ICFILPNT (3) : R/W : 11.3 IC card protocol Expansion Represents the storage area of the File Name data during file creation. All of the above data are valid only in DISK Mode and represent the address of the data set by the ISET function.



- F64AH : ICFILSZ (1) : = : 11.3 IC card Protocol Expansion Specifies data area size of the IC card as a disk. Unit is 128bytes and default value is 38H (7 kbytes). This variable is set by IDSK function.
- F64BH : ICDIRNO (1) : : 11.3 IC card protocol Expansion Specifies the number of directories of the IC card as a disk. Default value is O8H. This variable is set by IDSK function.
- F64CH : ICOPNFLG (1) : : 11.3 IC card Protocol Expansion
  Flag for distinguish whether the class code of the command data is File
  class code or System class code.

F64DH : ICDIRBOT (2) : - : 11.3 IC card protocol Expansion Points the bottom address of the directory area of the IC card as a disk. Default value is 100H. This variable is set by IDSK function. F64FH : ICDATTOP (2) : - : 11.3 IC card protocol Expansion Points the logical top address of the data area of the IC card as a disk. Default value is 400H. F651H : ICBLKSZ (2) : - : 11.3 IC card protocol Expansion Specifies the block size of the IC card. Default value is 100% (256bytes) F653H : ICWAIT4 (1) : - : 11.3 IC card protocol Expansion Specifies the waiting time between to supply the power to the IC card and to supply the clock to the IC card. Unit is mSec and default value is ClH(1mSec). F654H : ICSTATUS (3) F657H : ICOFFMD (1) - For future use. F658H : RZICCTL1 (1) F659H : RZICCTL2 (1) F65AH - F65BH : For future use. F65CH : ICMAXREC (2) : R/O : 11.3 IC card protocol Expansion F65DH : QTICCARD (1) : R/O : 11.3 IC card protocol Expansion Fo5FH : 1rICCARU (1) : R/O : 11.3 JC card protocol Expansion Indicates data for the IC card capacity. ICMAXREC --Indicates maximum number of physical records of IC card. Jefault is 40H (64records) Indicates physical capacity of IC card. Unit is -- **D***R*ADDITO specified by ICBLKS2. Default is 20H (32 x 256bytes). Indicates the number of records used for directory. TPICCARD --Default is O8H (8 records). F660H : ICONFLG (1) : R/O : 11.3 IC card protocol Expansion IC card power supply state is indicated. =00H : No power supply (closed) =OlH : During power supply (open) =FFH : No access for the proper time, power supply is temporarily stopped. F661H : ICCVFLG (1) : R/O : 11.3 IC card protocol Expansion Indicates the rear cover state while the IC card is in use. =00H : IC card rear cover closed =01H : ]C card rear cover open F662H : ICUSEDV (1) : R/O : 11.3 IC card protocol Expansion Indicates the IC card current use status. =FFH (-1) : Command through mode uses IC card. =00H : For future use =01H : Being used by BIOS ICCARD =O2H : Being used as a disk F663H : ICRSTDT (10) : R/O : 11.3 IC card protocol Expansion Area to store reset response data of the IC card, pointed to by ICRSTPNT (F1AEH). APPENUIX Page 21 - 42

F66DH : For future use.

- F66EH : ICWAIT3 (2) : R/W : 11.3 IC card protocol Expansion Specifies the interval (in 10-ms units) for a time out error for the response from the IC Card after command transmission. Default value = 00FFH (approx. 2.5 Sec)
- F670H : ICOPNSTS (1) : R/W : 11.3 IC card protocol Expansion Specifies the processing during File OPEN status (after the IC Card power is ON). (Valid only in DISK Mode)



F671H : ICFMTSTS (1) : R/W : 11.3 IC card protocol Expansion Specifies the processing during formatting. (Valid only in DISK Mode)



F672H : For future use.

F673H - F6B6H : Work area related to display See 4.4.7 Work Area Detail Related to Display for further information.

F6B7H : ERRSEC (2) : R/W : 2.5 Self Test Area to store error sector number when an error occurs at RAM disk checksum. Consecutive numbers are used from track 0 or sector 0 for sector numbers

F6B9H : FNBLOS (1) : - : -Work area used when BIOS is used in the system.

F6BAH : SFUNCCD (1) : - : -Work area used to select functions from the system menu initial screen. F6BBH : SYSMMOD (1) : R/W : 9.2.1 System Menu Control This flag disables the system menu functions.



F6C5H : TCNTDT (2) : - : 11.2 Communication Protocol Expansion Area to check the number of retries when Filink protocol is specified. F6C7H : TERRTIME (2) : - : 11.2 Communication Protocol Expansion Work area to check timeout time by BIOS TCAM. F6C9H : TCAMAREA (12) : R/W : 11.2 Communication Protocol Expansion Work area to pass parameter in BIOS TCAM application. It has the following meaning at Filink protocol. At file send .... Perform connect process after storing filename (11 bytes) in this area. At file receive ... After connect process, the filename received in this area is stored and returned. At no sequence It is not used at TCAM but used for filename memory in the DL/UL process or DLL process. (To equalize the interface to Filink.) At expansion protocol Using identical interface with Filink is preferable. F6D5H : TCAMIF (1) : R/W : 11.2 Communication Protocol Expansion Area to store whether connect is performed in the send/receive mode when Filink protocol is specified in BIOS TCAM. =00H : Send mode =01H : Receive mode F6D6H : DLLSVSP (2) : R/O : 11.2 Communication Protocol Expansion Work area to save the stack pointer when the DLL process starts. F6D8H : LOADAD (2) : R/O : 11.2 Communication Protocol Expansion This work area stores the next data loading address when loading a file to the DLL process TPA area. F6DAH : DLLRVS (1) : - : 11.2 Communication Protocol Expansion This flag indicates message reverse state of "Down Loading" and "Up Loading" in the DLL and DL/UL processes. F6DBH : DLLFILE (1) : R/O : 11.2 Communication Protocol Expansion Work area to store type of received file in the DLL process. =00H : COM file received =01H : BAS file received =FFH : For future receive =FEH : File receive other than COM/BAS (During open) F6DCH : ERRADR (2) : R/W : 11.2 Communication Protocol Expansion Indicates error the process execute address when a error occurs during the DLL or DL/UL process. F6DEH : ULDLSVSP (2) : - : 11.2 Communication Protocol Expansion Work area to save the stack pointer when the UL/DL process starts. F6EOH : ULDLTYPE (1) : R/O : 11.2 Communication Protocol Expansion This flag indicates whether either the UL/DL process is being executed. =00H : Down Loading =01H : Up Loading APPENDIX Page 21 - 45

- F6E1H : RENFLG (1) : : 11.2 Communication Protocol Expansion This flag indicates whether the rename process is executed at the disk write process in the UL/DL process.
- F6E2H : STESTWK (10) : : -Work area used in the system menu TEST process.
- F6ECH : BCDCB (60) : R/W : 11.5 Barcode Decoder Addition Work area used in barcode read routine. See 11.5.3 Barcode Reader Work Area Details for further information.
- F728H : SGRAPHWK (128) : : (4.2) BIOS (GRAPHICS) Work area used by BIOS GRAPHICS and KANJI.
- F7A8H : DBGWORK (18) : : (10.3) System Jump Table
  Work area used by EHT-10/2 debugger (WAD).
- F7BAH F7DFH : Work area used by EHT-10/EHT-10/2 debugger (WAD). However, when the development cartridge is not connected, it can be used by a user anytime.

(5) System Work Area V (RSYSAR5)

This area has a buffer and stack area which do not need initialization, and has a system hook table, system jump table and interrupt vector which are initialized at reset.

Address : Variable (Number of bytes) : Reference

F7EOH : KEY\_BUFFER (66) : 4.3 Key Input Key input buffer

F822H - F82CH : For future use

F82DH : SYSFCB (36) : (5) BDOS Description The FCB data specified by a user is copied to this area while BDOS is in use.

- F851H : SYSDMA (128) : (4.2) BIOS (READ) (WRITE) Data exchange with disk is performed through this area while BDOS and BIOS are in use.
- FBD1H : PKT\_TOP (9) : (6.5.4) Protocol, 11.3 IC card Protocol Expansion This EPSP data send/receive area is used in the system at EPSP data send/receive. This is also used as work area for the IC card command and data send/receive.
- FdDAH : SCRCH BUF (128) : (6) Disk System Description Scratch buffer at disk access.
- F95AH : COMBUF (256) : (7.2) Serial Interface Receive buffer which uses RS-232C with an I/O device (CON:, RDR:, PUN:, LST).
- FA5AH : BCDBUF (208) : (11.5) Bar Code Decoder Addition Buffer to store data read from the barcode reader.
- FB2AH : PRNBUF (128) : 4.5 Printer Output buffer to printer unit.
- FBAAH : SYSWORK (261) : -This is used as a common work area in the system.
- FCBOH : STRTSP (96) : (2.3) System Start and End Stack area for the system address O start (STARTER) process.
- FD10H : SPEXT (32) : (8) Interrupt Process Stack area for EXT interrupt process
- FD3OH : SPICF (48) : (8) Interrupt Process Stack area for ICF interrupt process
- FD6OH : SPOVF (56) : (8) Interrupt Process Stack area for OVF interrupt process
- FD98H : SP8251 (48) : (8) Interrupt Process Stack area for EXT interrupt process

- FDC8H : SP7508 (120) : (8) Interrupt Process Stack area for 7508 interrupt process
- FE40H : BIOSSK (176) : (4.1.2) BIOS Process Stack area for BIOS process
- FEFOH : BDOSSK (64) : (5) BDOS Stack area for the BDOS process
- FF60H : SYSHOKTOP (48) : 10.2.4 system Hook (II) Description System Hook Table (II)
- FF9OH : RSJMPTOP (48) : 10.3.3 Resident Jump Table Description Resident Jump Table
- FFCOH : RSHOKTOP (48) : 10.2.3 System Hook (I) Description
   System Hook Table (I)
- FFFOH : VECTBLTOP (16) : 8.2 Interrupt Vector Interrupt Vector Table

## (6) Label Table

Address	Label name	Address	Label name	Address	Label name
F07A	ADDLB1	FICD	ADDLB2	F6B9	ADDLB4
F42C	ADDLNK	F079	ADDLT1	FICA	ADDLT2
F6B7	ADDLT4	FOE 1	ADSKOPN	F065	AD RAM IN
F072	AD_RAM_OLD	F063	AD ROM CP1	F3CA	AFTER3
FOEO	AHSTWRT	F3DF	ALLOCA	FFC0	ALMHKO
FFC3	ALMHK1	FFC6	ALMHK2	FFC9	ALMHK3
FFCC	ALMHK4	F029	ALRMAD	F3C5	ALRMCT
F027	ALRMDS	F02C	ALRMFG	F32F	ALRAMSG
FOIE	ALRMSOND	F02B	ALRMST	FICA	ALRMTIME
F028	ALRMTP	F557	ALVO	F574	ALV1
F584	ALV2	F59C	ALV3	F5BE	ALV4
F407	AREC 1	F405	ARECORD	F3D1	ARET
Γ020	ATSHUTOFF	F021	ATSOTIME	F723	BARBIT
BC3A	BARCDBOT	1020	BARCDSIZE	AC1A	BARCDTOP
F6F8	BARFLG	F71C	BARTHR	FFEA	BASHOOK
F07F	BASICB	F07C	BASICF	F07D	BASICR
FA5A	BCDBUF	F6EC	BCDCB	F70C	BCGETP
F3BC	BCINTST	<b>F</b> 710	BCLOC	F711	BCLOF
F6FA	BCOPTN	F70E	BCPUTP	F714	BCTLOC
<i>F</i> <b>1</b> 5	HCTLOF	F712	BCIPTP	F6F9	BCTYPE
1001	BLOLB1	FOD6	BDCLB2	F219	BDOLB3
F40F	BDOLB4	2034	BDOLT 1	F0D6	BDOLT2
T209	BDOLT3	F3CD	BDOLT4	25A1	BDOSBOT
F412	BDOSFN	OEFE	BDOSSIZE	FF5F	BDOSSK
(.:)70	BDOSSKSZ	16A3	BDOSTOP	F414	BDSBNK
F005	BDSLAD	F642	BEEPBASE	F4D7	BEGDAT
F007	BIILAD	F63B	BIAS	F074	BIOLB1
F1B8	BIOLB2	F242	BIOLB3	F673	BIOLB4
FJ5C	BIOLT1	F0D6	BIOLT2	F219	BIOLT3
F42C	BIOLT4	38DB	BIOS 1BOT	12E0	BIOSISIZE
25 <b>F</b> R	BIOS1TOP	46CE	BIOS2BOT	0 <b>DF3</b>	BIOS2SIZE
38DB	BIOS2TOP	552C	BIOS3BOT	0E5E	BIOS3SIZE
46CE	BIOSSTOP	DC9D	BIOS4BOT	0F60	BIOS4SIZE
CD3D	BIOS4TOP	F417	BIOSERROR	F426	BIOSFN
FFE7	BIOSHK	0038	BIOSNUM	FEEF	BIOSSK
0630	BIOSSKSZ	F639	BITMAP	F722	BITPIN
F3E4	BLIMSK	F3E3	BLKSHF	F076	BLNKCNT
F077	BLNKRVRS	F075	BLNKSTAT	F078	BLNKT IME
F034	BNKDTBL	F23B	BPINTEBL	0100	BRBDOSD
F079	BTRYALM	F6B5	BTRYDSP	FOB3	BTRYFG
FCAF	BUFBOT	F3D9	BUFFA	0100	BUFSIZE
F097	BUZZHZ	F3B9	BUZZLNG	F094	BUZ_FLG
F41C	CALBNK	AC1A	CALCBOT	0C46	CALCSIZE
9FD4	CALCTOP	DFF5	CCPBOT	F003	CCPLAD
0358	CCPSIZE	DC9D	CCPIOP	F3D3	CDRMAXA
F3DD	CHECKA	F3EC	CHKSIZ	F30B	CHIKSUM
F20A	CHKSUMFG	F618	CHRMSK	F067	CKSMSIZE
F200	CNTNFG	F305	CNINILVL	F01C	CNTNSOND
F303	CNTNSP	F20B	COLUMN	F95A	COMBUF
F034	COMLB1	FOD6	COMLB2	F209	COMLB3
F3CD	COMLB4	<b>F00</b> 0	COMLT1	F090	COMLT2
F200	COMLT3	F300	COMLT4	F209	COMPCOL

Address	Label name	Address	Label name	Address	Label name
8DOF	CONFIGBOT	22BF	CONFIGSIZE	6450	CONFIGTOP
F6BC	CONFNOD	F0C8	CONSCRN1	FODO	CONSCRN2
F090	CPMSIZ	F42F	CRGDEV	FF6C	CRGHOOK
F220	CSTOPFLG	F1CB	CSTOPPRT	F574	CSV0
F584	CSV1	F58C	CSV2	F5AE	CSV3
F5D0	CSV4	F067	CS RAM IN	F423	CURBNN
F20F	CURDSK	F3D7	CURRECA	F3D5	CURTRKA
F633	DA	010 <b>B</b>	DATSIZ	F208	DBGFLG
F7A8	DBGWORh	F421	DBNKDT	FJOA	DCNT
F3E4	DIRBLK	F4D7	DIRBUF	F724	DIRECF
F3F4	DIRLOC	F3E8	DIRMAX	F062	DIRSIZE
F41B	DISBNK	F23F	DISBRK	F0E9	DISKROV
FOE4	DISKTBL	96D4	DISKUTYBOT	054B	DISKUTYSIZE
9189	DISKUTYTOP	F5E2	DISK_BNK	FF66	DLHK1
67D6	DLLBOT	F851	DLLBUF	F82D	DLLFCB
F6DB	DLLFILE	FF63	DLLHK1	F6DA	DLLRVS
0676	DLLSIZE	F <b>6</b> D6	DLLSVSP	6160	DLLTOP
F1DA	DLLTYPE	F213	DLOG	F215	DMAAD
F4D5	- DMAADR	F3F7	DMINX	F159	DPBO
F168	DPB1	F177	DPB2	F186	DPB3
F186	DPB4	F3DB	DPBADDR	F109	DPBASE
F109	DPEO	F119	DPE 1	F129	DPE2
F139	DPE3	F149	DPE4	F409	DETR
F1DD	DRCVBUF	FIDB	DRCVCNT	F40C	DREC
F062	DR_RAM_IN	F061	DR_ROM_CP1	F0E2	DSKDID
F1DF	DSKNUM	FOE3	DSKSID	F07A	DSKTFLG
F06D	DSK_R_W	F074	DSPFLAG	<b>F2E</b> 0	DSPTPSV
F2DF	DSPTYPE	FOOF	DUPLEX	0000	DXLTO
0000	DXLT1	0000	DXLT2	0000	DXLT3
0000	DXLT4	F210	EFCB	F204	ELCTLFLG
F3B7	ELOFFEND	F023	ELOFTIME	F5E2	ENDDAT
F3C0	ENTSINT	F3CD	ENTSP	F1AB	EPIBRTO
F62B	EPACKC	F1A8	ÉPATMO	F62C	EPBLCN
F627	EPDDEV	F62D	EPERMD	FIAA	EPETTL
F1A7	EPETMO	F626	EPFMT	F629	EPFNC
F1A9	EPMODE	F628	EPSDEV	F624	EPSIZ
F1A6	EPTIMO	F1A5	EPTRCN	F626	EPWKTP
F4D1	ERFLAG	F6DC	ERRADR	F416	ERRFLG
F6B7	ERRSEC	F019	EXECTYPE	F30D	ENESTRIC
FFDB	EXTHOOK	F3E5	EXTMSK	F402	ENTVAL
F3F2	FCBSCOPIED	F400	FCBDSK	F3C4	FG7508
FBAA	FILENAME	0088	FLAISIZE	015A	FLACCIZE
00ED	FLA3SIZE	04BA	FLA4SIZE	F637	FLGCNT
F6B9	FNBIOS	F05E	FORMATRAM	F702	FRCBUF
F201	FRCECNTN	F704	FRCSIZ	FOEB	FTSTAB
F45.1	FUNC_TBL	F22F	GET_KEYPIR	CD3D	CRAPHBOT
1103	GRAPHSIZE	BC3A	GRAPHTOP	F5E6	HCDATA
F5E5	HCN	F5E3	HCX	F5E4	HCY
FFD2	HK8251	0010	HOOKNUM	F1A3	HSARTCR
F1A1	HSARTMR	<b>F1A</b> 0	HSCTLR1	F1A4	HSSOUT
F1A2	HSSWR	F4CA	HSTACT	F4C5	HSTDSK
F4C8	HSTSEC	F4C6	HSTTRK	F4CB	HSTWRT
F651	ICBLKSZ	F1B3	ICCDPRM	F1B1	ICCDTIME
F1E4	ICCLASS	F202	ICCPWCNT	F203	ICCPWFLG
F026	ICCPWTM	F661	ICCVFLG	F64F	ICDATTOP
FF6F	ICDHK1	FF72	ICDHK2	FF75	ICDHK3
		APPENDIX	Page 21 - 50		

Address	Label na	me Address	Label name	Address	Label name
FF78	ICDHK4	FF7B	ICDHK5	FF7E	ICDHK6
FF81	ICDHK7	FF84	ICDHK8	F643	ICDIDENT
F64D	ICDIRBO	T F64B	ICDIRNO	F8D1	ICDWORK 1
F8D5	ICDWORK	2 F6EC	ICFDSP	Ē Ē Ē Ē	ICEHOOK
F647	ICFILPN	T F64A	ICFILSZ	<b>F1F3</b>	ICFICIS
F671	ICEMIST	S F1E6	ICHOKDT1	F65C	ICMASTREC
F657	ICOFEMD		ICONFI G	F61C	ICOPATIC
F670	ICOPNET	S F1F5	ICRECS7	FAAD	TOPETOV
F665	ICRSTAI	F666	ICRSTRI	F667	ICDSTCI
FIAD	ICRSTON	T F668	ICRSTD1	F663	ICOSTOT
FIAE	ICRSTPN	T F664	ICRSTIN	F663	ICOSTO
F1F2		F651	ICSTATUS	F672	ICASIIS
FIFG	ICTSDT	F662	ICLISEDV	FU72	ICISET
FIES	ICHAIT?	F66F	ICUSEDV	F1E7	TOWALLI
F65)	ICVORT	E100		P033	IEOSDBOT
0100	IEIOBI	ZE 0000	IEOSPIOD	6160	IFUSRBUI
0160	TECNODE		TECVODIOP		TISISKBOL
E23C	TLAIONOV	TTE 0000	INDETOP	FJCF	INPU
F2C2	INNEOPY	F JLU	THE FLOR	LUAI	TNIR102
r 302	INTEG	FU32	INTLEVEL		INTRABOL
	INTROMS.		INTROCTOP	FU33	INTS/SU8
FUSE	INIS625	I FUAL	LNISEAL	FUSD	INISICF
FUEF	INISOVE	6 EC33	INTITE	F038	INTVLFG
FUSI	1515:50	B FFAB	JUALLUX	FFAE	JINIOPR
FF-48	JJUMP.CX	FFSF	JLDAXX	FFAD	JEDIRAX
FI-B/	JNOP14	FFBA	JNOPIS	FFBD	JNOP16
FF93	JPREBIC	5 FF96	JPSTHIOS	FF90	JRBDOS
FFB4	JRDBGIO	X FF99	JSCALLX	FFB1	JSDBGIOX
FF9C	JSELBNK	FFA2	JSTAXX	F222	KALMMOD
F222	KANAFLG	F820	KBUFEND	F221	KB_FLG
F095	KDFLTBZ	F21B	KEYD	F21A	KEYF
F21C	KEYS	F7E0	KEY_BUFFER	F227	MODE
F20D	<b>KPCHAR</b>	E91A	KSYSTBL	F233	KTABPTR
E8D6	KUSER1TI	BL E8F8	KUSER2TBL	F7BA	LAB4END
F7BA	LABEND	F1C7	LALPHAYX	F25D	LATRCHK
OODO	LBCDBUF:	S F7E0	LEUFTOP	0100	LCBUFS
F673	LCHRFON	F F269	LCRWAIT	F267	LCURATR
F25F	LCURSOR	F242	LIDSPMOD	F6FE	LEDCNT
F6FD	LEDT IM	F26F	LESCCNT	F26E	LESCFLG
F270	LESCPRM	F24B	LESCTB1	F24D	LESCTB2
F26C	LFKADDR	F26B	LFKSTAT	F724	LFLAG
F3F6	LINFO	F20C	LISTCP	0002	LKBFES
0040	LKBUFS	F1C8	LKCODEYX	F244	LLCDMOD
F255	LLNKFLG	F26A	LLNKWAIT	F6D8	LOADAD
F06A	LOG_ADRS	5 <b>F268</b>	LOLDATR	0009	LPACHS
0800	LPRNBUFS	5 F24F	LSCADDR	F260	LSCCPOSX
F261	LSCCPOSY	F2DD	LSCMDSV	F2DC	LSCMODE
0800	LSCRBS	F25E	LSCROLMD	F249	LSCRTB1
F253	LSCRVRAM	Y F251	LSCSIZE	F257	LSCSIZEX
F258	LSCSIZE	Y 0080	LSDMAS	0024	LSFCBS
E11A	LSORBNK	F5F1	LSTERR	0105	LSYSWKS
000R	LTBUFS	F1C4	LTOUCHAD	F245	LUWDMOD
F25B	LUNDPOSY	Y F25C	LUWDSZY	F683	LVRAMADR
F685		F266	LVRAMYOF	F2DF	LVSVELG
7263	LIVICEDI	r F265	LWDCPOSY	F259	LWDPOSY
F25A		v E262	LADYMIN	F263	LWDYMIN
		F 2 11 2			
E21E	IWOPLAE	ר ד202 ר ד27ד	LWORKBEI	F2AD	LWORKBE2

Address	Label name	Address	Label name	Address	Label name
F6A2	LW_BPOS	F6A0	LW_DADDR	F6A4	LW_LAL
F69E	LW_SADDR	F3E6	MAXALL	FC18	MANFILE
F38D	MCURFILE	F38B	MCURLINE	F38C	NCURPAGE
F0B9	MDRV	F397	MDSPBUF	F391	MDSPTOP
F2E3	NEMD	F2E1	MEMK	F2E2	NEMO
15C7	MENUBOT	F38A	MENLMOD	06. <del>\</del> F	MENUSIZE
0F18	MENUTOP	F393	MFATTR	F395	MFILEBUF
F38E	MFILENUM	F390	MFSTSCRH	FOBE	MFTYP
F38F	MLINEMAX	FOOA	MODEFG	F726	MCDULS
F38F	MOLDDRV	F39F	MTIMEBUF	F71E	NRWBAR
F718	NRWSPC	F3EE	OFFSET	F418	OLDBNK
F3FF	OLDDSK	F641	CLUDVMD	F088	OS2LB1
FIEA	OS2LB2	F2ED	OS2LB3	F7BA	OS2LB4
F07A	OS2LT1	FICD	OS2LT2	F2E1	OS2LT3
F6 <b>B</b> 9	OS2LT4	8000	OSROMBOT	FFD8	OVFHOOK
F706	OVTIME	FBAA	PBUFEND	F4BF	PGETPTR
F23D	PKANJFLG	F5F0	PKFOFF	F8D2	PKT_DID
F8D6	PKT_DRV	F8D1	PKT_FMT	F8D4	PKTFNC
F8D6	PKT_RDT	F956	PKTRSTS	F8D8	PKT SEC
F8D3	PKT_SID	F8D5	PKT_SIZ	F8D6	PKTSTS
F8D1	FKT TOP	F8D7	PKT_TRK	F8DA	PKT WDT
F8D9	PKT_WTP	FSEF	PHALEN	FSEE	PENOFE
F700	PLSCNT	F6FF	PLSFTa	E4BD	PPUTFTR
25B5	PREBIOSBOT	0614	PREBI DSSIZE	25A1	PRFBIOSTOP
F92A	TRNBUF	T3BD	PRNPWCNT	<b>F3BE</b>	PRNPWFLG
F025	FRNPWTM	F23E	PRTFLG	F1CC	PRTINIT
25FB	PSTBIOSBOT	0046	PSTBIOSSIZE	25 <b>B</b> 5	PSTEIOSTOP
F235	PTR_KEYTBL	<b>F22D</b>	PUT_KEYPTR	F01A	PWONSOND
F701	PWRCNT	F3C9	PWSWOFFG	F3C8	PWSWONFG
F65E	QTICCARD	F060	QT RAM IN	F071	OT RAM OLD
F05F	QT ROM CP1	F62F	RO	F630	ROH
F62F	ROL	F631	R1	<b>5E00</b>	RIBDOSTOP
5F00	RIBIOSTOP	5600	<b>R1CCPDTOP</b>	016A	RIENCHRSIZE
E896	RIEXCHRTOP	F632	RIH	008C	RIKTBLSIZE
E80A	RIKTBLTOP	F631	R1L	002A	RILNKSIZE
E7EO	RILNKTOP	6000	RIRAMDSKINP	0400	R1SCR1SIZE
DC00	RISCRITOP	0150	R1SCR2SIZE	E690	R1SCR2TOP
DC00	RIUSERTOP	0690	RIVRAMSIZE	E000	RIVRAMIOP
F633	R2	6000	R2BDOSTOP	6100	R2BIOSTOP
5800	R2CCPDTOP	0004	R2EXCHRSIZE	E93C	R2E\CHRTOP
F634	R2H	0066	R2hTBLSIZE	E8D6	R2KTBLTOP
F633	R2L	0036	R2LNESIZE	E870	R2LNKTOP
6200	R2RANDShTOP	0200	R2SCR1SIZE	DEOO	R2SCR1TOP
0200	R2SCR2SIZE	E000	R2SCR2TOP	00.40	R2SCR3SIZE
E800	R2SCR3TOP	DEOO	<b>R2USERTOP</b>	0200	R2VRAMISIZE
<b>E</b> 2U0	R2VRAMITOP	0200	R2VRAM2SIZE	E-100	R2VRAM2TOP
0200	R2VRAM3SIZE	E600	R2VRAM3TOP	F635	R3
F636	R3H	F635	R3L	F637	R4
F638	R-IH	F637	R4L	F639	<b>R</b> 5
F63A	RSH	F639	R5L	F63B	R6
F63C	R6H	F63B	R6L	F63D	R7
F63E	R7H	F63D	R7L	F63F	<b>R</b> 8
F640	R8H	F63F	RØL	F5E2	RAMD_BNK
F068	RAMD_SIZE	F5F5	RAMSIZE	F07B	RAMIFLG
F069	RAM_SET	78A2	REDOSDBOT	009F	REDOSDSIZE
7803	RBDOSDTOP	EA00	REDOSLAD	0100	RBDOSLSIZE

Address	Label name	Address	Label name	Address	Label name
0100	RBIOS1SIZE	F001	RBITOP	F401	RCOUNT
F4D3	READOP	FOF5	READTAB	F62E	REGA
1673	RELOCBOT	00DC	RELOCSIZE	15C7	RELOCTOP
F6E1	RENFLG	F3FE	RESEL	7FE8	RESERVEBOT
0033	RESERVESZ	7FB5	RESERVETOP	F00 <b>9</b>	RESENO
F000	RETADD	F41E	RETBNK	F725	RFI AG
F415	RIOBYTE	F1BE	RLCGENG	F1C1	RICGENK
F1BB	RLCGENN	F246	RLCGENU	F188	RICGENS
F3F3	RMF	F22B	RM BOINCD	F454	RM BUF
F237	RM BLFFTR	F22A	RM FSIIN	F225	RM HATUFLG
F22C	RY SIINCD	F224	RM SIINFLG	F226	RAI SOUTLG
FIEO	RNDFIG	F211	RODSK	F424	ROMEVADD
F429	ROMEXBNK	F207	ROMEXQ	F428	ROMEXOON
3000	ROMIDBOT	0018	ROMIDSIZE	7FE8	ROMIDTOP
EFBB	RRSYSPRBOT	EB00	RRSYSPRTOP	F1 <b>9E</b>	RS2ARTCR
F19C	RS2ARTMR	F19B	RS2CTLR1	F19F	RS2SOUT
F19D	RS2SWR	F625	RSBYTE	F195	RSB\TED
F624	RSCALSW	F240	RSCLSF	F622	RSFDEV
F4D2	RSFLAG	FFCO	RSHOKTOP	F3BB	RSINTST
<b>FF</b> 90	- RSJMPTOP	F623	RSODEV	F61A	RSPALAD
F60E	RSPBITL	F60D	RSPBITR	F60F	RSPPAR
F609	RSPRBAD	<b>F60</b> 5	RSPRBGP	F607	RSPRBPP
F60B	RSPRBSZ	F611	RSPSPP	<b>F</b> 61 <b>0</b>	<b>RSPSTOPB</b>
F604	RSFSTS	F612	RSRBEAD	F61C	RSRDL
F619	RSSSPP	F621	RSXMODE	FOOE	RSXOFF
F616	RSXOFS2	FOOD	RSNON	F614	RSXONSZ
F090	RSYSAR1BOT	0090	<b>RSYSAR1SIZE</b>	F000	RSYSARITOP
F200	RSYSAR2BOT	0170	RSYSAR2SIZE	F090	RSYSAR2TOP
F300	RSYSAR3BOT	0100	<b>RSYSAR3SIZE</b>	F200	RSYSAR3TOP
F7D0	RSYSAR4BOT	04D0	RSYSARISIZE	F300	RSYSAR4TOP
FF5F	RSYSAR5BOT	0780	RSYSAR5SIZE	F7E0	RSYSAR5TOP
7CB5	RSYSPRBOT	0413	RSYSPRSIZE	78A2	RSYSPRTOP
F1E1	RINTYP	F05C	RXXLB1	F0D6	RVXLB2
F219	RXXLB3	F42C	RXXLB4	F034	RXXLT1
FOD6	RXXLT2	F219	RXXLT3	F40F	RVOLT4
FODA	RZARTCR	F0D9	RZARTMR	F42C	RZBANKR
FOD7	RZCMDR	F0D6	RZCTLR1	FOD8	RZCTLR2
FODE	RZCTLR3	F219	RZCTRL2	F658	RZICCTLI
F659	RZICCTL2	FODB	RZICCTLR	F42E	RZIER
FODD	RZIOCTLR	F42D	RZSBBNKR	FODC	RZSWR
F070	R W RETRY	F635	SA	F430	SAVEIX
F432	SAVEIY	F41F	SBNADT	F6FC	SCNERR
F084	SCONFMOD	F8DA	SCRCH_BUT	7803	SCREENBOT
22D7	SCREENSIZE	552C	SCREENTOP	F079	SCRLB1
FICA	SCRLB2	F2E1	SCRLB3	F6B7	SCRLB4
F69D	SCRLFG	F074	SCRLTI	F1B8	SCRLT2
F242	SCRLT3	F673	SCRLT4	F3 <b>F9</b>	SEASRCHA
F3 <b>F8</b>	SEARSCHL	F3E1	SECTPT	F4C1	SENDSK
F4C9	SEKHST	F4C4	SEKSEC	F4C2	SELTRK
FC19	SELPOS	F3F5	SEQIO	F6BA	SFUNCCD
F728	SCRAPHWK	0010	SHOKNUM	F3FD	SINGLE
<b>F61</b> E	SISOCNT	F631	SIZE	001F	SIZERAM
0000	SIZELSER	FOOB	SIZRAM	F21E	SKEYFLG
F0B8	SMENUFLG	F223	SMAFLG	FE3F	SP7508
0078	SP7508SZ	FDC7	SP8251	0030	SP8251SZ
F722	SPCBIT	<b>F7</b> 16	SPCTHR	FD2F	SPENT

Address	Label name	Address	Label name	Address	Label name
0020	SPEXTSZ	FD5F	SPICF	0030	SPICFSZ
FD97	SPOVF	0038	SPOVFSZ	F5FB	SRBADR
F5FF	SRBR	F5FD	SRBS12	F41D	SRCBNK
F600	SRDCHR	F241	SRMODE	F601	SRPRT
F010	SRSADR	F602	SRSB	F014	SRSPAK
F603	SRSPARA	F389	SRSTNODE	F196	SRTABI
F369	SSEDITWK	F620	SSAMODE	F082	SSYSIMOD
0768	STARTBOT	0668	STARTS17F	0100	STAPTTOD
F6F2	STESTWK	F351	STIMERIE	EC VE	
F20A	STRICOL	FDOF	STRTSP	0060	STDTSDS7
F3C3	STS7 508	F726	SUNCHK	F6.17	SI'CDSD
F6C2	SVERRETG	FABA	SVEMODE	E110	SUCI DRVIC
FARC	SVDEMOD	FARE	SVSFLDSK	FECI	SVSELSEC
F6BE	SVSFLTRE	7045	SVSADIBOT	0000	SVS (DIST7T
7095	SVSARIMOR	7585	SVS D2DOT	0170	SIGNOSCI
7045	SYSAD2TOP	7595	SVSAD2DOT	0100	SISAR2SIZE
7505	SVS 102000	F100	SISARSOUT SUS VETTE	5107	SISARSSIZE
F106	SUSCEPTIDI	F 1 5 5	SVEDMA	F 197	SIGARIAR
F190	SISUILKI	F001	SYSDMA	F82D	SISECB
FF00	* SISHUNIUP	F081	SISINFLG	FFSF	212120
F000	SISLII	FUSU	SISLIZ	F200	SISLI3
F 300	SYSLIA	6A50	SYSMENUBCI	027A	SYSMENUSIZE
PIDP	SYSMENUTOP	-PBB	STEPPOD	EUUU	SISROMBOT
	SYSKSVBOT	000B	SYSRSVEIZE	DFFD	SYSRSTOP
194	SYSSOUT	F198	SYSSWR	9FD4	SYCTESTBOT
1900	SYSTESISIZE	9604	SYSTESTIOP	FBAA	SYSWORK
TCOU	SZRAM	0000	SZUSER	F06E	S_CHK_SUM
E1D9	TISTFLG	F1D5	TISTTIME	F1D7	T2NDTIME
rud3	TBL7508	F851	TBUF	F3BF	TBUZ_FLG
F6C9	TCAMAREA	F6D5	TCAMIF	FICD	TCAMPRM
F6C5	TCNTDT	F1D4	TDFLTCNT	F6C7	TERRTIME
<u>FF60</u>	TEXHK 1	F6B3	THATRAD	F6AF	THCNTXY
F6AD	THPOSXY	F6B1	THSVSP	F1C9	THSYSFLG
F822	TIMBUF	F708	TIMENT	F5F9	TIMEEND
FOZD	TIMERO	F02F	TIMER 1	F032	TIMERIM
F369	TIMEWK	F70A	TIMLOC	F3FB	TINFO
FFDE	TMDT83	FFE1	TMDT85	FFE4	TMDT86
F206	TIFLAC	F205	TMFUNC	FFCF	TNHOOK
F3C6	TMSEC	FOSC	TOPRAM	F65F	TPICCARD
F3F0	TRANV	F6FB	TRMCHR	F6C3	TSPSAVE
E850	TSYSTEL	E80A	TUSERTBL	F086	UCONFMOD
9189	ULDLBOT	047A	ULDLSIZE	F6DE	<b>ULDLS\SP</b>
8D0F	<b>ULDLTOP</b>	F6E0	ULDLTYPE	<b>FF69</b>	<b>UTLHIKI</b>
F4CC	UNACNT	E4CD	UNADSK	F4D0	UNASEC
T4CE	UNATRK	F716	UPCBUF	+ F00C	USERBIOS
FUDF	USERCRG	F239	USERKIBL	F20E	USRCCDE
F217	USRDMA	F411	USRFCB	F40F	USRSBD
F424	USRSBI	F083	USYSMMOD	FFFO	<b>VECTBLTOP</b>
F403	VRECORD	F3CB	WFUNCFLG	F720	WIDBAR
F71A	WIDSPC	FOFF	WRITAB	F4D4	WRTYPE
F243	WTONLY	F300	XUSRBIOE	F0B6	YALMDS
F087	VALMST	F5F2	YLCOUNTRY	F5F4	YLCOLNTRY
E5E3	YLDFLTC	F308	YMAINST	F40E	YOLDDSK
F21D	YPECMET	F134	YPEKBLE	F229	YPEKCNT
F231	YPERPIR	FOR4	YPOFDS	F0B5	YPOFST
F307	YPWSWST	F5F7	YSIZERAM	F309	ZSTARTFG

## 5. BIOS FUNCTION LIST

- The meanings of symbols used in the columns of EHT-10 and EHT-10/2 are as follows:

- 0 : This function can be used in the same way as in HX-40 and PX-4.
- o: This function can be used in the same way as in HX-40 and PX-4, but the parameter range is different from that in HX-40 and PX-4.
- @: This function can be used in the same way as in HX-40 and PX-4, but the parameter is different from that in HX-40 and PX-4.
- x : This function is not supported in EHT-10/EHT-10/2.
- \$ : This function has been newly added to EHT-10/EHT-10/2.

Entry address		Name Function	Nodification for	EHT		
Offset from WBOOT	Absolute address	Name	Function	HX-40/PX-4	10	10 / 2
-03H	EBOOH	B001	Cold-bonts CP/M.		0	0
+00H	03H	WBOGT	Warm-boots CP/M.		0	0
+03H	06H	CONST	Checks the CON: input status.		0	0
+06H	09H	CONIN	Inputs one character from the CON: device.	Position code information is returned.	0	0
+09H	OCH	CONOUT	Outputs one character to the CON: device.	Function added and ESC sequence parameter partially modified	0	Q
+OCH	OFH	LIST	Outputs one character to the LST: device.		0	0
+OFH	12H	PUNCH	Outputs one character to the PUNCH: device.		0	0
+12H	15H	READER	Inputs one character from the READER: device		0	0
+15H	18H	HOME	Sets the disk seek track to 0.		0	0
+18H	184	SELDSK	Specifies a drive.	Only drives A to E are supported.	0	٥

Entry address						EHT	
Offset from WBOOT	Absolute address	Name	Function	Modification for HX-40/PX-4		10 / 2	
+18H	EB1EH	SETTRK	Specifies a track for read/write operations.	Depends on the drive.	0	٥	
+1EH	21H	SETSEC	Specifies a sector for read/write operations.	Depends on the drive.	0	٥	
+21H	24H	SETDMA	Specifies the DMA address for read/write operations.		0	0	
+24H	27H	READ	Reads 128-byte data.		0	0	
+27H	2AH	WRITE	Writes 128-byte data.		0	0	
+2AH	2DH	LISTST	Checks the LST: device status.		0	0	
+2DH	30H	SECTRN	Translates a logical sector to a physical sector.		0	0	
+30H	33H	PSET	Performs a logical operation for VRAM data.	Covered by GRAPHICS (WBOOT+90H)	X	X	
+33H	36H	SCRNDUMP	Dumps the VRAM contents to the LST: device.	No operation is performed in EHT-10.	X	0	
+36H	39H	BEEP	Sounds the buzzer.	The hardware beep function has been added.	9	0	
+39H +3CH +3FH +42H +45H +48H	3CH 3FH 42H 45H 48H 4BH		No function				
+4BH	4EH	TIMDAT	Sets and reads time, enables and disables the alarm/wake function and sets and reads the alarm/wake time.		0	0	

					-	_
Entry	address	Namo	Evection	Modification for	EI	ΗT
Offset from WBOOT	Absolute address	Name		HX-40/PX-4	10	10 / 2
+4EH	E851H	MEMORY	Reads the current bank information.	The subbank value is returned as the return informa- tion.	0	9
+51H	54H	RSIOX	Performs serial communications.	The DSR/DTR line control functions have been added.	0	0
+54H	57H		No function			
+57H	5AH	MASKI	Sets and resets the interrupt mask, allows and probibits interrupts by the 7508 CPU, and checks the current mask status.	The external interrupt control function has been added.	0	0
+5AH	5DH	LOADX	Reads one-byte data from the specified address in the specified bank.	The bank specification values have been modified because	0	0
+5DH	60H	STORX	Writes one-byte data at the specified address in the specified bank.	have been added.	0	0
+60H	63H	LDIRX	Transfers the specified length of bank data to the specified another bank.		0	0
+63H	66H	JUMPX	Jumps to the specified bank address.		0	0
+66H	69H	CALLX	Calls the specified bank address through a subroutine.		9	9
+69H	6CH	GETPFK	Reads the key code currently set.	The structure differs from	0	0
+6CH	6FH	PUTPFK	Registers a key code in the user table.	HX-40.	9	0
+6FH	72H	READSW	Reads the states of switches.		0	0

APPENDIX Page 21 - 57

Entry address		Namo	Evention	Medification for	El	T
Offset from WBOOT	Absolute address	name	Function	HX-40/PX-4	10	10 / 2
+72H	EB75H		No function			
+75H	78H	RDVRAM	Reads one character from the screen.	Only one-byte screen buffer data is read.	9	0
+78H	78H	MCMTX	Processes a microcassette.	No MCT is mounted	X	X
+7BH	7 E H	POWEROFF	Turns the system power off.		0	0
+7EH	81H	USERBIOS	Registers a user- created BIOS function.		0	0
+81H	<b>84</b> H	AUTOST	Specifies an automatic start string.	The automatic start string function is not supported.	X	X
+84H	87H	RESIDENT	Sets and resets the RESIDENT function.	The RESIDENT function is not supported.	X	X
+87H	8AH	CONTINUE	Sets and resets the CONTINUE flag.	Not discrimina- ted by the keyboard type (standard or item).	0	0
+8AH	8DH	BARCODE	Supports the barcode reader.	Newly aded functions.	S	\$
+8DH	90H	TCAM	Sends and receives data through a public line.		S	\$
+90H	93H	GRAPHICS	Supports graphic functions.		S	S
+93H	96H	TOUCH	Sets the indication for a touch-panel key block		\$	
+96H	9 <b>9</b> H	ICCARD	Exchanges data with an IC card.		\$	S
+99H	9CH	KEYIN	Initiates the system input functions and exchanges data.		S	S

Entry address			Fuending		EHT	
Offset from WBOOT	Absolute address	Name		Modification for- HX-40/PX-4		10 / 2
+9CH	9FH	KANJI	Displays and prints kanji characters.	Newly added functions.	S	s
+9FH	A2H	BACK LIGHT	Controls the backlight software.			\$
+A2H	A5H	INFORM	Checks the addresses of the work areas and jump tables used by the system.		\$	S

## 5. DISPLAY CONTROL FUNCTIONS

(1) List of display control functions

- The meanings of the symbols used in the columns of EHT-10 (window and fixed areas) and EHT-10/2 are as follows:

- O: This function can be used in the same way as in HX-40 and PX-4, but the parameter range is different from that in HX-40 and PX-4.
- 0: This function can be used in the same way as in HX-40 and PX-4, but the parameter is different from that in HX-40 and PX-4.
- X: This function is not supported in EHT-10/EHT-10/2.
- \$: This function has been newly added to EHT-10/EHT-10/2.

Code	Function	Remarks	EHT-10		EHT-10/2
			Window area	Fixed area	
02H	SCREEN LEFT	Function deleted	X	x	Ä
05H	ERASE END OF LINE		0	0	0
06H	SCREEN RIGHT	Function deleted	X	X	X
07H	BELL		0	0	0
08H	BACK SPACE		0	0	0
09H	ТАВ		0	0	0
OAH	LINE FEED		0	0	0
ОВН	НОМЕ		0	0	0
ОСН	CLEAR SCREEN & HOME		0	0	0
ОДН	CARRIAGE RETURN		0	0	0
10н	SCREEN UP		0	X	0
11H	SCREEN DOWN	t sin t	0	X	0
1AH	ERASE END OF SCREEN		0	0	0
1BH	ESCAPE	ESC sequence entry	0	0	0
1CH	CURSOR RIGHT		0	0	0
10H	CURSOR LEFT		0	0	0
1EH	CURSOR UP		0	0	0

Code	Function	Remarks	EHT-10		EHT-10/2	
			Window area	Fixed area		
1FH	CURSOR DOWN		0	0	0	
ESC '%'	ACCESS CGROM DIRECTLY	Attribute added	Ø	0	Q	
ESC '('	BLOCK REVERSE		0	0	0	
ESC '*'	CLEAR SCREEN & HOME	Same as code OCH	٥	0	0	
ESC 'O'	REVERSE ON		0	0	0	
ESC '1'	REVERSE OFF		0	0	0	
ESC '2'	CURSOR OFF		0	0	0	
ESC '3'	CURSOR ON		0	0	0	
cSC '≖'	SET CURSOR POSITION		0	٥	0	
ESC (C)	SET CHARACTER SET TARLE		0	IJ	Ü	
ESC 'P'	SCREEN DUMP	Supported only in EHT-10/2	X	X	0	
ESC 'T'	ERASE END OF LINE		0	0	0	
ESC 'Y'	ERASE END OF SCREEN		0	0	0	
ESC 78H	SECRET		0	0	0	
ESC 7DH	NON SECRET		0	0	0	
ESC 90H	PARTIAL SCROLL UP		0	0	0	
ESC 91H	PARTIAL SCROLL DOWN		0	0	0	
ESC 92H	SCROLL RIGHT N CHARACTERS	Function deleted	X	X	X	
ESC 93H	SCROLL LEFT N CHARACTER	Function deleted	x	x	x	
ESC 94H	SET SCROLL STEP	Function deleted	X	x	x	
ESC 95H	SET SCROLL MODE		0	X	0	
55C 96H	SCROLL UP 1 LINE		0	x	0	
ESC 97H	SCROLL DOWN 1 LINE		0	X	0	
ESC 98H	SET SCROLL MERGIN	Function deleted	X	X	X	

Code		Function	Remarks	EHT-10		EHT-10/2
				Window area	Fixed area	
ESC ESC ESC ESC ESC ESC	AOH A1H A2H A3H A4H A5H	KANA LED ON KANA LED OFF ALPH LED ON ALPH LED OFF CALC LED ON CALC LED OFF	Only in EHT-10/2 EHT-10 has not LED	X	X	0
ESC	BOH	FUNCTION KEY CHECK MODE ON		0	0	0
ESC	81H	FUNCTION KEY CHECK MODE OFF		0	0	0
ESC	DOH	SET SCREEN SIZE	The number of horizontal columns cannot be specified	0	X	Q
ESC	D1H	CHANGE ACTIVE SCREEN	Newly added	\$	\$	\$
ESC	D2H	DIRECT DISPLAY	Attribute added	9	0	9
ESC	D4H	LOCATE TOP OF SCREEN		0	x	0
ESC	D5H	LOCATE BOTTOM OF SCREEN		0	x	0
ESC	D6H	SELECT CURSOR KIND		0	0	0
ESC	D7H	FIND CURSOR		0	x	0
ESC	D8H	SET WINDOW	Newly added	\$	x	X
ESC	D9H	SET ATTRIBUTE	Newly added	\$	\$	x
ESC	DAH	SET DISPLAY TYPE	Newly added	\$	S	X
ESC	EOH	SET DOWNLOAD CHARACTER	Font size is different.	Ø	0	0
ESC	FOH	KEYBOARD REPEAT ON/OFF		0	0	0
ESC	F1H	SET KEYBOARD REPEAT START TIME	Function deleted	x	x	X
ESC	F2H	SET KEYBOARD REPAET INTERVAL TIME	Function deleted	x	X	X
ESC	F3H	SET ARROW KEY CODE	Function deleted	x	x	X
ESC	F4H	SET SCROLL KEY CODE	Function deleted	X	x	X
ESC	F5H	SET CONTROL KEY CODE	Function deleted	x	X	x

APPENDIX Page 21 - 62

Code Function		Remarks	EHT-10		EHT-10/2
			Window area	Fixed area	
ESC F6H	CLEAR KEY BUFFER		0	0	0
ESC F7H	SET KEY SHIFT	Function deleted	X	х	X

(2) Details on display control functions

The details on the display control functions listed in the above table are explained. These display control functions are mainly used by BIOS CONOUT and BASIC Print. For how to use these functions for BIOS CONOUT, see the explanation of "CONOUT" in Chapter 4 of the Software Part. For the BASIC Print statement, refer to the BASIC manual.

Code : Function name

05H : ERASE END OF LINE Clears the cursor and subsequent columns of the line to blanks.

07H : BELL

Sounds the 880-Hz buzzer for one second.

**O8H : BACK SPACE** 

Moves the cursor one column to the left on the screen. If the cursor is positioned at the first column, this command moves the cursor at the last column of the previous line. If the cursor is positioned at the Home position on the screen, this command performs no operation. Notes:

- 1. If the cursor is to be moved outside the window screen, this command conforms to the follow or unfollow mode.
- If the cursor is to be positioned inside the fixed screen which is hidden by the window screen, the cursor cannot be viewed on the LCD screen.
- 09H : TAB

Sets the cursor at the first tab position to the right of the current cursor position. If no tab position is detected on the current cursor line, this command sets the cursor at the first tab position on the next line. If the cursor is to be positioned outside the screen, this command sets the cursor at the first tab position on the last line on the screen.

Tab position = Column (1 + 8n) n = 0, 1, 2, ...

See Notes 1 and 2 on Back Space (O8H).

OAH : LINE FEED

Moves the cursor one line downward on the screen. If the cursor is positioned on the last line of the window screen, the window screen is scrolled down one line. If the cursor is positioned on the last line of the screen, the screen is scrolled up one line. Note:

If the cursor is positioned on the last line of the fixed screen, this command performs no operation.

- OBH : HOME Moves the cursor at the home position on the screen. See Notes 1 and 2 on Back Space (OBH).
- OCH : CLEAR SCREEN & HOME Clears all the screen contents to blanks and performs home(OBH) processing.

ODH : CARRIAGE RETURN

Moves the cursor at the first column of the line. Notes:

- 1. If the cursor is to be positioned outside the window screen, this command conforms to the follow or unfollow mode.
- 2. If this command is entered immediately after one character is displayed at the last column of the screen, the cursor is moved at the first column of the previous line (the line on which the last one character was displayed).
- 10H : SCREEN UP

Shifts up the window screen by one screen. If part of the shifted window screen is positioned over the home position, only the part of the window screen on the home and subsequent lines is displayed. Note:

The cursor remains at the original position on the screen.

11H : SCREEN DOWN

Shifts down the window screen by one screen. If the shifted window screen contents are to overflow the screen, the screen is displayed so that the last line of the screen matches the last line of the window screen. See Note 1 on Screen Up (10H).

- 1AH : ERASE END OF SCREEN Clears all screen contents starting from the cursor position to blanks.
- 1BH : ESCAPE Enables ESC sequence acceptance.
- 1CH : CURSOR RIGHT Moves the cursor one column to the right on the screen. If the cursor is positioned at the last column of the line, this command moves the cursor at the first column of the next screen. If the cursor is positioned at the last column of the screen, this command performs no operation. See Notes 1 and 2 on Back Space (08H).
- 1DH : CURSOR LEFT Same as Back Space (08H)
- 1EH : CURSOR UP Moves the cursor one line upward on the screen. If the cursor is positioned on the first line, this command performs no operation. See Notes 1 and 2 on Back Space (08H).
- 1FH : CURSOR DOWN Moves the cursor position one line downward on the screen. If the cursor is positioned on the last line of the screen, this command performs no operation. See Notes 1 and 2 on Back Space (OBH).

ESC '%' : ACCESS CG ROM DIRECTLY Reads the character corresponding to the specified code from the character generator and displays it at the current cursor position on the screen. The cursor then moves at the next column.



\*1 The attribute-byte configuration is as follows:



\* : Can be specified only in EHT-10 vertical display mode

In EHT-10 (horizontal display mode) and EHT-10/2, only the reverse and secret attributes are made effective. See Section 4.4 for details.

ESC '(' : BLOCK REVERSE

Inverts and displays the specified length of the window screen contents starting from the specified position.

	550		EHT-1	0(V	ertical)	EHT-10	(Hoi	rizontal)	EHT	r-10	)/2
2	'('	Y: Y axis X: X axis	1 1	to to	14 12	1 1	to to	10 25	1 1	to to	4 20
3	Y	n:No. of reversed characters	1	to	168	1	to	250	1	to	80
4	х	characters									
5	n(H)										
6	N(L)										

Note:

If the screen is scrolled and the inverted data overflows the window screen in EHT-10/2, inversion is released.

ESC '\*' : CLEAR SCREEN & HOME Same as Clear Screen & Home (OCH)

- ESC 'O' : REVERSE ON Turns the reverse display mode on, causing subsequent characters to be inverted and displayed. See Note 1 on Block Reverse.
- ESC '1' : REVERSE OFF Turns the reverse display mode off.
- ESC '2' : CURSOR OFF Does not display the cursor.
- ESC '3' : CURSOR ON Displays the cursor.
- ESC '=' : SET CURSOR POSITION

Moves the cursor at the specified position on the screen. Note:

If the cursor is to be positioned outside the window screen, this command performs different processing according to the follow or unfollow mode.

Follow mode: When the cursor moves upward, this command moves the cursor on the first line of the window. When the cursor moves downward, this command moves the cursor on the last line of the window.

Unfollow mode: The window remains at the same position.



Vertical direction:  $l \le m \le maximum \ screen \ line$ Horizontal direction:  $l \le n \le maximum \ screen \ column$ 

ESC 'C' : SET CHARACTER SET TABLE Specifies the character set of the specified country.

1	ESC
2	'C'
3	ID

ID: Country identification character

J:	Japan	D:	Denmark
U:	USA (ASCII)	W:	Sweden
<b>F</b> :	France	:1	Italy
G:	Germany	S:	Spain
Ε:	Great Britain	N:	Norway
Notes:

- 1. Default value Japan or USA (ASCII) is specified by the DIP switch.
- 2. The characters that have already been output remain unchanged even if the corresponding codes indicate different characters in the newly-specified character set.
- ESC 'P' : SCREEN DUMP Outputs the VRAM data being displayed to the printer.
  - Note:

In EHT-10, this command performs no operation.

- ESC 'T' : ERASE END OF LINE Same as Erase End of Line (O5H)
- ESC 'Y' : ERASE END OF SCREEN Same as Erase End of Screen (1AH)
- ESC 7BH : SECRET Sets the character output mode to secret. Notes: 1. In EHT-10, characters are output in secret mode.
  - 2. In EHT-10/2, spaces are displayed. If the screen is scrolled and the secret data overflows the window screen and then returns inside the window screen, the character data stored in the screen buffer is displayed.
- ESC 7DH : NON SECRET Releases the secret mode.
- ESC 90H : PARTIAL SCROLL UP Scrolls up the m lines starting from line n by one line. Line (n + m -1) becomes a blank line.

1	ESC
2	90H
3	n-1
4	m

n: Scroll start line

1 < n < maximum number of screen lines
m: Scroll width</pre>

1 < m < maximum number of screen lines</pre>

- Notes:
- If value (n + m) exceeds the maximum number of screen lines, value m is automatically adjusted so that value (n + m - I) becomes the maximum number of screen lines.
- 2. The cursor remains at the original position on the screen.

ESC 91H : PARTIAL SCROLL DOWN Scrolls down m lines starting from line n by one line on the screen. Line n becomes a blank line.

1	ESC
2	91H
3	n-1
4	m

```
ESC 95H : SET SCROLL MODE
```

```
Specifies whether automatic scrolling is to be performed.
```

1	ESC
2	95H
3	m

m: Mode =0: Follow mode (default) =1: Unfollow mode Note: In follow mode, the screen is automatically scrolled according to the cursor movement. In unfollow mode, the screen is not automatically scrolled according to the cursor movement. ESC 96H : SCROLL UP 1 LINE Scrolls up the window screen one line. Notes: 1. If part of the window screen is to be positioned outside the screen, this command performs no operation. 2. The cursor remains at the original position on the screen. ESC 97H : SCROLL DOWN 1 LINE Scrolls down the window screen one line. See Notes 1 and 2 on Scroll Up 1 Line. ESC AOH : KANA LED ON Turns kana LEDs on (only effective in EHT-10/2).

```
ESC AlH : KANA LED OFF
Turns kana LEDs off (only effective in EHT-10/2).
```

- ESC A2H : CALC LED ON Turns the calculator (CALC) LEDs on (only effective in EHT-10/2). ESC A3H : CALC LED OFF Turns the calculator (CALC) LEDs off (only effective in EHT-10/2). ESC A4H : ALPH LED ON Turns the alphabetic (ALPH) LEDs on (only effective in EHT-10/2). ESC A5H : ALPH LED OFF Turns the alphabetic (ALPH) LEDs off (only effective in EHT-10/2). ESC BOH : FUNCTION KEY CHECK MODE ON Sets a mode in which, if a function key is pressed, the function assigned to the key is not executed but the code discrete to the key is returned. (YPFCMFLG = FFH) Note: See the explanation of CONIN for the information obtained when a key is pressed in this mode. ESC BIH : FUNCTION KEY CHECK MODE OFF Releases the function key check mode. (YPFCMFLG = OOH)ESC DOH : SET SCRIEN SIZE Sets the virtual screen size. EHT-10 (Vertical) EHT-10 (Horizontal) EHT-10/2 ESC 1 n : Lines n of 14 to 42 10 to 20 4 to 25 2 DOH Screen 3 n If the virtual screen size is specified, an area of the specified size is allocated and Clear Screen & Home (OCH) processing is performed. Notes: If the virtual screen size does not exceed 28 lines in EHT-10 1. (vertical display mode), the fixed screen area is allocated, enabling the fixed screen to be used.
  - 2. Set Screen Size cannot be executed on the fixed screen.
  - 3. Even if the virtual screen size is modified, the CP/M size remains unchanged.

ESC D1H : CHANGE ACTIVE SCREEN Specifies an effective user screen type.



Notes:

- In EHT-10 (horizontal display mode), Change Active Screen is made ineffective. If a virtual screen (scroll screen) size not exceeding 28 lines is specified in Set Screen Size and the fixed screen area has not been allocated in EHT-10 (vertical display mode), Change Active Screen is made ineffective.
- 2. If Change Active Screen is executed, cursor movement and displaying of one character on the specified screen are made possible. If the user screen is changed in EHT-10/2, the old screen contents are saved, making each CONOUT code effective on the new screen. (Two screens can be used independently. See Section 4.4 for details.)
- ESC D2H : DIRECT DISPLAY

Outputs a character at the specified position in VRAM. The character can be directly output at any position on the LCD screen.

			EHT-10(Vertical)	EHT-10(Horizontal)	EHT-10/2
1	ESC		1 40 14	1 40 10	1 40 4
2	D2H	position	1 10 14		1 10 4
3	Y	X: Horizont	al 1 to 12	1 to 25	1 to 20
4	Х	posición			
5	n	n: Characte	r code   OOH < n <	FFH	
6	m	m: Attribut	es		

Notes:

- 1. A character code corresponding to the character generator must be specified.
- 2. See Note 1 on Access CGROM Directly (ESC+"%") for attributes.
- ESC D4H : LOCATE TOP OF SCREEN

Moves the window screen at the top of the screen. The cursor remains at the original position.

### **ISC D5H : LOCATE END OF SCREEN**

Moves window screen at the bottom of the screen. The cursor remains at the original position.

ESC D6H : SELECT CURSOR KIND Selects the cursor type.

1	ESC
2	D6H
3	n

n: Cursor type
 = 0: Block & blink (default)
 = 1: Block & nonblink

- = 2: Underline & blink
- = 3: Underline & nonblink
- ESC D7H : FIND CURSOR Moves the window screen at the cursor position so that the cursor is positioned on the first window screen line. If the cursor is positioned on the LCD screen, this command performs no operation.
- ESC D8H : SET WINDOW

Specifies the position and size of the window screen in EHT-1C (vertical display mode). The cursor remains at the original position on the screen.

1	ESC
2	D8H
3	m
4	n

m: Window start line  $(1 \le m \le 14)$ n: Window end line  $(1 \le n \le 14)$ Notes: 1. Value m must not be greater than value n.

- 2. If the window screen is modified, graphic data is lost.
- 3. Set Window is only effective in EHT-10 (vertical display mode). This command can change the fixed screen display area (screen part that can be viewed on the LCD screen) by changing the position and size of the window screen.

ESC D9H : SET ATTRIBUTE

Sets character attributes. If this command is entered, the subsequent one character is displayed with the specified attributed.



n: Attributes (0: Not specified 1: Specified)



Notes:

- 1. In EHT-10 (horizontal display mode) and EHT-10/2, only reverse and secret attributes are made effective. If the data displayed with these attributes overflows the window screen in EHT-10/2, the specified attributes of the overflowed part of the data are made ineffective.
- 2. The default value for n is 0.

### ESC DAH : SET DISPLAY TYPE

Specifies the character display type in EHT-10. After the character display type is changed, Clear Screen & Home (OCH) must be executed.

1	ESC
2	DAH
3	n

n: Display type =0: Vertical display (default) =1: Horizontal display

## ESC EOH : SET DOWNLOAD CHARACTER Defines an external character from EOH to FFH.

EHT-10 (vertical) EHT-10 (horizontal) or EHT-10/2

1	ESC	1	ESC
2	EOH	2	EOH
3	n	3	n
4	p(1)	4	p(1)
5	p(2)	5	p(2)
6	p(3)	6	p(3)
7	p(4)	7	p(4)
8	p(5)	8	p(5)
9	p(6)	9	p(6)
10	p(7)	10	p(7)
11	p(8)	11	p(8)
12	p(9)		

n: Character code (EOH  $\leq n \leq$  FFH) p(1) to p(11) and p(1) to  $\overline{p}(8)$ : Font patterns

Note 1:

p(10)

p(11)

13

14

An external character pattern of 7 x 11 dots must be specified for EHT-10 (vertical display mode) or that of 6 x 8 dots must be specified for EHT-10 (horizontal display mode) or EHT-10/2.



For the Japan version, default font patterns have been defined for external characters EOH to E3H.



Example: To register the following pattern in E4H, perform as shown below.





ESC FOH : KEYBOARD REPEAT ON/OFF Controls the repeat function of keyboard keys (including the touch-panel keys).

1	ESC
2	FOH
3	n

n: Switch
 =0: Repeat function ON
 =1: Repeat function OFF (default)

ESC F6H : CLEAR KEY BUFFER Clears the key input data buffer contents and deletes the pre-hit keys.

# 7. BDOS FUNCTION LIST

- The interface for calling BDOS in EHT-10/EHT-10/2 is the same as that for calling BDOS by CP/M. See Chapter 5 in Software Part for details.

- This table provides brief explanation of each BDOS function. Refer to the relevant CP/M manual for more details.

Nusber	Function Name	Input	Outreit
0	System Reset	C : OOH	None
1	Console Input	C : 01H	A : Input char
2	Console Outout	С : 02Н	None
		E : Output char	
3	Reader Input	C : 03H	A : Input char
4	Punch Output	C : 04H	None
		E : Output char	
5	List Output	C : 05H	None
		E : Outpur char	
6	Direct Console 1/0	С : 06Н	A : Input char (input)
		E : OFFH (input)	: None
		; Output char	
		(output)	
7	Get LOBYTE	С : 07н	A : LOBYTE
8	Set IOBYTE	C : 08H	None
		E : LOBYTE	
9	Print String	C : 09H	None
		DE : Address at which	
		the string is	
		stored.	
10	Read Console Buffer	C : OAH	Loads the buffer with
	_	DE : Buffer address	entry from the console,
11	Get Console Status	C ; OBH	A : Console Status
12	Get Version Number	C : OCH	HL : Version Number
13	Reset Disk System	C : ODH	None
14	Select Disk	C : OEH	None
		E : Disk number	
15	Open File	C : OFH	A : Directory code
		DE : FCB address	

Nusber	Function Name	Input	Output
16	Close File	C : 10H	A ; Directory code
		DE : FCB address	
17	Search for First	C : 11H	A : Directory code
		DE : FC8 address	
18	Search for Next	C : 12H	A : Directory code
19	Delete File	C : 13H	A : Directory code
		DE : FCB address	
20	Read Sequential	C : 14H	A : Return code
		DE : FCB address	
21	Write Sequential	C : 15H	A : Return code
		DE : FCB address	
22	Create File	C : 16H	A : Directory code
	_	DE : FCB address	
23	Rename File	C : 17H	▲ : Directory code
		DE : FCB address	
24	Get Lugin Vector	C ; 18ii	HL : Login vector
25	Get 2isk Number	C : 19H	A : Disk Nunber
26	Set DHA Address	C : 1AH	None
		DE : DHA address	
27	Get Allocation	C : 1BH	HL : Allocation address
	Address		
28	Write Protect Disk	C : 1CH	Nane
29	Get R/O Vector	C : 1CH	HL : R/O vector
30	Set File Attributes	C : 1EH	A : Directory code
		DE : FCB address	
31	Get DPB Address	C: 1FH	HL : DPB address
32	Set/get User Code	C : 20H	A : None (Set)
		E : OFFH (Get)	: User code (Get)
		: User code (Set)	
33	Read Random	C : 21H	A : Return code
		DE : FCB address	
34	Write Randca	C : 22H	A : Return code
		DE : FCB address	
35	Compute File Size	C : 23H	FCB r0, r1, r2
		DE : FCB address	
36	Set Random Number	C : 24H	FC8 r0, r1, r2
		DE : FCB address	

Nunber	Function Name	Input	Output
37	Reset Disk Drive	C : 25H	None.
		DE : Drive vector	× .
38			
39			
40	Random Write with	C ; 28H	A : Return code
	Zera File	DE ; FCB address	
251	Verify File	C : OFBH	
		DE : T~FCB	
252	Remove Tape	C : OFCH	
253	Hount Tape	C : OFDH	
254	Read Tape 10	C : OFEH	
255	Create Tape	C : OFFH	
	Directory	DE : T~FCB	





APPENDIX Page 21 - 81

#### 9. SAMPLE PROCRAM LISTS

(1) SAMPLE 1. BIOS CALL



(2) SAM	PLE 2. CONIN CONOUT	:				
				BIOS CONTS	CONOU	T CONST
						N ALTIN SANPLE FOULEAN
		;	NOTES			
				KEVIN READ	e prog SV and	ram uses CONIN CONDUT CONST displays I character on the LCD,
		1 1 1 1	() A888	able condie	ion	<)
			. 280			
			() load	ing address		< b
			· PHASE	LOON		
			<> cons	tent velues		43
£803		TDOSW		equ oe	803N	
2809		CONST			809N	
E 2 7 2		CONOUT READSW		20U 02	80CR 872H	
2896 2890		TOUCH		EQU 02	896N	
1000		1		201 01	0000	
		1		MAIN PROCE	۸M	
		1				
0100	31 1000	STABT:	LD	SP. HAINSP		; Sat stack polotar
0103	02 02 CE 2872		10 C/ 11	C.02N 814064		; Sead dip SV or ENT-10,10/2 : Sted
0108	26 90		NO	800000008		Cheak the bit 7
	rz au		L #	1010008		PHT-10/2
0100	28 14	:	3 8	5,85380D		; Jump to ENT-10/2 coutine
0105	21 013E	TCHEEY:	LD	NL.INPDAT		; else 207-10 ; Make input-date area for 207-10
0111	06 01 CD 889C			8,01N 8271N		; Set the Alphabet input wode 2 logut the Alphabet
0116	21 0132		LD	HL, INPDAT		
0114	CD ZBOC		CALL	CONOUT		; ; display the inputted data (1 character)
0110	08 01 13 0158		10	C, 018 Dr. 627812		; Nako ; TOUCH key block ; Sat the key block data
0122	CD 1996		CALL	TOUCH		; Display a key block
		:				
0128	06 01	VEAGO:	LD	8,01H		; Change the keyboard mode to Alphabet
0124	CD 289C CD 28C9		CALL	ETTIN C0519		; Read the data from CDB:
0130	4 P CD_ 8 80C		LD	C.J Cosout		; : Dieplay the data
0114						
0134	CD 1806	Stat.	CALL	CON 5 7		; Cheek another key input
0137	PE 00 26 P9		C P J B	308 2.5817		t tt any say is pressed, court
0138	CD 2803	:	CALL	NBOCT		7
					8 708	effected 32bvide
			•••••			
6135		SPEAT:			2 0 0	
0136			9.0	0,3,0,0,0,	a	
0142	00 00 00 00 00 00 00					
0148	00 00 00 00 00 00 00 00		0 8	0.0.0.0.0.	o.o.o.	<b>0.0</b> .0.0,0,0,0
0152	00 00 00 00 00					
0134	00 00 00 00					
		6 F	•••••			***************************************
		÷				
0158		: REYBLX :			··· ··-	
015E	04 00 02 01 20 07 00 45		08	4,13,2,1,2	a N , 07N	. 0 , " 0 % 0 '
0166	42 44					
		1				
		ENO				

				:				
						8109 L	ES7 FUNCH	BEADEN INFORM SAMPLE PROCEAM
					NOT			
				:	AUTE:	This p	rogram rec	eives 1 cheracter date from 85-2320
						10 100 100	B) , PUNCH printer u	the data on the LCD, and LIST the data nit.
				÷				
				i	()	eble co	ndition ()	
				i	- 280			
					c) load	ling add	****	
					. PHASE	100M		
					() con	stant ve	lues ()	
2803				VBOOT		2 Q U	0 8 8 0 3 N	
2800				CONCUT		200	OFBOCH	
2807				PUNCH		200	0 8 8 1 2 8	
8815				READER		290	028158	
882D				L15797		290	0 E 8 2 D H	
EBAS				INFORM		291	028428	
0003				I O B V T Z MAINSP		2 G U 2 G U	01000N 000038	
• • • • •								
						MAIN P	80CBAN	
0100	• •	1000		S (AET:		-	NGP	: Set the stack pointer
0100	21	0003			LD	NL, IOB	YTE	
0106	3.6	59			LD	4 8 2 3 , 5	8 M	; Change the PUN: device to the LCO
0106	0.8	0 5			L D	C,098		; Set the parameter of INFORM
010A	C D	EBAS			CALL	INFORM		; Look the LSTEBS address
0100	22	0150				(LSTER	EVD1.01	1 2448 F21588 GDALERA
0110				LOOP :				
0110	CD	2815			CALL	BRADER		Band 1 character from #5-232C
0113	4 7					C / A		Preserve the redister
0114	60	6812			CALL	PUNCR		Display the read character on the LCD
0112	C D	P 8 2 D			CALL	LISTET		Look the printer statum
0118	2.4	0150			LD	HL, (LS	TEBBADI	LOAD LSTERS address
0116	1 6				LD	A, (RL)		: Load the content of LSTIBE
0116	T E	00			C P	0		Printer connected 7
0121	2 0	06			18	N Z , U S P	356	connected
				1	•			
0123	21				101			Set the LIST parameter
0124	67				CALL	LIST		Print the reed character
0129	63	0110			3 P	LOOP		1
				Denner.				
0128	21	0140		038336:	٤0	HL,ERG	171 S C	Display the error esseage
0150								
0128				05910:				
0128	48				10	C ( UL )		- 16 data io 0
0127	* 5	0.0			C P	0		then the end of data
0130	25	0 9			10	2,2500	A T	
2124	2.5				2 U S H	HE		; Preserve register
0135	CD	1900			C # 1 1	CONOUT		; Display ] character or sessage
0135	E 3				POP	N Z		Next data
0138	2.3				19	0.5.910		
0134	63	UIEE		1	4 F	0.57.00		1
0130				ENDDAT				
0 3 3 D	с 3	2903			3 P	WBOOT		1
				4 1 19				
						£ R R O R	9855AG2	
				:				
0140				ERRMSG:				ODH GAN
0140	5 O 3 4	72 68	6 E 2 O		ud		LET CHAUS	
0144	15	52 52	4 7					
0140	5 2	00 0A						
014 F	0 0				0.8	008		



C 0 8 7 2 0 8 7 2 0 8 3		6	: : : : : : : : : : : : : : : : : : :	NOTE: 1) 2990 .180 () land .PBASE () cons	This eat This eat This is able cond ling addr 100H tant vel 20U 20U 20U	DIE PROG SELOSE, S DIE Prod diskcopy ditton 0 EB03H 0 EB03H 0 EB03H	BAM: ETTBE, SETSEC, SETDNA, BEAD, WRITE THE IS executed only on ENT-JO. program from D: to E:. ()
E B 1 B E B 1 E E B 2 1 E B 2 3 E B 2 3 E B 2 5 E B 2 5 E B 2 5 E B 2 5 E B 1 B E B 1 B E B 1 E E E E E E E E E E E E E E E E E E E			SELD5N SETTB& SETSEC SETDAA READ WRITP TOUCN			0 2 8 1 8 M 0 2 8 1 8 M 0 2 8 2 1 M 0 2 8 2 4 M 0 2 8 2 4 M 0 2 8 2 A M 0 2 8 2 6 M	
1000			MAINS P 784MAX SECNAR		2 G U 2 G U 2 G U	01000M 40 64	; Fleppy disk
							· • • • • • • • • • • • • • • • • • • •
				* • • • • • *	NAIN PHO		
0100	11 1000		START:	1.0	SP. MAIN	5 P	: Set stack pointer
0103	0 E 1 B CD EBOC		ī	10 CA11 10	C,13H CONOUT		Cursor off ; Esc + '2'
010A	CD 190C		7	CALL	CONCUT		;
0100 0110 0113 0158 0138	2 1 0 0 0 0 2 2 0 2 2 B 2 2 0 2 2 D 3 2 0 3 2 3 2 0 2 2 A			10 10 10 10	(DSENCU (DSENCU ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) )	], 81 ], 21 ], 4	; A=0 ; Set first track number ; Set first sector number ; Set frive oumber (D:) ;
0118			LOOP:	13	A. ( 85K5	631	: Select dagk
0 1 1 E 0 1 1 F 0 1 2 F 0 1 2 6 0 1 2 5	4F 4E 00 CD 2818 FC 85			1 D L J C A L 1 L D G B 1 D	C, A E, 000 SELDSE A, B L		first dias act+ss ; first cosck ; from cosck ; lf BL + 0 theo error ; Set error code
0126	28 <b>23</b> 36 18			38	2,05210	8	
012A 012E	CD 6816			10 CA11	BC, ITRE SETTBE	809)	; Set current track number 1 No. then read this track
C 11 0135	ED 48 022D CD 2821			CALL	SETSEC	SCH1	; Set BOCIDE
0135	01 022F CD 8824			LJ CALL	201047 201047		Set 2114 Address
0 3 2 8 0 1 4 1 0 1 4 3 0 1 4 5 3 3 4 5	3 A 0 2 2 A PE 0 3 2 0 0 5 C 0 E B 2 7 I 5 0 5		N817ES:	10 CP JR CALL JC	A, (3555 03H 52, -817 READ ERRCHN	12 m 1 E S	; Check read /vrite : If DSSSUN+3 then BEAD : else +BITE : Sead 121byte ;
A F 1 D	01 01			LD Call	C, OlH WRITE		; No blocking vrite ; Vrite 125 byte
014P 014P 0150	8 J 8 8 1		8 8 8 C H K -	0 R J R	A N 2 , D 5 K E	8 9	; Check return personator ; 16 Acou then arror
0   5 2 0   5 3 0   5 7 0   5 A 0   5 C 0   5 C 0   5 C 0   6 ] 0   6 3	3 A     0 2 2 A       6 E     0 7       3 Z     0 2 2 A       7 E     0 4       2 9     8 D       2 A     0 2 2 D       2 C     1 D       9 E     4 D		·	L D 1 C P 3 R L D L D L N C P C P	A, (DSNN 0000011 0058900 049 2, L009 N1, (SEC L A, L SECNAX	( MU 1 8 1 , A 1 , A 2 , A	Check read / write If DSKNUM=3 then DSKNUM=4 ejse DSKNUM=3 Write cycle ? Yee. then write Updete sector number Check sector number
			AP	PENDIX	Page 21	- 86	

C ) 6 S C ) 6 S C ) 6 S C ) 6 A C ) 6 B C ) 6 B C ) 6 B C ) 6 B C ) 7 J C ) 7 J C ) 7 7 C ) 7 7 C ) 7 7	20 10 2A 0228 2C 70 FE 2S CA E803 22 0228 - 71 0000 - 22 0228 - 0 118	J 8 L 0 1 5 C 1 0 C 7 J P L 0 t 0 N 8 X T : L 0 J P	N 2 . NEXT NL , (TBKKUN) L A , L TBKMAS Z , WBOOT (TRKNUM), NL NL , COM (SECNUM), NL LOOP	Update track number Check track number If over track max , then V800T Set new track number A+0 for soctor number Set new sector number 8 ged / write again
0 1 7 D 0 1 7 D 0 1 7 D 0 1 8 0 0 1 8 2 0 1 8 4 0 1 8 7 0 1 8 8 0 1 8 8 0 1 8 8 0 1 8 8	06 F9 SP CB 23 16 00 21 0183 19 SE 23 56 E8 CD 01A7	D S # 2 8 R : S U B L D S L A L D A D O L D I NC L D Z R C A L	0 P S N R, A R D.00 M M L, E B B T B L B L, D E R, (B L) M L D.(M L) D R, N L L D S P M S G	Change error code 0,1,2,3,4,5,6 Set error code to OE E = E = 2 Set error message table Cet error table address Cet low address RL = message address
018F 0191 0194 0197	C D E B O B C D E B D E D T D D Z Z D D E D D D E D D	LD LD CAL CAL	C,01H 02,027810 1. TOUCH 1. CONTN	; Make 1 touch bey ; Set beyblock ; Make 'END' bey ; Wait bey in
019A 019C 019F 01A3 01A4	0 E 18 C D E 80 C C D E 80 C C D E 80 C C J E 80 J	10 CAL 10 CAL 19	C,18M L CONOUT C,'3' 1 CONOUT WBOOT	Cursof of 25C + '3' ;
		***	DISPLAY MESSAGE	2 STEING
		• • •	,	
		101	2:	
		; ()	EL : Top addres	a of message string
			NONE	( )
		-	NONE	
0147	18	DSPNSG:	А.(ИL)	; Cat a data
01	37 C 9	0 B R 2 T	A 2	: Check and mark 000 : If data is 00% then return
	£ 5	i PUS	8 N1	; Save pointer
01/P	FEOC	L D C A !!	C,A L CONOUT	: Display a characier
0148	2 3	P C P 1 N C	R 1. 81.	; Spdate jointer
0181	19 #4	a t	DSPNSC	-
		•••	DATA ABEA	
0 1 8 3 0 1 8 3 0 1 8 3 0 3 8 5 0 3 8 7 0 3 8 6 0 3 8 6 0 3 8 6	0101 0100 0108 0107 0107 0107 0108 0108	200721: 30 00 00 00 00 00 00 00 00 00 00 00 00	D 1 5 5 1 5 R READIER S 1 1 1 2 R S L 2 1 7 2 8 R 0 4 9 2 5 R R 0 4 9 1 5 8 L 3 3 2 R	
010	13 60 73 6-	DISKERR:	blak ont mount	R <sup>1</sup> . 0 0 M
01C1 01C5 01C9 01CD	14 69 73 68 20 60 67 74 20 60 67 75 60 74 00	DE READERS -	DIET DOG GOOM	
. 1 <b>D O</b> 0 1 D -1	52 65 61 64 20 65 12 12	0.8	'Read error',00	0 M
0108	07 74 UU 87 77 60 74	NBITERS:	'Write error'.	0 D H
0108	65 20 65 72 17 66 73 00	08		
0127	47 55 AP 24	SLCTEBR:	Salact Arrors	. C Q N
0127	63 14 20 65	U 8	ANTER OLIMA	• •
012F 01P3	00			
0114	52 2E 4F 20	DOWFERE: DO	'8/0 or write	grotect error',00N
		APPEND	IX Page 21 - 87	

0168	6 F	72	20	77						
0160	72	69	74	65						
0200	20	70	12	6 7						
0204	2.4	65	63	7.4						
0208	20	6.5	72	12						
0200	6 2	12	0.0	-						
0207		-	-			ETSTERR				
0205	5 2	65	61	6.4			DB	- Read /	HELLS PREPARE OOH	
0213	2 9	57	72	6.9						
0217	7.4	6.5	20	6.5						
0218	7.2	12	67	12						
021P	0.0									
0220						REVELS:				
0220	0.4	0.5	02	0.1			<b>be</b>	a i a 7	1 ODB 078 008 'PND'	( Red here
0224	0.0	07	0.0	4.6			20		,1,000,000,000, 200	, End Edy
0220	4.5	4.4	00							
4440	42	4.4								
									3 L A	
						i .				
0224						USENUM:				
UZZA							05			
0228						TRENDA:		•		
0229							05	2		
0220						2 SCHONI				
0220							05	4		
0229						UNA:				
0 2 2 8							05	128		
							END			
Hecros:										
Symbole	2									
8908	CONT	N			E 80C	CONOUT		0101	DISETRR	
022 P	ARG				0170	JZKXBB		0228	DSENCH	
J 1 A 7	0595	s c			014 P	EBBCHR		0183	EBRTBL	
020 #	E 7 S 7	E 8 R			0220	REJOUR		0118	LOOP	
1000	MAIN	SP			0177	NEIT		2827	READ	
0100	2 E A D	8 A A			<1P4	BOVPERB		0040	SECHAN	
0 2 2 D	SECN	U M			6163	SELDSK		Z 8 2 4	SETONA	
2821	5875	80			6916	SETTBK		0127	SLCTERB	
0100	STAR	T			2 8 9 6	TOUCA		0028	TREMAX	
3228	TREN	U 71			£003	VEDOT		287A	WRITE	
0108	VAIT	888			0144	VBITES				

No Patal errortsi

.

SI SAMPI	F 2. RFFA					
		;				
				8105 BE	EP SAMPLE	E PROCRAM
			*******			
		1	NOTE:			
				This se	ople prog	gram plays music
		1		by usin	g peeb.	
		1.0	()	able con	dition	()
			. 280			
			() loadi	las adde		
					- 23	
			- PHASE	1006		
			() conet	tant val	106	0
P 2 3 8		BPINTERL		zqu	0 P 2 3 R N	
6803		i VROOT		701	058038	
8839		8 E E P		LOU	OEBJ9H	
		1				
1030		MAINSP		200	010000	
		:				*********
		1		MAIN PR	OCBAN	
						••••
0100		START:				
0100	21 1000		LD	SP, MAJN	SP	; Set stack pointer
0103	14 <b>F21</b> B	;	L D	A. ( 8 P 1 N	78821	. Cet been interrupt sable.
0106	F6 80		0.8	1000000	0.B	Dilable 1 sec interrupt table.
0106	32 7238		LD	(893972	8£),A	; Set new beep interrupt during beep.
	21 0118	;	וח	-		Set the top address of sone data
0100		:		11013040		, ser the rol end of a strug care
0172		LOOP:				
0102	4 6		10	8, (NL)		; Sound type
4110	4 2		10	C.(NL)		Sound Jength
0111	23		INC	NZ		Nast pointer
0112	79		10	A,C		; If sound langth is 0.
0113	87 CA 5903		19	7 NROOT		; then and ov obta. • Red of data then \$8001
				2,2		
0117	25		PUSH	11 L		Save song table pointer
0118	CD 2839		CALL	eerp 		Sound Bestore some take pointer
0110	15 0		18	LCOP		Loop
		;				
		1				***************************************
		:				
		;				
0112		SCNC:				
0112			08	11.6.11		د با تا تا با
0126	11 02 11 02					
0124	11 09					
0120	11 03 14 09		25	0 شرائي د (		<u></u>
0124	1) 09 11 03					
0139	11 05					
013A			38			
0142	16 02 14 02					
0146	12 06					
0145	12 02 12 02		9 <u>5</u>	15,2,15	.2.75.=.	1
0150	00 07					
2152	0 0 0 2 0 7 0 2		0.8	15,2,15		14, 1, 15, 1, 15, 2, 15, 5
8210	07 07 08 02					
0154	07 02 07 02					
3150	01 02 01 02		DB	15,2,15	5,2,17,2,	17,5,17,2,17,2,17,9
016.	13 02 11 06					
0165						
0165	11 03 11 08		08	17,3,17		17,2,15,6,13,2,15,2
0172	0 7 02 11 02					
0176	0 P 06 00 02					
0170	00 02 00 12		08	13,2,13	18,00,6	
0190	0006					
0162	19 02 19 02		08	25,2,25	2,25,2,	۲, ۲, ۲, ۲, ۲, ۲, ۲, ۲, ۲, ۲, ۲, ۲, ۲, ۲
J 1 8 6 0 1 8 ▲	19 02 19 02					
0182	19 08		0.2			
0190	18 02 19 02		DB	24, 2, 29	5,2,24,3,	22,3,22,3,22,3,24,2
0184	19 01 16 09					
				Dago 21	- 90	

0195	16	09	16	03			
0190	38	02					
0192	15	Q 2	18	0 2		0.8	24,2,24,2,24,2,24,2,24,2,24,2,24,5,22,2
0182	18	02	18	02			
0146	18	02	18	8 0			
0144	16	02					
0140	18	02	16	08		08	24,2,22,9,20,2,17,2,20,12
0180	1 a	02	1.1	02			
6184	14	0 C					
0186	19	0 2	19	0 2		0.8	25,2,25,2,25,2,25,2,25,2,25,2,25,2,25,3
018A	19	0 2	19	02			
0188	19	0 2	) 9	0 2			
0102	19	8 0					
0104	18	02	19	0 2		00	24,2,25,2,24,3,22,6,24,3,22,9,22,3
0108	18	0 3	16	06			
0)00	18	03	36	08			
0100	16	02					
0102	16	02	18	03		08	22,2,24,3,24,8,22,2,20,2,22,8,20,2
0106	18	08	16	3 Z			
0104	1.4	02	16	08			
0102	34	0 2					
0360	16	02	14	0 2		O R	22,2,20,2,20,16
0124	1.4	10					
0186	0 0	0 C				08	00,0
						END	

•				8105 1	INGA TAGRE	LE DEOCRAM
			NOTE:	This s	ample prog	tree reads the clock.
		, , ,		and d	isplaya ti	ne time.
			() 888	emble co	ndition	0
		•	280			
			+> loa	ding add		< >
		•	. P H A S E	100N		
			() con	stant va	JUAN	( )
2803		48001		EQU	4680JN	
£ 80 9		CONIR		EQU	06903N 019069	
2 80C 2 84 2		CONGUT T LNOAT		ZQU IQU	OEBOCN OEB4EN	
2872		READSW		200	088728	; Bead switch
E 8 9 C		<b>KEYIN</b>		5 G U	028968	
1000		MAINSP		EOU	010000	; Stack pointer
			••••	MAIN PI	8067AM	· · · · · · · · · · · · · · · · · · ·
			SOTE:	Displa	y time unt	il BETURN hey is pressed.
0100	31 1000	STABT:	1.0	SP, SAII	N S P	; Sot aloch pointer
0103	21 071A		6.0	81,C052	2071	; Cursor off data
0106	CD 01A9		CALL	DSPHSC		; Cursor off
0109	21 0220		LD GALL	BL,DATE DSPNSG	nsc	; Date message : Display 'DATE'
	0.5 0.3	7	1.0	C 0.2 H		Read D1P SW
0111	CD E8:2		CALL	BRADSV		; Read
0114	86 80 A 7		08	100000 \$		11 A = 0
0117	25 )7	:	38	2.10981	E Y	; Then the eaching is 221-10
0119	1 0102	b 2 Y 8 O D :	10	02.0102	2.9	5 Else ENT-10/2 : Set cursor (1.2)
0110	CD 3150		CALL	SETCUR		· · · · · · · · · · · · · · · · · · ·
0122	CD OJAS		CALL	059850		Display 'TIME'
0121	06 03 C = = = = = C		CILL	RTAIN 8'038		Set normal reyboard bode
0175 0178	NE 32 0257		10 10	A (8_1191	E1,4	TTPE flag - 0 ,then EBT-10/2
0122	15 29		38	LOOP		:
0130	11 01 2 3	TCHREY:	1.0	DE 0103	3 M	; This is EUT-10 ; Set cursor (1.3)
6133	CD 2102		CALL	SEICUR		
0136	21 921T CD 01A5		CALL	8L, 119: 359456		; 5190107 '7176'
1110	02 01		12	C,018		; set the key block - Set key block descriptor
1111	C3 ±356		CALL	TOUCE		Hade 'END' hey
2146	11 12:5		12	A.JETH	E 7 7 A	O , then ERT-10
2149		LOOP:				•
0149	CD EB06		CALL INC	C O X S T A		; Rey in check ; Ts any key presed?
0110	20 07		18 CALL	<pre> &lt; Z , SK ] I</pre>	P	· Ces inputted key.
0152	PE 0D		CP	CDN		is the RETURN hey pressed?
0154	29 44	1	18	2,1150		, 162 / CHON 200
0156	13 0249	5 K 1 P :	10	0 E , N T 1 P	12	i fime descriptor
0159	0 E 0 O C D E B 4 E		10 Call	C,00H 7150A7		; Read time function ; Read time.
0130	(0) 1011	7	CALT	TIMPENS		· conners the new A old time
n 1 5 E 0   6 ]	29 86		18	Z,LOOP	-	if those are the same , then loop
0163	CD 01C5		CALL	TIMESTY	r	; Set new time data
0166	3 A 0 2 5 7 R 1		L D O R	A. [N_T) A	YPE)	; Choch mochine type ; If A = 0 .

016A	28 1A		38	2,8873	; Then BHT-10/2
					This is ENT-10
0160	11 0402			DE, COZN Setcum	Set curmor (4,2)
0107	21 0220		10	UL,DATE	Dete dete
0175	CD 0149		CALL	DSPNSC	Display the date
0178	11 0404		LD	DE.0404H	Set cursor (4,4)
0178	21 0230		LO	HL.TIME	Time data
0181	CD 0149		CALL	DSPHSC	Display the time
0184	18 63		JR	LOOP	i -
		KEY2:			This is ENT-10/2
0180	11 0801		1.0	D£,0801N	1
0199	CB 01FB		CALL	SETCUE	Set cursor (8,1)
0180	21 0220		CALL	05 PNSC	; Dete dete Bigolay the dete
0187	11 0802		LD	82,0862N	
0195	CD 0170		CALL	SETCUR	<pre>Set cursor (8,2)</pre>
8 2 1 0	21 0238		CALL	DSPMSC	. Display the time
0138	18 49		18	LOOP	
0140		TINEEND	1	HL.CUBRON	: Cureor on data
0140	CD 0149		CALL	DSPHSC	. Cursor on
0145	C3 8803		38	10084	Jump WBCC7
		i.			
		*	• - •	DISPLAY NEBBAG	GE UNTIL PIND OOM
				••••	
			PUIL		
			() ent	ry paremeter	6.3
		:		HL : Ten addre	ese of message string
			•••••	NONE	
		1	<> pre	served registers	8 ()
				NONE	
n9 PAIn	78		LD	A_(HL)	; Get message date
014	8 9		08	4	End ears 7
<b>6144</b>	C 8			•	
0140	4.8		LD	C A	: Set display date
0140	85		PUSH	HL CONDUT	: Save pointer to compare
34(0	CD EBOC		POP	81	Regtore sessage poloter
0101	23		150	81	Pointer update.
0183	15 . P4		18	DSPSSC	; Loog watil find 000
					***************************************
				GEBGE GLD L N	51 1185
		:			
			NOTE:		
		:	() 001	ry parameter	4.3
				NUNE NUNE DAIMAGIAI	• •
		4 		27 : Meture :	70114410N
				• 1 : 5•	w time and old line are the same.
			13.010	94 : 9 + •101007 brovens	· 1186 18 Cifferent inte old time
				5058	
0185		1136288			New type date
3195	31 0249		12	DE.011.12	; 01d 11me 2018
0185	26 06		. >	5,36R	; lata counter
0180		TLOOP:	LD	A. ( DE )	; Cot old time data
0180	LA BT		C P	1 H L 1	Compare with new time
0164	Ċ Ū		RIT	NZ	; if those are different, theo return.
0100	13		186	N L.	t hoturet oboste
0101	23		DINE	11009	toop 6 time
0102	C 9		A T 7		
4-6-		-			
		0		SET TIME DATA	·
		1			
			UOTE:		
				try parameter	6 B
		4			0
		4	F.W.	nost	
			<1 pr	eserved register	10 CD

		i		NONE	
0105		TIMESET	:		
0105	21 0219		10 10	HL.NTIME DE.OTIME	Set the time deta to the time area
8 3 1 0 3 3 1 0	01 0006 20 NO		1018 1018	86,8	; Yeer/wonth/dete/hour/esnute/second ; Move new date to the ald area
0100	21 0249	1	1. 0	BL, STINE	; Set BCD data to the pessage area (ASC))
0103	11 0220		10	DE, DA®E 8. D3H	; Cobe; ; HL 10 Bource , DE is destination ; B is counter
0108	CD 018C	SET 10 :	CALL	SETASCII	CONVERT BCD to ASCII
0100	23		INC	н L D E	Pointer update
0100	10 #9	ı .	DJNZ	58710	; Loop 3 times.
0 1 D F 0 1 E 2	11 0236 06 03		10	8,03M 06,1176	; Time data setting area;
0124	CD 018C	58720:	CALL	58745CJ1	Convert BCD to ASCIT
0127 0128	13		3 N C	DE SP320	; forn 3 times
0128	C 9	•	827	3 2 3 2 4	trond o trades
		•			
			*****	SET ASCII DATA	- FECN 86 0 6 7 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6
			NOTE:		
		:	() #D1	ry parameter	
				BI : BCD deta DE ; ASCII dat	eddress A Belting Address
				NONE	
			t, pre	81	
0180	78	SETASCI	): ID	A.(8L)	; Get the PCD date
0120	P 5 0 P		PUSN BRCA	A F	Save the BCD data Nove the MSB 4 bits to the LSB 4 bits
0187 0170	0 P 0 P		88CA		
0101	0 P C D 0 1 P 6		39CA CALL	NEXT	Set 199 ASCIJ dala
0195	F1	N E X T :	POP	A P	; Restore the BCD data
0176	C6 30		ADD 1 D	A.309	, coeg the solo onto ; Change to the sSC11 data ; Solo the sSC11 data
0178	13		INC	02	; Setting pointer uponte
0410					
				SET CURSON POS	11710X
		a P	NOTE:		
		4	+) #DZ	ry perameter D 1 X directio	in .
			c) ret	UEN DATABELET VOSE	41
		н В	() pre	SONE	
0 1 F D 0 1 F D	δ	SETCUR:	PUSH	5 8	; Save cursor position
G 3 F E O 2 F F	ວຽ ດະເຄ		PUSH LO	0 C , 1 5 H	ξεε
0201	CD IBOC OC 10		LD	COSOCT C,'='	
0206	CD E805 D1 59		POP	CONDUT De	Rostore Y value
0208	26 1F		A 0 D	A , O 1 F M F A	Set y value to Cred.
020E	CD 280C		CALL		;
0212	7 A C G 1 F			а, 0 а, 01 гн	
0215	17 CD E80C		LO CALL	C , A CONDUT	Set I value to Cres.
0219	<b>C</b> 9		# E T		;

						:					
						CUEROFE					
0214						CUSEUIF		1.0.0 1.2.5			
0210	1.15	34						104.		;	CUPBOF OFF DATA
0210						CORECH:					
0210	18	33	0 0				08	1041.11	, GDN	:	Cursor on deta
0 2 2 0						DATHSG:					
0330	0 C		-				19	d C H		1	Clear acreen
0223	4 4	4.1	5.4	4.5			08	1 DATE:		1	
0225	3 A										
0226	0 0						D 18	0 О Н		;	End meck
0227						TINNSC:					
0227	5.4	4.9	40	4.5			08	17188:*		:	
0228	3 4									-	
023C	0.0						08	0.0 H		:	End mark
0220						DATE					
0220	3.0	30	2 2	10			n e	100/00/	00'		
0231	10	3.9	30	10							and another area
0235	0.0	•••					D.e.	0.0 H			
0233								000			
0230	2.0	3.0				1226		100.00	0.0.1		
0236	30	30	34	30			0.8	uu:uu	00		IIme meesage Aree
ALSU	70	JA	3.0	30							
0232	00						08	UGN		-	
023F						REABTR:					
023F	04	3 D	02	01			0 8	4,14,2	1,0DH,078,00,	1 S N O 1	
0243	σ0	07	0 0	4.5							
0247	4 2										
0349						NTIME;					
0249							05	7			
0250						07178:					
0250							0 S	1			
0257											
0767							85	1			
0237								•			
						230					
Tecroe:											
Symbule											
( 8 0 8	CONI	N .			EBOC	<b>TONOUT</b>		2306	CONST		
0218	CUS8	017			0210	CUSEON		0220	DATE		
022u	DATH	S C			9 4 1 0	DSPMSC		0186	# E Y 2		
3237	E E Y B	LE		1	0119	E EYBOD		289C	R E Y 3 H		
0149	LOOP				1000	MAENSP		0257	3_TTPE		
01/6	XEXT.				0249	NTINE		0250	OTIME		
8877	READ	SN			0108	SETIO		0124	5 E T 2 O		
0186	SETA	s c 1	1		0 1 # D	SETCUE		0156	5 X L P		
0100	STAR	7			0130	108521		2842	112047		
0236	TIME				0195	TISICHE		2130	TINEEND		
0105	7140	527			0227	113356		0190	71998		
	TOP	/ M			2803	SROOT					
	IVEL	4									

So Fetel error(a)

		7 Ø									
				8105 E	510X 540P 800880888		P20687W				
		:									
		•	NOTE: This secole				ananan diantaya aka ananis da ka				
		4 Ø		and set	nds the i	np	utled data.				
		5 6 6	()	mble com	ndition	e	>				
			. 280								
			11 1968	116	r e 8 8	۲	,				
		:	- PHASE	1008							
		-	() cone	tant va	luee	¢	>				
2903		FROOT		2 <b>8</b> D	860830						
6906		CONST		8 Q U	<b>EB0E1</b> 0						
£909		CONTR		E G U	028098						
2854				200	023063						
2869		CALLE		£ Q U	028698						
2872		READSW		100	028728						
6996		TOUCH		EQU	628968						
E89C		REAIN		800	028968						
		I I I I I I I I I I I I I I I I I I I		540							
0010		RSOPN		2 Q U	101	÷	85232C open function				
0030		N 3 C L 3		2 Q U	204		logut status function				
0040		ASCST		EQU	408	:	Output status function				
0050		8 S G Z T		290	5 O N	-	Cot function				
0060		RSPUT		290	608	1	Put function				
0181		83288		5CU	9 O N	;	Peror status function				
		1									
0000		CR		200	008		Carriage return code				
0018				201	LAN	- 1	Line feed code				
8010					0.001.04	•					
		242400		IQU							
003C		X L S R S C R X S Y 3 S C R H A 1 M S P	N	200 200 200	0 0 3 C M 0 0 3 C M 0 0 3 C M	;	Change to user screen Change to system screen				
		i		MAIN P	BOCRAN		******				
0100	11 1000	START:		CO 9419			Set criek on other				
		:									
0103	08 03		10	C,038		1	Get DISBNE address				
0108	22 0194		20	(01588)	K), AL	;	Save DISENE address				
0108	21 8010	:	12	R.L. 585.	A D 8		Copy open perseter from system area.				
0100	11 0128		L D	DE, OPN	P 9 N	-	Application persector eres.				
0111	01 0009		LD	8C,9		1	Parameter Number				
9714	08 03	4	1019			÷	Copy				
		-	1.0	8, <b>01</b> H		÷					
			CALL	6 Z Y I N		;	Change the beyboard mode to Alphabet				
0116	23 0328	•	1.0	SL.OPS	PRS	1	Open parameter				
0319	06 10		20	9,950P	S	-	BSZ32C open function				
0118	CD E354		CALL	85102			OPEN				
2118	87		6	4		÷	Sfror reiven				
011 -	CS 6803		2 P	NZ, 680	01		YES . then \$8007				
0122		NEVCHX:									
0122	CD 2806		CALL	CONST		;	Cer key inputted status				
0115	30		150	A		;	Input any key?				
0:26	CC 013F		CALL	2,207		÷	Yes, then put the data				
0130	21 0185	,	1.0		P 8 3		Get incus status				
0120	06 30		LD	8.8515	1		Input status function				
0128	CO E854		CALL	15103		:	Cat input status				
0131	30		IXC	A		:	If there is receiving data.				
0135	CC 0162		CALL	2,021		:	then get the date				
0135	19 69		1 .	REVCHE			reng				
0137		PEND									
0137	06 20		1.0	B. BSCL	s	;	Close #S10X				
0139	CD 2854		CALL	SICI NBCCI		-	Broaria and				
0130	CJ K903		1 A	- 8001			······································				

			:					
						PDT IKPUTTED 0/	4 T A	10 852320
			•					
					ROTE:			
					() entr	y parameter	6	)
					•) retu	IFN persector	¢	
			i		4	SON2		N
					( <i>, , ,</i> ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	NONE		,
0137			PUT	:				
013P	03				CALL	CONIN	-	Cet inputted date
0143		0 3			C P	038	1	If inputted key is OBR,
0145	CA	E 8 0 3			3.6	2, 98007	2	than end of program
0148	c s				PUSU	8 C	:	Save input key code
0149	C 5				PUSH	80		Save input hay code
0144	12	0.0			CP ID	CR CI7	-	If inputied Rey 18 BYTURN, then it console out
		2800			CALL	Z,CONOUT		
0191	C 1				POP	BC	1	Restore input key code
0192	CD	280C			CALL	CONCUT	;	Consols out inputting data
0195	<b>C</b> 1				POP	8 C	;	Restore input key code
0156	21	0188			10	BL, OPNPBM	1	Put inputting data to 15232C
0159	06	80			CALL	8,85PUT 8510X		Put deta Put deta
0158	C 4	0148			CALL	NZ,EBBDSP	- 1	If error is returned, then display error
0181	6 3				8 Z T		1	
						GET RECEIVED D		
					NOTE:			
			;					
					() enti	SONE	•	8
						TO DATABATAT		b
			:			SUDE		
					() pres	JONE SUBLICUTE		•
0162			G.E V	:				Chark error status
0102	60	2854			CALL	ROISE	:	Cat arror status
0167	Z 6	74			AND	011101008	;	freq happened ?
01 u d	C 4	0145	5 A.		CALL	NZ, ERECSP		105, than display ever
0160	21	0128			1 D	HL,OPSPEH	;	Cat received data
0157	06	10			10	1, BSCET	1	Col function
0171	C D	2854 0149			CALL	NZ. 288D5P	1	if error, then display error
01/4			1					
0177	4.1				10		- 1	Console out received data
0178	C 0	8800			CALL	CONODI	:	Display received data
0172	C D	0157			CALL	BVSCPP		Reverse off
0191	63				361		;	
						SEVERSE MODE O	N	
					5072:			
					<> ent:	ry parameter	¢	1
					() 781)	urn persector		>
					()	NONE None		3
			1		· µre	80		
0152	C 5				PUSH	8 C	;	Save BC register
0183	0 8	18			LD	C . ESC	÷	Beverse on command
0185	C 0	2 8 0 C			CALL	C. (0 *	1	235 - V
0190	C 0	E80C			CALL	CONQUT	1	
0160	C I				POP	8 C	ŧ	Restore BC register
0122	۶ ک				867		i i	

.

			• • • • • • • •	REVERSE RODE OF	• • • • • • • • • • • • • • • • • • •
			4) entr	y garageter	()
			·) retu	NONE En parameter	4 3
				NONE	
			() grau		
0187	cs	EVSOFF:	PUSN	80	: Seve BC register
0190 0192	02 18 C8 280C		LO	C. 25C	Beverse off command
0199	02 31		LD	c,'1'	:
0194	c1		CALL POP		Bestore BC register
0198	C 9		827		
		•			
				DISPLAY HESSACE	UNTIL FIND CON
		:			
			NOTE:		
			<> ente:	y persenter UL : Mescage de	() ta top addres=
			•• F850	en parameter NGNE	4 7
			() pres	erved registers	e 3
				JUBE	
3]9C 019C	71	DSPNSC :	10	A.(8L)	; Cet display data
0190	37 C 9		08 887	A 2	: Check and code : Yee , then return
	4.5	:	1.0	C.A. (11)	
0195	25		PLSU	#1	Save data pointer
0141	CD 880C 21			BL .	; Concoja out the data ; Seatore date pointer
0145	23 18 74		1 M C J 8	UL DSP:ISG	; Peinter update ; Loop
					***************************************
		4 3	******		
			NOTE:		
			() entr	y paremeter	0
			·	A : Erzor statu en persenter	•
				NONE	0
			· pres	ALL registers	
		; g R B D S P :		. —	
8410	P 5		7 U S M	3 C	Save 2)) registers
0 1 A 9 0 <b>] . A</b>	05		PUSH	D E 91	
6148	£ 5	1	21. C B	M	Save wrrdr status
3140	25	+			Abones 14 113162 SCTURD
			15	A, JETH	Set system tank
			LD CA11	(DISZNK),A CALLX	Call OS jump table
			12	с, эсн	Clear screen & howe
			CA11	CONOU <b>1</b>	
	A310 15A			N1, ERRMSC DSPHSC	Set error message address Display 'ERROB occured'
0180	CD 019C				Bestore error statue
C193	<b>F</b> 1		10	8,3	Set loop counter
0184	<u>ub ua</u>	NTAREC :	LD	с, 30н	Creg = '0'
0186 0186	0 E 3 O 0 7		я1Сл ] R	NC, WTBIT	Shift loft
0189	30 0) 8C		( » C	c	Creg = 1'
0180	-		PUSH	A P 8 C	Save Area Save Bree
0180	c 5		CALL	CONONL	Diapley   bit
0186	CD EBOC	APP	ENDIX	Page 21 - 37	

01C1 01C2 01C3	C ) P ) 1 () P )		0 q 0 q 0 q	0 P 80 0 P A1 1 N Z N 1	C F TAREC		; Gestore Bree ; Bestore Ares ; Loop S time
			LD CA AN DR JB	0 C 1 L B 1 S D 1 ( 2 A 8 S	, 02 H E A D S V 0 0 0 0 0 0 0 Z , N E Y B O	8	; Read DIP SW ; Read ; Check bit 7 ; If bit 7 < 1 then this is EHT-10/2 ; Juep to EHT-10/2 routine
			i LD i LD i CA i CA		,019 2,x2981 0UCH 2 <b>X</b> 4	M	; EHT-10 routine ; Nake 1 key bjock ; Set key block dete ; Gieplay s key block ;
0105			LEYBOD: LD		.01H 241N		; Change the keyboard sode to Alphabet
0105			I NEXT: ; CA	LL C	0 N I N		; Nait for key in
			: 10 : 10 : 10 : 10	1 11 2 A 2 (1 1	X, XUSAS , 0 P P H D 3 S BNK) A L 1 X	СЯN , А	; Change to user acreen ; OS bank ;
01C5 01C7 01C8 01C9	2) C1 C1 C9		ם סק סק אני אני	) P R) ) P 0) ) P B( ) P A) E T	L E C P		Bestore regletere
					47.4 A98	A A • • • • • • • •	
<pre></pre>	45 52 52 4P 52 2u 4P 43 43 55 52 45 44 00 0A 00 41 20 30 20		20935C: 08		TBFOR 0	CCURED',	00M , 0AM , 00H
0102	00		1				
010P 010P 0123 0127	03 02 03 01 00 07 00 50 72 65 73 73		EFTELE: De		,14,3,1	, 0 D H , 0 7 M	, 0 C M , ' P r e e e *
			••		0 8 X 4 A A A A A A A A A A A A A A A A A A	A A	••••••
0128 0128			028785: DS	s <u>e</u>			; 25303 open patameter area
0174 9174			DISBNE: DS	5 2			; Destinction bank area
Secros:			E N	60			
State of a second secon	: CALLX COVST 257MSC 257MSC 257MSC 252LK 2421X SfX2 PUT RSE88 SFX2 24PUT SRSAD8 V800T XSVS5CBS	E 9 0 9 0 0 0 0 1 2 0 1 2 0 0 0 0 1 2 0 1 2 0 0 0 0	C 2 % 1 % C 3 % 1 % C 5 % P C 5 % P % R	3 5 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	B O C 1 F 4 1 C A S A S 1 C 2 2 S 7 0 2 D B S 4 0 2 D B S 4 0 4 0 1 S 2 6 9 6 1 B C	CONOUT DISBNK ERRNSC INFORM SEVCUK AAINST PIND RSCLS BSIDX RSOST RVSON TOUCH UTBIT	

No Patel errorie)

				BIOS CETPER SAMPLE PROCEAM						
				NOTE:						
					This s define	ampla progra d hey code i	ib zi m A norma	aplaying present 1 mode		
			*	() magable condition ()						
				Icading address <3						
			i	· • • • • • • •	1008					
				<) con	stant va	lues ()				
2803			: V 8007		FQU	028030		VECOT entry address.		
8083			CONTR		UDI	828098 88806 N		CONIN entry address.		
288C			GETPPE		200	0286C8		CETPFE entry address.		
2872			BEADSW		20U	028728	:	BEADSW entry address.		
1000			MAINSP		200	1000M		TOUCH ENTry Barroom. Stack Deloter.		
0003			BREAR		291	038				
			1							
			i		NAIN PROCRAM					
				NOTE;						
0100			START:							
0100	31	1000		LD	5 <b>P</b> , MA EI	9 Z N	:	Set stack pointer.		
0103	0 2 C 2	880C		LJ CALL	C, CCB Conout		1	Clear acreen.		
0108 010a	0 2 C D	02 8872		LD CALL	C,2 BEADSW		: 7	Bladsy parameter. Read DIP swich.		
0 1 0 D	07			BLCA			:	Check 281-10 or 287-10/2		
Q 1 C K	18	08			C' 3E15.		:	If target machine to EBY-10/2 then		
							1	ship key block set.		
0110	0 2	02		10	C.2		-	The number of key black.		
0115	CD	2 2 9 8	r - 1 0 <b>b</b> - D	CALL	TOUCE		÷	Set hey block.		
0110			1							
				GET EET CODE TABLE DATA						
				NOTE :	This p	rocadure is ;	getting	all hey code table data.		
0115	0 2	01		1 0	C.1		;	Normal adde.		
0114	21	5 F 0 1 5 5		10	3.0778			Bead all bey code.		
0118	¢ 9	TBGC		CALL	CETPPE		:	Get hey code date.		
				••••••	DISPLAT					
			4 9							
				SOTE:	This p	rocedure is :	display	jng key code table data.		
0122	30	02	:	10	c.2		0 0	BEADSN parameter.		
2124		1015			82.AU34		:			
0127	35	04		18	C. TTPE	1.1		CHARA SHI-IN OF COL-INIA		
0124	06	4.6	19921:	1.10	B.70			Set loop counter for EBT-10		
0120	18	05		3.8	LOOPI			(2TR-10 has 70 key code#1		
0122	06	23		LD	8,34			Set loop counter for £87-10/2		
0130	23	0185		10	NL, PPR		:	Set diepley data buffer address.		
0133			L00P1:				;			
0133 0134	C 5			មជ 5 ស ម ជ 5 ស	8 C		;	Save loop coulter.		
0135	12			10	A.(HL)		:	Load display deta.		
0136	CD	0149		CALL	TADES		:	Display data.		

0139	C0 £109		CA11	COFIN	; Get any inputted key.
0130	FE 03		C P	BBZAX 7 NROOT	; STOP code7
0132	CA ENGI			2,68001	; two. then end:
0141	C 1		<i>P</i> G P	80	Bestors loop counter.
0142	E) 23		207 18C	8 L R L	, sestors pointer. ; Pointer update.
0143					i j
0144	) C D	1	0 J N Z	LOOPI	Loop.
0146	C3 2803		J P	TDOEV	tod.
0110					
					; • • • • • • • • • • • • • • • • •
		Р 1 1		DISPLAY DATA	
			NOTE:		
				Display data in A reg	ister in here image.
0149		SPDAT:			:
0149	75		PUSH	AP	; Save data.
0144	C8 34		28 L	*	•
014C	CB 37		SBL	A	
3460	CB 3F		32L 581	A	
0150	CD 0164		CALL	TBSDAT	Translate upper 4 bit to ASCII
					code. - Lond display data to 6 resistan-
0155	••		2.0		
0156	CD 230C	1	CALL	CONDUT	; Display ASCII data.
	<b>F1</b>		P 0 P	A.7	: Restore data
0154	26 01		AND	Q P H	
0150	CD 0164		CALL	TESDAT	; Translate 1 wer dolt to ASCII
35F	4.0		1.0	£ . A	; Lond display data to C register.
			e . 1 1	6050117	· Disolay ASCII data
0160	CB XBUC		227		
•					
				TRANSLATE BIHARY DATA	10 A5C11 DATA
		-	_	Translate binary data	to ASCII code.
		Ĩ			
		1		A : binary da	ta.
				() return perseter	• >
				A : ASCII dat	• •
0164	22 GA		C P	0 A N	Data ( GAN?
0166	30 03			NC.IBSDATS	50.
0105	PE 20		2 2	306	Data ( CAB
0164	٤ ٩		2 2 7		
0 ] 6 8	37.08	TESPATS:	10	C. 99H	Data > - GAN
0160	9.3		503	c	
0362	26 10			408	
0110	¢ 3				
					;
				COUCH SEY DISCHIPTON	
		-			
2111		Tattaat			
3171	02 08 32 31		3 8	1,14,2,1.0DH,7,0,"CNT	1
3175 0179	00 0+ 00 43 42 54				
3178	04 38 32 01		0 2	4.14.2.1.038.7.0, * 250	•
0172	03 07 00 45				
6133		;			
		1			
				MEI CONE HEAD BULLES	
010		PPNBUP:			
0185		-	3 0	7 0	; Key code deta reading area.
			7 N D		
			- ··· ··		

			*****	8105 P	UTPPK SAMPLE	PROCRAM			
			ROTE:	This e user k and se (for E	asple program by table in n ts the dete f NT-10/2)	este s orcel s or dieg	the data to the code, pleying it.		
			<) ase - 280	e=ble co	ndition ()				
			<> los	ding odd	F888 ()				
				100M					
			<) con	stant ve	lues ()				
2 8 6 7 2 8 6 7 2 8 6 7		60000 CONDU CONTU PUTPE CETPE		E G N E G N E G N E G N	0 2 8 0 C N 0 2 8 0 C N 0 2 8 6 C N		CONCUT entry address. CONUL entry address. PUIPFE entry address. CETPFE entry address.		
1000		928145		EQU	10008	:	Stack pointer.		
			*****	MAIN 91					
			BOTE:						
0100	31 .01	START;	10	5 P . 3 A 1 3	N S P		Set stack pointer.		
010	_2 0C	c	LD Call	C,OCA Conodt		:	Clear acreso.		
0			*****			;			
	•								
		* *	RCIE:	This p	rocedure is i	nitiati	ng key code table.		
0108 0104	02 00 CD 288/	,	10 CALL	C , 0 Putppk		4 9 1 9	Initiate key table function. Initiate key table.		
			*****	327 821	. TABLE	••••••			
			SOTE:						
				tbia gu	rocedure is s	ettin <b>g</b>	key ende table.		
010D 010P	06 01 02 01		1 D 1 D	8,1 C,1		;	Key 1 to key 34. Normal mode.		
0111	38 01	20091:	LD BCSA	A.1 A7			Secting data 1 to 34. Save secting data.		
0114	C S		P1 28	BC			Save hey position no.		
0111	CD 1861	7	CALL	PUTPPE			Set hey code table.		
0119	71		70 P	A P			Bestore setting data.		
C_1A 0116	3C 04		1 X C 1 X C	A B			Update satting data. Update hay position no.		
0110	35 P3		7 B C b	23 C, LOOP1	I				
			•••••	D ] S P L A 1	SETTING REY	C 0 0 0			
			NOTE :	This pr and dij	rocoduro in g aglaying it.	etting	key code table data,		
0120	08 01 06 01		L D L D	C,1 8,1		1	Sormel méde. Sey i to 34.		
0124	C1	L0097:	PUSB	8C		1	Save key position no.		
0128	CD 0135	9	CALL	OSPOAT BC		1	Display bey code. Restore bey code.		
0)20	04		INC	8		;	Update key position no.		
		APP	ENDLX	Page 21	- 101				

	7.8			L D	A . B	; Luop 34 times,
0120				<b>C</b> 0	16	
0125	1 6	23			5 10083	4
0130	30	12		18	C. LUUP2	¢
						. walting for inputting any key.
0132	CD	6009		CALL	CONIN	
						i tad
0135	с 3	6903		3.8	N BCOT	, the ,
			:	******	********************************	
					DISPLAY DATA	
					*********	
			1	NOTE:		
			:		Displaying date in A reg	leter in here smage.
			-			
0118			ISPNAT			:
0130	P 5			PUSH	A.7	save data.
0130		11		5.8.1	A	
0133		1.		6.81		
8630	C 0	1.			A	P
0130		3.		381	-	•
0137		37		381	4 	. Translate upper this to ASCII
0141	6.0	0121		CALL	TRUCET	
						coowi 
0144				10	C , A	; Long progray data to c redistor.
	-					
0145	C D	6 8 0 C		CALL	CONCUT	; Display ASCII data.
						•
0148				9 G P	A 2	; Bestore data
0149	<b>26</b>	0 P		AND	078	:
0148	CD	0153		CALL	TESDAT	; Translate lower (bit to ASCI)
						code.
3 1 4 2	47			10	C , A	; Lond display data to C register.
0147	C D	280C		CALL	CONDUT	: Display ASCII data
0152	<b>C 9</b>			827		
			:			* * * = * * * * * * * * * * * * * * * *
			1		CRANSLATE BINARY DATA TO	ASCII TATA
					*********************	******
				1078 -		
			1		Amonglate bioses data to	APRIL and a
			:		itenetate othero esta (0	ASCII COGN,
			1		<pre>** entry parameter</pre>	• >
			:		a consey date,	
			:		** revuro parameter	41
					A : ASCII data.	
0123			Insur:			
0153	11	0.3		C P	GAN	; Data C OAN7
0125	30	03		38	NC,TRSDATS	; 30.
						-
0157	16	30		0 2	3 0 M	; Data e GAB
0159	C 8			921		1
015A			TESDATS	-		:
0154	5.5	09		10	С,09И	Deta >+ GAN
0 1 S C	91			ちてき	c	· · · · · · · · · · · · · · · · · · ·
0150	<b>F</b> 6	40		0 8	408	
0157	6 9			32	-	•
						<b>P</b>
				232		•

.

			AUTO PONER OFF & AUTO BACKLICH OFF			
			NOTE:	This backl	sample pr light off t	Offen sets auto power off time and auto time and turns thes off when the time is up
				unless	any key	is pressed.
			()	able co	ndition	6.3
			() long	ding edd	lres6	()
		;	. PNASE	100H		
			() cons		lues	٤ >
5 8 0 6		; CON57		<b>FO</b> 10	02806W	' CONST ADDREW Address
6087		CONIN	•	8 <b>0</b> U	NEBOSN	CONIN entry eddress
237E 2842		PONEBOP BACHLIG	Р Н Т	20U 20U	OEB7EB OEBA2U	; PONRROFT entry address; ; BACKLIGNT entry address
802)		ATSOTIN	£	ຂຸດບ	070718	
2020		TIMERO		ZGU	090208	
7579		TIMPEND Eloptin	e	200	08263N 08023N	
2387		ELOTPEN	0	8 G U	0 P 3 8 7 N	
1000		MAINSP		του	010000	; Stack pointer
		1				
				NA1N		
		1				
0100	31 1000	: T & A T Z	LO	5 P . MA 1	NSP	: Set stack pointer
		7				
0103	2 A F021		LD	KL. (A1	(301102)	Set auto power off time
0106	ED 58 F020		LO	DE.ITI	(063H	; Cat ] sec couster
0104	22 8589		700	(11921	WB),HL	; Set auto pover off time
		î.				the sure backlight off time
0108	28 1023		1.0	HL.(21	(91110)	: Get auto backlight off time
0111	ED 58 F02D			0E, (T) HI DP	n#20)	Cat 1 sec couster
0116	22 1387		10	121011	FEND),8L	; Set auto becklight off time
0119		1000:				
0119	CD 8808		CALL	CONST		; Check key in ?
0110	28 26		186	2.E41	18	; if areg . Crim ; Then key in.
			Applica which with C	needs ( Needs ( CONST.	nerts the to disable	D BLOCOND IN THIS DATE AUTO PONES OFF OF AUTO BACKLICHT OFF
0117	24 8589		LD	88.173	MERNDI	: Else not bey in. : Cet euto power off time
0122	ED 58 2020		LD	DETIT	HERO1	; Get 1 sec counter
0128	87 PD 52		58C	э К1,08		; Reset carzy (lag ; (tlyczyd)-(tlyge0)
6.13	30 05	1.2	1.11	3C, 3K1	TCHE	; ) ( then not gaver off
0122	0 8 0 0	-	L D	C.00x		; Set continue made power off
0150	CD 2372		CALL	60 - 1 8 C		; Power off
0130		EKIICHK	1.0	21 (81		· for outo backlings off time
0130	20 28 2020		1.0	00,111	STROI	Cet 1 sec counter
0137	83 E0 63		C R C	A HI DE		- Aeset carty flag
0134	30 09		1.0	NC, SEE	E 2 P	; > 0 then not becklaght off
0116	0 8 00	Ť	LD	C,00H		set backlight off
0136	CD 2842		CALL	BACNLI	C H 7	. Bechlight off
0141		SLEEP:				
0141	76		NALT JP	100P		
u144		1				
0145	CD 2808	NEY_EN:	CALL	CONIN		
0148	0E 01		10	C, 01H		
014A 0140	CD 2842 C9		RET	BACKL	1681	; DALELIGNT UN
		6 N D				
(11) SAMPLE 11. CONTINUE

						8105 CO	NTINUE S	MPIE PROCEAN
					NOTE: T	hia prod	ren deter	veines the continue ands or the
					te	start =o	de.	
					()	able con	dition	¢
			;		. 280			
					() load	ing addr	6 = 8	
			:	:	. PHASE	1008		
					() CONS	taot vel		•
£063				800T			0 2 8 0 3 N 0 2 8 0 9 N	
288A						EQU EQU	028848	
1000				AIRSP		του	01000N	
1000								
						1111 98	06 8 A M	
0100			s	TART:				
6106	31 100	- 00			LD	SP,MAIN	5 P	; Set stack pointer
0103	CD 280	c			LD CALL	C, OCB		t Clear screen
0108 0108	CD 010 51 010	1 E 1 2			LD Call	BL,CONT DSPMSC	<b>M S C</b>	: Display 'CONTINUE'
0108	02 01			•		C.01A	1 8 3	Take 1 Couch hey
0113	CD IB9	9 6			CALL	TOUCH		
0 J 1 6 0 J 3 8 0 J 1 B	CD 698 77 076 05 05	3 ] 6			10 18 Call	C,028 D2,KFT8 Touch	L K 2	; Neke 2 touch keys 'ON','OFF' key
0112	CD 880	9 9	Ĩ		CALL			Wait bey in
0123	29 JA				18	2,2NDCO	N 1 7 M	: Tee, then end : Tee, then ert flute
0126	0 E 01 PE 01				10	С,1	• •	Check touched key
012c 012z	28 04	17			JR	E, SETAT	7 3 H	If 'ON' , then jump : flam 'OFF' touched
0131	0 0			SETATT:	DEC	c		; Set Creg.
0132	7A 32 016	8 8			1 D 1 D	LREABER V'3	2+51.4	Set 'ON' tev attribute
0136	18 19 011	1 2			L D L D	A, E (Geyble	2+151.6	; Set 'OFF' key attribute
0130 0134	CD 888 19 D7	5 🔺			CALL JR	CONTINU 10071	2	Set continue flag or reset ; Rewrite key
0137				NDCONT:				
9136				1	1 9	CEGOT		1020 4001
				;				NACARCHINTI PIND CON
				5 	NOTE:			
					<pre> entr</pre>	y parame HL : to	ter paddres	s of string meaning
					() retu	PR DATER NONE	eter	e 1
					() pred	erved re Sove	glaters	$\alpha$
0142	••		i	: O S P M S C :				
0143 0144	87 C8				L D D R R E <b>T</b>	A, (HL) A 2		; Lot a math of messever ; Chack and mark ; If find OOk then return
0145	25			÷	PUSH	H 1.		Save pointer to medeage
0147	CD 28(	0 C			CALL	C , A Condut		: Set data to LTES Diepisy deta
0148	23				POP 1 N C 1 D	HL HL Aspess		; sectore puinter : Pointer updata
	10 14							

				:			
						DATA AREA	
							***********
3110				ċ	0		
0148	18 3	2 4	3 47		D 8	IBN, '2', 'CONTINUE', OGN	: Cursor off .
0152	48 5	4 4	5 4 E				
0156	55 4	5 0	0				
							CONTINUE.
0150				ĸ	EYBLK1:		
0159	04 0	2 0	2 0 1		D 8	4,34,2,1,00H,07H,00H,'EXO'	
015D	000	7 0	0 45				
0161	4 8 4	4					
				:			
0)63				×.	6 Y B L H 2 I		
0163	0)0	3 0	2 0 1		D 8	3,3,2,1,01N,07H,00N,' ON'	
0167	61 0	7 0	0 20				
6368	42.4	2					
0180	04 0	3 0	2 0 1		08	4,2,2,1,028,078,00N,'0FF'	
0171	020	7 0					
0175	46.4	6					
					END		

(12) SAMPLE 12. BARCODE

		;	•••••		••••••
				BIOS BARCODE SAM	PLE
			NOT		
		;	Note:	This program rea	da the date from the barcode reader
				and displays the	data.
		4	() ASSO	while condition $\leftrightarrow$	
		:	. 280		
			() 1 and	(an address ()	
				TUN GOOLEEP ()	
			. PNASE	1008	
		1	() cons	cant values co	
£803		NBCOT		EGN OEBO3H	
8806		CONST		20U 07805H	
2800		CONCUT		EQU ORBOCH	
EBT1 PRAD		BARCODE		EQU OEB72M EQU OEBADM	
8896		TOUCH		EQU OFRS6N	
889C		XEYEN		IGN OFBACH	
1000		MAINSP		EGN 01000N	
				MAIN PROGRAM	
					***************************************
0100		START:			
0100	31 1000 31 1000		10	SP, MAINS2 C, 18H	; Set the stack pointer
0 0 5	CD 891C		CALL	CONDUT	CONOUT ESC + 12 CURMON SPP
0 14	CD 880C		CALL	CONQUT	• •
0.1 0.D	18 02	-	10	C. 02H	: Read dia SW of EBT-10.10/2
u 12	CD 2872		CALL	BEADSE	
. 112	E6 50 87		A N D O B	100000008	; theck the bit 7 ; if bit 7 • 1 theo the machine is
	-			N	ENT-10/2
0119	20 UA	:		SZ ( BEI DUS	' lost to the for-lost control
0117	0 8 0 1	TCHRET:	1.0	r 01H	; EBT-10 Touch key coutine : Nobe 1 rouch key black
0119	11 0103		10	OZ, KEYBLK	; Set the hey block descriptor address
6110	CD 2896 15 17		CALL JB	TOUCH	; Display the key block ;
0					
0123	26 03	XXXEOD:	LD	8,03R	; int-lu/2 ; Change the heyboard to the Algoabet mode
0123	CD 299C		CALL	SETIN C. INF	· Set corner
0128	CD 280C		CALL	COPOCT	; fit
0128	02 30 CD 280C		LD CAIL	C. ' + ' CONOL'T	• •
0120	02 23		10	C, 4+0178	•
0132	CD EBOC		LO	C,1+01P8	
0137	CD 280C		CALL	COROUT	; ]
0135	CD 0187		CALL	0897186	· · · · · · · · · · · · · · · · · · ·
a f		BAB1:			:
0140	CD 0175		CALL	BAROPN	Open the bercode teader
0143	0 2 0 3	5 A 8 4 .	1.0	C.038	; Read the parcode status
	13 ERSD		CALL	SARIDIE A R	: · Chara Steat.
0149	21		0.8	c	; BC to the number of data
014A	C4 0157		CALL	NZ, DSPOAT	; if any data exists, then display it
0140	6081 CD		CALL	CONST	; Check key in
0141	25 FO		35	A 2,8AR2	; II areg. = 0 ; Then not key in
013-	CO. 5800	1	C 4 I 1	CON1N	· Cot a kay in data
0157	7£ 00		CP	Q D N	11 code is CON
0158	20 89		J R C A 1 2	NZ, BARZ BARCIS	; Then END • Close the barcode reader
0154	CO 2803		CALL	¥8001	
		1			
0160	( 12 65 13	ENDHSC:			
0160	73 20 52 45			-thes wrings (0	
0168	54 55 52 4E 20 74 6V 20				
018C 0170	65 78 68 74				

0174 00

i i ÷ -----OPEN THE BARCODE READEB ÷ ; NOTE: This subrouting opens the barcode reader. -÷ CODE TYPE : JAN / EAN / UPC-A / EPC-E 0.071.03 SUZZER ON DELIMITER ON LED CYRL ZERO ADD O N OFP (only UPC-E) ÷ c) entry perameter () i ...... ÷ c) saturn parameter () none <) preserved registers () 1 ..... 0175 0175 0177 0179 0178 BAROPN: 0 2 0 0 3 £ 0 1 1 6 0 0 1 2 0 D : Code type JAN / EAN / UPC-A / UPC-E Set Option ; Delimiter is ODM 1 D с.сси A,01H D,00H F,0DH L D 1.0 10 017D 018C CD 2880 C9 CALL BARCODE ; Open 821 ; -----XOTE: i This subroutine closes the barcode reader. . () entry parameter () : none ;;;; () voturn persenter () none () preserved registers () 1 .... BABCISI 0161 ; Close function tweet; Close 01 01 CD 2250 15 C , C 1 9 0111 0193 C.N11 927 BARCODE 0156 69 ÷ 1 ATAT DABA INT YALAZIC SOTE: This subroutine reade the date from the budder, and displays the data on the 100. () entry parameter () none () resurs parameter () none o preserved registers <> ALL 0187 DSPDAT: A.F. ; Save registers 15 PUSH 8 C 0 E С \$ 8 \$ 0198 4 CD 280C PUSH LD 81. C,088 018A Set cursor (1,1) 0188 ; CALL CONOUT Number of chozactors to display 8.20 0190 06 14 10 LOOP : 0192 ÷ APPENDIX Page 21 - 107

0192	0 E 0 2 0 S		L D D E C	C,03N B	; Read function number ;
0195			PUSH	80	; Save counter Breg
0136			CALL	BABCOUE	; Bend the data
Q 1 9 A	39 16		18	2. 280052	· If 2P + 3 then no deta in the butter
U19C	FE OO		69	000	: Check the delimiter
019E	29 08		3.18	2,05959	
0140	4 F		LO	С,А	; Display data
0141	C S		PUSN	8 C	; Save counter
0142	CD EBOC		CALL	CONCUT	1
0145			POP	BC	
0140	13 EA			LOUP	i
0148		n5858:			A
0146	QE 20		1.0	C.20H	Display ' ' to clear old dote
0144	c s		PUSN	80	Save counter
0148	CD E80C		CALL	CONOUT	
3210	C 1		POP	8 C	Bestore Breg.
014 5	0 5		0 E C	8	
0130	20 26		J R	NZ, DSPSP	
		1			
0182		ENDDSP:			
0182			POP	NL	; Reatore registers
0103	01		POP	02	
0184	C 1		000	80	
0185			POP	87	
0180	63				i
				DISPLAY STRI	N C
					************************************
		;	NOTE:		
				Oisplay stri	ng data to the CON; until find OON.
		;			
		-	() ()	ry persector ·	)
		-		NT - 7	
				AL - 109 800	cash or arryul ande.
		:	() PET	urn parameter	()
		;		none	
		•			
			<) gre	seeved cedicts	P ()
		:			
0187		DSPHSC:			
0187	72		LD	A, (81)	; Cat data
0198	87		0 9	A	Check terminater
0189	C 8		2 2 2	2	; If find termineter them return
		:			
012.5	<b>E</b> 5		PUSN	NL	
0188	3 9		10	C, A	;
0180	C3 280C		CALL	CONCUT	; Display data
0130	El		POP	B T	
n1c0	21		250	NI	; Cpdate polater
0101	13 74		16	026220	
		;			
				DATA OF BET	BLOCK DESCRIPTOR
		:		************	**********************************
		\$			
01C3		REVELS:			
0303	04 08 02 01		2.5	04,14,2,1,00	N, 07N, 30, 11ND 1
0107	00 07 00 4S				
U ] C B	4 E - 4 4				
		i T K D			
		x 3 W			

			******			-
				1 2018 ••••••••	CAN SAMPI	
		+	NOTE:			
				This This	sample p	rogram is executed only on EHT-10.
		:		lf sa	ee named	program already exists in BAN disk.
		:		then a	ld file :	is eraned.
			228 61	estis co	ndition	4.5
		•	. 280			
		• •	() 104	ding add	8896	()
			. P N A S E	1008		
			() CON	stant va	lues	¢ 3
E 8 8 3		N8007		EQU	0 E 8 O 3 N	
2809		CONIN		E G U E G U	028098	
2880		10401		200	028908	
8886		10008		EQU	02896N	
EBAS		THEORS		290	0 Z B A 5 8	
0005				200	000058	
1000		SAINSP		tgu	a Tana M	
		- - -		4 RIAF		
0100		; 57487;				
0100	31 1000	:	LD	5 P . NA I	NSP	Set stack pointer
0103	02 18		LD	C.188		Clear screen
0105	CB 280C		CALL	CONCUT		ESC + '2'
0104	CD 280C		CALL	CONCUT		1
0100	21 0293		1.0	8 L . S 2 S	DMSC	Display 'SEND'
0110	CD 0101		CALL	DSPNSC		1
	<b>6e</b> 60	1		C 0.9 M		Car address of TCATAERA
0115	C2 2845		CALL	IPPORT		HL = TCATABEA address
0118	22 0289		LD	(TCANA	821,81	SAVE TEAMABES Address
		:				
0118	22 01		10			Receive tile
0117	CD 2390		CALL	TCAN		Connect
0122	39 14		3.8	C, TCAH	ERB	If CT=1 then error
	C.D. 016D		<b>C</b> • • • •			t Receive file to RAM disk
0121	38 OT 55		38	C. TCAH	288	If CT-1 then error accured is ACVFILE
		;				
0129	32 04		LD	A,048		Function disconnect Disconnect
0120	36 03		78	C, TCAN	288	If CTal then ertor
		;				
0130	21 0272			M1.END	IN S G	Disblek (Ituissed)
0135	15 17		38	ENDPRO		
		1				
0139		TCAMEBB	:			· (ANA ATTE 2568
0138			10	C.0C8		: Clear Screan
0138	CD 280C		CALL	CONOUT		
0131	F 1		POP	A P		; Sestore effor code
0138	21 4100		LD	HL.288	TPL	Set error table address
0142	5 F		10	t,Å		; Cot error code
0143	16 00		LD	D,00M		; Drog = 0 · shife foft fit s 2
0145	CB 23		SLA	2 31 0 P		, MARE GLEDI DESERE ACCURE
0167	26		10	E. ML		Save errer message address
0118	2.3		1 8 6	HL.		1
014A	58		10	D, ( H L )		
0148	68		2 X	D2,HL		: Divujo A BOUMANO ! NT 77 BOUNT 10 ALLOL MEBBAGO
0140	CO 0101		LALL	USPHAL	•	l erobret eros.e.
0147		ENOPRO:				
014 P	08 01		LD	C , 0 3 H		; Nake ] touch key block - cat hav bluck
0151	11 0248		LO	108,X21	COL N	: Make 'Press' koy
0154	CD 2009		CALL	CONIN		Walt for hey in
0154	C3 2803		1.5	NBOOT		; End progress

ACCENT SATA & MART TILE           MOTE:           Note           C: :::::::::::::::::::::::::::::::::::			:			
Access and write it to file.           Note:           Access and write it to file.           Access and write it to file. <td></td> <td></td> <td></td> <td></td> <td></td> <td>***********************************</td>						***********************************
Holds:         Holds:           10         Holds:           10         Holds:           10         Holds:           10         Holds:           11         Holds:           12         Holds:           13         Holds:           14         Holds:           15         Holds:           16         Holds:           17         Holds:           18         Hold:           18         Hold:           18         Hold:           18         Hold:           18			•		RECEIVE DATA &	MAKE FILD
Hoft:         Hoft:						
Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and write it to file.           Image: Second data and file.           Image: Second data and file.           Image: Second data and file.           Image: Second data and file.           Image: Second data and file.           Image: Second data and file.           Image: Second data and file.           Image: Second data and file.           Image: Second data and file. <thimage: and="" data="" file.<="" second="" th=""></thimage:>				NOTE:		
<ul> <li></li></ul>			;		Receive data and	d write it to file.
<ul> <li></li></ul>			:			
1000         1000 <th< td=""><td></td><td></td><td></td><td><pre>4) ente</pre></td><td>y parameter</td><td>()</td></th<>				<pre>4) ente</pre>	y parameter	()
$\begin{array}{cccccccccccccccccccccccccccccccccccc$					NONE	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $			1	() retu	CY = 1 · error	c) nerwrad in thie newline
CC + 0 : no second i preserved relations         CC + 0 : no second not relations           0160         C6 0101         CALL         BL BC YSIC LD         D Bablay 'Breaving' Second DSP SEC           0163         22 0188         LD         LD         Set 4: DSP SEC           0163         22 0188         LD         Set 4: DSP SEC         Set 4: DSP SEC           0164         22 0188         LD         Set 4: DSP SEC         Set 4: DSP SEC           0164         22 0188         LD         Set 4: DSP SEC         Set 4: DSP SEC           0164         10 0008         LD         Set 4: DSP SEC         Set 4: DSP SEC           0164         10 286         LD         Set 7: DSP SEC         Transat file sec 4: DSP SEC           0174         22 028         LD         Set 7: DSP SEC         Transat file sec 4: DSP SEC           0174         22 028         LD         CLN         File Sec 4: DSP SEC         Set 7: DSP SEC           0174         02 28         LD         CLN         Processor         Set 7: DSP SEC           0174         02 28         LD         CLN         Processor         Set 7: DSP SEC           0174         02 28         LD         CLN         DSP SEC         Set 7: DSP SEC <t< td=""><td></td><td></td><td>:</td><td></td><td>A : 622</td><td>or code</td></t<>			:		A : 622	or code
1:0         Descent of training of traini of traini of training of traini of training of training of train					CY = 0 : no erre	of
IDD         IDD <td></td> <td></td> <td>;</td> <td>() pres</td> <td>erved registers</td> <td>4.3</td>			;	() pres	erved registers	4.3
1150         2         3         2         10         2         10         2         3         5         4         1         5         4         1 <td></td> <td></td> <td></td> <td></td> <td>NONE</td> <td></td>					NONE	
100         21         000	4150		i Bevelle			
0         CO         0.0101         CALL         DSPRC         Control           0163         32         01         CALL         DSPRC         Control           0163         32         01         CALL         DSPRC         Set A::           0164         32         0238         1         Control         Set A::           0164         1.0238         1         Control         Set A::         Control           0164         1.0238         1         Control         Set A::         Control           0164         1.0238         1         Control         Set A::         Control           0174         32         Control         Control         File         Control         File           0174         61         Control         Control         File         File         Control           0174         61         1         Control         Control         Set Char         Control           0174         Control         Control         Control         Set Char         Control         Set Char           0174         Control         Control         Control         Set Char         Set Char           0177         Contre	015D	23 0298	activite	1.0	HL. BCV956	: Diaplay 'Recovering'
0163         32         01         10         4.01H         5.01 A:           0160         04000         00000         0         0.0000         0         0.0000         0         0.0000         0         0.0000         0         0.0000         0         0.0000         0         0.0000         0         0.0000         0         0.0000         0         0.0000         0         0         0.0000         0         0         0.0000         0         0         0.0000         0         0         0.0000         0	0160	CD 01D1		CALL	DSPHSG	i contration in the second sec
0163         32         01         10         4,01M         5 Set 4::           0165         32         0000         10         0000         Set 4::         Set 4::           0160         24         0000         10         0000         Set 4::         Set 4::           0160         24         0000         00::         Set 4::         Set 4::         Set 4::           0160         24         0000         00::         Set 4::         Set 4::         Set 4::           0160         25         00::         Set 7::         Set 4::         Set 7::         Set 7::           0174         02::         10         17 Contat         Set 7::         Set 7::         Set 7::           0174         02::         10         10::         10::         Set 7::         Set 7::           0174         02::         14         20::         Set 7::         Set 7::         Set 7::           0174         12::         10::         10::         C:::         Set 7::         Set 7::           0174         12::         10::         C:::         Set 7::         Set 7::         Set 7::           0174         2::         0::         Set 7::						
0165       01       00       10       17       17       10       17       10       17       10       <	0163	38 01		T 0	A_01H	; Set A:
0180       01       0008       10       01.0018       10       01.11       11       1000000000000000000000000000000000000	Q 1 6 5	35 0988		LO	[FC8],A	; Set drive code
110         24         0.0000         000000         0000000         00000000         00000000         00000000         00000000         00000000         00000000         00000000         0000000000000         0000000000000000000000000         000000000000000000000000000000000000	0148	01 0008		1.0	RC 11	I Number of fileness character
Disc         Disc <thdisc< th="">         Disc         Disc         <thd< td=""><td>0188</td><td>24 0389</td><td></td><td>LD</td><td>RL. (TCAMADR)</td><td>Received (i)s mans area</td></thd<></thdisc<>	0188	24 0389		LD	RL. (TCAMADR)	Received (i)s mans area
173         20         80         LD19         Present file pase           173         47         10         4.0         4.0           174         32         0227         10         (FCH+17),A         3.0           0177         32         0200         10         (FCH+12),A         3.0         5.00           0177         02000         10         (FCH+12),A         5.00         5.00         5.00           0177         02000         10         (FCH+12),A         5.00         5.00         5.00           0177         10         02000         1.0         (FCH+12),A         5.00         5.00           0177         10         02000         1.0         0.10         5.00         5.00         5.00           117         10         02000         1.0         0.118         7.00         5.00         5.00         5.00           118         20         0.0         0.0         7.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         5.00         <	0162	11 02BC		1.0	DE. TCB+1	FCB file DAMA Area
1174       32       0277       100       17       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       177       100       100       177       100       100       177       100       100       177       100	0171			LDIB		Transmit file base
1174       32       0.02.7       32       0.02.7       32       0.02.7       32       0.02.7       32       0.02.7       32       0.02.7       32       0.02.7       32       0.02.7       0.0	0173	AF		TOR	<b>A</b>	. <b>▲ - 0</b>
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0174	32 0207		10	(PCB+121,A	
0174         02 14 0177         02 04 0005         16 0 0005         16 0 0005         16 0 0005         17 0 0005         18 0 0005         Punction number Set DFA address           0127         00 0005         11 0 0005         10 0 0005         10 0 0005         11 0 0005	0177	71 012=		1.0	[ [ [ ] ] ] ] ] ]	Set 'Cr' U
11         02220         11         02220         11         02230         11         030         11         030         11         030         11         11         030         11         11         030         11 <th11< th=""> <th11< th=""> <th11< th=""></th11<></th11<></th11<>	0174	46 30		1.0	C 14H	Fuertion cumber
0137         CD 0005         CALL         BDOS         Set DAA address           0122         00 11         10         C,118         Function number           137         CD 0005         LJ         DC, CALL         BDOS         File directory search           137         CD 0005         CALL         BDOS         File directory search           137         CD 0005         CALL         BDOS         File directory search           138         28         CAL         BDOS         File directory search           0138         CAL         BDOS         Sec 7CB         Sec 7CB           0180         DE         DE         CALL         BDOS         Sec 7CB           0192         CD         CAL         BDOS         Sec 7CB         Sec 7CB           0180         DE         CAL         BDOS         Nake         Nake           0192         DE         CAL         BE         Sec 7CB         Sec 7CB	0170	11 0220		LD	DE DHA	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0177	CD 0005		CALL	BDOS	Set DHA address
0122       08       11       10       C, 118       Function number         0174       11       0288       Lu       DE, FCB       St FC3 address         137       C0       0069       CALL       BDOS       Flind       directory search         0138       Z6       03       JS       Z,ZSDDEL       No. Then oot file delets         0138       Z6       04       JS       Z,ZSDDEL       No. Then oot file delets         0139       C1       0288       LD       DF, CCB       Set FC3 address         0139       C1       0288       LD       DF, CCB       Set FC3 address         0139       C1       0288       LD       DF, CCB       Set FC3 address         0139       C1       LD       DF, CCB       Set FC3 address       Set FC3 address         0130       C2       C4       JS       Set FC3 address       Set FC3 address         0140       JC       Nake       Nake       Nake       Set FC3 address         0140       JC       JS       CALL       DBOS       Address       Set FC3 address         0141       JC       A       DS       Set FC3 address       Set FC3 address       Set FC3 address			7	+		
01:4       11       02.7CB       Set PCB address         01:87       CD       00.005       TNC       A       Final/T         01:87       CD       00.005       TNC       A       Final/T         01:87       CD       00.005       TNC       A       Final/T         01:80       02.02       00.005       TNC       A       Final/T         01:80       02.13       10       02.7CB       Set FCB address         01:80       02.16       TNDDZL:       D       Set FCB address         01:80       02.16       TNDDZL:       D       Set FCB address         01:81       02.80       CALL       BODS       Set FCB address         01:82       CD       00.005       CALL       BODS       Set FCB address         01:83       CD       00.7CB       Set FCB address       Set FCB address         01:84       CD       CALL       BODS       Set FCB address       Set FCB address         01:85       02.16       D       P.FCB       Set FCB address       Set FCB address         01:85       02.16       D       C.11.8       BODS       Set FCB address       Set FCB address         01:85       0	0192	08 11		LD	C,118	Functico number
137       36       1003       Find a fractory function         0133       28       88       38       2,2500EL       No. Then not file delete         0180       02       13       10       C,138       Delete file         0187       10       02.38       10       07.708       Delete file         0187       11       0288       CALL       BDOS       Delete file         0186       028       CALL       BDOS       Delete file       Set fCB address         0180       02       0405       10       07.708       Nake       Nake         0180       02       14       13       Nake       Nake       Nake         0181       12       04       38       X2,STARTN       Yes       Tas contious         0181       22       10       A,008       Function receive       10         0183       16       10       A,038       Function receive	0124			LA	DE. 1C8	Set FC3 address
0138         20         08         38         7_FENDEL         No. Then not file delete           0138         00         10         02.200         50         50         70           0197         10         02.80         10         02.200         50         50         70           0197         02.0005         C.1138         10         02.200         50         70           0195         02.16         10         02.700         50         50         70           0197         10         0200         00         70         70         70           0190         02         10         02.700         70         70         70           0190         10         02.700         10         70         70         70           0141         12         09         10         10         100         70         70           0141         12         09         10         10         100         100         70           0144         02         22         10         10         100         100         100         100         100         100           0144         02         22	_187	36		INC	8103 A	Plady
0180       02 13       10       07.35       10       07.758       Delete file         0180       10       07.758       Set fCB address       Delete         0191       02.86       CALL       BDOS       Delete         0192       CD 0005       CALL       BDOS       Delete         0193       02.16       TNDDZL:       DELETES       Set fCB address         0194       CD 0005       CALL       BDOS       Take         0195       02.00       C.168       Nake new file         0197       10.0280       CALL       BDOS       Take         0190       10       C.168       Nake new file       Take         0197       20.04       JB       N2.5TARTN       Test. Theo contious         0143       12.09       A.008       Set error code       Set error code         0144       02.22       CALL       CONVCY       ID       A.008       Set buffer address         0144       02.280       CALL       CONVCY       ID       A.038       Punction Proceive         0148       02.280       CALL       CASA       Reveive       ID       A.038         0183       10       A.038       Pu	0158	28 88		38	Z.ESDDEL	No. Then not file delete
0 0 0 1 3 0187       1 0 2 2 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0100		1			
0187       13       0280       10       08.7CB       Set FCB address         0192       CD       0005       CALL       BBOS       Delate         0195       02       14       10       07.7CB       Set FCB address         0195       02       14       10       07.7CB       Set FCB address         0196       02       10       07.7CB       Set FCB address         0197       11       0288       CALL       BD03       Take         0190       12       04       JB       N2.5TARTW       Yee. Theo contioue         0197       20       04       JB       N2.5TARTW       Set carry flag for error         0144       03       10       A.098       Set error code       10         0144       02       22       10       A.098       Set error code         0144       02       22       10       A.098       Set error code         0144       02       22       10       A.098       Set error code         0144       02       21       10       A.098       Set error code         0144       02       10       A.038       Set prici receive         0144	0180	02 13		LD	С.13Я	; Delete file
0195       CALL       DBOS       Delate         0195       OE       14       DES       Delate         0197       11       0288       DD       DF, FCB       Sat FCB address         0190       3C       DD       DF, FCB       Sat FCB address         0190       3C       DDS       Make       Sat FCB address         0190       3C       DDS       Make       Sat FCB address         0190       3C       JE       NC A       Make       Make success ?         0190       3C       JE       N.C.A       Make       Sat FCB address         0141       3E       OS       JE       No CA       Make       Sat FCB address         0141       3E       OS       JE       No OS       Sat error Code       Sat FCB address         0144       OF 2P       DD       A, 098       Sat error Code       Sat error code         0144       OF 2P       DD       A, 098       Sat error Code       Sat error code         0144       OF 2P       DD       A, 038       ; Function receive       Sat error scourse         0145       21       OF 800       CALL       TCAN       ; Faroive       Sat error scourse	0185	11 0288		10	02.208	Set FCB address
0195       02 16       LB       C,168       Nake new file         0195       02 0005       CALL       BD05       Nake       Nake         0190       3C       INC       A       Nake       Nake         0190       3C       INC       A       Nake       Nake         0190       3C       INC       A       Nake       Nake         0140       3C       JB       N2,STARTW       Tes. Theo contious         0140       3T       SCF       Set carry flag for error         0143       C6       D9       ID       A,098         0144       02 22       ID       A,098       Set error code         0145       02 03       ID       A,038       Punction receive         0146       03       ID       A,038       Punction receive         0147       10       Set B90       Set buffer addree	0192	[] 6663		CALL	9607 -	. Delete
0195       02 16       10       10       02.168       Nake new file         1197       11 0280       10       02.768       Set FCB address         0190       3C       1NC       A       Make success ?         0192       20       64       38       N2.51ARTW       Yes. Theo continue         0140       37       SCF       1D       A.098       Set carry flag for error         0141       32       09       1D       A.098       Set error code         0141       32       09       1D       A.098       Set error code         0144       02       22       014       STARTW       Yes. Theo continue         0144       02       22       014       Starty flag for error         0144       02       22       014       Starty flag         0144       02       28       1D       A.098         0145       32       038       1D       A.098         0145       22       1D       A.038       Punction receive         0145       02       1D       A.1038       Punction receive         0145       02       1D       A.13       Check 85 tee.         0181<	4185		TNDOTL			
197       11       0288       LD       07.9C8       Sat FCB address         0194       CD       0005       CALL       BD05       Nate         0190       32       1NC       A       Hake success ?         0190       37       SCP       Yes. Theo contions         0141       32       05       LD       A,098         0143       C6       BTARTN:       Set error code         0144       STARTN:       ID       A,098         0144       C2       ID       A,098         0144       STARTN:       ID       Call CONOUT         0144       C2       ID       A,038         0144       C2       ID       A,038         0144       C2       Call CONOUT       ID         0145       32       C210       ID       HLDRAL         0148       C1       C210       ID       HLDRAL       Sat buffer Address         0147       38       10       OB       Call       TCAN       Sat buffer Address         0148       21       C210       ID       A,138       Check 8C res.       Call         0181       28       OB       JB       JB	1195	02 16		10	C,168	1 Jake new file
019A     CD 0005     CALL     BD05     Nake       019D     3C     INC     A     Hake success ?       019F     20 04     JE     N2,STARTW     Yee. Theo continue       01A0     37     SCP     Set carry flag for error       01A1     3E 09     LD     A.09H     Set carry flag for error       01A4     CP     BTARTW:     ID     C.*.'     Dieplay *.* by 12S received       01A4     CP 22     ID     A.09H     Set error code       01A4     CP 22     ID     C.*.'     Dieplay *.* by 12S received       01A4     CP 280C     CALL CONOUTY     I     Dieplay *.* by 12S received       01A5     CD 280C     LD     A.03H     : Function receive       01A6     CD 280C     CALL CONOUTY     :     Set buffer address       01A5     CD 280C     CALL CONOUTY     :     Set buffer address       01A7     CD 280     CALL CONOUTY     :     Set buffer address       01A8     CD 21020     LD A.13     :     Function receive       01A8     CD 21020     LD C.15H     :     function sequential crite       0181     28     DB     JR 2.CLSFL     :     then receive end       0182     S1     DD C.15H	-197	11 0288		LD	02,768	Set FCB address
0100       32       iNC       A       Make success ?         0132       20       64       38       N2,STARTW       Yes. Theo continue         0140       37       SCP       Set carry flag for error         0141       32       09       LD       A,098       Set error code         0144       02       22       LD       A,098       Set error code         0144       02       22       LD       Call       CONOUT       Set error code         0144       02       22       LD       Call       CONOUT       Set error code         0144       02       28       Gall       Conout       Set error code       Set error code         0145       32       03       LD       A,038       Punction receive         0145       32       03       LD       A,038       Punction receive         0145       32       03       LD       A,038       Punction receive         0145       32       03       B27       C       If C'+1 then error cocured         0181       38       08       C       If Scregio       If C'+1 then error cocured         0182       31       08       CLST       If	019A	CD 0005		CALL	2005	3 3aké
0.157     2.0     0.4     JE     X2,STARTS     Yes. Then contains       01A0     37     SCP     Set carry flag for erior       01A1     32     09     LD     A,098     Set carry flag for erior       01A4     02     22     LD     A,098     Set carry flag for erior       01A4     02     22     LD     C.1.1     Display     by 125 received       01A4     02     22     CALL     COMOUT     I       01A4     02     20     LD     A,038     Punction receive       01A5     36     03     LD     A,038     Punction receive       01A6     02     200     LD     ML,DNA     Sat buffer address       01A6     28     CALL     COMOUT     I     I       01A6     02     280     CALL     COMOUT     I       01A8     21     02     D     LD     ML,DNA     Sat buffer address       01A8     21     02     D     NH,DNA     Sat buffer address       01A8     21     02     D     A,A     IC     Cast       01A8     21     02     D     A,A     IC     For eine       0181     28     D     D     Ca	0180	30		INC		Hake success 1
C1A0       JP       JCP       J	0192	20 04		18	NZ, STARTS	Yes. Then continue for energy flog for dr"Of
01A1       CF       01A1       01C	\$140	38 09		10	A.09H	Set error code
0144       02 22       10       C,'       Dieplay       by 125 received         0146       CD EBOC       CALL       CONOUT       1         0148       CD EBOC       CALL       TCLN       1         0148       CD EBOO       CALL       TCLN       1         0148       CD EBOO       CALL       TCLN       1         0148       CD EBOO       CALL       TCLN       1         0149       CD EBOO       CALL       TCLN       1         0140       CD EBOO       CALL       TCLN       1         0141       CD EBOO       CALL       TCLN       1         0141       CD EBOO       CALL       TCLN       1       1         0141       CD EBOO       CALL       TCLN       1       1       1         0141       CD EBOO       CLN       CLN       1       1       1       1         0161       DBO       DBO       CLN       1	0141	<b>C</b> 9		718		
0144 0144 0146 0146 0146 0146 0146 0146	01-0		;			
01A4       02 22       10       C_1.'       Display * by its recorded         01A6       CD EBOC       CALL       CONOUT       1         01A8       C1 0280       LD       A,038       ; Punction Peceive         01A8       C1 0280       LD       ML,DTA       ; Set buffer address         01A8       C1 0280       CALL       TCAN       ; Reveive         01B1       DB       BET       C       ; If CY+1 then error incured         01B2       T9       DB       A,B       ; Check 85 zee.         01B3       B1       OB       C       ; If Screg+0         01B4       25 OB       JR       JR       ; CLSFL       ; then receive end         01B6       OF 15       LD       C.15H       ; Function sequential write         01B6       OF 15       LD       C.15H       ; Function sequential write         01B6       OF 15       OB       CALL       BDOS       ; Write 1 sector         01B7       T3 92       ZS       ZS       Z, STABTW       ; L00p         01C1       r5       CLSFL:       PUSR       AP       ; If Areg(>0 then error         01C2       OT 10       LD       C.10N       ; Punc	0144		STARTV;			
01A6     CALL     CONCT     1       01A8     21     0210     LD     NL.DNA     Sat buffer Address       01A8     21     0210     LD     NL.DNA     Sat buffer Address       01A8     CD 8890     CALL     TCDN     Reveive       01B1     D8     CALL     TCDN     Reveive       0192     79     LD     A.8     Check 80 200.       0183     B1     OB     C     If 80reg+0       0184     25     OB     JR     Z.CLSFL     then receive end       0186     Of 19     LD     C.158     Function sequential write       0186     Of 19     CO     OB     CALL     BDOS     Write 1 sector       0187     C3     CALL     BDOS     Write 1 sector     Call       0186     OF 2003     CALL     BDOS     Write 1 sector       0187     C3     C     Sat STARTW     Loop       0181     CO     DA     Sat STARTW     Loop <t< td=""><td>0144</td><td>02 22</td><td></td><td>LD</td><td>C</td><td>Dieplay by 115 received</td></t<>	0144	02 22		LD	C	Dieplay by 115 received
61AS       3£ 03       LD       A,038       ; Punction receive         01AB       21 0280       LD       HL,DTA       ; Set buffer address         01AE       CD E890       CALL       TCAN       ; Reveive         01B1       D8       CALL       TCAN       ; Reveive         01B1       D8       BET       C       if CY=1 then error incoured         01B3       B1       OB       C       if Streg+0         01B4       25 0B       JE       LD       C,158       ; function sequential crite         01B6       0F 19       LD       C,158       ; function sequential crite         01B6       0F 19       LD       C,158       ; function sequential crite         01B6       0F 19       LD       C,158       ; function sequential crite         01B6       0F 19       LD       C,158       ; function sequential crite         01B6       0F 10 2288       LD       D2 Pf 28       ; function sequential crite         01B1       0288       LD       C,158       ; function sequential crite         01B2       37       O8       A       ; Check Ace8.         01B2       10       C3       Z, STA875       ; if Aregro0 then er	0116			CALL	EDNOCY	
01a8       21 0280       LD       NL,DTA       ; Set buffer address         01a8       20 0890       CALL       TCIN       ; Set buffer address         01a8       20 0890       CALL       TCIN       ; Reveive         01a1       28       BET       c       ; If CY=1 then error iscured         0132       79       LD       A,B       ; Check 8C ises.         0183       81       OB       C       ; If BCreg+0         0184       25 0B       JR       2,CLSFL       ; then receive end         0186       0f 19       LD       C.15M       ; Function sequential write         0188       11 0288       LD       C.15M       ; Function sequential write         0188       02 003       CALL       BDOS       ; Write 1 sector         0188       0303       CALL       BDOS       ; Write 1 sector         0188       0303       CALL       BDOS       ; Loop         0188       10 288       LD       2.5 X.STABTW       : Loop         0189       10       CLSFL:       PUSR       PUSR       Punction file close         0101       FS       ID       C.10N       : Punction file close	8 1 A C	36 03		LD	A,038	: Punction receive
Clar       CD 1890       CALL       TCIN       Reveive         0181       08       BET       C       11f CY+1 then error scoured         0181       08       1D       A.8       Check 87 see.         0182       19       1D       A.8       Check 87 see.         0183       08       C       11f BCreg+0         0184       25       08       18       2.CLSFL       then receive end         0186       0f 19       10       C.15H       Function sequential write         0186       0f 19       10       C.15H       Function sequential write         0186       0f 19       10       Df.FC8       Set FC8 address         0188       CD       08       Check Are6.       Clarf         0188       CD       08       Check Are6.       Clarf         0189       13       028       Check Are6.       Clarf         0181       CD       25       2.STABTW       Loop         0161       FS       0162       10       Clarf         0161       FS       10       Clarf       Function file clare         0161       FS       0162       10       Clarf       Function file c	0143	21 0280		LD	HL, DILA	Set buffer Address
0181       98       BE7       C       : If CY-1 then error inclured         0182       79       LD       A.B       : Check 8C ree.         0183       81       OB       C       : If StregtO         0184       25       0B       JB       2.CLSFL       : If StregtO         0186       01       10       2.S       0B       : If StregtO         0186       01       10       0288       JD       C.SFL       : then receive end         0186       01       0288       LD       C.ISH       : Function sequential write         0186       01       0288       LD       Df.FC8       : Set FC8 addrese         0188       CD       0405       CALL       8DOS       : Write 1 sector         0188       27       08       A       : Check Areg.       : isop         0188       07         : LO          0188       07         : LO          0188         : LO        : LO         0187         : CLSFL:        : LO         0101	0148	CD 2890		CALL	1013	Reveive
0132       7.9       1.0       A.8       ; Check 8.5 200.         0183       81       08       C       ; If BCred:0         0184       25       08       38       ; Check 8.5 200.         0184       25       08       38       ; Check 8.5 200.         0184       25       08       38       ; Check 8.5 200.         0184       25       08       38       ; function sequential write         0186       01       0288       1.0       DC.158       ; function sequential write         0186       01       0288       1.0       DC.768       ; Set FC8 addresse         0188       CD       0005       CALL       8D05       ; Write 1 secter         0188       CD       0005       CALL       8D05       ; Write 1 secter         0188       CD       08       A       ; Check Ares       ;         0182       27       25       2; STASTW       ; Loop         0197       15       2; STL       ; Strate       ; Loop         0101       75       PUSR       AP       ; If Areg(>0 then error         0102       10       LD       C, 10M       ; Punction file filese	0181	28		327	C	; If CY+1 then error secured
0183     31     008     C     11 DEregio       0184     35 0B     38     2,CLSFL     ; then receive end       0186     01 IS     LD     C.15B     ; function sequential crite       0186     01 IS     LD     C.15B     ; function sequential crite       0186     01 IS     LD     C.15B     ; function sequential crite       0186     01 IS     LD     C.15B     ; function sequential crite       0188     11 0288     LD     DE.FC8     ; Set FC8 address       0188     CD     0005     CALL     BDOS     ; Frite 1 secter       0188     37     O8     A     ; Check Ares       0187     23 22     25     2; STARTS     ; Loop       0101     F5     PUSR     AF     ; If Areg(:)O then error       0102     10     LD     C.10M     ; Punction file filse	0132	79		10	A , 3	Check BC 109.
0184     43     00     14     1,000     100       0186     01     10     0,000     100     100     100       0188     11     0288     100     020     100     100       0188     11     0288     100     020     100     100       0188     0100     041     8005     100     100       0188     0005     041     8005     100     1000       0188     0100     010     100     100     100       0101     10     10     010     100     100	0183	25 03		18	2 01471	i i peregro - then receive end
0186         0E         19         LD         C,158         : Function sequential crite           0186         11         0286         LD         DE,FE8         Set FE8 address           0188         CD         0055         CALL         BDOS         ; Vinction sequential crite           0188         CD         0055         CALL         BDOS         ; Check         Set FE8 address           0188         37         OR         ; Check         ; Check         Set FE8         Set FE8           0182         37         OR         ; Check         ; Check         Set FE8         Set FE8           0182         37         CE         SET SES         ; STABTE         ; L007           0161         FS         OUSER         SET SES         ; SET SES         ; SET SES           0162         016         LD         C, IDN         ; Function Sequential crite	0184	43 65		4.0	2,22372	, (101 10001 10
0186     11 0288     12     Df.FC8     Set FC8 address       0186     CD 0005     CALL 8D05     Write 1 sector       0182     37     OB     1     Check Ares       0187     15     2.5TABTW     1007       0101     F5     PUSR AF     16 Aregion then error       0102     01     10     C.10N     Punction file close	4186	OE 19		10	C,150	; function sequential vrite
0188         CD 3003         CALL         BDOS         Write 1 sector           0182         87         08         ;         Check Areg.           0182         87         08         ;         Check Areg.           0182         87         15         2,57A87%         ;         1007           0101         CLSFL:         0101         FS         PUSR         AP         ;         16 Aregro 0 then error           0102         0103         10         C,1004         ;         Punction file close	0188	11 0288		13	DE.FCB	; Set [C8 address
0182     37     08     A     ; Check Ares.       0182     15     2,STABTE     ; Loop       0101     CLSFL:	0188	CD 3005		CALL	8005	; Write 1 sector
CIBF     IS	0182	87		0.8		Check Areg.
D1C1         CLSFL:           D1C1         PS         PUSR         PUSR         ; If Areg(>0, then error           01C1         PS         LD         C, 10M         ; Punction file sloke           01C2         DT         LD         C, 10M         ; Punction file sloke	CIBF	_ 3 _ 4 -		- 8	*19:98.8	ا <b>و</b> ت هـ .
01C1 PS PUSH AP ; If Areging then error 01C2 Of 10 LD C,10N ; Punction file ilose			CLSFL:			
01C2 OT 10 LD C,10N ; Punction file slose	0161	P S		PUSR	N P	; If Aregia then error
	0102	0 1 10		LD	С,10М	; Function file close
	0101	11 0228		10	DE .ICB	; Set TCB address
31C7 CD UUDS CALL 8005 ; Close (110	3107	CI 0005		CALL	8005	; C1050 (110
OLCA FA FOR AF I I I I I I I I I I I I I I I I I I	A 2 1 0	E 4 8 7		0.8	Å	Chack Ares.
	0108	C.S.		RET	2	A=0 then no error
UTC JE CA LD A,10 ; Error code	0160	JE GA		10	A,10	Error code
ulcf J7 Scy Set corry flad		37		367		; Set corry flag
01D0 C9 A87	0100	C 3		8 <b>2 1</b>		i

			******		
		;		DISPLAY MESSACE	*****
			X0781		
		;	NGIE.	Display message	string until find OQN.
		;	() entr	y parameter	()
		1		H1 TOP address	01
			<> rutu	NOSE	()
		i	<> pres	NONE	()
0101	Ťε	OSPHSC;	LD	A. (HE)	: Cet e date
0102	A 7		0 A	A	; Check end mark DOH
	6.5		861	٤	; IL UUN then return
0104	25		PUSH	8 L	; Save pointer for address
0106	CP EBOC		CALL	CONOUT	; Display a pessaga
0109	E 1 2 3		POP INC	A L U L	; Restore pointer : Vodate pointer
010-	18 24		18	O S PM S G	
		:			
			***-**		**********************************
0100		IRRTBL:			
0100	0183		DV	PARAEBR	; Dummy
0165	0203		0 0 1	PANAEBB OPENERR	; Paremeter error : Already open error
C 1 E 3	0570		06	XOPNZEB	Not open error
0127	0557		0.6	BOVFIRR	; Forced and ; Buffer over flow e-ror
011 *	0215		05	TINEERR	; Time out error
0110	0240		0 4	CONNERR	; Protcol error ; Communication error
0,66	u 2 6 0		0 N	PLHKEBB	Pile make error
U	0212	;	<b>U</b> •	/1	; FILE WEITE SEVOF
0123	50 61 72 61	PARAERR	: 	PARABELET ATTOT	1.00N
0177	60 65 74 65				
0178	72 6 <b>P</b> 7 <b>2</b> 00				
0203		OPINEBR	:		
2207	61 64 79 20		UB	.ziseedā oben.'n	
02.0	6P 70 63 62				
3230	50	NOPNERR	:		
0233	- 2 67 74 20 58 70 53 68		08	'Nct cpen',00A	
0215					
0219	46 67 72 63	2 5 N D E 5 3	C 8	'Torced end'.00W	
0210	65 64 20 65				
0221	6E 64 00	30 V 7 E 2 B	:		
0224	42 75 66 66		08	"Buffer over flo	~',00R
0225	85 12 20 0F 75 65 72 20				
0230	56 6C 6F 77				
0235		TIMEERR	:		
0235	54 69 60 65 TO 58 75 71		3 8	17104 OUT1,00H	
023D	00				
0.25 0	50 11 65 14	PROINER	: 	Protoni errori.	0.0 м
0242	63 67 6C 20			Protegi atron 1	
0246	65 <b>12 72 67</b>				
0240		COMMERS	:		
0240	43 57 60 60 75 68 69 63		08	Communication e	F20F1,008
UZ 5 4	61 74 69 61				
0255 025C	6£ 20 85 72 72 68 72 00				
0260		FLHKERR	:		
0260	46 65 60 65 20 63 61 68		n n	LITE COU UUS OB	8 . UUM
0260	20 68 GF 74				
0270	65 00				
0172	67 73 60 T4	FLSTEBB	: 	INDIAN OPPOST AA	N
0276	65 20 65 12			HELLE BITTOR , UU	
0278	72 65 72 00				

0272			ENDRISC:	
0272	0C 46 69 6E		0.8	OCH. 'Pinished'. CON
0282	69 73 68 65			
0286	64 00			
0288			ACVMSC:	
0288	00 52 65 63		0.8	OCH. 'Receiving'. ODM
0280	65 69 76 69			
0290	62 67 00			
	-			
0293			SPNONSC	
0293	00 53 65 68		0.2	OCH 'Send file from HOSTL DOW
0297	64 20 66 69			cent send title title Host (00H
0298	65 65 20 20			
0298	20 20 66 72			
0243	67 60 20 48			
0247	48 53 54 00			
0248			KENBLK-	
0248	03 03 03 01		0.0	9 3 9 1 00N 078 00N 'Proco'
0247	00 07 00 50			eleteliteenterpteent steam
0283	72 65 73 73			
0287			PRMCADE	
0287			0.5	2 Price Bessage address save and
0289			TCAHADRE	- ' when appendix sources over side
0289			D S	2 : TCAHARRA addres area
				- I TRANAKEN SOOLEE SEES
0288			PCB:	
0288			0.5	37 : PCB Aren
0220			DHA:	
0220			DS	129 ; DMA Area
			END	
Hecroe:				
Symbols				
0005	8005	0224	80	OICI CLSTL
0245	CONTERR	E B C S	CONIN	ELOC CONLUT
0120	O H A	0101	DSPHSC	0195 ENDOBL
0272	ENDHSC	014 #	ENDPRO	0737 EBNGADB
0100	PREIBL	0228	FCA	0219 FZNDER0
0260	PLHETRA	0272	PLNTERR	RBAS INFORM
0248	KETBLN	1000	MAINSP	O210 NOPHERS
0203	0 P E N E 8 8	0183	PARAERO	0238 PROTERO
0150	RCVFILE	0286	8 C 7 H S C	0293 SENDHSC
0100	START	0144	STARTW	EB90 TCAN
0289	SCANADE	0138	TCAHERR	D235 TIMEEBR
2336	TOUCH	2803	VEGOT	

No Petel erroris)

.

		0 1	*****	8109 C	RAPHICS ( L	INE) SAMPLE PROCEAM
			*****			F & & & & & & & & & & & & & & & & & & &
			NOTE:	This s This p	sople pro rogram di	gree is executed only on 247-10. spinys the line pattern until "END"
		b		Rey 15	pressa.	
		•	()	esble co	ndition	6.5
		;	- 280			
		- 	$\leftrightarrow$ loa	bbe antb	8893	4.3
		•		100N		
			() con	stant va	Jues	4.3
£ 8 0 3		NBOOT		200	NE0830	
806 8809		CONST		5 G U 5 G U	028069 92808N	
2006		CONCUT	s	29U 29U	0 2 8 0 C H 0 2 8 9 3 8	
2896		TOUCB		IQU	028968	
1000		MAINSP		290	01000N	
		1				
0100		START:				
1100	31 1000		LD	58,941	N S P	; Set etack pointer
0103	02 18 CD 880C	·		C,188		; Cugao - off - FRC + '2'
0105	0g 32		LD	C, '2'		
0104	CD EBOC		CALL	CONDUT		
0100	CD 8883 G2 CD	,	LO CALL	C,00N (84PN)	C S	; Set Loit function ; Package initialize
0112	06 0C 02 05		1.D 1.0	8.12 C.018		Display 12 lines Line subction
0116	21 0148	10081	LD	31,740	E 3	; Set package address
0119	C 5	LUUP.		80		: Save counter
0118	CD 8983		CALL	C BAPHI	<b>c</b> s	; Display lice
0112	£ ) C 1		POP POP	H 1. B C		; Hestore pacinge addites Bestore counter
0120	1) 000c 19		10 400	02.13 H1.D2		; Nake nort package
0174	70 63		DJNZ	LOOP		1
0128	10 30	4	LD	C.018		; dake 1 touch key - maka (stat has
0128	CD 2896		CALL	10008	014	
0122	CD \$909	-	CAIL	CONIN		, walt for sey 10
0131	02 18 CB 220C		LD CALL	C, 388 Conout		ESC · '3'
0136	01 33 CD 5806			C, '3'		
0138	C3 1803		JP	¥8007		+
			• • • • • • •			
				A ATAL	********	
0138		<u>284978:</u> :				
0132 0142	04 02 02 01 00 07 00 49		8 0	4,14,2	,1,000.07	M, OOH, 'END"
0146	4E 44					
0145	000+ 0064	PACE1:	0 N	10 100	50 100	
014C	0032 0064				1201100	
0150	6 7 7 F F F		) W	0 <b>7 7 7 7</b> M	1	
0154	000A 003C	й	<b>0</b> 4	10,60,	\$0,60	
0   5 8 0   5 C	07 00 0035 003C			1.0		
0151	****	,	0 W	0 <b>7 7 7 7</b> 8	I	
0160	000A 0064			10,100	1.10.60	
0166	01 00		D 8	1.0		
010A	* * * *	;			•	

016c	0032 0064		Ðu	50,300,50,60
0170				
0174			88	1,0
01/0	****		1) M	UFFFF
		;		
0178	0016 0050		ÐW	30,45,65,45
0170	0041 0020			
0180	01 00		10 B	1,0
0182	EFTF		שמ	0 7 F F P N
		:		
0184	0041 0055		<b>D N</b>	65.85.65.45
A 1 9 A	0041 0020			
0180	01 00		n e	1.0
0180			55	0.98889
0102	****			UPPERH
0130	0041 0022		0 14	82'92'20'100
0194	0032 0064			
0198	01 00		00	1,0
A 2 1 0	* * * * *		0 4	0 <b>P 2 8 2</b> A
		;		
0190	0041 0020		0 5	65,45,50,60
0140	0032 003C			
0144	01 00		08	1.0
0146	****		D W	077778
BAID	0017 0020	,	<b>DN</b>	30 45 10 60
0140	0004 0030		<b>2</b> -	50,03,10,00
0180				
0100			6.0	
GIDA			0.	
0184	0012 0020		0.	30,45,30,85
0128	0018 0055			
0780	01 00		10 B	1,0
0186			D 🖌	BAAAAD
0100	001E 0055		10 W	30,95,65,85
0104	0041 0055			
0168	01 00		0.0	1,0
0104			0 W	DAAAAH
			-	
0100	0.012 0.055		n N	10 95 10 100
0100	0004 0054		-	30,33,10,100
0100	41 66 GU			
0104			0.0 D N	0 N
0100	****		1	UAAAAU
			END	

.

			******				
					*******	JHCLE) JAMPLY PROGRAM	
		;					
			NOTE:	This sag		eres is executed only on ENT-L	٥.
		5 5 5		This pro bey is pr	gran di enned.	aplays the circle pattern unti	I "END"
			()	able cond	ltion	()	
		•	. 280				
		8 	() load	ling addre	<b>R.</b> B	4.3	
		ė	PHASE	1 0 O H			
		1	() cons	tant valu	<b>e</b> 9	0.3	
2803		; ¥8001		200	ИСОВЗО		
806		CONST		EGU	028068		
2 2 U 3		CUNIA	•	2 Q U	OFBOSH OFBOCH		
693		CRAPH	105	ເດນ	028938		
6996		TOUCH		E G U	0 T 8 9 8 N		
1000		2 M C A B	p	EQU	010009		
		-			• • • • • • • • •	•••••••••••••••••	
		;				•••••	
0100		57481	:				
0100	31 1000		L D	SP, MAINS	P	; Set etach pointer	
01(3	02 18		Lu	с,18н		: Cursor off	
- 105	CD 280C		CALL	CONOUT		2 RSC + 121	
0100	CD 280C		CALL	COXOV1		;	
0100	0.8.00		1.0	C 0CN		t Class acrees	
010 P	CD 280C		CALL	CONDUT			
0112	06 24		1.0	8,36		Display counter	
0114	02 06		L D	C,068		Eircle function number	
0116	e 6	LOOP:	31.60	20		Save counter	
0112	21 0147		10	BT PACK		Set package address	
0114	CD 8893		CALL	62191C8		Display circles	
a ( ) D	C 1		2 O P	8 C		; Bestore coubter	
0115	2A 0148		10	SL. ILCS	• 4 1	Cet I direction radius	
0121	28		SEC	81		Ser Sem Pedius	
0125	22 0148		1.0	193C6+41	.ac	: Cet 7 direction redive	
0125	28		212	81			
0129	2.8		LEC	81		Decrement V radius	
0124	22 014D		10	PACE-6	. 별고	Set can cadlus	
0120	10 27		5581	1307		, COLIT A LOGICO 4 1	
0120	02 31		10	C,012	z	Set a bey black data	
0134	11 U120		-5	10103	•	Display a key block	
0137	C3 1909		CALL	(0818		; Neit for key in	
0134	0 E 1 B		1.0	с.19М		; Cuursor an	
0130	CD 880C		CALL	CONCUT		1 22 • 13.	
0137	08 33		13	C. 131			
0144	CJ 2807		I A L L I P	¥6001		1008 3001	
		1 1		384 ATAC	A 		
		4 7					
0147	0.03. 0.04.	PACE	2.6	42.54		; Center corcisate	
0148	0025 0015		54	10.72		S , Y TEDING	
01	01 00 00		D B	1.0.0		Attribute & flag	
0152	0000 0000		D 6	0.00		; Start & And angle	
8210							
6156	04 02 02 01		4,14,2.	1.008.079	. 0 0 N . * E	NO" ; END key	
0 3 S A	00 07 00 45						
0116	4 2 4 4						
			E N D				

## (16) SAMPLE 16. GRAPHICS (TILE)

		2 0 1 2 2 2 2 8		83 2018	APHJCS(T	DESCRIPTION DESCRIPTION	••••
			NOTE:	This so This pr bey is n	aple pro ograa di reased.	gram is executed only on ENT-1 aplays the tile pattern until	0. - E N D -
					dialaa		
		ь • d			dition	¢ 3	
		á	. 280				
		* 0 #	<) ]on	ding addr		4 3	
		:		1001			
		1	() 000	stant val		e 3	
2 80 3 2 80 9 2 80 C 2 8 9 5 2 8 9 6		CONTN CONTN CONDUT CRAPHIC FOUCN	5	20U 20U 20U 20U	0 2 8 0 3 M 0 2 0 0 3 M 0 2 0 0 3 M 0 2 0 0 3 M 0 2 0 3 M		
1070		T Tainsp		290	010008		
		:					
			•••••	NAIN PE	0CBAN ••••••		
0100		; START:					
0100	3) 1000	;	LD	SP,941N	5 9	; Set stack pointer	
610	JE 18 TD 2840		18 CALL	C.188 Concut		: Cursor off : ISC + '2'	
0 01	02 32		LD	2,121			
01.4	CD 280C	:	CALL	CONCUT			
0 1	OF OC CD RROC		LD CALL	C, OCH Comout		Clear screen	
0112	08 04		10	8,048 C.06H	;	; Number of circle : fircle function	
0116		100P1:					
0116 0117	C 5 2 1 0 1 6 B		PCSB LD	BC BL,PACK	1	; save counter ; sat package address	
0114	CD 2893		CALL	CBAPRIC	2	: Bisplay circle	
011"	11 000▲		1.0	DE,10			
0121	2A 016 P		12	BI, PAC	E1+4)	: Get I direction radius : Beset carry flag	
0125	zn \$2		5 8 C	82.92		8L - 10	
0127	22 016P 22 0171		10	(PACE1+	41,4L 61,9L	; set new V redius	
0120	16 27		0332	100#1		; Display 4 circle	
012 #	06 05		LD	8.058		Display 5 times	
0131	02 08 21 0195		13	C,058 81,7711	PTN	; Yile function ; Set tile pattern address	
0116		10012:				· favo eile perfern advess	
0135	E 5		PCSO	30		SAVE COUNTER	
0135	71 017A		10	81.3372 C242815		: Set pechage address :	
0135	C1		POP	90	-	Bestore Counter	
0125	ATED AT ADDD 41		10	2E1173C	1 <u>5</u> 1	Get Start Address	
0145	19		120	12.16		: Nake next packede Sak new traft & constraint	
0116	44 - 41 - N		A 6	(FALAS)	, = =		
9 1 4 9	E1 13 0004		9 <b>9 9</b> 1 D	91. 21.048		Skip 4 byte	4
0142	19		A D D	3C, JR	=	A new sext tile pattern address	
014E	10 E3		2358	10092	21, <u>21</u>	. 301 Hor tive petters societ	•
	<b>AR</b> (1)		1.0	С.01И		: faire 1 touch key	
0155	13 0198		10	00.6278	15	Set the key block oueber	
0125	CD 2396		CALL	TOUCS		; ATEBIEA & KAA DIOCE	
8110	CD 2809		CALL	CONIN		: Walt for touch : Curnor on	
0150	CO ENOC		CALL	CONCUT		ESC + '3'	
0103	08 33 CD 880C		10	CONUN1 C', 13,		1	
0192	få Engl					LAND NEGOT	
0368	C3 6803		JP	1001		, seeb secon	

				0.000000000000000000000000000000000000	
0168		PACK1:			
0168	A 1 0 0 A 5 0 0		0.9	42,74	; Center
0167	0028 0028		0 5	40,40	; I , Y direction redium
0173	01 00 00		0 8	1,0,0	; Attribute & flag
0176	0000 0000		0 W	0,0	; Start A and angle
		:			· ·
017A		PACH2:			
ATCO	0028 004A		0.4	43,74	; Start cordínate
3710	04		09	4	; Tile pattern length
0177	0188 019C		0.4	TILEPTN, TILEB	; Tile pattern Address
C 1 9 3	01		08	1	; Attribute
0184	0800 BA10		0 N	NORK, SON	Vork Ares
		1			
0188		TILEPTN	11		; Tile pattern
0139	3455 4455		11 4	GAASSN, GAASSB	
0180	CCCC 1333		10 M	000008,033338	
0190	2799 99FP		D 🖬	CFF69N, 099FTR	
0194	0303 3030		<b>D</b> N	00303N,030308	
0198	1144 1144		D 🖬	011448.011448	
				•••••	,
019C		T1128:			
0186	0000		<b>D</b> N	00000	Back ground tile nettern
			-		
9.019.0		REYBLE:			
0198	04 08 02 01		2.8	4.14.2.1.0DN.0	7H_008.18ND1
0142	00 07 00 45				
0146	45 44				
0.1.4.8		NORR:			
0140			23	80N	Nork area

17) SAMPLE 17. KANJI

		;				
		;		BIOS KANJI	SANPL	E PROCRAM
					*****	
			NOTE			
				This seech		rae displays and printe our kenil.
				(code 0400	N - 04	
			•) 868	emble condit	lon ()	
		•	. 280			
		8	c) loa	ding addreas	• >	
		4	. PHASE	100M		
			() con	stant values	4.2	
E 9 0 3		V BOOT		2QU 01	8038	
EBOC		CONOUT		EGN OF	BOCH	
E 8 9 P		KAN J1			8 6 F N	
1000		HAINSP		2QU 01	0 0 0 R	
				MAIN PROGR	Aff	
		1				
		1				
0100		STABT:				
0100	31 1000		10	C IBN		; ? ser rue arec's bordres
0103			CALL	CONCUT		. CONCUT ESC + '2' cursor OFF
0105	02 32		10	C, 328		
0104	CD 280C		CALL	CONCUT		;
0100	3.0.0	4	10	A.008		; Set the code(low) of a sji
0100	11 0141		10	DE, HANDAT		
0112	13		INC	0 8		i
0113	13		1 8 6	DE		; Calculate the low code setting address
0114	13		1 N C	DE		;
0115		LOUP:	L D	(02).4		: Set the low code to the data sectors
0115	21 0141		1.0	BL. BANDAT		; Set the address of kanji data
0118	06 01		LD	8,018		; Set the persenter to display on the LCD
0118	23		PUSN	A #		; Presezve registers
0116	0 \$		PUSB	20		
0110	CD 8883	-	CALL	<b>A Y K 3 3</b>		; usplay on the ICD
0120	21 0141		LD	BL, KAFDAT		; Set the address of kanji data
0173	06 02		10	8.072		; Set the parameter to prict out
0125	CD EBSP		CALL	EANJI		Print out
C 1 7 R	02 00		13	C, 0C 9		· · · · · · · · · · · · · · · · · · · ·
0134	CD 180C		202	18		
0120	# 1		202	AF		- -
0128	10		150			•
0130	7 E 00		2.2	0		; le all data displayed T
0132	20 21		13	SZ,100P		; ies then about also LOOP
	OT 18		LE	C,198		; CONOCT ESC - 121 curver 25
0116	CD 880C		CALL	CONCUT		1 8
0139	01 33		10	C,33A		
0132	C2 880C		CALL	CONOUT		9 1
0135	C3 6903		CAL-	- 8001		
						*********
		;		NANJI DATA		
		1		**********		
		4				
5 T 4 3		ZASSAI.				
0143			ΟΞ	4.91.4.914		
0145	U 4					
			<u> </u>			

(18) SAMPLE 10. MASKI

				• • • • • • • • •		а в а с а с а с а с а с а с а с а с а с	L 2 J	980684M
				NOTE :	This ea interpu	elpe pro opt diest	Dere ble.	am makes lace. interrupt and CVS
				()	le condi	tion	• 2	
					280			
			1					
				() loadi	ing eddre		•	
							100	â w
				() CONBI	lant valu		• •	
6 0 6 3 A 2 6 1 0 0 0 1			VBODT Maski Stackad		t e U Z e U 2 e U	и с о е то и а 2 е то и а 2 е то и а 0 о о 1		; NECOT entry address. ; NASEI entry address. ; Stack aves.
0077 0071			880V <i>P</i> 88352C		29U 29U	11130113	18	; OVF interrupt bit mashing data; ; lasc. interrupt bit mashing data.
C 1 C C C 1 C C	31	2000	START	LD	SP , STACE	( A D	; 1	let stack pointer.
0103		<b>S</b> 0		LD			; 1	lead current interrupt mode.
0105	C D	885A 43 011A		LD	HASEI IOLDINTI	11,80	1	lava current intertupt mode.
0100	78	21		AND	18077		1.1	fetest 047 interrupt bit fdissblei
0107	47			LD	8.4		7	
0110	7 9			LD	A.C			
0111	16	P 7		4 N D	BRISEC		11	least less. interrupt bit-(disable)
0113					-1-		•	
0114	CD	2854		CALL	JASE1		1 1	get gen internat mode.
0117	с 3	8803		3.8	¥8001		: 4	Juep MACOT
0114			OLDINTS	<b>I</b> :	2 0	2	; 4	31d interrupt mode anys area.
				E N D				

				This cand area of ble cond . 280	08X.LD182 mlps pro RAM disk	I SAMPLE PROCRAM ogram initializes the directory k. ()
		:		PHASE		7008
		0 	() cons	tant val	u <b>e</b>	< )
E 0 6 4 0 8 6 3 E 8 6 3 E 8 6 3		NBOOT Stori Ldiri Inform		2 Q U 2 Q U 2 Q U 2 Q U	К С В С З И К С В В З Й С В В З И С В В З И С В А 5 М	: VBOOT entry address. ; STABX entry address. ; LDIRX entry address. ; INFORM entry address.
1069		AD_BAH_	IN	8 Q U	020658	Top address of RAM disk
F068 F062		8 A M D _ S 1 D 1 8 S 1 8 B	22	EQU	0 P 0 6 8 M 0 P 0 6 2 M	; Estend RAM mize, (for RAM disk) ; directory area size (/125bytm)
0003 u007		A D O I 9 8 N A D S B C B N	R F	2 Q U	3	: INFORM parameter (DISONN) ; INFORM parameter (SBCONK)
0025		PHTPTN		8 Q U	0 2 S R	; Porent pattern.
1100		STACEAD		εου	10008	; Stack gros.
0100	39 1000	STABT:	LD	SP, STAC	EAD	; Sat stack pointer.
0103	CD 0147		CALL	CALDETO	p	; Calculave directry area top address A bank.
0106 Ci08 010C 010F	38 25 2D 48 0169 2A 0167 C3 2860		10 10 13 CALL	A, 22722. BC, 1018 BL, 1218. STORE	S BNE 1 SDB 1	; Pormat pattern (ORSb) ; Creg. : bank of directory area. ; Blreg.: top address of directory area. ; Set foreat pattern.
0112 0114 0117	66 03 68 5372		LD CAIL PCS3	C.40015 187622 81	<i>4</i> X 5	Cet DISCONK address. Save.
2115	02 27		L D	C.10585	256	; Get SBCBNK address.
0110 0120 0121 0122	77 2843 24 0159 77 21 77		10 10 20 10 10	INFC22 A, 12133 (01), A B1 (01), A	561	Get bank date of directory area. Set to SBCBNE Bestore DISBNE address. Set bank data.
0123	02 30		1.0	с, э		
0125 0125 0124 0125	- A POSZ - B 125 47 - C 25		5 2 L L 2 A P	A     2   2 S A 2 , A C	1281	Directory area size, (/1200) Nake number of transmitting date.
9120	ũ E		910	36		
0125	IA 8267 54 50		10	HL, :DIR I 2 P	ADBI	Hireg.: top address of directory area : Circg.: next A002068. :
0134	13 CD 2863		CALL	12183		Parman directory area with DESH.
0137 0135 139 0130	17 18 20 48 0165 20 2860		X O R X 1 X 2 1 3 1 1 4 3	A D£.31 SC,1218 STOBX	8 N K ()	Areg.=00H Hirog.: top eddress of checksus ares. Creg. Hanx deta of checksus ares. Set 00H to checksus ares top.h
0140	JA 7062		L D D E C	A , I D I 9 S A	<b>I Z R</b> ]	; (directory area size) / 128byte ;> checksum data number of director;
0144	4 P		10	C , A		
0147	13 13		190	8,0 02		(T)
0148	13 CD 2863		15C CALL	LDIEX LDIEX		; Direg. Hireq. 1] ; instalse checksum date of directory area.
0140	C] 1803		3 P	-		1005 gavt

					Calculate	top add	ress & bank of Directory area
							***************************************
			-				
0148			CALDET	0 P :			
0148	2.4	1065		10	NL. (AD BAT	1 180 - :	16 extend BAH not sounted, then
0152	11	6000		1.0	DE. 6000M		(AD #AN 19) is too address.
0155	3.4	P 0 6 8		LD	A.LBAND SI	1261	.)
0158	87			0.8	4		
0159	2.8	0.1		18	2.CALDRTI	0	
0158	8 3			£ 1	DE.NL		
0150			CALDRY	10:			
0150	22	0167		LO	(DIRADR),	HL :	Set top address of directory area.
015 0	0.7			BLCA			
0150	0.7			RLCA		1	heke benk data
0161	0.7			RICA			
0162	0.2			RLCA			
0163	12	0169		1.0	(DISSKE).	a :	Set bank data.
0166	6.9			827			
						,	
0167			DIRACE	;	09 2	;	Top address of directory area.
0169			018886	:	05 1		bank data of directory eres.

ZND

. 2 5 0 SELECT SEBIAL HODE ÷. 1 i NOTE : Check current eersel mode and if it is different from selected mode ::;; then set new mode parameters to ABT. If same mode is setting them return immediately. Control register are ... CTLBJ (POGN) ABTHR (PISH) 2 SK8 (P10R) ARTCR (P168) ÷ () entry persector () Creq. : Select code code. OON : SYSTEM Oin : User. Oin : Epop. () return persector () Areg. : Select information. 008 : Same device selet. 018 : Device changed. 2flag. : Depend on A register. ā () preserved registers () ÷ 8.0 <> Constant value <> 5 < port adcress + ÷ 101 0008 .... 201181 0015 248158 E Q U 0158 2 Q U ZARTSR 0015 0151 0 2 1 8 ZARICE 191 0169 0018 2568 2 O U 0188 ( bit address of port date ) : .... TTRENTT L Q U 000001008 0010 28RCO 1 Q U 280 001000008 28861 0020 0000 28862 101 210000008 1:250 28862 TOU 10000000 0004 25540 2 A U ......... 0008 25551 200 300010008 101 20000013 0001 1751 000000108 28078 0002 290 6644 2232 260 330001006 0020 28875 TCC 301000008 <> system wors area define () 200 F241 SENODE 0 2 4 1 8 SETABL B2CTL21 191 193 1184 371968 110068 ..... FODE STRATS. 298 0.0098 F004 BZARTCA 101 CTODAR 200 310208 1000 8255R 2023 1511223: Current serial mode. SRNODE - Selected mode? Tem. (No operation is done.) 1.0 A. I SESO221 0000 34 7311 0003 91 SL'B c z ..... 0004 C S : forced SELSER. 0005 PSELSER: 0033 08 15 1.5 A. IZASTSSI Transmit empty ? So. (wait) 28 04 23 7A 0 0 0 7 AND 21323971 18 Z, FSELSER 8000 LD Set new serial mode. .... 19 A.C (SEHODE),A 32 7213 10 000C RLCA 07 0007 RLCA 07 0010 A . C C . A 0011 **3 L** A 0 0 LD 0012 1.2 BE : Table offset. HL : Serial parameter table top. B, C NI, SETABL 06 00 LD 0013 21 7196 0013 10 : HL : New parameter table HL.BC 0018 0.9 address.

			:	4 9et 1	TIE and	ITE to 'Lo	w 1 - 2		
0010	JA FODA			1.0					
0010	EG FA			AXD	0778-2	122-1232			
0011	JZ FODA			1.0	(82481)	C 2 ] . A		Update	
	22.16			OUT	(24270	B ] , A		; Output	J/O register.(CT181)
			:	+ for (	CT181 I	Sot Baudra	te ) >		
0033	18			LD	A, CHEN				
0024	E6 P0			4 N O	288C3+		1 + 2 8 8 C 0		Clear baudrata data.
0027	34 1006			LD	INLI.A				4 #
0024	26 07				A.(87C)	7181) 8863-88862			; Gat new CTLE1 data.
0030	86			0.8	( 11 1		- 10861-		. Hats out deverate date.
0020	35 6008			LD	(82671)	111.4			- Codata Avitas area
6636	03 00			OUT	(ICTLE	1),A			Output J/O register.
			:	t Pre-	read ABT				
0032	2.3			186					
0033	46			1.0	8.(8L)			There	are don't care bits.
0.034	2.3		:	4 7ar :	SV8   S#	t leresl e	witch 1	,	
0035	78			190	H L			1	
8 6 8 0	28 OC				A. ( 0 L )			; Cet ne	w SVR data.
0038	17			LD	(81).4				de series seres ole.
0038	JA FODC			LD	A . ( B Z S	¥8)		Cat Cu	ryone SVA data.
0030	20 73			AND	0 F F R - Z	5 5 ¥ 1 - 2 5 5 ¥ 0		; Clear	eartal switch.
0031	32 PODC			0 8	(87)				
0042	D3 13				( 2 2 5 8 8	1 <u>,</u> A		; Update	eyetee Aree.
					(23-0)			, output	1/0 Postator.
6044	23		:	( Set /	A 87C8 -1				
0045	JA PODC			INC	81				
0048	28 08			LJ	A.(#25) 96661	⊨ #r		Cat eu	grant SSB
0044	78			1.0	A (81)				hauf anna is south
0048	20 09			3.8	NZ, SET.			; then a	et NL pointed date.
	26 00 4 P			AND	0228-20			; Clear	conscol (no.1078,878)
00,0	JA FODA			10	C.1			1 Save:	
0053	E6 22				280784			; cat cu	arbon current correctors.
0055	8.1			0.8	c			New ma	de and old control line.
0056									
0056	32 FODA		361AM11	10		-		. Undere	
0059	03 16			001	IZABTC	B1,A		Output	1/0 register.
			ī	· Set	ARTNB >				
0058	2.9			1.0					
385C	32 FODS			1.10	3,0 (B2ART)	• R I A		Seved	SNIME GAVE:
205 F	D3 15			027	LARTS			-	
				+ Set :	zetura 3.				
0063	1.7				-				
0043	30			103			AFPE	+ ) ( Ifl	ag OPF 1
3063	C 8			RET			i 1		
				END					
1126618	11								
0003	VSELSER	FDDA	BIARICE		2003	RETEASS			
1006		1000	825×R		0000	SELSEB			
0036	3 E 1 Y M I ] O	1241	JETODI		4136	JEATRE			
1010	288168	0013	288C1		36:3	23862			
0010	20000	0000	201681		2201	1207R			
0010	19979	0004	2838		0004	25540			
0008	25541	0015	2558		0001	2738			
0004	ZYXCHPTY								

No fatal errorial

(21) SAMPLE 21. CALLY

		; ••	•••••						
			2109	CALLE SANS	LE PROCE	A.1			
			** -						
			Thi	e cample pr		eye the	evete.		
				a prograe o	INGO MELO	UV FOUL		Adrem 1n	<pre>ep table.</pre>
		- C P	assemble c	ondition	6.)				
			. 280						
			antheol	ddrees	4.3				
			. <b>P</b> NA	3 B	1001				
		1.4.9	constant		6.3				
2803		NBCCT	EQU	028038		; 5100	. entry	addrees.	
EBAS		CALLX INFORM	EGU EGU	N 2 8 6 9 0 H 2 A 6 9 0		; CALL:	K entry . RM entry	addrees. addrees	
0025		MELODY	EGU	000358		; HELO	DY PONTI	ne in ey	etes juan
0003		ADDISBNE	Tev	3		1570	IN garas	eter (Dl	5 8 N E ]
8866		ADBPINT	EGC	1		INFC	IN param	eter (BP	INTEGL)
00 <b>pp</b> 1000		SYSBNK Stackad	I Q U	<b>отр</b> 1000и		; ; Stec			
0100	31 1800	START: LD	SP, 3	TACEAD	; 5	tack po	inter.		
0103	0 8 0 3	LD	C , A D	BPINT	; Diest	le 100c	Interr	ept in A	ELODY.
0105	CD ZBAS CB FZ	C A 5 E	LL 1770 1 7,(8	12 M					
A 0 1 0	02 03	LD	C,AD	DISBYK BH	Cet I	ISCONE	ddrese.		
3100	32 77		A . ST	SBNE	Areg	: 8787	tri bank	dete.	
0111	17	10	(81)		; 541 1	G 7138N			
0112	21 012E 11 3000	1.D 1.D	81, N DE, N	USICD7 USICTBL	Trene		ic deta	to upper	30009
0119	01 0120	LD	BC.N	USICSI					
	20 00					table			
0110	21 3000 DD 23 0025	10	11,5	ELO2Y	1 14814				
0124	02 62 CD E869	LO	C.2	.z	: LOOD ; CALL	SELODY.	•		
0179	(		¥800	1 1	Juep				
01.0									
C12C	0 6 2 9	MUSICS2; MUSICBT:	D M	41					
0178	06 30		28	10.0	4.27.	3.23.	4.25.	4.27.	5,29
0124	05 10 04 19		2.	• . = 3 .					
0129 012C	04 18 0 <b>9 10</b> 04 2 <b>3 04 10</b>		38	4,32.	4,29.	4,27,	4,25,	4,21.	4,29,
3140	04 13 04 19			5,27					
0144	04 18 04 10								
0145	04 10 G4 12		35	4,25.	4,21,	9.29.	4,25,	4,27,	9.25
6141	35 10 04 18 34 13 38 13								, 90
0116	04 20 34 18		28	4.32. 5.25	3.29.	4.27.	4 4 4 3 1	4,411	41441
013A	24 18 04 19			-					
0162	05 19			- 18		4 9 0	4 33	4 14	4.24.
2164	04 20 04 20		38	5,32	4.34.	•		4,201	
0165 0165	04 10 04 30 04 22 04 22								
0170	08 20			4.29.	4.29.	4.27.	4,27.	10.25	
0176	04 10 04 10								
017A 017C	10 19		0.8	Q					
-		23							

			•••••	8103 (	NAE MARVOI	PLE PBOCEAN
		0 7 8	NOTE:	This se data and	neple prog I displaye	rem reads the screen buffer the data.
		•	()	eeble co	ndition	4.9
		÷				
			<>> 10a	ding add	tress	٩.
		•	. PHASE	100N		
		:				
					HILLS.	
2 3 0 3 2 3 0 9		NBOOT Conin		5 G U	0 Z 8 0 3 8 0 Z 8 0 9 8	
280C		CONCUT PUTP <i>P</i> K		800	0 2 8 0 C 8 0 2 8 6 7 8	
2878		BOVEAN		290	028788	
				System	work add	
2257		LSCSIER	<b>r</b>		0 # 2 5 7 8	
1258 1572		7 2 C 2 3 2 5 1	r 9 v	20U 20U	072581 075728	
		4		Other	copetante	
		MAINGO		200	010008	
1000						
100A 100D		C B		2 Q U	008	
		:				
				HAIN 8		
0100	31 1000	310017	10	5 P , MA 1	NSP	; Set stack pointer
0103	07		LD Rlca	7 . ( 480	CUNTET	; if for-iu, then derive the sey-los- ; at the top of the screen.
0107	01 0101 32 01		10 10	8C,010 A,018	18	; (Rev code)
0100	04 886F		CALL	NC.PUT	275	(Kay define)
0107		100910:	13	80,010		, IGATICA DOMITICO
0112	79 CD 0142		LD CALL	A . C DSP10		; Display lice humber ;
0116	CD 0156		CALL	CELF		:
0119	~ \$	100970:	9759			
0114	CD 1878		CALL	B D V B A E	I	. Read the ecreen buffer data
011D 011E	87		707 01	80		- teror:
0117	20 16		38	#2,570		; 7
0121	CD 013D		CALL	05991		: Display attribute & character code
0124	3A 8257			A,(150	512221	. Jelt coloeu Donygrop
0128	88 30 82		C 9 1 8	B SC,100	) P 2 0	; Vithio screeo? ; Vee
	CD 4156	-	C 4 1 1	CRIF		: Tove cursor to sext line
0128	00		150	C		Nest line position
0131	3A F258		_D []	E, J18 A, (130	512111	, 10g of 10,0000
0134	89 30 D8		C P 1 9	C SC.100	0 1 0	; Mithin screep? ; 740
		5700				
0137	CD 8808	3100	CALL	CONIS		: Input any key
0124	C3 1803	:	1 9	CHOCY		; 20d 01 (08 D¥04r
		:	•••••	0 1 S P L A	Y THE NL	A TA BY NEX
			170H	laeid	Ly HL PAGI	ater deta by changing hex code.
		8	-> -===	ry peres		()
			()	at : 1 aurn pars	Dieplayed meter	data 4)
		•		NONE		<b>1</b> 2
		0 0	05.0	9C, D2,	. U L	
		APPE	NDLX	Page 21	- 125	

0110		DEDNIL I			
0110	3.0	USPAC.	1.0	A M	1 Decelow floor but a bullet
0135	CD 0142		C 1 1 1	05810	, utapiev tirst dyte dy neg
0141	10		10	A 1	) 1 Dicplay second byte by bey
			2.6	A15	" archima second offe oa wea
0142		05910:			
0142	F 5	027101	PUSB	AF	· Save entry data
0143	0 P		BRCA		1 Move MSR J har to 15h J his
0144	0 P		BBCA		i iore inter a pit to fin a pit
0115	0 6		RRCA		
0146	U F		RRCA		
0147	CD 0148		CALL	0 S P 2 0	Convert binary to her and display is
0144	P 1		POP	A.P.	; Restore entry data (Use 158 4 bits)
		:			
0 2 4 8		0 S P 2 0 :			
8 1 1 0	26 08		AND	0.6.8	Seglect SSB 4 bits
0140	C6 80		ADD	A , 9 D H	00008 11118 convert 0 F
0147	27		D A A		1
0150	C E 4 D		ADC	A,408	1
0 ) 5 2	5 )		D A A		1
0153	CJ 015D		J P	SCONDUT	Displey it
				CONSOLE ONE CE	A 12 CODE
			SOTE:		
				Sove the cursor	position to the top of pest lipe.
		8			• • • • • • • • • • • • • • • • • • • •
			() entr	y parameter	4.3
		-		NOFE	
		1	() cetu	rn parameter	4.3
		-		NONE	
		;	<) pree	erved registers	6.3
		;		BC,DE,ML	
		;			
U150		CBLP:			
0.15			LD	A , C B	; Carviage return
0:*	CD 3150		CALL	SCONDUT	
51.8	38 0.4		10	ALL	; Line feed
8 3		209001			
1.24			PUSH	86	, 3878 Telletere
0156	10.5		0050 PU20		
0150			1030	пь. С. А.	l de la constante de
0100	CO 200C		CATI	CON007	9 A
0163	r)		POP	NI NI	: : Restore cadiater#
0165	DI		POP	0.8	1 KBRDEBED
0166	<u> </u>		P D 9	ac	
0167	23		867		7° 4
			IND		

		; ••			
		i	8105	ICCARD SAN	IPLE PROCRAM
		; ••			
		;			
		- : NO	T.R.:		
			This	seenie oro	drog pools the data of the
				seepic pro	grow reade the date of IC-card.
			essentre c	01014100	15
		i .			
		. 2	80		
		8			
		; 0	loading ad	dress	4.3
		1			
			ASP 1000		
			constant v	=1400	••
			8103	8817188	
		•			
6803		MECOT	E Q U	059039	
8906		C 0 N S 7	ECU	023068	
6908		CONIN	5 G U	860830	
2962		CONDUT	2 Q U	0280C8	
4043		PUTEFE	2 Q U	028678	
2999		ICCABD	EQU	023998	
		1			
		7	Swata	a work add	P84488
				0.01010	
F 10J		ICCOVAN		001000	
		161501		071298	
<b>FEG4</b>		829313	200	0 F 6 0 4 8	
P 5 7 2		ABCONNER	IQU	075728	
		;	Other	CONSTANTS	
100.		3A133P	800	010008	
0.0.0			2011	0.4.8	
0007		C 8	100	0.0.0	
0000				000	
0000		PRON	rac	0008	
0049		TSKON	zeu	0788	
		1		-	
		; •••			******************************
		:	SALAE	PROCEAS	
		; •••			******************************
0100		START:			
0100	31 1000			NED	· Set stack counter
0103	CD 0164				, for default her table
0105		LAL	L BETSEI		, get cerauit sey table.
0100		C 3 1	I ICULUS		; close ic-care
0109	CD 0398	CAL	L ICOPES	4	; Open IC-card
010C		LOOP:			
910 <b>C</b>	CD 0138	CAI	L DIVEL	7 2	; Seod command or onta
0107	38 31	<b>מ</b> ג און און א	C. STO!	P	. Seco ettor
3111	01 0000	1.0	80.200	1 C E	Vain for some time
0114		100800-	,-		
0114	0.8	_00700,	ar .		
8115	3.6	DEC			
41.3	73	10	A , D		1
3413	51	0 2	L		
2117	20 78		SZ,100		a 4
2119		L00910:			
0119	CD 2806	CAL	I CONST		; Check inputted-key status
0110	30	180	A		:
0110	25 13		2.570	P	: ADY KOV 10
0117	CD 02FA		1 1657A	rus	Check received status
2122	30				ADV data recorved?
2113	20 84	1.50			
21.74	*** **		24140		
1 1 1 1 1	CE 2111	FOONSE:		n	- The Starlay the recorded terms
		C.A.:			
3125	10 00 C	28	C.5101		1 30 VIIV
0178	CJ UZYA	CAL	L ICSTA	103	TUNCE LOCATAON ALALAZ
3753	2 C	: > 0	. A		T. TUA. CUTO ISCATAGO
C 1 7 2	25 #5	18	2,100	P Z 0	1 7 e s
0123	15 DA	2.8	100P		a b
		1			
2122		STOP			
0112	CD 8809	C 1 1	L CONTR		; loput any key.
0115	() ()()		NROOT		
	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	47			-

÷ 1 SESD DATA TO IC-CARD NOTE: Send compand to 1C-card () entry personner () (CNDPNT) -- Sending data pointer (CNDPNT) -- 20000000 () return parameter () [CNDPNT) -- Next pointer CT : Return information >0 -- Normal and =1 -- Error or and of data NONE 0138 DTWRITE: 26 0220 1.5 4138 BL, (CHOPNE) ; Command data pointer ; Get data counter 0138 42 t b C.(81) 0130 79 LD. A,C ; Check and of data 0135 87 0.8 . 0132 37 SCF tnd of date 0137 C 8 z 8 I T : 0140 zs PUSN 8 L 0141 23 INC PUSE NL Data top Address H Z 0143 06 00 1.0 8,008 ; Set next data pointer 0145 0.9 ADD HL, BC 0146 22 0220 LD (CNDPST).BL 0149 8.1 POP 81 CD 02EA 1048178 0144 CALL Sond the data 81 0140 21 POP 0142 15 PUSH .... ; Save return parameter CD 0171 014 2 DSPD7 CALL Display the data 7) C9 0152 POP ... 0153 RET ÷. ; BECEIVE DAT, FROM IC-CAID -NOTE: 2 Bezeive data or status from IC-card i () entry parameter NCNE د ۽ 2 5 constance registers . SONZ 0154 CTELAD . 015 4 24 3144 12 S1. (DATARNT) ; Set received data storing area 91 1121AT 0157 22 1 N C : CA11 117 1 = 1 2ead received date 3158 25 \*\*\* SLI.I SLI.I SCI.I TORZO 10 10 1 015C Set Eaceived data count 0157 0160 22 0171 1140 Display the data 217 i i SET DEPACLY LET TABLE SOTE Set default hay table for EBT-10 >>> +ctty\_patameter \$205 • ) () tetru leroparez () // preserved registers // NONT 0164 EFYSET: 3 A FEF2 07 ; 15 2H1-10, then define the key-code ; at the top of the screen. 0164 LD A, IYKCOUNTRY I 0157 BLCA 8C,0101H 01 0101 10 0168 (Key code) (Xey define) A, OJH NC, PUTPFK 0168 30 31 01 8868 CALL 0160 BET 0170 C S

1 ÷ \*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\* DISPLAY THE DATA ÷ NOTE: Display the data by her code c) entry parameter ci HL ; Data address ÷ ••• â First byte is data length () return parameter 2 < 3 BONE () preserved registers () BC,DI,ML ; . 0171 DSPDT: 0171 85 05 PUSN н 1 ; Sava registers PUSH 3 0 8C 8,(81) N1 0113 C S 9 U S 8 ÷ LD Display data counter 0174 4.6 INC 0175 ; Data top address 2.3 DSPDTIC 0176 A,(HL) DSPHEX ; Get display data ; Convert to hea code & diaglay it ; Next data pointer t n 0176 78 CALL 0177 017A CD 0184 N1 DSPOTIO 23 DJAZ 10 19 0178 5 ÷ CALL C317 ; Carriage return & line feed ; Restore registers 0170 CD 0198 2 O P 80 C 1 D I 0160 0181 POP POP DE 0152 **E** 1 N L 827 0193 C 9 i -----• DISPLAY THE DATA BY BEE CODE i 1 NOTE: 1 Oisplay HL register data by changing hes code. i i () antry pacometer ٤) A : Displayed date () return persector () : 1 NONE () preserved registers EC, DE, Mt 63 2 SPHER: 0154 PUSH : Seve entry deta : Nove SSB 4 bit to LSB 4 bit 0154 P 5 .... J 1 8 5 0 1 5 6 07 07 0157 0 7 11CA .... 0199 0.5 Convert binary to ber and display it is sentore entry data (Ese 128 - Site) CALL POP 3199 0310 03 DSPZO A T 0180 7) 05920: 0180 Neglect 356 4 bits 00000 -- 11118 convert 0 -- 1 A50 0 2 9 0110 16 31 66 90 808,8 5 D D 27 CR 40 6191 0.4.4 100 R01,4 0192 9194 27 SCOSOUT 3.9 Display it C3 0197 0195 : \*\*\*\*\*\*\*\*\*\* ì CONSOLI DUT CE & LI CODE ÷ i SOTE: Nove the cursor position to the top of nest line. Sone ; c) return parameter NOSE ()NONE () preserved resisters () 2C.D2.VL 3195 CALE: ; Carriage return A,CR SCONDUT 0195 10 30 36 CALL CD 019P 019A ; Line feed 38 0A LD A, LF 0190 0198 SCONOUT ; Save registars 0191 C S PUSH 8 C 3.0 PUSH PUSH 0140 05 NL LACD 25 0345 4.2 LU CALL C , A CONOUT CD EROC 01A3 Restore registers H L D B POP 0146 E 1 0147 01 **90** P 8 C 8 A 1 0 C 3 POP 8 E T Page 21 - 129 APPENDIX

		1			
DIAA	01AC	DATAPST	t	0.0	OUFF
OLAC		8077:		D S	128
0225	0555	CHOPNE:		D.¥	ICCNDTB
		:			
		-			
				IC-CARD	COMMAND TABLE
				SESSERAS.	
			•		
		•			
		:	F0110¥1	ng data	la cossende for lC-card.
		1			
0222		ICCHDI8	:		
		LIST			
		1			
					***************
				OPER 1C	
					***************************************
		:			
		i 10	SOT:		
		:		This to	utine opens the IC-card.
		:		Paramet	ers for communication to IC-card are
		÷		add 0380	. 5 bits, even parity 1 stop bit in
		-		default	VALUE, So if communication personators
					tch to [Fight you should [CODDE as
				Lev like	cen to re-cara, you change recutan as
		•		you 1120	
		i i		AND 11	reset process of system is unmatch to
		:		your IC-	card, you estend it by using IC-card's
		1		hook. Po	r example, the cause of waiting too
		1		short, a	naver-te-reset being unmatch, or etc.
				This sa	aple routine changes to non perity.
					· · · · · · · · · · · · · · · · · · ·
		:	C) entr	y parame	ter ()
		-		NONE	
		1	()		eter ()
				NT AD	EMER-TO-PREET data tetting area ton add.
				69 1 84	
		1			
		:			al Koreany opened
		:			
					A reg. im error codm.
		•			•1 : VATABELET OTTUE
		•			4 ; Alrendy opened
		ī			J : Not opened
		1			4 : Force stop
		:			5 : Beceive buffer overflow
		;			6 : Tien out
		1			7 ; Betry error
					S : Communication Actor
		1			9 : Power off stop
		1			A : 10-cordin core in company
		1			
		i.			B : 30FIAL Alfendy used
		:			C : [C-card used by disk
		•	C) JIES	erved re	Sisters ()
		;		SONE	
		:			
0295		ICOPES:			
		;			
		;	lf yau	Vant to	execute by non parity, you must intert
			Dext fou	E STATON	ents in this program.
		:	-72efaul	1 is eve	C DATIES-)
		:*	10	A, PNON	; Set parity to non perity
		; •	2.0	(ICCDPS:	5+21,A
		1.	13	ALISNON	Change IS-byte checking sata
		-	1.3	I LCTSET	1.A 2
0228	CD 0308	4	CALL	10845	Doen IC-card
0298	63		877		
		4		CTOSE T	C-C:RD
		1			
		1			
			2016-	****	waves stars the 15 deed
				2015 201	AFFLE FTDREB TOA TFLEBED
		1	() BRTT	y garawe.	1 E E E E E E E E E E E E E E E E E E E
		1		AUXE	
		1	· · · · · · · · · · · · · · · · · · ·	an parami	Ermi ()
				3038	
		1	1) pres	erved ze:	SIAINE / P
		1		SOSE	
43.45		1			
		ICCLOSE	1		
1000	CA 030C		CALL	ICLOSE	; Close IC-card
0262	C 9		867		i

			**********************************
		READ FROM IC-	CABD
	*****	************	
b B			
	NOTE:		
		This routine	reade data from the JC-card,
		INTS FOUTINE	teres received data and set thes
		to appointed a	res. It judges the and of data
2			And anter to because where the
		he close the I	for error is naggened, you sust forest and start from (sitis)
		aten. Aut when	vou costinue the costation vou
		augt reset err	or-flas because ICCARD's READ
		routine reads	no data during error-flag being on-
;	() ent	ry parameter	6.3
:		BL : Received	data storing area top address
:	() 288	urn persector	4.3
:		CT : Belura 1	nformation
		» 0 — -	Normally opened
		BC	reg. is received data count.
		A	error
		A 7	all restor code.
1			- i rermoster erfog 2 : Alreedy ones-4
			3 that assad
•			
			1 : Barnive buffer overflow
			6 : Time out
			7 ; Batry error
			5 : Communication error
:			9 : Power off stop
			A ; 1C-cord's come is opened
:			B ; Secial already used
8			C ; IC-card used by disk
	< pre	derved reqister	· 9 4 3
0) 0000	10	8C.0000H	Data counter initialize
01 0000			
101	810:		
25	PUSM	H L	; Save date pointer
C 5	8 U S N	BC .	; Save data counter
CD 0310	TALL	TBEAD	; Sead data by 1 byts
C 1	POP	BC	
E 1	POP	HL	;
38 01	75	C'1CB750	; Bead error
11	10	()),A	; Store the read data
12	180	H L	; Pointer up
0.3	150	BC	Cosnier up
19 72	3.8	10010	,
ICA	1010		· Save error status
	1.0	. /	· Reset error ()an
PE BP	LU	100011118	: illeget perity, fleming, aver-run;
5.0 JT 39 9684	19	(899555).5	:
8	FOP	AP	lectore error flag
PE 08	C P	068	; Check error status
37	567		
C 0	987	NZ	Not time out erfor
15	10	3,8	; Check read data counter
31	3.8	c	i
38 06	10	A , 063	i B
11			
3 •	≦ C 7		
6 5	4 3 2 R 2 7	2	) No data received
C <u>5</u> A P	402 119 RCZ	2	Xo data :=celved

0 5 3 0 2 C 3

02DF 02E0 02E1 02E2 02E3 02E4 02E6 02E7 02E9 02E9

			NOTE; () ant () ret	This routine we ry parameter HL : Data top a BC : Data count urn parameter CY : Seturn iof •0 9 •1 8 A teg	<pre>item the data into IC-cerd. () iddrema for writing () iddrema for writing () iddremation iddremat</pre>
			tr pre	NONZ	
022A 022A 032B 022C 022D 02P0 02P1 02P1	25 C9 42 CD 0314 C1 21 21 21	1C#814	2: PUSN PUSN LD CALL POP POP B2f	H L B C I W Z I Y Z C B L C	: Save data pointer : Save data counter : Get writing data : Write it to IC-card : : : Etror
0 2 7 3 0 2 7 4 0 2 7 5 0 - F 6 0 2 7 7 0 2 7 %	23 08 58 81 68 18 PO		1 N C D E C 1 U 0 8 8 6 7 3 R	NL BC 4,8 C Z ICVRITZ	: Data pointer up : Deta counte, down : Check and of data : : Rod of data (Norma) zeturn) : Loop until data and
			<pre></pre>	This routine re JC-card. If error is hap atatus for nest ry parameter CT : 2eturn in (0 N A red (0 S (1 S A red (2 - S) (2 - S) (3 - S) (4 - S) (4 - S) (5 - S	ATUS ade the received status from speced, you must remet error access. () ormatico ormally opened () is received data status 8 No data in received buffer () is received data rount (bytec) req. is only active is Axffs) () () is error code. () is parameter error () is error code. () is serior code. () is ser
0 2 7 A 0 2 F A 0 2 F D 0 2 7 E 0 3 0 1 0 3 0 3 0 3 0 6 0 3 0 7	CD 0319 P5 3A P604 16 Af 32 F604 F1 C 9	1 C S T A T	US: CALL PUSN LO AND LD POP RET	15TATUS AF A.(@SPSTS) 100011118 (RSPSTS),A AP	; Read ]C-cord's status ; Save error status ; Beast error flag ; IBeset parity, flaming, over-runi ; Restore error flag

					8105 10	CABD ACCE	SS ROUTINE
							***************************************
			8		These e	outines c	ell BIOS ICCARD.
				() entry		lar	
			;		Depend	on each f	unction
			1		n pares	eter	()
			;		CA : 84	turn info	restion
			:			=0 No	really opened
			•			Other	registers depend on each function
						*1 Er	tor
			* :			A CURS	=) : Paramatar aryor
			•				2 : Altendy opened
			: .				3 : Not opened
			;				4 : force stop
			:				S : Receive buffer overflow
							6 : Time out
			•				<pre>/ : delry error A : Computering</pre>
			:				9 : Power off ston
			;				A : IC-card's case is opened
			1				8 : Sevial already used
			;				C : IC-card used by disk
			:	4) press	rved re;	glatere	¢ >
					NONE		
0306			; IODEN:				
3308	38.00		TUPEN	f D	A 0.0 H		: 0000 16-co.td
0304	18 02			28	ICCALL		; open it-card
0 3 0 C			101050:				•
0300	32 01			10	A.018		; Close If-card
0308	18 0.			18	ICCALL		;
0310			IBEAD:				
0310	38 02			LD	A,028		; Read from 1C-card
0312	18 06			3 8	ICCALL		;
0314	38.03		Iwella.	1 1	A . 0 3 H		· NELLA 1010 IFerard
0316	18 07			3 9	ICCALL		
0318			ISTATOS:				•
0318	32 04			LD	A,048		; Sead 1C-card's status
031 🔺	CD 1889		ICCALL:	CALL	ICCABD		;
0310	C 9			827			;
Necros:							
Symbols	:	499-					
		0222			5008	CR	
0190		0144	DATAPNT		0185	05720	
0171	DSPDT	0176	DSPDTIO		0184	DSPORX	
0154	DTBEAD	0136	DTERITS		031.4	ICCALL	
2899	JCCARD	F183	1000985		0287	ICCLOSE	
0222	1008078	0200	101052		0288	ICOPIN	
0268	102010	0204	1 C 2 D 2 O		0263	ICBEAD	
0294	ICSTATUS	1129	ICTSDT		032A	ICVEITE	
0306	IOPEN	0310	CATEL		0312	1574705	
0100	IVEITE	0154	LETSET		0129	100910	
0106	100820	1000	MAINER		0000	PNOF	
7867	PC-P98	2604	95P575		9191	SCONDUT	
0100	START	0132	5102		0078	TSNON	
2802	VBCOT	7572	TRCOUNTS	T			

No Petal error(a)

## (24) SAMPLE 24. User BIOS Area

		;				
			18832U	OS SAMPLE I	PROCRAM ••••••	• •
		SOT	<b>r</b> ·			
		5	TPIN	sample proj	gree shows how	to make USEBBIOS-ABEA.
		4.3		ndition	¢ 3	
		4	. 280			
			hha anthan	<b>5886</b>	< b	
		P				
		;			1.0.0 M	
		:	constant va	1.4	4 >	
		; • 8	JOS entry p	oint )		
E 803		N 80 07	EQU	0 E B O 3 B 0 E B O C M	; N8001 : CONON	entry address. T entry address.
2839 2839		BREP	EQU	028398	822P	entry address.
2869		CALLX	EGU	0296314	; CALLA	antry address.
		; • •	ystem servi	ce routine	s )	
0013		BIOSITEO	20U	AC100	; 88195 : Set 6	loader. ystom permeter.
0016		CHCBAND	EQU	00)9N	; Chang	a and disk address.
			yetes Area	define →		
		STERAM	EQU	070088	: SLADD	ard SAT also of RAN diab.
F00C		COLERICO	290	0700CU 07060H	; US288 : RAM d	105 area sí"e. Iok síze.
2080 7085		AD_RAM_IN	ZQU	010629	Start	eddress of EAM
		AD_BAH_OLD	280	0 2 0 7 2 3	: Start	address of BAN disk. (old)
8904		R A M D _ 9 1 2 2 D 1 3 2 N K	E Q U E Q U	000658 074138	; Exten ; Objec	d RAM SIZE t benk for CALLX.
1415						
		: • L	5298105 AP	a define +		
0000		LETOTION	2 Q U	000008		USERBIOS area bottom
5.8.2.O		HZADTOP	200	CEBOTTON	- )08	; Neader top address.
0 8 7 0		9010	29 U 29 U	HEADTOP	02R	; ( 2 ) ( 0 ; ( 5 ) Program name.
0872 0871		e 1 1 1	EGU	STADTOP.	0 4 8	(1) Program size.
23750		TEAVE DARLS	EGU	BEADTOP.	0 C 16	(12) Beless program
		TRESE	200	SEADTOP.	0 Z F	; address. ; ]; unused byte.
3878		C8#5CM	293	984DT08+	0 7 11	;[1] check sum
0010		2235128	200	16		: USERBIOS Area Bire = 46
						necesser]
0060		407219 <u>8</u> 207378	19C 20C	UBBOTTON UBBOTTON	- 1008 - 8185] ZX = 256	; Belasse progam ACdress. ; Program start Address.
0050		23252	ZOU	508		; >41 60
			tispely deta	1.)		
200C		21.5	EEU	0000		; Clear screen & home.
0013		252	SGU KQU	C 1 8 8 G 2 N		: ISC sequence. : Caridge tetutn.
3004		17	sev	N & D		: line feed.
				loutine		
		: • • • •				
0120		\'81CS:				
0100	31 0254	LO	SP,STA	CHAD	; Set stack pou	nter.
0103	CD 01C0	CAI	LL CHE_HI	0.41	; Check existen	ce of header.
0106	20 13	91	AZ, 814	<b>N B 4 0</b>	; Juep 14 0084	NOT 6310E.
0106	CD 0105	C & 1	LT CHRTAN	5 R 6 2 0	; Check used to ; Juep of used	nule :
0103	40 40			VRT 1	: Chack overver	te flan.
010D 0110	JA 0378 87	0	A		. id access	
0111	28 27	1.	Z, 2881	C 71 0	; is connot ave	rwrite ineu jueb.
01)3	21 0118	SEVOSO: Tu	N.L	A D 4 O	; Set return ad	dress.
0113	4 8 W 8 8 G					

0116	25		PUSH	8 L	;
0117	ZA DBEC		10	NL, (BESAD)	Do leresse process.
0114	ES		18	(NL)	
0118		HEAD40:			
0118	CO 0142		CALL	UBMAKE	hake VEIBBIOS Area.
0116	A P		20	at , the the	
0120	CD 01A9		CALL	UBLOAD	load object program.
0123	CD 0354		CALL	OPNUBLOS	Open process for USERBIOS area.
0129	21 01F2		LD	HL, HSCOK	Set O.K. easeage address,
0120	01 0010		εņ	9 <b>C</b> ,001GM	Wait 1.5sec.
0127		NAIN_BN	D ;		
0127	C5 CD 0176		PUSH	BC PRMSC	Save. Digolav meseage.
0133	c1		POP	80	Restore,
0134	CD 2838		CALL 1P	8229 Nagoz	BUZZEC ON 197's PELY PROM THIS DECERAM I
0121	C3 1993		••		all 5 carr raon lars recover,
0136	23.0307	ENDEND:	1.0		· Sat artas assess address
0130	01 2210		10	8C,22108	; 822F (880WZ, 1.658C.)
0140	18 20		18	HAIN_END	:
		;			
				8 80188920 afa <i>n</i>	) F ØÅ ) y y ø ø ø ø ø v v ø ø
			<) ret	uro parameter ()	at and
	-	:		Carry OFF; Brrd	br end.
0142		USHARE:			
0142	3A 700C		La	A. (DSEBBIOS)	; Check wile of USEBBIOS area migs.
145	30 02		2 P J B	NC.13NHE10	: Jued 10 OK
••••					
0149	34 2008		1 11		· Chack any size is evalable as our
0140	07		BLCA	81(314883)	+2 (/180> /512b)
0140	C6 08		4 D D	A, BUBSIZE/2	
0147			821	MAISI XC	; ; If not everiable.
					-
0153	32 10 32 700C		1D 17	A, BUBSIZE (NSEBBIOS). A	Set dew USEBBIOS Area elze.
0150				(	•
0157	D.D. 7.1 0016	CB4K10:			· Cat has everal nationalists
0158	CD 017A		CALL	JCALLE	
0112					- antand DAM size
0195	41		10	B.A	, FILEDE FAN EIZE.
0162	34 9080		LD	A.(QT_RAS_1N)	; Total EaS disk eize.
0165	- 7 A 7		13	E, A A	: No format.
0167	DD 21 0019		10	IX, CECBAND	; Set new BAH diek gize.
0168	CD 017A CD 0154		CALL	JCALLE	- Relocate RAM disk eres
0171	DD 21 0013		10	11, 81053710	Load B105 jump table.
0175	CD 017A		CALL	JCALLX	
0179	C 9		5 C F 8 <b>F</b> T		
C17A 017A	75	JEALLE:	P./' S.H	A.7	: Save
317B	32 PT		12	A. 0 F 7 B	; Set object benk data, (system bank)
0170	22 F418		10	DISENK),A	Featore
0151	CJ 2869		19	CALLY	
				th EAS stepping	sk
				*****	
			SOTE :		
				This program's	function is to relocate BAH disk contens
		:		according to ne	w RAN djeb to address.
					$\mathbf{O}$
		- 1 87		NONE	
			<pre>c) ret</pre>	Urn gereester NGNS	()
		;		17 W FT W	
0177		:			
0184	JA FOOB	AELUL:	LD	A. (5128A/1)	; Get BAR disk size of standard part.
0187	87		08	A	1 f = 1 g = 00A
0186	C 0		= E T	z	τ <b>υθη Γατωτώ</b> .
		APPE	NDLX	Page 21 - 135	

0189	07		8104		1
0184	07		BLCA		/lKbyte> /lbyte
0180	00 00		10	5 A	
			20	C,U	
0182	2 A P072		10	HL, (AD_RAN_OLD)	] ] Old RAM disk top address.
0191	KO 28 KO62		10	DE, (AD_RAH_EN)	: Nev SAM disk top address.
0195	25		PUSH	82	Save.
0196	87		0 R	A	Clear carry flag.
0197	ED 52		5 B C	н., де	Check transmit direction.
0194	E 1		POP	H L	. Bestore
UTJA	60			Z	If dose not change then return.
0198	30 33		38	C. BELOCID	i jump if old top adress ( een top
					address.
0190	20 80		LDIR		E Belocate
0136	Ca		821		1
0140		Attoc1	0:		
0140	0.8		DEC	B C	
GIAI	0 9		A P D	NL.PC	; Hireg.:bottom address of cid.
0142	2 B		E 8	D2,8L	
0144	2 S			N C . 8C	i Deed thattan oddaean of any
0145	0.3		ISC	80	
9 4 1 0	88 03		LDDB		Belocate.
0146	C 9		827		;
		1			
				*************	
				Load object pro	
			NOTE	1	
		ê		This progers 1	a for loading the programm to USER \$105
JIA9			:		
3149	21 3284		LD	HL, UBYOP	Load the user program.
0110			1.10		
0182	2D 80		1010	CC   BEATLE	
0284	21 0354		1.2	81,81570P	Lond the release-program
0;87	11 0800		12	DI, EBLSTOP	
0130	20 BG			80,803112	,
0138	29		821		
					a of dSTR-8105-824028
				ntty petameter ()	
				NOKE	
		a 1			
				2flag ON : Hea	der eziets
				Zfleg OFF : Res	der dose not exist
2100	21.2868	226_3E.	AGC .	NI SELOTOR	,
0102	36 40		1.2	3.138	Length of USE8-BIOS-MEADER
0101	<u>ь Г</u>		208	<b>A</b>	Areg. + 00H
· 3		-25_54°	21 : 		t fur checking
0126	2.6		A 3 U - K P	8 ( B L ) 2 1	' SAM CUNCTING.
0101	10 50		1252	CBN_SUS	
0154	87		38	A	tesult is CON ?
C1C8	C 0		282	N 2	; So:
					· · · · · · · · · · · · · · · · · · ·
01-10	IA 3850		10	HL, (HEADTOP)	; LNECA IU CODE ("UB")
0107	11 4455		5.84	NE, DE	
	C 9		811		

........... Check used module name 1 i entry parameter ()
NON ÷ 2 0105 CHE\_USA: 21 0276 J1 0892 06 08 NL, UBNANE SE, SAME 0185 : New module name. : Current module name : Jongth. ιο 10 0108 LD 8,08H 0100 UB\_CHN: 0100 1.4 17 A. ( DE ) ; Check sodule name. 38 IULI C P 0 1 D P C O 821 NZ : 0130 150 23 н х ÷ 13 DE 0122 13 19 DJNZ па сяк ; 108 ۵ 0185 C 9 827 -----; Diepaly a cosego () entry paremeter () RLreg. ; Start address of a seconds. The essence sust be terminated by "OOH" ÷ ÷ . : 0126 innse: 34(0 7.8 1.0 A. (8L) ; Bead display deca. 0127 23 1.10 LL 0122 8.7 0.8 A Z 1119 RIT 63 Ind of data. ÷ 0 1 2 A 0 1 2 B C . A N L 4.7 6 D 25 PGSR 0120 CD 180C CALL CONOUT Display it. 1 21 4 D S 8 L 0170 PBNSC 1. 18 . ÷ Jeesage area i 0112 15005 cls.cl.LP ESC.'2' ' Completed' 01/2 00 00 04 38 18 22 20 43 6P 6D 70 6C 65 74 0175 38 35 0178 0177 65 64 0201 0.0 003 0102 0C 0D 0A 18 23 43 51 6E 6E 6F 74 20 60 8F 75 6E 74 CIS,CB,IJ ISC,'2' 'Cannot aount' 0232 38 28 0207 38 0208 0207 0213 0.0 008 stack spea define > 0214 29 160 428 1 STACSAE 3254 ............................. : 

	; Open porcess for USESS area							
	;							
0254	OPNUBIOS:							
	<pre>************************************</pre>							
0254 C9	:							

0254

• • •			• •						•	• •								. 19	۰							• •		
			9	e t	1	1 8		d _	P	6	• ť	- 1	5	it	8	- 8	1 6	0	s	-	<b>A</b>		8 4					
		•			•		•	• •	•	• •			•		•	•	•			•	•	•	• •	•			•	
			м					• •	_																			
			y N	0 N	ĩ				E.																			
• •	t e		r n					e t				,																
			N	0 N	2																							

0255		MAKENEAD:		
0255	21 0800	10	NL, BELSTOP	; Set address of release couting.
0258	22 6250	۵.۳	(REL_IC), HL	1
0258	21 0274	1.0	HL, BEAD_INP	; Colcu_oto chack sum da.m.
023E	06 07	LD	3.078	; hender mirn.
0260	A P	209	A	Areg. = 001.
0261		CAL_SUM:		
6261	96	508	(BL)	:
0282	23	180	84	Sum check.
0263	10	D J N 2	CAL_SUN '	
0265	32 0293	10	(DB_CHI_STN),A	; set check sum data.
0268	21 0274	LD	BL, BEAD_INF	; Copy new header.
0269	11 0873	1.0	DE, HEAGTOP	:
0262	01 0010	2.0	8C,108	;
0271	10 90	LDIB		
0273	C 9	BET		
0274	55 42	dZAD_INE:	32 1181	IE OT BEAGET.
0276	54 45 23 54	CSNTR:	08 725771	BOC' ; NAME of program. (Please
0274	50 52 4P 47			, change to
	1.0			; CHR DAGGI
UZIE	10		53 6633120	z j size oz program, (126 ayrm
				; 30000ETY1
U 4 7 8		201 10.		, uvervelse ilæg-
0250		att_it:	25 2	access c: 1818860 FOULING.
0252	00		57 UU5	
0253		CHTCRRTRCH:	2.3	; CRECE ECE 1818.

User program define

LOU I . DEFEASE

		. PHASE		RUSTOP
0284	UBTOP	εQU	PAUSE	
6600		0 \$	1000	; ( DUNHA )

25058

				9 • • • • • • • • • • • • • • • • • • •	IP OWN FOUT	ing that will be relecated . age .
0100			UBSIZE	2 Q U	S-RUBTOP	
				. DEPHAS	5.2	
				Relese	routine	
			This :	routine wi	II be move	nted on HOOK ABEA.
0353					SPLSTOP	
0304			BLSTOP	EQU	PAUSE+UNS	5128
			;			
			1 · · · · · · · · · · · · · · · · · · ·		wies book	· · · · · · · · · · · · · · · · · · ·
			; in this cou	iline.		6
			:•			•
			; C.Nead	der null (	clear )	
0080	06 10		LD	8,10H		Counter set.
0802	31 Depo		1 D	BL,HEAD	D T G P	; Reader top address.
D 8 0 5			REL_LOOP:			
0835	35 66		LD	(NL).00	3 M	Bull cleater
0808	10 98		DINZ	RE1_100	D P	
D 8 0 4	C 9		111			
8 0 0 Ø				EQU	S-BRLSTO	9
					5	
			END			
JUCEOU.						
Symbols:			AD RAW 013	7819		
0013	BIOSITLD	1841	CALLY	0261	CAL_SUN	
3019	C B G B A N D	D	TBESTR	3100	CAIE_383	
	CARLICA	0000	C3	7413	2158NE	
0134	TRRESD	0018	250	2970	1010	
3112	BEADIO	0118	JCALIX	300A	LF	
0129	SALN_END	0255	"AEREEAD	3029	32245	
0202	352828	0122	NSCOE OVWRT	3872	NASE 7acse	
0126	PRNSC	1040	07_CA5_17	r 0 6 5	212_0EA5	1
0154	RELOC	0140	SELOCIO	5220	321_IC	
2384	BELLOOP	0 B F C	RELSTOP	0010	REBRICE	
cc00	RUBTOP	0016	SETEANAD	2874	\$122	
6001	SIZEAN	0254	STACKAD TRLCAD	5030	LOUTION CREAKE	
0157	C2103 C5%K10	0276	CUNADE	2100	2851II	
0254	10753	0000	19_190 	2162	TA_CME_S	C 3
3996	UNCSE	700C	CREASION	1363	-1001	

So fatal eccorial
		:				
				AUTO P	0VE2 077	SANFLE PROCRAN
		:	*****		*******	*****************************
		:				
			NOTE:	-		
				1012 00	aple prog	ras gets the sey-in wath and
						r-ort time.
			12	eeble co	ndition	()
			. 280			
			() 108	ding and	/	()
		•	- PHASE	1000		
		;				
			() 600	ataot va	1	63
		• •				
£ 8 0 3		wao.	37	260	068038	
2806		CON	57	860	830630	
2063		CON	19	200	023033	
EBOC		KOD	707	EQU	023008	
4377		PUI		reu	069618	
		POW.		FOU	028128	
		1		Lev		
				System	work add	201040
		1				
020		A T 5 .	ILTOPP	EQU	070208	
8021		A75(	373M2	200	070218	
1023		ELO	7 T 1 3 E	reu	070700	
2.2.		1.0	7 6 <b>8</b> 10 <b>0</b>	100	0 7 3 8 7 1	
13-2		TEC		200	0 2 5 7 2 8	
1214		111	CRAJ	Lac	075798	
		:				
		1		Other	constants	
10-2		141	424	tou	6166CH	
		•				,
0001		<b>STO</b> 1	•	EGU	0018	
				DATH P		
		÷			•••••	
		i.				
0100		TTA	17;			
0.00	63 0114		C A 1 1	25.241	226	Set default bey table.
	CB 0114	:				
0166	CD 0122	1001	CALL	52Y13P		Get key-in date.
0106	- F		t D	С.А		1
3104	FE 31		⊆ <b>P</b>	S T O P		le it stop code?
010C	CA 2803		39	5,4800	1	Tes. (End of program)
010	CD 180C		CALL	CONDUT		
0111		:	* #	1007		
			******			
		à		SBT OE	FAULT SET	TABLE
		8 8	******	••••••	••••••	
		a i	SOTE			
		-		Set de	fault bar	table by each machine.
		i	<> ent	LY DATAM	eter	4 9
		-	() 2011	SUNE NES Safa		4.5
		1.0		NONE		
		;	5.2 pp.m.	8 * 7 × * 0 - 7	*#1#1#18	4.3
				SONE		
0114						
0114	34 9572		10	ALITAC	OUST 8 7 1	; Check machine type
3117	0 7		BLCA			; ]e it 187-10
0119	33		12	56 . 311	018	;
0118	21 0173		LD	31.KEY	781	· See all default heve
611 <b>2</b>	34 2267		CALL	50,90Y		,
0121	6.3		# # T			ø

÷ CET KET-IN BATA NOTE: Get bey-in deta from console. This routine uses BIOS CON11, so you can check other statum () entry paremeter 6.3 NONE veturn paremeter A : Key-in deta e ) < ) () preserved registers () NONE PRVINP : 0122 NL, (ATSOTIME) DE, (TIMERO) UL, DE 24 .021 LD set auto power-off time (Stop 1) 0122 20 38 2020 0125 10 0129 ADD 19 (TINEEND), HL 22 75 79 10 ; ιΟ HL, (ELOPTINE) 2.4 1023 ; Set auto backlight-off time (Step 2) 0120 19 AC 0 1.0 HL, DE (ELOPPEND), NL 0130 22 1387 0131 : BETT10: 6134 CD 2806 CALL CONST ; Check console inputted statue 0134 ; logutted any key? ; Yee 3C 28 INC 18 2.827190 0138 ÷ A. LATSHUTOFFI ; Check auto power-off time (Step 5) LD 0134 . Bisable Auto power-off? 87 0130 28 07 1.0 TAS 0131 LD HL. (TINEEND) 0140 24 7579 2 D 58 1020 10 LE, (TIMERO) 0143 014\* 87 39 5 8 C Auto power-off time been reached? H1,D8 10 52 0148 0144 08 01 1 0 C.018 N. PONEROPP CALL Tee (Power off by continue mode) PC 2875 0140 SET120: 0117 BL, (ELOPTINE) ; Check cuto backlight-off time (Step 7) 2A 1023 LD 0152 10 1.0 A . H ī. Disable auto backlight-off? 0.0 ÷ 6113 15 26 07 1. 0.6 EY23. S 145 0154 HL. (PLOPPEND) BR. (TISERO) 26 UF 26 P38" 20 58 P020 Check suto backlight-off time (Step 7) 0156 12 2 1.0 0159 0.8 87 . 0150 81,02 Auto bachlight-off time has been 20 52 580 015E researd? C.008 0110 02 30 10 Tes (Backlight off) CALL S. BACKLIGT TC 2842 0 1 8 2 1 Tou can check snother status. For exemple, check printer-ready-status of receive-interrupt status, etc. ÷ 1 ETTI30: 0185 AALT 7.6 1 0185 011738 15 CC 33 1 0166 ÷ REV192: 0165 C O N 2 N C 3 1 1 Set key-in data. CD 2808 0165 17 C,S18 P.5. PUSH 0188 02 01 12 0160 CALL CD IBA2 BACELIGT Sachlight op 0182 0171 11 .0.2 11 127 0172 6.9 ÷ JEAN TELE 1919 INTERNET ..... Default hey table for INT-19 SETT81: 0173 . 4110 0119. C 7 7 8 3.8 0018. OFFR. 0113 0177 .... 17 17 77 78 ..... n n 0 F F N . 0774 0.5.7.0. ..... 0175 6170 .... 0. ..... 0778. ..... ..... ..... .... 0170 ..... 0131 \*\* \*\* \*\* \*\* 0 F F N . OFF8. 0 F F N . .... 0 7 7 8 0.8 6182 11 0186 ..... **GFTH**, . 6110 ..... ..... 8 0 17 75 17 77 0187 0188 18 ..... 0758. OTPN, 0778. 17 17 17 17 0.8 OF78. 0180 0190 **P F** 0 7 7 8 ..... 10 11 10 11 n e 0 F F H . ...... 6 P 7 N 0181 17 0195 -----DA ..... ..... 077R. DFFN. 0 7 F N 0196 019A

0198		<b>T F</b>	• •	rr	08	0	0 F F H .	07TH _	0 F F N .	0718
019 F										
0140			FT	P	08	OFFN,	OFFN.	0 <b>7 7</b> N ,	0 P 7 N .	075H
0144	8 E									
0145	TI			P F	08	0 <i>5</i> <b>F</b> N .	0 <i>71</i> H .	а <b>гт</b> н,	0 P Y H +	0 7 F N
PALD										
JIAA -	TP			77	08	0 <b>F F N</b> ,	OFTN.	OPFN,	OPPN,	OPTN
0142	E P			- •						
OIAF				F F	08	OFEN.	UIFN.	<b>U 3 P M</b> 1	UPPW,	UPFN
0183			_							
0184	PA	. 6.8	TC	70	08	OFAW,	UFBM,	UPCB,	<b>ЧУДИ</b> ,	area
0189	T E									

(26) SAMPLE 26. BOOS ERROR A.

			NOTE:	SAMPLE This as Its fun able con .280	PROGRAM PRO	9	EBROR EBROR To change BDOS error vector. SETERR.
			() los	TDDe BALD	1009	¢ 3	
			4) CON	*00 ×61		4 3	
2 8 0 3 7 7 8 8		JJUNPIX		50U	077468		; JUTPE address in regident
0010		GOBACE		8 <b>Q</b> U	801000		1 BOGS and process
r418 r3D1 0005		0   5 8 6 K 4 8 6 7 8 0 0 5 2		200 200 200	0 F 4 1 8 8 0 F 3 E 1 8 0 0 0 5 8		; JONDE parameter address. ; BDOS error code save stem. ; BDOS entry address.
		;	•••••	5 e t 8 0 0			••••••••••••••••••••••••••••••••••••••
		•					
0100 0100 0103 0107	21 0310 20 58 0006 13	SETVECT	LD LD 1#C	EL.X888 DS.(300 DE	v <b>tn</b> 92+11		; Sew BBOS error vector address. ; BBDOS1 entry address.
	13 01 0008 20 30 C 3			DE DE 1C,0000	8		Dêreg.; Error vector top. COPT.
			() <b>8</b> 720	De Vestor	Table ()	)	
			:				
0110 0112 0114 0114	0 1 1 9 0 1 1 D 0 1 2 2 0 1 2 7		24 24 24	IBADSEL IBADSEL IBCDSE IBCPILE		3 ad ee 3 ad ee 3 3/0 da 3 /0 fi	ctor error. lect error. sk error. le error.
0115		ISADSEC : • • • • • • • : • Tou : • rout:	con se Los 10	this stor	e BAD 520 2ge.		**************************************
8 E E O A I E O	2601 23012A	: • • • • • • • •	LD ; P	EOBASE	• • • • • • • • •	8et er	••••••••••••••••••••••••••••••••••••••
0110		384D581					
		;+ Tou ;+ Pouts	CAO 90	this stor	π 340 52) A <b>q</b> €.	_KLT 1880	
011D 011F	26 92 C3 0124	; • • • • • •	1.D J.P	9.2 388808		5 et er	zar code
0172		1.00555; 	ceu 20	••••••••••• ••••••••••••••••••••••••••	n 8/0 DI: age.		• • • • • • • • • • • • • • • • • • •
0122	C2 015V 36 03		10	С. И В 0 8 8 3 1		Set er	for code.

.

APPENDIX Page 21 - 143

.

0127		AROFILE: :•••••••••••••••• :• You can be :• Fouting in	this storage,	PILE EBOR recovery
0127	21 0004	to	RL,4	; Set error code.
6174				
0174	28.88	10	1 0 P P H	· Set etter code to estude economic
0120	12 8383	1.0		, set error coos to return persector.
0120			(ABEI],OL	
0158		LD	JI, GUBACE	; BDO3 exit process routing.
0123	32 77	LD	A. 075H	; Set to system bank.
0135	32 7418	LD	(DISBNE).A	:
0138	C3 FT48	9 C	<b>J J U M P X X</b>	JUNPX

END

				SANPLE P	BOCBAN	61712 AO4	981728, 8
				This sas	bje bre	gram show	a how to use SITERS and BSTERS.
		n D	() a.e.e	eble cond	ition	0	
				. z 8 0			
		:	<> load	ling addre	• •	e 3	
				. PHASE 1	008		
		:	() cone	tant valu	• •	•	
1003		NBCOT CallX		2 Q 0 2 Q 0	0 2 9 0 3 m		; SBOOT entry address.
2842		1 > 70 8 1		200	028A58		INFORM entry address.
0004				260			; settau adgesu in syctam ; jusp table.
0000				RGU	0000W		; ESTERE address to system ; juap table.
		à					
		:	******	2008 1 a P	9 7 9 7 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	7 8 C 0 - 6 7 Y	
0,01	DD 21 000A	- ETHODE	. D	I I . S B T Z B			address in system jump table.
01-6	18 04		3.8	C B C Z B B		;	
0116	DD 21 0000	ESTHODE	20	1X,257EB			andress to system jump table.
		CNGEBR:					
0104	02 03	•	LD	C,3		: Cas 01	SBNK address.
0107	SE FF		LJ	A , 0778		; Set sy	stem bank data to DISBKE
0111	77 CD 8869		LD CALL	(BL),A Calle		Do SET	ERE or SSTIRE.
0119	C 9		NET			•	
		1		Check BD	05	r	
		1	******			*******	
0116		C85228:	_				
0116 0137	87 C S		0 8 9 7 T	A 2		; It sor	se] setus3 1300 tetoro.
0115	25		PUSR	81		; save :	eturs ;erabeters,
0119	D 5 C 1		PUSN	8 C			
0118	P 5		2033	A P			
3:1C	11 0120		10	DE JEREVE	8	Erroz	vector top Artress.
	19		100	81.38			ALL ADVART AFTOR SAEdiar address
6122	7 E		LD	A. (HL)		: Careor	afa golett titet menetet
0123	6 6 2 3		1 D	ИL У,(ВL)			
0125	6F 13 012B		10	L,A DI,EBREE	т	) Elzen- 1 Set Pe	turn address.
0124	2 S 9		PCSB JP	0 E ( H L )		CALL .	rror hendler.
						,	
0128		ERBBET:					
0128	73 C 1		POP	A P B C		Restor	. spos return persector.
0120			POP	D 2			
0137	69		RET	Π.		ė	

() Error Vector Table ()

0138	C 8	867
		; • • • • • • • • • • • • • • • • • • •
		;• routing in this strage.
		: You can set your own B/C FILE ERBJE recovery *
0128		
0134	C 9	887
		; • • • • • • • • • • • • • • • • • • •
		; · roution in this strage.
		; · You can set your own #/O DISK EREOB recovery ·
0134		14 C D S K :
0738	C 9	327
		,
		- Fourtine in this areas
		Tou can set your and BAR SPIRCY TROOP
0138		BADSEL
0138	C 9	827
		· · · · · · · · · · · · · · · · · · ·
		· routine in this strage.
		: I Tou can set your own BAD SECTOR BRADE recovery .
0138		BADSEC :
0136	6129	DW ROFILE ; R/O file error.
0134	013A	DW ROBSE ; 8/0 disk error.
0132	8610	BABSEL BAB CALACTOR
0130	0135	DW BADSEC ; Bad sector error.
0130		299473

2 N D

		•••••	A 110 81404	ALABH CONTROL SANDLE PROCEAN
		NOTE	; This program interrupt, ()	shows how to disable and check caused by power off and slars)
			seeble condition	n «v
			. 280	
		: «» la	ading address	
		: · · co	netect velues	< 3
F G 3 4		7 P O P O S	2GU 0703	18 : Power Off Lot. disable flag.
7086		YPOPST YALMOS	EGN	10 : Power OPS jot. status. 98 : ALARM int. disable flag.
F087		YALNST	Edn obor	ELAEM Int. status.
/ ? * * *		5018124L	29U 07790	18 ; Post-Blüß entry (resident ; system)
0100	c3 0137	3 P	5 T A B Y	; Jump to sample routine.
		•••••		
		••••		r GFF & Alere
		FOIL	:	
			THIS Program	
		; () en	NOSE	< p
			XOSE	
		• «> pr	All preserved	ra () 1.
0133		DISABLE:		
0103 0104	85 21 8084	PUSR 10	el Nl, ypopds	; Save register. ; Set power OPP disable bit for user.
0107	C 8 96	SET	6,(8L)	1
0 1 0 9 0 1 0 9	21 2086 C8 26	L D 5 # T	81.VALMD8 6,(81)	Set alarm disable bit for user.
010E 010F	8 1 C 9	POP	81	
			Check Joser.	131 & STOCHTS
		sort	: This parers	's function is cleak and enable Jawer Off
		8 10 11	ANG ALARN IN	- AlaRM ist has occured execute power off
			or display 1	alara sessage.
			SOSE	
			SOST	
			All preserved	3
0110	25	СНК <b>ЈИТ:</b> Рш <b>с</b> м	HE	; Save Peristers.
0111	r 5	N2 U S	5 P	
0113	23 8084 C2 85		8 L , YPOFOS 6 , (ML)	; Seest power Off dishels bit for user-
0117	21 2085	L D	NL, YPOPST	; Power CPP interrupt status bit.
0114	CD 0122	CALL	BITCOPY	Copy user bit to \$105.
0110	21 7086 CB 88	1 D 8 2 S	HL, TALMDS 6. (HL)	Reset siarm disable bit.
		APPENDIX	Page 21 - 147	

0122 0125	2 1 C D	F0 8 1 0 1 2 E			10 Call	#1.TALN #17COPY	12	; Alarm interrupt status bit. ; Copy user bit to stog.
0123					POP	A.F.		; restore registere.
0129	e 1				POP	NL IBSTRIO	•	
0124	CD	FF96					a	; Cell PSTBIOS.
								; is occured,
								the PSTBIOS executes Power OFF
								i or displays
0120	<b>c</b> 1				RET			
0120								1
0122				BITCOPT	1.0	A.(N1)		
0148	2.6				AND	0100000	0.8	; sack out user bit.
0131	07				RLCA			Copy to BIOS bit.
0132	86				450	1011111	18	; Norge, · Annar waam bit
0123	71				1 D	( UL ) . A		; Set new status.
0136	C 1				8 2 T			1
				;	******			
				1		Sample		
					KOTE :			
				1		In th	is progra	m, ALARN A power QFF int. is disabled
						become	a sero.	i e crebreatuñ antil fue roch conutet
E863				CONST		290	028038	; SAUDT BOTFY. : âlos const estry.
8808				CONTN		200		BIOS CONIX entry.
210C				COMOUT		20L	023068	: BIDS CONCUT motery.
8886				YOUCA		260	OXDACS	; #103 focts entry:
0137				START:				
								· ber define
0137	0.7	03			10	C.2		
0130	CD	8898			CALL	TOUCE		
								A
0137		0103		STABTS :	CALL			: Fower GFF & Alarm los.disable.
0137	0.6	0 🛦			LD	8,10		less conster.
0144				L00#2:				AANA 'OTD CODDIAT.
0144	C 3				PC 30			
0145				L00P1:				
0145	()	Z B C 6			CALL	CONST		Check Mey lopor statos.
0145	20	7.4			190	32.100	1	loop of not key in.
01112								
0148	62	2005			CAIL	CONIS		; Get Rey. · If and hey then V1007
2110	- 7 Z - C J	2903			29	2.28007		1 'T AAR Ba' rear -sage
01:3	47				10	C , A		
0334	د ع	2305			CALL	130403		display inputted tade.
5197	- 51				POP	10033		leon
0135	- 0				0452	10012		1 1000
0 3 5 A	62	2113			CALL	CAE131		: Clacs jower off and alars int.
								I I INT. AND ANTONRY DESUIC, TEAS
- 1 4 -	11	3127			19	\$11872		1 1 1 2 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2
101.4								
0.60			••	ATACTSE			41m	
0160	10	05 05	19		0.8	7,2,2,7	, 0, T, N, T, O,	TOACT WER
6169	÷ £	10 35	54					
3186	2.3	18 45	5.9					
2170	16	07 05	10		0.8	1,7,5,1	,eam,1,0,	6 N U
0174	20	20 45	4 8					
0170	4.4	20 20	2 0					
					EN D			

(28) SAMPLE 28. Mount Check of the cartridge device

				CRCNOC	E SAMPLE PROCES	n • • • • • • • • •	
			NOTE				
		-		This This	program shows h program is cons	ow to use Ci tructed of r	IGNOOK. Next J modules.
					). CECEGOL es 2. Setup modu 3. Ecleman mo	pand module. Je.(CRESITU dule.(CRERIS	. (CRCPROC) 9) 9)
				if ya execut	e use uses alos e usios program	area for the	nam program, you must ter 1.61
		0 10 10					
		é	**		NG)7100 ()		
				. 230			
		:	() 108	ding add	2000 ()		
				- 7 N A S E	100M		
		;	< + con	51401 V4	luas +>		
P P 8 C F 0 0 0 P 4 2 P F 0 0 P		C B C B C B Z Y S I C B C D I U S Z B C		200 200 200 200	0 / 7 6 C 8 6 7 0 0 0 8 6 7 4 2 7 H 6 7 0 D / H	CACHOOR 18271 Cartrid User ca	t ontry address. Instruction address. Age device code address ortridge check flag.
2010 0013 0311 0019		Z 105 Z C M 5 Z C M 5 Z 1 0 C	re 300 50 71 B	290 290 290 290 290	C 1 6 8 C 1 C B C 1 2 8 C 1 9 R	; for pri ; for out ; for out	inter ouny iput register. iput register status.
600C		5 1 9 0 8 5 C		E G U I G U	0728 0038	Device Printe:	cade for printer upit. : BSC sequence code.
0100	A 7428	: : : : : : : : : : : : : : : : : : : :		2 x p a n d	bodula for CBG	8005 	TITOR JONICO LA BEL
0103 1	P 03		<b>C 7</b>	N150		2010102 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
0105 0	:0		821	N 2		1548 21	tura.
0106 2 0109 4	6		10	N E , P B N 9 , 1 3 E 1	DATA	: Printer : Get oAt	A SUBDER
0104		28510	00 P :			·	on conster-
0108 2	3		1 NC	HL			
0100 4	.D 0174		CALL	PENOLT		: 3etpet	 SATA 
0121 1	1 0 P7		2 J N 2	985100	P	1008	
0113 1	. 3					đ	
3114 2	18 16	/ 2 × C .	18	A. 1210	S T B 1	5110 11	primer reacy status.
0116 4	17 18 <u>1</u> 1		1 D 2 N	8, A A : 1 2 C E	9581	5110 11	autput register status
0119 3 511A 0	3 0 ) P		08 28CA	9		5.10	· Carry
0118 3	56 TT			C , 2890	UT	; 100 <b>P</b> 11	cnaž.
6110 7 0118 0 0120 6	5 10 13 10		LD GMT RET	A   C   2CHST	A ( R O	Cutput	
		:	e inte	ial data	• •		
0121		PRKD		023×			data number.
0122 1	18 26		00	25C, '4	, '   P A N		: Set down load character : address DEtH - DEBH
0124 E	15 50 15 30 4 59		08	04EN.0	200,0180,0200,0	4 E H , 0 0 0 N	Tuesday
012A 4 012C 7 0130 7	1 E 00 24 14 7 F 14 22 00		8 0	024N.C	)14N,07FN,014N,0	22N,000H	: Nedneedey

		END		
0120	C 3	864		
0168	78	21		; 82228 27 88888
0165	32 2007	LD	(USPBCBC), A	; Beest Leer flag.
0, 7	A P	801	٨	;
0164	22 7760	LD	(CBGBOOH+1),HL	
5161	21 000	LD	UL, BPTADD	Set return address.
0. 2	32 PP6C	1.0	(CBCHOOK),A	
0110	32.25	10	N. 0.8.2 H	: Set avsten hank.
0158	• 3	r euera: Di		·
0168		C 0 C D . S .		
				~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~
			Belease espand proc	edurø
0153	0 3 0 0	CBCA0DE:	DE CSCABOC	Izgand procedure addrese.
0158	0 0	CRCBANE :	08 008	Expand procedure bank (RAM 010)
0127	са	RET		1
0 1 5 6	84	E 1		; 28882 El 28833
0153	32 9009	LD	(USEBCRC),A	1
0151	32 02	LD	A, 3150	Set User code.
0142	22 9260	LD	(CSCB008-1), 8L	; Sat
0148	24 0159	1.0	HL. (CZCADDA)	; set expand procedure address
0148	32 5860	1 1	A, LENGONNA) /CRENGON1.A	; cet expend procedure bank. 'Ser
0145	14 0158	21 7 m	A ACREBANKS	; IIII DI INNA
0144	11	CRCSETUP;		
		;		~~~~~
		;	CRGHOOM Setup andu)	e
			<b> </b>	*****
0141	40 00			
0132	40 44 7E 44	8 đ	040H,044H,07PH,044H	,040N,000N ; Saturdey
0130	54 00			· · · · · · · · · · · · · · · · · · ·
8660	54 6A 79 6A	D 8	054H,06AH,079H,06AH	,054H,000H ; Friday
0136	24 00	18	erau'niau'niig'nidK	, uzin, uuun ; Inutsday
0132	24 24 77 14	5.8	0748 0138 0758 0148	0.21M 0.00M

	<pre>     SOTE:     SOTE:         .250         () long         .250         () long        </pre>	ABT(Bu-Ready) BOOK SAMPLE PROCEAN This sample program does as following. 1. Nove extend-part to 5000M in EAN. 2. Be-write ART(Bu-Ready) hoak address to 600 3. Ettend-part is to receive data A set it in receive-buffer. 4. Neglect system's process. Extend part beglects ROM/XOFF. DTB/DSB, RTS/CT and S1/SO controll. This progres doesn't check whether 6000M of RA free or not. Tou must decrease BAM disk to gels User-BlOS eres.(If you use User-BlOS eres, you change loading eddress.) stie coodition ()	CN to S. N is enough
	. PUASE	100M	
		Alos entries	
C 8 G 3	N B G Ø T	50N 05903N	
	5 5 5	System work addresses	
P385 P142 P420 F605 P607 (609 P612 P612 F61C	; 85JNTST 8238ANER 82533NNE 85PSTS 85P8BCP 85P8BCP 85P83AD 85888AD 85888AD	2GU       073888         2U       074208         8GE       074208         2GU       076048         2GU       076058         2GU       076098         2GU       076098         2GU       076128         2GU       076128	
	:	Other constants	
0 0 0 5 0 0 1 4 0 0 1 5 0 0 1 6 n 0 2 2	: 2 B A N K B 2 A R T D I R 2 A R T S B 2 3 0 S T R 2 5 0 B F R P	20U 05R 20U 148 20U 15R 20U 15R 20U 16R 20U 228	
0 0 0 2 0 0 5 0 0 0 1 0	2 # 3 # D T 2 # D 5 # 2 # C D	EGU 00000000 EGU 0000008 EGU 300100008	
1000 9000 PPD2 0042	341559 10404000 487900k 84786	ZQT C13068 EQU 350008 EQU CTTD28 EQC 000428	
0       8         0       4         0       2         0       1         0       0	8 3 5 0 5 8 8 5 5 7 2 C 8 5 5 0 8 C 8 5 5 0 8 C 8 5 5 0 8 C 8 5 5 8 0 0 8 5 5 8 8 0 8 5 5 8 8 0 8 5 7 8 C 9 5 0 8 C 9 5 0 8 C 9 5 7 7 8 C 9 5 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	200       10000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       0000008         200       00000008         200       00000008         200       000000008         200       000000008         200       000000000000000000000000000000000000	
			••
0100       31       3080         0103       73         0104       21       0132         0107       11       9902         0104       63       663         0105       80       80	START: LO DI LO LO LO LDIR	SP.MAINSP Set stack pointer ; Diamble all interrupt NL,HOOKOATA ; Load hook data DE,ARTHOOK ; BC,0003N ;	
010P 21 0123 0112 11 8660 0115 03 0678	10 10 10	NL.EITEND ; Load extend progree date DE.LOADADD ; BC.EITEGY-EITTOP	
	APPENDIX	Page 21 - 151	

2113	8 D 8 G		LDJA		
011A 011B	78 C3 7803		21	NBOOS	Bestore interrupt status
		:		*5001	1
		;	New ho	ok date	
0118 -		ноокри	TA:		
0110	CJ 8000		) b	EXTTOP	
			New ext	tend-program data	
0121		DYPDNA			
0121		ENTENL			
		:	() load	ling address	( )
			. DEPHAS	5 P	
			PHASE	LOADADDR	
8000		EXTION			
		;	******		
				EXTEND PART FOR	NEN ANT INTERBUTT ROUTINE
			NUTE:	This couting is	the process for new art-interrupt.
					the process for new stt intertop.
			() enti	Y DAFAmeter Noxe	()
			() T	IIN perseter	• •
		4	()	NONE	
			· · pion	NONE	
8000					
8000	21 2604		LD	HL, ESPSTS	•
5003	DB 15		1 N 7 D	A, (ZARTSR)	; Check Ri-ready status
8006	26 02		AND	TRIEDY	, sava status registor
8008	28 56		3 R 1 0	Z, EITD90	; So data in
9008	07		BLCA	a   0	Cet a ror status
600c	26 70		AND	855782+855082+85	
8005	18		LO	A , B	; Cot DSR line status
8010	26 80		AND	ERDS	
8013	D2 18		1 N	A.(2105TB)	Cet CD line status
6015	26 10		AND	EBCD	
5078	82		OB	a	
9019	83		0.8	£	è
9034	86 58		101	(#L) #550 <b>58+#55CD</b>	
501D	77		LD	(RL), A	Set 14De ALALUS
9012	DB 14		13	A.(2487018)	: lead received data
5020	11 PS04		Lb	81,257575	; Check mecalve buffer
5023	CB 4X 28 04	-	117	25287,(21) 2.277010	: : Buffer impfs full
5027	CB 06		SET	35880,(BL)	: Set buffer-overflow bit
5029	19 33		25	2 2 2 0 9 0	
5028		EXTDIO	:		
5028	21 7288			BL, SSINTST	; Set interrupt flag
5030	TA 2607		10	BL,(25P28PP)	Stare receive data into buffer
5033	CD \$062		CA11 180	155142	: • Neve huffer accress
5037	54		LD	0.2	
1029	30 70 70 76		10	E.L.	: Bottom of buffer:
5020	37	â <b>a</b>	09	A	
5334	12 42		120	≝1,8C	
9041	20 03		18	NZ, EXT320	50
5043	2.4 7609		1 D	UL.(BSPEBAD)	; Change put-pointer to buildr-top
5048		EX1020	-		
5046	22 8601		10	(25288991,RL	; See most data put-pointer · Chack buffor full
2043	10 15 761	u a	OR	ок, ткалянсят А	* ***** ******************************
504E	ED 42		58C	RL,8C	- Buller Isn's full
5052	2005 217604		LD	52,227030 HL,85 <b>P575</b>	<sup>1</sup> OMEEME - ON F EMEE
5055	CB CE		511	BSEBP,(NL)	; Set buffer-full bit
5057			:		
5057	2 A F61C		LD	NL, (RSBOL)	Increase receive-date counter
905A 905B	23 22 761C			HL (RSEDL),NL	i •
905E	18 40		18	DOGTXS	; Check nest data being received
8060		2 2 1 0 9 0	:		

8060	E 1			P 3 P	19 <u>1</u>		Seglect OS process
5061	C 9			821			Ind of estend process
			• i				
			÷		STORE	DATA TO BA	A ABEA
						********	
				NOTE:			
			:		This r	outine ete	res the data into \$43.
			1		You eu	et call th	is routing in D1 status.
			1	() ====	ry perse	- 1	( )
					01 : S	etting add	Iraas
			÷		A : S	etting dat	•
				1) 646	NEO DEFE		15
			÷,	()	BUNZ		
			р 1		ALL	afrecate	
9062	6.5		I SATEEI	PUSH	: c		
8463	<b>#5</b>			PUSN	44		
5064	P.5			9138	A P		
3065	ED 48 742C			10	BC,(BZ		; Save current bank data
8069	38 42			1.0	A, 8438	E	:
	D3 65			OUT	(ZBAHE	81,4	8
8060	A7			ION			0 0
8062	9.1 4.4 9.1			007	122083	**1.*	
9871	71			10	1811 4		: 1 Ant dasa sasa Ball.
8072	7.8			LD	A.C		: Bestore old bank
9073	03 05			001	I ZBAHE	81.A	
8075	78			1 D	A . B		•
8016	03 22	-		0 U T	122888	141 A	1
8018	73			POP	A 7		anstore registers
5018	C 1			909 909	BC		
807B	6.4		EXTROT:				i
				E in C			
Nacros:							
symbols.	1						
7605	ARTBOOK	0003	3863		0087	85033	
		0003			8033	13782	
10001	83987 877000	102 P	83880		8046	211020	
8057	217010	6010	827090		0121	EXTEND	
9000	EETTOP	0112	BOOKDAT	<b>A</b>	8062	LSSTAR	
8 8 8 8		1000	RAINEP		0042	ZANBE	
7328	3518757	F 6 0 8			F 6 G 5	82558865	
2607	* \$ \$ \$ \$ \$ \$ \$	7684			F612	<b>GA38828</b>	
1110	BSBDL	6003	BESCD		0080	855058	
0840	# 5 5 <b>7 % 2</b>	0020	110224		0010	******	
9475	***	0100	87427		780?	12007	
0014	*******	8015	248138		0005	284868	
0010	210578	0010	2800		0660		
0002	281801	0022					

Bo Patal error(e)

(30) SAMPLE 30. EXTHOOK

	••••••	<b>E X 7</b> 1 N 1	ZRRUPT NOOK SAMPL	• • • • • • • • • • • • • • • • • • •
	NOTE :	Than r	roffee about how	10 HEE BITHON'
		This p	program in constru	cted of next ] modules,
			1. EXTHOOK expan 2. Setup module, 3. Belease modul	d module, (EXTINT) (EXTSETUP) •.(EXTBLS)
		lf you execute	une USEB 8105 or V8105 program. (	ea for this progree, you must ees Chapter 4.6)
	() A680	able con	dition ()	
		. 2 9 0		
1	<> loss	tine addr	4.6 4.6 4.6	
		. P H A S 2	100 M	
:	() Cons	stant val	u :	
77D8 277 ODE8 ENC	HOOK	2 Q U 2 Q U	0 7 7 8 8 8 0 0 7 2 8 8	; EIT book addrees. ; Carttidge interrupt enable
9 D E 1 D S C	RCINT	<b>191</b>	000238	; module. ; Cartridge isterrupt disable
7000 B27	408	* <b>Q</b> U	070003	; module. ; "BET" instruction address.
PP4/ EZT	A0	2 G U	0 E F 4 PN	all interrupt end process.
PCD8 PST	CCTLB	1 Q U	000088	; 'JS BETAJ' address. ; P198 output data sava address
0018 210	STB	2 G U	0168	; Flöbifor CEG & IC card int. ; status]
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	38	2 G U	0231	; \$230(for timer int. statum)
C013 38C	57D 70	EGU Egu	0 1 8 N	; 'ORT' instruction. ; 'JO' idetruction.
:	••••••	3178008	eatend program	
0100 EXT 0100 PD 73 0142 0164 31 0186	197; 10 10	( 3¥SZZT 39, 2375	SP), SP 2	; Save system stack pointer. ; Sat bee stack pointer.
6107 11 P606 010A 22 0144	1.0 1.0	88,887A 194873	. D D 2 D 1 , 8 L	; Set gost procedure. ; ( Cartzidge int, control )
0100 21 0125 0110 £5	1 D 9 C 5 8	81,8272 81	TT	; Set feturn eddress. ;
0111 08 22	13	A.(Z175	5 3	; if timer itt. bas secures.
0113 0F 0114 BC 0123	CALL	C.18775	2	; the call in the process.
911/ 28 16	13	A, [ 2 1 0 S	<b>T</b> E 1	; if cartridge I/F int. ses : secured.
0119 CF	8368			; then call cartridge process.
0114	8236	- 5.2 - KO - 1870	5.0	
	POP	11		9 
JAIP OF				
0120 07 0121 03 0311 0124 66	2 P 8 2 Y	9C,1ST1	cc	; IT IC CAPE I/F IEI, MAG OCCUPAD; ; Then tall IC card process: ;
0125 BZT				
0125 28 58 0244	6.0	DE . ( SVB	2 T A D D )	; Get cartinge 1/8 lot. control 7 ADDB.
0129 28 78 0142	10	SP.(SVS	(EXTSP)	contorn system stack pointer.
012D 21 EF4F	1. U 2. X	(SP).81	1 M 5	; cnange return address : Lo post Drocess.
0121 02	PUSH	02		; Set catridge 1/7 Int control1
0132 28	3.0	C # 1. 1		; routine address. ; lump to return address.

0133       1	8 ym bo ) e 0 D E l 0 1 0 0 0 1 3 3 2 F 4 F 0 1 2 5	OBCRGINT EXTINT EXTSP INTTR ETAD BETAT SVSERTSP	C D 2 8 C 1 5 4 C 0 1 3 4 C 0 1 8 F C 0 0 F C 0 8 C 0 1 6	ENCRC]NT EXTRLS JNTCRC JNCND BTADD B21CCTLB 2103T0	F P C B 0 1 4 6 0 1 4 J 2 P B 9 0 C C 9 0 1 4 4 0 0 2 3	[ X 7 H DOX E X 7 5 8 7 U P [ N 1 1 C C P S 7 8 7 7 B E T C M D S V B E T A D D 2 1 T S R	
0133       INTTE:         0133       C3         0133       C3         0133       C3         0134       C4         0135       C4         0136       C4         0137       C4         0138       C4         0139       C4         0131       C4         0132       C4         0133       C4         0134       C4         0135       C4         0136       C4         0137       C4         0138       C4         0139       C4         0139       C4         0130       C4         0131       C4         0132       C4         0133       C4         0134       C4         0135       C4         0136       C4         0137       C4         0138       C4         0139       C4         0130       C4         0131       C4         0132       C4         0133       C4         0144       C4         0144	lecroe:			l S D			
0133       19775:	0154 c154 -55 0157 0158 0158 0150 0160	P3 27 13 37 7799 31 7600 22 PPDC 69		217315: 31 10 13 10 13 23 23 257	2 8 7 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	12 13 14 14 15 14 14 14 14 14 14 14 14 14 14 14 14 14	tele ***********************************
0133       1NTTM:         11 MTTM:       11 MTTM:         12 MTTM:       11 MTTM:         133       C9         0134       34 PODB         0135       C9         134       34 PODB         0134       34 PODB         0135       C9         134       34 PODB         0134       34 PODB         0135       C9         135       C9         136       10 A.(BTICCI         0137       C9         138       C9         139       C9         1314       34 PODB         1315       P4 40         1316       A.(BTICCI         1317       P2 40         1318       C9         1319       C9         1310       P2 40         1311       P2 40         1312       P2 40         1313       P2 40         1314       P2 40         1315       P2 40         1316       P2 400         1317       P2 400         1318       P2 400         1319       P2 400         1319       P2 400 <td>0153</td> <td>C 9</td> <td></td> <td>882</td> <td></td> <td></td> <td></td>	0153	C 9		882			
0133       INTIN:         1 If you want to espend the imac/Same interrupt procedure :         1 you can insert your our runting in this area.         0133       C9         0134       arr         0135       arr         0136       arr         0137       c For Certridge 1/F Interrupt :         0138       arr         0139       C9         0131       arr         0132       arr         0133       arron         0134       arron         0135       arron         0136       arron         0137       arron         0138       arron         0139       con         0130       arron         130       con         131       arron         132       con         133       arron         134       arron         135       arron         136       arron         137       con         138       arron         139       con         130       arron         131       arron         132       broad	0146 0146 0147 0149 0145 0147 0147	P3 32 C9 32 8788 21 0100 22 970C 78		EXTSETUP: DI LD LD LD LD LD LD LD LD LD LD LD LD LD	A.8270 (95781 91,517 (81750	77D 77).4 157 902+11.81	: #=:: D] 444 ; Set RET sectruction. ; Set RET hook jump eddress. ; sez RE 664
0133       INTTN:         1 1f you want to espand the lame/Sence interrupt procedure -         1 you on reactine in this error, interrupt procedure -         1 10 000000000000000000000000000000000				••••	008713	setup modul	
0133       INTTN:         (* If you cant to expand the lase/dame interrupt procedure *         (* If you cant to expand the lase/dame interrupt procedure *         (* For Certridge 1/F interrupt *         0133       C9         0134       (* For Certridge 1/F interrupt *         0134       (* For Certridge 1/F, you can the state can be append the interrupt procedure fore the *         0134       (* For Certridge 1/F, you can heart your own routine to this erea. *         0135       (* For you can interrupt procedure fore the *         (* If you want to expand the interrupt procedure fore the *         (* If you want to expand the interrupt procedure fore the *         (* If you want to expand the interrupt procedure fore the *         (* If you want to expand the interrupt of you can select post process         0130       (* For fc card interrupt *         0141       (* For fc card interrupt *         0141       (* for you want to expand the interrupt procedure fore the *         (* for fc card interrupt *       *         0141       (* for fc card interrupt *         0142       <	0184			EXTSP	890	408	atack eres.
0133       INTTN:         0133       INTTN:         0133       C9         0134       INTCRC:         0135       C0         0134       SA PODB         0134       SA PODB         0135       C0         0134       SA PODB         0135       C0         0136       SA PODB         0137       SA PODB         0138       C0         0139       C0         0130       C2         0131       SA PODB         0132       INTCRC:         0133       C0         0134       SA PODB         0135       C0         0136       C0         0137       C0         117       You can (B21CCTLB)         117       You can (B21CCTLB)         118       Contraide (1/P, you can Loser: your own routine is this area:         119       You can Loser: your own routine is this area:         1110       C1         1111       LD         1111       LD         1111       LD         1111       LD         1112       LD         11311 <td>3 1 4 2 0 4 4</td> <td></td> <td></td> <td>SVSRETSP: SVBETABD:</td> <td>2 C 2 G</td> <td>2 2</td> <td>: svetem stack pointer save ar ; returo method.</td>	3 1 4 2 0 4 4			SVSRETSP: SVBETABD:	2 C 2 G	2 2	: svetem stack pointer save ar ; returo method.
0133       INTN:         0133       C9         0134       SET         0134       INTCHC:         0137       C9         0138       INTCHC:         0134       INTCHC:         0135       C9         0136       INTCHC:         0137       C6 40         0138       C0         0139       C0         0134       INTCHC:         115       LD         115       SET         115       C6 40         115       C6 40         115       SET         115       C6 40         115       SET         115       C1         116       C1         117       C1         118       C1         119       C1         1110       C1         11110       C1	6141	29		INTICC: :• If you wa :• card 1/7. :- 377	nt to esp you can i	and the loter neart your o	rrupt procedura form tha 1C · n routine in this erea. ·
0133       INTTN:         0133       INTTN:         0133       If you want to expand the imac/Semac interrupt procedure         0133       C9         0134       INTCRC:         0135       C6         0134       INTCRC:         0135       C6         0136       INTCRC:         0137       C6         0138       C0         INTCRC:       INTCRC:         0139       C0         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt procedure form the         If you want to expand the interrupt want to expand the interrupt want to expand the interrupt want to expand the interrupt want to expand the interrupt want to expand the interrupt want to expand the interrupt	0143	69		RET C For	[C card	intertugt (	3. BZTADD.,(nothing)
0133       iNTTN:         : ' 1f you want to expand the lease/Sense interrupt procedure ·         : you can insert your oun routine in this eres.         : you can insert your oun routine in this eres.         : ' for Certridge 1/7 interrupt >         0134       INTERC:         0137       R6 40         0138       CO         INTERC:       Check CBC int. seebing states         0139       CO	013A	21 9000		: If you wa : cartyidge : .	nt to exp 1/V, you 	and the inter can inser: yo	trugt procedure form the
0133 INTTN: : ' If you want to expand the lease/Seese interrupt procedure · : you can insert your oun routine in this sree. : ************************************	0134 0134 0137 0139	3 A PODB 26 4 0 C 0		INTERC: LD AND BET	A, (323 010000 92	CCTLB) 008	; Check CBG int, mambing statu ; ; Beturn if disable.
0133 INTTN: ;************************************				i far	Cortride	e [/F interro	upt >
0133 INTEN: - If you want to expand the lassc/Sesso interrupt procedure + - you can insert your oun routine in this sres.	0133	C 9		917			
	0133			INITH: - · · · · · · · · · · · · · · · · · · ·	nt to exp	and the lease	c/Semac interrupt procedure . In this area.

(31) SAMPLE 31, BIOSHOOK

				******			•
		;			COF 24265	E PHOCHAN **********	•
		-					
			NOTE				
		4		This	progres s Dtogres i	nove now to exter • constructed of	next 3 modules.
					1. Ente	slubom 2018 beba	(228)056)
		1 b			2. Betu	p module (OPSUB10	1 2 0
					3. reje	ace module (BLSP)	1001
				These	modules	Are parts of UR	OS DOPATAR (
				Chept	er 4.6).		
				You .	vel esecu	te the UBIOS proj	irne when you use this
		-		progr			
		:	()	abla co.	0011100	< )	
				. 2 5 0			
					1		
			() LUN			()	
0000		PAOS	Z	2 G U	000000		; dumey lebel.
****		8105	36	ZQU	0		; BIOS BOOK address.
0200		0880	TUA	IGU	CUC001		- address
0010			1 2 2	880	16		
C C O O		1001		201	C 8 8 0 7 7 0	3-8U8S12£+236	: expend procedure top.
0007		8105	7 F T B	ZQU	000078		: BlOS Jump table top.
000		TARC	19102	x G U	000078		; L'SI FUNCTION.
		•				******************	
					970C <b>0.0</b> 70	GF U3288(U3 AP68 8 2 3 4 4 6 6 8 4 4 6 9 8 8 9 2 2	
0000'		OPHEI	: 2016				
0000.	23 2200		11	(31014)	1052 K+11.NL	; Z11+00	ad Utus entry address. a 8105 8008 Address.
2006.	63		RET	• • • • • • • •			
		;		Extend	BIOS prog	gron .	
		:					
0000		CATO	•	EQU	PAUSE		
		i	40 LOAD	ing add:		€ 3	
				. 29AS8		ECBIOP	
• •							
0000		ZX310	152:				tech soloter.
6604	31 6671		13	57.233	10552	: Set new stack	2016147-
C207	25		PCSE	81		: seve estry par	3881076.
6000	23		PUS 8	DE		1	
CC08	P 5		PCSB	4 <b>F</b>			
	24 0017		10	21.(SA)	YESP)	: Get this BIOS	function entry address.
CC 0 0	2.2		1×c	9 L		* 2	
CCOP	3.3		150	EL (DI)		i	
CC 0 F	18		1.0 2.5 m	2 ( d L ) M L		1 •	
CC11	26		1.0	J.(B1)		:	
							A.4
C::2	TA COCS		12	- RL 810	023412+11	<pre>.; #IOS jump lack .;</pre>	e idp accress,
6	8		5.8	A		A	
CC17	51 33		\$ 3 C	81,31		1	
CC19	7.0		10	A		C APAGA C TOIS 3	135 function code.
CC1A	PE 37		C P 1 P	7 ANGLII	122	1 II this side a then lymp.	S terior concline ;
			<del>.</del> .				
CC17		E 7 3 1 5	358:				
CC17 CC10	F 1 3 1		207	3 F		. Tecovel eurig	,
CC21	21		P02	HL			
C C 2 2	20 78 2028	,	LD	S.P., I.S.A.V	VESP)	; Recover old st	ach pointer.
CC 2 8	C 8		821				
CC27		2 x 8 1 (	DS:	_			
CC 27	FL .		PO P	A 7 D 7		: Becover entry	parameters
6628	81		<b>PO P</b>	ЯL		•	

			; • • • • • • • • • • • • • • • • • • •	can 108	ert yaur	0	onded-8105 routine in this strage.
CC31	8 D C 9	78 CC2P		L D 8 E T	5 P . ( 5 A V	<b>8</b> \$ P 1	; Recover old entry address.
0021			SAVESPI		05	2	
((3)					05	4 D M	
6671			IXBIOSS	P	<b>EQ</b> U	1	
0071			UBSIZE		<b>10</b> U	1-RUATO	
						_	
					. UEPBAS	ε	
					Belesse	TOUTINE	
				This so		II he sould	and an ROOF LEFT.
					ocrue er		HEEVE ON ROOK AREA.
0071			LSTOP		TOC	PACSEAU	11128
0 0 0 0			BRISTOP		200	008008	
0 8 7 9			NEADTOP		201	008908	
7000			827400		EGU	00001	
				an lead			
				() 1080			
					. <i>P</i> N A S E	RESTOP	
0800			8159800	:			
0080	21	• • • •		10	HL, BETA	00	; initial adiress.
0303	22	1718		10	(810288	+11,81	; BICS ROOK initialize.
				< Heade	e oull e	lear )	
			*				
	0 6	12		6.0	8.108		; Counter set.
8080	21	0990		LD	HL, HEAD	TOP	; Seader typ address.
				_			
0101			857 7001	P :			
8080	36	6 0			1811,00	8	Sever BAIT Clear.
0300	2.3			136	871 TOO	2	4
	10			BET		-	6.
0011			81552		29C	8-281870	) P

			250		
Sacros:					
Symbols	12				
2727	310596	2007	SIDSIPTS	CC.7	II31CS
CC80	EXBLOSE	0117	SCIEIZ	0211	EXRICSSP
0.070	SCIDASS	2300	OPNEBIOS	0000	PAUSZ
3808	RT1 1.00P	1200	066133	3900	8197206
0011	81552	1071	111707	3930	8815758
10190	arestar	1111	117707	1111	SAVESE
0007	TARCET3105	3600	72867757	0071	CB512E
0000	C 870 P				

So fatal erroria:

## (J2) SAMPLE J2. Extend communication protocol

	;			
	;	EITEND CO	MHUNICATION S.	ANPLE PROCRAM
	1			
	; NOTE:			
	;	This sampl		and communication-sethod.
	;	This uses	STAN-SO DBELL	ally, so this doesn't work
		by itself.	If you use t	his you sust and fy this and
	•	dahua it		nie, jou wort addity this she
		00004 111		
	å			
	2 41 88	seeble condi	1308 41	
	. 250			
	:			
	() 10	ading addres	s ()	
		060548		
	•			
		SIUS ence	3 6 8	
E 0 0 3	WEDOT	200 0	2 8 0 3 N	
806	CONST	50n 0-	E 8 0 6 N	
2009	CONTN	£67 0	23098	
EBOC	CONOUT	200 0	83063	
0020	BTAN	<b>ZQU</b> 0	00208	
		5×4148		
	:	aystee ou	LE NOULBERS	
2324 7160	ICAMPIN	Een o	80.01	
7104	TOFLICHT	200 0	71048	
7105	TISTTIME	50U 0	71058	
F1D7	T2NOT192	tqu O	71078	
7109	TISTPLC	200 0	F1018	
7108	DECVCNT	200 0	71088	
7100	DECVBUP	200 0	F10D8	
P42C	BZBANER	20U 0	P 4 3 C H	
P420	B7SBBER	100 0	P 4 7 D R	
1461		800		
	1373446	E 4 U U		
78C3	TCNIDT	280 0	F6C38	
7867	TEBBTIMP	500 0	PECTR	
P6C7	TOSEESZ	EQU T	******	
76C8	TCAMAREA	100 0	F6C98	
P6D5	754817	200 0	/6D58	
P.5. D.6	104340	F.G.F. 0.1	REDAN	
1600	PP9.50	8.310 01		
201C		1. UL U.	repus	
7844	FILENARE	200 0	FBAAB	
7000	ELTADE		10008	
5 M D C	158	200 91	ETADB -45	( ) X X )
LIDC	DCT	20U 14	CB +12	; ( 20 )
2724	TIISI	202 20	C7 + D 5	11241
2772	TZIIK	290 11	LIST +14	;:40)
2773	37.447.8	600 X1	LJS7 -06	((43))
	1	Other con		
1000		1 C II 0	10000	
1000	24.358	160 0	10008	
1660	TTIBEL	560 01	56608	
0022	A 2 1 8 0 M	EEU 01	8 7 M	
0042	24486	200 0-	428	
0090	BECSI	790 01	5 O A	
0005	E3×L2:	10 UG1	<u>s</u> N	
1011	2829882	201 21		
0000	X014X9	7.C.I. 0.	0000000	Concertion type (leased line)
0000	NEXEAN	ECU 01		
4003		<u>160</u> 01		
0002	JULKA	1 1 L UI		
0000	TEXTS	Let at		THI TTI COLDER (STOL
0041	TINTSD	720 QI	0000018	2.1 III 105001 (FT7)
0000	RJTATZ	ZGU O		3101na'''''TC2+1
2008	TIECNT	260 S		
	1			
	007 84	arameters.		
0005	RYR	<b>261</b> 01	00001018	; SYN count (+\$)
0030	BAUD	201 0	01100008	9600 BPS
				1
	1 			
	Lines	attribute par	ra081878	
	i .			
0000	KAISEN	εου Ο	0000008	; Leased Line
0002	TRACZ	500 Q	00000108	; TPDCe BOOB OP
0004	57510N	2QU 01	0000000	; Secondary elation
	:			
0000	TFX_ETB	RGU OI	00000000	; Chack code (Receive 178)
	ADDENDTY	Page 21 - 1	58	

0001		TEX_ETX	200 000	000018	Chack code (Receive ETI)
0001		2 7 8 1 2 7 8 8	200 01) 200 05พ	1	
		••••			
			MAIN PROCEA	đ	
6080	90 00	8 8	000H,00EH		CI elevie 10
6082		De	000W,000M,0	ICON :	
6085		STABT:			
6083	33 1000 P3	L D 0 1	SP, MAINSP		Set stack pointer Disable all intersunt
6089	21 60 . 8	LD	NL. BOOSDATA	· ; ;	Leed hook date
608C	11 7660	10	DE, TÊKHKI 86 Jej		
6092	20 20	LDIN	00,0-4	;	
6094	21 6384	LD	ML,LCEDT		Copy initialize data
6094	01 0028		80,01801-10	- TOS	
1090	20 80	LDIB		:	
6092	38 03	LD	A.03N		Set protocol to extend type
60 4 1	32 9100	LD	(TCANPBH),A		Jot protocor to ertend type
6044	/8	E) 12	NECOT	1	Restore interrupt status
0443	C3 8803	;			THO OF UNICIALIZE
		E New I	hook date		
6048		NOOKDATA:			
6048	82	DB	A P 1 8 0 11	: 1	TCAM
60A9 60A8	6 D D P 2 2	D 6	APLRON		
60AC	6142	DW	REDLL		
8042	E 2	08	AFLECH	1	01861
6081	22	28	APLBON	+ 1	NTRET
6082	6161	D 🖬	EIUL	-	
					•
		-			•
					• • • • • • • • • • • • • • • • • • • •
		LCB			
			•		-
6084			0		commando Status
6086	0000	DW	0	1	Buffer size
6015	0000	34	0		Buffor adčiece Recorvoć
608C	0000	04	Ô		Seletves
6 C 8 2	0000		٥	1	8 - C - F - + C
			•		
		;	•		Mada
50C0 60Cl		8 J	C		
60C2	2 2 2 4	D 16	71157	;	Timer table address
6004	2772	36	S L I S T		iot stri Settà troutet legie egoicee
		;		,	
		. +++ 11MEB 1	TABLE		
60C 9	0006	í (5%	6		• •
600.	0006	36	8		a -
60CE	000 <b>C</b>	2 *	0		
ēara	000	25	3	÷	- 1 - <i>0</i>
6004	0000	0.4	40		• U • •
		: +++ 21121 ·	LICATER TABLE -		
50D6	07	1 22	;	:	9 L
6007	07	21			N 2 N 3
6009	07	05	· · · · · · · · · · · · · · · · · · ·	1	84
6 0 D A	0.0	8 0	0	1	N 5 N 6
8908	0 0		U		A U
		. +++ 8TAH E	/F PARAMETERS	• • •	
6000	0.0	) 0.8	Q	1	81 (Sanding mode)
6000	0080	0.4	RECS2		82 (User's receive size)
600 -		01801:			

		i				
			NOTE			
				Following pro protocol.		is extend part for another
				This program	work	a under BIAM-SO (by using
				BTAM celling).	So	if 8TAM-50 doesn't install,
				menual of STAM	- 60	for further information.
				lo this progr	•	STAT-90 is modified to 8105
				working system	RON	IS extended by hook.
		0 0 1		The protocol	of 8	TAM-60 is used BSC-1 torminal.
				Jump table		
		I USTCAN	:			
600 F	C3 917C		3.8	TCAN	÷	luep new TCAM program
			8105 7	CAff Jung table		
	67 6145	CPEN:	39	BOPEN		Connect line
6025	C3 628A	10132	3.0	BSENO	ĩ	Send 1 record
6018	C3 6101	ELOSE:	18	BHCV		Disconnet line
60 8 9	C] 0JL1				. '	
		9 0	(This	table destroyed	181	enguster)
		1041	100	00100018		IN HI SYSN INCLUSION
0021	0 2 9 5	BTINT:	1.0	C1	;	Initialize BTAN
6070	21		08	1081		
6071	0 E 0 O	91400:	08	13AL	Þ	concracion bian
.01	JE 01	,019T:	L	C , 1	;	Define 10 list
8	21	CORR -	0 B 1 D	LD8L . 2		Open line
6371	21		D 8	LDØI	P	
607A	0 2 0 3	LCLOSE	: LD	C, 3	;	Close line
6-00	21	RINITL	: 10	C , 4	;	Beed initsel
607F	21		0.8	LDHL		tood coostowo
6100	02 95	ECATHZ	01	LDNL		san continue
#103	02 06	BCONCT	: LD	C,6	ĩ	tonos beel
810S	21 02 07		LD	<b>c</b> .1	;	Read interrupt
.105	71	BINIQ:	00 10	LDUL C,S		Read initial inquiry
8109	21		80	LD.L		Neuro (ottio)
810C	0 Z 0 A	#1311L	28	1081	•	
6107	01 38	NCNTNE	: LD DP	C,11 LD81	i	<u>este contione</u>
6112	1 02 0C	WINEST	: LD	C.12	i	Write initial and reset
6114	21	*CNEST	1 1 D B	C.12	;	Grite connect and reset
6117		NCOLC	0.8			Svite connect
6115	0 Z 0 Z 0	CONC.	58	1DNL	,	
6118	01 07	windc:	1.0	C.15 1081	;	Crite inquiry
611D 6112	21 38-19	-×87:	10	C.16	4	Stite wast becore transmission
8120	21	SIIDL:	10	C.17	a a	Frite TTD
6123	11	6848:	3.8 1.3	1001 C,15	i	Neite NAK
5124 5176	C E		13	1281		5-118 F0401
6127	51 12	##1361	2.5	IDAL		
1123	00 11	-21207	: 13	2.10 1981	1	Naite disconect
6117	11 38 17	CVATCB	: 10	C,13	i	Control Satch
6127	21	CESALL	: 12	C, TI	à	Control enable
1120	36	C35481		LDHI		CONTROJ GISADIO
\$133	11 14	[]]) []] []] []] []] []] []] []] []] []]		LONL	P	
ET 25 2136	27 I.A	BINAIT	: 10	C,26	ê	1100 walt
6135	21		LD	C,27	:	Scope
6139	CO 0020		CALL	8745	1	Execute BTAM Set return rode (For error)
6138	47		10	3.A C,R	1	ter foreto rose con ecces
6137	4C 87		0.9	A	1	
8143	<b>c</b> 9		WET		é	

÷ ÷ \* EXTENO DLL ROVINE ÷ NOTE: this routine estends DLL for new TCAN. CAUTION: : If TCAN roceive size ion't } record ()29 bytes), you must make new DLL routine. But it is very difficult to extend DLL for this. () entry parameter 6.5 NONE +> zetuen pozometor NONE 6.1 () preserved registers () NONE 6142 PERALS. 21 6159 11 75AA 01 0008 20 80 6142 BL.FILEBAS DT.FILEBAS BC.11 1.0 Copy new file name IAlways eceive BAS file: 6145 LD 6148 LD 6148 1018 6140 63 821 1 6142 FILECON: 6142 6152 54 45 55 54 20 20 20 20 -----0.8 COH! 6156 43 47 40 FILEBAS: DB 6159 54455654 2020202020 424153 6159 ...... 845 \* 615D . 6161 : TA09119 6164 54 43 58 54 20 20 20 20 4' 41 54 80 TEXT DAT \* 6164 616S 616: i EIIEND UL ROLTINE ----ï NOTE: ÷ This routine extends UL for new TCAE. But new TCAN routine supports system's require, so in this program only return to system. -1 : () entry parameter () SONE () return narabeler ()SONE c) preserved registers <> ÷ NONE 61 E F żxul: BET 616F 69 EXTEND DE BOUTINE SOTE: This routine extends DL for new TCAH. CA0110N: e 3 entry parameter
SONE • ) c) seturn persecter. NONE 4.1 >> preserved registers NONE 6170 EXDL: 19 15 12 1218 5154 11 1984 31 UCCE 20 90 HU.PILEDAT DE\_FILENAME ; Copy new file name ; calways receive DAT file)  $\begin{array}{c} 6 170\\ \overline{0} 173 \end{array}$ 6176 E179 6173 30,11 111 Сэ ÷

		••••• NOTE () •	NEW TCAN ROUTI This routine j When BIOS TCAN sending or rece ntry parameter A : Function -01 03 04 Other register TUT parameter CY : Beturn in =0 -1	s new blos TCAM soutine. is extended, it is very sepertant iving by 125 bytes unit. () number Connect Send Receive Disconnect s are depend on each function () formation Stormal return Other registers depend on each function. Error veturn A reg. is more datail information. B reg. 4 STAM's A reg. ()
6170		I TCAN:	NONE	
6 ] 7 C 6 ] 7 C 6 ] 9 D 6 ] 5 D 8 ] 5 D 8 ] 5 D 8 ] 5 D 8 ] 5 D 8 ] 5 D 6 ] 9 D 6 ] 9 D 6 ] 9 D 7 D 7 D 7 D 7 D 7 D 7 D 7 D 7	CS         21       6194         E3         3D         CA       6022         3D         CA       6025         2D         CA       6025         3D         CA       6025         3D         CA       6025         3D         CA       6025         3C         CA       6025         CO       C         CA       6025         CO       C         CA       6025         CO       C         CA       6025         CO       C	פעמו גע גע גע גע גע גע גע גע גע גע גע גע גע	Η Ε Η L , Τ C A Η R P P (S P ) , Η L A 2 , Ο P E N A C , S E N D A C , S E N D A C , C L O S E A , Z E B J	Sat return address Chack function Connect function Sand function Becaive function
		TCAN	I return process	
			A reg. is retu flamon ei RAJI	irn status, if a res. is equal to zero, by endel, else error return
6194	o 7	TCARET:	•	
6185	C 9	BET	ž	Sormal .eturn
6 . 9 6 6 1 9 7 8 1 5 5 6 1 9 7 6 1 5 5 6 1 9 7 8 1 4 6 6 1 4 5 6 1 4 4 6 1 4 6 8 1 4 6 8 1 4 7	C 5 D 5 2 A 5 C 3 2 B 5 C 0 2	PUSR PUSR LD DEC LD DEC LD POP POP FOP FOP SCF 827	ML DS SISPSAVE) DI.SPSAVE) AL INLI.D ML MLI.T DE AL A.ERES	Seglect system error process Thange return address Flomeunication error code) from flag
			CONNECT 1100- CONNECT 1100- This scent 1 Intry parameter C Connection C C Connection C C Connection C C Connection C C Connection C C Connection C C C Connection C C C C C C C C C C C C C C C C C C C	<pre>inclusted file name. if in type (Esn't care) i Sending mode i Seceiving sode file name if neec if creation yormal return forer return forer return forer return so</pre>
6145			SOSE	
6   A 5 6   A 6 6   A 0 6   B 0 6   B 1 6   B 4 6   B 7	CD 60EE 3E 35 32 EFDC AF 32 EFDD 11 EFDC CD 60F1	C A L I L D X D R L D L D L D C A L I	. 811 NT A, 5 YN 8 8 4 ° D I 0 C 1 1 , A A ( D C T + 1 ) , A D F, D C 7 L 87 M 0 D	; Instablize BTAN ; ETT clock, ER-DR on, 9600 bps, SYN=5 ; Set DCT mode ; Set BSC-] (or BSC-2) ; Ceneration BTAN (Modify BTAN)

6 1 8 A 6 1 8 C 6 1 8 E 6 1 8 E 6 1 C 1	16 09 )1 06 CD 60P7 C2 63C0		10 10 <b>CA11</b> JP	D, TIKONT E.TBACE+STSICN LOPEN NZ, PATAL	; Timer counter ; Leased Jina, trace on, 2nd STA., 4₩ ; Open Jine ; Fatal arror
61C4 61C5 61C5 61C2 61C4 61C0 61D0	AT 32 F1D9 67 22 F1D5 22 F1D7 C9		30 Α L D L D L D L D L D R <b>Č T</b>	A   TISTFLG   , A H , A L , A { TISTTTHE   , H L { T2NDT1ME } , H L	Initialize parometers Remet first access flag
6100				BECZIVE 1 BECOR	2 D
			NUTE.	Becaive data fo This uses follo IISTFLC(1) TISTTIME(2) T2NOTIME(2)	the host computer by 1 record. Noting work area. Statum flag Bit 7 : Receive and flag Bit 3 : Pirst receive flag Data counter in system buffer Current data satting address
		4 •	<) =0 E	TUSINSZ (2) TY parameter	() ()
			<) rat	HL : Beceive bu uro pavameter	ffer top address ()
				a : Betern ist = OCH -	ormation - Normal feluth - BC reg. means received data counter 16 BC reg. equal to sero, then and of receiving. (Text end)
				4 > 0 0 N -	- irror return 30 reg. oeens more detail information
			() pre	served registers. NONE	¢ 3
61D1 61D1 81D4 61D6	3 A 2109 28 50 28 06	8 A C J :	1 D 4 5 O J B	A , ( T 1 S T P L C ) ] 0 0 0 0 0 0 0 0 Z , R C V 0 0 0	Check and of receiving (353 is receive EOT flag) Not and
(105	01 0000	1	1 D	8C.0000H	; Set return parameter
6108 610C 610P	78 32 P109 C8		LD LD BET	A, 2 (TISTPLG), A	(Clear receive flag) End of receiving (BC + 6)
8120		P. C V O O O :			
6120	22 F1D7 24 2779			01,(07A317+1)	; Sat user area sotting size
6126	22 7667		L D L D	(TUSERSZ),81 A,=T1STF1C)	: Sbeck first time secons
6120	£6 02		AND JB	00000108 NZ, RCV100	Not first time
0126	20 31	7	1.0	A.2256AN	Cade conversion an
6170	32 2000		LD	(1C3),A	"Tear status
6195	35 6403		LD	(LCB+1),A	sar huffar erre
6179 617C	2A F108 22 2892		10	(LCB+1),8L	
6189	2A E100 12 EF04		13	8L.(DXCV8CY) (109+4),8L	; 3et buiter start address
6205	11 3363 27 EFD6		10	(TCB+6)'RT 61'3	; Clear reserved area
6292	11 1100		10 6 11 1	818171	2ead initial
5211	C2 825C		3 P	SZ, JEBCHSO	: 3748 ev:07
6133	JA EIDS		13	A,:1111716; 600000128	i fer first receive flag
5227 8219	21 2129		10	171217107,4	
611C 6717	21 1105 CD 5253		LD CALL	(TISTTIME), ST NORMAL	; Seed status check
8733		30100:			
6122	50 / 2 F105		10	(125171217),36 E.A	Beceive data counter in system Suffer Check data counter
6227	31		0.8	C 2 2 5 7 7 0 0	No data received
8113	72.12	;			507 <sup>1</sup> 5 3700
		1	сару г	OCETAGO ONEN EO A	
622A 622E	ED 58 F10D 28 F107		L D L D	DE, (DECVOUT) NL, (IZNDI)MEI	; Data setting address ; Data setting address
6231		RCV110:			
6231	1.4		LD	A ,   DE   ] 5 5 7 8 X	; Received data : Set it to RAM erem
6235	53 CT 0103		INC	H L	
6236 6237	1308	let-	1 N C 0 E C	U E B C	
6238	e 5		PUSH	HL	;

6239	2A 86C1		LD	HL, (TUSERS2)	: Chack user buffer also
6230	28		DEC	HL	;
8230 8240	22 E6C7		1.0	(TUSERSZ),ML	; Neer's buffer full?
6241	85		0 A	L	
6242	£ 1		POP	N1	t j
6243	25 09		1 H	Z, RCV150 A. B	; Yes : Data remain in buffer?
6246	81		OR	c	
8247	20 28		JR	KZ, ACV110	; Yes
6219		ACV120	:		
6249	22 8107		10	(725071ME), HL	; Save data setting address in user buffer
5240	18 14		J.R.	8 C V 2 0 0	: Beceive cent TEXT
	10 10	;			
			Nove r	ecsived data in r	eceive-buffer
5242		801120	:		
6238	10 43 810	3.5	LD	(TISTTIME), 8C	; Save remain data count in buffer
6253	81		OR	A   8 C	, Any cata resideut
6254	28 08		3 R	Z, 8CV160	; So
0256	24 7100		LD	NL, (DECVBUT)	; Receive-buffer top sources • Di rog is repain-dats top address
625A	ED 80		LDIR		; Noval
6250	20 48 277	19 KC 1 5 U	:	BC. (BTASIF+1)	: deturo permenters (Beceived deta eize)
6260	AP		XOB	A	(Normel recurn)
6261	C 9		RET		i
		17 1	Receivo	e newt text	
e 1 A 3					
6262	5.8		IOB	A	; Status clas=
6263	32 2801		10	(LC8+1),A	: · Sar new tereive huffer size
269	22 8802		10	(108+31,81	
626C	24 9100		10	HL, (DECVBUP)	; Set new receive buffer address
8272	11 2200		LO	DZ,LCB	Set LCB address
6275	CD 6100		CALL	BCNTNE N.Z. FRACHNI	
04+5			4 6		
6278	23 8105		10	(TISTTINE), NL	; Data count in buffer
6281	18 86		18	BCV100	9 2 8
			Charl		
		,	CHELL		
8233		NORMAL	:	4 (169-1)	- Fand Fratus
6226	28 01		AND	TEI_III	572278 9
6299	32 7605		1.9	TCATIFI, A	: Ses (Setting 1)
6258	2.8	:	61.1		•
			treer	check for read 10	1 t A L
6250		i EBECHS	0		
C21C	T S		: >	A . B	; Error code
6250	75 77		CP	-1 7 7 8 7 8 8	; JALA) errov7 · Tes
5292	FE 01		67	1	; Isception error?
6254	C2 638C		2.9	NZ, LINERR	; Sc. lice error
6297	C 8		5 5 7		j stadž s WTMEN
		-			
			trrer	cnach fer read co	ntiave
6755		EBRCBS	Crrer	cneck for read co	
6755	TE C1	EBECBE	Errer	cneck fer road co	Rtinue ; Exception error?
6755 6755 134 5 <b>29C</b>		EBBCSS	Cree 1: 22 13 12	cneck fer read co I <1.falcex0 A,C	Ations ; Exception error? ; fo ;
6 7 5 5 6 7 5 5 1 2 9 C 6 7 9 D	TE 51 25 TS 75 87	E R B C S S	Crrer SP SP CR CR	cneck fer read co	Atiove ; Exception error? ; To ; ; Receive 2012 ; No
6 2 5 5 6 2 5 5 5 2 9 C 5 2 9 C	TE 01 TE 75 75 87 20 20	EBBCBK	Creer 17 23 19 CR 12	-neck for reed co - - - - - - - - - - - - - - - - - - -	Atiove ; Exception effor? ; To ; ; Receive £07? ; No
6755 6753 290 529 6753 80 529 6	TE 01 TE 01 TS BT 20 EC	EBBCBC	Crer 13 13 13 13 13 CR 13 CR 13 CR	nech for read co : : : : : : : : : : : : :	Ations Exception stor? Fo Receive 2072 T No -11061
6755 6757A 6779C 6779C 6779C	TE 01 20 TS 31 20 EC 24 F6D5	5 B B C B 6	Crrer CP 23 29 23 20 CR 12 Receiv LD	nech for read co : : : : : : : : : : : : :	Ations ; Exception stor? : Fo : Receive for? : No =iing) ; Already received fTT?
67.55 67.55 67.53 67.53 67.43 67.43	TE 01 20 TS 30 EC 24 F605 3D	E B B C B C	Creer CP CP CP CR CR CR CR CR CR CR CR CR CR	I CONCENSION OF THE CONCENSION	Ations ; Exception synds? ; Fo ; Fo ; Receive for? ; No =iind: ; Already received ETI? ; ; Yo (illeged for)
6755 6755 6729C 6743 6743 6744	TE 01 20 TS 75 87 20 EC 24 F605 3D C0	E B # C 8 6	Creer 1 27 13 CR 13 Receiv LO BZC RET	CRACH SCT FORD CO	Ations ; Exception error? ; fo ; Receive for? ; No -ling) ; Already received ETI? ; No (lliggel for)
6755 6755 673A 529C 6743 6743 6743 6743	TE 01 20 TS 75 87 20 EC 24 (605 3D 0 0 24 (109	EBRCBK	Crrer ; ; ; ; ; ; ; ; ; ; ; ; ;	CRACE SCT FOR CO 1 C.SELCENS A.C A SZ.RECREO 501. (Free Sate A.(YISTFLC) 10000000	Ations : Exception error? : To : Receive fOT? : No : ling) : Already received ETI? : No (lliggel fOT) : Set end of text flag
6755 6757 5290 67.4 67.4 67.4 67.4 67.4 67.4 67.4 67.4	TE 01 20 TS 75 87 20 EC 24 F605 3D C0 34 F109 P6 90 32 9100	EBRCBK	Errer ; ; ; ; ; ; ; ; ; ; ; ; ;	CRACE SCT FOR CO 1 C.SELCENS A,C A SZ.CRECHED 501. (Free Sate A.(TISTFLC) 10000000 (TISTFLC).A	Atiove ; Exception error? ; fo : Receive fOT? : No -ling) ; Already received ETI? ; No (lliggel fOT) ; Set end of text flag
6755 6755 529C 6743 6743 6743 6743 6743 6743 6743	TE 01 20 TS 75 B7 20 EC 2A F605 3D C0 2A F109 P6 90 32 F109 24 FF9	EBRCSK	Errer ; ; ; ; ; ; ; ; ; ; ; ; ;	CROCH SCT FOOD CO 1 C.SELCENS A,C A SZ.CRECHEO 201. (Free Sata A.(TISTFLC) 10000000 (TISTFLC),A HL,(OTAWIF+1)	Ations ; Exception error? ; fo ; Receive for? ; No =ling) ; Already received ETI? ; No (lliegal for) ; Set end of text fleg ; ; Calculate received date else
6 7 5 5 6 7 5 5 5 2 9 C 6 7 4 6 7 5 6 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5	TE 01 20 TS 75 87 20 EC 24 F605 3D C0 34 F108 P6 90 32 F108 24 CFF9 ED 58 F60	E R R C 8 6	Crrer ; ; ; ; ; ; ; ; ; ; ; ; ;	CROCH 102 FOOD CO 1 C.1.21CEN2 A,C A 52.CRECREO 201. (Free Inte A.1.1CA71[] A 52 A,(T1STFLC) 10000000 (T1STFLC),A ML,(OTAWIF+1) 0C,(TUSES2)	Ations ; Exception error? ; fo ; Receive for? ; No = ling) ; Already received ETI? ; No (lliggel for) ; Set end of text fleg ; ; Calculate received date size ; User buffer size1 - (Free size)
6 7 5 5 6 7 5 5 5 2 9 C 5 2 9 C 6 7 4 6 7 5 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7	TE 01 CC TS 75 BT 20 EC 24 (605 3D C0 34 (109 76 90 32 (109 24 (FF9) ED 52 43	E R R C 8 6	Crrer ; ; ; ; ; ; ; ; ; ; ; ; ;	CROCH 102 FOOD CO 1 C.1.22[CEN2 A,C A SZ.CRECREO 201. (Free 2218 A.(1157FLC) 100000000 (1157FLC),A HL,(01AW1FO1) 00,1105ES21 HL.OF 0.W	Atious : Exception error? : 'o : Receive for? : No -ling: : Already received ETI? : Sot (lliggel for) : Set end of text flag : : Calculate received date size : User buffer size! - (Free size) : ot received date size
6 2 5 5 6 2 5 5 5 2 9 C 5 2 9 C 6 2 4 3 6 2 4 3 6 2 4 3 6 2 4 4 6 2 8 8 6 2 8 4 6 2 8 7 6 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7 8 7 7	TE 01 CC TS 75 BT 20 EC 24 [605 3D C0 24 [605] 24 [605 24 [605]	E R R C 8 6	Crrer ; ; ; ; ; ; ; ; ; ; ; ; ;	CROCH SCT FOOD CO 1 C.1.22CCENS A, C A SZ. CORCAEO 507. (Free Sata A, (TISTFLC) 100000000 (TISTFLC), A HL, (BTAHIFOI) 0C, (TUSEBS2) HL. OF 0.W C.L	Ations : Exception error? : 'o : Receive for? : No -ling) : Already received ETI? : So (lliegal tot) : Set end of text fleg : : Calculate received date size : (User buffer size) - (Free size) : Set received date size
6 7 5 5 6 7 5 5 5 2 9 C 6 7 4 3 6 7 4 4 6 7	TE 01 CC TS 75 BT 20 EC 24 (605 3D C0 34 FID9 P6 90 32 FID9 24 CFF9 ED 58 F60 ED 52 44 40 AF	E R R C B K	Crrer ; ; ; ; ; ; ; ; ; ; ; ; ;	CRACH SCT FORD CO	Ations : Exception error? : 'o : Receive for? : No -ling) : Already received ETI? : No (lliggel tOT) : Set end of text fleg : : Calculate received date size : User buffer size) - (Free size) : Set received date size : Normal return

		;		• • • • • • • • • • • • • • • • • • • •	
		-		SEND 1 BECORD	
		0 4 8	SOFE:	Send data to hra	at computer by 1 record
				This uses follow	ving work area. Data ocidier in weer buffer
		-		T2NDTINE(2)	Deta counter in user buffer
			<	TCXTOT (1)	VACK COUNTER
		1		HL : Receive but	ffer top address
			() ret	BC : Sending dat Uff permeter	La count (Bytes) ()
		•		A : Return info	ormation
				• CICINI	- Normal return
				4 D 0 4 D -	- Error return
			<) pre	served registers	4) 3C Log: Manual adda garait reactantion.
		1		NONE	
6 2 B A	22 9105	03630.	10	(TISTTIME), HL	Data pointer in user buffer
6280	20 43 2101		LD	(T2NDTIME), BC	Data coupter in user buffer
		1	Copy d	ata from RAM to sy	yates area
62C)		IND100:			
4263	20 58 9180		LD	DE, (DBCVBUP)	; Sending data address in buffer
6208	7A P195 ED 48 P198			BC, (DECVENT)	; Deta pointer in Gast Satter
A 3 6 6					
6200	CD 6392	240110:	CALL	ISLDAN	; tend data
6207	12			(DC),A H'	: Set it to system buff=f
f 3 D 1	12		185	DE	;
6752	1 B		0 8 C • C S N	8C 71	
62 4	2A P107		LD	BL, (T2NDTINE)	Decrement data counter
8207	28 22 F1D7		DEC LD	8L (12xD1138),8L	
62.8	10		LD	A , H	All data bufferilg?
6200	83 £1		POP	RL	• •
5202	25 04		18	2,5×0120	
6221	81		01	c	System buffer full?
6222	20 25		38	NZ, SND110	; 30
8224		SND120:			
8284	27 8105 24 8108		10	(IISTTIMEL, WL BL. (DECVENT)	Calculate data length in buffer
6 2 E A	6D 42		5 8 C	NL.BC	(Buffer size - Buffer space)
			3+1 LC	8	
4376	1. 0.	\$	10	A	70 L 6 A
6211	J2 1750		15	:102),4	STIITE . BENSAN flag on
8271	A.F. 3.2. 5.P.D.1		102	A   168•11.4	; Clear status
6275	22 8802		15	(2C3+21,HL	; Set data length
6275 6278	ZA FIDD IZ EFD4		10	ICB ( JECVEL F )	Set buffer address
6278	C1 0000		LD	9C,0000H	Class Peserve arda
6301	2D 43 2136		1.0	( [ [ 3 + 6 ] + 3 [	,
6305	1A P1D9		19	A.(1151F1C) 200000038	; Check first time access :
630A	20 11		18	N2, SND200	Soi first time
e 10 -		550143:			
- 21C	▲ <b>Г</b>		138	٤	- Initialize HaCk counter
5350 5314	32 7605	550160:	10	TECTOTIA	i
6310	11 2700		LD	92,108	; Mrite shitlel
5212	CD 610C C2 6362		3.6	SZ, ERRCHKZ	file stire alle
6319	14 4159		LD	A, ( 1157FLG) 000000018	Set first write 1145
6310	12 <i>1</i> 109		10	TISTELGI,A	
2331	19 09	-	18	5 N N 3 O O	; Go to check of and
			Weste	continue	
6171		: 5 # D 2 D 0 =			
6353	11 8000	3.00000	LD	DELCB	Set LCb
6328	CD 810F C2 8382		CALL JP	NENTRE NZ.ERRCHKZ	White stor
#		\$		and the shark	
			200 01	25VOIN <b>G CUACE</b>	
6320		SHB300:	LP	H.E., 17250775973	; Check data count in buffer.
0356	4 VID7		5 <b>1</b>		•
		APPE	NDIX	Page 21 - 165	

6121	70		LD	А, Н	
6330	85		0.8	L	; Any data resain?
6331	20 98		38	NZ, SND100	; )'a =
		;			
		1	Write #	n d	
		;			
6333		SND400:			· · · · · · · · · · · · · · · · · · ·
6333	14 6112		LU 	A . ( B I A H I E I	; chack sanning acca
6336	66 01			200000018	; CONTINUE MODE/
0333	C 3			2	,
6119		SND410:			
6139	3 5 0 8		10	A. TEXTHONHENKAN	+ TOUKA
6338	32 6600		LD	(LCB1,A	; STXETX, MENKAN on
6328	A P		XOB	A	; Clear status
6335	35 ELD1		LD	(LCB+1),A	
6342	23 0000		LD	HL,0000N	; Set data length to zero
6345	22 2702		10	(LCB+2),HL	
6345	11 8900				-; set LLD Keite continue (Send STR., 873)
6348	C3 6107		19	X7 700CHK7	- Write effor
5348	L D D D K		• •		
6361	2 A 2109		_ D	A.ITISTPLG1	: Besot write-first flag
8354	26 78		AND	11111100	
6356	32 /109		10	IT1STFLC),A	
0359	CD 63C2		CALL	VRESETX	; Vrite reset
635C	C2 63C0		39	NZ,FATAL	
6357	32 00		10	v'one	NOTAT LACALU
6361	C 9		E E I		ì
		ă a	REFORE F	hack for writing	initiai/continue
				a	
6362		EBRCHK2	:		
6362	F 2 F 2		CP	- 2	: Tatal error:
6364	CA 63C0		16	Z, PATAL	; Yes
6267	FE 01		CP		Exception error?
6369	C2 63BC		30	NI,LINEEB	NO
			1.0		
6385	79		OR .	A	Receive EDT:
6365	6 6		RET	2	: Yee
6362	PP 01		<b>C</b>	1	Beceive DISC?
6371	c 8		TIE	2	; 7
6312	FC 03		CP	3	; Receive RV17
6374	28 11		38	Z, 8V1	; <b>Tee</b>
6378	FC 03		CP	3	; Receive VACE7
6373	29 21		18	Z, VACE	; 188 : Conteneton?
637A	PE 04		1.0	* * * * * * * * * * * * * * * * * * *	The (Only SIXIT)
6370	28 32		C P	3	ENG Receive SACE 1
6360			1.8	2, 540500	
6382	78 06		CP	6	ENG Beceive EOT 7
6384	23 56		. 8	Z, SND150	; Yes :Check end)
6368	C 9		BET		
		4	_		
			88C6146	evi data	
8387	1		1.0	A. (T157513)	; Set write-first fleg
6364	36 2100		0 8	00000019	
4350	32 7179		10	ITISTTLGI,A	
6357	24 7107		10	91,(T2NOTIME)	; Check data count to buffer.
6392	7 C		2.0	A L S	
9383	8.5		08	<u>L</u>	[ Fisianed]
6394	C2 62C1		23	S2,280700	) No (Seug sezt ters)
			t -	LPECT-V	· Verte teelt
6397	CD 63CZ		AI:	- 11 a. of 6 + 10	
0.3.3.8	C 3	i			
		1	20C01V0	MACE data	
		1			
6368		WACE:			and an and the first first
5293	2A 7125		0.8	3.12.217247	301 -2
4346	75 0)		LD	(7157516).A	
UALD FAFD	22 1109		LD	SL. (72NDTIME)	theck data tount in buffer.
6346	18 F19F		CALL	HINEU	: Write incust*
63A9	02 5362		3 P	NZ, SRRCHNS	Sene ertor
63 A C	C3 632C		; P	SND300	Ind Thees
6 J A F		WACKOO:	1.0	A TESTET	- Chark WAEK COURTER
6 3 A F	3 A 86C5		1.00	5,1105161] 6	, there warm coonter.
6382	30		1.0	TCSTDT1.A	INACK COURT UD!
6 ) B -	JZ 76C5		C P	17	BACK COUNS 1 16 7
92F0	FE 13		3.0	C , SND160	Seite instand
638B	0110		827		i -
	~ *				

		1	1100.0	***			
			CINE .	C Leg		rtor status.	
		-			= 0	Date error	(Write only)
		i			2	ACKU/ACKI error Time out error	(Grite only)
		1			3	Data check error	(Bead only)
					4	10 erpor	
638C	60 636t	LINERHI	CALL	VRESETX		; write reset	
638E	CU 6362		RIT			:	
		:	Retal				
			ratal	BTAM CIG		ne autometically.	
		1		C reg	eans e	rtor status.	A
		1			0	Stop by esterney 11)eest megro	Q6A7Ce
					1	BSC check (Nodeo	not ready!
		i			2	Longth overrun	
					4	Carrier detect	
		-			5	Not used	
		1			6	Not opened Double open	
63 <b>60</b>		PATAL:			•		
8300	C 9		8 6 1			; So operation	here
	-						
				CLOSE L)	N 8		
		2					*************
			VOT P :				
		-	3016.	this tou	tine c	loses line.	
		;		IF BTAMI	7's bit	t 0 is zero and a	ending data mode.
				The net stars		4) 10CE -	
				NONE			
		1	() 2011	ure parame	167	¢ )	
			() 0785	NUNE Berved feg	1 S T G T #	4.3	
				NONE			
6307		BCLOSE:	1.0		1	· Chack appding	
6361	34 EFF9		450	00000001	8	Continue mod	•7
6354	C1 2338		CALL	N Z , S N D 4 1	0	; Tes (Seod STI	
6309	CD 60PA		CALL	LCLOSE		<pre>: Cluse line. : Kormal return</pre>	
6300	1		881	8			
6363	63	:					
			vrite i	Colu IC	and .		
		-		0111 00	red	, <u>, , , , , , , , , , , , , , , , , , </u>	
6368		VPESETX	*				
8367	c S		2058 CALL	9C 685587		SAVE PERISTER	
6367	20 03		18	YZ, WRESI	0	; fatal error	
6304	c 1		202	80		: Bentore regan	ter
A 3 D 5	79		22	a. 3		RESTORE STOR	2368
6326	C 9	¥82510:					
6307	D I		209	28		Soe Amene	
6305	C 9		YEY				
							*********
				STORE CA	7A 10 J	145 ABE_	
			N072:				
				This row	1104 41	CT04 120 Éata 13 Nue ipuise (= 21	
		1		100 2011 19 2572401	- E # 1 A - 3 • P	AND LEUTIE 12 21	
		-			ting as	22911	
				t : 5et	ting da		
			() ZEEG	NOSI	100		
			() pres	cerved reg		* 2	
		ISSTAX:		ALL			
6339	C5		2 C 5 N	8 C		; Save register	0
KUL3 AGE3	25		PUSB	AE			
8 3 0 8	rs		LD	8C. (8284)	NKEI	; Save current	ass data
6300	20 48 FV2C		6.0	A.RAHBK			
6365	03 05		OUT	IZBASHB1	, A	1	
6321	A 8		OUT	4 ( 250 0 H K B	1,4	-	
6363	1) 44 fl		POP	AF	-		
631.9	11		L D	INEL.A		: Set date into	- EA2 -
6359	19		OUT	I ZBANKBI	. 4	' sessors of 0	
638A	18 va		LO	A , B		1	
6320	D3 22		OUT POP	4 Z S 8 8 × N R A F	1 , A	1: Restore rette	
6 3 E P	Ø 1					, assiste regte	
		APPE	NDLX F	Page 21 -	167		

6361 63E0	C 9			P 0 P 8 2 7	8C		
			- - - - - - - - - - - - - - - - - - -	•••••	CET DAT	A 7807 8A	
				NOTE:			
			2	NOTE.	This co	utine get	s the data from BAN.
				() entr	A Dataed	it call th iter	is routine in DI status. ()
					BL : Ce	tting add	re
				<) zetu	A : Co	itten data	
			:	<> pres	erved re	gisters	
0372			; ISLOAI:		BC,DE.I	IL	
6373			•	PUSH	8C		; Save register
6317	38 42			LD	90,(#Z3 	IASER 1	; Save current bank data
6328	D3 05			OUT	128ANES	1), A	9 
6370	03 22			13	A,008 1258251	881.A	•
6400	18			LD	A,(81)		Get date from TAM.
6401	79			PUSN	A P		. Restore old book
6462	D3 05			017	(2BANES	E 1 , A	, settie die dama
6404	78			LD	A , 8		
5407	F1			POP			Bentore gotten data
6408	C 1			808	BC		Bostore register
				821			•
Nacros:				END			
Symbole							
0022 6148	APLEON	0030			6301	BCLOSE	
0020	8TA3	8779	874517		6022	11151	
6071	87M08	6139	BISCOP		6136	ATNAIT CLOST	
6003	CONIE	2083	CONOUT		1906	CONST	
612D	CVATCH	890C	001		7100	BECVBUP	
0008	8883	F60C	EBBADE		623C	I DACORO	
6299	EBECSE3	6362			6170	EXAL	
6159	PILEBAS	6167	FILZCON		6164	PILZDAT	
FBAA	FILENAME	0006	RENKAN		6 Q 🔺 5	SOCEDATA	
6074	IDIST	6372	ISLDAI		63D9 99D0	155743	
60.84	10801	6 G F A	LCLOSE		0021	LDBL	
6380	1 1 5 7 8 8	F609	LOADAD		6077	LOPZN	
1000	OPEN	0042	BANBE		6100	BCNTNE	
6103	BCONCT	6029	BCY		6120	BCV000	
6222	SCA100	825C	20180		5282	809200	
0080	82CS2	FOGO	RETADR		5109	81118	
6070	EINITL RIBANER	F42D	8258853		8025	SEND	
8201	90100	8100	SND110		6124	SND120	
630C	SNB190	6310	5 N O 1 E O		8338	5×0410	
6053	START	0000	STATUS		0004	STSION	
0005	S Y N	P106 6170	TISTELC		F109 F6C9	TISTTIME TCAMARZA	
F107 F603	12501176	FICD	TCAMPRS		6164	TCAMEET	
5 C E	TENTOT	P1D4	TOPLICS TYTER	7	F6C7	TEBBTIME TEBBTIME	
5663	- 2X 815	000)	123_211		0005	TIECXT	
		0032	TOUMA		2000	IIAII	
TECJ	TSPSAVE	63 A P	SACEOD		2803	VEODT	
6398	NENEST	530+	WENTNE		6115	VCONCT	
512A	SP18CT	610C	WINITL		6327 	N82510	
6112	-15885 -68887	6362	VEESETX		6121	WITEL	
6115	SN8T	3005	ZBANEE		0022	5299265	

So fatal errortal

## (33) SAMPLE 33. Extend IC card protocol

.

331 SAMPLE 33.	. Extend IC card	protocol				5-1 C
		:	******			
		1		5 A 17 P 2	LE PROCRAM	FOR EXTENDED IC-CARD
		;				
		÷	3 T O K	This s	sample prog	ram extends IC-card executing
				the pa	ert of send	ling or receiving commands.
				Thim	sample pro	gram in divided into two parts.
				DART 1	te initieli Cor enother	IZING Bystem eree. Another is estended r protocol. Extended part executes in
		4		this p	program ous	st be located in an application ROM
				1256 1	Rbit), Plens	se write this program into P-BOH.
		;	()	eble d	condition	()
		:				
		;	. 200			
			O lond	ing ad	dress	•
		ē	. PHASE	06050	0 8	
		;	() CONA	tent	Alues	4.3
		;		8105	entries	
		1			088038	
2809		CONIN		200	669420	
830C		CONDUT		E G U	069068	
				Syste	an work add	resses
		;		7.011	020544	
POBS		YPOPST		EQU	070858	
9009 1006		8248738		200	CFOD3N	
		8288308	_	IOC	9 LOD VR	
2008 2000		RZSVR		200		
F000		RZJOCTL	2	201	OPODON	
FODE		R2C7183		19U	070028	
7026		ICCPWTM Setabl		100	050268	
2197		STSARTS	1	280	071978	
F1A8		EPIBETO	-	200	071488	
F125		1088052		EQU	071258	
#136		ICHONDI	1	EQU	071268	
129		ICCPNCN	T	200	072021	
F 2 0 3		1003011	C	zqu	072038	
1241		283005		ZQC	065413	
5462		52618E		203	074628	
F4C4 F4D5		SPESPC		290	374248	
785C		1034332	<b>C</b>	200	076508	
1012		•••••	-			
2660		ICONFLG		2 G U	07660R	
-662		CUSEDV		TEL	076628	
8663		ICIDDI		1 G C	018238	
:177		3982		101	071779	
562*		TPICCAR	0	202	076579	
				Othe	r constants	S
1000		MAINSP		rou	010008	
575F		HOOKADD	8	201	077678	
0 C = 0		SAITTIS	E	EQU	000500	
5000		758¥78		EQU	0358	
6001		3002		EQU	0848	
0002 000A		ANSCNT		EQU	0 0 A H	
0020		RICSIZE		<b>IO</b> U	020M	
0001		RDCHD		<b>EQU</b>	C I N	
0 0 0 2		NTCHD		8 Q U	0214	
0000		ZCTLB1		EQU	008	
0004		2128		8 Q U	041	
0014 0014		2487008		290	1 4 M	
0015		ZARTHB		EQU	3 S H	
0015		ZARTSR		E G U	124	
		APPE	NDIX P	Page 2	1 - 169	

0016		210978 2487C8	201 16N EQU 16N	
0018		2100113	EQU 17N EQU 15H	
0019		ZIOCTLR ZCTLR3	421 U91 1125 U93	
0006		ERRG	EQU 06N	
0007		EBRJ	EQU 07M	
0009		2889	KOU DAN	
		18811	EMID GRM	
			HAIN FROGRAM	
		Nes	statemento	are header for 305-execute program.
6080	na ne		0005 0078	Vender for ADM second
6082	00 00 00	0.8	0000,0000,00	
60.85		STAR7 :		
6085	31 1000	LD	SP, MAINSP	Set stack pointer
		:		
0000	21 6080	10	HE, HOOKDATA	Change NOOK jump address
408E	01 0018	LD	86,349	somb to begin by approacted the
6 6 8 3	P3	D I		Disable all interrupt
6023	20 80	L01	8	
	F 8			
6095	38 99		A. 111110008	ezesbba esuseas sood se?
5097	33 9126	L D .	(10306011),4	
		The	following parame	sters are depend on each IC-card.
		; So 1	f you need, pleas	ne ent them na you like.
		+		
A 8 8 8	32 9199	1.0	A, ISBITE (ICTSDT).A	, set to byte comparing data
6097	32 84	LD	A, 5002	Set parity, deta length, stop bit.
6 C A 3	32 \$197	LD	(SYSARTHR), 4	
0 A 4	38 20	LD	A, BECSIEE	Set 1 record physical eize
A D A D	3 P 10	10	A. 16	: Set 10-cord stre
6048	32 F652	1.0	(OTICCARD), 4	
6 C A Z	23 0078	LO	81,16=1024/1	36 ; Set saminum record symbols
8081	22 F65C	10	(ICMAIBEC), E	
6087	22 71sB	10	(EP18870),81	
4 5 0 B	C3 8803	3 5	VECCT	i.
		: ***	************	
		;	10 9RJL 8008	
1182		90083474:		
0000	02	0 8	APLEON	; for 8005 90. 1
3603	6005	2 M	LEDITOP+008	Te- 3006 No. 2
6001	8005	24	CEDTTOP+038	
80CJ	0 2	20	APLRON	: for 9006 No. 3
6004	5008	2 N	C80170P-06P	
6006	50 D.R	10 B 11 M	(801139+96+ 767962	TET AGGE AGG T
6669	02	28	APLEON	Ter HOOK No. 1
6 0 C A	6021	54	(807:0P+0C2	
6000	02	08	APLEON TRATERON	; 707 400k 107
6000	02	28	ABLICH	tar 2001 No. 1
3666	6217	26	19331735+125	
6002	92	2.8	151802	107 300E to 1
6003	SOZA		CODITOPIIE	i i
			****************	
			anto another are	serve for extended it-card protocol.
		TOV	following product	an 18 dor 18.
			***************	
8085		UBDATA:		
		Þ		
			HOOR JUNP T	ABLE FOR EXTEND IC-CARD
			_	
		: NO1	E:	a part with all interrupt disable.
		:	somb fo fur	

6005		UBOTT	0 P :		
6005	0100 CJ		4 6	DICONEJ	Besot IC-cord
6008	CJ 617F		10	UICONKJ	
3008	C3 617P		J P	UICDHL4	
60EL	C3 6159		1 P	DICONKS	; Brad ] record
6024	CJ 6191		3 P	nicowre	; Vrite 1 record
60EA	CJ 6209		39	UICDUKS	; formet IC-card
				BESET JC-CABD	AND BECEIVE RESPONCE
		-	() ent	try parameter NONE	4.3
		1	<> P+1	urn parameter	
				= CT : Betorn 101 = 0 : 92	creally opened
				11 : 06	Den ertor
				A	leror code 186 line out error
					•07 Protocol ezcor
					+OA Cover is opened
			() Dre	NONP	()
6 0 E D		NJC081		PORT	
6020	83		DI		
60 <u>2</u> 2	08 16		AND	00000180N	; check it-card cover status. : Cover is obsoed?
6072	31		567		
6073 6079	32 08 CA 617P		1 D 3 P	A. FREII 2, BOOGRET	; Error retuza code. ; Yes
	34 6008	1	1.0	A. (82152718)	: Enable Cover Interrupt
6075	P6 26		0.8	001001308	Set to reception mode
60 00	D3 17		0 C T	A, (ELTCOIL)	: Supply Vec to IC-card
- 0	T6 27		10	0C1001118 (ZICCTER).A	Sumply clock to IC-card
6103	32 7008		LD	(REICCTLE).A	
	•••	:			
6106	01 0050 256 05		LD CALL	WAIT MAIT	; WAIE UNELL VEC AND ELDEN ATATIONARY
e 1.0 c	3. 80.0.	;	LD	A. (BZARTCB)	; genet error of serial line
6102	26 10		0.8	000100008	
6111	03 15		007	( ZARTCE ), A	; - Cat BST to low level
e113	JA FODD			111110113	. 297 527 fD 104 16467
6119	32 1000		LD	(BEIOCTIB),A	
6 . 1 9	33 18		OST	(Z10CT12),A	;
A) N B	0.0.16		1.8	A. (210518)	; Check 1C-card cover status.
6112	£6 04		AND	000001008	Cover is opened?
6121	37		SCF		: Force return code
6122	32 03		38	Z, BOOKRZI	Yes
6126	32 01		<u>_ D</u>	A.018	Set IC-catd power-on flag
6125	22 2660	4	15	(ICONFLU),A	;
			lecely	e apswer-to-ress	, using the subsoutine 180004.
		1	Sext tw	.2100407678.0	
		1			I No poor of the lites actwor-to-teent
6129	87		39	HCONBET	Ind of hook.
812C	7.2.3.1				
6178	ZA FJAE		10	SL. (JCRSIPNT)	Cjevi lecejnjuč vleg
513:	54		1 D	I, 1	
6132	36 30		20	(RI),009	;
4124	12		180	PC INSCRT-1	
6126	27 000a		LDIB		
9173	[D 30				
6129	IA FLAE		12 2 4 1 1	21.0152515510 202007	Pecoive is and set it to RAM
6138	CD 5110 18 76			0,2100120	fror (Perhaps time out)
6112	24 1129		10	A .: ICISDI'	; Check IS byte cooe
6146	3.6		10	A. 2981	: Irror code
6147	32 07		SCF		
5149 5148	20 33		JR	NZ, HOONBET	; Unwetch
			1.9.0	N T	Beceive TO byte
614C	23		CALL	168001	
6150	38 24		1 B	C, UJCD190	; Receive error
8152	23		1 N C	ні С.А	
8153	47		AND	000011118	
6154	57		10	D , A	; Save complementary byte
		UICDII	0:		
6157		A <b>P</b> W		$P_{0,0,0}(2) = 1.21$	
		API	CNULX	rage 21 - 1/1	

6157	2.0				) Charle Labordona buta
6118	13		8824		i Check Intoriace ayte
6150	0.0		BBCA		
6154	0.2		RECA		1
6158	0.5		ARCA		
615C	00 04		10	8,04N	loop counter
				•	
6158		U1C0120:			
615E	0 6		ERCA		
615F	P 5		PUSH	A F	
6160	OC 628C		CALL	C.ICBODT	; Receive data if interface byte exists
6163	38 16		18	C, U1C01A0	; Receive error
6165	4 7			C 1 A	
6166	P I		150	# P	
6167			D 1 X 7	0100120	: 
6764	10 24		18	C. 01CD110	: The hore exists
0.01	30 00				
616C	7.4		23	4,5	; Check comlementary byte
6160	87		38		
6162	25 01		4 B	I, BOORRET	; So complementary byte
6170	47		15	8 , A	1
		1			
6171		CICD130:			
6171	CD 626C		CALL	108007	Receive complementary byte
6174	38 06		38	C'AIC0180	
6178	23			81	
6177	10 28		U J N Z		
8173	18 04			SUURBEI	; NOPUAL PECUER
<		4100160			
6178	<b>F</b> 1		POP	47	1
6170	•••	C1CD190-			
6170	32 06		L D	A . 2226	: Error code (Time out)
6178	37		5 C 7		tredr return
			10110-	ing BOOIs are	beau foo
F ± 7 P		LICORES:			
9115		U_CDHEJ.			
6177		CICDAR4:			
		:	80.05 -		
6178		8006427:		÷.	
6177			9058	A.7	Save return code
6190	34 8422		1.0	A.(B2128)	; Bestore interrupt status
\$193	D3 04		0.01	(2168),4	1
6185	71		POP	A.P.	1
6156	78		21		[famb]e interrupt
8187	C 9		827		1
				BECRIVE DATA	
		:			
			9078:		
				Secoive data	by 1 record from 15-card.
				Calculate re	ad-statt address by SEETBE & SEESEC,
		2		and read into	DBAADR from 15.
		1			
		;	() ent	ry persector	()
		;		(SEETEE) :	Track for reading stars address
				(SEESEC) :	Sector for reading start address
				(DTAADE) :	SALLeL TOLLER ISL LERGING TURO
				uro parameter	
				- CI : serven	Normally concert
				41	
		1		- 4 -	TTTOT COOP
					=06 time cut error
		1			agt Broscect attos
		:			nea laver is spened
		1	() pre	eerven registe	EB (1)
		1.		All without	A P
6198		UICONES:			
6155	13		01 0		
2159	25		YUSN	30	· ····································
EIEA	35		51°54	10 E	
6158	13		- L 3 H ( A 1 1	108800	: Read 1 record
26136			18	ICADIWT	, neee 1
0137	10 41				•

SEND DATA BY 1 RECORD -----19702 Send data by 1 record to 10-cerd. Calculate write-start addrose by SENTRE & SENSEC, and write the data of DSAADR. () entry parameter () (SEKTRK) ; frack (or reading start address (SEKSEC) : Sector for reading start address (DHAADE) : Buffer address for reading into () return parameter CY : Teturo : Teturo information =0 : Normally n +1 : Open error Normally opened ign -- Cover 12 obesed
ign -- Lime on; ellos
ign -- Lime on; ellos
ign -- Lovel 1
ign -- Covel 1
ign -- Co А ved registers 4.5 All without AP 6191 1C0886: 13 01 6191 6192 C \$ PUSN 8 C Save registers 8193 8.5 8058 D 2 PESB ..... 23 81 102403 9195 CD 618C CALL Wyite 1 record 6119 ICROSVT: 6196 8.1 POP R L : lestore registers 6199 . 10 POP 3.0 C) 19 EI 6194 POF 80 8005877 6192 3.8 1 CODSE 1 VE ATLE CASSE .......... NOTET Same as DICDUES () entry parameter Same as OICDBKS () return parameter Same as OICDBKS () preserved registers < 1 • NON 5190 ICRBOO · 5 2 0 C LD C . 00M ; Select serial line to IC-cerd 6190 13 8348 12 8274 6197 CALL SELSER Fill null code to DMA buffer CALL 6142 FILLNULL ICALBCD C, ICB820 6148 C2 6344 Calculate record number Virtual record . 6135 28 10 19 ÷ 6144 LA PADS LD BL. (DMAADR) : and buffer address 51AD ICB810: BC C, BDCMD ICMDFNC Save loop counter Receive 1 physical record - 2 PUSH 1 32 01 6148 LD 6130 CD 6441 CALL 6123 C 1 POP 36 2 6194 35 121 Error return : Set next record humber 6125 IBLNCHN CD 545A CALL 19 73 108510 D252 112820: \$134 £114 : Sorsel return  $^{2.7}$ 0.8 . -125 2.9 857 ............ VEITE JATA 25 1 BECGBD ............... \_\_\_\_ 1010: SABE AS CICDENS () entry personter 1.5 antro parameter Saam as CICDNK6 return parameter Somo am UICDNK6 preserved requisters 4.1 • • • 4.3 NON 618C 100400: : Select serial line to IC-card 0.0 3.0 C. 00H 61 B C 1.10 CALL SELSER CD 6346 1618 ; Calculate record number 6 1 C 1 CD 8344 CALL ICALRED C, 1CR620 Victual record 6164 38 10 18 1 HE, IDHAADRI ; DHA buffer address 6166 Ľυ 168910: 6109 : Save loop counter 6109 C S PUSH 80 APPENDIX Page 21 - 173

C. NTCHD ; Send | physical record LD 0E 02 61CA CALL 1030180 CD 6441 6 ) C C 80 909 881 6108 **C** 1 C ; Errot return 6100 D S IBLKCHK Set next record number CALL CD 645A 6101 DJNZ 108410 10 ÷. 6104 ICRV20: 6106 08 A Sormal return 6106 81 ..... 6127 6.9 ÷ 10-CABD NOUXT CHECK ......... : NOTES C) entry persector NONE <.) 5 Y ---NONE FD paremeter + D + Normally opened (G7T1CABD) -- 1C-card size (ICMAXEEC) -- 1C-card max record No (TPICCABD) -- 1C-card directory size (DF82) -- Dieb parameter block () return paremeter (DP82) -- Dieb par: \*1 Open error \* 2 Error code \*06 -- Time out error \*07 -- Protocol error =0A -- Cover is opened () preserved registers < NONP U1CDNK7: 6109 6106 F 3 01 ÷ In this part, you must make IC-card softwars-op. 1-status. and determine the size of 10-card and set it t- GTICCARD, and set IC-card max record size to 10MAIREC. following part is matting the system area oy IC-curd's size. 590 0982+5 **D5M2L** P17C 8 Q U DP82+7 08121 117E 0992+11 CK 921 8 Q U ; IC-cerd size (Fer kilo byte) 34 1658 1.0 A. LOTICCARDI 6109 27 08 6100 . 1,0100700 ; No size J B 6130 | Nes IC-card size - 1 6101 20 020 . 11100700: 6120 ; Set disk eize met. 21 F17C 01 C2C9 FE 02 LD (DS7211,A 6120 1.5 BC, 2+2=6+3 6123 6125 C P 7 : 1068 than 2 kild bytes ; 8 reg. ts 4, C reg. 16 16 C, C1C3720 26 0C C8 00 19 6125 BLC 812A 9111 8122 11 31 F2 09 21C C7 C Less than 5 kilo bytes B reg. 18 5, C reg. is 32 6170 2.5 04 2.8 C, CICD720 ÷ alc. 8172 \$3.20 8 28 21 C 6 L C 0100720: 5775 TS 1= PS≤P T9 61FC 61F7 ; Set sector ausber of directory 1.0 A . 3 (TPICCABD),A 10 13 A . C Set directory number 9175 10 32 2175 DIC ٨ 1 DRM211,A 8175 LD 1.0 Set check sup vector \$129 A . C BRCA 6230  $\subseteq \Gamma$ FEC 3 \$101 3.7 12 7141 10 80 CES2LI.A Sormal return 6105 6205 HOOKRET 11 5177 2.0

.

				SEAD-DI TANEON	) • • • • • • • • • • • • • • • • • • •
		5 17	-		
			HOTE	This Poutine (	s formating lC-card for disk use.
		•	() ent	ry parameter	6 P
		4 9		NOSE	
		:	., ,	CT ; Beturn in	formation
		•		+0 ; 5	igreelly forestied
		•		+1 ; r	Cornal Grrov C Errov code
					IOL Soc mounted
				annual contact	•02 Cannot formet
			t pre	FORE	
		:			
6209	34 866C	CICUBES	10	A. (ICONTLO)	; Check IC-card gover-on status
620C	מנ		DEC	<b>A</b>	
6200	25 06		18	I, UICDAGG	; Algeady power on - If-rard mount check
6212	DA 517F		18	C , 800%887	Nount error
6213	01 0064	CICTAGO	LD	BC,100	; Neit 100 millisecond (Yor key-sound)
8219	CD 62C6		CALL	WAIT	
621B	JA FODE			A,(#ZCTL#3) 001)]]]]	Stop bey sound
621E 6220	32 PODE		1.0	(BZCTLE31.A	
6223	03 23		007	(207183),A	;
	CB 5214	ī.	eatt	# 1 L E N 10 L *	: Fill null code to 37A buffer
6223	CD 6465		CALL	ICCETHAL	Calculate max track & relord number
		-			
6228	24 PRC2		PT SH	NT (SKELEP)	Save Correct track & sector
6228	34 9464		LD	A,(STESTC)	
6232	PS		PUSU	A7	. Cat any leach A sector
6233	22 7402		LO	(S287881.NL	sat nam traca e sector
6239	<b>▲</b> ℓ		108	A	
6234	32 8464		LD	(SINSEC),A	:
6230		0100510	1		
623D	CD 631A		CALL	1 CCHERAE	; Check max track 4 sector
6240	10 22 cb 6181		CALL	0100100	; Uver mag focord ; Vrite nul] code into IC-card
6245	39 21		<u>2</u> B	C, TICDAJO	Srite error
	2. 2025		1.0		· Check power-off interrupt
6147	21		0.8	A	
6148	10 19		18	\$2,5103830	; Pewer-off happened
	14 P468		10	A. (SEESEC)	Tedeus tofoet field the tel ;
6250	36		: 50	Δ	
6791	32 7464		LD	(SEESEC),A	-
5254 / 164	20 85		28	NE, UICD010	
0430	• -	7			
6251	32 2464		10	(SENSEC),A El.(SENTRE)	; Sector No. 18 V : Set Dext track Number
5728	12		15 C	HL	
611F	11 1402		LD	(SIETER), BL	
5262	- s 38		. 4	L1C3310	8
3164		8128920	1		
6254			POP	A.F.	· Kormal Parking
6265	51.27		18	0122540	
6253	6514	0100530	CALL	1001500	: Close IC-card
5165	10 02		10	2,012	; free code
6282	1		PGP	14	
0168	2.4		201		, «FFOF FOTUIN
62.57		11123540	=		
6:61	32 7464		1.0	(SEESEC),A	. Bestere track & sector humber
6272	21 22 74C2		10	#1. (SEKTOK).AL	
6273 6276	78		LD	A , B	: Set return code
6277	c] 617F		3 0	HOONBET	:
ŝ UTILITY SUBBOUTINES NOTES The following soutines are utilities. : FILL HULL CODE 70 ONA BUFFER NOTE Fill null code (ESN) to DEA buffer. () entry persector RONE • 0 seture parameter ()
SkOk () preserved registers () NOVE \$27A FILLNELLS. 627A ; DMA buffer eddress ; 129 bytee ; Null code 24 9405 LD 8,50A A,025R 06 30 10 6277 32 15 10 6291 PILYULIG: 6291 77 (RL),A ; Set null code to BBA buffer LD 6292 23 INC 81 91130130 10 10 0283 D 3 N 2 6255 **c 1** BECEIVE 1 BYTE PBOT IC-CARD . -NOTE: Beceive deta from 1C-card. This routine must be called by all interrupt disable. 1 <> entry parameter NONE 4.5 NUME () FRUIC parameter () CT : Beturn information +0 : Normally received A Feq. is received data -1 : Time out error () preserved registers BC, DE, BL • 6286 120008 6256 CD 8288 ; lesst write mode ; Receive 1 byte from serial (IC-card) CALL IRESMAT 8289 11 SEVETT 2 This routine receives the data and note it to work. 525C ICEDET: 529C 62-P 6290 CE 6236 CALL 180005 : Becejve data Time out error ; Save received data 38 827 2 1811.4 1978 **C 1** SEND 1 BYTE 10 IC-CARD ......... XOTE: Send data to IC-card () entry parameter < > L : Sending data - C 3 SOSE () preserved register# () ST,DI.%1 ivecas: 5292 34 7004 10 A. | 22497C81 5492 15 16 75 8275 AP 111110118 5283 Disable Sz-coable status 122 8923 1923 21 200A 23 16 7.0 IZTARTCRI,A IZARTCRI,A 0 U J 6200 1 CALL 6290 CD 8284 ISPANET ; Set write mode 1.0 62A0 A, C SSDBYT 19 CD 6283 Send 1 byte to serial (IC-card) CALL INSCID: 62.44 0 B 15 g 6 04 29 7 A ? Check TR-empty ; (wait until send the data to carial) A. ( ZABTSB | 000001008 1.0 62A4 AND 6246 2,1WRC30 3.0 ; 62A8 1 CALL CD 6288 IRESWAT : Resat write mode .... AF (RZABTCB),A Restore Sx-enable status P 1 202 1 62 A D 32 PODA D3 16 LD 63 A E 041 A. (ZABTCR).A 6281 APPENDIX Page 21 - 176

.

121 : **C 3** 6783 -----SEI ON REET GRITE NODE NOTE: i There are two routines. One is setting write mode and shother is resetting write mode. () entry paremeter • ) ÷ RONZ 1 () return parameter • ROXE . preserved registers 4.8 • BC.DE.NL ISETNET: 6284 1.0 A. (BEICCTLB) ; Set write mode 3A 2008 AND 111110118 ISETSRES ; 28 PB 8297 6289 I BESNAT : 6238 1.0 A. (BZJCCTLB) ; Reset write mode 3. FODB F5 04 ..... 000001003 0.8 1 6287 ISZTARES: 6200 LD (BEICCTLBI.A ; 32 FODB D3 17 62C0 92C3 001 (ZICCTLE),A 2 EET ÷ 6205 C B ; ..... 2 į : NOTE: Softwald timer. If interrupt to harpeded, then the valting time is longer. ÷ ī 4) entry parameter ()
8C : Time (Per millimecond)
() return parameter () 1 RONE : () preserved registers . . AF.DE.81 VAIT: 6206 ; Seve register P098 P 5 6266 :0171A2 4267 A,230 1.0 : Loop counter 32 26 6107 WAITIO: 3 <u>1</u> 6209 Vert 1 millisecond 020 3.0 NZ . WA1720 20 20 2.8 62CA 7 ; Sait X sillisecood DEC 8 C 8200 0.8 **1**0 A , 3 79 0.8 C 82CE 62CT 81 10 20 .... SZ, NA1710 2 202 327 6201 11 6.8 6202 7 SEND DATA TO SERIAL NOTE: Send late to serial. () entry persector A : Send: • > A : Sending date ( totat pathelof SCN2 ) There () preserved registers All registers SSDBYT: 1213 5103 4204 Save sending data. Set IC-card power off time Set IC-card power off time Set IC-card power off check flag PUSH FE SA F026 SZ F2C2 SE 01 51 A. ( ECOPATRI A. ( ECOPATRI A TECPACATILA 10 10 613A 5.013 1.0 (ICCPMPLG),A LD -32 [203 1256 550810: 6207 West until Tr-ready 1.3 A.IZABISBI 08-15 6307 AND 00000018 26 01 29 PA 62E1 18 1.550810 62E3 101 P1 53 6225 Send the data to serial [ZABTBO81.A 001 6226 .... 6 9 6228

. . i RECEIVE DATA (ROM SERIAL ÷ 8017-Receive dota from serial. 2 ŝ () entry parameter NONE ÷. . . . ÷ () foturn parabetar () CT : Beturn Information 0 : Normally received
 A reg. in received data
 =1 : Time out argor ÷. 41 procerved registers () BC,DE,HL 5119 SEVETT: 6229 c £ PUSR 5 C ; ; 1et time-out counter ; (EPI2ATO) = 4 jg counter velue ED 48 2142 BC, (EP15810) 6224 1.0 6711 11 21 S1.4 C 6:70 C# 10 91 8 62 23 6222 5 L A e 6274 C3 10 8 8. 1 6276 38V810: 0.8 D 8 C 80 : Check time-out 62 1 79 A . B 1.0 6278 81 12 06 0.8 10 Time-out error code Time-out error 6279 ; 6278 29 14 38 2.584880 . 6270 08 15 1 3 Veit until Barrendy -6277 6301 16 02 25 23 4 N D J B 801000000 Z, S8VB10 : 3A P026 32 202 3E 0\_ 32 7203 ; Sot IC-card power off time ; tDefault is 30 seconds) 6303 LD 8. (ICCPVT3) 630 -LP (ICCPWCKT)... 1.0 A. C18 ; Set 1C-card power off check flag (ICCPNTLG).A 6308 LD 08 14 . Read received date 6367 1.8 A, (ZASTDIR) 5-1 6210 37 SATASS: 6311 31 .... 1. 6311 : Set return information 6312 c : c 9 POP 8.0 6313 887 . ÷ CLOSE IC-CABD ÷ i SOTE: Close IC-card. This routine must be run in interrupt disable. ÷ () entry parameter • > KONE Teturn persetter NONE 4.3 4.5  $\mathbf{C}$ preserved registere () SOKE 6334 :0521230: 13 13 2004 A. (BEARTER) : Save cuttent ABI-command testater 5314 5217 9USH AP 111110113 : 3 16 -9 ASE ; issat 3x-onacle 0 2 1 5 - CARTCRI, A 031A 217 . 120: AE 14 10: BE 10: B 10 N. (82300118) 000003009 (K1:00112).a ; Set 957 to high level 531C 621J 9311 9314 12 187 A.(EIICCILA) 11111008 (SIICCILA),A (IICCILA),A 12 522E : Disable cover interrupt 14 2002 33 20 35 33 27 35 33 27 26 AND Stop power-supply Stop clock-supply 5229 ì. 5312 5218 10 1 ; Reset IC-card flags ; Power control flag - 37. 6321 101 3.6 10 E102 11 7662 12 7660 2.2 LICCPRPLEN,A Using device flag Prwer on flag Cover status flag 9334 10 IICUSEDV),A (CONFLC),A ιD 6337 32 1661 ۱D (JCCVFLC),A 633A ÷ 02 01 C0 6346 10 Set serial port to 85-2320 C. 0 I H 6330 6337 CALL SELSER POP 6342 P 1 AP (ZARTCR),A 03 16 001 6345 **c** 9 827

		1	*****	***********	
				SELECT SERIAL	
			SOTE:		
				Select serial	line & change it.
				TV December	
				C : Line numb	• · ·
		1	() ret	urn paremeter	()
		:		NONE	
			() 986	NONP	
6346		SELSER:			
6346	34 8241		LD	A. (SENODE)	; Check current serial status.
6344	C 9		827	2	
				•	, 1000 D1 HEG 000
634B		SELSIO:			
6348	08 15		19	A ,   2 A & T S R }	; Sait until Tx-empty
6340	25 PA		18		
6353	19		LD	3,6	Set oew excludione
6352	32 2341		l D	(SENODE).A	
6355	07		BLCA		Cot now serial parameter table address
6316	6 1		ADD	A . C	
6358	4 8		LD	C , A	<b>0</b>
6359	06 00		LD	8,008	
6358	21 P196		10	BL.SETABL	8 8
6355	39		A D D	91,8C	* *
6358	3A PODA		LD	A. (BRARTCR)	: Disable Az-enable & Tz-enable
6 3 6 2	EG FA		A 5 D	8470	• • • • • • • • • • • • • • • • • • •
6364	CD 6344		CALL	S Z L S 8 0	2
6367		1	1 11	A (BL)	· Set For a rediater
6368	86 80		AND	0208	: (Bit "Ate/
6364	27		LD	(HL),A	
8368	JA POD6		LD	<b>A</b> ,(RZCTIR])	1
6362	26 OP		AND	8100	
63.0	33 8006				
6374	D3 00		001	(ZCTL81),A	
		7			-
6376	23		INC	UL .	
6377	4 B 2 3			8, (82)	
6379	72		LD	A. (HL)	Set SWR requeter
6314	26 00		AND	0008	; (Serial Line)
£37C	6 P		1.2	(81).4	,
6373	JA FODC		1.0	A.(825V8)	
6390			1 A A D 1 B	1811	1
6363	32 7000		12	(RZSVR) A	
6366	03 15		CCT	(2568), 4	- -
		:			
6388	23		CALL	5E 5F1550	
6250	15 15		LD	A. 3	Sot AETHE register
635D	12 2009		LD	A. (BRTSAIS)	(Parity, Sit Length, Stop bit)
5390	23 11		961	(ZASTNEL,A	*
5392	C 9		827		:
\$ 2 9 3		SELS50:			
5293	JA - CDC		15	A., A25¥8.	; Check serial line
5296	26 25		AND	820	
6282	TR		10	A. ( 8 L ) 57. 571 590	Not 29-232C
0-13	** +#	1.0			, . <i></i>
6392	t6 30		150	05 D N	; Set RTS & DIR control line
625D	11			\$1.4 	6 8
629E	1A PODA		12	A.IZZARTCRI	
62A1 -	20 22		13 N U 3 R	2 - 3 C	
9 <b>- 6</b> -	5 A			-	
4014		511530:			
2234	32 103A		13	(EIARTCRI,A	; Set ARTCE register
6347	33 36		ULT BIT	V Z A H I U N I , A	<i>i</i>
C J A J	6.3				

					*******
				CALCULATE BECO	80
		1			
			SOTE:		
				Calculate phys by seek track/s	lice) record number & loop counter motor.
		2	()	EV Decemeter	4.5
		•		(ICNECS2)	l physical record size
		•		( [ [ ] ] ] ]	Birectory record number
		-		(3EKTBE)	Seek track number Seek tector number
			() ret	urn parameter	()
		•		CY : 1 Virt	unl record
		-		0 Phys b2 : 8	IICAL FOCOPO Incord Anabar
		:		1 1 1	.oop coulter
		:	() pre	served regasters	
6344		ICALROD	14	NUNE	
6 3 A A	3A 2852		10	A, ITPICCARD)	; Chack wartul record
6340	47		10	8, <u>A</u>	; Directory record number
6381	88		CP	8	TPICCARD sector to 1 sector of 0 track
6362	39 08		18	C, TOLKIO	; then virtual record
8384	FE 05		69	068	
8388	JU UT 24 24C2		18	NC.IULKIU RI.(SPRTRK)	
0380	76		LD	A , B	
6380	85		01	L	•
0380	37		367	,	: • Virtual record
			•••	•	
6387		IBLEIG:			10 s
6362	36			A_(DPUZ+1) A	Directory Compet
6363	CD 6379		CALL	CALKADE	Calculate logical address
6368	2 A 908 A		10	HL, (LOCADB)	
6365	61 6JCR			A ICHECCHE	celculete DoAelcel equites
6300	C B		8 2 T	-	
		4 7 8 9 1		CALACULATE PBY	SICAL BRCOBD VOYBRB
			NOTE:	Calculate phys	ncel secord pueber
		6 		77 DATABOLOT	()
		•		BL : Logical a	ddroee
			() ret	urn parameter	4)
				B : Loop coun	record number
		:	<+ pte	served registers	
				NONE	
1301		ICHICON	£ :		
6202			ICE	A	Calculate record number (Per 128 bytes)
6301	CB 14		8L 8L	L	1 DF / 123 ++> 9F
6303	17		1 L A		
6304	57		LD	9 , A	
8733	36		L D	x.8	
6306		ICRECIO	:		
6306	62		10	8,3	Copy data
6305	14 7125			A. ( 2082052 )	: 1 Inveicel record 1110
8368	41		LD	C . A	Check physical fecore number
1300	2 2 50		LS	A	
3 J Ü E	46 41		1.7	8.010	i
6360		ICREC20	:		
8380	91		51.8	c	; § req. is loop counter
6 J E I 6 J F 2	53		487	2 F	- 1 11 108: 18 14Azycyv 100010 Unbool
4363	0.4		150	3	
1324	E 8		22	D2.8L	;
6362	13		A U U 7 R	HL,DE DE BI	; Secord number (Fer [CBECS2] -
6387	0 9		82.1	6	; lecord number overflow
6328	18 76		3.8	1688620	:

			CHECK TRACE/SE	CTOR OVERFLOS
		K I	370: This coutine c But it neglects couct be always	hecks overflow of track/sector. Liret byte of track, so first byte zero.
		•	) ONTRY perametar DE : Mammada ( (SEKTBK) Cu (SEKSPC) Cu	() teck/record number frent track number frent sector number
			CY: 1 Trac CY: 0 Addr	t) k/sector overflow ees O.K.
			BC, DE, NL	
632A 632A 6320 6328	3 A 7 4 C 2 B A 3 P		D A. (SENTRE) P D E P	; Check trock number ; Trock number unmatch? ;
6227 6360 6371 6374	D 0 C 0 J A F4C 4 B B		ET SC ET N2 D S,(SEKSEC) P P	Yes Trock number overflow Check eector number
6375 6376 6377	C S 3 P C 9		ET E	Match
		•	CALCULATE 10C1	CAL ADDRES⊆
		N	Calculate log:	cal address by seeking track/sector.
		4	) entry persector A : Directory (SE6786) S	() No. ()f correct) est track number
		<pre></pre>	ISTESTC) 9 Toturn parameter (1064081) 1	est sector number () ndca) address
		<pre></pre>	preserved registers	
6379	<b>7</b> 5	CATRADS: :	USB	; Save ail requesters
6379 6378 6279 6274 6378	<b>75</b> C3 D5 E5	L L L L L L L L L L L L L L L L L L L	USB AF CSN 8C CSN DE USE BL	; Save all Peqisters ; ;
6379 6378 627A 6378 6370 6370 6370	<b>r</b> 5 C 5 D 5 E 5 <b>2</b> 1 <b>0</b> 0 0 0 <b>3</b> 7 <b>2</b> 5 <b>1</b> 7	0   0   0   0   0   0   0   0   0   0	USB AF USN BC USN DE USE BL D BL.000CN B A B 2,CALZO	Save all requesters Check offset record Von need of correct
6379 6378 6.279 6378 63778 6377 6377 6377 6400 6402 6402 6402 6407	<b>75</b> <b>C</b> 5 <b>D</b> 5 <b>E</b> 5 <b>21</b> 0000 <b>37</b> <b>29</b> 17 <b>47</b> <b>24 T4C2</b> <b>87</b> <b>20</b> 0 <b>7</b>	11 C V T R V D B : C V T R V D B : D I D I D I D I D I D I D I D I	USB AF USB BC USB BC USE BL D BL,000CN B A C ALZO B A C ALZO B A C ALZO B A C ALZO B A C ALZO	Save all Peqisters Check offset record Non need of correct Current record is in directory area; No
6379 6378 627A 637A 637C 637C 640C 640C 640C 640C 640C 640C 640C 640	r S         C 5         D 5         E 5         2 1       0 0 0 0         3 7         2 9       1 7         4 7         2 A       F 4 C 2         8 7         2 0       0 7         3 A       F 4 C 4         P 2       0 0	C Y T R Y D B : C Y T	US8 AF CSN BC CSN DE USE BL D UL.000CN A A C ALZO B A A A A C ALZO B A C ALZO B A C ALZO B A C ALZO C A C C ALZO C A C C C C C C C C C C C C C C C C C C C	Save all Peqigters Check offset record Non need of correct Current Pecord is in directory area; No
6379 6378 6.29 6274 6377 6377 6377 6377 6377 6402 6402 6402 6407 6407 6407 6407 6402 6402	rs         cs         ds         ts         21       0.0.0.0         37         29       1.7         47         5.4       f.4.0.2         87         20       0.7         3.4       f.4.0.2         87       0.0.0         3.5       0.5         3.5       0.5	CALIG:	US8 AF CSN BC CSN DE USE BL D BL.000CN 8 A 8 2,CAL20 D B.A 0 A,(SESTES) 9 A 8 NZ,CAL10 D A.(SESEC) 9 OSN 8 C.CAL20	Save all Peqisters Check offset record Non need of correct Current Pecord is in directory area: No Tes
6379 6378 6.29 6274 63778 63778 63778 63778 63778 6377 6402 6402 6402 6402 6402 6402 6402 6402	rs         C5         D5         t5         21       0000         97         25       17         47         25       17         47         20       07         34       P4C2         87         20       07         34       P4C4         PE       05         35       09         3f       IC         90       5P	: CALKADB: P P P P P P P P P P P P P P P P P P P	US8 AF CSN BC CSN DE USE BL D BL.000CN 8 A 9 A, (SESTES) 9 A 9 A, (SESTES) 9 A 9 A, (SESTES) 9 A 9 C, CAL10 0 A, (SESEC) 9 OSM 9 C, CAL20 0 A, (SESTES) 9 A 9 A 9 A 9 A 9 A 9 A 9 A 9 A	Save all Peqisters Check offset record Non need of correct Current record is in directory area: No Tes Calculate correct time -32 - Directry No.1 = 12 +++ H1
6379 6378 6274 63778 63778 63778 63778 63778 63778 6402 6402 6402 6402 6402 6402 6402 6402	rs         cs         D5         ts         21       0000         97         25       17         47         25       17         47         20       07         34       P4C2         87         20       07         34       P4C4         PE       05         35       09         3f       IC         90       6         6       05         CE       6450	: CALKADB: P P L L L L L L L L L L L L L L L L L	US8 AF CSN BC CSN DE USE BL D UL.000CN 9 A 9 2,CAL20 0 8,A 0 4,(SESTES) 9 A 9 NZ,CAL10 0 4,(SESTES) 9 A 9 NZ,CAL10 0 5 M 8 C.CAL20 0 5 M 8	Save all Peqisters Check offset record Non need of correct Current record is in directory area: No Tes Calculate correct time -22 - Directry No.1 = 11 +++ H1
6379 6378 6378 6378 63778 63778 6377 6402 6402 6402 6402 6402 6402 6402 6402	<pre>rs cs 21 0000 97 29 17 47 20 07 34 r4C2 87 20 07 34 P4C4 PP 05 39 09 3f IC 90 6F 66 05 CE 6435 </pre>	: CALKADB: P P P P P P P P P P P P P P P P P P P	USB AF USB BC USP BL USP BL USP BL USP BL USP BL USP BL USP CALTO D A, (SESTES) B A B VZ, CALTO D A B A B A B A B A B A B A B A B	Save all Peqisters Check offset record Yon need of correct Current record is in directory area: Yo Tes Calculate correct time -32 - Directry Yold * II *** HI Save correct size
6379 6378 6274 6378 63778 6377 64002 64002 64002 64002 64002 64002 64009 64009 64009 64009 64009 64009 64009 64009 64009 64009 64009 64109 64002 64000 64000 64000 64000 640000000000	r S         21       0000         25       15         25       17         27       17         29       17         47       147         20       07         34       9404         90       90         38       20         38       20         39       09         38       20         39       05         38       25         29       15         38       25         39       05         38       25         39       05         38       25         39       05         38       25         39       25         39       25         39       25         39       35         39       35         39       35         39       35         39       35         39       35         39       35         39       35         39       35         39       35	: CALKADB: P P P P P P P P P P P P P P P P P P P	US8     AF       US8     8C       CSM     DE       US8     BL       D     HL,00000M       B     2,04120       D     A,05857253       B     2,04120       D     A,05857253       B     V2,0410       D     A,058520       P     OSM       B     C,04120       D     A,1572       C     A       D     A,22       US8     C,058       D     L,358       D     L,1528       D     NL,1528       D     NL,1528	Save all requisiers Check offset record Non need of correct Current record is in directory area; No Tes Calculate correct time ( 22 - Directly No 10 HL Save intrect size Change tracs number to sector number Trace number = 54
6379 6378 627A 6378 6378 63770 64002 6402 7406 74057 6402 84002 6402 84002 6402 84002 6402 84002 6402 84002 6402 84000 8400000000	rs         21       0000         97         29       17         47	: CALKADB: P P P P CALVO: CALV	US8     AF       US8     8C       CSM     DC       US8     BL       D     HI.00000M       B     2,CAL20       D     A,GSTTES!       B     2,CAL20       D     A,GSTTES!       B     2,CAL20       D     A,GSTTES!       B     2,CAL20       D     A,GSTSTES!       B     VZ.CAL10       D     A,GSTSSEC)       P     OSM       B     C.CAL20       D     A,GSTSSEC)       D     A,STSSEC)       D     A,STSSEC)       C     A,SSSEC)       C     HL,GSESEC)       C     HL,GSESEC)       C     HL,GSESEC)       D     HL,GSESEC)       C     HL,OF	Save all requirers Check offset record Non need of correct Current record is in directory area; No Tes Calculate correct time 132 - Directry No.1 * TI *** HI Save intrect size Change track number to sector number Trace number * S4 faiculate total sector busber
6379 6378 6378 6378 6378 63770 6400 6400 6400 6400 6400 6400 6400 6	rs         21       0000         97         29       17         47       14         54       r4c2         87       20         70       07         34       94c4         90       90         35       09         36       20         37       20         38       09         38       09         38       09         38       09         38       09         38       15         20       6         37       17         18       14         19       0         19       0         10       000         06       07	: CALKADB: P P CALKADB: P P CALIO: CA	USB     AF       USB     BC       USP     BL       USP     A       SQ     C       USP     A       USP     C       SQ     A       USP     C       SQ     A       SQ     C	Save all requirers Check offset record Non need of correct Current record is in directory area; No Tes Calculate correct time -32 - Directry No.1 * TI *** H1 Save infrectry No.1 *** H1 Save infrectry No.1 *** H1 Save infrectry No.1 *** H1 Save infrectry No.1 *** H1 Save infrectry No.1 *** H1 Save infrectry No.1 *** H1 Save infrectry No.1 *** H1 S
6379 6378 6274 6378 6378 63778 63778 63778 63778 64002 64000 64000 64000 64000 64000 64000 64000 64000 64000 64000 64000 64000 6400000000	rs         21       0000         37         25       17         47         20       07         34       r4C2         37         20       07         34       r4C2         37         20       07         34       r4C4         72       0         35       09         36       2C         64       35         C5       64         74       r;         75       7.4         76       0.5         77       20         76       0.5         77       20         78       r;         79       0.5         74       7;         75       7.4         76       0.5         75       7.4         74       7;         75       7.4         74       7;         76       0.0         76       0.7         78       7.4         74       7.6         75       7.4	: CALKADB: P P P P CAL20: CAL2	USB     AF       USB     BC       USB     BC       USE     BL       D     SL.00000       B     A       B     2.00000       B     A       B     2.00000       B     A       B     2.00000       B     A       B     2.00000       B     A       B     YZ.00000       D     A.127       B     YZ.00000       D     A.27       USB     3       D     I.35000       C     C.01700       D     A.27       USB     3       D     I.35000       C     S.0000       S     2.0000       S     0.0000       S     0.0000       S     0.00000       S     0.00000       S     0.00000000	Save all requirers Check offset record Non need of correct Current record is in directory area: No Tes Calculate correct tite Change track number to sector number Trace number = 54 fairwlate total sector number Calculate logical address Total sector number = 129 (BLDE) = 2 (NLDE)
6379 6378 6378 6378 6378 63778 63778 6402 6402 6402 6402 6402 6402 6402 6402	r S         2 1       0 0 0 0         3 7         2 5         1 7         3 7         2 9         1 7         3 1         7 4 7         2 0         3 7         2 0         3 4         7 4 7         2 0         3 7         2 0         3 7         2 0         3 4         7 4 7         2 0         3 7         2 0         3 7         2 0         3 7         2 0         3 7         2 0         3 7         2 0         3 8         3 9         3 8         3 7         7 8         7 8         7 9         2 8         2 1         3 9         2 1         3 9         2 1         3 9         2 1         3 1         3 1         3 1         3 2         3 3	: CALMADB: P P P P CALMO: CALMO: CALMO: CALMO: CALMO: S C S C S C S C S C S C S C S C S C S	USB     AF       USB     BC       USP     BL       USP     A       SQ     CALTO       USP     A       SQ     CALTO       USP     SE	Save all requirers Check offset record Yon need of correct Current record is in directory area: Yo Tes Calculate correct time (32 - Directory Yold = TD +++ HI Save intrectory Yold = TD ++++ HI Save intrectory Yold = TD ++++ HI Save intrectory Yold = TD ++++ HI Save intrectory Yold = TD +++++ Save intrectory Yold = TD ++++++ Save intrectory Yold = TD ++++++++++++++++++++++++++++++++++

6438 C 1 .... 10 : Correcet Address ; BC reg. im offsot value ; (MLDE) - (BC) --> (MLDE) 08.81 0430 83 6.2 643D 87 80 A H L . 8C 6432 20 42 580 .... 8.3 02.81 80,0000M 23 01 0000 LD 6441 .... ED 42 580 81,8C . ED 13 POGA LD .... (LOCADE), DE : Set logical address 644A 644 D 7 11 LD A . L 32 1060 (LOCADR+2),A 10 . 20.2 Bestors all registers 644E 21 N.L. POP 6447 D 1 3.0 6450 **C**1 POP 30 6451 71 63 POP .... T38 : : Shift HL register by 8 register. 1453 411 8 7 7 -8453 8455 6457 6459 1 ; H1 + 2 ++ 8 --> 2L CB 25 5 L A CB 14 10 PA BL PJNJ 90171 6.9 887 5 \_\_\_\_\_ ..... 1 UPDATE IC-CABO BECORD NUNEEB 2 ÷ NOTE: ÷ Select serial line & change it. 1 () entry persector . i ; . 26 IBLECON: .... PUSH 6 5 A ٢5 38 i ] physical record size 34 1115 LD A, (ICBECSZ) Calculate next logical address 6452 5.0 L D 2,4 0,00M 16 00 845 P LD 81,28 01 6461 19 ADD \$462 01 POP Sent physical record number 8463 1 8 6 13 3 0 .... 6 8 111 : CET IC-CARD NAX IBSCE/SECTOR ï 3078: Cot essione track/sector of iC-card. -3707 \$465 ICCETSAX: ; Check IC-cars power on status 8443 10 34 7560 A, (1005715) 82,C1CD861 78 01 C4 81D5 : Power off then sound thech .... CALL j deuns error 6460 2.5 EIT. C 2A 765C 6461 L D R1, (ICMAI2EC) ; Check eax record number 6471 6472 8473 6474 1.0 A . Y 1 CE 2.5 3.7 5 C 7 1 30 100310 z C 5 511 1 0.75 0 C C 81 2 5 - 2 add the ease of variabl record 6416 34 7657 A. (TPICCARD) L 10 6473 647A 8 . A A , S 4.2 L D 38 08 LD 6470 9 0 5 U B h 6470 C . A 4.8 1.0 8472 06 00 в, сон 1.0 8430 0.9 ADD HL, BC 2 1 ; Calculate max track/sector с,40н 6481 06 40 L D 0,00H 8483 16 00 10 5 1CCET10 58C 8483 NL.8C ; Subtract records per track 20 42 APPENDLX Page 21 - 182

35 03 6152 1.0 C, ICCET20 1 14 INC Track counter 1 up 6459 D 18 89 648A 1.0 1001710 1002720: 6480 09 64 A C A TI D NL BC 1 6460 5.0 C,L Set record number 1.0 6452 87 OB A 6458 6.9 8.5.7 - 1 : CHECH IC-CARD INTERRUPY NOTE: Check IC-card status () entry parameter () (1CSAIBEC) -- Jazizwe record number ī () return parameter ()
CT 1 +1 -- Error AC, DE, BL 6490 ICINTCRS: 34 7661 LD A. (ICCVPLG) Chack IC-card cover status 6490 87 6493 .... Error code A . 22211 20 6494 37 SC 7 6496 \$ 7 6497 r 0 BZT Cover 10 opened 1 Check JC-card power status 6499 3A 2880 LD A. ( ICONPLC) INC 6198 10 . 32 09 LD 6491 teror coos 649E 65 8 Z T z Power is off 0. FASP 8. ۸ 5 9 ; Normel return 6440 ł : ŝ SEND CONDAND & RECEIVE ANIMER i ş NOTE: Send command and receive answer. ÷ <> entry parameter • > C : Command code ÷ = 61 -- Read record = 02 -- Vrite tecord 2 DB : Record number N1 : Buffer for data sending/reciving () return parameter CT : +1 -- Prror +0 -- Normal end 1 () preserved registers () BC, DR, B1 1 ICHDPSC: 6441 ; Checs Il-card .sterrupt CD 6490 CALL ICINTCEE. 64A1 64A4 88: с А. 5 36 Check legical commany come-79 /2 01 LE 6445 1 C 7 016 5440 75 06 2,1050130 028 38 Read function 6445 c 9 6 4 <u>4</u> A 15 1A 3 8 2.1080200 Frite Lunction 6440 1 502 S 4 A E 387 6445 ÷ following programs are reading from or writing inte Friend, but they are concerned to in-carrie pressed in they are not a sector. You are these routimes as your in-carr working. . . ÷. 11151 SND Sactosi Syabole: 6410 CALIO 0004 ASSCNT 6419 CAL20 0002 APLOON 6431 CA130 6315 CALKADB 280C 5177 CEST 2809 CONTR CONDUT T 1 5 2 1405 BOANEO 0 P 8 2 1000 CRC16 1172 0 8 M 2 L F17C D 5 M 2 L F1AB 0001 0008 E 8 8 1 1 0006 8886 E B B T FILNUL10 627A PILLNULL 6391 0009 6883 .... HOOMADDR 6080 HOOKDATA 6 1 7 P HOOKBET I BLEIO 6454 1818688 6 J A A ICAL BCD 6387 6314 1001500 1202 ICCPVCNT 6 J E A ICCPWFLG ICCETIO F 2 0 3 1026 ICC PNTH F 5 6 1 ICCVPLC 648C 1001720 6465 ICCETHAN 6485 6490 64C8 F126 I CHONDY L ICINTERS. F 6 5 C 10143820 6441 ICHDENC. 100200 6480 1001001 APPENDIX Page 21 - 183

1001	ICONBED	0010	ICONTRN	0002	ICONVET
-660	ICONFLG	6595	108010	657E	1080016
< 1 8 8	1080147	625C	1 C R D D T	6306	1088010
1320	1 C B E C 2 D	6 3 C E	ICAECCHK	F185	ICRECSZ
4534	1080010	6538	1 C R E D 2 O	6523	ICREDDAT
ô 1 9 D	) C & A O O	61 A D	103810	618A	1 C R R 2 0
TIAE	ICESTPNT	6 1 8 C	108500	6169	168910
6106	108620	7663	1 C T D D T	F1E9	167507
r 6 6 2	ICUSEDV	654E	ICVETIO	6542	ICWRTDAT
6578	LODICEC	6256	180005	6566	180686
651F	1820520	6478	1860375	6288	IRESWRT
6200	15279825	6284	ISETVRT	64DE	ISHDCOM
6575	IFSTCSC	62A4	148610	6292	IN BCOS
6550	) WBCBC	206A	LOCADA	1000	MAINSP
0064	NODE	6453	HULTI	565E	GTICCARD
0001	80CMD	0020	RECSIZE	FODA	#ZARTCR
	E ZARTHB	7006	RZCTLAI	#0D#	BICTLB3
FODB	BZICCTLE	F426	RZTEN	7000	BILOCTIB
f00c	BISNB	F4C4	SEKSEC	74C2	35735
6348	SELS10	6393	SE1550	E D A 4	521590
6346	SELSEB	P 2 4 1	585002	\$196	SETABL
6276	58v810	6311	587890	6289	SEVBTT
6200	550810	6 2 D J	SSDavt	6055	START
0002	STI	F197	STSARTHR	F651	TPICCARD
0038	TSBTI	6 C D S	ATAGSU.	6549	0807807
.005	CBDTTOP	6157	UIC0110	615E	0100120
6171	U1CD130	6178	U1CB150	617C	<b>VICD190</b>
6120	U1C370D	6176	CICD720	6215	0100900
623D	0100910	6264	0100620	0165	0160820
6167	CICDS40	6010	LICDNE1	6177	CICDHE2
8179	UICONEJ	6177	UICONE4	6198	VICOBES
6191	DICDNEO	6109	UICONE7	0209	CECONKS
82C 6	VAIT	62C7	VAIT10	6269	NAIT20
0050	MAITTIME	5903	¥8001	0002	W163D
P 0 8 5	TPOPST	0016	2 # 8 T C R	0014	ZARTDIR
0014	2421008	0015	ZARTHR	0015	ZARTSR
0000	101141	0023	207183	0017	ZICCTLE
0004	2128	0019	ZIOCTLB	C Q 1 6	872013
0039	Z S 6 2				

No Fatal errortal

.

```
(34) SAMPLE 34. FDD Format/Copy utility for the EHT-10/EHT-10/2
        < Overview >
        This program enables FDD format/copy by using PF-10 or TF-15 via
        RS-232C I/F of EHT-10/EHT-10/2. This program does not work
        by itself. It should be included in an application.
        < Function >
        This program has the following functions.
        (1) Format "D" drive.
        (2) Copy from "D" drive to "E" drive. ( In this case, you should
        use Dual type TF-15 )
        < How to use >
        (1) Include this program in your application.
        (2) Added the process of extended subroutine. (CALSUB1)
        You can do the special process by using CALSUB1 (Ex. Display the
        track No. on the LCD. )
        (3) Set the address of CALSUBI to CALLADI, and call FDDUTY.
                           .
                                            () Assemble condition ()
                                   i
                                            . 280
                                   ÷
                                           () Stort address
                                                                      ()
                                   1
                                            . PRASE
                                                              100N
                                            . CONNENT
                                   ;
                                                     Copyright (c) 2950H 1986
                                   2
                                                     Created
                                                              bу
                                                                       T. Teneka
                                                                       1986. 7.25
                                                                                         ver 1.0
                                   [COMMAND NAME]
                                                     (D, E)
                                                          . OISK VOLUME COPY
                                   : FUNCTION 1
                                                                      Copy from D drive to E drive
                                                          DI PORMAT
                                                                       Foreat 0 drive
                                            SUBTTE SYSTEM CONSTANT
                                   ÷
                                       .....
                                                     PROGRAM CONSTANT
                                                                            OOJPH : Data cend to 19 & receive troe IF
OOJ2H : Only receive data ... receive ACM
 001 .
                                   2 P S P S N D
                                                     του
 0022
                                   Z P S P B C V
                                                     EQU
                                                             OPOBAN : Power off interrupt disable (lag
OPOBSN : Power off status flag
OPOBSN : Alaro/Wabs interrupt disable flag
OFOB7N : Alaro/Wabs status flag
OF720N : CTRL/STOP flag
OF41BN : Destination bank
                                                     2 Q U
 .....
                                   YPOPDS
                                                     200
 1985
                                   YPOPST
 P11 8 6
                                                     8 O U
                                   YALMDS
 1087
                                   VALMET
                                                     £ D U
                                   CSTOPFLC
                                                     8 Q U
 r 220
  F418
                                   DISBNK
                                                     101
 E 8 6 9
                                                     2 Q U
                                                              CALLX
 JPST8109
                                                     8 Q U
                                                              022988
 0029
                                   TRKHAT
                                                     E Q U
                                                              40
                                                                     i Track ess nue
                                            Vork area srange
                                   UNSTART.
                                                     6 Q U
                                                              OPBAAH ; System work area
 .....
                                                     E Q U
                                                              WKSTART +00
                                   CALLADI
 C B A A
                                                                       ; 1 track access
                                   :
                                   2
```

			:	Followi	ng ere t	e + porary	work area	1	
FRAC			DINTPLC		8 <b>G</b> U	CALLADI	• 0 2		
							Interru	ipt status (A)weye use	1
2 8 A D			TBACE		2 Q U	DINTPLC	• 0 1		
			:				; current	track number (Always	
			2		F data e	end/zece:	ive		
					PAD				
I O A E			PACELT		E.C.C.	TRACE	• 6 1		
7842			F 51 T	EQU	FACERT		; format	0= (con master to ala	
			010	EQU	PHT+L		; Destina	tion device 10 Jlh-D	Ι, Ε
7880			518	ZQU	610+1		: Source	device ID 23hourseler	•
			812	700	196+1		Data et		
7883			0 4 7	ZQU	912+1		; T.8.N., D.8	W. SEC. TTPE, 1 escto	r data
							; etc		
1030			801	200	DAT+137				
					3-/// 66				
			1	******				1	
					374810	<b> </b>	,,		
			•						
1000			5 <b>7</b> 4 C E	1 G U	10000				
			-	2011					
2203			CONTR	100	019088				
£ 10C		+	CONDUT	290	820620				÷ .
2857			PUTPJE	EQU	028678				
					0.4.8				
0000			C B	200	000				
0020		-	SPACE	290	208				
									5
0100	3.1	1000	5 T 4 U T :		SP.STAC	6	: Set sta	ch pointer	
0103	3 O	01		10	C,01		Define	key table for HC-10	
0105	0.0	01			8,01		1		
0107	50			CALL	PUTPPE		:		
0103									
0100	21	0160		10	MI,MSCO	٥	Bisplay	opening measure	
010 /	CU	0133		CALL	836436		•		
			;	Hate par	chet date	•			
			•				Subsout	ine call	
0112	21				/CALLAD		3001000		
011-	C D	0193		CALL	FDDUTT		60 11		
0118	38	8 0		38	C, 28808		: Error		
							- Dicolay	anding message	
0110	21	01/9		CALL	DSPHSCE	1 a	i unipreș		
6123	63	8803		3.8	¥8007				
				Dieplay	erfor co				
0128									
0124	25			PUSN	A7	_	; Save er	ror code	
0.27	21	0188		LD	NI,NSCO	2		arror accorde	
0124	C 8	0123			A.F.		7		
0120	C D	0143		CALL	DSFUEI		; Display	error code by here c	ade
6131	CD	2809		CALL	CONTR				
0134	<b>C J</b>	E B O 3		3.6					
			;	Subrout	ine 1 (C4	alled whe	in 1 track	ACCERS)	
0137			SUB1:			K 1	: Current	track number	
0127	3 4	0143		CALL	DSPHEX		Display	it by hear code	
0134	02	20		LO	C, SPACE		; Dieplay	apace code	
0137	C D	2065		CALL	CONQUT				
0142	C 9						*		
				Display	binary (	data by h	esa code		
0143			DSPNEX:	RELE	AP		:		
0143	6 ¥ 0			BRCA			;		
0145	0 P			BRCA			1		
0146	C P			RRCA					
0147	0 7	0100		CALL	05920		Display	upper 4 bits	
0148	C D 9 1			POP	AF				
0140			05920;						
0140	26	0 P		AND AD0	4.90P		:		
0148	27	90		DAA	2,308		*		
0130	ct	4.0		A B C	A,40M		•		
0153	21			0 4 4			•		
			הפסע		age 21 -	186			
			rur r Ka						

				C.A	i
0154	4 F C.D. 2 B G C		CALL	CONCUT	
0155	C 9		857		
0123		1			
			Velgaio	strings until	find DO code
		1			
0159		DSPMSC;			
0159	72		10	A L ( N L )	; Get data
0154	87	(	0 8	A	
0158	C 3		A E T	Z	: Delimeter
0156	4 1		L D	€.▲	
0150	ε.5		PUSH	иТ	• •
0158	CD 288C	(	CALL	CONCUT	
0151	E 1		POP	HL	• •
0162	23		INC	B L	4 9
0163	18 74		3 8	DSPHSG	a a
9100		;			1. 1
		1	Display	strings & wai	t until any Key in
0161		DSPMSCsl	N :		
0105	CD 0159	1	CALL	DSPMSC	
0165	CD 8809		CALL	CONIN	
0168	C 9		8 <u>7</u> 7		- -
0100					
		;		date	
0160		HSCOO:			
0180	20 46 68 72		0 8	' Pormating',	CE, LF, 00
0100	60 61 74 69				
0170	68 67 00 0A				
0174	0.0				
01/0		MSC01:			
0113	00.04		08	CR, LF	
01/3	20 43 88 60		08	' Completed',	C8, L7, 00
0110	20 66 65 74				
0177	55 64 0D 0A				
0103	0,				
0191		15 C J 2 :			
0198	0 2 0 4		8 0	CR LP	
0-75	46 22 27 68		0.6	'Error'	
ALLO	1 20 20 20				
0.42	0.0	1	8 8	0 0	
0141	66		SUBTTL	JAIN BOUTINE	
				Floppy disk u	tility
					**********
		10 A			
			ED Entry	parameter	$\langle \rangle$
		-		None	
			E Betu	n sereeter	4.5
				CY-flag : Bet	ven information
				• 0	: Normally and
				+ 1	: Error bappened
		:			A res, is error code
					ant : Force stop
					61 : Device non-connect
					62 Computication error
					By ; force stop
		:			to ' prime cuject curve Lo ' attra stint
		:			AD - Nairo anoice, couce AD : Alian deiser alles
		:			AN : AAlfa Biorary attol
		:			PL I
		i	C> Pres	erved register	9 ()
		;		BC'DS'HF	
6193		P00UTT:			
0193	E 5		PUSH	8 L	2 20A0 LG8/9fele
0194	23		PUSN	DZ	; · · · · · · · · · · · · · · · · · · ·
0195	C S		NZUSH	BC	
		:			
		\$	leset di	lak system	
		:			
0196	CD 017D	4	CALL	RESET	BODGE OLIAS
0199	38 03		3.8	C, FDDUTY90	zeror
		:	_		
			Start of	i disk operati	0 N S
		1			
		- P		If you want t	o copy diskette, you replace
		1		the Statement	CALL POSHAT' to 'CALL DISKCOPY'
		;			
0198	CD 0193		CALL	FORMAT	; Do formating::
1910		PDDUTT90	:		
0192	C 1		POP	80	; Bestore registers
0190	01		POP	9.0	;
0140	 F 1		POP	нL	:
0141	c 9		T 3 8		:
0 4 4 4					

			SUBTTL	FORMAT	
		-		Pormat D driv	•
		1	******	************	
		i	-		
		•	() Eot	ry parameter	0
				3000	
			() 895	orn garageter	Normal and
				CT-FINE : 40	- Prroz bananed
				- 1	(A ref. la error code)
					• ()
			.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	All without A	100.
			() Xot		()
				If you want t	a format other drive,
				you must chan	ge "DAT+0" dete.
0142		FORMAT	:		
0142	e S		PUSM	B.L.	; Save registers
0143	D 5		PUSH	0 2	
0144	C 5		PUSH	80	
			HAZO E	eresting packet	
	31 6263	÷			· Potest backet address
O J A S			C . ! !	CORRA	· Conv compand data
0140		•	CALS	CUPTE	
		•	formet	connend (Forme	t first track)
6148					
0140			101	A	; Set initial track counter
0140	32		LD	(TBACK).A	(to formating, it isn't used)
0142	C0 0215		CALL	DENTPWALN	; Disable power off & alars interrupt
0182	CD 02A0		CALL	CALSUBI	
0185	87		0.8	<b>A</b>	<b>i</b>
0116	CD 0262		CALL	SENDSLAVE	: Send command to FOD & Cot O track ACE
0189	30 33		3.8	C, /08HA190	; EPSP error ·
0188	34 1482		LO	A . ( DAT+2 )	
0132	0.		0.	•	; Check return code
0187	37		367		
0100	20 20			NI, FUENAISU	; .DD error
0103	CD 0111		CAT.		· Poshie newer off interver
0162	34 37		18	C PORMATSO	, Porce dien
0101					,,
			Track	oumber 0139 £	crmstios
0167	08 21	-	LO	B, TREMAX-1	; Loop counter
0109		FORMAT	10:		
0169	JA FBAD		LD	A,(TBACE)	; Set next access track number
0166	3 C		INC	<b>A</b>	:
0100	32 78AD		LD	(TBACH).A	:
0100	CO 0215		CALL	DINTPWALM	; Disable interrupt (power off & alars)
0103	CD 03AD		CALL	CALSUBI	;
0106	37		967		
0107	CO 0262		CALL	SENDSLAVE	; Becelve ACE cade
01DA	38 12		12	C.FURNATSU	EFSP ertot
GIDC	JA VEES			A.(DALV-)	
0107	87		508	-	, CDACE FALWED COUR
0180	30 08		3.0	SPTARON TR	
	20 00				, who mitted
0123	CD 0221		CALL	RENTPH	: famble mover off interrupt
0176	18 06		38	C. POBRATSO	
0128			DJNZ	POBMAT10	Loop 39 times
					· · · · · · · · · · · · · · · · · · ·
OJEA	CD 0750		CALL	EINTALN	; foable alars int-crupt
0)80	87		0 2	<b>A</b>	; Ndrmal raturo
0122		POBNAT	90:		
0188	C 1		POP	#C	; Restore registers
0127	D 1		POP	0 6	
0190	F1		FC F	# L	
4171	- Y		HE I		0
				81386097	

÷ Copy from D drive to E drive ÷ ÷ c) Entry parameter None • • ----() Beturn parameter 0 n persmeter () CY-flag ; +0 -- Normal end +1 -- Error happened : () Preserved registers All without A reg. · Note 6.5 а. 11 you want to format other drive, you must change '351'0' deta. DISECOPY: 0172 8 L ; Save registers PUSH εS 0182 5 PIISN 0 1 c 1 PUSH B C 0174 1 Nabe volues copy command packet ÷ ÷ 1.0 : Volume copy command 1 02C9 H.L. PACKCP 0175 COPYA CALL 1 Sand volume copy command & Cet soewer ٠ ; Common program with formating 3.8 19 48 0178 SUBTT1 Boset dish drive 1 Seset diak drive ï 1 C) Entry parameter None . . ÷ None () Beturn persenter () -- Normal ert () -- Trror heppened (A reg. (s e.ror code) î 1 i . () Preserved registere () All without A reg. . 0100 нL ; Save registers PUSH 85 9.0 PUSH 0192 0.5 PUSH 8 C C S 5 0127 LD HL, PACERS ; Besat drive packat 21 0280 0200 CALL CD 02A2 COPYA ..... 0203 1 ÷ Seent disk drive system subroutine i : 08 87 0206 SENOSLAVE CD 0268 Send command to slave CALL 0 2 0 7 8 C **P** O P : Restore registers 0204 **c** 1 POP 0 8 0208 n 1 20 P N L 81 0200 C 0200 0.6 A, (DAT) Check return code 3A 7883 1 D 0 9 0208 A Z 0211 87 2 Normal return C 8 0212 0213 SCI 37 887 grear return 0214 6 3 SUBTTL Other subroutines 3 Interrupt consideration 1 ì Disable power-off & elses interrupt 2 () Entry pareneter د ، None  $(\cdot)$ ÷ Rone () Preserved registers () All without A regi 5 DINTPHALME 0215 PUSH R.L. 25 21 2084 0215 LD NL, YPOFDS ; Disable power off interrupt 0216 C8 75 587 6.(HL) HL,YALNDS 0219 LD : Disable alore interrupt 0218 21 2086 5 8 7 8,(8L) C8 76 0212 Disable flag on 38 01 1.0 A.1 0220 (DINTPLC),A LO a z z z POP 81 0225 81 0228 **C** 9 827

:

			Enable	power-off int	erupt
			() Eot	ry parameter	( $)$
			() 841	uto parameter	e 3
				CT-flag ; Be	turn information
				• 1	: Force stop ( Areg. 10 1)
			4) Pre	All without	88 () A req.
0227	82	# 1 M T	PW: PUSN	H L	SAVA registers
0228	05		PUSN	DE	and tryincere
0229	6.5		PUSN	8 C	
022A 0220	21 F084 C8 86		1.D 8.2.3	HL,YPOPDS 6,(HL)	Enable power off
			Check	power off inte	rrupt
0227	21 2035	•	10	NL, YPOPS7	
0232	CB 76		817	6,(HL)	; Check flag
0234	29 06		885	6,(NL) 8,21NT20	; Reset flag : Rower-off oot baccaned
0230		1	• •		t
			Power	off bu using P	\$78)OS toutine
0238	C8 TE		SET	7 (EL) PINTALM	; followe do power off ; followe lore (preprint
0234	CO . #798		CALL	JPSTB105	Da power off th:
			Check	force-stop	
		-			
0240	34 2720	ET NI	10: 10	A. (CSTOPPLG)	: Check force-stop flag
0243	87		O R	A	1
0244	28 06	:	N.C.	2, 21N790	Not force-stop.
0246	CD 0250		CALL	BINTALM	; Eusble slare interrupt
0249	32 01			A, 01	; Force-stop error code
024C	C1-	EINTS	90: <b>202</b>	ac	: Anntore registers
024D	D 1		POP	0 8	
0248	C 9		POP	HL	201
		:	NC.		•
			Eneble	alara interru	
			() Ent	ry parameter	6 b
		Ĩ		None	()
		*		None	
			() Pre	served regiate	F8 ()
0210		E 1 N T		All stendet	
0251	8 S D S		PUSR	R1_	Save registers
0232	C S		POSR	8C	
0293	21 7086				· Proble slave interrupt
0336	CB 86		885	6.(BL)	
036.0		-	Chees	Alare interrup	
0258	21 POBJ CB 76		10	HL, YALNST	
0250	C8 86		811	6.(NL) 5.(NL)	; \$eeet {]A8
0 4 3 4	28 05		- 9 C	Z, EINTA90	; Not happened elarm interrupt
		4 8	Alara	acreen display	
0261	C.8. ##				
0 2 6 3	CD ##96		SET CALL	7,(HL) JPSTBIOS	; Do alere
0266					
0366	AP	EINT	S DEA ROX		; Diesble flag off
4401	32 FBAC		10	(DENTPEC),A	-
026A	C 1		202	ac	: Restore registers
0280	DI		POP	De	
0360	C 9		POP	HL	
			# E F		1

		4	> Ent	CY-flag : Comm	() and type
				= 0 = 1	: Send command & receive d.ta 1 Only receive data
			> Reti	CY-flag : Retu	()
				*0 : =1 :	Normal return Error
		4	> Pres	served registers	(A reg. is error code)
0268		SENDSLAVE		All without A	reg.
0268	DD ES	P	USH	IX	; Save registers
0270	E 5 D 5	P 9	USH	DE	1
0272	C S	P	USH	BC	1
0273	DD 21 0622 38 06	L	R	IX, EPSPRCV C, SENDSSO	Use (receive ACK only routine> !
0279 0278	3E 01 BD 21 001F	L	D	A, 1 IX, EPSPSHD	; Send and receive data
0278		SENDS50:			
0272	21 7418	Ļ	D	HL, DISBNK	; Set bank to OS RON
0284	21 FBAE	L	0	HL. PACKET	Communication buffer address
0257 028A	CD EB69 B7	0	R	A	1
0288	25 07		R	Z, SENDS90	; Normal end
			PSP .	rear	
0280	F5	P	USH	AF A. (DINTFLG)	; Save return code . ; Check in, errupt disable ?
0291	87	0	8	A	
0292	23 06			L, SERDSAU	
0294	CD 0227 CD 0250	c	ALL	EINTALM	: Enable power off interrupt : Enable alarm interrupt
029A		SENDSSO:	OP	AF	
0298	37	5	CF		; Error return
0290	e1	SENDS90:	OP	ac.	: Restore registers
0290	D1	2	OP	DE	
0297	DD E1		OP	1X	
0241	C 9		ET		
				Copy data into	command buffer
			Ent		0
				HL reg. : Sour	rce address
			,	None	
		4	> Pre	BC BC	
0242	e 5	COPYA:	USH	BC	i de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la contra de la
EASO	11 FBAE	-	D	DE, PHT	: Command buffer address : Copy length
0246	ED 30	1	DIR		Copy
024.0	C1 C9	R	8T	ac.	1
			****		
				Subroutine ca.	] • • • • • • • • • • • • • • • • • • •
				ry parameter	0
				None	
			> Bet	None	.,
		•	> Pre	served register: BC.DE.HL	. 0
0240	*5	CALSUB1:	USH	HL.	; Save registers
OZAD	b 5	1	USH	DE	
OZAF	c s		D		· Execute address
0280	2A FBAA	CALSUBI0:	.0	HE, (CALLADI)	, sarress audress
0283	E5 21 0789	P	D	HL.CALSUB20	; Return address
0.2	83	1	x	(SP),HL	; Return addr> (SP)
0287					

2239       CALSUB20: POP DE POP DE C1 BC 2380 C1       Restore registers         02380 C1       SUBTIL COMMENT DATA AREA         COMMUNICATION DATA & VORK COMMUNICATION DATA & VORK COMMUNICATION DATA & VORK         02380 00 10 23 0D       PACKES: DO Slave wend packet data CO PACKES: DE OOH, JLh. 23h, ODH, ODH         02230 00 11 23 0D       PACKES: DE OOH, JLh. 23h, TCh. 00h, 0DH         02230 00 01 23 0D       PACKES: DE OOH, JLh. 23h, TCh. 00h, 0DH         02230 00 01 23 0D       PACKES: DE OOH, JLh. 23h, TCh. 00h, 0DH         0223 00 01 23 0D       PACKES: DE OOH, JLh. 23h, TCh. 00h, 0Lh         0224 00 01 23 0D       PACKES: DE OOH, JLh. 23h, TCh. 00h, 0Lh         0225 00 01 23 7A       DB OOH, JLh. 23h, TAH, 00h, 0Lh         0226 00 01 23 T       PACKECP:       : Copy (0Lh means from D drive to E drive)         0226 00 01 02 03 123 7A       DB OOH, JLh. 23h, TAH, 00h, 0Lh         0226 00 01 02 05 07 78AA       CALSUBIO 588 CALSE         7300 CALSUBI 28 CAUSTIC 280 CAU		0288	E 9		39	(HL)	; Go	subroutine	
0255       C1       POP       BC       Restore registers         0258       E1       POP       BC         0258       E1       POP       BC         0258       E1       POP       BC         0258       E1       POP       BC         0258       E1       SUBTL COMMENT DATA AREA         0250       060       31       23       OD         0261       000       D       PACKES:       : Reset         0262       00       01       23       TO       DE         0263       00       31       23       OD       DE       OOH, 31h, 23h, OD, 00h, 00h, 00h         0263       00       31       23       TO       DE       OOH, 31h, 23h, TOh, 00h, 00h         0263       00       12       TO       DE       OOH, 31h, 23h, TOh, 00h, 01h         0263       00       12       TO       DE       OOH, 31h, 23h, TOH, 00h, 01h         0264       ATLUENIA       CALLADI       END       END         Nacrost:       Frack       F220       CANIN       EDG         741.0       0150       CALVENIA       0150       DEFNO         741.0		0739			CALSURZO:				
0224       01       POP       D2         02280       C9       EI       FOP       RL         02280       C9       EIT       SUBTIL COMMENT DATA AREA         COMMUNICATION DATA & VORK       COMMUNICATION DATA & VORK       COMMUNICATION DATA & VORK         COMMUNICATION DATA & VORK       COMMUNICATION DATA & VORK       PACKES:         COMMUNICATION DATA & VORK       COMMUNICATION DATA & VORK         C2020       OO 12 23         C2021       PACKEP:       : Format (01h means D drive)         C2020       OO 31 23 7A       DB       OOh, 31h, 23h, 7Ah, 00h, 01h         C2030       CALUDIN       DB       OOh, 31h, 23h, 7Ah, 00h, 01h         C2040       O 31 23 7A       DB       OOT         C205       OO 31 23 7A       DB       OOT         C31001       D320       CALLADI       DB       OOT         C320       CALUDI       CALUDI       CAL		0789	C1		POP	80		store resistars	
02380       01       020       00       NL         02380       C9       NL       SUBTL       SUBTL       COMMENT DATA AREA         0280       C9       SUBTL       COMMENT DATA & VORK       () Slave send packet data ()         0220       00 01       D       D       D       D         0221       00 01       D       D       D       D         0223       00 01       D       D       D       O         0223       00 01       D       D       D       O       O         0223       00 01       D       D       O       O       O       O         0223       00 01       D       D       O		0284	D1		POP	DR		atore regreters	
0220C         CG         HIT           SUBTL         SUBTL         COMMENT DATA AREA           COMMUNICATION         DATA & VORK         COMMUNICATION           0220D         00011         DATA & VORK           0220D         00011         PACKES:         : Rest           02201         00001         PACKES:         : Rest           02201         0001         DB         000h,31h,23h,70h,00h,00h           02202         0001         DB         00h,31h,23h,70h,00h,00h           02203         0001         DACKEP:         : Format (01h means D drive)           02204         0001         DB         00h,31h,23h,70h,00h,01h           02205         0001         DB         00h,31h,23h,70h,00h,01h           02204         001         DB         00h,31h,23h,70h,00h,01h           02205         0001         DB         00h,31h,23h,70h,00h,01h           02205         0001         DB         00h,31h,23h,70h,00h,01h           02205         0001         DB         00h,31h,23h,70h,00h,01h           02205         001         DB         0210           02205         001         DB         0210           02205         001         DB <td< td=""><td></td><td>0788</td><td>F1</td><td></td><td>POP</td><td>H L</td><td></td><td></td><td></td></td<>		0788	F1		POP	H L			
CIRC         SUBTTL         COMMENT DATA AREA           SUBTTL         COMMENT DATA AREA           COMMUNICATION         DATA & VORK           COMMUNICATION         DATA           COMMUNICATION         CALLIADI           CALLIADI         ESSG           CALLIADI         ESSG		0280	<b>C</b> 9						
SUBTIL COMMENT DATA AREA           COMMUNICATION DATA & VORK           COMMUNICATION DATA           PACKEP:         COMMUNICATION DATA           PACKEP:         COMUNICATION DATA           COMUNICATION DATA           PACKEP:         COMUNICATION DATA           PACKEP:         COMUNICATION DATA           PACKEP:         COMUNICATION     <							4		
COMMUNICATION DATA & VORK O Slave send packet data O PACKES: : Rest 0000 0					SUBTT	L COMMEN	T DATA AREA		
COMMUNICATION DATA & VORK           0 Slave send packet data 0           0280         00 31 23 0D         PACKES:         : Reset           0201         00 00         PACKES:         : Reset           02023         00 31 23 0D         PACKET:         : Porsat (0in seams D drive)           0203         00 31 23 7C         PACKET:         : Copy (0in seams D drive)           0203         00 31 23 7C         PACKET:         : Copy (0in seams from D drive to E drive)           0205         00 31 23 7A         PACKET:         : Copy (0in seams from D drive to E drive)           0205         00 31 23 7A         PACKET:         : Copy (0in seams from D drive to E drive)           0205         00 31 23 7A         PACKET:         : Copy (0in seams from D drive to E drive)           0205         00 31 23 7A         PACKET:         : Copy (0in seams from D drive to E drive)           0205         00 31 23 7A         PACKET:         : Copy (0in seams from D drive to E drive)           0206         00 11         0203         CALLADI         PACKET:           10000         CALSUBIO         0289         CALSUBIO         CALSUBIO           02000         CR         7220         CSTOPT:         PACKET         PACKET           0140					:				
CONHUNICATION DATA & VORK O Slave send packet data O PACKRS: : : Reset 0280 00 31 23 0D 0201 00 00 0203 00 31 23 7C 0200 00 01 0203 00 31 23 7A 0200 00 01 0209 PACKFT: : Porest (01h means D drive) 0209 00 01 0209 PACKCP: : Copy (01h means from D drive to E drive) 0209 00 01 0209 00 0209 00 01 0209 01 0200 01 0200 01 0200 01 0200 01 0200 01 0200 01 0200 01 0200 00 01 0200 01 0200 00 00 0200 0000 00									
Collary send packet data Collary           0280         00 31 23 0D         PACKES:         : Reset           0200         00 00         D         D         D         D         D         D           0201         00 00         D					COMMUNICATI	ON DATA	A WORK		
01350       00 31 23 0D       PACKET       : Rest         0201       00 00       00, 31h, 23h, 0Dh, 00h, 00h         0202       00 00       00, 31h, 23h, 0Dh, 00h, 00h         0203       00 31 23 TC       DB       00h, 31h, 23h, 7Ch, 00h, 01h         0202       00 01       DE       00h, 31h, 23h, 7Ch, 00h, 01h         0202       00 01       PACKET:       : Copy (01h means D drive to E drive)         0202       00 31 23 7A       DB       00h, 31h, 23h, 7Ah, 00h, 01h         0202       00 01       END       END         Nacros:       Symbols:       rcsc cony (01h means from D drive to E drive)         0200       CONIN       FBAA       CALLADI         FC3C       DA       TEND       CALMANIA         Nacros:       FFACKET       CALLADI       TEG9         CONIN       FBAA       CALLADI       TEG9         CONIN       FBAA       CALLADI       TEG9         TFACK       PACKET:       CALLADI       TEND         Nacros:       FBAA       CALLADI       TEG9         CONIN       FBAA       CALLADI       TEG9       CALLADI         TFAC       PACKET:       CALLADI       TEG9       CALLADI     <									
0130       040       31       23 0D       D       D       00h,31h,23h,0Dh,00h,00h         02C1       00       00       0       00h,31h,23h,0Dh,00h,00h       00h         02C3       00       31       23 TC       D       D       00h,31h,23h,7Ch,00h,01h         02C7       00       01       23 TC       D       D       00h,31h,23h,7Ch,00h,01h         02C9       00       31       23 TA       D       00h,31h,23h,7Ah,00h,01h         02C0       00       01       D       D       00h,31h,23h,7Ah,00h,01h         02C0       00       01       D       D       D       00h,31h,23h,7Ah,00h,01h         02C0       00       01       D       D       D       D       02h         02C0       00       01       D       D       D       D       D       D         02C0       00       D <td></td> <td></td> <td></td> <td></td> <td>() 51</td> <td>ave send</td> <td>packet data ()</td> <td></td> <td></td>					() 51	ave send	packet data ()		
01800       06 31 23 0D       PACKES         02100       00 00       PACKES         02101       00 00       PACKES         02101       00 00       PACKES         02101       00 01       DB       00h,31h,23h,00h,00h,00h         02102       00 01       DB       00h,31h,23h,7Ch,00h,01h         02103       PACKES       DB       00h,31h,23h,7Ch,00h,01h         02103       PACKES       Copy (01h means from D drive to E drive)         02103       00 01       DB       00h,31h,23h,7Ah,00h,01h         02103       00 01       DB       00h,31h,23h,7Ah,00h,01h         02104       DB       00h,31h,23h,7Ah,00h,01h       DB         02105       00 01       DB       00h,31h,23h,7Ah,00h,01h         02105       00 01       DB       DB         02105       00 01       DB       DB         02105       CALSUB10       022B       CALSUB20         02105       CALSUB10       022B       CALSUB20         10105       FRAU       DISTFG       0215         10140       DFRAU       DISTFG       0215         10140       DFRAU       DISTFG       DSFRGSIM         0140		0290			B. CYPS.				
0 00 00         0 00 00         0 00         0 00         0 00         0 00           0 0 00         0 0 00         0 00         0 00,01         0 00,01,00,01,00,01           0 0 00         0 0 01         0 00,01,00,01,00,01,00         0 00,01,00,01,00           0 0 0 01         0 00,01         0 00,01,00,01,00         0 00,01,00           0 0 0 01         0 00,01         0 00,01,00,01,00         0 00,01,00           0 0 0 01         0 00,01         0 00,01,00,01,00         0 00,01,00           0 0 0 01         0 00,01         0 00,01,00,01,00         0 00,01,00           0 0 0 01         0 00,01         0 00,01,00,01,00         0 00,01,00           0 0 0 01         0 0 0,01         0 00,01,00,01,00         0 00,01,00           0 0 0 0 01         0 0 0,01         0 0 0,01         0 0 0,01         0 0 0,01           0 0 0 0 01         0 0 0,01         0 0 0,01         0 0 0,01         0 0 0,01           0 0 0 0 0 0         0 0 0 0         0 0 0,01         0 0 0,01         0 0 0,01         0 0 0,01           0 0 0 0 0         0 0 0 0         0 0 0,01         0 0 0,01         0 0 0,01         0 0 0,01           0 0 0 0 0         0 0 0 0         0 0 0,01         0 0 0,01         0 0 0,01         <		0280			PACARDI	00b 31	h 23h 00h 00h	0.0 h	
0101       0103       PACKFT:       : Format (01h means D drive)         0103       00 01       00 h, 31h, 23h, TCh, 00h, 01h         0103       010       PACKCP:       : Copy (01h means from D drive to E drive)         0103       00 01       DB       00h, 31h, 23h, TCh, 00h, 01h         0104       010       01       DB       00h, 31h, 23h, TCh, 00h, 01h         0105       00 01       DB       00h, 31h, 23h, TAh, 00h, 01h         0105       00 01       DB       00h, 31h, 23h, TAh, 00h, 01h         0105       00 01       DB       00h, 31h, 23h, TAh, 00h, 01h         0105       00 01       DB       00h, 31h, 23h, TAh, 00h, 01h         END         Nacros:         Symbols:         FYBAA         CALLADI         END         Nacros:         Symbols:         FYBAA         CALLADI         END         Symbols:         FYBAA         CALLADI         FYBAA         CALLADI          CALLADI		0201	00 31 23 0	D .					
0203       00 31 23 TC       PACKFT:       : Forest (01h means D drive)         0207       00 01       DB       00h,31h,23h,7Ch,00h,01h         0209       00 31 23 TA       DB       00h,31h,23h,7Ah,00h,01h         0209       00 31 23 TA       DB       00h,31h,23h,7Ah,00h,01h         0209       00 01       END       END         Macrosi       Symbols:       Forest (01h means from D drive to X drive)         7C3C       BCT       PBAA       CALLADI         FC3C       BCT       PBAA       CALLADI         Symbols:       FC3C       BCT       PBAA         7C3C       BCT       PBAA       CALLADI       END         Macrosi       Statusia       0289       CALUS         Symbols:       F1C3C       BCT       PBAA         7C3L       BC       CANUN       EBC         0200       CR       F220       CSTOPF:       CPB3         0210       CR       F220       CSTOPF:       PB3         1211       DISSNK       01F1       D'TKOPFV       014C       DSF20         0143       LSPHEX       0159       DSF:SG       0155       DSFHESSIN         0240       EINTAL			00 00						
D2C3         00 31 23 TC         DB         00h,3lh,23h,TCh,00h,0lh           02C9         01         PACKCP:         : Copy (0lh means from D drive to K drive)           02C9         00 31 23 TA         DB         00h,3lh,23h,7Ah,00h,0lh           02C9         00 31 23 TA         DB         00h,3lh,23h,7Ah,00h,0lh           02C9         00 31 23 TA         DB         00h,3lh,23h,7Ah,00h,0lh           02C0         00 0l         END         END           Macrosi         END         END         02AD           Symbols:         FBAA         CALLADI         EB60           CALSUBI         02B3         CALSUBIO         02B9           CALSUBI         D2B3         CALSUBIO         02B9           CALSUBIO         CNOUT         02A2         COPYA           0000         CR         F12AC         CSTOPF:C         FB3 DAT           FMIF         DIS         FAAU         DINTF'G         0215         D.NTPVALM           0143         LSPMEX         0159         DSF:SG         0165         DSFNSGSIM           024.4         ELST20         0216         FINAD         0221         EINFAD           0143         LSPMEX         0165         DSFNSG		02C3			PACKFT:		; Fo	reat (01h means D drive)	
02C7         00 01           02C3         PACKCP:         ; Copy (01h means from D drive to E drive)           02C9         00 31 23 7A         DB         00h,31h,23h,7Ah,00h,01h           02C0         00 01         END         END           Racros:           Symbols:           FC3C         BOT           PBAC           CALLADI           END           Symbols:           FC3C           SOUT           PBACCALLADI           END           South colspan="2">Calladi           CALSUBI           CALLADI           CALSUBI           CALSUBI           CALSUBI           CALSUBI		02C3	00 31 23 7	с	0.8	00h.31	h.23h.7Ch.00h.	01h	
D2C3         PACKCP:         ; Copy (01h means from D drive to E drive)           02C0         00 01         DB         00h,31h,23h,7Ah,00h,01h           END           Symbols:           FC3C         SOT YBAA         CALLABI         END           Symbols:           FC3C         SOT YBAA         CALLABI         END           Sot YBAA         CALLABI         END           Sot YBAA         CALLABI         END           Sot YBAA         CALLABI           YBADOLS:           FC3C         SOT YBAA           CALLABI         END           SOT YBAA         CALLABI           CALSUBI         Q2B3           CALLABI         END           SOT YBAA           CALLABI           PODTYSO           CALLABI           CALLABI           CALLABI           CALLABI           CALLABI		02C7	00 01						
DICG         PACKCP:         Copy (01h means from D drive to E drive)           DICG         00 01         DB         00h,31h,23h,7Ah,00h,01h           END           END           Macros:           Symbols:           FC3C         BOT           FBAA         CALLADI         EB69           CALSUB1         OZAS           OZAS         CALLADI         EB69           CALSUB1         OZAS           OZAS           CONSIN         EB0C CALLX           CONSIN         EB0C CALLX           OZAS           CALSUB1         OZAS           CONSIN         EB0C CALLX           OZAS           OPYA           OZAS           CASINS           OPYA           ODIT           OZAS           OPYA           OZAS           OZAS           OZAS									
D2C9         00 31 23 7A         DB         00h,3lh,23h,7Ah,00h,0lh           END           Symbols:           FC3C         BOT         FBAA         CALLADI         END           Symbols:           FC3C         BOT         FBAA         CALLADI         END           CALSUB1         CBB         CALSUB20           CONIN         EBCC         CONOT         02A2         COPYA           CONIN         FBAC         CALSUB10         02B5         CALSUB20           CONIN         FBAC         CONOT         02A2         COPYA           OCONIN         FBAC         CINTY-G         02A2         COPYA           OCONIN         FBAC         CINTALH         02A2         FINTHAN           FINTHO         0142         FINTHAN         0159         SPNDUTY            FINTHAN		0209			PACECP:		; Co	py (Olh means from D drive to E drive)	
02C9       00 01 21 23 7A       05       000,310,230,740,000,010         END         END         END         Symbols:         FBAA CALLADI       EB69       CALLX         CALSUBI       0200         CALSUBI       0200         CALSUBI       0200         CALSUBI       0200         CALSUBI       0200         CALSUBI       0200         CALSUBI         0200         CALSUBIO          CALSUBIO <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>									
END EXD EXD FC3C BOT FBAA CALLABI EB69 CALLX GC3AD CALSUBI DEB3 CALSUBIO DEB9 CALSUB20 EB09 C7NIN EB0C CONUT DEA2 COPYA 0000 CR F220 CSTOPFC FBB3 DAT FB1F DID FBAU DINTFLG DEIS DAT FB1F DID FBAU DINTFLG DEIS DAT FALL BISSNK OIFC D'NCOPY DI4C DE720 0143 LSPREX OIS9 DSP.SG OI65 DSPNSGSIM 0240 EINTALH 0227 EINTPV 0022 EFSRCV 071F L'SPS'D 0126 ERROR 0133 FDDUTY 0182 FDBUTY90 FBAE FNT FBB1 FNC 0182 FDBUTY90 FBAE FNT FBB1 FNC 0182 FDBUTY90 FBAE FNT FBB1 FNC 0182 FDBUTY90 BAELFNT FBB1 FNC 0182 FDBUTY90 FBAE FNT FBB1 FNC 0182 FDFMAT 01C9 FORMAT10 DIRE FORMAT90 0198 FDDUTY90 FBAE FNT FBB1 FNC 0198 FDBUTY90 FBAE FNT FBB1 FNC 0198 FDBUTY80 FBAE FNT FBB1 FFB FBAE NESSEN FFB FFF FFFF FFFF FFFFFFFFFFFFFFFFFFF		0209	00 31 23 7.	*	0.8	000,31	n, zan, ran, oon,	UTN	
Symbols:       FC3C     BOT     FBAA     CALLADI     EB69     CALLX       02AD     CALSUBI     02B3     CALSUBIO     02B9     CALSUB20       0200D     CR     F220     CSTOPF:C     FBB3     DAT       FB1F     D1D     FBAL     DINTF:G     0215     D.NTFWALM       0143     LSFMEX     0159     DSF.SG     0165     DSFNSGSIM       024.0     EIX/20     024C     EINTPU     0122     EPSPRCV       0143     LSFMEX     0159     DSF.SG     0165     DSFNSGSIM       024.0     EIX/20     0126     EINTPU     0022     EPSPRCV       011F     L'SPS'D     0126     ERBOR     0133     FDBUTY       0122     PORMAT     0129     FORMATIO     0182     FORMAT90       0179     MSC01     0138     MSC02     0279     PACKERS       E86F     PUTPFK     0170     DISS     MSC01     0183       0179     MSC01     0185     MSC02     0279     PACKERS       E86F     PUTPFK     0170     RESST     0277     SENDS10       0234     SENDS50     0290     SENDS10     0286     SENDS10       0294     SENDS50     0297 <td></td> <td>0100</td> <td>00 01</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>		0100	00 01						
Symbols:         PC3C       BOT       PBAA       CALLADI       EB69       CALLX         02AD       CALSUBI       02B3       CALSUBI0       02B3       CALSUB20         2B09       CALSUBI       02B3       CALSUB20         2B09       CANN       EB0C       CONOUT       02A2       COPYA         000D       CR       F220       CSTOPP.C       FBB3       DAT         Fhir       DI5       FBA4       DISTF.G       015       D.STWALM         0141       LSPEX       0159       DSF.SG       0145       DSPNSGSIM         024u       EIA/20       024C       EINTPW       0022       EPSPNSCSIM         0250       EINTALM       022T       EINTPW       0022       EPSPNSCSIM         0192       FDBUTY90       FBAC       FNT       FBB1       FNT         0117       E'SPS'D       0126       ERCOR       0133       FDDUTY         01182       FORMAT       0109       FORMAT90       FORMAT90         0117       MSG01       0185       MSG02       0209       PACKCP         FBAE       PACKET       02C3       FACKFT       02B0       PACKEP <t< td=""><td></td><td></td><td></td><td></td><td>END</td><td></td><td></td><td></td><td></td></t<>					END				
Macros:         FC3C       BOT       FBAA       CALLADI       EB69       CALLX         O2AD       CALSUBI       02B3       CALSUBI0       02B2       CALSUB20         D00D       CR       F220       CSTOPF'C       FB3       DAT         F11.F       D1D       FBAC       D1F'SCOPF'       O11C       D5F20         O141       LSPHEX       0159       D5F'SC       D10       D5F20         O141       LSPHEX       0159       D5F'SC       O11C       D5F20         O141       LSPHEX       0159       D5F'SC       O126       D5F1010         O159       D5F'SC       0126       FRNCV       O122       EFSPRCV         0132       FORMAT       0126       FRNCP       D126       MSC00         0132       FORMAT       0127       FB51       FNC       MSC01       O138       MSC02       O277       SEND50         0179 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>									
Symbols:       FBAA       CALLADI       EB69       CALLX         02AD       GALSUBI       02B3       CALSUBI0       02B9       CALSUB20         0500       CONIN       EB00       CONUT       02A2       COPYA         0600       CR       F22C       CSTOPFIC       FBB3       DAT         FBAF       DID       FBAU       DINTFIG       0115       DAT         F41_       DISSNK       01F1       D'NCOPY       014C       DSP20         0143       LSPHEX       0159       DSF.SG       0165       DSPMSGSIM         0240       EINTA20       024C       EINT90       0266       EINTA30         0250       EINTAM       0227       EINTPW       0022       EPSPRCV         071F       E'SPS'D       0126       ERROR       0193       FDUTY         0182       FORMATIO       0126       ERROR       0193       FDUTY         0182       FORMATIO       0126       ERROR       0193       FDUTY         0182       FORMATIO       0126       KROR       0193       FDUTY         0182       FORMATIO       0126       MSG02       0209       PACKCF         FBAE </td <td>1</td> <td>seros:</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	1	seros:							
Symbols:       FC3C       BOT       FBAA       CALLADI       EB69       CALLX         G2AD       CALSUBI       02B3       CALSUBI0       02B9       CALSUB20         EB09       C7NIN       EB0C       CONOUT       02A2       COPYA         000D       CR       F220       CSTOPF:       FB3       DAT         FB1F       D1D       FBAU       DINTF'G       0215       D.NTPNALM         F41_       D1SNK       01F1       D'KCOPY       014C       DSP20         0143       LSPMEX       0159       DSF:SG       0165       DSPNSGSIN         024u       E1.720       024C       EINT90       0266       FINTA90         0250       EINTALM       0227       EINTPN       00262       EFSPRCV         019E       FDDUTY90       FBAE       PAT       FBB1       FNC         019E       FDDUTY90       FBAE       PAT       FBB1       FNC         0143       MSG01       0185       MSG02       0269       PACKCP         PF96       JPSTB105       000A       LF       016C       MSG00         0119       MSG01       0185       MSG02       0269       PACKCP     <									
FC3C       B0T       FBAA       CALLADI       EB69       CALLX         02AD       CALSUB1       02B3       CALSUB20       CALSUB20         2B09       CANNN       TB0C       CONOUT       02A2       COPYA         000D       CR       F220       CSTOPP.C       FBB3       DAT         FB1F       D15       FAL       D17F%C       0215       D.NTPWALM         0141       LSPREX       0191       D'3KCOPY       014C       DSP20         0143       LSPREX       0199       DSF.3G       0165       DSPMSGSIM         024u       E1.720       024C       EINT90       0226       EPSPRCV         071F       ESFSTD       0126       ERKOR       0193       FDDUTY         0112       FORMAT       0126       ERKOR       0193       FDDUTY         0112       FORMAT       0126       ERKOR       0193       FDDUTY         0122       FORMAT       0126       ERKOR       0193       FDDUTY         0132       FORMAT       0126       ERKOR       0193       FDUTY         0142       FORMAT       0126       ERKOR       0197       CACKF         FP36		Symbols	:						
02AD       CALSUB1       02B3       CALSUB10       02B3       CALSUB20         2B09       CYNIN       EB0C       CONOUT       02A2       COPYA         000D       CR       F220       CSTOPPTC       FBB3       DAT         FB1F       D10       FBAU       DINTFTG       0215       D.NTFWALM         F41_       D153NK       01F1       D'NCOPY       014C       D5P20         0143       LSPREX       0159       DSP.SG       0165       DSPMSGSIM         024u       E1XFALM       0227       EINTPW       0026       EINTA90         0250       EINTALM       0227       EINTPW       0022       EPBRCV         019E       PDUTY90       FBAE       PMT       FBD1       FNC         0142       FORMAT       01C9       FORMAT10       01EE       FORMAT90         PF96       JPSTB10S       000A       LF       016C       MSG00         0113       MSG01       018S       MSG02       0220       PACKCP         PBAE       PACKFT       02B0       PACKEP       SENDS10       029C       SENDS10         0294       SENDS10       029C       SENDS10       029C       S		FC3C	BOT	7 BAA	CALLADI	EB69	CALLX		
EB09         CYNIN         EB0C         COMOUT         02A2         COPYA           000D         CR         F220         CSTOPF.C         FB33         DAT           FAL         DISNK         01F1         D'NCOPF.C         FB33         DAT           F41         DISNK         01F2         D'NCOPF.C         DAT           0143         LSPHEX         0159         DSF.SG         0165         DSPNSGSIM           024u         EIS.120         024C         EINTAPO         0222         EFSPRCV           0250         EINTALH         0227         EINTPW         0022         EFSPRCV           011F         FYSS'D         0126         ERGR         0193         FDDUTY           011F         FYSPS'D         0126         ERGR         0193         FDDUTY           0112         FORMAT         0126         ERGR         0193         FDDUTY           0112         FORMAT         0126         ENTATIO         DIEE         FORMAT96           01179         MSG01         G155         MSG02         02C9         FACKEF           PBAE         PACKET         02C3         PACKFT         02BD         FACKEF           PBAE <td></td> <td>02.×D</td> <td>CALSUB1</td> <td>0283</td> <td>CALSUBIO</td> <td>0289</td> <td>CALSUB20</td> <td></td> <td></td>		02.×D	CALSUB1	0283	CALSUBIO	0289	CALSUB20		
000D         CR         F220         CSTOPPLC         FB30         DAT           FB1F         DID         FBAU         DINTFLG         0215         D.NTFVALM           P41_         DISSNK         01FL         D'NCOPY         014C         DSP20           0143         LSPMEX         0159         DSFLSG         0165         DSPNSGSIM           024u         EINTALM         0227         EINTPW         0022         EFSPRCV           0r1F         E'SPSTD         0126         ERROR         0193         FDDUTY           019E         FDDUTY90         FBAE         FNT         FBB1         FNC           0142         FORMAT         01C9         FORMAT10         01EE         FORMAT90           0179         MSG01         0188         MSG02         0209         PACKEP           FBAE         PACKET         02C3         PACKFT         02BD         PACKEP           FBAE         PACKET         02C3         PACKFT         02FF         SENDS0           0179         MSG01         0182         SENDSO         02C3         FACKFT         02FF           FBAE         PACKET         02C3         PACKFT         02FF         SENDS		EB09	CONIN	EBOC	COMOUT	0 Z A Z	COPYA		
PALF       DID       PRAC       DINTF'G       D215       D.NTFNALM         P41L       DISSNK       01FC       D'NCOPY       014C       DSP20         014J       LSPHEX       0159       DSP:SG       0165       DSPHSGSIM         024u       EINTAD       0227       EINTPO       0266       EINTADO         0250       EINTALM       0227       EINTPN       0022       EPSPRCV         019E       PDDUTY90       FBAE       FMT       FBB1       FNC         0142       FORMAT       0109       FORMAT10       01EE       FORMAT90         0179       MSG01       0185       MSG02       0270       PACKEF         PBAE       PACKET       0223       PACKEF       SENDS50         0294       SENDSS0       029C       SENDS1AVE       SENDS1AVE         PBB0       SID       FBB2       SIZ       0020       SPACE         PBB0       SID       FBB2       SIZ       0020       SPACE         PBA0       SACK       0100       START       0137       SUB1         PBA0       SACK       0100       START       0137       SUB1         PBA0       TACK		0 0 0 D	CR	1220	CSTOPP	FBB3	DAT		
P41       DISSNK       01PC       D'ACOPY       D14C       DSP20         0143       LSPHEX       0159       DSP.SG       0165       DSPHSGSIM         024u       EISTALM       0227       EINTPO       0266       EINTA90         0250       EINTALM       0227       EINTPV       0022       EPSPRCV         011F       FSPS*D       0126       ERROR       0193       PDUTY         019E       PDUTY90       PBAE       PMT       7BB1       FNC         0142       FORMAT       0109       FORMAT10       01EE       FORMAT90         0142       FORMAT       0109       FORMAT10       01EE       FORMAT90         0179       MSC01       0185       MSG02       0209       PACKCF         PBAE       PACKET       0260       PACKCF       0260         PBAE       PACKET       027F       SENDS50       0294         SENDSS0       029C       SENDS90       026E       SENDSLAVE         PBBO       SID       FBB2       SIZ       0020       SPACE         1000       STACK       0100       STAKT       0137       SUB1         FBAA       VKSTART       01AS		LB7L	DID	FBAC	DINTF'.G	0215	DINTPWALM		
0143         LSPMEX         0159         DSP.3G         0165         DSPRAGEIM           0240         E15720         024C         EINTPO         0266         EINTAPO           0250         EINTALH         0227         EINTPV         0022         EPSPRCV           0r1F         E'SPS'D         0126         ERBOR         0193         PDDUTY           019E         PDDUTY90         FBAE         PMT         7BB1         FNC           0142         FORMAT         0109         FORHAT90         FE         FORHAT90           0179         MSC01         0185         MSG02         0229         PACKEF           7BAE         PACKET         0223         PACKFT         028D         PACKCF           7BAC         SENDSS0         029C         SENDS90         0262         SENDS1AVE           0294         SENDS80         029C         SENDS90         0262         SENDS1AVE           7BAO         SID         FBB2         SIZ         0020         SPACE           1000         STACK         0100         START         0137         SUB1           7BAA         VKSTART         01AS         XXXXXX         F0B6         TALMDS		F41	DICSNK	0171	D-JECOPT	01+C	DSP20		
0240         E15720         024C         E18750         0266         E18780           0250         E1874LM         0227         E1875V         0022         EP\$PRCV           0191         E1875V         0126         ERROR         0193         FDDUTY           0192         FDDUTY90         FBAE         FMT         FBB1         FNC           0142         FORMAT         0109         FORMAT10         0122         FORMAT90           0179         JSSTBIOS         G00A         LF         016C         MSG00           0179         MSG01         0185         MSG02         0209         FACKEF           FBAE         PACKET         02C3         FACKFT         02BD         PACKES           EB6F         PUTPFK         01FD         RESET         02FF         SENDSS0           0294         SENDSS0         029C         SENDS90         026E         SENDS1AVE           FB80         S1D         FB82         S1Z         0020         SPACE           F000         STACK         0100         START         0137         SUB1           F8AA         VKSTART         01AB         XXXXXX         F086         TALMDS		0143	LSPHEX	0159	DSPLSG	0165	DEPRESIA		
0250         EISTALH         0227         EISTPW         0022         EPSPRCV           0°1F         E'SPSTD         0126         ERROR         0193         FDDUTY           019E         FDDUTY90         FBE         FMT         FBB1         FNC           0142         FORMAT         0109         FORMATIO         0122         FORMAT90           0142         FORMAT         0109         FORMAT10         0122         FORMAT90           0179         MSC01         0185         MSC02         0209         PACKCF           788AE         FACET         0203         FACKRS         EB6F         PUTFFK         01FD         RESET         027F         SENDS50           0294         SENDSS0         029C         SENDS90         0268         SENDSLAVE           7880         SID         FB82         SIZ         0020         SPACE           7890         SID         FB82         SIZ         0020         SPACE           7880         SID         FB82         SIZ         0020         SPACE           7880         SID         FB82         SIZ         02165         SUB1           7880         SID         FB82         SI		024u	E17150	0240	EINTSO	0266	EINTAGO		
OPIF         USDST         USDST           019E         PDDUTY90         PBAE         PMT         PBB1         PNC           01A2         FORMAT         01C9         FORMAT10         01EE         PORMAT90           01A2         FORMAT         01C9         FORMAT10         01EE         PORMAT90           0179         MSC01         01S5         MSC02         02C9         PACKCP           PBAE         PACKET         02C3         PACKCP         PACKCP           PBAE         PACKET         02C3         PACKCP           0294         SENDSS0         029C         SENDS90         026E           0294         SENDSS0         029C         SENDS90         026E           1000         STACK         0100         START         0137           1000         STACK         0100         STAKNAX         EB03           FBAA         VKSTART         01A5         XXXXXX         F0B6           FDBT         YALHST         F084         YPOFDS         F0B5		0250	EINTALH	0227	EINTPW	0022	EFSPRCV		
019E         FDDDTT90         FDAE         FMT         FDBT         FAC           01A2         FORMAT         01C9         FORMAT10         01EE         FORMAT90           0179         JPSTBIOS         000A         LP         016C         MSG00           0179         MSG01         0185         MSG02         02C9         PACKEF           PBAE         PACKET         02C3         PACKFT         02BD         PACKES           EB6F         PUTPFK         01FD         RESET         02FF         SENDS50           0294         SENDSS0         029C         SENDS90         026E         SENDSLAVE           FB80         S1D         FB82         S1Z         0020         SPACE           1000         STACK         0100         START         0137         SUB1           F8AA         WKSTART         01AB         XXXXXX         F0B6         TALMDS           F0B7         YALMST         F084         YPOFDS         F0B5         YPOFST		0018	I'SPS!D	0126	ERROR	0193	PDDUTT		
01A2         PORMAT         01C9         PORMATIO         01E2         PORMATIO           PF96         JPSTBIOS         000A         LP         016C         MSG00           0179         MSG01         615S         MSG02         02C9         PACKEP           PBAE         PACKET         02C3         PACKFT         02BD         PACKES           EB6F         PUTPFK         01FD         RESET         02FF         SENDS50           029A         SENDSS0         029C         SENDS90         026E         SENDSLAVE           PBB0         S1D         FB12         S1Z         0020         SPACE           1000         STACK         0100         START         0137         SUB1           FBAD         TRACK         002S         TRKMAX         EB03         WB00T           FBAA         WKSTART         01AS         XXXXXX         F0B6         TALHDS           F0B7         YALHST         F0B4         YPOFDS         F0B5         YPOFST		0196	FDDUTY90	PBAE	PAT	0188	FORMATOR		
P96         JPSTBIOS         000A         LF         0100           0179         MSG01         0155         MSG02         0209         PACKET           PBAE         PACKET         0203         PACKFT         028D         PACKES           EB6F         PUTPFK         01FD         RESET         027F         SENDS50           0294         SENDS80         029C         SENDS90         0268         SENDSLAVE           PBB0         S1D         FBB2         S12         0020         SPACE           1000         STACK         0100         START         0137         SUB1           FBAD         TRACK         0028         TRMAX         EB03         WB00T           FBAA         WESTART         01A8         XXXXX         F0B6         TALHDS           F0B7         TALHST         F084         YP0FDS         F085         YP0FST		0142	FORMAT	0109	FORMATIO	0160	FORMAL SU		
0119         MSC01         0185         MSC02         0265         PACKEF           PBAE         PACKET         02C3         PACKET         02BD         PACKES           EB6F         PUTPFK         01FD         RESET         027F         SENDS50           0294         SENDSS0         029C         SENDS90         026E         SENDSLAVE           FBB0         S1D         FBB2         S1Z         0020         SPACE           1000         STACK         0100         START         0137         SUB1           FBAA         VKSTART         01AS         XXXXXX         F0B6         TALMDS           F0B7         YALMST         F084         YPOFDS         F0B5         YPOFST		1196	JPSTBIOS	0004	LF	0266	BACKCR		
PBAE         PACKET         02C3         PACKET         PACKET         PACKET         <		0179	M5601	0105	BACKET	0280	PACKRS		
EBEF         FOTPTR         OIFD         RESET         OIFF         SENDSO           0294         SENDSSO         029C         SENDS90         0262         SENDSLAVE           PBB0         SID         FB2         SIZ         0020         SPACE           1000         STACK         0100         START         0137         SUB1           FBAD         TRACK         0028         TRKMAX         EB03         WB00T           FBAA         WKSTART         01AB         XXXXXX         F0B6         TALHDS           F0B7         YALHST         F0B4         YP0FDS         F0B5         YP0FST		FBAE	PACAET	0203	BRCER	0277	SENDSSO		
0294         5ENDSSD         0496         5ENDSSD         0496         5ENDSSD         6496		EBEF	PUTPPE	0110	E E N D E B A	0267	SENDSLAVE		
FBRO         SID         FBRO         SID         FBRO         SID         FBRO         SID         FBRO         SID         SID         FFROM		0294	5230560			0070	SPACE		
FBAD TRACK 0028 TRKHAX EB03 WB00T FBAA WKSTART 01AB XXXXXX F0B6 TALHDS F0B7 YALHST F0B4 YPOFDS F0B5 TF0F5T		7880	510	0100	CTART	0137	SUB1		
FBAA WESTART 01AB XXXXXX F0B6 TALHDS F0B7 YALHST F0B4 YPOFDS F0B5 YPOFST		1000	STACE	0028	TEXHAY	8803	WROOT		
FOB7 YALHST FOB4 YPOFDS FOB5 YPOFST		FRAD	UNCTART	0148	XXXXXX	7036	TALHDS		
TODY TALADE TOTAL TOTAL TOTAL		FORT	VALVET	F084	YPOPDS	F085	TPOFST		
			TALAPL						

No Fatal error(s)

